# **DRAFT FINAL**

# REMEDIAL INVESTIGATION/FEASIBILITY STUDY FOR SITE-WIDE GROUNDWATER AT THE FORMER BADGER ARMY AMMUNITION PLANT, BARABOO, WISCONSIN

Prepared for: U.S. Army Environmental Command 2950 Connell Road JBSA – Fort Sam Houston, TX 78232

# **NOVEMBER 2019**

Prepared by: SpecPro Professional Services, LLC S7560 U.S. Highway 12 North Freedom, WI 53951

# **EXECUTIVE SUMMARY**

This Remedial Investigation/Feasibility Study (RI/FS) presents updated groundwater investigation results, human health risk assessment findings, and the analysis of remedial alternatives for contaminated groundwater at the former Badger Army Ammunition Plant (BAAP). The RI/FS is prepared in accordance with the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 (CERCLA) as amended by the 1986 Superfund Amendments and Reauthorization Act (SARA), the National Oil and Hazardous Substances Pollution Contingency Plan (NCP) and the Defense Environmental Restoration Program (DERP) requirements.

BAAP was constructed in 1942 to produce smokeless gunpowder and solid rocket propellant as munitions components for World War II. The former BAAP is located on the Sauk Prairie, between the Baraboo Range and the Wisconsin River. Because of production and waste disposal practices that were common at the time, soil and groundwater at the former BAAP were impacted. The Department of the Army (Army) has transferred a majority of the total 7,275 acres of BAAP to other Federal agencies.

The Army began assessing potential waste management areas that may be sources of soil and groundwater contamination in 1980. When the Army applied for a Resource Conservation and Recovery Act (RCRA) permit in 1988, the State of Wisconsin did not have authorization to implement certain elements of RCRA, also known as the Hazardous and Solid Waste Amendments of 1984, so the Army operated under a dual federal-state permit, where the Wisconsin Department of Natural Resources (WDNR) regulated the RCRA operating and/or closure requirements and the United States Environmental Protection Agency (USEPA) addressed RCRA corrective action requirements.

RCRA closure and post-closure requirements were managed through an *In-Field Conditions Report* (IFCR), which WDNR issued in 1987. As required by the IFCR, the Army has been conducting groundwater monitoring of both monitoring wells and residential wells since 1987. The current site-wide groundwater monitoring program follows the IFCR dated September 4, 2013 and subsequent revisions up through July 24, 2018. Currently, the Army is sampling 166 monitoring wells and 54 residential wells at varying frequencies.

In 2011, the Army submitted a Revised Alternative Feasibility Study, Groundwater Remedial Strategy report to the WDNR. The selected groundwater remedy was Monitored Natural Attenuation (MNA). Due to the relatively long remedial timeframe for the MNA remedy to achieve the proposed cleanup levels, the proposed remedy included construction and operation of a municipal drinking water system that would provide residents in the communities surrounding the former BAAP with drinking water while groundwater contamination continued to diminish over time. During an evaluation by the Army's Office of General Counsel it was determined the Army did not have the legal or funding authority to procure and operate a municipal water system as identified in the 2011 Revised Alternative Feasibility Study.

While a draft Decision Document (DD) for Site-Wide Groundwater was being prepared in 2012, the Army identified several areas where the draft DD did not meet both legal and policy requirements. Specifically, a human health risk assessment was not prepared, incorrect legal standards were identified for the selected groundwater remedy and key components of the proposed response action were outside the Army's authority. In 2017, the Army coordinated with the WDNR and informed the public regarding the need to align the Badger Site-Wide Groundwater remedy selection to comply with legal,

policy, and funding authorities. The Army communicated the need to reevaluate the groundwater remedy at BAAP in a letter dated July 25, 2017.

# **Remedial Investigation**

The Army has conducted numerous site investigations and remedial actions at the former BAAP property. Groundwater investigation activities at BAAP began in 1980. Site-wide groundwater investigations identified four groundwater plumes: Propellant Burning Ground (PBG) Plume, Central Plume, Deterrent Burning Ground (DBG) Plume, and Nitrocellulose Production Area (NC Area) Plume.

The regional groundwater flow direction in the BAAP area is south-southeast towards the Wisconsin River. The Wisconsin River acts as a discharge point for groundwater east and south of BAAP. Based on historical groundwater sampling data, groundwater is contaminated by chlorinated solvents and explosives. While other contaminants of concern were detected, it is unlikely these contaminants are site related.

The Army has replaced seven residential drinking water wells due to groundwater impacts associated with the BAAP groundwater plumes. Three residential wells were impacted by the PBG Plume where volatile organic compounds (VOCs) were detected above WDNR NR 140 Enforcement Standards (ES). Three residential wells were impacted by the Central Plume where total DNT concentrations exceeded the NR 140 ES. The final residential well that was replaced was impacted by the DBG Plume where total DNT concentrations exceeded the NR 140 ES. All seven residential wells withdrew water from the shallow sand and gravel aquifer.

Remedial activities addressing source areas for the four groundwater contaminant plumes have been implemented. Soil remedial actions addressed the source areas to the maximum extent possible and minimized the potential exposure to human health based on anticipated future land use at the former BAAP. The Army has received site closure from the WDNR on all soil related investigations and remedial actions at BAAP.

### **Risk Assessment**

A groundwater human health risk assessment (HHRA) was conducted in 2018. The HHRA evaluated whether groundwater contamination originating from the BAAP poses a current or hypothetical future risk to human health. The HHRA evaluated two potential ways people could be exposed to chemicals in groundwater, through vapor intrusion and domestic use of groundwater. The HHRA is based on conservative screening level risk calculations using maximum groundwater chemical concentrations detected in each groundwater plume. These conservative calculations overestimate the actual risk.

Based on previous vapor intrusion investigations, groundwater contamination at the BAAP does not pose a current or potential future risk to area residents due to vapor intrusion from any of the four groundwater plumes.

The groundwater risk evaluation was conducted to estimate the potential risk associated with the domestic use of groundwater. Groundwater quality data (residential wells and monitoring wells) from 2015, 2016, 2017 and 2018 were used for the initial screening level risk evaluation to represent current

and hypothetical future groundwater quality. The default risk-based screening values provided in the United States Environmental Protection Agency's (USEPA) Regional Screening Levels (RSLs) Resident Tapwater Generic Table (November 2018) were used to calculate both the cancer and non-cancer risks. The Tapwater RSLs incorporate exposure to chemicals in groundwater associated with ingestion (drinking water and food preparation), as well as dermal contact (hand washing and bathing) and inhalation (bathing, food preparation, and dishwashing) during use of the groundwater. When making risk management decisions, the Army considered a cumulative cancer risk above 1x10<sup>-6</sup> (one in a million) for off-site residential wells and groundwater monitoring wells (current risk) and 1x10<sup>-4</sup> (one in ten thousand) for on-site groundwater monitoring wells (hypothetical future) where existing property transfer documents are restricting access to groundwater. For both the off-site and on-site risk evaluations, the RI/FS identifies potential remedies when the cumulative non-cancer risk hazard index exceeds 1.0.

The risk-based COCs identified in the PBG Plume were chloroform, CTET, ethyl ether, TCE, and 2,6-DNT. For the PBG Plume, the risk evaluation identified unacceptable cancer risks and non-cancer hazards associated with current (off-site) groundwater access, as well as hypothetical future (on-site) cancer and non-cancer risk above the risk management criteria.

The risk-based COCs identified in the DBG Plume were chloroform, 1,1,2-TCA, TCE, and total DNT. For the DBG Plume, the risk evaluation identified unacceptable cancer risks and non-cancer hazards associated with current (off-site) groundwater access, as well as hypothetical future (on-site) non-cancer risk above the risk management criteria.

The risk-based COCs identified in the Central Plume were 1,2-dichloroethane, benzene, chloroform, and 2,6-DNT. The risk evaluation indicated that the Central Plume has current (off-site) cancer risk above the risk management criteria.

There were no risk-based COCs identified in the NC Area Plume. The current and future hypothetical cancer risks and non-cancer hazards associated with the NC Area Plume are below the risk management criteria.

The COCs that are identified in the RI have an associated current or hypothetical future risk. The Feasibility Study evaluates potential response actions for the identified risks.

# Feasibility Study

The FS identifies and provides a detailed evaluation of potential remedial alternatives that could reduce, control or mitigate exposure to groundwater COCs. Remedial alternatives considered in the FS must be protective of human health and the environment for the PBG Plume, DBG Plume, and Central Plume and meet Applicable, or Relevant and Appropriate Requirements (ARARs), which are CERCLA threshold criteria for remedy selection. The HHRA did not identify COCs for the NC Area Plume. Therefore, remedial alternatives are not being considered for the NC Area Plume.

The FS includes remedial action objectives (RAOs), which provide a general description of what the cleanup will accomplish, serves as the basis for evaluating each remedial alternative, and provides an understanding of how the unacceptable risks will be addressed by each remedial alternative.

Groundwater RAOs require the remedy to protect human health by preventing exposure to contaminated groundwater, to restore groundwater to the extent practicable, and minimize the impact of the contaminant plumes on the environment. Specifically, the RAOs for any individual plume are achieved when the risk-based groundwater COCs are below cleanup levels. The FS includes the identification and evaluation of general response actions (GRAs), remedial technologies, and process options with respect to effectiveness, implementability, and cost. Appropriate remedial technologies and process options were carried forward and combined to develop remedial alternatives for each individual plume. The remedial alternatives and a brief description are listed below.

# **Remedial Alternatives – PBG Plume**

Based on site conditions and the screening of process options, six remedial alternatives were developed for the PBG Plume to address the presence of contaminants in groundwater at the BAAP.

# Alternative 1: No Action

The No Action Alternative is a mandatory evaluation that provides a baseline to evaluate the other alternatives. It would have no impact on the contaminant plume, require no groundwater monitoring but would include on-site groundwater access restrictions.

# Alternative 2: Monitored Natural Attenuation and Alternate Water Supply

The Monitored Natural Attenuation (MNA) and Alternate Water Supply Alternative would include continued groundwater monitoring, on-site groundwater access restrictions, and a provision for an alternate water supply, where necessary.

### Alternative 3: Active Groundwater Remediation – Pump and Treat

The Active Groundwater Remediation – Pump and Treat Alternative would target areas remediating the impacted groundwater with elevated 2,6-DNT concentrations. It would include groundwater removal through four extraction wells and treatment units located both on-site and off-site. It would also include continued groundwater monitoring, on-site groundwater access restrictions, and a provision for an alternate water supply, where necessary.

### Alternative 4: Active Groundwater Remediation – Anaerobic Bioremediation

The Active Groundwater Remediation – Anaerobic Bioremediation Alternative would target remediating the impacted groundwater with elevated 2,6-DNT concentrations. It would include insitu biochemical treatment utilizing permanent injection wells and temporary injection points to administer the biochemical product into the contaminant plume. The injection locations would be located both on-site and off-site. It would also include continued groundwater monitoring, on-site groundwater access restrictions, and a provision for an alternate water supply, where necessary.

# Alternative 5: Well Replacement – Plume Area

The Well Replacement – Plume Area Alternative would involve replacing shallow aquifer residential wells, meeting replacement criteria, near the PBG Plume with deeper bedrock aquifer wells. It would also include continued groundwater monitoring and on-site groundwater access restrictions.

# Alternative 6: Source Area Treatment

The Source Area Treatment Alternative would target remediating the impacted groundwater with elevated 2,6-DNT concentrations directly downgradient of the source areas. It would include in-situ biochemical treatment utilizing permanent injection wells to administer the biochemical product into the contaminant plume. In addition, the alternative would include continued groundwater monitoring, on-site groundwater access restrictions, and a provision for an alternate water supply, where necessary.

# **Remedial Alternatives – DBG Plume**

Based on site conditions and the screening of process options, six remedial alternatives were developed for the DBG Plume to address the presence of contaminants in groundwater at the BAAP.

### Alternative 1: No Action

The No Action Alternative is a mandatory evaluation that provides a baseline to evaluate the other alternatives. It would have no impact on the contaminant plume, require no groundwater monitoring but would include on-site groundwater access restrictions.

### Alternative 2: Monitored Natural Attenuation and Alternate Water Supply

The MNA and Alternate Water Supply Alternative would include continued groundwater monitoring, on-site groundwater access restrictions, and a provision for an alternate water supply, where necessary.

### Alternative 3: Active Groundwater Remediation – Pump and Treat

The Active Groundwater Remediation – Pump and Treat Alternative would target remediating the impacted groundwater with elevated total DNT concentrations. It would include groundwater removal through three extraction wells and treatment units located both on-site and off-site. The alternative would also include continued groundwater monitoring, on-site groundwater access restrictions, and a provision for an alternate water supply, where necessary.

### Alternative 4: Active Groundwater Remediation – Anaerobic Bioremediation

The Active Groundwater Remediation – Anaerobic Bioremediation Alternative would target remediating the impacted groundwater with elevated total DNT concentrations. It would include insitu biochemical treatment utilizing temporary injection points to administer the biochemical product into the contaminant plume. The temporary injection points would be located both on-site and off-site. It would also include continued groundwater monitoring, on-site groundwater access restrictions, and a provision for an alternate water supply, where necessary.

### Alternative 5: Well Replacement – Plume Area

The Well Replacement – Plume Area Alternative would involve replacing shallow aquifer residential wells, meeting replacement criteria, near the DBG Plume with deeper bedrock aquifer wells. It would also include continued groundwater monitoring and on-site groundwater access restrictions.

### Alternative 6: Source Area Treatment

The Source Area Treatment Alternative would target remediating the impacted groundwater with elevated total DNT concentrations directly downgradient of the source area. It would include in-situ

biochemical treatment utilizing temporary injection points to administer the biochemical product into the contaminant plume. In addition, the alternative would include continued groundwater monitoring, on-site groundwater access restrictions, and a provision for an alternate water supply, where necessary.

# **Remedial Alternatives – Central Plume**

Based on site conditions and the screening of process options, five remedial alternatives were developed for the Central Plume to address the presence of contaminants in groundwater at the BAAP.

### Alternative 1: No Action

The No Action Alternative is a mandatory evaluation that provides a baseline to evaluate the other alternatives. It would have no impact on the contaminant plume, require no groundwater monitoring but would include on-site groundwater access restrictions.

### Alternative 2: Monitored Natural Attenuation and Alternate Water Supply

The MNA and Alternate Water Supply Alternative would include continued groundwater monitoring, on-site groundwater access restrictions, and a provision for an alternate water supply, where necessary.

# Alternative 3: Active Groundwater Remediation – Pump and Treat

The Active Groundwater Remediation – Pump and Treat Alternative would target remediating the impacted groundwater with elevated 2,6-DNT concentrations. It would include groundwater removal through eight extraction wells and treatment units. The alternative would also include continued groundwater monitoring, on-site groundwater access restrictions, and a provision for an alternate water supply, where necessary.

### Alternative 4: Active Groundwater Remediation – Anaerobic Bioremediation

The Active Groundwater Remediation – Anaerobic Bioremediation Alternative would target remediating the impacted groundwater with elevated 2,6-DNT concentrations. It would include insitu biochemical treatment utilizing temporary injection points to administer the biochemical product into the contaminant plume. The temporary injection points would be located both on-site and off-site. It would also include continued groundwater monitoring, on-site groundwater access restrictions, and a provision for an alternate water supply, where necessary.

### Alternative 5: Well Replacement – Plume Area

The Well Replacement – Plume Area Alternative would involve replacing shallow aquifer residential wells, meeting replacement criteria, near the Central Plume with deeper bedrock aquifer wells. It would also include continued groundwater monitoring and on-site groundwater access restrictions.

Each alternative was evaluated based on criteria identified the USEPAs 1994 National Oil and Hazardous Substances Pollution Contingency Plan (NCP) and includes overall protection of human health and the environment, compliance with Applicable or Relevant and Appropriate Requirements (ARARs), long-term effectiveness and permanence, reduction in toxicity, mobility and volume, short-term effectiveness, implementability, and cost.

The Army's preferred alternative or remedy will be presented in the Proposed Plan; the remedy will be based on the results of this RI/FS. The Proposed Plan will briefly summarize the remedial investigation and the remedial alternatives evaluated in this RI/FS, highlighting the key factors that led to identifying the preferred alternative. The Army will submit the Proposed Plan to the regulatory agencies and then the public for review. After this review, the Army will release a Decision Document that documents the selected remedy, certifies that the remedy selection process was carried out in accordance with CERCLA, and addresses public comments on the Proposed Plan.

# **CERTIFICATION PAGE**

### Certification Statement

In accordance with NR 712.09, a registered professional engineer, a hydrogeologist, or a scientist from the State of Wisconsin shall certify this report. The required certification statements are presented, and signed and sealed as follows:

"I, **Joel L. Janssen**, hereby certify that I am a Professional Geologist as that term is defined in s. NR 712.03(1), Wis. Adm. Code, and that, to the best of my knowledge, all of the information contained in this document is correct and the document was prepared in compliance with all applicable requirements in chs. NR 700 to 726, Wis. Adm. Code."

Joel L. Janssen, P.G. License No. 1325 Geologist, SpecPro Professional Services, LLC

<u>11/05/19</u>



"I, **Brian S. Jacobs**, hereby certify that I am a professional engineer in the State of Wisconsin, registered in accordance with the requirements of ch. A-E 4, Wis. Adm. Code; that this document has been prepared in accordance with the Rules of Professional Conduct in ch. A-E 8, Wis. Adm. Code; and that, to the best of my knowledge, all information contained in this document is correct and the document was prepared in compliance with all applicable requirements in chs. NR 700 to 726, Wis. Adm. Code."

Date





# TABLE OF CONTENTS

1.0		INT	RODUCTION	1
2.0		SIT	E BACKGROUND	3
2.1		Site	Description	3
2.	.2	Site	History	4
	2.2.	1	Production and Standby Periods	4
	2.2.	2	Waste Disposal Practices	4
	2.2.	3	Demolition and Restoration	5
2.	3	Env	rironmental Setting	5
	2.3.	1	Topography	5
	2.3.	2	Climate	6
	2.3.	3	Surface Water Hydrology	6
	2.3.	4	Geology	6
	2.3.	5	Hydrogeology	8
3.0		SOU	URCE INVESTIGATIONS AND REMEDIAL MEASURES	10
3.	1	Pro	pellant Burning Ground	10
	3.1.	1	Interim Remedial Measures/Modified Interim Remedial Measures	16
3.	.2	Dete	errent Burning Ground	17
3.	3	Cen	tral Plume Area	19
	3.3.	1	DNT Source Investigation	19
3.	.4	Nitr	cocellulose Production Area	20
	3.4.	1	DNT Source Investigation	20
4.0		GR	OUNDWATER CHARACTERIZATION	22
4.	.1	Gro	oundwater Quality Regulations	22
	4.1.	1	Federal Groundwater Quality Regulations	22
	4.	.1.1.1	National Primary Drinking Water Regulations	22
	4.	.1.1.2	2 National Secondary Drinking Water Regulations	22
	4.	.1.1.3	3 Tapwater Regional Screening Level	22
	4.1.	2	State Groundwater Quality Standards	23
	4.	.1.2.1	Enforcement Standards	23
	4.	.1.2.2	2 Preventive Action Limits	23
4.	.2	Gro	undwater Sampling Program	24

4.3	We	l Identification and Designation	42
4.4	Gro	undwater Properties	42
4	.4.1	Water Level Elevation and Flow Direction	
4	.4.2	Hydraulic Conductivity	43
4	.4.3	Hydraulic Gradient	46
4	.4.4	Groundwater Flow Velocity	50
4.5	Nat	ure and Extent of Groundwater Contamination	50
4	.5.1	Propellant Burning Ground Plume	50
	4.5.1.	Carbon Tetrachloride	60
	4.5.1.	2 Ethyl Ether	61
	4.5.1.	3 Trichloroethene	62
	4.5.1.4	Total Dinitrotoluene	63
	4.5.1.	5 Concentration Graphs	
	4.5.1.	6 Monitored Natural Attenuation	66
4	.5.2	Deterrent Burning Ground Plume	69
	4.5.2.	Total Dinitrotoluene	
	4.5.2.	2 Sulfate	
	4.5.2.	3 1,1,2-Trichloroethane	
	4.5.2.4	Trichloroethene	
	4.5.2.	5 Concentration Graphs	
4	.5.3	Central Plume	76
	4.5.3.	Total Dinitrotoluene	
	4.5.3.	2 Benzene	
	4.5.3.	8 Concentration Graphs	79
4	.5.4	Nitrocellulose Production Area Plume	79
	4.5.4.	Total Dinitrotoluene	79
	4.5.4.	2 Concentration Graphs	80
4.6	Res	idential Well Replacement	80
4	.6.1	Propellant Burning Ground Plume	80
4	.6.2	Deterrent Burning Ground Plume	80
4	.6.3	Central Plume	80
5.0	GR	OUNDWATER HUMAN HEALTH RISK ASSESSMENT	82
5.1	Ris	x Assessment Overview	82

5.2	2 Identification of Exposure Pathways83				
5.3	Scre	eening Level Groundwater Risk Evaluation	83		
5.	3.1	Data Collection and Evaluation	83		
5.	3.2	Chemicals of Potential Concern Selection Process			
5.	3.3	Exposure Assessment and Assumption			
	5.3.3.1	Cancer Risk Characterization	85		
	5.3.3.2	2 Non-Cancer Risk Characterization	85		
	5.3.3.3	Risk Calculations			
	5.3.3.4	Risk Evaluation Results			
5.4	Pro	pellant Burning Ground Plume	86		
5.	4.1	Characterization of Exposure Settings			
5.	4.2	Exposure Quantification – Vapor Intrusion Pathway Analysis	89		
5.	4.3	Exposure Quantification – Groundwater Pathway Analysis			
	5.4.3.1	Current and Potential Future Uses of Groundwater			
5.	4.4	Risk Evaluation Summary			
	5.4.4.1	Hypothetical Future On-Site Groundwater Risks			
	5.4.4.2	2 Current Off-Site Groundwater Risks			
5.5	Dete	errent Burning Ground Plume	92		
5.	5.1	Characterization of Exposure Settings			
5.	5.2	Exposure Quantification – Vapor Intrusion Pathway Analysis			
5.	5.3	Exposure Quantification - Groundwater Pathway Analysis			
	5.5.3.1	Current and Potential Future Uses of Groundwater			
5.	5.4	Risk Assessment Summary			
	5.5.4.1	Hypothetical Future On-Site Groundwater Risks			
	5.5.4.2	2 Current Off-Site Groundwater Risks			
5.6	Cen	tral Plume	97		
5.	6.1	Characterization of Exposure Settings			
5.	6.2	Exposure Quantification – Vapor Intrusion Pathway Analysis	100		
5.	6.3	Exposure Quantification - Groundwater Pathway Analysis	100		
	5.6.3.1	Current and Potential Future Uses of Groundwater	100		
5.	6.4	Risk Assessment Summary	101		
	5.6.4.1	Hypothetical Future On-Site Groundwater Risks	101		
	5.6.4.2	2 Current Off-Site Groundwater Risks	101		

5.7	Niti	rocellulose Production Area Plume	102			
5.7	7.1	Characterization of Exposure Settings	102			
5.7	2.2 Exposure Quantification - Vapor Intrusion Pathway Analysis					
5.7	7.3	Exposure Quantification - Groundwater Pathway Analysis	104			
:	5.7.3.	1 Current and Potential Future Uses of Groundwater	104			
5.7	7.4	Risk Assessment Summary	105			
	5.7.4.	1 Hypothetical Future On-Site Groundwater Risks	105			
	5.7.4.2	2 Current Off-Site Groundwater Risks	105			
6.0	RE	GULATORY REQUIREMENTS	106			
7.0	CO	NTAMINANTS OF CONCERN	109			
7.1	Pro	pellant Burning Ground Plume	109			
7.2	Det	errent Burning Ground Plume	112			
7.3	Cen	ntral Plume	112			
7.4	Niti	Nitrocellulose Production Area Plume1				
8.0	RE	REMEDIAL ALTERNATIVE DEVELOPMENT PROCESS				
8.1 Previous Soil Remedial Activities		vious Soil Remedial Activities	114			
8.2 Previous Groundwater Remedial Activities		vious Groundwater Remedial Activities	114			
8.3 Groundwater Remedial Action Objectives		oundwater Remedial Action Objectives	115			
8.4	Ger	General Response Actions				
8.5	Ide	ntification and Screening of Potentially Applicable Technologies	115			
8.6	Pro	cess Option Screening Criteria	116			
8.7	Eva	luation and Selection of Representative Process Options	116			
8.7	7.1	Land Use Controls	118			
8.7	7.2	Development of New Water Resources	118			
8.7	7.3	Groundwater Treatment	119			
8.7	7.4	Groundwater Containment	123			
8.7	7.5	Summary of Process Options for Groundwater	123			
8.8	Alte	ernatives Analysis Process	123			
9.0	RE	MEDIAL ALTERNATIVES – PBG PLUME	126			
9.1	Alte	ernative 1 – No Action	127			
9.2	Alte	ernative 2 – Monitored Natural Attenuation and Alternate Water Supply	128			
9.3	9.3 Alternative 3 – Active Groundwater Remediation – Pump and Treat		130			
9.4	9.4 Alternative 4 – Active Groundwater Remediation – Anaerobic Bioremediation		133			

9.5	Alternative 5 – Well Replacement – Plume Area137
9.6	Alternative 6 – Source Area Treatment
9.7	PBG Plume Remedial Alternative Summary142
10.0	REMEDIAL ALTERNATIVES – DBG PLUME 144
10.1	Alternative 1 – No Action145
10.2	Alternative 2 – Monitored Natural Attenuation and Alternate Water Supply146
10.3	Alternative 3 – Active Groundwater Remediation – Pump and Treat
10.4	Alternative 4 – Active Groundwater Remediation – Anaerobic Bioremediation151
10.5	Alternative 5 – Well Replacement – Plume Area154
10.6	Alternative 6 – Source Area Treatment156
10.7	DBG Plume Remedial Alternative Summary159
11.0	REMEDIAL ALTERNATIVES – CENTRAL PLUME 162
11.1	Alternative 1 – No Action
11.2	Alternative 2 – Monitored Natural Attenuation and Alternate Water Supply164
11.3	Alternative 3 - Active Groundwater Remediation – Pump and Treat166
11.4	Alternative 4 – Active Groundwater Remediation – Anaerobic Bioremediation169
11.5	Alternative 5 – Well Replacement – Plume Area172
11.6	Central Plume Remedial Alternative Summary174
12.0	REMEDY SELECTION
13.0	REFERENCES

# LIST OF FIGURES

Figure 1	Site Location Map
Figure 2	Geology of Sauk County, Wisconsin
Figure 3	Generalized Geologic Cross Section West-East
Figure 4	Generalized Geologic Cross Section North-South
Figure 5	Bedrock Surface Map
Figure 6	Location and Orientation of Geologic Cross Sections
Figure 7	Geologic Cross Section A-A' – Propellant Burning Ground
Figure 8	Geologic Cross Section A1-A1' – Propellant Burning Ground Source Areas
Figure 9	Geologic Cross Section B-B' – Propellant Burning Ground
Figure 10	Geologic Cross Section C-C' – Propellant Burning Ground
Figure 11	Geologic Cross Section D-D' – Propellant Burning Ground
Figure 12	Geologic Cross Section E-E' – Deterrent Burning Ground
Figure 13	Geologic Cross Section F-F' – Deterrent Burning Ground

Figure 13Geologic Cross Section F-F' – Deterrent Burning GroundFigure 14Geologic Cross Section G-G' – Central Plume

Figure 15	Geologic Cross Section H-H' – Nitrocellulose Production Area
Figure 16	September 2017 Groundwater Contours
Figure 17	September 2017 Groundwater Contours – Propellant Burning Ground
Figure 18	Monitoring Well Locations
Figure 10	Propellant Burning Ground Layout
Figure 20	Posidential Well Locations
Figure 21	Well Sempling Frequency Man
Figure 22	Corbon Tatrashlarida Isosonaantration Man Dranallant Purning Ground
Figure 22	Carbon Tetrachloride Isoconcentration Map – Propenant Burning Ground
Figure 25	Carbon Tetrachloride Isoconcentration Cross Section A-A – Propenant Burning Ground
Figure 24	Carbon Tetrachloride Isoconcentration Cross Section $B-B - Propentiant Burning Ground$
Figure 25	Carbon Tetrachloride Isoconcentration Cross Section C-C – Propellant Burning Ground
Figure 26	Carbon Tetrachloride Isoconcentration Cross Section D-D' – Propellant Burning Ground
Figure 27	Ethyl Ether Isoconcentration Map – Propellant Burning Ground
Figure 28	Ethyl Ether Isoconcentration Cross Section A-A' – Propellant Burning Ground
Figure 29	Ethyl Ether Isoconcentration Cross Section B-B' – Propellant Burning Ground
Figure 30	Ethyl Ether Isoconcentration Cross Section C-C' – Propellant Burning Ground
Figure 31	Ethyl Ether Isoconcentration Cross Section D-D' – Propellant Burning Ground
Figure 32	Trichloroethene Isoconcentration Map – Propellant Burning Ground
Figure 33	Trichloroethene Isoconcentration Cross Section A-A' – Propellant Burning Ground
Figure 34	Trichloroethene Isoconcentration Cross Section B-B' – Propellant Burning Ground
Figure 35	Trichloroethene Isoconcentration Cross Section C-C' – Propellant Burning Ground
Figure 36	Trichloroethene Isoconcentration Cross Section D-D' – Propellant Burning Ground
Figure 37	Total Dinitrotoluene Isoconcentration Map – Propellant Burning Ground
Figure 38	Total Dinitrotoluene Isoconcentration Cross Section A-A' – Propellant Burning Ground
Figure 39	Total Dinitrotoluene Isoconcentration Cross Section A1-A1' – Propellant Burning
	Ground Source Areas
Figure 40	Total Dinitrotoluene Isoconcentration Cross Section B-B' – Propellant Burning Ground
Figure 41	Total Dinitrotoluene Isoconcentration Cross Section C-C' – Propellant Burning Ground
Figure 42	Total Dinitrotoluene Isoconcentration Cross Section D-D' – Propellant Burning Ground
Figure 43	Total Dinitrotoluene Isoconcentration Map – Deterrent Burning Ground
Figure 44	Total Dinitrotoluene Isoconcentration Section E-E' – Deterrent Burning Ground
Figure 45	Total Dinitrotoluene Isoconcentration Section F-F' – Deterrent Burning Ground
Figure 46	Sulfate Isoconcentration Map – Deterrent Burning Ground & Landfill #5
Figure 47	Total Dinitrotoluene Isoconcentration Map – Central Plume
Figure 48	Total Dinitrotoluene Isoconcentration Section G-G' – Central Plume
Figure 49	Total Dinitrotoluene Isoconcentration Map – Nitrocellulose Production Area
Figure 50	Total Dinitrotoluene Isoconcentration Section H-H' – Nitrocellulose Production Area

# LIST OF TABLES

Table 1	Propellant Burning Ground – Volatile Organic Compounds Soil Sample Results (2005)
Table 2	DNT-Impacted Soil Contaminant Mass Estimate – PBG and DBG Source Areas
Table 3	Monitoring Well Construction Information – Sampling Required by WDNR
Table 4	Monitoring Well Construction Information – Sampling Not Required by WDNR
Table 5	Residential Well Construction Information – Sampling Required by WDNR
Table 6	Residential Well Construction Information – Sampling Not Required by WDNR

- Table 7Field Hydraulic Conductivity Test Results
- Table 8Horizontal Groundwater Gradient
- Table 9Vertical Groundwater Gradient
- Table 10Residential Well Groundwater Analytical Results August 2018
- Table 11Groundwater Analytical Results September 2018
- Table 12Groundwater Analytical Results April 2018 (DBG Plume)
- Table 13Groundwater Analytical Results June 2018
- Table 14
   Potential State and Federal Applicable or Relevant and Appropriate Requirements
- Table 15Groundwater Contaminants of Concern
- Table 16Groundwater Cleanup Levels
- Table 17Technology Screening

# LIST OF APPENDICES

- Appendix A Summary of WDNR Conditions of Approval
- Appendix B PBG Waste Pits Soil Investigation Information (2005)
- Appendix C Groundwater Quality Regulations
- Appendix D Groundwater Sampling Schedules (2019-2020)
- Appendix E Plume Concentration Over Time Graphs
- Appendix F Vapor Intrusion Investigation Reports (2012)
- Appendix G Screening Level Groundwater Risk Evaluation (Draft) Exponent
- Appendix H Groundwater Conceptual Site Models Exposure Routes
- Appendix I Remedial Alternative Cost Summaries
- Appendix J Remedial Alternative Treatment Area Drawings

# LIST OF ACRONYMS

1,1,1-TCA	1,1,1-Trichloroethane
1,1,2-TCA	1,1,2-Trichloroethane
2,4-DNT	2,4-Dinitrotoluene
2,6-DNT	2,6-Dinitrotoluene
μg/l	Micrograms per liter
ARAR	Applicable or relevant and appropriate requirement
atm	Atmosphere-meters
Army	Department of the Army
ATSDR	Agency for Toxic Substances and Disease Registry
bgs	Below ground surface
BAAP	Badger Army Ammunition Plant
BEST	Biologically Enhanced Subsurface Treatment
BNA	Base Neutral Acid
BSD	Bluffview Sanitary District
CERCLA	Comprehensive Environmental Response, Compensation, and Liability Act of
	1980, also known as Superfund: Amended in 1986 by the Superfund Amendments
	and Reauthorization Act (SARA).
cm/sec	Centimeters per second
COC	Contaminant of Concern
COPC	Contaminant of Potential Concern

CSM	Conceptual Site Model			
CTET	Carbon Tetrachloride			
DBG	Deterrent Burning Ground			
DD	Decision Document			
DERP	Defense Environmental Restoration Program			
DNT	Dinitrotoluene			
DoD	Department of Defense			
EBS	Enhanced Biodegradation System			
ES	Enforcement Standard			
٥F	Degrees Fahrenheit			
FS	Feasibility Study			
ft/ft	Feet per foot			
GAC	Granular Activated Carbon			
GEMS	Groundwater and Environmental Monitoring System (WDNR)			
gpm	Gallons per minute			
GRA	General Response Action			
HDPE	High-Density Polyethylene			
HHRA	Human Health Risk Assessment			
HI	Hazard Index			
НО	Hazard Ouotient			
HWTTU	Hazardous Waste Thermal Treatment Unit			
IFCR	In-Field Conditions Report			
IRM	Interim Remedial Measures			
MCL	Maximum Contaminant Level			
MCLG	Maximum Contaminant Level Goal			
MIRM	Modified Interim Remedial Measures			
mg/l	Milligrams per liter			
MNA	Monitored Natural Attenuation			
MSL	Mean Sea Level			
OSWER	Office of Solid Waste and Emergency Response			
NC	Nitrocellulose			
NC Area	Nitrocellulose Production Area			
NCP	National Oil and Hazardous Substances Contingency Plan			
NG	Nitroglycerin			
NPS	National Park Service			
NR	Natural Resources			
NPDWR	National Primary Drinking Water Regulations			
NSDWR	National Secondary Drinking Water Regulations			
O&M	Operation and Maintenance			
PAL	Preventive Action Limit			
PBG	Propellant Burning Ground			
PDS	Prairie du Sac			
PP	Proposed Plan			
ppt	Inches of Precipitation			
PSTS	Pilot-Scale Treatability Study			
RA	Remedial Action			

RAO	Remedial Action Objective
RCRA	Resource Conservation and Recovery Act
RME	Reasonable Maximum Exposure
ROD	Record of Decision
RSL	Regional Screening Level
RI	Remedial Investigation
SPS	SpecPro Professional Services, LLC
RI/FS	Remedial Investigation/Feasibility Study
SMCL	Secondary Maximum Contaminant Level
SVE	Soil Vapor Extraction
SVOC	Semi-volatile Organic Compounds
TBC	To be considered
TCE	Trichloroethene or Trichloroethylene
THQ	Total Hazard Quotient
USACE	United States Army Corps of Engineers
USDA	United States Department of Agriculture
USEPA	United States Environmental Protection Agency
VOC	Volatile Organic Compounds
WDHFS	Wisconsin Department of Health and Family Services
WDNR	Wisconsin Department of Natural Resources
WDOT	Wisconsin Department of Transportation
Wis. Adm. Code	Wisconsin Administrative Code
WPDES	Wisconsin Pollutant Discharge Elimination System
WP&L	Wisconsin Power and Light
WWTP	Wastewater Treatment Plant

# **1.0 INTRODUCTION**

This Remedial Investigation/Feasibility Study (RI/FS) was prepared by SpecPro Professional Services, LLC (SPS), for the Department of Army (Army) for investigation and remediation activities at the former Badger Army Ammunition Plant (BAAP) in Sauk County, Wisconsin. This RI/FS presents updated groundwater investigation results and the analysis of remediation alternatives for contaminated groundwater at the BAAP in accordance with the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 (CERCLA).

Environmental cleanup decision-making under CERCLA follows a prescribed sequence: Remedial Investigation (RI), Feasibility Study (FS), Proposed Plan (PP), and Record of Decision (ROD). Under the Defense Environmental Restoration Program (DERP), the Department of Defense (DoD) has conducted investigation and cleanup activities at BAAP. The DoD Manual, DERP Management, dated March 9, 2012 outlines the policies and procedures the Army must follow when conducting environmental restoration.

The RI serves as the mechanism for collecting data to characterize site conditions, determine the nature and extent of the contamination, and assess risks to human health and the environment from this contamination.

This RI/FS was prepared to serve as a principal source for decision-making relating to remediation of groundwater impacts from the BAAP. The report provides a summary of historic and current groundwater investigation and remediation efforts by the Army and describes the development and re-evaluation of groundwater remedial action alternatives for the BAAP.

The Army's preferred alternative or remedy will be presented in the PP; the remedy will be based on the results of this RI/FS. The PP will briefly summarize the remedial investigation and the alternatives evaluated in this RI/FS, highlighting the key factors that led to identifying the preferred alternative. The Army will submit the Proposed Plan to the regulatory agencies and then the public for review. After this review, the Army will release a Decision Document (DD) that documents the selected remedy, certifies that the remedy selection process was carried out in accordance with CERCLA, and addresses public comments on the Proposed Plan. Included within the DD is the Army's ROD.

The In-field Conditions Report (IFCR), issued by the Wisconsin Department of Natural Resources (WDNR) in 1987, and subsequent amendments, calls for groundwater monitoring and reporting at the BAAP. The current site-wide groundwater monitoring program follows the IFCR dated September 4, 2013 and subsequent revisions up through July 24, 2018.

The initial site-wide RI and FS were completed in 1993 and 1994 (ABB-ES, 1993 and 1994). Soil and groundwater remedial alternatives were analyzed, selected, and approved by the Army and state and federal regulators for the PBG and Deterrent Burning Ground (DBG) areas, and their associated groundwater contaminant plumes. In addition to the PBG and DBG areas and their associated plumes, the Central Plume and Nitrocellulose Production Area (NC Area) Plume have since been identified through further groundwater investigations; however, remedial

investigations and actions were previously completed in these areas. These activities were documented and reported to the WDNR. These efforts have met soil remediation action goals and have received regulatory closure. Investigation of groundwater has been ongoing at the BAAP from 1980 to the present. The interim groundwater remedial action for the Propellant Burning Ground (PBG) Plume began in 1990 and continued through 2015. Groundwater monitoring associated with the current sites addressed under Resource Conservation and Recovery Act (RCRA) closure will continue indefinitely (30 years or more) until the WDNR approves case closure.

In December of 2011, the Army completed and submitted to the WDNR, a Revised Alternative Feasibility Study, Groundwater Remedial Strategy report. The selected groundwater remedy was Monitored Natural Attenuation (MNA). Due to the relatively long remedial timeframe for the MNA remedy to achieve the proposed cleanup levels, the proposed remedy included construction and operation of a municipal drinking water system that would provide residents in the communities surrounding the former BAAP with drinking water while groundwater contamination continued to diminish over time. During an evaluation by the Army's Office of General Counsel it was determined the Army did not have the legal or funding authority to procure and operate a municipal water system as identified in the 2011 Revised Alternative Feasibility Study.

While a draft Decision Document (DD) for Site-Wide Groundwater was being prepared in 2012, the Army identified several areas where the draft DD did not meet both legal and policy requirements. Specifically, a human health risk assessment was not prepared, incorrect legal standards were identified for the selected groundwater remedy and key components of the proposed response action were outside the Army's authority.

Since 2012, the Army has monitored groundwater, which included installing new monitoring wells and continued evaluation of the contaminant plumes through groundwater monitoring. In addition, the Army has completed the systematic shutdown of the Interim Remedial Measures (IRM) and Modified Interim Remedial Measures (MIRM) being conducted at the Propellant Burning Ground. A summary of the Army's actions to address the WDNR's conditions of approval is provided in Appendix A.

In 2017, the Army coordinated with the WDNR and the public regarding the need to align with Badger Site-Wide Groundwater remedy selection to comply with legal, policy, and funding authorities. The Army communicated the need to reevaluate the groundwater remedy in a letter dated July 25, 2017. This RI/FS includes the HHRA which is based on groundwater data collected from 2015 through 2018.

# 2.0 SITE BACKGROUND

# 2.1 Site Description

The BAAP, located in south-central Wisconsin within Sumpter and Merrimac Townships in Sauk County, was constructed in 1942 to produce smokeless gunpowder and solid rocket propellant as munitions components for World War II by the Army. BAAP is located on the Sauk Prairie, between the Baraboo Range and the Wisconsin River.

Production of nitric acid, sulfuric acid, oleum (also known as fuming sulfuric acid), nitrocellulose (NC), and nitroglycerin (NG) occurred in support of munitions components production. Production periods were as follows: World War II (1942 to 1945), Korean War (1951 to 1958), and Vietnam Conflict (1966 to 1975). A portion of the BAAP property was transferred post-World War II under the Formerly Used Defense Sites (FUDS) program. BAAP was maintained on stand-by status during the non-production eras and determined to be excess in 1997. Excess hazardous substances were disposed at primarily two locations on-site: the PBG and the DBG. The production and waste disposal practices during operational periods were burning and burial (landfilling), and this impacted the soil and groundwater at the BAAP with multiple contaminants.

After the closure, BAAP land consisted of 7,275 acres that the Army has transferred and divided between the United States Department of Agriculture (USDA), Wisconsin Department of Transportation (WDOT), United States Department of Health Services on behalf of the Bluffview Sanitary District (BSD), Bureau of Indian Affairs (BIA) on behalf of the Ho Chunk Nation and the National Park Service (NPS) on behalf of the WDNR. The property that comprised BAAP is being used as agricultural and grazing land (USDA), Highway 78 (WDOT), recreational land (NPS/WDNR), agricultural and industrial land (Ho Chunk), and a wastewater treatment plant (BSD). The Army still maintains ownership of two cemeteries on the former BAAP.

The primary land use to the north of the BAAP is recreational at Devil's Lake State Park, managed by the WDNR. This area is not impacted by past activities at BAAP as it is hydrologically upgradient. Lake Wisconsin and the Wisconsin River, to the south and southeast of the BAAP, are hydraulically connected to the groundwater beneath the BAAP. Lake Wisconsin was formed in 1914 by the Wisconsin Power and Light (WP&L) dam on the Wisconsin River, near Prairie du Sac (see Figure 1).

Agricultural and residential property is located to the east, south, and west of the BAAP. The agricultural and residential property is in the townships of Merrimac, Prairie du Sac, and Sumpter. The 2016 United States Census estimated the Township of Merrimac population at 1,010 residents, the Township of Prairie du Sac population at 1,132 residents, and the Township of Sumpter population at 1,224 residents.

# 2.2 Site History

# 2.2.1 Production and Standby Periods

During World War II, BAAP employed approximately 7,500 workers and produced approximately 271 million pounds of single- and double-base propellant. Oleum and smokeless powder production began in 1943. Rocket paste powder production began in 1945. The solvent less extrusion smokeless propellant process was installed in 1944 and 1945. A portion of the BAAP property was transferred during 1945 under the FUDS program. From 1945 to 1951, the BAAP was in standby status.

BAAP was reactivated for the Korean War in 1951. Reactivation activities were completed by 1954. Facilities for the manufacture of Ball Powder<sup>®</sup> propellant were constructed during 1954 and 1955. A facility to recycle old cannon powder as a source of NC for the new propellant was also constructed in 1954 and 1955. BAAP remained in production until the Korean War ended and the propellant magazines were full, approximately 1958. During the Korean War, approximately 286 million pounds of single- and double-base propellant were manufactured with a peak production employment of 5,022 employees. The BAAP was in standby status again from 1958 to 1966.

BAAP was reactivated in 1966 for the Vietnam Conflict. The BAAP manufactured Ball Powder<sup>®</sup> propellant, rocket propellant, and smokeless propellant from 1966 to 1975. In 1972, construction included new sewage treatment systems, new acid production, and new nitroglycerin (NG) production facilities. During the Vietnam Conflict, approximately 487 million pounds of single- and double-base propellant were manufactured with a peak production employment of 5,400 employees. The BAAP was placed in standby status in 1975 and was declared excess in 1997, which began the dismantling/demolition process.

# 2.2.2 Waste Disposal Practices

The PBG area has been identified as a source area of groundwater contamination. The PBG Plume source area includes Landfill #1, PBG Waste Pits, 1949 Pit, and the Racetrack Area (see Figure 1). During production periods, the PBG Waste Pits, 1949 Pit, and the Racetrack Area were used for disposal of waste and excess production chemicals, primarily solvents [benzene, carbon tetrachloride (CTET), and trichloroethene (TCE)], and explosives [dinitrotoluene (DNT)]. Excess chemicals and munitions components were placed in open pits and burned to dispose of them. Ash, asphalt, concrete, slag, wood, and other metallic and nonmetallic wastes were disposed of in Landfill #1. Additional information about source area investigations, remedial actions, and regulatory acceptance is provided in Section 3.1.

The DBG area has been identified as a source area of groundwater contamination. The DBG Plume source area includes the DBG Waste Pits, Landfill #3, and Landfill #5 (see Figure 1). During production periods, the DBG Waste Pits were used to dispose of waste and excess production chemicals, primarily explosives (DNT). Excess chemicals and munitions components were placed in open-topped metal tanks and burned to dispose of them. Coal ash, construction rubble, trash, and burned garbage were disposed of in Landfill #3. Coal ash,

demolition debris, laboratory waste, and office waste were disposed of in Landfill #5. Additional information about source area investigations, remedial actions, and regulatory acceptance is provided in Section 3.2.

The Nitroglycerin (NG) Production and Rocket Paste Production areas have been identified as source areas of groundwater contamination for the Central Plume (see Figure 1). Process wastewater was conveyed in open ditches from the north-central to the south side of BAAP where it subsequently flowed to the Settling Ponds and Spoils Disposal Areas, and eventually to the Wisconsin River (see Figure 1). The wastewater may have contained various production chemicals (i.e., DNT, lead, nitrocellulose, and nitroglycerin). Additional information about source area investigations, remedial actions, and regulatory acceptance is provided in Section 3.3.

The Smokeless Powder and Nitrocellulose (NC) Production areas have been identified as source areas of groundwater contamination for the NC Area Plume (see Figure 1). The Smokeless Powder and NC Production areas manufactured single-base propellant across approximately 800 acres of land. DNT was a component of the manufacturing process. Process wastewater (containing production waste) was conveyed through a network of underground piping that lead to an open ditch near the wastewater treatment plant (WWTP), see Figure 1. The process wastewater may have leaked into the soil beneath the piping network or beneath the production buildings. Additional information about source area investigations, remedial actions, and regulatory acceptance is provided in Section 3.4.

Sanitary wastewater was conveyed through a network of underground piping and treated at the on-site WWTP, see Figure 1. Some remote production areas treated sanitary wastewater in small leaching systems.

# 2.2.3 Demolition and Restoration

Environmental investigation and restoration activities began at the BAAP in 1977. Between 2002 and 2012, most of the structures at the BAAP were demolished and placed into the on-site Landfills 3118 and 3646, located on the east-central portion of the BAAP (see Figure 1). Landfills 3118 and 3646 are State of Wisconsin licensed facilities that were permitted to accept asbestos, demolition debris, and contaminated soil. Landfills 3118 and 3646 were closed in 2008 and 2013 in accordance with State of Wisconsin regulatory approval, respectively. Demolition activities included: removal of all process chemicals, equipment, piping, process and storage tanks, munitions and explosives. Many of the concrete slabs that laid underneath these structures have been removed and have either been disposed of or recycled for beneficial reuse.

# 2.3 Environmental Setting

# 2.3.1 Topography

The land surface features at the BAAP is the result of glaciation. The BAAP is located on the southern edge of the Baraboo Range, also commonly referred to as the Baraboo Hills. The terminal moraine, deposited by the leading edge of the last glacier as it moved from east to west,

extends from north to south across the central portion of the BAAP. The topography in the eastern two-thirds of the BAAP consists of gently rolling hills with numerous depressions. The western third of the BAAP is an outwash plain that is nearly level to gently sloping towards the southwest.

# 2.3.2 Climate

The climate in the BAAP area is typically continental with some influence from the Great Lakes system. Average annual temperatures in the region vary from 39 degrees Fahrenheit (°F) to 50°F. The freeze-free season is typically 80 to 180 days. From 1971 to 2000, the Southwest Wisconsin Divisional Climate Summary included the following averages: Winter: 19.7°F, 3.44 inches of precipitation (ppt); Spring: 45.8°F, 9.24 ppt; Summer: 69.2°F, 13.14 ppt; Fall: 48.0°F, 8.10 ppt (Wisconsin State Climatology Office Website, 2010). Precipitation for the area averages approximately 30 inches annually. Typically, 70% of this rainfall occurs during the growing season; April through September. The one year and ten year predicted maximum 24-hour rainfall totals for Sauk County are 2.3 and 4.1 inches, respectively.

# 2.3.3 Surface Water Hydrology

Surface drainage consists of overland flow to the west, south, and east. Much of the run-off collects in isolated depressions on-site and infiltrates or evaporates. The ditches in the northwest portion of the BAAP drain toward the Ballistics Pond and subsequently to the west of the BAAP and Highway 12 (see Figure 1). The surface water from the Nitroglycerin, Rocket Paste, and Magazine Areas, located in the central and southeast areas of the BAAP, discharges to the Settling Ponds in the south-central portion of BAAP (see Figure 1). The Settling Ponds are manmade areas that received wastewater from production but are now almost entirely dry except in severe rain events. The Settling Ponds area drains to the south and east and discharged into Gruber's Grove Bay, on Lake Wisconsin (see Figure 1).

# 2.3.4 Geology

A thick sequence of unconsolidated sediments was deposited during multiple glaciation events. A glacial terminal moraine transects the BAAP from north to south, as shown in Figure 2. Figure 2 is a map depicting the geological features at the surface. This map was adapted from the Geology of Sauk County by Attig and Clayton, 1990. Bisecting BAAP from north to south is the terminal moraine shown in dark green (gj) and classified as thick till of the Johnstown Moraine. Thinner glacial till, shown in light green (gd), is found east of the terminal moraine.

On the far eastern side of BAAP is a unit classified as a collapsed meltwater-stream sediment (sc). West of the terminal moraine is stream sediment (sj) of the Johnstown Moraine, shown in pink. There is also a unit of stream sediment (ss) shown cutting through the terminal moraine in the southern portion of BAAP. This stream sediment unit is younger than the Johnstown sediment, contains ice rafted boulders, and was deposited by floodwater during the drainage of glacial Lake Wisconsin during the Elderon Phase of glaciation.

Based on the borings advanced at BAAP, the glacial till varies in thickness from 10 to 90 feet. Outwash sand and gravel or fluvial deposits (stream sediment) lie beneath the till. The water table does not intersect the till beneath BAAP, only the outwash is in contact with the groundwater. West of the terminal moraine, a thick sequence of glacial outwash sand and gravel was deposited (sj). Glacial tills to the east are primarily silty sands with cobbles and boulders. Several feet of clay and silt (loess) overlie the glacial sediments.

Figures 3 and 4 are generalized geologic cross sections that show thickness of the unconsolidated sediment (sand and gravel) overlying bedrock. These two cross sections were adapted from figures in Hydrogeology and Simulation of Groundwater Flow in Sauk County, Wisconsin (Gotkowitz et al, 2005). The unconsolidated sediment and bedrock unit thicknesses were derived by reviewing boring logs from wells at and near BAAP. Bedrock geology at BAAP is dominated by the Eau Claire Formation (Cambrian age) beneath most of BAAP, with some Precambrian metamorphosed quartzite, granite, and rhyolite. The Eau Claire Formation consists of sandstone/shale/siltstone/dolomite. The Baraboo Range to the north of the BAAP contains Precambrian conglomerate and quartzite, which are part of the Baraboo Syncline, rising approximately 500 feet above the BAAP. The bedrock surface dips steeply toward the south, where soil deposits quickly thicken to a maximum of approximately 250 feet. Along the northern BAAP boundary, soil deposits are thin or absent and bedrock outcrops are common. Figure 5 illustrates the bedrock surface beneath and surrounding BAAP. This bedrock surface map was based on available monitoring well, production well, and residential well construction logs. The bedrock surface drops 200 feet in the northern third of BAAP and flattens out in the southern two-thirds of BAAP.

Figure 3 shows the Bluffview Well #3 on the far left penetrating the entire Eau Claire Formation and entering the Baraboo quartzite. A layer of shale is shown to underlie the western half of BAAP. The shale layer acts as an aquitard, which retards groundwater in the sand and gravel aquifer and the upper sandstone aquifer from moving downward into the lower sandstone aquifer. The Eau Claire Formation is shown to thin out to the east and acts as both an aquitard and an aquifer based on the thickness of the sandstone. The Bluffview Well #3 draws its water from the Eau Claire Formation.

Figure 4 is a cross section that runs from the Baraboo Range south to the Village of Prairie du Sac (PDS). This section also shows the Bluffview Well #3 on the far left and the PDS Well #3 on the far right. The PDS Well #3 penetrates through the Eau Claire Formation and a layer of shale before entering the Mt. Simon Formation (sandstone). The shale layer is shown to be present from just north of the Bluffview Well #3 down to PDS. The shale layer was also found in the Bluffview Well #4 well located at the Bluffview Sanitary District's wastewater treatment plant (WWTP). This shale layer acts as an aquitard, which restricts groundwater from migrating deeper into the Mt. Simon Formation. Based on the well log, the PDS Well #3 has a water depth at the ground surface, whereas the local water table is located 45 feet below ground. This implies that the PDS Well #3 is a flowing or artesian well. The thick sequence of the Eau Claire Formation and the shale layer protect the PDS Well #3 from contaminants on the surface and in the sand and gravel aquifer. Monitoring well PBN-1405F is shown at the BAAP boundary and penetrates through the layer of shale. PBN-1405F was installed in 2014 by the Army to verify that contaminants have not migrated through the shale layer.

Geologic cross sections depicting stratigraphic relationships between the various soil units, bedrock units, and water table are orientated in Figure 6. Figures 7, 8, 9, 10, and 11 are geologic cross sections that are orientated through the PBG area. Figures 12 and 13 are geologic cross sections that are orientated through the DBG area. Figure 14 is a geologic cross section orientated through the Central Plume area. Figure 15 is a geologic cross section orientated through the NC Area Plume area. The terminal moraine is shown in many sections, represented as glacial till (SP-SM or SM-SP), and consists mostly of varying grain sizes of sand with fines and some gravel/cobbles/boulders. Based on the cross sections, the glacial till is not present beneath the water table. Beneath the glacial till lies sand of varying grain sizes that was deposited by glacial fluvial processes (glacial outwash). The sand outwash contains many pockets of gravel with some being localized and others interconnecting. The gravel layers have been encountered up to 40 feet thick. A uniform layer of gravel exists near the bedrock surface, south of the PBG. A layer of clay and silt (CL-ML), up to 30 feet thick, is present in the DBG area. As shown in Figure 12, the fine-grained layer appears to pinch out approximately 1,300 feet east of the DBG. Both Figure 12 and 13 show the fine-grained unit located beneath the water table. The bedrock shown in each cross section consists of the Eau Claire Formation.

# 2.3.5 Hydrogeology

Two major aquifers and one minor aquifer are present beneath the BAAP: the surficial sand and gravel aquifer, the Eau Claire Formation, and the deeper Mt. Simon Formation (sandstone aquifer), respectively. The sand and gravel aquifer and the Eau Claire are un-confined to semiconfined and possibly hydraulically connected. The Eau Claire Formation varies between 80 to 280 feet below ground surface (bgs). The Mt. Simon Formation is located approximately 400 feet bgs and is mostly present to the east and south of BAAP. The general direction of groundwater flow is south to southeast. Steep gradients exist along the northern boundary of the BAAP. The gradient flattens substantially in the central and southern portions of the BAAP. Recharge to the sand and gravel aquifer is limited by infiltration through a fine-grained loess unit (silt/clay) in some areas.

As previously mentioned, Figures 3 and 4 shows that the Eau Claire Formation contains at least one uniform shale layer that acts as an aquitard, which retards groundwater in the sand and gravel aquifer from moving downward into the lower sandstone aquifer (Mt. Simon Formation). The Eau Claire Formation also contains many thinner layers of shale and thick sequences of dolomite that act as an aquitard.

The regional groundwater flow direction in the BAAP area is south-southeast towards the Wisconsin River as depicted in *Water-Table Elevation Map of Sauk County, Wisconsin* (Gotkowitz and Zeiler, 2003) and *Hydrogeology and Simulation of Groundwater Flow in Sauk County, Wisconsin* (Gotkowitz et al, 2005). This direction of flow correlates well with the groundwater contours generated by collecting water levels in the BAAP monitoring wells. Figure 16 depicts the groundwater contours at BAAP during September 2017. Figure 17 depicts the groundwater contours near the PBG during September 2017.

The Wisconsin River acts as a discharge point for groundwater east and south of BAAP. As depicted in *Water Resources of Wisconsin Lower Wisconsin River Basin* (Hindall and Borman, 1974) groundwater on both the west and east sides of the Wisconsin River discharges into the Wisconsin River. The Lake Wisconsin reservoir, caused by the hydroelectric dam on the Wisconsin River, influences groundwater flow across the BAAP. Lake Wisconsin is north of the dam where there is an approximate 40-foot surface water drop. The water level in Lake Wisconsin is elevated above the water table for much of the southeastern portion of the BAAP. Anywhere the elevation in Lake Wisconsin is higher than the water table, the water in Lake Wisconsin will discharge to the groundwater. Subsequently, Lake Wisconsin discharges to the groundwater until it reaches the WP&L dam. The net result is groundwater flow parallel to Lake Wisconsin with discharge to the Wisconsin River south of the dam. Groundwater in the northeast portion of the BAAP is higher in elevation than Lake Wisconsin; therefore, the groundwater discharges to Lake Wisconsin in this area.

# 3.0 SOURCE INVESTIGATIONS AND REMEDIAL MEASURES

The Army has conducted numerous investigations and remedial actions at the former BAAP property. Groundwater investigation activities at BAAP began in 1980 and continue today. The Army proposed and the WDNR approved the site-wide groundwater investigations for RCRA licensed units which included the PBG, DBG, Central Plume, and NC Area Plume areas. Figure 18 shows the locations of the monitoring wells, the four groundwater plumes, and the source areas for each plume.

The RCRA licensed units served as source areas of the four groundwater contaminant plumes being addressed through the Army's CERCLA authority. These remedial activities have been well documented and documentation has been provided to the WDNR. The closure activities conducted by the Army have minimized the potential exposure to contaminated soil at BAAP. The Army has received closure approval from the WDNR on all soil related investigations and response actions at BAAP.

# 3.1 Propellant Burning Ground

The PBG is located in the southwestern portion of the BAAP. The PBG is comprised of the following areas: Waste Pits, 1949 Pit, Racetrack/Hazardous Waste Thermal Treatment Unit (HWTTU) area, and Landfill #1. The location and layout of the PBG is shown in Figure 19.

The PBG Waste Pits consisted of three waste pits (WP-1, WP-2, and WP-3) and an open burning area. The Waste Pits were approximately 40 feet in diameter and 12 to 15 feet deep. The PBG Waste Pits became active sometime between 1942 and 1949 and were last used in 1983. Approximately 2,280 cubic yards of soil were removed from the Waste Pits, from ground surface to approximately 23 feet deep in 1999. The soil was transported off-site and incinerated by a licensed hazardous waste contractor. The PBG Waste Pits were filled with clean soil to grade.

The 1949 Pit was a waste disposal area active between 1949 and 1962 located adjacent to the PBG Waste Pits (see Figure 19). The 1949 Pit contains approximately 58,080 cubic yards of waste, propellant, and construction materials. The area was no longer used, covered, and vegetated by 1968. A clay and geomembrane barrier cap was installed at the 1949 Pit in 1998 to inhibit the movement of contaminants in the soil. The *1949 Pit Phase One Cap, Final Construction Report* (Olin Corporation, 1999) was submitted and approved by the WDNR in 1999.

The Racetrack/HWTTU area consisted of an oval gravel road, three refuse pits, and burning plates, as well as the HWTTU. In 1995, three-fourths of the Racetrack/HWTTU area was closed with a soil cover to prevent contact with residual lead in the soil. The *Final Documentation Report For Soil Cover Construction Racetrack And Thermal Treatment Unit Closure* (Olin Corporation, 1996) was approved by the WDNR. Contaminated soil from the remaining portion of the Racetrack area was excavated and disposed in 1997 and the WDNR letter dated December 2, 1997 indicated that no additional remedial actions were required for this area.

Landfill #1 is a closed demolition debris disposal facility located northeast of the PBG that was used between 1942 and 1959. The area was covered with soil and vegetated by 1974. The facility contains approximately 19,500 cubic yards of ash, slag, asphalt, concrete, wood, and other metallic and nonmetallic wastes. To reduce infiltration, a composite cap including two feet of clay and geomembrane barrier cap was installed and completed in September 1997. The *Landfill #1 Final Cap Construction Report* was submitted to WDNR in January 1998. Regulatory approval of the Landfill #1 cap was received in a Liability, Clarification and Current Environmental Conditions letter report dated August 27, 2014.

DNT and organic solvent-containing materials are known to have been disposed of at the PBG through open burning and burial during production periods. Subsequently, localized impacts to soil consisted of 2,4-DNT, 2,6-DNT, polycyclic aromatic hydrocarbons, benzene, CTET, TCE, arsenic, chromium, lead, selenium, and zinc above soil cleanup remedial action objectives.

A soil vapor extraction (SVE) system operated at the PBG Waste Pits from 1997 to 1999. SVE wells were installed within each of the three waste pits. Approximately 1,600 pounds of solvent-related volatile organic compounds (VOCs) were successfully removed from within the vadose zone. The SVE system was shut down after achieving satisfactory removal of VOCs from the waste pits.

A pilot biotreatment system was installed at Waste Pit 1 in 1999. A Pilot-Scale Treatability Study was conducted in 2000 to evaluate the effectiveness of bacterial degradation of DNT by naturally occurring bacteria in the soil (in-situ). The system extracted groundwater beneath Waste Pit 1, treated the water with phosphate, and reinjected it into the soil column above the waste pit. Oxygen was added to the vadose zone by injecting air through the former SVE system wells, which now served as air injection wells. Carbohydrate (ethanol) injection wells for the control of nitrate byproduct were installed downgradient, but never used. Monitoring results indicated the indigenous bacteria were aerobically biodegrading DNT in the soil column successfully; therefore, the Army decided to go full-scale with the biotreatment system.

The Biologically Enhanced Subsurface Treatment (BEST) system was installed in 2000 to reduce the soil and groundwater contaminants beneath the PBG Waste Pits. The BEST system operated from 2001 to 2005. From 2001 through 2003, additional air injection wells were installed to aid bacterial degradation of DNT in the groundwater. The air injection wells were in operation until 2006. Evaluation of the BEST system indicated effective DNT reduction in soil and groundwater occurred during the operation of the system.

In 2005, Shaw Environmental, Inc. (Shaw) investigated the PBG source area to evaluate the existing soil conditions beneath each PBG Waste Pit (WP-1, WP-2 and WP-3) and evaluate the BEST system performance. Investigation activities are presented in further detail in the *January 2005 Field Activities Technical Memorandum, Propellant Burning Ground* (Shaw Environmental, Inc., 2005). This investigation included drilling borings through each waste pit and collecting soil samples for laboratory analysis. These soil borings and samples were collected at pre-specified intervals corresponding to previous borings, thus allowing for direct comparison to previous concentrations of VOCs and DNTs. Soil sample results for DNT were compared to previous soil samples collected beneath the waste pits during 1991, 1997, 2002, and

2003. Appendix B contains a site map and tables comparing the DNT concentrations from the January 2005 memorandum prepared by Shaw. This investigation determined that DNT concentrations had been reduced and CTET, chloroform, TCE, and other VOCs were no longer present in the soil beneath the PBG Waste Pits. A summary of the VOC soil sample results is provided in Table 1. No additional soil sampling has been conducted.

Based in the soil boring data provided in the *January 2005 Field Activities Technical Memorandum*, the DNT contaminated soil beneath the PBG Waste Pits has vertically migrated into the groundwater. Table 5 in Appendix B provides a comparison of the DNT soil results beneath Waste Pit 2.

In 2006, a draft Alternative Feasibility Study was completed to re-evaluate the interim remedial actions for soils at the PBG and determine the final remedy. The selected remedy included the previous remedial actions: soil vapor extraction, partial soil excavation and incineration, and full-scale bioremediation. The final remedy chosen included removal of the bioremediation system, installation of an impermeable cap, and continued groundwater monitoring and remediation. On March 17, 2008, the WDNR approved the final remedy for the PBG subsurface soil.

Removal of the BEST system was completed in 2008. The PBG Waste Pits were then capped with a clay and geomembrane barrier cap, according to regulatory requirements. The *Construction Documentation Report, PBG Phase 2, Cap and Construction* (SpecPro, Inc., 2009) report was submitted to the WDNR and approved in a letter report dated March 25, 2009.

Based on the 2005 Shaw soil investigation data, the DNT soil contaminant mass was calculated to be 34,810 pounds. This DNT contaminated soil is located beneath the PBG Waste Pit cap. Input parameters and calculations are provided in Table 2.

The Waste Pits, 1949 Pit, Racetrack/HWTTU, and Landfill #1 areas are regularly inspected. Signage, fencing, and vegetation are inspected and maintained. Cap and cover areas are inspected annually for erosion, settlement, undesirable vegetation, and other deficiencies. Annual cap and cover maintenance reports are submitted to the WDNR and United States Environmental Protection Agency (USEPA).

In addition to the annual inspection and in accordance with condition of the final approval, the U.S. Army Corps of Engineers (USACE) completed a five-year review of the PBG in 2013 and 2019. Results of the 2013 review were provided to the WDNR on June 26, 2013. The 2013 five-year review focused on groundwater contaminant concentration trends, cap maintenance activities, and possible modifications to the maintenance and monitoring of the PBG site remedy. These three items were evaluated for the period from 2008 to 2012. The 2013 five-year review concluded that concentration trends for some individual wells were either increasing or probably increasing but the overall plume stability was found to be stable, decreasing, or did not exhibit a trend. The 2013 five-year review also concluded that maintenance records showed that the PBG cap system was being properly maintained and in acceptable condition. Results of the 2019 five-year review are not currently available.

# Table 1 Propellant Burning Ground - Volatile Organic Compounds Soil Sample Results (2005) Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

		Ve	olatile Organic Compour	nds
Sample Number	Sample Interval (ft bgs)	Carbon Tetrachloride	Trichloroethylene	Chloroform
PBB 0501 010	20 - 30	< 0.060	< 0.060	<0.060
PBB 0501 022	21 - 22	< 0.060	< 0.060	<0.060
PBB 0501 026	25 - 26	< 0.060	< 0.060	<0.060
PBB 0501 030	20 - 30	< 0.060	< 0.060	<0.060
PBB 0501 031	30 - 31	< 0.060	< 0.060	<0.060
PBB 0501 040	30 - 40	< 0.060	< 0.060	<0.060
PBB 0501 041	40 - 41	< 0.060	< 0.060	<0.060
PBB 0501 050	40 - 50	< 0.060	< 0.060	<0.060
PBB 0501 051	50 - 51	< 0.060	< 0.060	<0.060
PBB 0501 060	50 - 60	< 0.060	< 0.060	<0.060
PBB 0501 061	60 - 61	< 0.060	< 0.060	<0.060
PBB 0501 070	60 - 70	< 0.060	< 0.060	<0.060
PBB 0501 071	70 - 71	< 0.060	< 0.060	<0.060
PBB 0501 080	70 - 80	< 0.060	< 0.060	<0.060
PBB 0501 080	90 - 91	< 0.060	< 0.060	<0.060
PBB 0501 090	80 - 90	< 0.060	< 0.060	<0.060
PBB 0501 091	90 - 91	< 0.060	< 0.060	<0.060
PBB 0501 100	90 - 100	< 0.060	< 0.060	<0.060
PBB 0502 010	104 - 105	< 0.060	< 0.060	<0.060
PBB 0502 023	22 - 23	< 0.060	< 0.060	<0.060
PBB 0502 029	28 - 29	< 0.060	< 0.060	<0.060
PBB 0502 030	20 - 30	< 0.060	< 0.060	<0.060
PBB 0502 035	34 - 35	< 0.060	< 0.060	<0.060
PBB 0502 040	30 - 40	< 0.060	< 0.060	<0.060
PBB 0502 050	40 - 50	< 0.060	< 0.060	<0.060
PBB 0502 053	52 - 53	< 0.060	< 0.060	<0.060
PBB 0502 060	50 - 60	< 0.060	< 0.060	<0.060

# Table 1 Propellant Burning Ground - Volatile Organic Compounds Soil Sample Results (2005) Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Sample Number	Sample Interval (ft bgs)	Carbon Tetrachloride	Trichloroethylene	Chloroform	
PBB 0502 070	60 - 70	< 0.060	<0.060	<0.060	
PBB 0502 080	70 - 80	< 0.060	<0.060	< 0.060	
PBB 0502 080	80 - 90	< 0.060	<0.060	< 0.060	
PBB 0502 090	80 - 90	<0.060	<0.060	<0.060	
PBB 0502 100	90 - 100	<0.060	<0.060	<0.060	
PBB 0502 105	104 - 105	<0.060	<0.060	<0.060	
PBB 0503 010	60 - 70	<0.060	<0.060	<0.060	
PBB 0503 013	12 - 13	<0.060	<0.060	<0.060	
PBB 0503 020	10 - 20	<0.060	<0.060	<0.060	
PBB 0503 030	20 - 30	<0.060	<0.060	<0.060	
PBB 0503 040	30 - 40	<0.060	<0.060	<0.060	
PBB 0503 050	40 - 50	<0.060	<0.060	<0.060	
PBB 0503 055	54 - 55	<0.060	<0.060	<0.060	
PBB 0503 060	50 - 60	<0.060	<0.060	<0.060	
PBB 0503 070	60 - 70	<0.060	<0.060	<0.060	
PBB 0503 080	70 - 80	<0.060	<0.060	<0.060	
PBB 0503 090	80 - 90	<0.060	<0.060	<0.060	
PBB 0503 100	90 - 100	<0.060	<0.060	<0.060	
PBB 0503 105	100 - 105	<0.060	<0.060	<0.060	

ft bgs - feet below ground surface

All results are expressed in milligrams per kilogram (mg/kg)

Samples collected by Shaw Environmental, Inc., in January 2005

Samples were analyzed by CT Laboratories using method SW8260B

Boring PBB 0501 was drilled beneath Propellant Burning Ground (PBG) Waste Pit 1.

Boring PBB 0502 was drilled beneath PBG Waste Pit 2.

Boring PBB 0503 was drilled beneath PBG Waste Pit 3.

### Table 2

**DNT-Impacted Soil Contaminant Mass Estimate** 

# **Propellant Burning Ground and Deterrent Burning Ground Source Areas**

### **Remedial Investigation/Feasibility Study**

# **Badger Army Ammunition Plant**

Source Location	Average Concentration (mg/Kg)	Interval Depth (ft bgs)	Interval Thickness (ft)	Length (ft)	Width (ft)	Total Volume (ft3)	Mass Volume (lbs)			
Propellant Burning Ground										
Waste Pit #1 (upper zone)	7,776	22 - 31	9	50	40	18,000	17,493			
Waste Pit #1 (lower zone)	45.6	31 - 91	60	50	40	120,000	684			
Waste Pit #2 (upper zone)	3,746	23 - 43	20	40	30	24,000	11,236			
Waste Pit #2 (lower zone)	191	43 - 105	62	40	30	74,400	1,776			
Waste Pit #3 (upper zone)	1,618	13 - 20	7	30	20	4,200	849			
Waste Pit #3 (lower zone)	528	20 - 90	70	30	20	42,000	2,772			
Total DNT Soil Contaminant Mass										
Deterrent Burning Ground										
Waste Pit #1	1,950	10 - 30	20	70	35	49,000	11,942			
Waste Pits #2 and #3	1,050	20 - 50	30	280	40	336,000	44,093			
Total DNT Soil Contaminant Mass										

cm - centimeters

cm3 - cubic centimeters

ft - feet

ft3 - cubic feet

lbs - pounds

mg/Kg - milligrams per kilogram

bgs - below ground surface

Deterrent Burning Ground estimate is based on data from 1991 to 1998.

Propellant Burning Ground estimate is based on data from 2005.

Soil bulk density = 125 lbs/ft3 = 0.002002 Kg/cm3

Mass volume (lbs) = average concentration (mg/kg) x soil bulk density (kg/cm3) x 28,317 (cm3/ft3) x total volume (ft3) x  $(1 \text{ kg}/10^6 \text{ mg}) \text{ x } 2.204586 (lb/kg)$ 

# 3.1.1 Interim Remedial Measures/Modified Interim Remedial Measures

Groundwater contamination in monitoring wells at the PBG was first detected in 1982 (Tsai, 1988). In 1989, the Army evaluated interim remedial measures (IRM). The goals of the early groundwater remedial action were to: 1) curb the advancement of the plume, 2) reduce contaminants within the plume, and 3) be compliant with local, state, and federal regulations.

The IRM groundwater pump and treat system began operations during June 1990 by pumping approximately 350 gallons per minute (gpm). The IRM groundwater treatment system originally consisted of one source control well (SCW-1) and three boundary control wells (BCW-1, BCW-2, and BCW-3) located within the BAAP boundary. The groundwater was treated with liquid phase granular activated carbon (GAC) treatment. BCW-4 was installed in 1993 but was never connected to the IRM. By April 1998, BCW-1, BCW-2, and BCW-3 were shut down and SCW-1 and SCW-2R (installed in 1997) were pumping approximately 310 gpm. Figure 19 shows the locations of the former IRM extraction wells and the former boundary control wells.

Extracted groundwater from the IRM extraction wells was pumped through a GAC system that removed VOCs and DNT from the water by adsorption. The treated water then flowed through a pipeline and discharged into Lake Wisconsin near Gruber's Grove Bay.

An evaluation of the IRM was conducted in 1993 to address new regulatory requirements. This evaluation concluded that the PBG Plume was not being entirely captured by the IRM system. The PBG Plume was extending beneath and east of the three original boundary control wells. A groundwater treatment system was designed to augment the existing IRM system.

This augmented groundwater treatment system called the MIRM system began operations on June 20, 1996. The MIRM groundwater treatment system originally consisted of six boundary extraction wells (EW-161, EW-162, EW-163, EW-164, EW-165, and EW-166) pumping a combined 3,000 gpm. These MIRM extraction wells were located along the southern BAAP boundary (see Figure 19). Four additional extraction wells (EW-167, EW-168, EW-169, and EW-170) were installed along the axis of the plume in 2005 (see Figure 19). The pumping of these extraction wells was refined over the years to optimize removal of groundwater contaminants including the replacement of EW-163 with EW-163R and EW-170 with EW-170R. Until use of the MIRM was discontinued in 2015, the five pumping MIRM extraction wells (EW-163R, EW-167, EW-168, EW-169, and EW-170R) extracted groundwater from the PBG Plume at a combined rate of approximately 2,400 gpm. The water from the MIRM extraction wells flowed through air strippers for treatment of VOCs then passed through a pipeline and discharged into Lake Wisconsin.

Since the PBG Waste Pits were capped in 2008, the DNT concentrations in monitoring wells near the PBG Waste Pits have dropped significantly. The mass of DNT being removed from the groundwater by the IRM system had also reduced dramatically, indicating the IRM system had effectively removed most of the available contaminant mass in the groundwater near the source area. This implied that further operation of the IRM system would not be cost-effective. The Army submitted an *Interim Remedial Measures Shutdown Plan (Badger Technical Services* 

(*BTS, LLC*), *October 2012* that outlined a systematic approach to restoring natural groundwater conditions so that PBG Plume dynamics and attenuation could be evaluated. Based on the WDNR's December 11, 2012 approval letter, the IRM was shut down on December 17, 2012. The Army's June 17, 2014 letter to the WDNR summarized the monitoring activities conducted during 2013 and 2014 and requested that the IRM system be dismantled. The WDNR's August 4, 2014 letter approved the dismantling of the IRM system. During 2014, the IRM extraction wells were abandoned and the IRM treatment building was demolished.

Like the IRM, the MIRM system had reached its limitations of effective contaminant removal and its operation would no longer be cost efficient. To that end, the Army submitted a *Modified Interim Remedial Measures Shutdown Plan (Badger Technical Services, LLC), January 2014* that outlined a systematic approach to restoring natural groundwater conditions so that PBG Plume dynamics and attenuation could be evaluated. Based on the WDNR's August 4, 2014 approval letter, the MIRM was completely shut down on August 31, 2015. The Army's June 27, 2016 letter to the WDNR summarized the monitoring activities conducted between 2014 and 2016 and requested that the MIRM system be dismantled. The WDNR's July 15, 2016 letter approved the dismantling of the MIRM system. During 2016, the MIRM extraction wells were abandoned. Ownership of the MIRM treatment building was transferred from the Army to the BSD in July 2016.

# **3.2** Deterrent Burning Ground

The DBG area consists of seven acres and is located in the northeastern portion of BAAP. The DBG area was used as a sand borrow pit from the 1940s until the early 1960s, and a waste disposal site from the 1940s to the 1970s. The DBG consisted of three burn areas within a manmade depression, approximately three acres in size and 20 feet deep.

Coal ash from the power plant, construction rubble, trash, and burned garbage were deposited in Landfill #3, which is part of the DBG. From 1966 through 1971 the remaining portion of the DBG was used for open burning in open-topped metal tanks of deterrent, a liquid organic extract from surplus propellant, composed mostly of DNT and di-n-butyl phthalate, as well as minor amounts of diphenylamine, benzene, and nitrocellulose. Structural timbers, asphalt shingles, cardboard, paper, and office waste were also burned in the pits. Subsurface soils at the DBG were found to be impacted with DNT, n-nitrosodiphenylamine, arsenic, and chromium. The majority of the impacts were found in the shallowest portion of the pit, with arsenic and chromium in limited areas of the site. Investigations also showed DNT spread vertically in the subsurface soils and reached groundwater.

Landfill #5 is located to the northeast of the DBG. During operations, the landfill reportedly received solid waste, including office waste, demolition debris, laboratory waste, and coal ash from the power plant between 1979 and 1988. No hazardous materials were reported to have been disposed in Landfill #5. In 1988, the landfill was closed with a clay barrier cap which received regulatory approval from the WDNR on September 20, 1989.

An interim corrective action consisting of the removal and off-site incineration of DBG waste pit soil occurred in 1999 and 2000. Impacted soil from the three pits was excavated to a depth of approximately 15 feet. The total volume of the excavated and incinerated soil was

approximately 4,260 cubic yards. Each pit was backfilled with clean fill to pre-excavation grades. This removed the surface soil contaminated with the highest DNT levels and metals.

In 2001, the backfilled area was covered with an interim geomembrane cap, facilitating additional soil and groundwater studies to better understand site conditions. On April 24, 2002, the *Draft Alternative Feasibility Study - Deterrent Burning Ground Waste Pits Subsurface Soil* (Stone & Webster, Inc., 2002) was submitted to request a permit modification to perform the remedial action, including partial excavation and incineration (completed in 2000), geosynthetic clay and geomembrane barrier cap installation, institutional controls, and groundwater monitoring. In accordance with conditions set forth in the WDNR *Final Determination of Remedy for the Deterrent Burning Ground*, dated October 14, 2002, an Enhanced Biodegradation System (EBS) and a geosynthetic clay and geomembrane barrier cap use installed at the DBG during 2003. The cap also encompassed Landfill #3. Due to limited groundwater contamination and the low risk to potential receptors, active groundwater remediation was not required by the WDNR.

The EBS was installed beneath the cap in the area of the three DBG waste pits. The EBS was designed to enhance naturally occurring biodegradation of DNT in subsurface soil by maintaining soil moisture, nutrients and soil gas oxygen beneath the cap. Water and nutrients were introduced into the soil column through a network of piping. The water infiltration rate was kept below the average annual percolation rate.

The Army suspended all operation and monitoring associated with the EBS following the infiltration event in June 2008. This decision was based on the lack of a water resource sufficient to provide the volume needed for continued treatment, problems with the soil moisture and respirometry monitoring equipment and a lack of consistent evidence to show that the EBS was effectively enhancing degradation beyond what was occurring naturally. The WDNR was notified of the EBS discontinuance in a letter report from the Army dated November 17, 2011. This letter provided information on the operation and monitoring of the EBS from 2003 to 2008.

Based on investigation data presented in the *Draft Alternative Feasibility Study - Deterrent Burning Ground Waste Pits Subsurface Soil* (Stone & Webster, Inc., 2002), the DNT soil contaminant mass was calculated to be 56,035 pounds. This DNT contaminated soil is located beneath the engineered cap. Input parameters are provided in Table 2. Concentrations and volume data were used to derive a mass volume in pounds. It should be noted that the soil data used in the calculation was collected from soil borings conducted between 1991 to 1998. No additional soil sampling has been conducted since 1998. Based in the soil boring data, DNT isoconcentration map, and cross sections provided in the *Draft Alternative Feasibility Study*, the DNT contaminated soil beneath the DBG cap is estimated to be 26 feet above the water table. This separation distance implies that the water table (groundwater surface) does not currently intersect with soil contaminated with DNT. Prior to the cap being constructed in 2003, rainwater would have mixed with contaminated soil in the DBG waste pits and vertically infiltrated down towards the groundwater table.

The DBG cap and Landfill #5 areas are regularly inspected. Signage, fencing, and vegetation are inspected and maintained. Cap areas are inspected annually for erosion, settlement, undesirable
vegetation, and other deficiencies. Annual cap and cover maintenance reports are submitted to the WDNR and USEPA.

In accordance with the condition of the final approval, the USACE completed a five-year review of the DBG in 2013 and 2019. Results of the 2013 review were provided to the WDNR on June 26, 2013. The 2013 five-year review focused on groundwater contaminant concentration trends, cap maintenance activities, and new technologies potentially applicable to address remaining impacted soil beneath the DBG cap. The 2013 five-year review concluded that most of the groundwater concentration trends were either stable or decreasing with some individual wells indicating increasing or probably increasing trends. The 2013 five-year review concluded that maintenance records showed that the DBG cap system was being properly maintained and in acceptable condition. No new technologies were identified that were not previously evaluated during the remedy selection or could be implemented without negatively impact to the DBG cap. Results of the 2019 five-year review are not currently available.

### 3.3 Central Plume Area

Based on the knowledge of groundwater flow and monitoring results, the detection of DNT in groundwater during 2004 at the Water's Edge Subdivision indicated another source of DNT groundwater contamination existed besides the PBG and DBG. The Water's Edge Subdivision is located on the north side of Gruber's Grove Bay and at the southern portion of the Central Plume (shown in inset B on Figure 20). Based on the groundwater flow direction and the groundwater contaminant detections, the source of 2,6-DNT contaminated groundwater was believed to be in the north-central portion of BAAP where nitroglycerin, rocket paste, and rocket propellant were produced (see Figure 1). However, several investigations/excavations to date have not determined a specific source of DNT contamination (e.g., landfill or disposal area). It is believed that the broad production area may have caused the groundwater impacts. In 2004, 2,6-DNT was first detected within two residential wells located in the Water's Edge Subdivision. The 2,6-DNT concentration in two residential wells exceeded the Chapter NR 140 Enforcement Standard (ES). In 2005, the Army replaced these two residential wells, WE-RM385 and WE-RR541.

In 2006, the USDA installed a well (USDA 6) in the southeast portion of BAAP to provide water to cows. The USDA 6 well is located approximately 4,300 feet north of the Water's Edge Subdivision (see Figure 20). Sampling results indicated 2,6-DNT exceeded the Chapter NR 140 ES. The following is a summary of the DNT source investigations that were conducted in the Central Plume area.

### 3.3.1 DNT Source Investigation

Groundwater data and historical production standard operating procedures were reviewed. Based on these reviews, the investigation of the source of DNT contamination focused on the Rocket Paste production area (see Figure 1). Containers of production chemicals, which contained DNT, were transported by rail to each Pre-Mix House from the Bag Loading House. Nitrocellulose and nitroglycerin were added to the chemical mixture in each Pre-Mix House. The resulting slurry was then pumped to the Final Mix Houses. The Rocket Paste production area was not connected to the main industrial sewer network, so production related wash waters were discharged to open ditches. The surface water from the Nitroglycerin, Rocket Paste, and Magazine Areas, located in the central and southeast areas of the BAAP, discharges to the Settling Ponds in the south-central portion of BAAP (see Figure 1). The Settling Ponds are manmade areas that received wastewater from production. The Settling Ponds area drains to the south and east and discharged into Gruber's Grove Bay (see Figure 1).

From 2007 to 2010, multiple soil boring investigations were conducted at locations where releases of DNT may have occurred. Soil samples were analyzed for semi-volatiles including DNT. Soil removal activities were conducted around production buildings and along ditches and drainage pathways leading from the Rocket Paste and Nitroglycerin production areas. In addition, sewer removal and adjacent soil excavation was completed in this area. All contaminated soil and sewer piping were disposed of in the on-site licensed Landfill 3646. The WDNR was provided with multiple reports on the investigation and remediation activities. The WDNR provided the Army with multiple case closure letters. Based on these activities, there remains no source of DNT contaminated soil in the Central Plume.

#### 3.4 Nitrocellulose Production Area

Based on the groundwater flow direction and the groundwater contaminant detections, the source of DNT contaminated groundwater is believed to be from the northwestern section of BAAP where nitrocellulose (NC) was manufactured. The completed NC was used to manufacture single-base propellants such as smokeless powder or double-base propellants such as rocket grains or Ball Powder. DNT was added to the manufacturing process in various production buildings. Investigations have determined that there are several potential sources of DNT contamination and it appears that the broad production area may have caused the groundwater impacts.

During 2007, the Army conducted a site-wide investigation into potential sources of DNT contamination in the groundwater. Several monitoring wells, including RIM-0705, were installed within the NC Production Area. Groundwater sampling determined that DNT was present within RIM-0705. This prompted soil investigations into the source of the DNT contamination. The following is a summary of the DNT source investigations that were conducted in the NC Production Area.

### 3.4.1 DNT Source Investigation

The former DNT Screen House (located just north of monitoring well RIM-0705) was used in the production of smokeless powder. Containers of solid DNT were brought to the DNT Screen House and the solid DNT was broken up and screened to remove foreign material. The screened DNT was then distributed to mixing operations within NC Production Area. As part of the daily operating procedures in the DNT Screen House, accumulated residue on the floors was washed into a floor drain, which discharged out to a concrete process sewer sump. During 2008, 2009, and 2010, soil investigations were conducted within and beneath the sump along with the soil surrounding and beneath the DNT Screen House. These investigations determined that DNT contaminated soil was present. Remediation activities during 2008, 2009, and 2010 included the removal of sewer piping along with the surrounding contaminated soil, removal of the concrete

sump along with the surrounding contaminated soil, and the contaminated soil surrounding and beneath the DNT Screen House. All contaminated soil and sewer piping were disposed of in the on-site licensed Landfill 3646.

Nine Hydro-jet Houses (located north of the DNT Screen House) were used during production of smokeless powder. During 2008, 2009, and 2010, soil investigations were conducted beneath the building basement concrete slabs. These investigations determined that DNT contaminated soil was present beneath the concrete slabs. Expansion joints and cracks within the concrete slabs were believed to be migration pathways for the DNT to penetrate beneath the basement slabs. Remediation activities during 2010 included the removal of the concrete slabs and the surrounding contaminated soil. All contaminated soil and concrete were disposed of in the onsite licensed Landfill 3646.

Additional soil investigation and removal activities were conducted around other NC Production Area buildings and the sewer piping network.

The WDNR was provided with multiple reports on the investigation and remediation activities. The WDNR provided the Army with multiple case closure letters. Based on these activities, there remains no source of DNT contaminated soil near the NC Production Area.

### 4.0 GROUNDWATER CHARACTERIZATION

#### 4.1 Groundwater Quality Regulations

Both the USEPA and WDNR have published groundwater quality regulations related to groundwater associated with public drinking water systems and residential wells.

#### 4.1.1 Federal Groundwater Quality Regulations

The regulatory requirements described below, are the most relevant requirements as they relate to groundwater access for domestic purposes. These requirements are considered to be protective of human health.

#### 4.1.1.1 *National Primary Drinking Water Regulations*

Through the Safe Water Drinking Act, the USEPA has established National Primary Drinking Water Regulations (NPDWRs) that set mandatory water quality standard for drinking water contaminants. These are enforceable standards called "maximum contaminant levels" (MCLs) which are established to protect the public against consumption of drinking water contaminants that present a risk to human health. A copy of the NPDWRs (May 2009) is provided in Appendix C.

Maximum Contaminant Level (MCL) - The highest level of a contaminant that is allowed in drinking water. MCLs are set as close to Maximum Contaminant Level Goals (see below) as feasible using the best available treatment technology and taking cost into consideration. For this reason, MCLs are not always risk based values and may be higher than purely risk-based goals or screening criteria. MCLs are enforceable standards for public water systems.

Maximum Contaminant Level Goal (MCLG) - The level of a contaminant in drinking water below which there is no known or expected risk to health. MCLGs allow for a margin of safety and are non-enforceable public health goals.

#### 4.1.1.2 *National Secondary Drinking Water Regulations*

The USEPA has also established National Secondary Drinking Water Regulations (NSDWRs) that set non-mandatory water quality standards for 15 contaminants. These are non-enforceable standards called "secondary maximum contaminant levels" (SMCLs). They are established as guidelines to assist public water systems in managing their drinking water for aesthetic considerations, such as taste, color, and odor. These contaminants are not considered to present a risk to human health at the SMCL. A copy of the NSDWRs (May 2009) is provided in Appendix C.

#### 4.1.1.3 *Tapwater Regional Screening Level*

The USEPA has developed tapwater regional screening levels (RSLs) using risk assessment guidance from CERCLA. The tapwater RSLs are risk-based concentrations derived from

standardized equations combining exposure information assumptions with USEPA toxicity data. The screening levels are considered by the USEPA to be protective for humans (including sensitive groups) over a lifetime. These values are derived solely on the basis of risk and do not consider the cost or feasibility of treating groundwater to these risk-based limits. A copy of the tapwater RSLs (November 2017) is provided in Appendix C.

The tapwater RSLs were developed considering potential exposure to chemicals in groundwater associated with domestic use of the groundwater as a drinking water source, as well as other normal domestic water uses, such as bathing, doing laundry, and washing dishes. Exposure to chemicals in groundwater are incorporated into the tapwater RSLs for both ingestion and dermal contact with the water, as well as inhalation of the portion of the chemicals in groundwater that are volatilized from the water as it is used (e.g., for bathing).

### 4.1.2 State Groundwater Quality Standards

Chapter NR 140 establishes groundwater quality standards referred to as Enforcement Standards (ES) and Preventive Action Limits (PAL) for groundwater beneath the State of Wisconsin. These Chapter NR 140 groundwater quality standards are also used for evaluating groundwater monitoring data. The Chapter NR 140 ESs and PALs are listed within Table 1 - Public Health Groundwater Quality Standards and Table 2 - Public Welfare Groundwater Quality Standards (see Appendix C). The Public Welfare Groundwater Quality Standards listed in Table 2 (e.g., sulfate) are guidelines established to address cosmetic and aesthetic effects of substances present in drinking water supplies (e.g., taste). A copy of the Chapter NR 140 Groundwater Quality standards (February 2017) is provided in Appendix C.

### 4.1.2.1 *Enforcement Standards*

The groundwater NR 140 ESs are protective of public health and welfare on the premise that the groundwater may be ingested through use as drinking water. All NR 140 ESs listed in Table 1 of Chapter NR 140 are Public Health Groundwater Quality Standards. The Chapter NR 140 ES concentrations are equal to or more stringent than the federal MCLs. Further references to groundwater standard exceedances will reference the NR 140 ES.

### 4.1.2.2 *Preventive Action Limits*

The Chapter NR 140 PALs serve "to inform the WDNR of potential groundwater contamination problems (and to) establish the level of groundwater contamination at which the WDNR is required to commence efforts to control the contamination". The Chapter NR 140 PALs are used early in the investigation process given the uncertainty over the nature and extent of contamination. The Chapter NR 140 ESs are used to define contaminants potential of concern and areas warranting remedial action where the current or future groundwater is used for drinking water purposes.

#### 4.2 Groundwater Sampling Program

The Army has been monitoring the nature and extent of groundwater contamination since the early 1980s. Based on the current understanding of the BAAP groundwater plumes, not all monitoring wells are currently being used to define the current plume areas. Figure 18 identifies monitoring well locations which were initially installed to characterize groundwater quality and which wells are being monitored by the Army to define the nature and extent of groundwater contamination. Any area outside the property transferred with groundwater access restrictions may be used for residential use. Figure 20 identifies the residential well locations currently being monitored by the Army. Both Figures 18 and 20 show the boundaries of the four groundwater contamination plumes. Figure 21 displays the current monitoring well and residential well sampling frequencies, groundwater plumes, and groundwater flow directions. The groundwater plumes are displayed in two ways on Figure 21: areas that exceed the Chapter NR 140 PAL and areas that exceed the Chapter NR 140 ES. Groundwater areas exceeding the Chapter NR 140 PALs are provided for informational purposes since this data is not used for remedy selection. These plume boundaries displayed on Figure 21 are approximate and based on DNT and VOC groundwater data collected during 2018. The DBG Plume boundaries do not include sulfate groundwater data. The sulfate data will be discussed in Section 4.5.2.2 of this report.

The current groundwater sampling program including monitoring wells and residential wells is being conducted according to sampling plans agreed upon by the Army and WDNR. Sampling plans are routinely modified based on requests from the WDNR.

A total of 166 monitoring wells are sampled at varying frequencies: 5 quarterly (four times per year), 119 semi-annual (twice per year), 35 annual (once per year), and 7 biennial (once every two years); see Figure 21. Table 3 provides the location, well construction information, and sample frequency for the 166 monitoring wells currently being sampled by the Army. Appendix D details the groundwater sampling program. Table 4 provides the location and well construction information for the 137 monitoring wells that are not currently being sampled. There are currently 303 monitoring wells associated with BAAP (see Figure 18).

A total of 54 residential wells are sampled at varying frequencies: 2 quarterly (four times per year) and 52 annually; see Figure 21. Table 5 provides the well construction information and sample frequency for the 54 residential wells currently being sampled by the Army. Table 6 provides the well construction information for the residential wells that are not currently being sampled and shown on Figure 20. Well construction and depth information was not available for many residential wells due to the lack of information provided on well logs. Information regarding the construction and depths of residential wells near BAAP in 1993 was included in the *Final Remedial Investigation Report* (ABB Environmental Services, Inc., 1993).

Concurrent with this RI/FS report preparation, yet independent of this effort, the United Stated Geological Survey (USGS) is performing a comprehensive review of the BAAP groundwater monitoring program. The intention of the review is to evaluate the existing program and determine if modifications could be made to strengthen the value of the data generated from the monitoring effort. No modifications are being proposed, at this time, to the previously approved

monitoring program; however, results of the USGS evaluation may result in suggested modifications to enhance the program.

### Table 3 Monitoring Well Construction Information – Sampling Required by WDNR Remedial Investigation/Feasibility Study

**Badger Army Ammunition Plant** 

Well Name	Well ID	License	Date Installed	NAD83 Northing (feet)	NAD83 Easting (feet)	Well Depth (feet)	Top of Casing Elevation	Ground Elevation	Well Diameter (inches)	Screen Length (feet)	Bedrock Depth (feet)	Well Type	Aquifer	Screen Level	Sample Frequency	Plume Area
S1111	751	3038	1/2/80	487,414	2,044,310	99.0	848.79	846.80	4.0	20.3	n/a	OW	Sand	A	Annual	Central
NLN-8203A	258	3118	5/5/82	494,954	2,045,545	115.5	884.12	881.80	4.0	10.0	n/a	OW	Sand	А	Annual	Central
NLN-8203B	259	3118	5/6/82	494,946	2,045,534	127.5	884.87	882.70	4.0	2.0	n/a	PZ	Sand	В	Annual	Central
NLN-8203C	260	3118	5/5/82	494,954	2,045,532	138.5	885.17	882.70	4.0	2.0	n/a	PZ	Sand	С	Annual	Central
NPM-8901	506	3487	10/25/89	497,388	2,041,526	100.0	862.92	861.50	4.0	20.0	n/a	OW	Sand	Α	Annual	Central
RIM-1003	491	3487	5/3/10	492,555	2,043,661	114.3	885.06	882.78	2.5	15.0	n/a	OW	Sand	A	Annual	Central
RIM-1004	494	3487	5/5/10	489,552	2,044,244	70.5	836.40	833.60	2.5	15.0	n/a	OW	Sand	Α	Annual	Central
RIN-0701C	443	3487	10/12/07	497,385	2,041,541	180.0	863.86	860.76	2.5	5.0	n/a	PZ	Sand	С	Annual	Central
RIN-0702C	444	3487	10/16/07	494,729	2,042,699	201.0	887.98	885.81	2.5	5.0	n/a	PZ	Sand	С	Annual	Central
RIN-0703C	445	3487	10/17/07	489,062	2,044,835	207.0	857.55	854.83	2.5	5.0	n/a	PZ	Sand	С	Annual	Central
RIN-1002A	492	3487	5/4/10	492,556	2,046,082	92.2	862.81	860.46	2.5	15.0	n/a	OW	Sand	Α	Annual	Central
RIN-1002C	493	3487	6/1/10	492,569	2,046,079	179.8	862.95	860.86	2.5	5.0	n/a	PZ	Sand	С	Annual	Central
RIN-1003A	495	3487	5/5/10	489,061	2,044,797	90.5	857.10	854.66	2.5	15.0	n/a	OW	Sand	Α	Annual	Central
RIN-1004B	498	3487	5/13/10	486,645	2,044,721	146.7	859.31	856.74	2.5	5.0	n/a	PZ	Sand	В	Semi-Annual	Central
RIN-1005A	496	3487	5/17/10	489,311	2,045,864	60.5	828.61	826.74	2.5	15.0	n/a	OW	Sand	А	Annual	Central
RIN-1005C	497	3487	5/17/10	489,317	2,045,865	147.0	828.75	826.49	2.5	5.0	n/a	PZ	Sand	С	Annual	Central
RIN-1501B	538	3487	10/23/15	492,538	2,046,945	123.5	845.87	842.86	2.5	10.0	n/a	PZ	Sand	В	Annual	Central
RIN-1501C	539	3487	10/27/15	492,538	2,046,939	165.2	845.86	842.80	2.5	5.0	n/a	PZ	Sand	С	Annual	Central
RIN-1501D	540	3487	10/30/15	492,578	2,046,076	237.8	863.54	860.86	2.5	5.0	n/a	PZ	Sand	D	Annual	Central
RIN-1502B	541	3487	9/22/15	489,765	2,046,626	103.4	824.29	821.41	2.5	5.0	n/a	PZ	Sand	В	Annual	Central
RIN-1502C	542	3487	9/25/15	489,768	2,046,631	143.1	824.40	821.44	2.5	5.0	n/a	PZ	Sand	С	Annual	Central
RIN-1502D	543	3487	10/2/15	489,772	2,046,636	213.3	824.33	821.35	2.5	5.0	213	PZ	Sand	D	Annual	Central
RPM-8901	507	3487	10/16/89	494,718	2,042,698	124.3	888.62	886.20	4.0	19.5	n/a	OW	Sand	Α	Annual	Central
NLN-1001A	331	3646	4/21/10	495,613	2,044,708	111.5	882.62	880.28	4.0	15.0	n/a	OW	Sand	Α	Annual	Central
NLN-1001C	332	3646	4/19/10	495,615	2,044,701	154.5	882.52	880.36	4.0	5.0	n/a	PZ	Sand	С	Annual	Central
SEN-0501A	580	4330	1/27/05	484,159	2,043,454	32.0	784.56	784.64	3.8	15.0	n/a	OW	Sand	Α	Semi-Annual	Central
SEN-0501B	581	4330	1/27/05	484,158	2,043,458	87.0	784.71	784.87	3.8	10.0	n/a	PZ	Sand	В	Semi-Annual	Central
SEN-0501D	582	4330	1/27/05	484,156	2,043,462	190.0	784.98	785.22	3.8	10.0	194	PZ	Sand	D	Semi-Annual	Central
SEN-0502A	583	4330	1/28/05	484,107	2,044,412	33.0	786.46	786.47	3.8	15.0	n/a	OW	Sand	Α	Semi-Annual	Central
SEN-0502D	584	4330	1/12/05	484,103	2,044,417	187.0	786.24	786.76	3.8	10.0	190	PZ	Sand	D	Semi-Annual	Central
SEN-0503A	585	4330	1/26/05	484,524	2,044,148	55.5	809.56	809.63	3.8	15.0	n/a	OW	Sand	Α	Semi-Annual	Central
SEN-0503B	586	4330	1/25/05	484,518	2,044,150	110.0	809.17	809.39	3.8	10.0	n/a	PZ	Sand	В	Semi-Annual	Central
SEN-0503D	587	4330	1/19/05	484,514	2,044,152	213.0	809.31	809.31	3.8	10.0	214	PZ	Sand	D	Semi-Annual	Central
ELM-8901	216	2813	1/18/89	501,113	2,043,592	165.0	922.57	920.50	4.0	19.5	n/a	OW	Sand	Α	Semi-Annual	DBG
ELM-8907	220	2813	4/18/89	500,500	2,044,492	150.3	916.21	913.70	4.0	20.0	n/a	OW	Sand	Α	Semi-Annual	DBG
ELM-8908	221	2813	4/1/89	500,503	2,044,033	145.0	906.05	903.00	4.0	20.0	n/a	OW	Sand	А	Semi-Annual	DBG
ELM-8909	222	2813	4/13/89	501,298	2,043,256	155.0	921.86	919.60	4.0	20.0	n/a	OW	Sand	Α	Semi-Annual	DBG
ELM-9501	234	2813	6/27/95	498,219	2,046,902	69.0	843.28	840.70	4.0	15.0	n/a	OW	Sand	A	Semi-Annual	DBG
ELN-0801B	455	2813	4/15/08	498,220	2,046,894	105.0	843.87	841.37	2.5	5.0	n/a	PZ	Sand	В	Semi-Annual	DBG
ELN-0801C	456	2813	4/15/08	498,213	2,046,896	150.5	843.82	841.42	2.5	5.0	n/a	PZ	Sand	С	Semi-Annual	DBG

### Table 3 Monitoring Well Construction Information – Sampling Required by WDNR Remedial Investigation/Feasibility Study

Badger Army Ammunition Plant	

Well Name	Well ID	License	Date Installed	NAD83 Northing (feet)	NAD83 Easting (feet)	Well Depth (feet)	Top of Casing Elevation	Ground Elevation	Well Diameter (inches)	Screen Length (feet)	Bedrock Depth (feet)	Well Type	Aquifer	Screen Level	Sample Frequency	Plume Area
ELN-0801E	457	2813	10/23/08	498,221	2,046,909	207.7	842.70	840.10	2.5	5.0	187	PZ	Rock	E	Semi-Annual	DBG
ELN-0802A	458	2813	10/28/08	498,661	2,045,219	107.5	878.47	876.20	2.5	15.0	n/a	OW	Sand	Α	Biennial	DBG
ELN-0802C	459	2813	10/30/08	498,663	2,045,211	180.8	878.47	876.10	2.5	5.0	n/a	PZ	Sand	С	Biennial	DBG
ELN-1001B	460	2813	5/11/10	497,078	2,047,480	96.1	809.31	806.98	2.5	5.0	n/a	PZ	Sand	В	Semi-Annual	DBG
ELN-1001C	461	2813	5/12/10	497,094	2,047,476	160.2	809.24	806.58	2.5	5.0	n/a	PZ	Sand	C	Semi-Annual	DBG
ELN-1001E	462	2813	6/23/10	497,110	2,047,472	245.5	809.34	806.46	2.5	5.0	230	ΡZ	Rock	E	Semi-Annual	DBG
ELN-1002A	463	2813	6/8/10	496,066	2,049,181	70.3	835.13	832.55	2.5	15.0	n/a	OW	Sand	Α	Semi-Annual	DBG
ELN-1002B	464	2813	6/9/10	496,056	2,049,188	116.2	835.15	832.39	2.5	5.0	n/a	PZ	Sand	В	Semi-Annual	DBG
ELN-1002C	465	2813	6/15/10	496,075	2,049,195	164.1	835.15	832.13	2.5	5.0	n/a	PZ	Sand	С	Semi-Annual	DBG
ELN-1002E	466	2813	6/17/10	496,063	2,049,200	236.5	834.75	831.97	2.5	5.0	219	PZ	Rock	E	Semi-Annual	DBG
ELN-1003A	467	2813	7/7/10	497,862	2,048,208	31.2	801.87	799.89	2.5	15.0	n/a	OW	Sand	Α	Quarterly	DBG
ELN-1003B	468	2813	7/6/10	497,867	2,048,198	96.5	801.40	798.74	2.5	5.0	n/a	PZ	Sand	В	Quarterly	DBG
ELN-1003C	469	2813	7/6/10	497,873	2,048,186	160.1	801.82	799.24	2.5	5.0	n/a	PZ	Sand	С	Quarterly	DBG
ELN-1003E	470	2813	7/1/10	497,876	2,048,172	230.6	801.62	799.12	2.5	5.0	213	PZ	Rock	E	Quarterly	DBG
ELN-1502A	533	2813	10/19/15	499,322	2,046,218	130.3	902.15	899.20	2.5	15.0	n/a	OW	Sand	Α	Semi-Annual	DBG
ELN-1502C	534	2813	10/14/15	499,317	2,046,221	203.0	902.36	899.30	2.5	5.0	n/a	PZ	Sand	С	Semi-Annual	DBG
ELN-1503A	535	2813	10/8/15	499,385	2,047,058	88.7	862.42	859.26	2.5	15.0	n/a	OW	Sand	Α	Semi-Annual	DBG
ELN-1503C	536	2813	10/7/15	499,377	2,047,057	162.6	862.29	859.54	2.5	5.0	n/a	PZ	Sand	С	Semi-Annual	DBG
ELN-1504B	537	2813	9/11/15	497,531	2,048,387	39.8	780.51	778.34	2.0	5.0	n/a	PZ	Sand	В	Quarterly	DBG
ELN-8203A	210	2813	3/24/82	501,516	2,044,336	157.5	927.79	925.20	4.0	10.0	n/a	OW	Sand	Α	Semi-Annual	DBG
ELN-8203B	211	2813	3/25/82	501,502	2,044,325	166.0	927.43	925.50	4.0	2.0	n/a	PZ	Sand	В	Semi-Annual	DBG
ELN-8203C	212	2813	3/24/82	501,517	2,044,323	176.0	926.93	925.30	4.0	2.0	n/a	PZ	Sand	С	Semi-Annual	DBG
ELN-8902B	224	2813	4/18/89	501,013	2,044,130	178.5	920.38	918.00	4.0	5.0	n/a	PZ	Sand	В	Semi-Annual	DBG
ELN-9107A	227	2813	11/10/91	500,568	2,045,411	126.0	897.72	895.30	3.8	10.0	n/a	OW	Sand	Α	Semi-Annual	DBG
ELN-9107B	228	2813	11/9/91	500,527	2,045,437	145.0	895.96	893.90	3.8	10.0	n/a	OW	Sand	В	Semi-Annual	DBG
ELN-9402AR	231	2813	2/15/94	501,014	2,044,060	145.0	920.92	919.00	4.0	15.0	n/a	OW	Sand	Α	Semi-Annual	DBG
S1134R	236	2813	6/8/95	501,504	2,043,991	151.0	922.06	920.60	4.0	15.0	n/a	OW	Sand	Α	Semi-Annual	DBG
DBM-8201	301	3037	3/23/82	500,846	2,043,148	174.7	918.76	916.70	4.0	20.0	n/a	OW	Sand	Α	Semi-Annual	DBG
DBM-8202	302	3037	3/20/82	501,147	2,042,937	157.4	920.35	917.80	4.0	20.0	n/a	OW	Sand	Α	Semi-Annual	DBG
DBM-8903	306	3037	2/16/89	500,499	2,043,488	133.0	898.94	896.40	4.0	20.0	n/a	OW	Sand	Α	Semi-Annual	DBG
DBN-1001B	472	3037	5/25/10	501,062	2,043,113	159.5	912.07	909.77	2.5	5.0	n/a	PZ	Sand	В	Semi-Annual	DBG
DBN-1001C	473	3037	5/27/10	501,063	2,043,094	197.0	912.00	909.78	2.5	5.0	n/a	PZ	Sand	С	Semi-Annual	DBG
DBN-1001E	474	3037	6/30/10	501,065	2,043,076	279.9	912.50	909.95	2.5	5.0	258	PZ	Rock	E	Semi-Annual	DBG
DBN-1002C	476	3037	6/17/10	500,487	2,044,488	210.1	916.12	913.72	2.5	5.0	n/a	PZ	Sand	С	Semi-Annual	DBG
DBN-1002E	477	3037	7/12/10	500,511	2,044,485	280.6	916.24	913.84	2.5	5.0	265	PZ	Rock	E	Semi-Annual	DBG
DBN-9501A	314	3037	10/24/95	500,312	2,043,686	120.0	889.10	886.70	3.8	10.0	n/a	OW	Sand	Α	Semi-Annual	DBG
DBN-9501B	315	3037	10/20/95	500,315	2,043,703	172.5	889.65	887.00	3.8	10.0	n/a	PZ	Sand	В	Semi-Annual	DBG
DBN-9501C	316	3037	10/18/95	500,298	2,043,710	228.5	890.03	887.50	3.8	10.0	n/a	PZ	Sand	С	Semi-Annual	DBG
DBN-9501E	317	3037	10/10/95	500,286	2,043,697	255.5	890.17	887.90	3.8	10.3	229	PZ	Rock	E	Semi-Annual	DBG
S1121	755	3038	1/18/80	496,303	2,047,578	59.3	815.58	813.90	4.0	20.2	n/a	OW	Sand	А	Semi-Annual	DBG

### Table 3 Monitoring Well Construction Information – Sampling Required by WDNR **Remedial Investigation/Feasibility Study**

Badger	Army Ammunition Plant	

Well Name	Well ID	License	Date Installed	NAD83 Northing (feet)	NAD83 Easting (feet)	Well Depth (feet)	Top of Casing Elevation	Ground Elevation	Well Diameter (inches)	Screen Length (feet)	Bedrock Depth (feet)	Well Type	Aquifer	Screen Level	Sample Frequency	Plume Area
RIM-0703	440	3487	10/4/07	499,282	2,034,376	113.0	889.23	886.53	2.5	15.0	n/a	OW	Sand	Α	Annual	NC
RIM-0705	442	3487	10/10/07	497,844	2,035,152	106.0	884.38	881.30	2.5	15.0	n/a	OW	Sand	A	Semi-Annual	NC
RIM-1002	478	3487	4/29/10	499,282	2,034,869	110.2	891.01	888.51	2.5	15.0	n/a	OW	Sand	A	Semi-Annual	NC
RIN-1001A	480	3487	4/28/10	497,066	2,035,221	106.8	884.38	882.05	2.5	15.0	n/a	OW	Sand	A	Semi-Annual	NC
RIN-1001C	481	3487	5/24/10	497,097	2,035,225	181.4	884.02	882.01	2.5	5.0	n/a	PZ	Sand	C	Annual	NC
RIN-1007C	479	3487	6/15/10	497,858	2,035,155	175.3	883.81	881.41	2.5	5.0	n/a	PZ	Sand	C	Annual	NC
S1125	504	3487	12/26/79	496,508	2,036,418	126.5	895.93	894.90	4.0	20.3	n/a	OW	Sand	Α	Semi-Annual	NC
PBM-0001	367	2814	7/14/00	491,611	2,035,455	134.5	890.23	887.54	4.0	25.0	n/a	OW	Sand	A	Semi-Annual	PBG
PBM-0002	368	2814	8/4/00	491,527	2,035,422	131.5	886.46	884.75	4.0	25.0	n/a	OW	Sand	A	Semi-Annual	PBG
PBM-0006	372	2814	8/1/00	491,477	2,035,323	124.5	879.02	875.89	4.0	25.0	n/a	OW	Sand	А	Semi-Annual	PBG
PBM-0008	374	2814	8/12/00	491,355	2,035,323	122.0	876.62	874.66	4.0	25.0	n/a	OW	Sand	A	Semi-Annual	PBG
PBM-8907	637	2814	3/3/89	487,689	2,034,443	92.7	849.45	846.60	4.0	10.0	n/a	OW	Sand	А	Annual	PBG
PBM-8909	639	2814	3/1/89	492,402	2,035,472	124.4	883.66	880.60	4.0	20.0	n/a	OW	Sand	A	Biennial	PBG
PBM-9801	360	2814	10/13/98	491,877	2,035,466	123.5	890.46	887.85	4.0	15.0	n/a	OW	Sand	A	Annual	PBG
PBN-1001C	595	2814	6/8/10	485,968	2,035,767	199.7	840.01	837.71	2.5	5.0	n/a	PZ	Sand	C	Semi-Annual	PBG
PBN-1003C	592	2814	6/3/10	487,681	2,034,448	189.6	848.21	846.51	2.5	5.0	n/a	PZ	Sand	C	Annual	PBG
PBN-1302A	770	2814	10/16/13	484,705	2,036,460	84.7	830.23	828.30	2.5	15.0	n/a	OW	Sand	A	Semi-Annual	PBG
PBN-1302B	771	2814	10/17/13	484,705	2,036,453	136.2	829.65	827.60	2.5	5.0	n/a	PZ	Sand	В	Semi-Annual	PBG
PBN-1302C	772	2814	10/22/13	484,705	2,036,448	187.6	828.98	827.00	2.5	5.0	n/a	PZ	Sand	C	Semi-Annual	PBG
PBN-1302D	773	2814	10/29/13	484,705	2,036,442	245.1	828.35	826.50	2.5	5.0	245	PZ	Sand	D	Semi-Annual	PBG
PBN-1303A	774	2814	11/5/13	484,651	2,036,981	130.5	884.88	883.00	2.5	15.0	n/a	OW	Sand	А	Semi-Annual	PBG
PBN-1303B	775	2814	11/12/13	484,651	2,036,968	176.5	883.71	881.60	2.5	5.0	n/a	PZ	Sand	В	Semi-Annual	PBG
PBN-1303C	776	2814	11/20/13	484,652	2,036,963	232.0	883.67	881.60	2.5	5.0	n/a	PZ	Sand	C	Semi-Annual	PBG
PBN-1303D	777	2814	11/22/13	484,652	2,036,958	287.0	883.42	881.60	2.5	5.0	287	PZ	Sand	D	Semi-Annual	PBG
PBN-1304A	778	2814	12/3/13	484,642	2,037,502	116.0	871.81	869.40	2.5	15.0	n/a	OW	Sand	А	Semi-Annual	PBG
PBN-1304B	779	2814	12/10/13	484,642	2,037,496	163.1	871.49	869.80	2.5	5.0	n/a	PZ	Sand	В	Semi-Annual	PBG
PBN-1304C	780	2814	12/17/13	484,642	2,037,489	218.0	872.00	869.70	2.5	5.0	n/a	PZ	Sand	C	Semi-Annual	PBG
PBN-1304D	781	2814	1/14/14	484,642	2,037,484	273.0	872.03	869.50	2.5	5.0	273	PZ	Sand	D	Semi-Annual	PBG
PBN-1401A	782	2814	2/19/14	491,036	2,035,501	132.2	887.30	884.57	2.5	15.0	n/a	OW	Sand	А	Semi-Annual	PBG
PBN-1401B	783	2814	2/12/14	491,035	2,035,494	163.7	887.09	884.57	2.5	5.0	n/a	PZ	Sand	В	Semi-Annual	PBG
PBN-1401C	784	2814	2/10/14	491,035	2,035,488	203.3	887.08	884.57	2.5	5.0	n/a	PZ	Sand	С	Semi-Annual	PBG
PBN-1404B	791	2814	3/11/14	487,745	2,035,891	179.5	895.08	892.18	2.5	5.0	n/a	PZ	Sand	В	Semi-Annual	PBG
PBN-1404C	792	2814	3/4/14	487,742	2,035,888	239.3	895.04	892.18	2.5	5.0	n/a	PZ	Sand	С	Semi-Annual	PBG
PBN-1404D	793	2814	2/26/14	487,737	2,035,885	299.8	894.49	892.18	2.5	5.0	300	PZ	Sand	D	Semi-Annual	PBG
PBN-1405F	794	2814	3/25/14	484,824	2,035,411	319.7	806.29	803.77	2.5	5.0	212	PZ	Rock	F	Biennial	PBG
PBN-8202A	613	2814	5/1/82	491,539	2,035,491	118.5	886.15	884.09	4.0	10.0	n/a	OW	Sand	А	Semi-Annual	PBG
PBN-8202B	614	2814	3/9/82	491,537	2,035,480	133.0	885.49	883.48	4.0	2.0	n/a	PZ	Sand	В	Semi-Annual	PBG
PBN-8202C	615	2814	3/8/82	491,529	2,035,490	141.2	885.43	882.47	4.0	2.0	n/a	PZ	Sand	С	Semi-Annual	PBG
PBN-8205A	622	2814	3/13/82	490,334	2,035,262	112.5	878.52	875.80	4.0	10.0	n/a	OW	Sand	A	Semi-Annual	PBG
PBN-8205B	623	2814	3/11/82	490,343	2,035,252	124.3	877.80	875.88	4.0	2.0	n/a	PZ	Sand	В	Semi-Annual	PBG

Well Name	Well ID	License	Date Installed	NAD83 Northing (feet)	NAD83 Easting (feet)	Well Depth (feet)	Top of Casing Elevation	Ground Elevation	Well Diameter (inches)	Screen Length (feet)	Bedrock Depth (feet)	Well Type	Aquifer	Screen Level	Sample Frequency	Plume Area
PBN-8205C	624	2814	3/11/82	490,330	2,035,250	133.5	878.31	875.80	4.0	2.0	n/a	PZ	Sand	С	Semi-Annual	PBG
PBN-8502A	632	2814	10/1/85	489,416	2,035,667	138.1	898.88	895.80	5.0	9.0	n/a	OW	Sand	А	Semi-Annual	PBG
PBN-8503A	633	2814	10/3/85	489,407	2,034,266	94.8	851.45	848.10	5.0	9.0	n/a	OW	Sand	А	Semi-Annual	PBG
PBN-8902BR	795	2814	3/24/14	489,418	2,035,684	160.0	898.87	896.82	2.5	5.0	n/a	PZ	Sand	В	Semi-Annual	PBG
PBN-8902C	645	2814	3/19/89	489,415	2,035,630	193.3	897.12	894.50	4.0	5.2	n/a	PZ	Sand	С	Semi-Annual	PBG
PBN-8903B	646	2814	3/8/89	489,457	2,034,281	125.0	847.93	844.90	4.0	5.0	n/a	PZ	Sand	В	Semi-Annual	PBG
PBN-8903C	647	2814	3/9/89	489,457	2,034,316	160.0	846.96	844.10	4.0	5.0	n/a	PZ	Sand	С	Semi-Annual	PBG
PBN-8912A	654	2814	3/2/89	486,338	2,034,980	103.4	855.86	852.60	4.0	20.0	n/a	OW	Sand	А	Semi-Annual	PBG
PBN-8912B	655	2814	4/15/89	486,312	2,034,979	138.0	856.34	852.60	4.0	5.0	n/a	PZ	Sand	В	Semi-Annual	PBG
PBN-9112C	665	2814	10/24/91	486,280	2,034,972	183.4	854.48	852.20	3.8	10.0	n/a	PZ	Sand	С	Semi-Annual	PBG
PBN-9112D	666	2814	10/16/91	486,253	2,034,965	231.0	853.31	851.20	3.8	10.0	n/a	PZ	Sand	D	Semi-Annual	PBG
PBN-9301B	668	2814	3/19/93	489,365	2,036,994	160.5	875.03	872.20	3.9	10.0	n/a	PZ	Sand	В	Semi-Annual	PBG
PBN-9301C	669	2814	3/16/93	489,353	2,037,006	227.5	874.64	872.22	3.9	10.0	n/a	PZ	Sand	С	Semi-Annual	PBG
PBN-9303B	673	2814	3/9/93	486,123	2,036,945	93.5	816.16	813.49	3.9	10.0	n/a	PZ	Sand	В	Semi-Annual	PBG
PBN-9303C	674	2814	3/14/93	486,126	2,036,969	164.5	815.05	812.45	3.9	10.0	n/a	PZ	Sand	С	Semi-Annual	PBG
PBN-9303D	675	2814	3/11/93	486,127	2,036,990	224.5	813.98	811.41	3.9	10.0	223	PZ	Sand	D	Semi-Annual	PBG
PBN-9304D	687	2814	10/19/93	484,890	2,035,315	210.0	806.09	804.10	4.0	10.0	210	PZ	Sand	D	Semi-Annual	PBG
PBN-9902D	691	2814	7/1/99	484,798	2,035,025	222.5	811.53	809.50	4.0	5.0	217	PZ	Sand	D	Semi-Annual	PBG
PBN-9903A	692	2814	6/23/99	483,859	2,035,680	76.0	826.91	825.18	4.0	15.0	n/a	OW	Sand	А	Semi-Annual	PBG
PBN-9903B	693	2814	7/8/99	483,859	2,035,687	112.0	827.17	825.00	4.0	5.0	n/a	PZ	Sand	В	Semi-Annual	PBG
PBN-9903C	694	2814	7/15/99	483,861	2,035,693	163.0	827.33	824.99	4.0	5.0	n/a	PZ	Sand	С	Semi-Annual	PBG
PBN-9903D	695	2814	7/13/99	483,861	2,035,698	208.0	827.52	825.10	4.0	5.0	196	PZ	Sand	D	Semi-Annual	PBG
PBM-9001D	981	3485	8/25/90	477,175	2,038,945	210.5	831.52	829.00	4.0	10.0	n/a	PZ	Sand	D	Semi-Annual	PBG
PBM-9002D	982	3485	8/18/90	475,994	2,038,132	204.5	821.31	818.70	4.0	10.0	n/a	PZ	Sand	D	Biennial	PBG
PBN-9101C	561	3493	10/25/91	477,125	2,038,954	152.5	830.11	828.00	3.8	10.0	n/a	PZ	Sand	С	Semi-Annual	PBG
PBN-9102B	562	3493	9/28/91	476,019	2,038,141	115.0	821.19	819.00	3.8	10.0	n/a	PZ	Sand	В	Biennial	PBG
PBN-9102C	563	3493	9/30/91	476,028	2,038,105	161.3	821.90	819.90	3.8	10.0	n/a	PZ	Sand	С	Biennial	PBG
SWN-9102C	569	3493	10/27/91	479,341	2,035,141	152.5	836.41	834.40	3.8	10.0	n/a	PZ	Sand	С	Annual	PBG
SWN-9102D	570	3493	10/23/91	479,341	2,035,185	185.0	836.66	834.50	4.0	10.0	n/a	PZ	Sand	D	Annual	PBG
SWN-9103B	571	3493	10/4/91	479,353	2,036,656	113.4	836.63	834.70	3.8	10.0	n/a	PZ	Sand	В	Semi-Annual	PBG
SWN-9103C	572	3493	10/2/91	479,351	2,036,622	162.8	836.80	834.60	4.0	10.0	n/a	PZ	Sand	С	Semi-Annual	PBG
SWN-9103D	573	3493	10/1/91	479,352	2,036,701	209.1	837.10	835.00	4.0	10.0	210	PZ	Sand	D	Semi-Annual	PBG
SWN-9103E	574	3493	11/10/91	479,352	2,036,753	237.9	837.38	835.00	3.8	10.0	210	PZ	Rock	E	Semi-Annual	PBG
SWN-9104C	575	3493	10/13/91	479,357	2,037,722	164.0	834.87	832.80	3.8	10.0	n/a	PZ	Sand	С	Semi-Annual	PBG
SWN-9104D	576	3493	10/9/91	479,359	2,037,678	197.0	835.33	833.50	3.8	10.0	n/a	PZ	Sand	D	Semi-Annual	PBG
SWN-9105B	577	3493	10/12/91	478,954	2,038,812	112.5	832.73	830.50	3.8	10.0	n/a	PZ	Sand	В	Annual	PBG
SWN-9105C	578	3493	10/11/91	478,924	2,038,828	147.0	832.88	830.80	3.8	10.0	n/a	PZ	Sand	С	Annual	PBG
SWN-9105D	579	3493	10/10/91	478,885	2,038,855	200.5	833.35	831.20	3.8	10.0	n/a	PZ	Sand	D	Annual	PBG
S1147	709	3499	10/10/83	484,928	2,034,512	70.8	817.07	815.70	5.0	25.0	n/a	OW	Sand	А	Semi-Annual	PBG
S1148	710	3499	10/10/83	484.691	2.035.563	56.7	803.72	802.10	5.0	25.0	n/a	OW	Sand	Α	Semi-Annual	PBG

Well Name	Well ID	License	Date Installed	NAD83 Northing (feet)	NAD83 Easting (feet)	Well Depth (feet)	Top of Casing Elevation	Ground Elevation	Well Diameter (inches)	Screen Length (feet)	Bedrock Depth (feet)	Well Type	Aquifer	Screen Level	Sample Frequency	Plume Area
SPN-8903B	718	3499	3/22/89	484,935	2,034,532	93.7	818.14	815.10	4.0	5.0	n/a	PZ	Sand	В	Semi-Annual	PBG
SPN-8903C	719	3499	4/13/89	484,907	2,034,501	127.7	818.13	815.30	4.0	5.0	n/a	PZ	Sand	С	Semi-Annual	PBG
SPN-8904B	720	3499	3/9/89	484,691	2,035,540	75.0	804.23	801.60	4.0	5.0	n/a	PZ	Sand	В	Semi-Annual	PBG
SPN-8904C	721	3499	3/30/89	484,694	2,035,642	106.5	803.25	800.70	4.0	5.0	n/a	PZ	Sand	С	Semi-Annual	PBG
SPN-9103D	725	3499	10/8/91	484,909	2,034,440	200.5	819.29	816.70	3.8	10.0	n/a	PZ	Sand	D	Semi-Annual	PBG
SPN-9104D	726	3499	10/1/91	484,693	2,035,601	206.0	802.61	800.80	3.8	10.0	212	PZ	Sand	D	Semi-Annual	PBG

<u>Notes</u>

OW = Water Table Observation Well

PZ = Piezometer

DBG = Deterrent Burning Ground Plume

Central = Central Plume

NC = Nitrocellulose Production Area Plume

PBG = Propellant Burning Ground Plume

Screen Level references the typical well depth configuration

### Table 4 Monitoring Well Construction Information – Sampling Not Required by WDNR Remedial Investigation/Feasibility Study Redeer Army Amounitien Plant

Bauger Army Ammunición Plant	

Well Name	Well ID	License	Date Installed	NAD83 Northing (feet)	NAD83 Easting (feet)	Well Depth (feet)	Top of Casing Elevation	Ground Elevation	Well Diameter (inches)	Screen Length (feet)	Bedrock Depth (feet)	Well Type	Aquifer	Screen Level	Sample Frequency	Plume Area
S1112	752	3038	1/4/80	490,050	2,045,210	91.7	838.03	836.40	4.0	20.3	n/a	OW	Sand	А	Not Sampled	Central
S1113	753	3038	11/23/79	491,611	2,048,037	66.1	821.58	820.00	4.0	20.2	n/a	OW	Sand	А	Not Sampled	Central
S1114	754	3038	11/20/79	491,603	2,048,038	105.4	821.46	820.10	4.0	5.0	n/a	PZ	Sand	С	Not Sampled	Central
NLM-9202R	270	3118	12/21/92	494,989	2,046,317	118.2	885.15	882.90	4.0	15.0	n/a	OW	Sand	А	Not Sampled	Central
NLN-8201A	252	3118	4/23/82	495,556	2,045,494	120.3	890.65	888.60	4.0	10.0	n/a	OW	Sand	А	Not Sampled	Central
NLN-8201B	253	3118	4/22/82	495,566	2,045,487	132.5	891.28	889.00	4.0	2.0	n/a	PZ	Sand	В	Not Sampled	Central
NLN-8201C	254	3118	4/7/82	495,552	2,045,485	142.0	890.54	888.60	4.0	2.0	n/a	PZ	Sand	С	Not Sampled	Central
NLN-8202A	255	3118	4/30/82	495,648	2,046,075	102.9	873.61	872.53	4.0	10.0	n/a	OW	Sand	А	Not Sampled	Central
NLN-8202B	256	3118	4/23/82	495,646	2,046,087	115.0	873.69	871.97	4.0	2.0	n/a	PZ	Sand	В	Not Sampled	Central
NLN-8204A	261	3118	5/8/82	494,911	2,045,873	125.5	892.72	891.00	4.0	10.0	n/a	OW	Sand	А	Not Sampled	Central
NLN-8204B	262	3118	5/8/82	494,899	2,045,877	137.5	893.44	891.60	4.0	2.0	n/a	PZ	Sand	В	Not Sampled	Central
NLN-8204C	263	3118	5/7/82	494,901	2,045,867	150.0	893.54	891.60	4.0	2.0	n/a	PZ	Sand	С	Not Sampled	Central
NLN-8205B	265	3118	5/10/82	494,905	2,046,159	136.5	899.28	896.90	4.0	2.0	n/a	PZ	Sand	В	Not Sampled	Central
NLN-8205C	266	3118	5/10/82	494,917	2,046,156	147.5	897.99	896.30	4.0	2.0	n/a	PZ	Sand	С	Not Sampled	Central
NLN-9205AR	269	3118	11/13/92	494,913	2,046,170	132.0	897.82	895.30	4.0	15.0	n/a	OW	Sand	А	Not Sampled	Central
RPM-9101	509	3487	10/26/91	492,702	2,045,303	105.8	874.04	871.80	3.8	10.0	n/a	OW	Sand	А	Not Sampled	Central
S1120	502	3487	1/17/80	493,313	2,044,061	122.8	880.14	877.40	4.0	20.2	n/a	OW	Sand	А	Not Sampled	Central
S1150	505	3487	10/10/83	496,772	2,037,797	138.0	897.56	895.60	5.0	25.0	n/a	OW	Sand	А	Not Sampled	Central
NLM-0301R	271	3646	7/23/03	495,613	2,045,778	112.0	881.20	877.92	4.0	15.0	n/a	OW	Sand	А	Not Sampled	Central
NLM-0302R	272	3646	1/9/04	496,404	2,045,533	127.0	894.50	891.70	4.0	15.0	n/a	OW	Sand	А	Not Sampled	Central
NLM-0401	296	3646	8/3/04	495,912	2,046,255	112.0	869.29	866.66	4.0	15.0	n/a	OW	Sand	А	Not Sampled	Central
NLM-1001	330	3646	4/14/10	496,509	2,044,604	106.0	880.22	878.00	4.0	15.0	n/a	OW	Sand	А	Not Sampled	Central
NLN-0701A	297	3646	6/6/07	495,491	2,045,250	125.0	887.47	884.87	4.0	15.0	n/a	OW	Sand	А	Not Sampled	Central
NLN-0701C	298	3646	6/5/07	495,491	2,045,242	155.0	887.29	884.79	4.0	5.0	n/a	PZ	Sand	С	Not Sampled	Central
ELM-9110	229	2813	11/13/91	501,635	2,044,708	154.0	923.03	920.80	3.8	15.0	n/a	OW	Sand	А	Not Sampled	DBG
ELN-8904A	225	2813	3/30/89	501,790	2,044,600	162.0	926.34	924.10	4.0	20.0	n/a	OW	Sand	Α	Not Sampled	DBG
ELN-8904B	226	2813	4/2/89	501,721	2,044,645	199.0	926.61	924.80	4.0	5.0	n/a	PZ	Sand	В	Not Sampled	DBG
LOM-8901	656	2814	2/17/89	492,014	2,036,131	157.5	918.08	915.90	4.0	20.0	n/a	PZ	Sand	А	Not Sampled	PBG
LOM-9101	661	2814	10/10/91	492,618	2,036,184	151.0	917.76	915.50	3.8	10.0	n/a	PZ	Sand	А	Not Sampled	PBG
LOM-9102	662	2814	10/25/91	493,326	2,036,375	148.0	912.46	910.30	3.8	10.0	n/a	OW	Sand	А	Not Sampled	PBG
LON-8902A	657	2814	2/19/89	491,571	2,036,136	159.0	927.95	918.50	4.0	20.0	n/a	OW	Sand	А	Not Sampled	PBG
LON-8903A	659	2814	2/20/89	491,581	2,036,311	158.0	926.36	919.20	4.0	20.0	n/a	OW	Sand	Α	Not Sampled	PBG
LON-8903B	660	2814	2/20/89	491,579	2,036,275	198.0	927.41	919.50	4.0	5.0	n/a	PZ	Sand	В	Not Sampled	PBG
LON-9502BR	683	2814	6/1/95	491,573	2,036,166	203.5	927.54	919.30	4.0	18.5	n/a	PZ	Sand	В	Not Sampled	PBG
PBM-0003	369	2814	8/8/00	491,440	2,035,388	120.5	875.95	876.89	4.0	25.0	n/a	OW	Sand	Α	Not Sampled	PBG
PBM-0004	370	2814	7/25/00	491,356	2,035,354	125.5	877.62	875.64	4.0	25.0	n/a	OW	Sand	Α	Not Sampled	PBG
PBM-0005	371	2814	7/19/00	491,566	2,035,322	128.0	883.58	881.22	4.0	25.0	n/a	OW	Sand	Α	Not Sampled	PBG
PBM-0007	373	2814	7/24/00	491,417	2,035,323	120.9	874.47	872.56	4.0	25.0	n/a	OW	Sand	А	Not Sampled	PBG
PBM-1201	764	2814	11/15/12	, 491,516	2,035,458	118.5	882.56	880.24	2.5	15.0	n/a	OW	Sand	А	Not Sampled	PBG
PBM-1202	765	2814	11/19/12	491,507	2,035,442	118.5	881.48	879.01	2.5	15.0	n/a	OW	Sand	A	Not Sampled	PBG

### Table 4 Monitoring Well Construction Information – Sampling Not Required by WDNR Remedial Investigation/Feasibility Study Redeer Army Amounitien Plant

	Badger A	Army Ammur	nition Plant	

Well Name	Well ID	License	Date Installed	NAD83 Northing (feet)	NAD83 Easting (feet)	Well Depth (feet)	Top of Casing Elevation	Ground Elevation	Well Diameter (inches)	Screen Length (feet)	Bedrock Depth (feet)	Well Type	Aquifer	Screen Level	Sample Frequency	Plume Area
PBM-1203	766	2814	11/20/12	491,496	2,035,425	118.4	880.18	877.69	2.5	15.0	n/a	OW	Sand	А	Not Sampled	PBG
PBM-8201	605	2814	3/18/82	491,409	2,034,559	100.7	857.36	855.70	4.0	20.0	n/a	PZ	Sand	Α	Not Sampled	PBG
PBM-8203	607	2814	3/16/82	490,778	2,034,771	108.8	868.42	862.70	4.0	20.0	n/a	PZ	Sand	Α	Not Sampled	PBG
PBM-8204	608	2814	3/17/82	490,553	2,035,006	115.5	875.72	869.00	4.0	20.0	n/a	PZ	Sand	А	Not Sampled	PBG
PBM-8205	609	2814	5/3/82	490,547	2,035,178	123.8	877.11	874.50	4.0	20.0	n/a	PZ	Sand	Α	Not Sampled	PBG
PBM-8501	625	2814	9/22/85	489,712	2,034,851	121.6	862.73	859.30	5.0	9.0	n/a	OW	Sand	Α	Not Sampled	PBG
PBM-8502	626	2814	9/17/85	489,417	2,034,654	101.7	849.42	845.40	5.0	9.0	n/a	OW	Sand	А	Not Sampled	PBG
PBM-8503	627	2814	9/18/85	489,414	2,035,277	150.5	886.29	882.90	5.0	9.0	n/a	OW	Sand	А	Not Sampled	PBG
PBM-8504	628	2814	9/24/85	488,819	2,035,043	125.4	866.47	863.80	5.0	9.0	n/a	OW	Sand	Α	Not Sampled	PBG
PBM-8505	629	2814	9/28/85	488,223	2,035,056	111.0	863.97	861.30	5.0	9.0	n/a	OW	Sand	А	Not Sampled	PBG
PBM-8506	630	2814	10/4/85	487,043	2,035,032	98.2	848.18	845.10	5.0	9.0	n/a	OW	Sand	А	Not Sampled	PBG
PBM-8905	635	2814	3/6/89	489,403	2,033,827	98.1	855.64	852.30	4.0	20.0	n/a	OW	Sand	А	Not Sampled	PBG
PBM-8906	636	2814	4/30/89	489,509	2,036,227	136.0	886.34	883.70	4.0	20.0	n/a	OW	Sand	А	Not Sampled	PBG
PBM-8908	638	2814	3/14/89	487,520	2,035,745	125.0	888.68	885.50	4.0	20.0	n/a	OW	Sand	А	Not Sampled	PBG
PBM-8911	640	2814	3/7/89	493,411	2,035,391	111.0	884.45	881.60	4.0	20.0	n/a	OW	Sand	А	Not Sampled	PBG
PBM-9803	526	2814	10/7/98	491,595	2,035,352	121.7	885.16	882.64	4.0	15.0	n/a	OW	Sand	А	Not Sampled	PBG
PBM-9901	361	2814	6/4/99	491,934	2,035,484	130.0	891.56	888.90	4.0	105.0	n/a	OW	Sand	А	Not Sampled	PBG
PBM-9902	362	2814	6/4/99	491,664	2,035,482	132.0	890.94	888.35	4.0	110.0	n/a	OW	Sand	А	Not Sampled	PBG
PBM-9903	363	2814	6/4/99	491,628	2,035,319	126.0	882.42	880.87	4.0	105.0	n/a	OW	Sand	А	Not Sampled	PBG
PBN-1001A	593	2814	5/3/10	485,984	2,035,770	79.3	840.37	838.17	2.5	15.0	n/a	OW	Sand	А	Not Sampled	PBG
PBN-1001B	594	2814	6/2/10	485,976	2,035,768	139.9	839.93	838.23	2.5	5.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-1002A	589	2814	5/20/10	488,451	2,035,897	130.8	893.90	891.70	2.5	15.0	n/a	OW	Sand	А	Not Sampled	PBG
PBN-1002B	590	2814	5/19/10	488,447	2,035,927	176.5	894.27	892.27	2.5	5.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-1002C	591	2814	6/9/10	488,450	2,035,908	216.8	893.48	891.48	2.5	5.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-1301A	767	2814	9/16/13	491,295	2,035,639	130.0	899.97	897.35	2.5	15.0	n/a	OW	Sand	А	Not Sampled	PBG
PBN-1301B	768	2814	9/12/13	491,310	2,035,602	159.5	897.32	894.58	2.5	5.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-1301C	769	2814	9/10/13	491,265	2,035,609	200.0	897.14	894.54	2.5	5.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-1402A	785	2814	2/4/14	490,204	2,035,272	113.6	878.31	876.47	2.5	15.0	n/a	OW	Sand	А	Not Sampled	PBG
PBN-1402B	786	2814	2/10/14	490,204	2,035,277	132.9	878.77	876.47	2.5	5.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-1402C	787	2814	2/18/14	490,204	2,035,282	162.8	878.74	876.47	2.5	5.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-1403A	788	2814	2/27/14	489,290	2,035,682	135.7	901.24	899.00	2.5	15.0	n/a	OW	Sand	А	Not Sampled	PBG
PBN-1403B	789	2814	2/26/14	489,290	2,035,687	157.2	901.22	899.05	2.5	5.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-1403C	790	2814	2/20/14	489,290	2,035,693	192.0	901.64	899.27	2.5	5.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-8201A	610	2814	3/18/82	492,093	2,035,482	117.8	884.59	881.50	4.0	10.0	n/a	OW	Sand	А	Not Sampled	PBG
PBN-8201B	611	2814	3/10/82	492,091	2,035,469	131.5	883.77	881.50	4.0	2.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-8201C	612	2814	3/10/82	492,101	2,035,476	141.0	883.98	881.50	4.0	2.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-8203A	616	2814	3/15/82	490,314	2,034,600	96.5	860.01	857.60	4.0	10.0	n/a	OW	Sand	A	Not Sampled	PBG
PBN-8203B	617	2814	3/15/82	490,311	2,034,613	108.5	860.26	857.60	4.0	2.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-8203C	618	2814	3/15/82	490,300	2,034,606	117.5	860.17	857.60	4.0	2.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-8204B	620	2814	3/13/82	490,027	2,035,049	120.5	874.74	873.00	4.0	2.0	n/a	PZ	Sand	В	Not Sampled	PBG

### Table 4 Monitoring Well Construction Information – Sampling Not Required by WDNR Remedial Investigation/Feasibility Study Redeer Army Amounitien Plant

Badger Army Ammunition Plant	

Well Name	Well ID	License	Date Installed	NAD83 Northing (feet)	NAD83 Easting (feet)	Well Depth (feet)	Top of Casing Elevation	Ground Elevation	Well Diameter (inches)	Screen Length (feet)	Bedrock Depth (feet)	Well Type	Aquifer	Screen Level	Sample Frequency	Plume Area
PBN-8204C	621	2814	3/12/82	490,026	2,035,062	131.5	875.59	873.00	4.0	2.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-8501A	631	2814	9/18/85	489,413	2,035,044	121.9	874.51	871.30	5.0	9.0	n/a	OW	Sand	Α	Not Sampled	PBG
PBN-8504A	634	2814	9/30/85	487,634	2,035,066	112.7	860.03	857.20	5.0	9.0	n/a	OW	Sand	А	Not Sampled	PBG
PBN-8901B	641	2814	1/22/89	489,397	2,035,022	159.9	872.55	870.00	4.0	5.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-8901C	642	2814	4/19/89	489,395	2,035,102	198.1	878.03	875.50	4.0	5.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-8901D	643	2814	1/21/89	489,397	2,035,047	238.2	874.19	871.50	4.0	5.0	n/a	PZ	Sand	D	Not Sampled	PBG
PBN-8904B	648	2814	3/19/89	487,673	2,035,060	144.0	859.32	856.70	4.0	5.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-8904C	649	2814	4/16/89	487,651	2,035,092	180.5	859.87	857.70	4.0	5.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-8910A	650	2814	2/22/89	491,156	2,035,501	128.0	889.82	886.80	4.0	20.0	n/a	OW	Sand	Α	Not Sampled	PBG
PBN-8910B	651	2814	2/28/89	491,159	2,035,539	166.7	892.09	889.10	4.0	5.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-8910C	652	2814	2/3/89	491,154	2,035,464	192.0	887.11	884.70	4.0	5.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-8910D	653	2814	4/29/89	491,142	2,035,388	237.0	884.42	880.90	4.0	5.0	n/a	PZ	Sand	D	Not Sampled	PBG
PBN-9106C	663	2814	10/22/91	487,104	2,035,032	201.0	848.71	846.10	3.8	10.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-9106D	664	2814	10/12/91	487,107	2,035,008	251.0	847.53	845.80	3.8	10.0	n/a	PZ	Sand	D	Not Sampled	PBG
PBN-9302B	670	2814	3/5/93	487,005	2,036,974	154.5	873.31	871.26	3.9	10.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-9302C	671	2814	2/26/93	487,017	2,036,966	204.0	873.76	872.24	3.9	10.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-9302D	672	2814	3/7/93	487,001	2,036,953	289.5	874.93	870.72	3.9	10.0	288	PZ	Sand	D	Not Sampled	PBG
PBN-9304A	684	2814	10/12/93	484,886	2,035,343	50.0	805.93	804.00	4.0	15.0	n/a	OW	Sand	Α	Not Sampled	PBG
PBN-9304B	685	2814	10/19/93	484,897	2,035,329	86.0	805.77	804.00	4.0	10.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-9304C	686	2814	10/21/93	484,866	2,035,315	115.0	806.41	804.50	4.0	10.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-9306C	667	2814	3/22/90	489,507	2,036,238	227.5	886.51	884.06	3.9	10.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-9401B	677	2814	8/8/94	486,957	2,038,337	127.7	852.23	850.50	4.0	10.3	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-9401C	678	2814	8/9/94	486,981	2,038,338	167.8	852.96	851.00	4.0	10.4	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-9401D	679	2814	8/3/94	486,971	2,038,337	267.0	853.01	850.90	4.0	10.0	277	PZ	Sand	D	Not Sampled	PBG
PBN-9402B	680	2814	8/24/94	485,560	2,038,160	95.5	816.36	813.90	4.0	10.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-9402C	681	2814	8/22/94	485,560	2,038,150	135.0	816.35	813.80	4.0	10.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-9402D	682	2814	8/18/94	485,557	2,038,140	225.0	816.14	813.70	4.0	10.0	n/a	PZ	Sand	D	Not Sampled	PBG
PBN-9404AR	676	2814	2/18/94	490,017	2,035,038	118.0	873.63	871.30	4.0	15.0	n/a	OW	Sand	А	Not Sampled	PBG
PBN-9901A	696	2814	6/22/99	484,812	2,034,889	59.0	810.38	808.39	4.0	15.0	n/a	OW	Sand	Α	Not Sampled	PBG
PBN-9901B	697	2814	6/29/99	484,808	2,034,889	107.0	809.93	808.46	4.0	5.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-9901C	698	2814	6/28/99	484,799	2,034,890	163.0	810.00	808.45	4.0	5.0	n/a	PZ	Sand	С	Not Sampled	PBG
PBN-9901D	699	2814	6/23/99	484,790	2,034,891	216.0	810.95	808.52	4.0	5.0	216	PZ	Sand	D	Not Sampled	PBG
PBN-9902A	688	2814	6/22/99	484,805	2,035,024	60.0	811.54	808.91	4.0	15.0	n/a	OW	Sand	А	Not Sampled	PBG
PBN-9902B	689	2814	7/8/99	484,803	2,035,020	111.0	810.72	808.41	4.0	5.0	n/a	PZ	Sand	В	Not Sampled	PBG
PBN-9902C	690	2814	7/7/99	484,800	2,035,029	168.0	811.23	809.16	4.0	5.0	n/a	PZ	Sand	С	Not Sampled	PBG
S1109	600	2814	2/14/80	488,537	2,032,975	107.3	856.64	855.10	4.0	20.4	n/a	OW	Sand	А	Not Sampled	PBG
S1117	601	2814	2/13/80	490,355	2,034,837	119.1	867.92	862.30	4.0	20.2	n/a	OW	Sand	A	Not Sampled	PBG
SWN-0501B	237	3493	12/15/05	480,635	2,039,879	155.6	860.07	860.40	4.0	10.0	n/a	PZ	Sand	В	Not Sampled	PBG
SWN-0501C	238	3493	12/13/05	480,634	2,039,894	206.6	860.28	860.60	4.0	10.0	n/a	PZ	Sand	C	Not Sampled	PBG
SWN-0501D	239	3493	12/9/05	480,635	2,039,906	262.9	860.38	860.50	4.0	10.0	n/a	PZ	Sand	D	Not Sampled	PBG

Well Name	Well ID	License	Date Installed	NAD83 Northing (feet)	NAD83 Easting (feet)	Well Depth (feet)	Top of Casing Elevation	Ground Elevation	Well Diameter (inches)	Screen Length (feet)	Bedrock Depth (feet)	Well Type	Aquifer	Screen Level	Sample Frequency	Plume Area
SWN-0501E	240	3493	11/30/05	480,635	2,039,917	290.3	860.53	860.70	2.0	10.0	253	PZ	Rock	E	Not Sampled	PBG
SWN-0502B	241	3493	12/22/05	479,887	2,039,265	155.8	856.10	856.30	4.0	10.0	n/a	PZ	Sand	В	Not Sampled	PBG
SWN-0502C	242	3493	12/20/05	479,885	2,039,280	201.5	856.39	856.50	4.0	10.0	n/a	PZ	Sand	С	Not Sampled	PBG
SWN-0502D	243	3493	12/7/05	479,886	2,039,273	244.9	856.19	856.30	4.0	10.0	n/a	PZ	Sand	D	Not Sampled	PBG
SWN-0502E	244	3493	12/13/05	479,893	2,039,267	260.0	856.27	856.50	2.0	10.0	240	PZ	Rock	E	Not Sampled	PBG
S1102	701	3499	11/5/79	484,693	2,036,063	64.6	809.25	807.70	4.0	20.0	n/a	OW	Sand	А	Not Sampled	PBG
S1103	702	3499	11/2/79	484,689	2,036,056	120.1	809.02	807.50	4.0	5.1	n/a	PZ	Sand	С	Not Sampled	PBG
S1106	705	3499	11/14/79	484,794	2,039,567	135.7	839.91	838.10	4.0	5.0	n/a	PZ	Sand	C	Not Sampled	PBG
S1133	708	3499	2/19/80	484,746	2,032,920	97.0	828.28	828.20	4.0	5.2	n/a	PZ	Sand	В	Not Sampled	PBG
S1149	711	3499	10/10/83	485,128	2,036,476	60.8	807.75	806.10	5.0	25.0	n/a	OW	Sand	А	Not Sampled	PBG
S1152AR	727	3499	4/12/95	484,582	2,036,036	56.0	812.48	809.80	4.0	15.0	n/a	OW	Sand	А	Not Sampled	PBG
S1152B	713	3499	9/26/85	484,582	2,036,049	73.6	813.26	810.30	4.0	5.0	n/a	OW	Sand	В	Not Sampled	PBG
SPN-8901C	714	3499	3/29/89	484,722	2,032,922	121.0	830.09	827.80	4.0	5.0	n/a	PZ	Sand	С	Not Sampled	PBG
SPN-8902A	715	3499	2/22/89	484,748	2,033,808	71.0	823.67	820.80	4.0	20.0	n/a	OW	Sand	А	Not Sampled	PBG
SPN-8902B	716	3499	3/15/89	484,741	2,033,827	98.8	823.61	820.30	4.0	5.0	n/a	PZ	Sand	В	Not Sampled	PBG
SPN-8902C	717	3499	4/14/89	484,745	2,033,868	129.0	822.48	820.00	4.0	5.0	n/a	PZ	Sand	С	Not Sampled	PBG
SPN-9102D	724	3499	10/9/91	484,733	2,033,650	182.8	824.11	821.60	3.8	10.0	n/a	PZ	Sand	D	Not Sampled	PBG

#### <u>Notes</u>

OW = Water Table Observation Well

PZ = Piezometer

DBG = Deterrent Burning Ground Plume

Central = Central Plume

PBG = Propellant Burning Ground Plume

Screen Level references the typical well depth configuration

Well Name	Well ID	License	Date Installed	Well Depth (feet)	Well Diameter (inches)	Bedrock Depth (feet)	Aquifer	Sample Frequency	Plume Area
USDA 1	828	3497	11/4/79	575	12	263	Rock	Annual	Central
USDA 2	829	3497	7/18/96	227	6	n/a	Sand	Annual	Central
USDA 3	126	3497	10/21/80	270	6	235	Rock	Annual	Central
USDA 6	128	3497	3/7/06	140	8	n/a	Sand	Annual	Central
WE-QN039	158	3497	11/15/01	100	6	n/a	Sand	Annual	Central
WE-QR441	157	3497	1/29/02	118	5	n/a	Sand	Annual	Central
WE-RD430	159	3497	12/10/02	80	6	n/a	Sand	Annual	Central
WE-RM383	153	3497	6/10/03	81	6	n/a	Sand	Annual	Central
WE-RR542	156	3497	9/20/03	100	6	n/a	Sand	Annual	Central
WE-RR598	169	3497	3/10/04	106	6	n/a	Sand	Annual	Central
WE-SQ001	165	3497	1/15/05	179	6	n/a	Sand	Annual	Central
WE-SQ002	170	3497	1/20/05	100	6	n/a	Sand	Annual	Central
WE-SQ017	164	3497	3/10/05	180	5	n/a	Sand	Annual	Central
WE-TF023	174	3497	2/22/06	178	5	n/a	Sand	Annual	Central
WE-TM599	129	3497	10/2/06	120	5	n/a	Sand	Annual	Central
WE-UA297	433	3497	7/17/07	180	6	n/a	Sand	Annual	Central
WE-UK125	431	3497	12/29/07	283	5	243	Rock	Annual	Central
WE-XD828	434	3497	8/19/13	80	6	n/a	Sand	Annual	Central
WE-XK342	435	3497	8/27/14	80	6	n/a	Sand	Annual	Central
WE-YW972	436	3497	5/14/18	121	6	n/a	Sand	Annual	Central
WE-ZE512	437	3497	12/22/18	324	6	205	Rock	Quarterly	Central
Anderson-R	411	3497		26	n/a	n/a	Sand	Annual	DBG
Brey	817	3497		85	6	n/a	Sand	Annual	DBG
Curto	412	3497		n/a	n/a	n/a	unknown	Annual	DBG
Gibbs	839	3497		n/a	n/a	n/a	unknown	Annual	DBG
Grosse	415	3497		110	n/a	n/a	Sand	Annual	DBG
Groth	842	3497	1/5/89	219	6	169	Rock	Annual	DBG
Gruber-D	417	3497		n/a	n/a	n/a	Sand	Annual	DBG
Hendershot	418	3497		20	n/a	n/a	Sand	Annual	DBG
Howery	419	3497		n/a	n/a	n/a	unknown	Annual	DBG
Kopras	874	3497	5/28/88	260	6	217	Rock	Annual	DBG
Lukens	860	3497	7/25/08	29	1	n/a	Sand	Annual	DBG
Melum	423	3497	7/6/06	100	5	n/a	Sand	Annual	DBG
Nowotarski	891	3497	11/16/99	88	2	n/a	Sand	Annual	DBG
Olah	904	3497		30	n/a	n/a	Sand	Annual	DBG
Osterland	422	3497		n/a	n/a	n/a	unknown	Annual	DBG
Purcell-D	163	3497	7/26/19	344	6	216	Rock	Quarterly	DBG
Purcell-G	916	3497		n/a	n/a	n/a	unknown	Annual	DBG
Raschein	424	3497		n/a	n/a	n/a	unknown	Annual	DBG
Reif	427	3497	- /- /	n/a	n/a	n/a	unknown	Annual	DBG
Revers	425	3497	5/8/89	80	6	n/a	Sand	Annual	DBG
Roll	426	3497		n/a	n/a	n/a	unknown	Annual	DBG
Schumann	428	3497		n/a	n/a	n/a	Sand	Annual	DBG
Spear	803	3497	3/1/93	159	n/a	n/a	Sand	Annual	DBG
Wenger	414	3497	- / /	n/a	n/a	n/a	Sand	Annual	DBG
Zurbachen-A	967	3497	8/28/78	176	6	173	Rock	Annual	DBG
Apel	998	3497	11/21/92	178	6	n/a	Sand	Annual	PBG
Delaney	152	3497	8/25/99	301	6	265	Rock	Annual	PBG
bbul	862	3497		180	n/a	n/a	Sand	Annual	PBG
Krumenauer	875	3497	4/8/90	156	6	n/a	Sand	Annual	PBG
Mittenzwei	800	3497	- /··· /	131	n/a	n/a	Sand	Annual	PBG
PDS-3	911	3497	6/11/91	554	15	186	Rock	Annual	PBG
Ramaker-J	917	3497		310	n/a	n/a	Rock	Annual	PBG
Schlender	931	3497		280	n/a	n/a	Rock	Annual	PBG

<u>Notes</u> DBG = Deterrent Burning Ground Plume Central = Central Plume PBG = Propellant Burning Ground Plume

Well Name	Well ID	License	Date Installed	Well Depth (feet)	Well Diameter (inches)	Bedrock Depth (feet)	Aquifer
Anderson	804	3497		n/a	n/a	n/a	unknown
Andres	631	3497	2016	n/a	n/a	n/a	Sand
Andres	130	3497		n/a	n/a	n/a	unknown
Askey-1	178	3497	8/6/49	166	6	n/a	Sand
Askey-2	932	3497	3/12/77	256	6	200	Rock
Ballweg	131	3497		n/a	n/a	n/a	unknown
Bauer	807	3497	9/18/79	65	6	n/a	Sand
Behrens	197	3497		n/a	n/a	n/a	unknown
Bender	119	3497		n/a	n/a	n/a	unknown
Bickford-1	809	3497	2/13/67	187	18	n/a	Sand
Bickford-2	810	3497	2/13/67	152	18	n/a	Sand
Bickford-D	808	3497	6/30/65	152	18	n/a	Sand
Block	117	3497	7/26/01	101	6	n/a	Sand
Bluffview #1	813	3497	5/31/60	280	8	175	Rock
Bluffview #2	n/a	3497	1/1/60	n/a	8	n/a	Sand
Bluffview #3	n/a	3497	4/22/42	435	16	199	Rock
Brabender	171	3497		n/a	n/a	n/a	unknown
Bram	168	3497	1/1/94	n/a	n/a	n/a	unknown
Carlson	124	3497		n/a	n/a	n/a	unknown
Checky	132	3497		126	5	n/a	Sand
Christie	820	3497		n/a	n/a	n/a	unknown
Clark-M	821	3497		n/a	n/a	n/a	unknown
Clark-S	822	3497		n/a	n/a	n/a	unknown
Co-op County Partners	948	3497	6/20/88	276	6	n/a	Sand
Coves Court	147	3497		372	8	196	Rock
Cramer	825	3497		n/a	n/a	n/a	unknown
YR846	628	3497	10/18/16	120	5	n/a	Sand
Crow	160	3497		n/a	n/a	n/a	unknown
Dahir	827	3497		n/a	n/a	n/a	unknown
Danube	830	3497		n/a	n/a	n/a	unknown
Delaney-L	175	3497	10/1/74	263	6	225	Rock
Deppe	413	3497		n/a	n/a	n/a	Sand
Dischler-B	926	3497		n/a	n/a	n/a	unknown
Dorman	182	3497		n/a	n/a	n/a	unknown
Dybul	133	3497		n/a	n/a	n/a	Sand
Dyrud-Witte	907	3497	7/23/74	178	6	140	Rock
E12671	609	3497		n/a	n/a	n/a	Sand
E12680	611	3497		n/a	n/a	n/a	Sand
E12690A	613	3497		n/a	n/a	n/a	Sand
E12734	621	3497		n/a	n/a	n/a	Sand
E12742	622	3497		n/a	n/a	n/a	Sand
Eilertson-N	929	3497	10/17/69	120	6	n/a	Sand
Eilertson-S	834	3497	9/14/83	290	6	n/a	Rock

Well Name	Well ID	License	Date Installed	Well Depth (feet)	Well Diameter (inches)	Bedrock Depth (feet)	Aquifer
Elsing	194	3497	8/2/99	275	6	n/a	Sand
Emery	167	3497	1/1/91	n/a	n/a	n/a	unknown
Engh	184	3497	10/21/74	288	6	240	Rock
Fehn	121	3497		n/a	n/a	n/a	unknown
Fenbert	902	3497		n/a	n/a	n/a	unknown
Fentress	195	3497		n/a	n/a	n/a	unknown
Ferry	836	3497	10/12/44	178	6	n/a	Sand
Franks	134	3497		n/a	n/a	n/a	unknown
Ganser	187	3497	10/20/89	273	6	228	Rock
Gasner	113	3497		n/a	n/a	n/a	unknown
Gentz	865	3497		100	n/a	n/a	Sand
Gjertson	196	3497		n/a	n/a	n/a	unknown
Gleason	135	3497	8/8/92	242	5	n/a	Sand
Goelz	173	3497		n/a	n/a	n/a	unknown
Goette	845	3497	10/31/01	570	6	n/a	Rock
Greimel	841	3497		n/a	n/a	n/a	unknown
Grosse-garage	416	3497		n/a	n/a	n/a	Sand
Gruber-North	970	3497	8/25/75	240	6	170	Rock
Haasl	189	3497	11/29/73	270	6	245	Rock
Halweg-J	846	3497		n/a	n/a	n/a	unknown
Hankins	847	3497	9/16/08	221	6	206	Rock
Hannah	848	3497		n/a	n/a	n/a	unknown
Hanson	963	3497	4/9/90	307	6	278	Rock
Harpold	918	3497		206	6	198	Rock
Hasheider	852	3497	7/8/87	198	6	n/a	Sand
Heidenreich	853	3497	3/25/82	235	6	n/a	Rock
Henning	854	3497		n/a	n/a	n/a	unknown
Henry	855	3497	10/15/82	75	6	n/a	Sand
Herr	136	3497		191	6	n/a	Sand
Hill	137	3497		97	6	n/a	Sand
Hutter-R	857	3497		n/a	n/a	n/a	unknown
IA214	n/a	3497	10/31/94	625	6	169	Rock
Jackson	176	3497	12/7/95	255	6	225	Rock
Jacobson	185	3497		n/a	n/a	n/a	unknown
Jannenga	188	3497	7/17/74	272	6	230	Rock
Jewell	859	3497		n/a	n/a	n/a	unknown
Johnson	138	3497		102	6	n/a	Sand
Johnson-K	139	3497		150	6	n/a	Sand
Jonas	115	3497		n/a	n/a	n/a	unknown
Jones	889	3497		n/a	n/a	n/a	unknown
Kamps	863	3497	8/19/77	267	6	225	Rock
Kaufman/Schmitz	183	3497	10/3/76	263	6	237	Rock

Well Name	Well ID	License	Date Installed	Well Depth (feet)	Well Diameter (inches)	Bedrock Depth (feet)	Aquifer
Kindschi-1	867	3497	5/26/77	140	14	n/a	Sand
Kindschi-3	n/a	3497		n/a	n/a	n/a	Sand
Kindschi-3	868	3497	2/19/82	181	14	n/a	Sand
Kindschi-4	n/a	3497		n/a	n/a	n/a	Sand
Kindschi-A	866	3497		n/a	n/a	n/a	unknown
Kindschi-J	869	3497		n/a	n/a	n/a	unknown
Kindschi-V	870	3497		n/a	n/a	n/a	unknown
Kirner	843	3497	2/27/91	534	5	226	Rock
Klepper	140	3497		101	5	n/a	Sand
Kohlman	109	3497		n/a	n/a	n/a	unknown
Kowalke	181	3497		n/a	n/a	n/a	unknown
Kyori	826	3497		n/a	n/a	n/a	unknown
Lang	877	3497	6/12/97	325	6	202	Rock
Lautenbach	600	3497	8/7/18	80	6	n/a	Sand
Lenerz	193	3497	7/13/05	276	6	230	Rock
Lins-2	n/a	3497		162	n/a	n/a	Sand
Lins-4	n/a	3497		190	n/a	n/a	Sand
Lins-K	878	3497	3/21/96	288	6	248	Rock
Lins-R	879	3497		275	n/a	240	Rock
Lochner	880	3497		n/a	n/a	n/a	unknown
Lohr	881	3497		n/a	n/a	n/a	unknown
Lund	420	3497	8/7/06	100	5	n/a	Sand
Lytle	915	3497		n/a	n/a	n/a	unknown
Maple Park Condos	166	3497	9/24/64	270	8	165	Rock
Markgraf	885	3497	10/8/57	236	6	223	Rock
Maschman	120	3497	10/30/96	117	5	n/a	Sand
Matz-Gary	179	3497	8/11/77	248	6	210	Rock
Matz-Terry	886	3497	8/10/59	122	4	n/a	Sand
McAuliffe-J	887	3497	12/20/88	300	6	242	Rock
McClaren	890	3497		n/a	n/a	n/a	unknown
МсСоу	177	3497	6/8/81	249	6	n/a	Sand
Meier	953	3497	1/6/44	187	6	n/a	Sand
Mittenzwei-2	141	3497		n/a	n/a	n/a	unknown
MK967	n/a	3497	10/19/98	122	6	n/a	Sand
Moely-B	979	3497		100	n/a	n/a	Sand
Mohrbacher	142	3497		n/a	n/a	n/a	unknown
Mueller-A	896	3497	2/6/80	164	12	n/a	unknown
Mueller-C	897	3497	12/23/67	240	6	n/a	unknown
Mueller-J	899	3497	2/20/91	523	5	224	Rock
Mueller-S	895	3497		n/a	n/a	n/a	unknown
Mueller-SM	894	3497		n/a	n/a	n/a	unknown
Mullen	900	3497		n/a	n/a	n/a	unknown

Well Name	Well ID	License	Date Installed	Well Depth (feet)	Well Diameter (inches)	Bedrock Depth (feet)	Aquifer
Nelson	905	3497	2/28/74	280	6	250	Rock
Nelson-D	901	3497		179	6	n/a	Sand
Nolden	n/a	3497		198	n/a	n/a	Sand
Ohlsen	903	3497		n/a	n/a	n/a	unknown
Orbitec	324	3497	9/15/08	160	6	100	Rock
Paulson	123	3497		n/a	n/a	n/a	unknown
PDS Dam	961	3497	5/15/95	285	6	192	Rock
PDS-4	n/a	3497	3/23/12	580	30	n/a	Rock
Peetz	906	3497		n/a	n/a	n/a	unknown
Pierce	143	3497		n/a	n/a	n/a	unknown
Powell	833	3497	6/21/78	191	6	n/a	Sand
Premo	801	3497		122	n/a	n/a	Sand
Price	180	3497		n/a	n/a	n/a	unknown
Priebe	913	3497	4/15/57	76	9	n/a	Sand
Raetzke	940	3497	9/28/79	100	6	n/a	Sand
Ramaker	144	3497		82	6	n/a	Sand
Raschka	148	3497		120	5	n/a	Sand
Richards	118	3497		n/a	n/a	n/a	unknown
Riley	122	3497		n/a	n/a	n/a	unknown
Riley-M	145	3497		n/a	n/a	n/a	unknown
Robertson	640	3497		n/a	n/a	n/a	Sand
Rodgers	146	3497		140	5	n/a	Sand
Roth-G	924	3497	5/26/88	298	6	231	Rock
Roth-John	192	3497	9/26/88	298	6	250	Rock
Ruhland	927	3497		n/a	n/a	n/a	unknown
SC375	637	3497	6/22/04	142	6	n/a	Sand
SC388	610	3497	7/19/04	82	6	n/a	Sand
Schwarz	198	3497		n/a	n/a	n/a	unknown
Sereg	933	3497		175	n/a	n/a	Sand
Shimniok	934	3497		n/a	n/a	n/a	unknown
Sinklair-1	110	3497	4/10/00	130	6	n/a	Sand
Sinklair-2	111	3497		n/a	n/a	n/a	unknown
Sinklair-3	112	3497		n/a	n/a	n/a	unknown
SMD	172	3497		n/a	n/a	n/a	unknown
Smith	114	3497	6/13/96	35	2	n/a	Sand
Spurgeon	190	3497	9/16/76	256	6	215	Rock
Stensberg	162	3497	6/16/69	265	6	200	Rock
Stepenske	858	3497	9/13/86	150	6	n/a	Sand
Steuber	944	3497		n/a	n/a	n/a	unknown
Stima	942	3497		n/a	n/a	n/a	unknown
Stratton	943	3497		n/a	n/a	n/a	unknown
SU393	641	3497	4/5/05	129	6	n/a	Sand

Well Name	Well ID	License	Date Installed	Well Depth (feet)	Well Diameter (inches)	Bedrock Depth (feet)	Aquifer
Summer Oaks	945	3497	9/1/81	320	6	185	Rock
SWS2	n/a	3497		n/a	n/a	n/a	unknown
Tesch	947	3497		n/a	n/a	n/a	unknown
TG671	633	3497	1/26/06	141	6	n/a	Sand
TR267	615	3497	4/26/04	97	5	n/a	Sand
Troestler	186	3497	6/1/77	279	6	255	Rock
TS854	626	3497		100	5	n/a	Sand
Tschudy-Herman	950	3497		n/a	n/a	n/a	unknown
Tschudy-Herman 2	108	3497	5/5/99	79	6	n/a	Sand
TU541	638	3497	4/4/05	140	5	n/a	Sand
TU813	635	3497	5/17/05	120	5	n/a	Sand
TV887	604	3497	8/2/05	76	5	n/a	Sand
Unger	199	3497		n/a	n/a	n/a	unknown
Urban	161	3497	4/29/80	275	6	225	Rock
USDA 4	127	3497	3/15/94	273	6	238	Rock
Valley of Our Lady	954	3497		565	6	190	Rock
VM039	632	3497	2/22/06	120	5	n/a	Sand
Volker	952	3497	4/8/75	258	6	175	Rock
Wells	991	3497	5/23/89	108	6	n/a	Sand
Werderits	116	3497	10/2/90	67	6	n/a	Sand
Weum	802	3497	7/7/01	158	5	n/a	Sand
Weynand	939	3497		n/a	n/a	n/a	Sand
Wicklund	957	3497		n/a	n/a	n/a	unknown
Wiley	958	3497		n/a	n/a	n/a	unknown
Witte	962	3497		n/a	n/a	n/a	unknown
Woods	429	3497	9/12/98	86	6	n/a	Sand
WW440	n/a	3497	3/12/12	505	n/a	53	Rock
XE308	939	3497	6/14/13	35	1	n/a	Sand
XG515	618	3497	11/25/13	119	6	n/a	Sand
XG526	614	3497	3/6/14	120	6	n/a	Sand
XG527	639	3497	3/7/14	142	6	n/a	Sand
XI081	620	3497	12/3/13	99	5	n/a	Sand
XL970	617	3497	9/9/14	122	6	n/a	Sand
XP869	616	3497	8/5/15	118	6	n/a	Sand
XR620	602	3497	9/1/15	86	4	n/a	Sand
XT998	605	3497	2/23/16	71	4	n/a	Sand
XU003	606	3497	3/30/16	80	4	n/a	Sand
XW317	608	3497	7/20/16	92	4	n/a	Sand
XW533	627	3497	5/4/16	120	5	n/a	Sand
Yanke	191	3497	9/13/77	307	6	285	Rock
YF504	612	3497	3/31/11	100	5	n/a	Sand
YJ530	636	3497	7/21/13	120	5	n/a	Sand

Well Name	Well ID	License	Date Installed	Well Depth (feet)	Well Diameter (inches)	Bedrock Depth (feet)	Aquifer
YL970	645	3497	11/14/14	100	5	n/a	Sand
YM397	n/a	3497	5/12/15	180	6	n/a	Sand
YP340	623	3497	2/23/16	79	5	n/a	Sand
YQ555	630	3497	6/9/16	118	5	n/a	Sand
YR160	601	3497	8/17/16	100	5	n/a	Sand
YR845	629	3497	10/18/16	119	5	n/a	Sand
YR846	628	3497	10/18/16	120	5	n/a	Sand
Zander	849	3497	9/16/45	229	6	n/a	Sand
Zeck	964	3497	10/11/72	240	6	n/a	unknown
Zick	965	3497	8/26/97	141	6	n/a	Sand
Zick-2	125	3497	1/23/06	117	6	n/a	Sand
ZS447	619	3497	7/26/18	93	5	n/a	Sand
Zurbachen-D	968	3497		n/a	n/a	n/a	unknown

### 4.3 Well Identification and Designation

All sampled monitoring wells and residential wells are given a unique three-digit numeric well ID, i.e. 360. This well ID is used to track the well data in the on-site groundwater databases as well as the WDNR's on-line accessible Groundwater and Environmental Monitoring System (GEMS) database.

In general, groundwater monitoring wells are identified by a three-part alphanumeric code, i.e. PBN-1404B. The first two letters of the well identification are determined by the source area or waste management unit, i.e. BG, DB, EL, NL, NP, RI, PB, SE, and SP. The exception to this is the "S" series wells installed in the 1980s. The third letter determines if the well is part of a well nest "N" or a stand-alone water table monitoring well "M". The next two numbers determine what year the well was installed, i.e. 2010 = 10 or 2015 = 15. The last two numbers indicate the order that well was installed during that year, i.e. 05 is the fifth well installed that year for that source area. The last letter determines the vertical positioning of the well screen. Wells labeled "A" are screened at or near the water table surface. Wells labeled "B" are screened below the water table, approximately 1/3 of the depth between the water table and bedrock. Wells labeled "C" are screened below the water table, approximately 2/3 of the depth between the water table and bedrock. Wells labeled "D" are screened below the water table and just above the top of the bedrock. Wells labeled "E" are screened below the water table and below the top of the bedrock. Wells labeled "F" are screened below the confining layer of bedrock (shale) in a lower bedrock aquifer. The static groundwater level in an "F" well is higher than the water table and indicates an artesian condition. There are exceptions to the well depth labeling as some monitoring wells installed during the 1980s were drilled shallower than the 1/3 or 2/3 distance between the water table and bedrock.

### 4.4 Groundwater Properties

### 4.4.1 Water Level Elevation and Flow Direction

Water level data collected from BAAP monitoring wells indicate groundwater depths ranging from 22 to 144 feet bgs or 744 to 788 feet above mean sea level (MSL). Figure 16 is a representation of the groundwater elevation surface in September 2017. The groundwater contours shown in Figure 16 are drawn at 5-foot intervals. The groundwater flow direction is generally to the south-southeast. In the southeast corner of BAAP, groundwater flow is deflected slightly to the south, due to influences from Lake Wisconsin. Due to the large number of monitoring wells, the elevation measurements for a sampling round are taken within a 30-day period. Due to the groundwater being highly conductive, the groundwater table does not radically change after precipitation and snowmelt events. The Lake Wisconsin Reservoir, located to the east and southeast of BAAP, is formed by the WP&L dam, which results in a constant lake elevation of approximately 774 feet MSL. Below the dam, the water elevation drops abruptly to 736 feet MSL as the lake reverts to the flowing Wisconsin River. The rapid change in water elevations at the dam results in a dramatic hydraulic drop in groundwater elevations around the dam. Groundwater discharges to the Reservoir in the northeastern portion of BAAP. The Reservoir discharges to the sand and gravel aquifer when adjacent groundwater levels are lower than the Reservoir level. About three miles north of the WP&L dam, the

Reservoir transitions from recharging to discharging to the underlying sand and gravel aquifer. Directly south of the WP&L dam, the Wisconsin River resumes with groundwater discharging to the river.

Figure 17 depicts the groundwater contours near the PBG during September 2017. The groundwater contours shown in Figure 17 are drawn at 0.5-foot intervals. This small contour interval was chosen to show the variability in the groundwater surface. The engineered cap (geomembrane barrier and compacted clay) of the 1949 Pit and PBG Waste Pits influences the local groundwater flow. The engineered cap is diverting surface water from percolating into the subsurface. Depression contours 778 and 778.5 feet MSL are shown through the PBG Waste Pits. The groundwater contours shown in Figure 17 show that the cap is protecting the subsurface.

### 4.4.2 Hydraulic Conductivity

Hydraulic conductivity values were calculated based on aquifer testing at two former MIRM extraction wells located near the PBG in 2005. The aquifer tests, which were comprised of a pump test followed by a step test, were conducted at former extraction wells EW-169 in February 2005 and at EW-167 in March 2005. The tests were conducted by continuously pumping the extraction wells over a period of time and measuring the drawdown in nearby observation wells. Observation wells (PBN-8504A, PBM-8505, and PBM-8904C) were monitored for the test at EW-169, which lasted two- and one-half days. The aquifer test at EW-169 yielded a hydraulic conductivity value between  $1.39 \times 10^{-02}$  to  $6.27 \times 10^{-02}$  centimeters per second (cm/sec). The aquifer test at extraction wells (PBM-8503, PBN-8502A, PBN-8901C, and PBN-8902C). The results of this testing yielded a hydraulic conductivity value between  $4.85 \times 10^{-02}$  and  $9.60 \times 10^{-02}$  cm/sec. Testing methodology is presented in further detail in the *Draft Corrective Measures Implementation Report, MIRM Extraction Well Realignment Project* (Shaw Environmental, Inc., 2006).

During the RI (ABB-ES, 1993), slug tests were performed on monitoring wells across the BAAP. The 1993 RI report included hydraulic conductivity values for many monitoring wells. Table 7 summarizes the hydraulic conductivity data for each of the four groundwater plumes. The average calculated hydraulic conductivity of 25 monitoring wells in the PBG Plume was  $4.2 \times 10^{-02}$  cm/sec. The hydraulic conductivity values obtained during the MIRM pump tests correlated well with the average value obtained from the 1993 RI slug tests. The average calculated hydraulic conductivity of 17 monitoring wells in the DBG Plume was 2.5x10<sup>-02</sup> cm/sec. There was limited slug test data from the 1993 RI report for monitoring wells in the Central Plume. Slug test data was collected during 2010 from three monitoring wells associated with Landfill 3646 (Feasibility Report Contiguous Addition to Landfill 3646, SpecPro, Inc., October 2010). These three monitoring wells are located in northeast corner of the Central Plume (see Figure 18). The average calculated hydraulic conductivity of three monitoring wells in the Central Plume was  $3.7 \times 10^{-02}$  cm/sec. There is no available hydraulic conductivity data for monitoring wells associated with the NC Area Plume. Due to the similarities in soil types between the PBG Plume and NC Area Plume, the PBG Plume hydraulic conductivity value of  $4.2 \times 10^{-02}$  cm/sec is being used for the NC Area Plume.

# Table 7Field Hydraulic Conductivity Test ResultsRemedial Investigation/Feasibility StudyBadger Army Ammunition Plant

Plume Area	Well	Level Type	Hydraulic Conductivity (cm/sec)	Soil Type at Screen Interval	Reference
PBG	PBM-8911	А	$4x10^{-2}$	Sand	1993 RI
PBG	PBN-8203B	В	$1 \times 10^{-3}$	Sand	1993 RI
PBG	PBN-8203C	С	$7x10^{-4}$	Sand	1993 RI
PBG	PBN-8901B	В	$3x10^{-2}$	Gravel with sand	1993 RI
PBG	PBN-8901C	С	$3x10^{-2}$	Sand	1993 RI
PBG	PBN-8901D	D	$5 \times 10^{-2}$	Sand	1993 RI
PBG	PBN-8902B	В	$1 \times 10^{-2}$	Sand	1993 RI
PBG	PBN-8902C	С	$2x10^{-2}$	Sand	1993 RI
PBG	PBN-8903B	В	$1 \times 10^{-2}$	Sand	1993 RI
PBG	PBN-8903C	С	$4x10^{-2}$	Sand	1993 RI
PBG	PBN-8904C	С	$2x10^{-2}$	Sand	1993 RI
PBG	PBN-8910B	В	$2x10^{-1}$	Gravel with sand	1993 RI
PBG	PBN-8910C	С	$2x10^{-2}$	Sand	1993 RI
PBG	PBN-8910D	D	$5 \times 10^{-2}$	Sand with gravel	1993 RI
PBG	PBN-9106C	С	$2x10^{-2}$	Sand	1993 RI
PBG	PBN-9112C	С	8x10 <sup>-3</sup>	Sand	1993 RI
PBG	PBN-9112D	D	$3x10^{-2}$	Sand	1993 RI
PBG	LON-8902B	В	$4x10^{-2}$	Gravel with cobbles	1993 RI
PBG	LON-8903B	В	$1 \times 10^{-1}$	Sand and gravel	1993 RI
PBG	SPN-8901C	С	$4x10^{-2}$	Sand and gravel	1993 RI
PBG	SPN-8902B	В	$1 \times 10^{-2}$	Sand	1993 RI
PBG	SPN-8902C	С	$3x10^{-2}$	Sand	1993 RI
PBG	SPN-8903B	В	$4x10^{-2}$	Sand and gravel	1993 RI
PBG	SPN-8904B	В	$2x10^{-2}$	Sand and gravel	1993 RI
PBG	SPN-8904C	С	$2x10^{-2}$	Sand	1993 RI
Av	verage - PBG Plun	ne	4.2x10 <sup>-2</sup>		
DBG	DBM-8901	А	3x10 <sup>-2</sup>	Sand	1993 RI
DBG	DBN-8902A	А	$8 \times 10^{-2}$	Silt and clay	1993 RI
DBG	DBN-8904A	А	$3x10^{-2}$	Sand	1993 RI
DBG	DBN-8904B	В	$5x10^{-2}$	Gravel with sand	1993 RI
DBG	DBM-8905	А	$6x10^{-3}$	Sand	1993 RI
DBG	DBM-8201	А	$7x10^{-3}$	Silty clay	1993 RI
DBG	ELN-9107A	А	$5x10^{-3}$	Sand	1993 RI

# Table 7Field Hydraulic Conductivity Test ResultsRemedial Investigation/Feasibility StudyBadger Army Ammunition Plant

Plume Area	Well	Level Type	Hydraulic Conductivity (cm/sec)	Soil Type at Screen Interval	Reference
DBG	ELN-9107B	В	$2x10^{-2}$	Sand	1993 RI
DBG	ELM-9110	А	$2x10^{-2}$	Sand	1993 RI
DBG	ELM-8901	А	8x10 <sup>-3</sup>	Silty sand	1993 RI
DBG	ELN-8904A	А	$4x10^{-2}$	Sand	1993 RI
DBG	ELM-8905	А	$1 \times 10^{-2}$	Sand with gravel	1993 RI
DBG	ELM-8906B	В	$5 \times 10^{-2}$	Gravel and sand	1993 RI
DBG	ELM-8908	А	$4x10^{-2}$	Sand with gravel	1993 RI
DBG	ELM-8909	А	$3 \times 10^{-2}$	Sand	1993 RI
DBG	ELN-8203C	С	$6 \times 10^{-3}$	Sand	1993 RI
DBG	ELN-8204A	А	$3 \times 10^{-4}$	Silty sand	1993 RI
Av	erage - DBG Plur	ne	$2.5 \times 10^{-2}$		
Central	NLM-1001	А	1x10-2	Sand	Landfill 3646
Central	NLN-1001A	А	1x10-2	Sand	Landfill 3646
Central	NLN-1001C	С	6x10-2	Sand	Landfill 3646
Ave	rage - Central Plu	ime	$3.7 \times 10^{-2}$		

PBG - Propellant Burning Ground Plume

DBG - Deterrent Burning Ground Plume

Central - Central Plume

cm/sec - centimeters per second

Level Type - typical well depth configuration

1993 RI - Final Remedial Investigation Report (United States Army Environmental Center, April 1993)

Landfill 3646 - Feasibility Report Contiguous Addition to Landfill 3646 (SpecPro, Inc., October 2010)

There is no hydraulic conductivity data for the Nitrocelluose Production Area Plume wells; assume same value as PBG.

### 4.4.3 Hydraulic Gradient

Monitoring wells are screened at various depths and assigned an alphabetical designation after the number of the well ID. Letter designation A is the shallow water table interval, and B, C, D, E, and F are piezometric intervals that increase in depth from B to F. The piezometers ending in E were constructed so that the screen was located in the bedrock. It should be noted that the unconsolidated sand and gravel aquifer is unconfined vertically.

As evident from the groundwater elevation map showing the September 2017 data (Figure 16), the area south of BAAP has a much steeper horizontal hydraulic gradient than the area to the north. Data sets from each groundwater plume were used to calculate horizontal hydraulic gradient. Groundwater elevations from the sampling periods of September 2017, April 2018, and September 2018 were used to calculate an average hydraulic gradient for each plume area shown in Table 8. The average hydraulic gradient calculated for the PBG area wells was 0.00183 feet per foot (ft/ft). The average hydraulic gradient calculated for the DBG area wells was 0.00108 ft/ft. The hydraulic gradient calculated for the NC area wells was 0.00079 ft/ft.

Vertical groundwater movement is evaluated by comparing groundwater levels from the different aquifer layers to determine vertical gradient. Monitoring well clusters, where two or more wells have screens positioned at different depths within the aquifer, are used to examine differences in the potentiometric groundwater surface between different layers of the aquifer. Vertical hydraulic gradients were evaluated for nested well pairs in the four plume areas. Table 9 summarizes the vertical groundwater gradients for the chosen well nests. Gradients were evaluated from the groundwater elevation data collected during the September 2017, April 2018, and September 2018 monitoring events. Positive vertical gradients indicate groundwater is flowing upward and negative vertical gradients indicated groundwater is flowing downward.

Four of the seven well pairs in the PBG exhibited an upward vertical groundwater gradient between deep to shallow wells; which would hinder groundwater contaminants from migrating deeper into the sand aquifer. All the DBG well pairs exhibited a downward vertical groundwater gradient between shallow to deep wells; which would allow groundwater contaminants to migrate deeper into the sand aquifer. Three of the four well pairs in the Central Plume exhibited an upward vertical groundwater gradient between deep to shallow wells; which would hinder groundwater contaminants from migrating deeper into the sand aquifer. The two well pairs in the NC Area Plume exhibited an upward vertical groundwater gradient between deep to shallow wells; which would hinder groundwater contaminants from migrating deeper into the sand aquifer.

# Table 8Horizontal Groundwater GradientRemedial Investigation/Feasibility StudyBadger Army Ammunition Plant

Plume Area	Well Pair	Well Distance (ft)	Sept 2017 Well Elevation (ft msl)	Sept 2017 Elevation Difference (ft)	Sept 2017 Gradient (ft/ft)	Apr 2018 Well Elevation (ft msl)	Apr 2018 Elevation Difference (ft)	Apr 2018 Gradient (ft/ft)	Sept 2018 Well Elevation (ft msl)	Sept 2018 Elevation Difference (ft)	Sept 2018 Gradient (ft/ft)	Average Gradient (ft/ft)	Average Hydraulic Conductivity (cm/sec)	Average Hydraulic Conductivity (ft/day)	Effective Porosity	Average Groundwter Flow Velocity (ft/day)	Average Groundwter Flow Velocity (ft/yr)
	PBN-1401A	6 3 4 0	778.96	0.80	0.00156	780.05	9.96	0.00157	778.82	10.02	0.00158						
PRG	S1148	0,540	769.07	9.09	0.00150	770.09	9.90	0.00137	768.80	10.02	0.00138	0.00183	$4.2 \times 10^{-2}$	110	0.26	0.84	306
TDO	SPN-8904B	5 440	769.37	10.70	0.00108	770.00	11.81	0.00217	768.71	11 56	0.00213	0.00105	4.2810	117	0.20	0.04	500
	SWN-9103B	5,440	758.58	10.79	0.00198	758.19	11.01	0.00217	757.15	11.50	0.00215						
	DBM-8202	6 200	786.35	6.03	0.00097	787.13	7 10	0.00116	786.62	6 88	0.00111						
DBG	ELN-1003A	0,200	780.32	0.05	0.00097	779.94	7.19	0.00110	779.74	0.00	0.00111	0.00108	$2.5 \times 10^{-2}$	72	0.26	0.30	109
DDO	DBN-1001B	6.030	785.73	5.05	0 00000	786.29	6.03	0.00115	785.72	6 50	0.00100	0.00108	2.3x10	12	0.20	0.50	109
	ELN-1003B	0,050	779.78	5.95	0.00099	779.36	0.95	0.00115	779.13	0.59	0.00109						
	NPM-8901	6 630				783.04	5 46	0.00082									
Central	RIN-1002A	0,050				777.58	5.40	0.00082				0.00097	$2.7 \times 10^{-2}$	105	0.26	0.39	1/13
Central	RIN-1002A	8 260				777.58	0.20	0.00111				0.00097	5.7x10	105	0.20	0.39	145
	SEN-0503A	8,200				768.38	9.20	0.00111									
NC	RIM-1002	2 210	787.47	1.97	0.00085	788.52	1.66	0.00075	787.13	1 71	0.00077	0.00070	$4.2 \times 10^{-2}$	110	0.26	0.26	122
INC	RIN-1001A	2,210	785.60	1.87	0.00085	786.86	1.00	0.00075	785.42	1./1	0.00077	0.00079	4.2X10	119	0.20	0.50	152

ft - Feet

ft msl - Feet Mean Sea Level

ft/ft - Feet per Foot

Central Plume elevations were collected during June 2018

Groundwater flow velocity = (hydraulic conductivity)(hydraulic gradient)/effective porosity

Hydraulic conductivity conversion: 1 cm/sec = 2834 ft/day

# Table 9Vertical Groundwater GradientRemedial Investigation/Feasibility StudyBadger Army Ammunition Plant

	Woll Pair	Well	Lovor	Screen Midpoint	Groundw	ater Elevatio	on (ft msl)	Vertical Groundwater Gradient (ft/ft)						
		ID	Layer	Elevation	Sep-17	Apr-18	Sep-18	Sep-17	Apr-18	Sep-18	Average			
	PBN-1401A	782	А	759.90	778.96	780.05	778.82							
	PBN-1401B	783	В	723.39	778.98	780.07	778.83							
	PBN-1401C	784	C	683.78	778.94	780.05	778.82	-0.00026	0.00000	0.00000	-0.00009			
	PBN-8205A	622	Α	768.30	778.07	779.14	777.87							
	PBN-8205B	623	В	752.63	778.04	779.13	777.86							
	PBN-8205C	624	C	743.30	778.12	779.20	777.90	0.00200	0.00240	0.00120	0.00187			
	PBN-8502A	632	Α	762.21	776.65	777.73	776.49							
	PBN-8902BR	795	В	739.37	776.71	777.77	776.54							
ри	PBN-8902C	645	C	703.80	775.77	776.87	775.59	-0.01507	-0.01472	-0.01541	-0.01507			
uno	PBN-1404B	791	В	715.18	775.20	776.12	774.79							
Gr	PBN-1404C	792	C	655.34	775.14	776.06	774.72							
gu	PBN-1404D	793	D	594.89	774.38	775.29	773.98	-0.00682	-0.00690	-0.00673	-0.00682			
rni	S1148	710	A	757.90	769.07	770.09	768.80							
Bu	SPN-8904B	720	В	729.10	769.37	770.00	768.71							
ant	SPN-8904C	721	C	696.70	769.47	770.09	768.83							
ella	SPN-9104D	726	D	599.80	769.40	770.10	768.78	0.00209	0.00006	-0.00013	0.00067			
rop	PBN-9903A	692	A	756.68	767.56	768.12	766.84							
Ρ	PBN-9903B	693	В	715.50	768.02	768.59	767.30							
	PBN-9903C	694	C	664.49	768.08	768.65	767.35							
	PBN-9903D	695	D	619.60	768.07	768.61	767.31	0.00372	0.00357	0.00343	0.00357			
	SWN-9103B	571	В	726.30	758.58	758.19	757.15							
	SWN-9103C	572	C	676.80	758.73	758.36	757.30							
	SWN-9103D	573	D	630.90	758.56	758.18	757.15							
	SWN-9103E	574	E	602.10	758.67	758.30	757.21	0.00072	0.00089	0.00048	0.00070			
	PBN-9101C	561	C	680.50	744.75	744.33	744.40							
	PBM-9001D	981	D	623.50	745.03	744.60	744.66	0.00491	0.00474	0.00456	0.00474			
	DBM-8202	302	A	770.45	786.35	787.13	786.62							
	DBN-1001B	472	В	752.77	785.73	786.29	785.72							
	DBN-1001C	473	C	715.28	784.11	783.97	783.36							
	DBN-1001E	474	E	632.55	784.54	784.38	783.71	-0.01313	-0.01994	-0.02110	-0.01806			
	DBN-9501A	314	A	771.70	783.49	783.38	782.79							
pu	DBN-9501B	315	В	719.50	783.51	783.38	782.76							
nou	DBN-9501C	316	C	664.00	783.49	783.36	782.78							
Ģ	DBN-9501E	317	E	637.55	783.38	783.26	782.66	-0.00082	-0.00089	-0.00097	-0.00089			
ing	ELN-8203A	210	A	772.70	783.79	783.64	783.00							
urn	ELN-8203B	211	В	760.50	783.43	783.12	782.52							
t B	ELN-8203C	212	C	750.30	783.46	783.14	782.54	-0.01473	-0.02232	-0.02054	-0.01920			
ren	ELM-9501	234	Α	779.20	781.23	780.88	780.36							
teri	ELN-0801B	455	В	738.87	781.33	781.05	780.49							
De	ELN-0801C	456	C	693.42	781.38	781.04	780.54							
	ELN-0801E	457	E	634.93	780.25	780.98	780.44	-0.00679	0.00069	0.00055	-0.00185			
	ELN-1003A	467	Α	776.19	780.32	779.94	779.74							
	ELN-1003B	468	В	704.74	779.78	779.36	779.13							
	ELN-1003C	469	С	641.64	779.92	779.50	779.23							
	ELN-1003E	470	Е	571.02	779.50	779.09	778.83	-0.00400	-0.00414	-0.00444	-0.00419			

# Table 9Vertical Groundwater GradientRemedial Investigation/Feasibility StudyBadger Army Ammunition Plant

	Woll Dair	Well	Lovor	Screen Midpoint	Groundw	ater Elevati	on (ft msl)	Vertical Groundwater Gradient (ft/ft)							
	wen i an	ID	Layer	Elevation	Sep-17	Apr-18	Sep-18	Sep-17	Apr-18	Sep-18	Average				
	RIN-1002A	492	Α	775.76		777.58									
	RIN-1002C	493	C	683.56	no data	777.57	no data								
	RIN-1501D	540	D	625.56		777.54			-0.00027		-0.00027				
al Plume	RIN-1005A	496	A	773.74	no doto	775.19	no doto								
	RIN-1005C	497	C	681.99	no data	775.24	no data		0.00054		0.00054				
	SEN-0501A	580	Α	760.14		767.37									
ntr	SEN-0501B	581	В	702.87	no data	767.51	no data								
Ce	SEN-0501D	582	D	600.22		767.77			0.00250		0.00250				
	SEN-0503A	585	Α	761.63		768.38									
	SEN-0503B	586	В	704.39	no data	768.61	no data								
	SEN-0503D	587	D	601.31		768.67			0.00181		0.00181				
ne	RIM-0705	442	Α	782.80	786.26	no doto	786.04								
lun	RIN-1007C	479	C	708.61	786.30	no data	786.06	0.00054		0.00027	0.00040				
СР	RIN-1001A	480	Α	782.75	785.60	no doto	785.42								
Ŋ	RIN-1001C	481	C	703.10	785.61	no uata	785.41	0.00013		-0.00013	0.00000				

Layer designation

A = shallow zone in sand and gravel aquifer

B = intermediate zone in sand and gravel aquifer

C = deep zone in sand and gravel aquifer

D = bottom zone in sand and gravel aquifer

E = top of bedrock aquifer

ft msl - Feet Mean Sea Level

ft/ft - Feet per Foot

Central Plume elevations were collected during June 2018

Gradient determined between shallow and deep well for each well cluster

Vertical Groundwater Gradient = (h2 - h1) / (z1 - z2)

h1 = shallow well groundwater elevation

h2 = deep well groundwater elevation

z1 = shallow well screen midpoint elevation

z2 = deep well screen midpoint elevation

### 4.4.4 Groundwater Flow Velocity

The advective groundwater flow velocity is derived from the hydraulic conductivity value, horizontal gradient, and effective porosity. Advective groundwater movement does not take into account dispersion, diffusion, or chemical retardation of groundwater contaminants, which can increase or decrease the rate of groundwater flow. It is a calculated value that provides an estimate of the rate of groundwater flow over time. The mathematical formula for determining advective groundwater flow velocity (v) is:

$$v = Ki/n_e$$
 Where:

K = hydraulic conductivity (feet/day) *i* = hydraulic gradient (feet/feet) n<sub>e</sub> = effective porosity

The average hydraulic conductivity values found in Table 7 were used in the groundwater flow velocity calculations. The average hydraulic conductivities for the PBG, DBG, Central, and NC Area Plumes are  $4.2 \times 10^{-02}$  cm/sec or 119 ft/day,  $2.5 \times 10^{-02}$  cm/sec or 72 ft/day,  $3.7 \times 10^{-02}$  cm/sec or 105 ft/day, and  $4.2 \times 10^{-02}$  cm/sec or 119 ft/day, respectively.

The effective porosity is estimated at 0.26 or 26%. Average horizontal gradients of 0.00183 ft/ft for the PBG, 0.00108 ft/ft for the DBG, 0.00097 ft/ft for the Central Plume, and 0.00079 ft/ft for the NC Area Plume were used to calculate the groundwater flow velocities.

The calculated average groundwater flow velocities as shown in Table 8 equal 0.84 ft/day for the PBG, 0.30 ft/day for the DBG, 0.39 ft/day for the Central, and 0.36 ft/day for the NC Area. These groundwater flow velocity values equate to 306 ft/year for the PBG Plume, 109 ft/year for the DBG Plume, 143 ft/year for the Central Plume, and 132 ft/year for the NC Area Plume.

### 4.5 Nature and Extent of Groundwater Contamination

Groundwater investigation activities at BAAP began in 1980 and continue today. Site-wide groundwater-related assessment activities, agreed upon by the Army and WDNR, include the following: soil vapor surveys; monitoring well drilling, installation, and surveying; water level measurements; pump testing; and monitoring well and residential well sampling.

The groundwater sampling results were compared to the Wisconsin NR 140 PAL Groundwater Standards to identify contaminants of potential concern (COPCs) for the four groundwater plumes at BAAP. The COPCs for each groundwater plume are further discussed in the following sections.

### 4.5.1 Propellant Burning Ground Plume

Groundwater contamination in monitoring wells associated with the PBG was first detected in 1982 (Tsai, 1988). The draft final (Phase 1) RI report (January 1990) indicated that groundwater contamination had migrated beyond the southern BAAP boundary. An off-site groundwater monitoring program was initiated in January 1990. In late April 1990, sampling results from

residential wells south of BAAP showed that two residential wells had been contaminated with CTET and one residential well contaminated with chloroform. The maximum concentrations of CTET and chloroform in the residential wells were 80 micrograms per liter ( $\mu$ g/l) and 9.9  $\mu$ g/l, respectively. It was determined that a VOC plume (PBG Plume) had migrated south from the PBG Waste Pits, past the BAAP's southern boundary, and then easterly to the Wisconsin River below the WP&L dam. The Army replaced the three impacted residential wells. Prior to well replacement, bottled water had been provided to the affected residences.

The PBG Plume originates at the PBG and extends south beyond the BAAP boundary. South of BAAP, the plume turns southeast towards the Wisconsin River due to the influence of the WP&L dam, just north of Prairie du Sac. The PBG Plume shown in Figure 21 represents the area where groundwater concentrations exceed a NR 140 PAL or ES for one or more of the following compounds: CTET, ethyl ether (diethyl ether), TCE, 2,4-DNT, 2,6-DNT, or total DNT. All six DNT isomers (2,3-DNT, 2,4-DNT, 2,5-DNT, 2,6-DNT, 3,4-DNT, and 3,5-DNT) have been detected in the PBG Plume, mostly in the PBG Waste Pits. The PBG Plume boundaries shown in Figure 21 are approximate and based on total DNT and VOC groundwater data collected during 2018 from both monitoring wells and residential wells. Table 10 summarizes the groundwater analytical results from the August 2018 residential well sampling event. Table 11 summarizes the groundwater analytical results from the September 2018 monitoring well sampling event. Isoconcentration maps and cross sections were prepared for CTET, ethyl ether, TCE, and total DNT. The isoconcentration maps were prepared using all groundwater data collected during 2018. The isoconcentration cross sections were prepared mainly using groundwater data collected during August and September 2018. Supplemental groundwater data from November 2014 was used to complete the isoconcentration cross sections. These contaminants, CTET, ethyl ether, TCE, and total DNT, have shown consistent exceedances of the NR 140 ES in multiple monitoring wells to facilitate the construction of isoconcentration maps.

During 2015, 2016, 2017, and 2018, bromodichloromethane, CTET, chloroform, 2,4-DNT, 2,6-DNT, total DNT, ethyl ether, nitrate, and TCE have been COPCs in the PBG Plume. The PBG Plume groundwater results from the August and September 2018 sampling events were evaluated for COPCs that exceeded a Chapter NR 140 PAL Groundwater Standard.

Three monitoring wells had NR 140 PAL exceedances for bromodichloromethane during September 2018. Bromodichloromethane was not detected in any residential wells that were sampled during August 2018.

A total of six monitoring wells had NR 140 ES exceedances for CTET during September 2018. In addition, thirty-one monitoring wells had NR 140 PAL exceedances for CTET during September 2018. Three residential wells (Apel, Krumenauer, and Schlender) had CTET detections that were below the NR 140 PAL during August 2018. Since 2010, CTET has been detected in these three residential wells, that are located east of the PBG Plume (see Figure 20).

A total of ten monitoring wells had NR 140 PAL exceedances for chloroform during September 2018. One residential well had a chloroform detection that was below the NR 140 PAL during August 2018.

#### Table 10 Residential Well Groundwater Analytical Results August 2018 Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

										All	l results	are exp	ressed a	as µg/l (I	nicrogra	ms per l	iter)							
AdderPicture <th< th=""><th colspan="3">August '18 Round         Level of Detection         Level of Quantitation           2.3-DNT         0.0057         0.029           2.4-DNT         0.0076         0.029           2.5-DNT         0.0029         0.029           2.6-DNT         0.0038         0.029           3.4-DNT         0.0038         0.029           3.5-DNT         0.0038         0.029           *Level of detection and level of quantitation may change each round.         ************************************</th><th colspan="4"><pre>     = Under PAL and ES     = Over Preventive Action Limit (PAL)     = Over Enforcement Standard (ES)     = No PAL or ES established     = Not Tested     ND = Compound was not detected     Sample     Date     Date</pre></th><th>Dichlorodifluoromethane</th><th>Chloromethane</th><th>Chloroform</th><th>Carbon Tetrachloride</th><th>Trichloroethene</th><th>Ethyl Ether</th><th>1,1,1-Trichloroethane</th><th>1,1,2-Trichloroethane</th><th>2,6-Dinitrotoluene</th><th>2,4-Dinitrotoluene</th><th>2,3-Dinitrotoluene</th><th>3,4-Dinitrotoluene</th><th>2,5-Dinitrotoluene</th><th>3,5-Dinitrotoluene</th><th>Dinitrotoluene, Total</th></th<>	August '18 Round         Level of Detection         Level of Quantitation           2.3-DNT         0.0057         0.029           2.4-DNT         0.0076         0.029           2.5-DNT         0.0029         0.029           2.6-DNT         0.0038         0.029           3.4-DNT         0.0038         0.029           3.5-DNT         0.0038         0.029           *Level of detection and level of quantitation may change each round.         ************************************			<pre>     = Under PAL and ES     = Over Preventive Action Limit (PAL)     = Over Enforcement Standard (ES)     = No PAL or ES established     = Not Tested     ND = Compound was not detected     Sample     Date     Date</pre>				Dichlorodifluoromethane	Chloromethane	Chloroform	Carbon Tetrachloride	Trichloroethene	Ethyl Ether	1,1,1-Trichloroethane	1,1,2-Trichloroethane	2,6-Dinitrotoluene	2,4-Dinitrotoluene	2,3-Dinitrotoluene	3,4-Dinitrotoluene	2,5-Dinitrotoluene	3,5-Dinitrotoluene	Dinitrotoluene, Total		
by by by by consing498consing598809809800<	Anderson	411	Anderson-R		CT Lab	8/21/2018	ND	ND	ND	ND	ND	0.22	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Conversion442 ConversionConversionCT also972/00 MONO <th>Apel</th> <th>998</th> <th>Apel</th> <th></th> <th>CT Lab</th> <th>8/20/2018</th> <th>ND</th> <th>ND</th> <th>ND</th> <th>ND</th> <th>0.18</th> <th>ND</th>	Apel	998	Apel		CT Lab	8/20/2018	ND	ND	ND	ND	0.18	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Oardo912OardoSeture9142OardoSeture9142OardoNo	Cornelius	426	Cornelius		CT Lab	8/21/2018	ND	ND	0.11	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Destanger1542ObtangerObtangerChab62200*ND	Curto	412	Curto	Nimmow	CT Lab	8/21/2018	ND	0.17	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
BasisBasi	Delaney	152	Delaney		CT Lab	8/20/2018	ND	ND	ND	ND	ND	ND	0.16	ND	ND	ND	ND	ND	ND	ND	ND	ND		
OrestarOrestarOrestarOrestarOrestarOrestarOrestarNo<	Gibbs	839	Gibbs		CT Lab	8/22/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
SinchSinchSinchSinchNo </th <th>Grosse</th> <th>415</th> <th>Grosse</th> <th></th> <th>CT Lab</th> <th>8/21/2018</th> <th>ND</th>	Grosse	415	Grosse		CT Lab	8/21/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
condercond	Groth	842	Groth		CT Lab	8/21/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
GalesGrade					CT Lab (D)	8/21/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Index <th< th=""><th>Gruber</th><th>417</th><th>Gruber-D</th><th></th><th>CT Lab</th><th>8/20/2018</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th></th<>	Gruber	417	Gruber-D		CT Lab	8/20/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
individeindiv <d table="">AdditionalAdditi</d>	Hendershot	418	Hendershot		CT Lab	8/21/2018	ND	ND	ND	ND	ND	2	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
JuidGradGradCrubCrubCrubRoNo </th <th>Howery</th> <th>419</th> <th>Howery</th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th>Р</th> <th colspan="12">mp not working; well not sampled</th>	Howery	419	Howery									Р	mp not working; well not sampled											
MoreaMileOFTLADSP20718ND	Judd	862	Judd		CT Lab	8/20/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Krunenaer6775Krunenaer6716877870 <th< th=""><th>Kopras</th><th>874</th><th>Kopras</th><th>Miller</th><th>CT Lab</th><th>8/20/2018</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th></th<>	Kopras	874	Kopras	Miller	CT Lab	8/20/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Lukeens680LukeensCPL Lab872/10ND	Krumenauer	875	Krumenauer		CT Lab	8/21/2018	ND	ND	ND	ND	0.16	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Methum423MethumMethumCT Lab87/21018No </th <th>Lukens</th> <th>860</th> <th>Lukens</th> <th></th> <th>CT Lab</th> <th>8/21/2018</th> <th>ND</th>	Lukens	860	Lukens		CT Lab	8/21/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Mitterzwei       600       Mitterzwei       Cr       Cr       Cr       Vol       No       No <th< th=""><th>Melum</th><th>423</th><th>Melum</th><th></th><th>CT Lab</th><th>8/21/2018</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th></th<>	Melum	423	Melum		CT Lab	8/21/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Newstarki         B0         Novetarki         Novetark         Novetark         Novetar	Mittenzwei	800	Mittenzwei		CT Lab	8/20/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Olah         964         Olah         Image of the set of th	Nowotarski	891	Nowotarski										Not ava	ailable;	well not	sampled	ł							
Oster land         422         Oster land         Oster land         Otic land         Otic land         Otic land         No         <	Olah	904	Olah		CT Lab	8/21/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Peckoh         Perkosh         CTLab         8/21/201         ND         ND <th>Osterland</th> <th>422</th> <th>Osterland</th> <th></th> <th>CT Lab</th> <th>8/21/2018</th> <th>ND</th>	Osterland	422	Osterland		CT Lab	8/21/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Priaire data Su Utilities       911       PDS-3       CT Lab       62/12/08       ND       ND <t< th=""><th>Peckosh</th><th>817</th><th>Peckosh</th><th></th><th>CT Lab</th><th>8/21/2018</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th></t<>	Peckosh	817	Peckosh		CT Lab	8/21/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Purcelic       163       Purcelic       CT Lab       8202078       ND       ND      ND       ND      <	Prairie du Sac Utilities	911	PDS-3		CT Lab	8/21/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Pricel         916         Purcell-G         Purcell-G         CT Lab         82/2018         ND         ND        ND        <	Purcell	163	Purcell-D		CT Lab	8/20/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Purcelled916Purcell-GPurcell-GCT Lab82/20/18ND					CT Lab (D)	8/20/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Ramager       G1 and Mer       G1 and Mer       G1 and Mer       G1 and Mer       ND       ND <t< th=""><th>Purcell</th><th>916</th><th>Purcell-G</th><th></th><th></th><th>8/20/2018</th><th>ND</th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th>ND</th><th>ND</th><th></th><th></th><th></th><th></th><th></th></t<>	Purcell	916	Purcell-G			8/20/2018	ND									ND	ND							
Reschering         Rescher	Ramaker	917	Ramaker-J			8/20/2018	ND																	
NeiAlgNeiNe	Raschein	424	Raschein		CTLab	0/21/2010	ND				ND													
Revers425ReversReversCT Lab8/2/2018ND <th>Reli</th> <th>427</th> <th>Reil</th> <th></th> <th>CT Lab (D)</th> <th>8/22/2018</th> <th>ND</th> <th>ND</th> <th></th> <th>ND</th> <th></th> <th>ND</th> <th></th>	Reli	427	Reil		CT Lab (D)	8/22/2018	ND	ND		ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		ND			
NormalizationNormal	Revers	425	Revers		CT Lab	8/22/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
SchunderSchunderNoteng, BainlegOf LabOf LabOf DationND <th< th=""><th>Schlender</th><th>920</th><th>Schlender</th><th>Koopig Ballwog</th><th>CT Lab</th><th>8/20/2018</th><th>ND</th><th>ND</th><th></th><th>0.13</th><th>0.48</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th></th><th>ND</th><th></th><th>ND</th><th></th><th>ND</th></th<>	Schlender	920	Schlender	Koopig Ballwog	CT Lab	8/20/2018	ND	ND		0.13	0.48	ND	ND	ND	ND	ND		ND		ND		ND		
Spear803SpearMilligmann, LaytonCT Lab8/20/2018ND <th>Schumann</th> <th>428</th> <th>Schumann</th> <th></th> <th>CT Lab</th> <th>8/22/2018</th> <th>ND</th>	Schumann	428	Schumann		CT Lab	8/22/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Water's Edge Group158WE-QN039Hilgemann, LaytonCT Lab87/2018ND<	Spear	803	Spear	1	CT Lab	8/20/2018	ND	ND	ND	ND	ND	ND	ND	ND	0.2	ND								
Water's Edge Group157WE-QR441Hemberger, Pattarozzi, HeathCT Lab87/2018NDNDO.1ND	Water's Edge Group	158	WE-QN039	Hilgemann, Lavton	CT Lab	8/7/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Water's Edge Group159WE-RD430Ford, Madden, Bastien/EddyCT Lab8/7/2018MMM<	Water's Edge Group	157	WE-QR441	Hemberger, Pattarozzi, Heath	CT Lab	8/7/2018	ND	ND	ND	0.1	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
Water's Edge Group         153         WE-RM383         Good, Rossing         CT Lab         8/7/2018         ND         ND <th< th=""><th>Water's Edge Group</th><th>159</th><th>WE-RD430</th><th>Ford, Madden, Bastien/Eddy</th><th>CT Lab</th><th>8/7/2018</th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th>-</th><th></th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th><th>ND</th></th<>	Water's Edge Group	159	WE-RD430	Ford, Madden, Bastien/Eddy	CT Lab	8/7/2018								-		ND								
Water's Edge Group         164         WE-SQ017         Thompson         CT Lab         8/7/2018         ND         ND         1.7         0.12         ND         N	Water's Edge Group	153	WE-RM383	Good, Rossing	CT Lab	8/7/2018										ND								
Water's Edge Group 165 WE-SQ001 Rosenau, Schwarz CT Lab 8/7/2018 ND	Water's Edge Group	164	WE-SQ017	Thompson	CT Lab	8/7/2018	ND	ND	ND	1.7	0.12	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		
	Water's Edge Group	165	WE-SQ001	Rosenau, Schwarz	CT Lab	8/7/2018	ND	ND	ND	1.4	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND		

#### Table 10 Residential Well Groundwater Analytical Results August 2018 Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

						All results are expressed as µg/l (micrograms per liter)															
August '18         Round         Level of Detectic           2.3-DNT         0.0057           2.4-DNT         0.0076           2.5-DNT         0.0029           2.6-DNT         0.0038           3.4-DNT         0.0038           3.5-DNT         0.0038           'Level of detection and level of quantita	n <u>Level of C</u> 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	20antitation 129 129 129 129 129 129 129 129 h round.	<ul> <li>= Under PAL and ES</li> <li>= Over Preventive Action Limit (PAL)</li> <li>= Over Enforcement Standard (ES)</li> <li>= No PAL or ES established</li> <li>= Not Tested</li> <li>ND</li> <li>= Compound was not detected</li> </ul>			luene	chlorodifluoromethane	loromethane	loroform	arbon Tetrachloride	ichloroethene	hyl Ether	1,1-Trichloroethane	1,2-Trichloroethane	6-Dinitrotoluene	4-Dinitrotoluene	3-Dinitrotoluene	4-Dinitrotoluene	5-Dinitrotoluene	5-Dinitrotoluene	nitrotoluene, Total
Last Name	Well No.	Well Name	<u>Shared With</u>	Analyzed By	<u>Date</u>	ř	ā	Ū	Ū	Ű	F	Ш	<del>,</del>	7	3.	'n.	3.	З,	2,	з,	ā
Water's Edge Group	156	WE-RR542	Cairnes, Sherpe	CT Lab	8/7/2018										ND						
Water's Edge Group	169	WE-RR598	Hall, Chow, Hartmann, Wenger	CT Lab	8/7/2018										ND						
Water's Edge Group	170	WE-SQ002	Neumaier, Ramaker	CT Lab	8/7/2018										ND						
Water's Edge Group	174	WE-TF023	Hilgemann	CT Lab	8/7/2018										ND						
Water's Edge Group	129	WE-TM599	Riordan	CT Lab	8/7/2018										ND						
Water's Edge Group	431	WE-UK125	Gust, Haag, Lochner	CT Lab	8/7/2018										ND						
Water's Edge Group	432	WE-UK124	Whalen	CT Lab	8/7/2018										ND						
Water's Edge Group	433	WE-UA297	Krisko	CT Lab	8/7/2018										ND						
Water's Edge Group	434	WE-XD828	Riethmiller	CT Lab	8/7/2018										ND						
Water's Edge Group	435	WE-XK342	Brandherm	CT Lab	8/7/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND
				CT Lab (D)	8/7/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND
Water's Edge Group	436	WE-YW972	Dietzen	CT Lab	8/7/2018	0.12	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND
Wenger	414	Wenger		CT Lab	8/20/2018	ND	ND	ND	ND	ND	1.4	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND
Zurbachen	967	Zurbachen-A		CT Lab	8/22/2018	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND
Dairy Forage Res Ctr	828	USDA 1		CT Lab	8/21/2018										ND						
Dairy Forage Res Ctr	829	USDA 2		CT Lab	8/21/2018										ND						
Dairy Forage Res Ctr	126	USDA 3		CT Lab	8/21/2018										ND						
Dairy Forage Res Ctr	128	USDA 6		CT Lab	8/22/2018										ND						

(D) = Duplicate CT Lab = CT Laboratories, LLC

# Table 11Groundwater Analytical ResultsSeptember 2018Remedial Investigation/Feasibility StudyBadger Army Ammunition Plant

						Volatile Organic Compounds (VOC) - SW8260C											
Well	Well ID	Sample Level	Sample Depth (feet)	Sample Date	Dinitrotoluene, Total	1,1-Dichloroethane	1,1-Dichloroethene	1,1,1-Trichloroethane	Bromodichloromethane	Carbon Disulfide	Carbon Tetrachloride	Chloroform	Chloromethane	Ethyl Ether	Tetrachloroethene	Trichloroethene	Nitrate, Total
DBG PLUME AREA WELLS																	
DBM-8201	301	А	154.6-174.6	Sep-18	3.005	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
DBM-8202	302	А	137.3-157.3	Sep-18	0.608	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
DBM-8202 (dup)	302	А	137.3-157.3	Sep-18	0.562	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
DBM-8903	306	Α	113-133	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
DBN-1001B	472	В	154.5-159.5	Sep-18	0.5978	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
DBN-1001C	473	С	192-197	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
DBN-1001E	474	Е	274.9-279.9	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
DBN-1002C	476	С	205.1-210.1	Sep-18	0.7705	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
DBN-1002E	477	Е	275.5-280.5	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
DBN-9501A	314	А	110-120	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
DBN-9501B	315	В	162.5-172.5	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
DBN-9501C	316	С	218.5-228.5	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
DBN-9501E	317	Е	245.2-255.5	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELM-8901	216	Α	145.5-165	Sep-18	1.427	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELM-8907	220	Α	130.3-150.3	Sep-18	0.668	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELM-8908	221	А	125-145	Sep-18	0.263	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELM-8909	222	А	135-155	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELM-9501	234	А	54-69	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-0801B	455	В	100-105	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-0801C	456	C	145.5-150.5	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-0801E	457	Е	202.6-207.6	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-0802A	458	A	92.5-107.5	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-0802C	459	C	175.8-180.8	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1001B	460	В	91.1-96.1	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1001C	461	C	155.2-160.2	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
									Volati	le Organic C	Compounds (	(VOC) - SW	8260C				
-----------------	---------	-----------------	------------------------	----------------	-----------------------	--------------------	--------------------	-----------------------	----------------------	------------------	----------------------	------------	---------------	-------------	-------------------	-----------------	----------------
Well	Well ID	Sample Level	Sample Depth (feet)	Sample Date	Dinitrotoluene, Total	1,1-Dichloroethane	1,1-Dichloroethene	1,1,1-Trichloroethane	Bromodichloromethane	Carbon Disulfide	Carbon Tetrachloride	Chloroform	Chloromethane	Ethyl Ether	Tetrachloroethene	Trichloroethene	Nitrate, Total
ELN-1001E	462	Е	240.5-245.5	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1002A	463	А	55.3-70.3	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1002B	464	В	111.2-116.2	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1002B (dup)	464	В	111.2-116.2	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1002C	465	C	159.1-164.1	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1002E	466	Е	231.5-236.5	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1003A	467	А	16.2-31.2	Oct-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1003B	468	В	91.5-96.5	Oct-18	0.192	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1003B (dup)	468	В	91.5-96.5	Oct-18	0.171	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1003C	469	С	155.1-160.1	Oct-18	0.1327	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1003E	470	Е	255.6-230.6	Oct-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1502A	533	А	115.3-130.3	Sep-18	0.627	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1502A (dup)	533	А	115.3-130.3	Sep-18	0.801	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1502C	534	С	198-203	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1503A	535	А	73.7-88.7	Oct-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1503C	536	C	157.6-162.6	Oct-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-1504B	537	В	34.8-39.8	Oct-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-8203A	210	А	147.5-157.5	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-8203B	211	В	164-166	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-8203C	212	C	174-176	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-8902B	224	В	173.5-178.5	Oct-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-9107A	227	Α	116-126	Sep-18	<0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-9107B	228	В	135-145	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
ELN-9402AR	231	A	130-145	Oct-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
S1121	755	A	39.11-59.3	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
S1134R	236	Α	136-151	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT

					Volatile Organic Compounds (VOC) - SW8260C												
Well	Well ID	Sample Level	Sample Depth (feet)	Sample Date	Dinitrotoluene, Total	1,1-Dichloroethane	1,1-Dichloroethene	1,1,1-Trichloroethane	Bromodichloromethane	Carbon Disulfide	Carbon Tetrachloride	Chloroform	Chloromethane	Ethyl Ether	Tetrachloroethene	Trichloroethene	Nitrate, Total
NC PRODUCTION	N PLUME A	AREA WEI	LLS	-			-	-				-		-	-		
RIM-0703	440	А	98-113	Sep-18	0.029 (J)	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
RIM-0705	442	А	91-106	Sep-18	0.089	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
RIM-1002	478	А	95.2-110.2	Sep-18	0.21	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
RIM-1002 (dup)	478	А	95.2-110.2	Sep-18	0.22	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
RIN-1001A	480	А	91.8-106.8	Sep-18	0.073	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
RIN-1001C	481	С	176.41-181.41	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
RIN-1007C	479	С	170.3-175.3	Sep-18	< 0.008	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
S1125	504	А	106.25-126.5	Sep-18	0.0061(J)	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT	NT
PBG PLUME ARE	A WELLS																
PBM-0001	367	А	109.5-134.5	Sep-18	12.98	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	<b>0.14</b> (J)	<0.1	< 0.1	<0.1	< 0.1	0.31	3.6
PBM-0002	368	Α	106.5-131.5	Sep-18	2.33	< 0.1	<0.1	<0.1	<0.1	< 0.2	0.29	<0.1	< 0.1	<0.1	< 0.1	0.69	3.4
PBM-0006	372	А	99.5-124.5	Sep-18	1.841	< 0.1	<0.1	< 0.1	<0.1	< 0.2	0.34	< 0.1	< 0.1	<0.1	< 0.1	0.73	3.1
PBM-0008	374	А	97-122	Sep-18	1.603	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	<b>0.19</b> (J)	< 0.1	< 0.1	< 0.1	< 0.1	0.41	NT
PBM-1201	764	А	103.5-118.5	Sep-18	23.66	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	0.25	< 0.1	< 0.1	< 0.1	< 0.1	0.65	NT
PBM-1202	765	А	103.5-118.5	Sep-18	3.45	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	0.34	< 0.1	< 0.1	< 0.1	< 0.1	0.8	NT
PBM-1203	766	А	103.4-118.4	Sep-18	0.201	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	0.25	< 0.1	< 0.1	< 0.1	< 0.1	0.32	NT
PBM-8907	637	А	82.72-92.72	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	0.27	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	NT
PBM-8909	639	А	104.4-124.4	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	<b>0.18</b> (J)	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	NT
PBM-9001D	981	D	200.5-210.5	Oct-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	15	1.4	< 0.1	< 0.1	< 0.1	5.5	NT
PBM-9801	360	Α	108.5-123.5	Sep-18	3.582	< 0.1	<0.1	<0.1	<0.1	< 0.2	<b>0.16</b> (J)	<0.1	<0.1	<0.1	<0.1	0.35	NT
PBN-1001C	595	C	194.7-199.7	Sep-18	0.038	< 0.1	<0.1	0.49	<0.1	< 0.2	0.86	0.65	<0.1	<0.1	<0.1	<0.1	NT
PBN-1003C	592	C	184.6-189.6	Sep-18	< 0.008	< 0.1	<0.1	<0.1	<b>0.11 (J)</b>	< 0.2	< 0.1	0.22	<0.1	<0.1	<0.1	< 0.1	NT
PBN-1302A	770	А	69.7-84.7	Sep-18	< 0.008	< 0.1	<0.1	0.62	<0.1	< 0.2	2.2	<b>0.14</b> (J)	<0.1	< 0.1	<0.1	< 0.1	NT
PBN-1302B	771	В	131.2-136.2	Sep-18	<b>0.011</b> (J)	<0.1	<0.1	0.55	<0.1	< 0.2	2.5	<b>0.14</b> (J)	< 0.1	<0.1	<0.1	<0.1	NT
PBN-1302C	772	С	182.6-187.6	Sep-18	0.0087 (J)	< 0.1	<b>0.1</b> (J)	0.68	< 0.1	< 0.2	3.8	0.69	< 0.1	< 0.1	< 0.1	< 0.1	NT

						Volatile Organic Compounds (VOC) - SW8260C											
Well	Well ID	Sample Level	Sample Depth (feet)	Sample Date	Dinitrotoluene, Total	1,1-Dichloroethane	1,1-Dichloroethene	1,1,1-Trichloroethane	Bromodichloromethane	Carbon Disulfide	Carbon Tetrachloride	Chloroform	Chloromethane	Ethyl Ether	Tetrachloroethene	Trichloroethene	Nitrate, Total
PBN-1302D	773	D	240.1-245.1	Sep-18	< 0.008	<0.1	< 0.1	< 0.1	<0.1	< 0.2	<0.1	<0.1	< 0.1	1.3	<0.1	<0.1	NT
PBN-1303A	774	А	115.5-130.5	Sep-18	< 0.008	<0.1	< 0.1	0.32	< 0.1	< 0.2	0.47	< 0.1	< 0.1	<0.1	< 0.1	<0.1	NT
PBN-1303B	775	В	171.5-176.5	Sep-18	< 0.008	<0.1	< 0.1	0.51	<0.1	< 0.2	0.83	<0.1	< 0.1	<0.1	<0.1	<0.1	NT
PBN-1303B (dup)	775	В	171.5-176.5	Sep-18	< 0.008	< 0.1	< 0.1	0.51	< 0.1	< 0.2	0.81	< 0.1	< 0.1	< 0.1	< 0.1	<0.1	NT
PBN-1303C	776	С	227-232	Sep-18	< 0.008	<0.1	< 0.1	0.77	<0.1	< 0.2	1.4	0.31	< 0.1	<0.1	<0.1	<0.1	NT
PBN-1303D	777	D	282-287	Sep-18	< 0.008	<b>0.13</b> (J)	< 0.1	< 0.1	<0.1	< 0.2	<0.1	< 0.1	< 0.1	<0.1	<0.1	<0.1	NT
PBN-1304A	778	А	101-116	Sep-18	< 0.008	<0.1	< 0.1	< 0.1	<0.1	< 0.2	<0.1	< 0.1	< 0.1	<0.1	<0.1	<0.1	NT
PBN-1304B	779	В	158.1-163.1	Sep-18	< 0.008	<0.1	< 0.1	<b>0.13</b> (J)	<0.1	< 0.2	<b>0.16</b> (J)	< 0.1	< 0.1	<0.1	<0.1	<0.1	NT
PBN-1304C	780	C	213-218	Sep-18	< 0.008	< 0.1	< 0.1	<0.1	< 0.1	<0.2	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	<0.1	NT
PBN-1304D	781	D	268-273	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	<b>0.11 (J)</b>	<b>0.14</b> (J)	< 0.1	< 0.1	< 0.1	<0.1	NT
PBN-1401A	782	А	117.2-132.2	Sep-18	0.742	< 0.1	<0.1	<b>0.14</b> (J)	< 0.1	< 0.2	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	<b>0.16</b> (J)	NT
PBN-1401B	783	В	158.7-163.7	Sep-18	0.552	< 0.1	< 0.1	<b>0.14</b> (J)	< 0.1	< 0.2	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	<b>0.13</b> (J)	NT
PBN-1401B (dup)	783	В	158.7-163.7	Sep-18	0.47	< 0.1	< 0.1	<b>0.12</b> (J)	< 0.1	< 0.2	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	<b>0.13</b> (J)	NT
PBN-1401C	784	C	198.3-203.3	Sep-18	0.058	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	NT
PBN-1404B	791	В	174.5-179.5	Sep-18	< 0.008	< 0.1	< 0.1	<b>0.19</b> (J)	< 0.1	< 0.2	2.8	0.97	< 0.1	< 0.1	< 0.1	1	NT
PBN-1404C	792	C	234.3-239.3	Sep-18	0.0044 (J)	< 0.1	< 0.1	<b>0.18</b> (J)	<b>0.16</b> (J)	< 0.2	0.73	0.84	< 0.1	< 0.1	< 0.1	0.2	NT
PBN-1404D	793	D	294.8-299.8	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	< 0.1	< 0.1	< 0.1	480	< 0.1	< 0.1	NT
PBN-1405F	794	F	314.7-319.7	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	<b>0.29</b> (J)	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	NT
PBN-8202A	613	Α	108.5-118.5	Sep-18	116.42	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	0.28	< 0.1	< 0.1	< 0.1	< 0.1	0.8	NT
PBN-8202A (dup)	613	А	108.5-118.5	Sep-18	103.32	< 0.1	<0.1	<0.1	< 0.1	< 0.2	0.31	< 0.1	< 0.1	< 0.1	< 0.1	0.84	NT
PBN-8202B	614	В	131-133	Sep-18	14.612	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	0.64	< 0.1	< 0.1	< 0.1	< 0.1	0.82	NT
PBN-8202C	615	С	139.2-141.2	Sep-18	0.77	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	< 0.1	<0.1	< 0.1	< 0.1	< 0.1	<b>0.14</b> (J)	NT
PBN-8205A	622	А	102.5-112.5	Sep-18	0.837	< 0.1	< 0.1	0.29	< 0.1	< 0.2	2.8	<0.1	< 0.1	< 0.1	< 0.1	0.71	NT
PBN-8205B	623	В	122.2-124.2	Sep-18	0.962	<0.1	<0.1	0.29	<0.1	<0.2	3.1	<0.1	<0.1	<0.1	<0.1	0.76	NT
PBN-8205C	624	C	131.5-133.5	Sep-18	1.094	<0.1	<0.1	0.42	<0.1	< 0.2	3.8	<0.1	<0.1	<0.1	<0.1	0.91	NT
PBN-8502A	632	A	129-138	Sep-18	< 0.008	<0.1	< 0.1	0.83	<0.1	<0.2	14	0.12 (J)	< 0.1	<0.1	< 0.1	2	NT

						Volatile Organic Compounds (VOC) - SW8260C											
Well	Well ID	Sample Level	Sample Depth (feet)	Sample Date	Dinitrotoluene, Total	1,1-Dichloroethane	1,1-Dichloroethene	1,1,1-Trichloroethane	Bromodichloromethane	Carbon Disulfide	Carbon Tetrachloride	Chloroform	Chloromethane	Ethyl Ether	Tetrachloroethene	Trichloroethene	Nitrate, Total
PBN-8503A	633	А	85.82-94.82	Sep-18	0.068	<0.1	< 0.1	<0.1	< 0.1	< 0.2	1.7	< 0.1	<b>0.14 (J)</b>	< 0.1	<0.1	<0.1	NT
PBN-8902BR	795	В	155-160	Sep-18	< 0.008	<0.1	< 0.1	<0.1	< 0.1	< 0.2	1.6	0.31	<b>0.12</b> (J)	< 0.1	<0.1	0.96	NT
PBN-8902C	645	С	188.1-193.3	Sep-18	0.017 (J)	< 0.1	< 0.1	<b>0.11 (J)</b>	< 0.1	< 0.2	1.5	< 0.1	< 0.1	< 0.1	<0.1	0.67	NT
PBN-8903B	646	В	120-125	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	<b>0.13</b> (J)	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	NT
PBN-8903C	647	С	155-160	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	< 0.1	< 0.1	< 0.1	< 0.1	<0.1	< 0.1	NT
PBN-8912A	654	А	83.4-103.4	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	< 0.1	< 0.1	< 0.1	< 0.1	<0.1	< 0.1	NT
PBN-8912B	655	В	133-138	Sep-18	< 0.008	< 0.1	< 0.1	<0.1	< 0.1	< 0.2	0.66	< 0.1	< 0.1	< 0.1	<b>0.11 (J)</b>	0.44	NT
PBN-9101C	561	С	142.5-152.5	Oct-18	0.08	< 0.1	< 0.1	<b>0.16</b> (J)	< 0.1	< 0.2	19	1.7	< 0.1	< 0.1	< 0.1	8.5	NT
PBN-9112C	665	С	173.4-183.4	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	0.96	<b>0.13</b> (J)	< 0.1	< 0.1	< 0.1	0.33	NT
PBN-9112D	666	D	221-231	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	NT
PBN-9301B	668	В	150.5-160.5	Sep-18	< 0.008	< 0.1	< 0.1	0.31	< 0.1	< 0.2	3.1	0.37	< 0.1	< 0.1	< 0.1	<b>0.13</b> (J)	NT
PBN-9301C	669	С	217.5-227.5	Sep-18	< 0.008	< 0.1	<b>0.1</b> (J)	1.1	<b>0.15</b> (J)	< 0.2	1.7	1.1	< 0.1	< 0.1	< 0.1	0.34	NT
PBN-9303B	673	В	83.5-93.5	Sep-18	< 0.008	< 0.1	< 0.1	0.75	< 0.1	< 0.2	2.4	<b>0.16</b> (J)	< 0.1	< 0.1	< 0.1	0.2	NT
PBN-9303B (dup)	673	В	83.5-93.5	Sep-18	< 0.008	< 0.1	< 0.1	0.65	< 0.1	< 0.2	2.1	<b>0.15</b> (J)	< 0.1	< 0.1	< 0.1	<b>0.18</b> (J)	NT
PBN-9303C	674	С	154.5-164.5	Sep-18	< 0.008	< 0.1	0.37	1.9	< 0.1	< 0.2	3.5	1.1	< 0.1	< 0.1	< 0.1	< 0.1	NT
PBN-9303D	675	D	214.5-224.5	Sep-18	< 0.008	0.39	< 0.1	< 0.1	< 0.1	< 0.2	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	NT
PBN-9304D	687	D	200-210	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	< 0.1	< 0.1	< 0.1	560	< 0.1	< 0.1	NT
PBN-9902D	691	D	217.5-222.5	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	< 0.1	< 0.1	< 0.1	0.26	< 0.1	< 0.1	NT
PBN-9903A	692	А	61-76	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	1.5	< 0.1	< 0.1	< 0.1	< 0.1	<b>0.12</b> (J)	NT
PBN-9903B	693	В	107-112	Sep-18	< 0.008	< 0.1	< 0.1	0.27	< 0.1	< 0.2	5.2	0.28	< 0.1	< 0.1	< 0.1	1.2	NT
PBN-9903B (dup)	693	В	107-112	Sep-18	< 0.008	< 0.1	< 0.1	0.28	< 0.1	< 0.2	5.2	0.29	< 0.1	< 0.1	< 0.1	1.2	NT
PBN-9903C	694	С	158-163	Sep-18	0.078	< 0.1	< 0.1	<b>0.12</b> (J)	< 0.1	< 0.2	10	0.23	< 0.1	< 0.1	< 0.1	1	NT
PBN-9903D	695	D	203-208	Sep-18	< 0.008	<0.1	<0.1	<0.1	<0.1	< 0.2	<0.1	<0.1	<0.1	440	<0.1	<0.1	NT
S1147	709	A	45.8-70.8	Sep-18	<0.008	< 0.1	< 0.1	< 0.1	<0.1	< 0.2	<0.1	<0.1	<0.1	< 0.1	< 0.1	<0.1	NT
S1148	710	A	31.7-56.7	Sep-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.2	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	NT
SPN-8903B	718	В	88.7-93.7	Sep-18	< 0.008	< 0.1	< 0.1	0.16 (J)	<0.1	<0.2	0.98	0.12 (J)	<b>0.12</b> (J)	< 0.1	< 0.1	0.18 (J)	NT

					Volatile Organic Compounds (VOC) - SW8260C												
Well	Well ID	Sample Level	Sample Depth (feet)	Sample Date	Dinitrotoluene, Total	1,1-Dichloroethane	1,1-Dichloroethene	1,1,1-Trichloroethane	Bromodichloromethane	Carbon Disulfide	Carbon Tetrachloride	Chloroform	Chloromethane	Ethyl Ether	Tetrachloroethene	Trichloroethene	Nitrate, Total
SPN-8903C	719	С	122.7-127.7	Sep-18	< 0.008	< 0.1	<b>0.19</b> (J)	1.5	< 0.1	< 0.2	0.54	< 0.1	<b>0.11 (J)</b>	<0.1	< 0.1	0.92	NT
SPN-8904B	720	В	70-75	Sep-18	0.061	< 0.1	< 0.1	<b>0.18</b> (J)	< 0.1	< 0.2	3.3	<b>0.15</b> (J)	< 0.1	<0.1	< 0.1	1	NT
SPN-8904C	721	С	101.5-106.5	Sep-18	< 0.008	< 0.1	< 0.1	0.24	< 0.1	< 0.2	4.5	< 0.1	< 0.1	<0.1	< 0.1	0.37	NT
SPN-9103D	725	D	190.5-200.5	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	NT
SPN-9104D	726	D	196-206	Oct-18	< 0.008	<0.1	< 0.1	< 0.1	< 0.1	< 0.2	<0.1	< 0.1	< 0.1	2,200	< 0.1	< 0.1	NT
SWN-9102C	569	С	142.5-152.5	Sep-18	< 0.008	<0.1	< 0.1	< 0.1	< 0.1	< 0.2	<0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	NT
SWN-9102D	570	D	175-185	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	NT
SWN-9103B	571	В	103.4-113.4	Sep-18	< 0.008	< 0.1	< 0.1	<b>0.13</b> (J)	< 0.1	< 0.2	1.4	<b>0.11 (J)</b>	< 0.1	< 0.1	< 0.1	<b>0.18</b> (J)	NT
SWN-9103B (dup)	571	В	103.4-113.4	Sep-18	< 0.008	< 0.1	< 0.1	<b>0.14</b> (J)	< 0.1	< 0.2	1.9	<b>0.13</b> (J)	< 0.1	< 0.1	< 0.1	0.2	NT
SWN-9103C	572	С	152.8-162.8	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	0.51	0.34	< 0.1	< 0.1	< 0.1	< 0.1	NT
SWN-9103D	573	D	199.1-209.1	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	2.8	0.33	< 0.1	< 0.1	< 0.1	0.32	NT
SWN-9103E	574	Е	227.9-237.9	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	NT
SWN-9104C	575	С	154-164	Sep-18	< 0.008	< 0.1	< 0.1	<b>0.12 (J)</b>	< 0.1	< 0.2	3.8	0.5	< 0.1	< 0.1	< 0.1	< 0.1	NT
SWN-9104D	576	D	187-197	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	5.1	0.79	< 0.1	< 0.1	< 0.1	< 0.1	NT
SWN-9105B	577	В	102.5-112.5	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	<b>0.1</b> (J)	0.2	< 0.1	< 0.1	< 0.1	< 0.1	NT
SWN-9105C	578	С	137-147	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	<b>0.19</b> (J)	0.65	< 0.1	< 0.1	< 0.1	< 0.1	NT
SWN-9105D	579	D	190.5-200.5	Sep-18	< 0.008	< 0.1	< 0.1	< 0.1	< 0.1	< 0.2	0.52	0.53	< 0.1	< 0.1	< 0.1	< 0.1	NT
Chapter NR 140 Preventive Action Limit (PAL) 0.005					85	0.7	40	0.06	200	0.5	0.6	3	100	0.5	0.5	2	
Chapte	Chapter NR 140 Enforcement Standard (ES) 0.05					850	7	200	0.6	1000	5	6.0	30	1000	5	5	10

Notes:

The Sample Level references the typical well depth configuration

Dinitrotoluene, Total (DNT) & VOC results are expressed in micrograms per liter (µg/l)

Nitrate, Total results are expressed in milligrams per liter (mg/l)

Bold values are detected results

Wells listed with (dup) after the name were duplicate samples Results for Dinitrotoluene, Total were analyzed by SW8270DSIM

J = Analytical result is between the Limit of Detection (LOD) and Limit of Quantitation (LOQ)

NT = Not Tested

A total of ten monitoring wells had NR 140 ES exceedances for 2,4-DNT during September 2018. 2,4-DNT was not detected in any residential wells that were sampled during August 2018.

A total of seventeen monitoring wells had NR 140 ES exceedances for 2,6-DNT during September 2018. In addition, eight monitoring wells had NR 140 PAL exceedances for 2,6-DNT during September 2018. 2,6-DNT was not detected in any residential wells that were sampled during August 2018.

A total of twenty-one monitoring wells had NR 140 ES exceedances for total DNT during September 2018. In addition, four monitoring wells had NR 140 PAL exceedances for total DNT during September 2018. Total DNT was not detected in any residential wells that were sampled during August 2018.

One monitoring well had a NR 140 ES exceedance for ethyl ether during September 2018. In addition, three monitoring wells had NR 140 PAL exceedances for ethyl ether during September 2018. One residential well had an ethyl ether detection that was below the NR 140 PAL during August 2018.

Three monitoring wells had NR 140 PAL exceedances for nitrate during September 2018. Residential wells are no longer being sampled for nitrate due to historically low detections.

A total of two monitoring wells had NR 140 ES exceedances for TCE during September 2018. In addition, seventeen monitoring wells had NR 140 PAL exceedances for TCE during September 2018. TCE was not detected in any residential wells that were sampled during August 2018.

### 4.5.1.1 *Carbon Tetrachloride*

The horizontal distribution of CTET is illustrated in Figure 22. The green shaded area displays where CTET was detected above the NR 140 PAL ( $0.5 \mu g/l$ ). The blue shaded area displays where CTET was detected above the NR 140 ES ( $5 \mu g/l$ ). These same color designations are also used in each CTET cross section. The highest concentration of CTET detected during September 2018 was 19  $\mu g/l$  in PBN-9101C, which is located 2,300 feet upgradient of the Wisconsin River. The horizontal boundary of the CTET plume covers the largest area compared to ethyl ether, total DNT, and trichloroethene.

Figure 6 shows the orientation of the isoconcentration cross sections for CTET, which are illustrated in Figures 23, 24, 25, and 26. As shown in Figures 23 and 25, there is a dolomite/shale layer beneath the contamination plume that retards groundwater contamination from migrating into the lower Mt. Simon Formation (sandstone).

Figure 23 (A-A') illustrates the estimated vertical extent of CTET, along the centerline of the PBG Plume, from the PBG (north) towards the Wisconsin River (south). The CTET concentrations are highest south of the BAAP boundary and in wells screened approximately 65 to 140 feet below the water table. The CTET plume extends north to south from the PBG to the

Wisconsin River with an average thickness of 90 feet beneath BAAP and 150 feet south of BAAP. The maximum depth of CTET is 150 feet below the water table at monitoring well PBM-9001D, which is screened in the gravel and sand just above the sandstone. Based on Figure 23, CTET has potentially entered the upper portion of the bedrock aquifer near PBM-9001D.

Based on the *Surface Waters Impact Investigation Report (BTS, LLC), November 2013*, CTET concentrations diminish as the PBG Plume migrates vertically and discharges into the Wisconsin River. CTET concentrations above the NR 140 ES were only identified in the sand aquifer; therefore, CTET does not migrate vertically into the bedrock. Isoconcentration sections for CTET (Figures 4 and 5) provided in the *Surface Waters Impact Investigation Report*, show the CTET plume boundaries in relation to the bedrock and Wisconsin River. More studies in this area would be helpful to further define the CTET concentrations above the bedrock.

CTET concentrations beneath the PBG (source area) are much lower than what is found downgradient of the PBG. The estimated boundary of the CTET plume is shown to approach the Wisconsin River. The groundwater mixes with the saturated sediment beneath the Wisconsin River. This zone is the groundwater/surface water interface. Dilution and volatilization of the CTET plume is expected to occur at the groundwater/surface water interface.

Figure 24 (B-B') illustrates the width and depth of the CTET plume approximately 2,000 feet south of the PBG. Figure 25 (C-C') illustrates the width and depth of the CTET plume approximately 6,600 feet south of the PBG and at the BAAP boundary. Figure 26 (D-D') illustrates the width and depth of the CTET plume, but off-site and approximately 12,000 feet south of the PBG. The CTET plume in Figure 24 is estimated to be approximately 3,200 feet wide and a maximum depth of 135 feet below the water table at PBN-9301C. The CTET plume in Figure 25 is estimated to be approximately 2,800 feet wide and a maximum depth of 150 feet below the water table and below PBN-1302C. The CTET plume in Figure 26 is estimated to be approximately 2,500 feet wide and a maximum depth of 120 feet below the water table at monitoring well SWN-9103D.

The following residential wells are shown on either Figure 23 (A-A') or Figure 26 (D-D'): Judd, Lins-K, Lins-R, Mueller-J, and Urban. These residential wells represent all the residential wells located near the PBG Plume. As shown in the figures, the Judd well is screened in the sand and gravel aquifer and the Lins-K, Lins-R, Mueller-J, and Urban wells are screened in the bedrock aquifer. There are several residential wells that were drilled through the CTET plume and then screened beneath the CTET plume.

### 4.5.1.2 *Ethyl Ether*

The horizontal distribution of ethyl ether (diethyl ether) is illustrated in Figure 27. The green shaded area displays where ethyl ether was detected above the NR 140 PAL (100  $\mu$ g/l). The blue shaded area displays where ethyl ether was detected above the NR 140 ES (1,000  $\mu$ g/l). These same color designations are also used in each ethyl ether cross section. The highest concentration of ethyl ether detected during September 2018 was 2,200  $\mu$ g/l in SPN-9104D, which is located at the BAAP Boundary. The horizontal boundary of the ethyl ether plume

covers the smallest area compared to CTET, total DNT, and trichloroethene. The ethyl ether plume is shown in two small areas downgradient of the PBG.

Figure 6 shows the orientation of the isoconcentration cross sections for ethyl ether, which are illustrated in Figures 28, 29, 30, and 31. Figure 28 (A-A') illustrates the estimated vertical extent of ethyl ether, along the centerline of the PBG Plume, from the PBG (north) towards the Wisconsin River (south). The ethyl ether concentrations are highest at the BAAP boundary and in wells screened approximately 170 feet below the water table. The ethyl ether plume is approximately 60 feet thick beneath BAAP. The maximum depth of ethyl ether is 190 feet below the water table and below PBN-9304D, which is screened just above the top of the bedrock. Figure 28 shows ethyl ether to only be detected in monitoring wells near the bottom of the sand aquifer. Based on Figure 28, ethyl ether has likely entered the upper portion of the bedrock aquifer near PBN-9304D and SPN-9104D. Ethyl ether was not detected in PBN-1405F, which was constructed 110 feet beneath PBN-9304D and below the dolomite/shale layer.

Figure 29 (B-B') illustrates the width and depth of the ethyl ether plume approximately 2,000 feet south of the PBG. Figure 30 (C-C') illustrates the width and depth of the ethyl ether plume approximately 6,600 feet south of the PBG and at the BAAP boundary. Figure 31 (D-D') illustrates the width and depth of the ethyl ether plume, but off-site and approximately 12,000 feet south of the PBG. There is no ethyl ether plume shown in Figure 29 because there were no detections above 100  $\mu$ g/l. The ethyl ether plume in Figure 30 is estimated to be approximately 750 feet wide and a maximum depth of 190 feet below the water table and below PBN-9304D. There is no ethyl ether plume 31 because there were no detections above 100  $\mu$ g/l.

The following residential wells are shown on either Figure 28 (A-A') or Figure 31 (D-D'): Judd, Lins-K, Lins-R, Mueller-J, and Urban. There are also five residential wells located over 1,400 feet east of the ethyl ether plume shown in Figure 27. The highest ethyl ether concentration in these five wells was only  $0.16 \mu g/l$ .

### 4.5.1.3 Trichloroethene

The horizontal distribution of TCE is illustrated in Figure 32. The green shaded area displays where TCE was detected above the NR 140 PAL ( $0.5 \mu g/l$ ). The blue shaded area displays where TCE was detected above the NR 140 ES ( $5 \mu g/l$ ). These same color designations are also used in each TCE cross section. The highest concentration of TCE detected during September 2018 was 8.5  $\mu g/l$  in PBN-9101C, which is located 2,300 feet upgradient of the Wisconsin River. The horizontal boundary of the TCE plume extends from the PBG to the Wisconsin River but is much narrower than the CTET plume.

Figure 6 shows the orientation of the isoconcentration cross sections for TCE, which are illustrated in Figures 33, 34, 35, and 36. Figure 33 (A-A') illustrates the estimated vertical extent of TCE, along the centerline of the PBG Plume, from the PBG (north) towards the Wisconsin River (south). The TCE concentrations are highest at the BAAP boundary, south of the BAAP boundary, and in wells screened approximately 65 to 140 feet below the water table. The TCE plume has an average thickness of 110 feet. The maximum depth of TCE is 145 feet below the

water table and below PBM-9001D, which is screened just above the top of the bedrock. Based on Figure 33, TCE has likely entered the upper portion of the bedrock aquifer near PBM-9001D. TCE concentrations near the PBG (source area) are much lower than what is found downgradient of the PBG. The estimated boundary of the TCE plume is shown to approach the Wisconsin River.

Figure 34 (B-B') illustrates the width and depth of the TCE plume approximately 2,000 feet south of the PBG. Figure 35 (C-C') illustrates the width and depth of the TCE plume approximately 6,600 feet south of the PBG and at the BAAP boundary. Figure 36 (D-D') illustrates the width and depth of the TCE plume, but off-site and approximately 12,000 feet south of the PBG. The TCE plume in Figure 34 is estimated to be approximately 1,200 feet wide and a maximum depth of 80 feet below the water table at PBN-8902C. The TCE plume in Figure 35 is estimated to be approximately 1,400 feet wide and a maximum depth of 85 feet below the water table and below SPN-8903C. There is no TCE plume shown in Figure 36 because there were no detections above  $0.5 \mu g/l$ .

The following residential wells are shown on either Figure 33 (A-A') or Figure 34 (D-D'): Judd, Lins-K, Lins-R, Mueller-J, and Urban. There are several residential wells that were drilled through the TCE plume and then screened beneath the TCE plume.

### 4.5.1.4 *Total Dinitrotoluene*

The horizontal distribution of total DNT is illustrated in Figure 37. The total DNT concentration is the sum of all six DNT isomers (2,3-DNT, 2,4-DNT, 2,5-DNT, 2,6-DNT, 3,4-DNT, and 3,5-DNT). The green shaded area displays where total DNT was detected above the NR 140 PAL (0.005  $\mu$ g/l). The blue shaded area displays where total DNT was detected above the NR 140 ES (0.05  $\mu$ g/l). The red shaded area displays where total DNT was detected above 1.0  $\mu$ g/l. These same color designations are also used in each total DNT cross section. The highest concentration of total DNT detected during September 2018 was 116.42  $\mu$ g/l in PBN-8202A, which is immediately downgradient of the PBG. The total DNT plume is shown in three separate areas, near the PBG, near the BAAP boundary, and farther downgradient of the PBG to the Wisconsin River. The separation of the total DNT plumes maybe related to the extensive groundwater pumping conducted by the MIRM treatment system.

Figure 6 shows the orientation of the isoconcentration cross sections for total DNT, which are illustrated in Figures 38, 39, 40, 41, and 42. Figure 38 (A-A') illustrates the estimated vertical extent of total DNT, along the centerline of the PBG Plume, from the PBG (north) towards the Wisconsin River (south). The total DNT concentrations beneath the PBG (source area) are higher than what is found downgradient. The total DNT concentrations are much lower south of the BAAP boundary than what is found on BAAP. The total DNT concentrations are highest in wells screened approximately 0 to 30 feet below the water table. The total DNT plume has an average thickness of 100 feet. The maximum depth of total DNT is 100 feet below the water table at PBN-9903C, which is screened 40 feet above the top of the bedrock. Based on Figure 38, total DNT has not entered the bedrock aquifer beneath or downgradient of BAAP. The estimated boundary of the total DNT plume is shown to approach the Wisconsin River.

Figure 39 (A1-A1') illustrates the estimated vertical extent of total DNT beneath the capped PBG Waste Pits to the southeast corner of the Racetrack Area at PBN-8205A, B, C. The total DNT concentrations are highest in the shallow wells located beneath the PBG Waste Pits (source area). The highest concentrations of total DNT are shown in well nest PBN-8202A, B, C, which is downgradient of waste pit 2 (WP-2). The vertical depth of the total DNT plume beneath the PBG Waste Pits can only be estimated as there are no deeper monitoring wells.

Figure 40 (B-B') illustrates the width and depth of the total DNT plume approximately 2,000 feet south of the PBG. Figure 41 (C-C') illustrates the width and depth of the total DNT plume approximately 6,600 feet south of the PBG and at the BAAP boundary. Figure 42 (D-D') illustrates the width and depth of the total DNT plume, but off-site and approximately 12,000 feet south of the PBG. The total DNT plume in Figure 40 is estimated to be approximately 1,800 feet wide and a maximum depth of 100 feet below the water table at PBN-8902C. The total DNT plume in Figure 41 is estimated to be approximately 1,300 feet wide and a maximum depth of 130 feet below the water table and below PBN-1302C. There is no total DNT plume shown in Figure 42 because there were no detections above  $0.005 \,\mu g/l$ .

The following residential wells are shown on either Figure 38 (A-A') or Figure 42 (D-D'): Judd, Lins-K, Lins-R, Mueller-J, and Urban. There is one residential well that was drilled through the total DNT plume and then screened beneath the total DNT plume.

### 4.5.1.5 *Concentration Graphs*

To evaluate contaminant trend data for the PBG Plume, concentration over time graphs were prepared for select monitoring wells within the plume. Graphs showing PBG Plume contaminant concentration over time are presented in Appendix E. The primary COPCs used for trend analysis were CTET, chloroform, ethyl ether, TCE, and total DNT. In the source area, data from eight monitoring wells were graphed. Graphs were prepared for 17 on-site monitoring wells located downgradient of the PBG. Graphs were prepared for 17 off-site monitoring wells located downgradient of the PBG.

The source area wells PBM-0002, PBM-0008, and PBN-8202A show a large decrease in DNT concentrations after 2002. These sharp decreases are related to the operation of the BEST system from 2001 to 2005. During December 2012, the IRM ceased groundwater pumping directly downgradient of the PBG Waste Pits and PBM-0002. Between 2012 to 2017, the total DNT concentrations in the source area wells stabilized between 1 to 5  $\mu$ g/l. During April 2018, a noticeable increase in total DNT concentration was identified in PBN-8202A. PBN-8202A is located directly south and downgradient of the PBG Waste Pits (see Figure 18). The total DNT concentration in PBN-8202A increased from 1.469  $\mu$ g/l during September 2017 to 94.65  $\mu$ g/l during April 2018 to 420.294  $\mu$ g/l during May 2018 to 116.42  $\mu$ g/l during September 2018.

Between April 2016 and April 2018, the groundwater table near the PBG Waste Pits rose 6.9 feet. Provided below is a graph depicting both the total DNT concentration and groundwater elevation in PBN-8202A from 2007 to 2018. The graph shows a peak in the groundwater elevation in 2009 but not an increase of total DNT. During 2009, the IRM was still operating a groundwater pumping well approximately 125 feet southwest of PBN-8202A. The graph shows

another peak in groundwater elevation in 2018 along with a sharp increase of total DNT in PBN-8202A. During 2018, the groundwater elevation in PBN-8202A ranged from 777.4 to 778.5 feet MSL. Based on the 2005 soil investigation data presented in Appendix B (Table 5), soil boring PBB-0502 (Waste Pit 2) had detectable concentrations of both 2,4-DNT and 2,6-DNT at a depth of 105 feet or 776.80 feet MSL. The following graph displays the DNT soil contamination depth in relation to the groundwater elevation. During the 2005 soil investigation, the groundwater was at an elevation of 774.5 feet MSL, below the soil contamination. Based on the above information, the groundwater beneath the PBG Waste Pits has risen above the DNT contaminated soil. The recent increase in total DNT concentrations in PBN-8202A appears to be related to the recent rise in groundwater coming into contact with the soil contamination.



The VOC compounds of CTET, chloroform, and TCE have been declining near the source area since the 1980's. The VOC compounds have declined to levels at or below the NR 140 ES.

In the on-site portion of the plume, the VOC compounds of CTET, chloroform, and TCE show decreasing trends in both the shallow and deep wells. The exception is that chloroform in PBN-8502A had a peak during 2015 but declined in 2016, 2017, and 2018. The ethyl ether concentrations in PBN-1001C and PBN-9304D have been decreasing. The DNT concentrations in the wells downgradient of the PBG show either stable or decreasing trends.

As the plume extends off-site, the VOC compounds of CTET, chloroform, and TCE show either stable or decreasing trends in both the shallow and deep wells. There are several monitoring wells that have seen peaks followed by decreases.

- The CTET concentration in SWN-9103C had a sharp peak during 2010 (90.1 µg/l) followed by a sharp decrease below 5 µg/l by 2014. The CTET concentration in PBN-9101C had a peak during 2012 (44.8 µg/l) followed by a decrease to 19 µg/l during 2018. The CTET concentration in PBM-9001D had a peak (above 25 µg/l) during 2011 followed by a slight decrease to 15 µg/l during 2017. During 2018, the CTET concentration in PBM-9001D peaked again at 25 µg/l and then dropped back to 15 µg/l. The CTET concentration in SWN-9104C has increased from 2015 to 2018, reaching 3.8 µg/l during September 2018. The CTET concentration in SWN-9104D has increased from 2015 to 2018, reaching 5.1 µg/l during September 2018.
- The chloroform concentration in SWN-9103C had a peak during 2007 (above 7 μg/l) followed by a decrease to below 0.5 μg/l during 2018. The chloroform concentration in PBN-9101C had a peak during 2011 (above 6 μg/l) followed by a decrease to 1.7 μg/l during 2018. The chloroform concentration in PBM-9001D had a peak (above 3 μg/l) during 2011 followed by a decrease to 1.4 μg/l during 2018.
- The ethyl ether concentration in PBN-9903D peaked during 2014 (above 3,500 µg/l) but decreased during 2015 and has remained stable with a concentration of 440 µg/l during 2018. Ethyl ether is not detected in the off-site monitoring wells located south (downgradient) of PBN-9903D.
- The TCE concentration in SWN-9103B had a peak during 2000 (above 7  $\mu$ g/l) followed by a steady decrease to below 0.5  $\mu$ g/l by 2014. The TCE concentration in SWN-9103D had a peak during 2014 (near 5  $\mu$ g/l) followed by a decrease to below 0.5  $\mu$ g/l by 2018. The TCE concentration in PBN-9101C had a peak during 2011 (above 14  $\mu$ g/l) followed by a decrease till 2017. Between 2017 and 2018, the TCE concentration in PBN-9101C increased from 6.5 to 8.8  $\mu$ g/l. PBN-9101C was not sampled between 1999 to 2010; therefore, no data was available. The TCE concentration in PBM-9001D had a peak during 2011 (above 5  $\mu$ g/l) followed by a decrease till 2015. Between 2015 and 2018, the TCE concentration in PBM-9001D has increased from 3.1 to 8.6  $\mu$ g/l.
- The DNT concentrations in the off-site monitoring wells have been stable or decreasing. The exception is that PBN-9101C had a peak (above 0.1  $\mu$ g/l) during 2013 followed by a slight decrease to 0.08  $\mu$ g/l during 2018.

### 4.5.1.6 Monitored Natural Attenuation

An evaluation of existing site information and groundwater data was conducted to illustrate that monitored natural attenuation (MNA) of chlorinated solvents or VOCs has been occurring within the PBG Plume. Based on groundwater monitoring data collected between 2015 and 2018, the following VOCs have been detected above the NR 140 PAL or ES routinely in the PBG Plume: CTET, chloroform, ethyl ether, and TCE.

The *Draft Technical Report Natural Attenuation Screening Study for the Propellant Burning Ground* (Stone & Webster, August 1999) provided evidence that VOCs are naturally attenuating in the PBG Plume. The Stone & Webster report summarized that the concentrations of chlorinated solvents in the groundwater are declining over time, along the length of the plume, and decrease with separation from the source area. This indicates that, overall, the chlorinated solvents are leaving the groundwater by some natural attenuation mechanism. Stone & Webster documented that no chlorinated solvent degradation products or transformation products have been detected in the groundwater. Based on groundwater monitoring data over the past 30 years, the more toxic TCE degradation product, vinyl chloride, has not been detected.

During December 1998, Stone & Webster collected groundwater samples from 38 monitoring wells located within or near the PBG Plume. Monitoring wells were chosen upgradient of the source area, in the source area, and downgradient of the source area. The samples were laboratory analyzed for VOCs and semi-volatile organic compounds (SVOCs). The samples were also analyzed for the following geochemical parameters: chloride, dissolved oxygen, iron II, methane, nitrate, nitrite, oxidation reduction potential (ORP), pH, sulfate, sulfide, temperature, total dissolved solids, and total organic carbon. Based on this 1998 data, there is no evidence to suggest that reductive dechlorination has occurred in the PBG Plume. The PBG Plume is a well-oxygenated groundwater system (aerobic) with little or no organic matter.

Stone & Webster documented that CTET, chloroform, and TCE concentrations dropped between 1990 and 1998 in six monitoring wells that are located along the axis (centerline) of the PBG Plume (PBN-8910A, PBN-8205A, PBN-8501A, PBN-8504A, PBN-8912B, and SPN-8903B). A generalized summation of the Stone & Webster groundwater data findings is shown below.

Monitoring Well	Distance from Source Area (feet)	Date Sampled	Carbon Tetrachloride	Chloroform	Trichloroethene
DDN 2010A	700	Mar-90	31.0	5.6	103.0
r div-0910A	700	Dec-98	11.0	1.6	48.0
DDN 8205 A	1.540	Mar-90	88.0	5.5	112.0
PDIN-8203A	1,340	Dec-98	42.0	2.6	41.0
PBN-8501A	2 520	Mar-90	43.0	14.0	30.0
	2,320	Dec-98	17.0	3.3	20.0
DDN 9504A	2 020	Mar-91	21.0	6.9	11.0
PDN-6304A	5,920	Dec-98	0.8	<0.2	<0.2
DDN 9012D	5 600	Mar-90	51.0	7.8	20.0
PDIN-0912D	3,000	Dec-98	<0.4	< 0.2	<0.2
	7 000	Mar-90	130.0	11.0	<5.0
SPIN-8903B	7,000	Dec-98	24.0	2.1	1.3

### Summary of 1990 - 1998 VOC Groundwater Data Propellant Burning Ground Plume

Note: All results expressed in micrograms per liter  $(\mu g/l)$ 

Shown on Figure 19 are locations of these six monitoring wells (shown above) in relation to the former IRM and MIRM extraction wells. Even though the four IRM wells (BCW-1, BCW-2, BCW-3, and SCW-1) were running from 1990 to 1998, they were only pumping a combined 350 gpm. Four of the six monitoring wells were isolated from the IRM wells and therefore not influenced by their pumping. The MIRM became operational in 1996 with six boundary extraction wells (EW-161, EW-162, EW-163, EW-164, EW-165, and EW-166) pumping a combined 3,000 gpm. Extraction wells EW-167, EW-168, EW-169, and EW-170 (EW-170R) were operational between 2006 and 2015. Five of the six monitoring wells were isolated (located far north) from the MIRM wells and therefore not influenced by their pumping. Shown in the above summary table are reductions in VOCs that clearly indicate that the PBG Plume was undergoing natural attenuation between 1990 and 1998.

Concentration over time graphs for monitoring well nests PBN-8205A, B, and C; PBN-8502A, PBN-8902BR, and PBN-8902C; and PBN-8912A, PBN-8912B, PBN-9112C, and PBN-9112D are provided in Appendix E. These 10 monitoring wells are located south of the former IRM wells and north of the original (1996) MIRM wells. These monitoring wells were not influenced by pumping operations until the MIRM was realigned in 2005. CTET, chloroform, and TCE concentrations for all 10 monitoring wells show decreasing trends. A more thorough discussion of concentration graphs for these 10 monitoring wells and 24 other monitoring wells associated with the PBG Plume is provided in Section 4.5.1.6. Ethyl ether concentrations in PBN-1001C,

PBN-9304D, and PBN-9903D have been declining since 2013, graphs are provided in Appendix E.

Based on the historic groundwater data, MNA has a reasonable probability of bringing the VOCs in the PBG Plume into compliance with Chapter NR 140 groundwater quality standards within a reasonable period of time.

### 4.5.2 Deterrent Burning Ground Plume

The DBG Plume originates at the DBG and extends southeast beyond the BAAP boundary. East of BAAP, the plume continues southeast towards Weigand's Bay which is connected to the Wisconsin River. The DBG Plume shown in Figure 21 represents the area where groundwater concentrations exceed a NR 140 PAL or ES for 2,4-DNT, 2,6-DNT, or total DNT. All six DNT isomers (2,3-DNT, 2,4-DNT, 2,5-DNT, 2,6-DNT, 3,4-DNT, and 3,5-DNT) have been detected in the DBG Plume. Because total DNT encompasses all six DNT isomers, total DNT was used to map the DBG Plume boundaries. The DBG Plume boundaries shown in Figure 21 are approximate and based on total DNT groundwater data collected during 2018 from both monitoring wells and residential wells. Table 12 summarizes the groundwater analytical results from the April 2018 sampling event for the monitoring wells associated with the DBG Plume. The April 2018 sampling round includes results for total DNT, sulfate, and VOCs. Table 10 summarizes the groundwater analytical results from the August 2018 residential well sampling event. Table 11 summarizes the groundwater analytical results from the September 2018 monitoring well sampling event. Monitoring wells associated with the DBG Plume were not sampled for VOCs during September 2018. An isoconcentration map and two cross sections were prepared for total DNT. The isoconcentration map was prepared using all groundwater data collected during 2018. The isoconcentration cross sections were prepared using groundwater data collected during August and September 2018.

During 2015, 2016, 2017, and 2018, 2,4-DNT, 2,6-DNT, total DNT, sulfate, and 1,1,2-trichloroethane (1,1,2-TCA) have been COPCs in the DBG Plume. The DBG Plume groundwater results from the April, August, and September 2018 sampling events were evaluated for COPCs that exceeded a Chapter NR 140 PAL Groundwater Standard.

A total of three monitoring wells had NR 140 PAL exceedances for 2,4-DNT during April 2018. 2,4-DNT was not detected in any residential wells that were sampled during August 2018.

A total of two monitoring wells had NR 140 ES exceedances for 2,6-DNT during April 2018. In addition, eight monitoring wells had NR 140 PAL exceedances for 2,6-DNT during April 2018. One monitoring well had a NR 140 ES exceedance for 2,6-DNT during September 2018. In addition, eight monitoring wells had NR 140 PAL exceedances for 2,6-DNT during September 2018. 2,6-DNT was not detected in any residential wells that were sampled during August 2018.

						Volatile Organic Compounds (VOC) - SW8260C										
Well	Well ID	Sample Level	Sample Depth (feet)	Sample Date	Dinitrotoluene, Total	1,2-Dichloroethane	1,2-Dichloropropane	1,1,1-Trichloroethane	1,1,2-Trichloroethane	Dichlorodifluoromethane	Ethyl Ether	Tetrahydrofuran	Trichloroethene	Sulfate, Total		
DBG PLUME ARE	A WELLS	5														
DBM-8201	301	А	154.6-174.6	Apr-18	2.216	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	18		
DBM-8202	302	А	137.3-157.3	Apr-18	0.578	<0.1	<0.1	0.97	<0.1	<0.1	<0.1	<1	<0.1	31		
DBM-8903	306	А	113-133	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT		
DBN-1001B	472	В	154.5-159.5	Apr-18	0.48	<0.1	<0.1	1.5	<0.1	<0.1	<0.1	<1	<0.1	NT		
DBN-1001C	473	С	192-197	Apr-18	0.024 (J)	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT		
DBN-1001E	474	Е	274.9-279.9	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT		
DBN-1002C	476	C	205.1-210.1	Apr-18	0.772	<0.1	<0.1	0.23	<0.1	<0.1	<0.1	<1	<0.1	19		
DBN-1002E	477	Е	275.5-280.5	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	18		
DBN-9501A	314	А	110-120	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT		
DBN-9501B	315	В	162.5-172.5	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT		
DBN-9501C	316	C	218.5-228.5	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT		
DBN-9501E	317	Е	245.2-255.5	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT		
ELM-8901	216	А	145.5-165	Apr-18	1.409	<0.1	<0.1	1.2	<0.1	<0.1	<0.1	<1	<0.1	76		
ELM-8907	220	А	130.3-150.3	Apr-18	0.57	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	17		
ELM-8908	221	А	125-145	Apr-18	0.345	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	15		
ELM-8909	222	А	135-155	Apr-18	<0.008	<0.1	<0.1	0.76	<0.1	<0.1	<0.1	<1	<0.1	13		
ELM-9501	234	А	54-69	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT		
ELN-0801B	455	В	100-105	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT		
ELN-0801B (dup)	455	В	100-105	Apr-18	< 0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT		
ELN-0801C	456	C	145.5-150.5	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT		
ELN-0801E	457	Е	202.6-207.6	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT		

						Volatile Organic Compounds (VOC) - SW8260C									
Well	Well ID	Sample Level	Sample Depth (feet)	Sample Date	Dinitrotoluene, Total	1,2-Dichloroethane	1,2-Dichloropropane	1,1,1-Trichloroethane	1,1,2-Trichloroethane	Dichlorodifluoromethane	Ethyl Ether	Tetrahydrofuran	Trichloroethene	Sulfate, Total	
ELN-1001B	460	В	91.1-96.1	Apr-18	< 0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1001C	461	С	155.2-160.2	Apr-18	< 0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1001E	462	Е	240.5-245.5	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1002A	463	А	55.3-70.3	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1002B	464	В	111.2-116.2	Apr-18	< 0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1002B (dup)	464	В	111.2-116.2	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1002C	465	С	159.1-164.1	Apr-18	< 0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1002E	466	Е	231.5-236.5	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1003A	467	А	16.2-31.2	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1003B	468	В	91.5-96.5	Apr-18	0.232	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1003B (dup)	468	В	91.5-96.5	Apr-18	0.225	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1003B	468	В	91.5-96.5	May-18	0.186	NT	NT	NT	NT	NT	NT	NT	NT	NT	
ELN-1003C	469	C	155.1-160.1	Apr-18	0.074	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1003C	469	C	155.1-160.1	May-18	0.108	NT	NT	NT	NT	NT	NT	NT	NT	NT	
ELN-1003E	470	Е	255.6-230.6	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1502A	533	А	115.3-130.3	Apr-18	0.594	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1502A (dup)	533	А	115.3-130.3	Apr-18	0.537	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1502A	533	А	115.3-130.3	May-18	0.69	NT	NT	NT	NT	NT	NT	NT	NT	NT	
ELN-1502C	534	C	198-203	Apr-18	0.022 (J)	<0.1	<0.1	0.71	<0.1	0.14 (J)	<0.1	<1	<0.1	NT	
ELN-1503A	535	А	73.7-88.7	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1503C	536	C	157.6-162.6	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	
ELN-1504B	537	В	34.8-39.8	Apr-18	< 0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT	

								Volatile Or	ganic Compo	unds (VOC) -	SW8260C			
Well	Well ID	Sample Level	Sample Depth (feet)	Sample Date	Dinitrotoluene, Total	1,2-Dichloroethane	1,2-Dichloropropane	1,1,1-Trichloroethane	1,1,2-Trichloroethane	Dichlorodifluoromethane	Ethyl Ether	Tetrahydrofuran	Trichloroethene	Sulfate, Total
ELN-8203A	210	А	147.5-157.5	Apr-18	< 0.008	<0.1	0.28	<0.1	0.3	<b>0.15</b> (J)	0.49	<1	<0.1	1100
ELN-8203B	211	В	164-166	Apr-18	< 0.008	<0.1	0.36	<0.1	0.98	0.36	0.27	<1	<0.1	990
ELN-8203B (dup)	211	В	164-166	Apr-18	< 0.008	<0.1	0.35	<b>0.11</b> (J)	0.87	0.38	0.31	<1	<0.1	1000
ELN-8203C	212	С	174-176	Apr-18	< 0.008	<0.1	<0.1	<0.1	0.38	<0.1	< 0.1	<1	<0.1	55
ELN-8902B	224	В	173.5-178.5	Apr-18	< 0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	18
ELN-9107A	227	А	116-126	Apr-18	< 0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	21
ELN-9107B	228	В	135-145	Apr-18	< 0.008	<0.1	<0.1	<0.1	0.25	<0.1	<0.1	<1	<0.1	34
ELN-9402AR	231	А	130-145	Apr-18	< 0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	11
S1121	755	А	39.11-59.3	Apr-18	<0.008	<0.1	<0.1	<0.1	<0.1	<0.1	<0.1	<1	<0.1	NT
S1134R	236	А	136-151	Apr-18	< 0.008	<0.1	<b>0.11</b> (J)	<0.1	0.42	<0.1	<0.1	<1	<0.1	750
Chapter	Chapter NR 140 Preventive Action Limit (PAL)				0.005	0.5	0.5	40	0.5	200	100	10	0.5	125
Chapte	r NR 140 E	Enforcement S	Standard (ES)		0.05	5	5	200	5	1000	1000	50	5	250

Notes:

The Sample Level references the typical well depth configuration

Dinitrotoluene, Total (DNT) & VOC results are expressed in micrograms per liter (µg/l)

Sulfate, Total results are expressed in milligrams per liter (mg/l)

Bold values are detected results

Wells listed with (dup) after the name were duplicate samples

Results for Dinitrotoluene, Total were analyzed by SW8270DSIM

J = Analytical result is between the Limit of Detection (LOD) and Limit of Quantitation (LOQ)

NT = Not Tested

A total of ten monitoring wells had NR 140 ES exceedances for total DNT during April 2018. In addition, two monitoring wells had NR 140 PAL exceedances for total DNT during April 2018. A total of eleven monitoring wells had NR 140 ES exceedances for total DNT during September 2018. Total DNT was not detected in any residential wells that were sampled during August 2018.

Three monitoring wells had an NR 140 ES exceedance for sulfate during April 2018. Residential wells are no longer being sampled for sulfate due to historically low detections.

Two residential wells had NR 140 PAL exceedances for TCE during August 2018. No monitoring wells had a TCE detection during 2018. Historically, TCE has not been detected in monitoring wells associated with the DBG Plume. Because TCE can be a residential contaminant and the contamination related to past Army operations has not been present in this area, these residential wells are assumed to not be contaminated by the DBG Plume. Based on Army investigations, some residential wells pumps have been found to contain TCE.

One monitoring well had a NR 140 PAL exceedance for 1,1,2-TCA during April 2018. One residential well had a 1,1,2-TCA detection but below the NR 140 PAL during August 2018.

### 4.5.2.1 *Total Dinitrotoluene*

The horizontal distribution of total DNT is illustrated in Figure 43. The green shaded area displays where total DNT was detected above the NR 140 PAL ( $0.005 \mu g/l$ ). The blue shaded area displays where total DNT was detected above the NR 140 ES ( $0.05 \mu g/l$ ). The red shaded area displays where total DNT was detected above  $1.0 \mu g/l$ . These same color designations are also used in each total DNT cross section. The highest concentration of total DNT detected during 2018 was  $3.005 \mu g/l$  in DBM-8201, which is immediately downgradient of the DBG. The horizontal boundary of the DBG Plume extends from the DBG towards Weigand's Bay but does not reach it. There are two wells shown in Figure 43 that are between the DBG Plume and Weigand's Bay. Total DNT was not detected in these two wells (ELN-1504B and Purcell-D) during 2018. During April 2019, there was a detection of 3,4-DNT and total DNT in residential well Purcell-D that was above the NR 140 ES.

Figure 6 shows the orientation of the contaminant plume isoconcentration cross sections for total DNT, which are illustrated in Figures 44 and 45. Figure 44 (E-E') illustrates the estimated vertical extent of total DNT, along the centerline of the DBG Plume, from the DBG (northwest) towards Weigand's Bay (southeast). The total DNT concentrations adjacent to the DBG (source area) are higher than what is found downgradient. The highest total DNT concentrations are found in wells screened approximately 0 to 30 feet below the water table. The total DNT plume extends northwest to southeast approximately 6,700 feet with an average thickness of 80 feet. Figure 44 shows that the DNT plume is only present in the sand and gravel aquifer and has not migrated downward into the bedrock.

Figure 45 (F-F') illustrates the width and depth of the total DNT plume between 200 to 1,200 feet south of the DBG. The total DNT plume is estimated to be approximately 1,000 feet wide

and a maximum depth of 55 feet below the water table in Figure 45 (F-F'), which is close to the source area.

The Purcell-D residential well is shown on Figure 44 (E-E') at the leading edge of the DBG Plume. The Purcell-D residential well was chosen based on its location along the cross section. The Purcell-D residential well is 112 feet deep and screened in the sand and gravel aquifer. Most of the residential wells in the Weigand's Bay area (downgradient of the DBG Plume) are screened in the sand and gravel aquifer along with a few bedrock wells. Based on the depth and location in relation to the DBG Plume and the historic groundwater monitoring data results, the DBG Plume containing total DNT is migrating toward the residential wells. Results from the August 2018 sampling of 23 residential wells located east and southeast of the DBG Plume did not detect total DNT (see Table 10). However, during April 2019 there was a detection of 3,4-DNT and total DNT in residential well Purcell-D that was above the NR 140 ES.

### 4.5.2.2 *Sulfate*

The horizontal distribution of sulfate is illustrated in Figure 46. The sulfate isoconcentrations are interpreted from the April 2018 groundwater data. Annually during April, 16 monitoring wells are sampled for sulfate. Table 12 summarizes the sulfate groundwater analytical results from the April 2018 sampling event. Since 2013, residential wells are no longer sampled for sulfate due to the historically low detections and the stability of the sulfate near Landfill #5. The green shaded area displays where sulfate was detected above the NR 140 PAL [125 milligrams per liter (mg/l)]. The blue shaded area displays where sulfate detected during April 2018 was 1,100 mg/l in ELN-8203A, which is immediately downgradient of Landfill #5. The limits of the sulfate isoconcentrations are approximately 500 by 850 feet. Due to the limited extent of sulfate detections, cross sections were not prepared. Wisconsin has a "secondary" NR 140 Public Welfare Groundwater Quality Standard for sulfate. The sulfate Chapter NR 140 Groundwater Standard is based on a taste threshold and not based on risk to human health.

### 4.5.2.3 *1,1,2-Trichloroethane*

Concentrations of 1,1,2-trichloroethane (1,1,2-TCA) exceeded the NR 140 PAL in monitoring well ELN-8203B, which is downgradient of Landfill #5 (see Figure 21). Table 12 summarizes the 1,1,2-TCA groundwater analytical results from the April 2018 sampling event. The April 2018 concentration of 1,1,2-TCA in ELN-8203B was 0.98  $\mu$ g/l. 1,1,2-TCA is detected in several other monitoring wells but below the NR 140 PAL (0.5  $\mu$ g/l). 1,1,2-TCA is routinely detected (below the PAL) in the Spear residential well, which is located 2,600 feet southeast of ELN-8203B. Due to the limited extent of 1,1,2-TCA detections, an isoconcentration map or cross section were not prepared.

### 4.5.2.4 *Trichloroethene*

The groundwater results from the August 2018 sampling event show that TCE was detected in three residential wells (Anderson-R, Hendershot, and Wenger). Two of those residential wells had TCE detections above the NR 140 PAL ( $0.5 \mu g/l$ ). All three residential wells are located in

the Weigand's Bay area (downgradient of the DBG Plume), see Figure 20. All three residential wells are screened in the sand aquifer, see Table 5. TCE has routinely been detected in the Anderson-R and Hendershot residential wells since 2007 and the Wenger residential well since 2010. The maximum concentration of TCE ( $4.7 \mu g/l$ ) was detected in the Hendershot residential well during August 2016. This TCE concentration is below the federal MCL and the NR 140 ES of 5  $\mu g/l$  for drinking water. The TCE concentration in the Hendershot residential well during August 2018 was 2.0  $\mu g/l$ .

Table 12 summarizes the TCE groundwater analytical results from the April 2018 monitoring well sampling event. No monitoring wells had a TCE detection during April 2018. Historically, TCE has not been detected in monitoring wells associated with the DBG Plume. There has been no source of TCE identified at BAAP that is upgradient of the Weigand's Bay area. A potential source of TCE is the shallow well jet pump at each residence. During 2012, the Army investigated TCE contamination in the Goelz residential well adjacent to Gruber's Grove Bay. The investigation determined that the well jet pump was the source of TCE contamination in the shallow residential well. Due to the absence of TCE in monitoring wells, TCE is not a COPC in the DBG Plume; therefore, an isoconcentration map or cross section was not prepared.

### 4.5.2.5 *Concentration Graphs*

To evaluate contaminant trend data for the DBG Plume, concentration over time graphs were prepared for select monitoring wells within the plume. Graphs showing DBG Plume contaminant concentration over time are presented in Appendix E. The primary COPC in the DBG Plume is total DNT; therefore, concentrations of total DNT were evaluated for trends. In the source area, data from wells DBM-8201, DBM-8202, and DBN-1001B, C, E were graphed. DBM-8201 shows a generally stable trend with some periods of elevated concentrations. DBM-8202 shows a spike in 2003 followed by a stable to decreasing trend. DBN-1001B shows a decreasing trend from 2014 till 2017. Between 2017 and 2018, the total DNT concentration in DBN-1001B has increased from 0.162 to 0.5978 µg/l. At the center of the DBG Plume, data from wells ELM-8901, ELM-8907, ELM-8908, and ELN-1502A, C were graphed. ELM-8901 shows a steep decreasing trend since 2009. ELM-8907 showed a generally stable trend followed by an increase between 2008 and 2011 then a steadily decreasing trend. ELM-8908 showed some variability then noticeable increases in 2008 and 2013 followed by a decreasing trend since 2014. Data from four wells (ELM-9501 and ELN-0801B, C, E) were used to evaluate the downgradient portion of the plume. All four wells have shown either stable or decreasing trends since 2009. During 2018, total DNT was not detected in ELM-9501 and ELN-0801B, C, E.

ELN-1502A and ELN-1502C were installed in 2015 to provide additional definition of the center of the DBG Plume near the BAAP boundary (see Figure 21). The total DNT concentration in ELN-1502A has been steadily increasing from 0.0087  $\mu$ g/l during December 2015 to 0.801  $\mu$ g/l during September 2018. Data from a nest of wells (ELN-1003A, B, C, E) located at the leading edge of the plume indicates a steady increase of total DNT in ELN-1003B and ELN-1003C. The total DNT concentration in ELN-1003B increased from 0.051  $\mu$ g/l during April 2017 to 0.32  $\mu$ g/l during November 2018. The total DNT concentration in ELN-1003C increased from 0.0085  $\mu$ g/l during April 2017 to 0.278  $\mu$ g/l during November 2018.

### 4.5.3 Central Plume

The source of the Central Plume is suspected to be related to production waste water, which was discharged to open ditches in the rocket paste and rocket propellant areas. The soil in this area has been thoroughly investigated and remediated. No groundwater contamination source was clearly identified. The Central Plume shown in Figure 21 represents the area where groundwater concentrations exceed a NR 140 PAL for 2,6-DNT or total DNT. 2,6-DNT has been routinely detected in either monitoring wells or residential wells in the Central Plume. Because total DNT encompasses all six DNT isomers, total DNT was used to map the Central Plume boundary. The Central Plume boundary shown in Figure 21 is approximate and based on total DNT groundwater data collected during 2018 from both monitoring wells and residential wells. Total DNT has been detected at shallow depths in the sand and gravel aquifer. Table 13 summarizes the groundwater analytical results from the June 2018 monitoring well sampling event. Table 10 summarizes the groundwater analytical results from the August 2018 residential well sampling event. An isoconcentration map and cross section were prepared for total DNT. The isoconcentration map was prepared using all groundwater data collected during 2018. The isoconcentration cross section was prepared using groundwater data collected only during June and August 2018. Since there has been no historical NR 140 ES exceedances for chloroform, an isoconcentration map or cross section were not prepared.

During 2015, 2016, 2017, and 2018, 2,6-DNT, total DNT, and chloroform have been COPCs in the Central Plume. The Central Plume groundwater results from the June and August 2018 sampling events were evaluated for COPCs that exceeded a Chapter NR 140 PAL Groundwater Standard. Six monitoring wells and one residential well had NR 140 ES exceedances for both 2,6-DNT and total DNT during 2018. Three monitoring wells and two residential wells had NR 140 PAL exceedances for chloroform during 2018.

Well	Well ID	Sample Level	Sample Depth (feet)	Sample Date	Dinitrotoluene, Total	Chloroform
DBG PLUME AREA WELLS						
ELN-1003B	468	В	91.5-96.5	Jun-18	0.179	NT
ELN-1003C	469	С	155.1-160.1	Jun-18	< 0.008	NT
ELN-1003C (dup)	469	С	155.1-160.1	Jun-18	< 0.008	NT
ELN-1504B	537	В	34.8-39.8	Jun-18	< 0.008	NT
CENTRAL PLUME AREA WE	CLLS	1				
NLN-1001A	331	А	96-111.5	Jun-18	0.209	NT
NLN-1001C	332	С	149-154.5	Jun-18	0.203	NT
NLN-8203A	258	А	105.5-115.5	Jun-18	< 0.008	NT
NLN-8203B	259	В	125.5-127.5	Jun-18	< 0.008	NT
NLN-8203B (dup)	259	В	125.5-127.5	Jun-18	< 0.008	NT
NLN-8203C	260	С	136.5-138.5	Jun-18	< 0.008	NT
NPM-8901	506	А	80-100	Jun-18	< 0.008	NT
RIM-1003	491	А	99.3-114.3	Jun-18	< 0.008	NT
RIM-1004	494	А	55.5-70.5	Jun-18	< 0.008	NT
RIN-0701C	443	С	175-180	Jun-18	< 0.008	NT
RIN-0702C	444	С	196-201	Jun-18	< 0.008	NT
RIN-0703C	445	С	202-207	Jun-18	< 0.008	NT
RIN-1002A	492	А	77.2-92.2	Jun-18	< 0.008	NT
RIN-1002C	493	С	174.8-179.8	Jun-18	0.062	NT
RIN-1003A	495	А	75.5-90.5	Jun-18	< 0.008	NT
RIN-1004B	498	В	141.7-146.7	Jun-18	0.066	NT
RIN-1004B (dup)	498	В	141.7-146.7	Jun-18	0.066	NT
RIN-1005A	496	А	45.5-60.5	Jun-18	< 0.008	NT
RIN-1005C	497	С	142-147	Jun-18	0.063	NT
RIN-1501B	538	В	113.5-123.5	Jun-18	< 0.008	NT
RIN-1501C	539	С	160.2-165.2	Jun-18	< 0.008	NT
RIN-1501D	540	D	232.8-237.8	Jun-18	< 0.008	NT
RIN-1502B	541	В	98.4-103.4	Jun-18	< 0.008	NT
RIN-1502C	542	С	138.1-143.1	Jun-18	< 0.008	NT
RIN-1502D	543	D	208.3-213.3	Jun-18	< 0.008	NT
RPM-8901	507	А	104.8-124.3	Jun-18	< 0.008	NT
S1111	751	А	78.75-99	Jun-18	< 0.008	NT
SEN-0501A	580	А	17-32	Jun-18	< 0.008	<b>0.17</b> (J)
SEN-0501B	581	В	77-87	Jun-18	< 0.008	0.46
SEN-0501D	582	D	180-190	Jun-18	< 0.008	0.91
SEN-0502A	583	А	18-33	Jun-18	< 0.008	<0.1
SEN-0502D	584	D	177-187	Jun-18	<0.008	0.63
SEN-0503A	585	А	40.5-55.5	Jun-18	<0.008	<0.1
SEN-0503B	586	В	100-110	Jun-18	0.065	<0.1
SEN-0503B (dup)	586	В	100-110	Jun-18	0.064	<0.1
SEN-0503D	587	D	203-213	Jun-18	<0.008	1.2
Chapte	er NR 140 Preven	tive Action Limit (	(PAL)	-	0.005	0.6
Chap	pter NR 140 Enfoi	cement Standard (	(ES)		0.05	6

Notes:

The Sample Level references the typical well depth configuration

All results are expressed in micrograms per liter (µg/l)

Bold values are detected results

Wells listed with (dup) after the name were duplicate samples

Results for Dinitrotoluene, Total were analyzed by SW8270DSIM

J = Analytical result is between the Limit of Detection (LOD) and Limit of Quantitation (LOQ)

NE = Not Established NT = Not Tested

### 4.5.3.1 *Total Dinitrotoluene*

The horizontal distribution of total DNT is illustrated in Figure 47. The green shaded area displays where total DNT was detected above the NR 140 PAL (0.005  $\mu$ g/l). The blue shaded area displays where total DNT was detected above the NR 140 ES (0.05  $\mu$ g/l). These same color designations are used in the total DNT cross section. The highest concentration of total DNT detected during June or August 2018 was 0.209  $\mu$ g/l in NLN-1001A, which is located in northeast corner of the Central Plume (see Figure 21). Figure 47 shows that total DNT in the northern section of the Central Plume, near the source area, has been depleted. Prior to 2017, the Central Plume encompassed a larger area that stretched up to the source area (NPM-8901) and further west towards RIM-1003 and RPM-8901.

Figure 6 shows the orientation of the contaminant plume isoconcentration cross section for total DNT, which is illustrated in Figure 48. Figure 48 (G-G') illustrates the estimated vertical extent of total DNT, along the centerline of the Central Plume, as it migrates towards Gruber's Grove Bay. The total DNT concentrations are highest in the northern portion of the Central Plume and in wells screened within 60 feet below the water table. The total DNT plume extends from the north to the south with an average thickness of 100 feet in the northern and southern sections. The total DNT plume is thinner, 60 feet, within the middle section. Figure 48 indicates that the DNT plume is only present in the sand and gravel aquifer and has not migrated downward into the bedrock.

The WE-UK125 residential well is shown on Figure 48 (G-G'). The WE-UK125 residential well was chosen based on its location along the cross section. The WE-UK125 residential well is screened in the bedrock aquifer, but the majority of the residential wells in the Water's Edge Subdivision are screened in the sand and gravel aquifer. Many of the residential wells located in the Water's Edge Subdivision are screened at the same depth (60 feet below the water table) that the DNT plume occurs. The DNT plume encompasses a portion of the residential wells located in the Water's Edge Subdivision.

### 4.5.3.2 Benzene

The groundwater results from the June 2017 sampling event indicated that benzene was detected in SEN-0503B at a concentration of 10  $\mu$ g/l, which is above the NR 140 ES of 5  $\mu$ g/l. SEN-0503B is located in the Water's Edge Subdivision (see Figure 21). None of the other seven monitoring wells or 16 residential wells in the Water's Edge Subdivision had detections of benzene during 2017. Between 2005 and 2016, benzene had not been detected in SEN-0503B. Since June 2017, SEN-0503B has been sampled twice during 2017 and twice during 2018 with no benzene detections. Benzene was also not detected in any monitoring wells or residential wells that were sampled during 2018. The source of the benzene is unknown. However, there is no evidence to suggest that these past benzene detections are attributable to the Army. Therefore, benzene is not considered to be a COPC.

### 4.5.3.3 *Concentration Graphs*

To evaluate contaminant trend data for the Central Plume, concentration over time graphs were prepared for select monitoring wells within the plume. Concentration over time graphs are provided in Appendix E. The primary COPC in the Central Plume is total DNT; therefore, concentrations of total DNT were evaluated. Seventeen of the 23 wells selected showed stable to decreasing trends throughout the plume. The other six wells (NLN-1001A, NLN-1001C, RIN-1002C, RIN-1005C, RIN-1004B, and SEN-0503B) have shown increasing total DNT concentrations. The total DNT concentrations in these six wells increased from below the NR 140 ES during June 2017 to above the NR 140 ES during June 2018.

### 4.5.4 Nitrocellulose Production Area Plume

The source of the NC Area Plume is believed to be from various nitrocellulose production buildings in the northwest section of BAAP. The NC Area Plume shown in Figure 21 represents the area where groundwater concentrations exceed a NR 140 PAL or ES for 2,6-DNT or total DNT. 2,6-DNT has been routinely detected in monitoring wells in the NC Area Plume. Because total DNT encompasses all six DNT isomers, total DNT was used to map the NC Area Plume boundary. The NC Area Plume boundary shown in Figure 21 is approximate and based on total DNT groundwater data collected during 2018 from only monitoring wells. There are no residential wells located near the NC Area Plume. Total DNT has only been detected at shallow depths in the sand and gravel aquifer. Table 11 summarizes the groundwater analytical results from the September 2018 monitoring well sampling event. An isoconcentration map and cross section were prepared for total DNT. The isoconcentration cross section was prepared using groundwater data collected during 2018. The isoconcentration cross section was prepared using groundwater data collected during 2018.

During 2015, 2016, 2017, and 2018, 2,6-DNT and total DNT have been COPCs in the NC Area Plume. The NC Area Plume groundwater results from the September 2018 sampling event were evaluated for COPCs that exceeded a Chapter NR 140 PAL Groundwater Standard. Three monitoring wells had an NR 140 ES exceedance for both 2,6-DNT and total DNT. In addition, two monitoring wells had a NR 140 PAL exceedance for both 2,6-DNT and total DNT.

### 4.5.4.1 *Total Dinitrotoluene*

The horizontal distribution of total DNT is illustrated in Figure 49. The green shaded area displays where total DNT was detected above the NR 140 PAL (0.005  $\mu$ g/l). The blue shaded area displays where total DNT was detected above the NR 140 ES (0.05  $\mu$ g/l). This same color designation is used in the total DNT cross section. The highest concentration of total DNT detected during September 2018 was 0.22  $\mu$ g/l in RIM-1002, which is located in the northern section of the NC Area Plume.

Figure 6 shows the orientation of the contaminant plume isoconcentration cross section for total DNT, which is illustrated in Figure 50. Figure 50 (H-H') illustrates the estimated vertical extent of total DNT, along the centerline of the NC Area Plume, as it migrates south. The total DNT concentrations are highest in wells screened at the water table. The total DNT plume extends

from the north to the south with an average thickness of 30 feet. Figure 50 indicates that DNT has not migrated vertically into the monitoring wells screened 80 feet below the water table.

### 4.5.4.2 *Concentration Graphs*

To evaluate contaminant trend data for the NC Area Plume, concentration over time graphs were prepared for select monitoring wells within the plume. Concentration over time graphs are provided in Appendix E. The primary COPC in the NC Area Plume is total DNT; therefore, concentrations of total DNT were evaluated. The total DNT concentration in RIM-1002 has increased from 0.045  $\mu$ g/l during September 2016 to 0.22  $\mu$ g/l during September 2018. Four of the five wells selected showed stable to decreasing trends throughout the plume. These four wells are located downgradient of RIM-1002. There are no residential wells located near the NC Area Plume.

### 4.6 Residential Well Replacement

### 4.6.1 Propellant Burning Ground Plume

The Army has replaced three residential wells due to impacts from chlorinated solvents. All three residential wells were located in the southern portion of the PBG Plume (see Figure 20). CTET was detected above the NR 140 ES in the Kirner (former Gruber-South) residential well during April 1990. The Kirner (former Gruber-South) replacement residential well, located on Hwy 78, was installed by the Army in 1990. CTET was detected above the NR 140 ES in the Mueller-J residential well during April 1990. The Mueller-J replacement residential well, located on Hwy 78, was installed by the Army in 1990. The Mueller-J replacement residential well, located on Hwy 78, was installed by the Army in 1990. CTET and chloroform were detected above the NR 140 ES in the Lins-K residential well during April 1990. The Lins-K replacement residential well, located on County Road Z, was installed by the Army in 1996. Prior to well replacement, bottled water had been provided to the affected residences.

### 4.6.2 Deterrent Burning Ground Plume

The Army has replaced one residential well due to impacts from total DNT. The Purcell-D residential well is located at the southeastern extent of the DBG Plume (shown in Inset A on Figure 20). The Purcell-D residential well, located on Hwy 78, was replaced by the Army in July 2019. During April 2019, 3,4-DNT and total DNT were detected at 0.056  $\mu$ g/l in the Purcell-D residential well. This total DNT concentration was above the NR 140 ES. The contaminated residential well was screened in the sand aquifer down to 112 feet deep. The replacement well was drilled into the lower bedrock aquifer, sealing off the upper sand aquifer with a grouted steel casing. Prior to well replacement, bottled water was being provided to the affected residence.

### 4.6.3 Central Plume

The Army has replaced three residential wells due to impacts from 2,6-DNT. All three residential wells were located in the southern portion of the Central Plume and in the Water's Edge Subdivision (see Inset B on Figure 20). During 2004, the 2,6-DNT and total DNT

concentrations in two residential wells exceeded the NR 140 ES. In 2005, the Army replaced the WE-RM385 and WE-RR541 residential wells with WE-SQ017 and WE-SQ001, respectively. The original residential wells were screened in the sand and gravel aquifer down to 100 feet deep. The replacement residential wells were also screened in the sand and gravel aquifer but at 180 feet deep. During June 2018, the 2,6-DNT and total DNT concentrations in the WE-UK124 residential well exceeded the NR 140 ES. In 2018, the Army replaced the WE-UK124 residential well with WE-ZE512. The original residential well (WE-UK124) was screened in the sand and gravel aquifer down to 100 feet deep. The replacement well (WE-ZE512) was drilled into the lower bedrock aquifer, sealing off the upper sand aquifer with a grouted steel casing. Prior to well replacement, bottled water had been provided to the affected residences.

### 5.0 GROUNDWATER HUMAN HEALTH RISK ASSESSMENT

A groundwater human health risk assessment (HHRA) was performed to determine and document whether groundwater contamination originating from BAAP posed a potential current or hypothetical future risk to human health. Because the RI only pertains to groundwater contamination, the focus of the HHRA was only groundwater. Soil remedial actions conducted by the Army and property restrictions have minimized the potential exposure of soil contamination to human health based on the anticipated future land use at the former BAAP.

Source areas and associated contaminant plumes for BAAP are shown on Figure 1. This HHRA provides risk managers information for determining whether unacceptable human health risks (cancer and non-cancer health hazards) might be caused by exposure to contaminants in the groundwater such that additional evaluation or action is necessary. The HHRA addresses human exposure pathways related to groundwater including the potential for vapor intrusion and for potable use or other domestic purposes. The Army will use the HHRA results in determining the scope of any response action(s) undertaken to address contaminants in the groundwater caused by past Army activities at BAAP.

### 5.1 Risk Assessment Overview

A HHRA is required to be completed as part of a remedial investigation/feasibility study under CERCLA to evaluate the potential human health risks associated with chemical exposure to environmental media (e.g., groundwater). This HHRA was conducted using standard USEPA risk assessment guidance, exposure assumptions, and toxicity factors. The USEPA HHRA process uses conservative assumptions about exposure to chemicals and their toxicity so that risks reported within this HHRA will not be underestimated. In all circumstances, priority is given to evaluating the potential human health risk regardless of the impact.

Risk assessments generally make risk estimates for defined groups or populations. The term receptor is often used to designate people who may be exposed to an environmental hazard and to whom the HHRA would be directed. Identification of receptor location and pathways by which they might be exposed is an integral part of any HHRA.

The focus of this HHRA is related to groundwater and the risk it may pose to humans. The HHRA does not address any potential risks associated with the direct exposure to contaminated soil or ecological receptors. For some media such as soil, the potential for exposure does not currently exist.

A screening level groundwater risk evaluation was conducted for each of the four plumes using USEPA human health risk assessment methods (USEPA 1989, 1991). The screening risk evaluation was conducted in two steps. First, site chemical concentrations were compared to health-based screening levels to identify chemicals of potential concern (COPCs). Second, risk estimates were calculated for COPCs that exceeded screening levels. The risk estimates were then compared to risk management criteria to put the magnitude of the risks into perspective.

The following four sections provide plume-specific screening level groundwater risk evaluations for each of the four plume areas. The risks for all four plume areas are assessed for the on-site portion of the plume and the off-site portion of the plume.

### 5.2 Identification of Exposure Pathways

As defined in the Risk Assessment Guidance for Superfund, Volume 1, Human Health Evaluation Manual, Part A (USEPA 1989), an exposure pathway is composed of the following elements:

- A source and mechanism of chemical release to the environment
- An environment transportation medium (e.g. groundwater) for the released chemical and/or mechanism to transfer the chemical from one medium to another
- A point of potential contact by humans with contaminated medium
- A route of exposure (i.e., ingestion, inhalation or dermal)

The exposure routes associated with domestic use of water include ingestion, inhalation, and dermal exposure.

- **Ingestion** Potential chemical exposure because of groundwater use for potable purposes for drinking water and preparing food.
- **Inhalation** Potential exposure due to groundwater use for domestic purposes resulting in inhalation of the contaminants during activities such as bathing, food preparation, and dishwashing.
- **Dermal** Potential use of groundwater resulting in chemical absorption through skin during activities such as washing hands and bathing.

### 5.3 Screening Level Groundwater Risk Evaluation

A Screening Level Groundwater Risk Evaluation (Draft) was conducted by Exponent, see Appendix G. The objectives of the risk evaluation were to estimate current and hypothetical future risks using groundwater quality data from existing residential and off-site monitoring wells (current risk) and on-site groundwater monitoring wells (hypothetical future risk). Exponent's screening level groundwater risk evaluation was conducted using standard USEPA risk assessment methods. The conservative calculations for this risk screening overestimate the actual risk. The maximum concentrations of analytes associated with each plume for current (residential wells and off-site monitoring wells) and hypothetical future (on-site monitoring wells) scenarios were used to estimate the risks.

### 5.3.1 Data Collection and Evaluation

Groundwater quality data for residential wells and off-site monitoring wells monitored by the Army from the past four years (2015, 2016, 2017, and 2018) were used to evaluate current groundwater exposure and risk. To evaluate hypothetical future groundwater exposure and risk,

on-site monitoring well data collected over the past four years (same period) were used. The past four years of groundwater quality data was selected to best represent current groundwater quality conditions.

The maximum concentrations of contaminants in the wells associated with each plume were used to estimate the risks. In other words, the highest concentrations of each chemical in each plume (i.e., many wells) were evaluated as if they had occurred in any one well. Consequently, the risks represented should be viewed as upper bound estimates of potential groundwater risks within a specific plume area.

### 5.3.2 Chemicals of Potential Concern Selection Process

The first step in the screening level groundwater risk evaluation is to select those chemicals that exceed screening levels for groundwater. Both USEPA's tapwater RSLs and Wisconsin NR 140 Groundwater Standards (ES and PAL) were used to screen chemicals. A comprehensive summary of the groundwater screening levels is provided in the Screening Level Groundwater Risk Evaluation (Draft) included in Appendix G. The determination of which groundwater standard was used to screen each chemical is further defined in Table 1 of Appendix G. A Total Hazard Quotient (THQ) Table THQ = 0.1 was used for screening as it is recommended by the USEPA for sites with multiple chemicals. Chemicals or analytes that exceeded the lowest available groundwater screening value are referred to as the COPCs. The maximum concentration of the chemicals detected in each plume were compared to the lowest of the groundwater screening values. These COPCs were retained for further risk evaluation and calculations.

### 5.3.3 Exposure Assessment and Assumption

USEPA's tapwater RSLs are risk-based concentrations developed using specific generic exposure assumptions that represent reasonable maximum exposure (RME) to groundwater. Exposure to chemicals in groundwater are incorporated into the tapwater RSLs for both ingestion and dermal contact with the water, as well as inhalation of the portion of the chemicals in groundwater that are volatilized from the water as it is used (e.g., for bathing). Tapwater RSLs based on non-cancer effects are also developed separately for adults and children, and then the lower of the two RSLs is selected for evaluating risks to people. RSLs based on cancer incorporate exposure during both childhood and adulthood. For this reason, the tapwater RSLs are considered a conservative risk-based benchmark on which to estimate risk associated with groundwater chemical exposure.

The groundwater risk evaluation was performed using tapwater RSLs that incorporate RME factors that characterize how adults and children are assumed to be exposed to groundwater. Some of the key exposure assumptions used to develop the tapwater RSLs are listed below.

Assumption	Adult	Child
Exposure Duration (cancer effects)	20 years	6 years
Exposure Duration (non-cancer effects)	26 years	6 years
Exposure Frequency	350 days/year	350 days/year
Water Ingestion Rate	2.5 liters/day	0.78 liter/day
Hours/day Air Inhaled	24 hours/day	24 hours/day
Body Weight	80 kilograms	15 kilograms
Averaging Time (cancer effects) [averaging of exposure is integrated over a person's lifetime]	25,550 days (i.e., 365	5 days/year x 70 years)
Averaging Time (non-cancer effects) [averaging occurs over the adult or child's exposure duration]	9,490 days (i.e., 365 days/year x 26 years)	2,190 days (i.e., 365 days/year x 6 years)

<b>USEPA</b> Tapwater	r Regional Screeni	ng Levels (RSLs)	) Exposure	Assumptions
-----------------------	--------------------	------------------	------------	-------------

### 5.3.3.1 Cancer Risk Characterization

Cancer risks were estimated for each COPC related to each plume. The cancer risk is the probability that an individual will develop cancer due to chemical exposure in the groundwater over their lifetime. This probability of contracting cancer due to chemical exposure represents the incremental increase in the probability of developing cancer during one's lifetime above and beyond the background probability of developing cancer. For example,  $1x10^{-6}$  represents a one in a million chance of contracting cancer. This cancer risk is in addition to the general background level risk of contracting cancer of any kind during one's lifetime unrelated to groundwater chemical exposure. Based on the USEPA's National Contingency Plan, cumulative carcinogenic risk below  $1x10^{-6}$  are generally considered to represent a negligible risk, cumulative risks between  $1x10^{-6}$  and  $1x10^{-4}$  are within a range considered acceptable under most conditions and cumulative cancer risk above  $1x10^{-4}$  indicate unacceptable levels of risk where potential action or further evaluation needs to be considered.

In off-site areas, where the Army does not have control over the use of the groundwater as a drinking water source, a cumulative cancer risk greater than  $1 \times 10^{-6}$  is cause for potential action or additional evaluation. For areas within the BAAP property, where the Army has control over the use of groundwater as a drinking source, a cumulative cancer risk greater than  $1 \times 10^{-4}$  is cause for potential action or additional evaluation.

### 5.3.3.2 Non-Cancer Risk Characterization

Non-cancer risks were estimated for each COPC related to each plume. For non-cancer effects, the likelihood that a receptor will develop an adverse effect other than cancer (e.g., kidney disease) is estimated by comparing the predicted level of exposure for a chemical with the highest level of exposure that is considered protective. The chemical-specific non-cancer risk is represented by a hazard quotient (HQ) value, which is derived by comparing the groundwater chemical concentrations to the chemical-specific tapwater RSLs. If an HQ value is less than or

equal to one, then adverse health effects associated with exposure to that chemical in the groundwater are unlikely to occur even among sensitive individuals (e.g., children). An HQ greater than one indicates that there is the potential for a health effect and that additional analysis is necessary. The sum of all non-cancer risks (i.e., each HQ for each COPC) within an area or plume is referred to as the hazard index (HI). An HI greater than one indicates a level of exposure that needs to be evaluated further to determine if a health concern exists.

### 5.3.3.3 *Risk Calculations*

The default tapwater RSL values provided in the USEPA's RSL Resident Tapwater Generic Table (November 2018) were used to calculate the risks. Groundwater risk estimates were calculated for each plume using a simple scaling method developed by the USEPA. For each COPC, the calculations described below were used to estimate potential cancer and non-cancer risks.

Cancer Risk: (Groundwater Concentration x Target Cancer Risk)/RSL for Tapwater Non-cancer HQ: (Groundwater Concentration x Target Hazard Quotient)/RSL for Tapwater

The target cancer risk that the RSL is based upon is  $1 \times 10^{-6}$  and the target hazard quotient is 0.1 as recommended by USEPA since multiple contaminants are present at the site.

### 5.3.3.4 *Risk Evaluation Results*

A comprehensive summary of the groundwater risk calculations is provided in the Screening Level Groundwater Risk Evaluation (Draft) included in Appendix G. The total DNT concentration represents the sum of all isomers of DNT detected in the water sample. The risk associated with DNT was evaluated for both total DNT and individual isomers. The higher of the two risk estimates (i.e., based on total or the sum of the individual isomers) were used in calculating the total risk for each plume area.

### 5.4 **Propellant Burning Ground Plume**

### 5.4.1 Characterization of Exposure Settings

The sources of the PBG Plume are in the southwestern portion of BAAP, see Figure 1. The PBG sources are comprised of the PBG Waste Pits, 1949 Pit, Racetrack Area, and Landfill #1. The Army has covered each of the PBG source areas with either an engineered cap or soil cover to inhibit the movement of contaminants in the soil to the groundwater. The PBG Waste Pits and 1949 Pit became active sometime between 1942 and 1949 and were last used in 1983. A clay and geomembrane barrier cap was installed over the 1949 Pit in 1998 and the PBG Waste Pits in 2008. The Racetrack Area consisted of a series of burning pads, plates and pits that were used from 1949 to 1994. In 1995, three-fourths of the Racetrack Area was covered with soil to prevent contact with residual lead in the soil. Contaminated soil was removed from the remaining portion of the Racetrack Area in 1997. Landfill #1 is a closed demolition debris disposal facility located east of the PBG Waste Pits that was used between 1942 and 1959. A

composite cap (two feet of clay and geomembrane barrier) was installed over Landfill #1 in 1997. Section 3.1 provides additional details on remediation activities of the source areas.

DNT and volatile organic solvents (e.g. carbon tetrachloride, chloroform, and trichloroethene) are known to have been disposed of at the PBG through open burning and burial during production periods. Contamination from these disposal and open burning activities migrated through the soil and into the groundwater. Groundwater beneath the PBG source areas is approximately 105 feet deep. The contaminated groundwater in the PBG Plume has migrated south to southeast (off-site) and then discharges into the Wisconsin River, see Figure 1. As the PBG Plume migrates away from the source area, it sinks lower into the sand aquifer. The offsite portion of the PBG Plume sinks deeper into the sand aquifer. Groundwater in the PBG Plume travels approximately 306 feet per year. Contaminants in the PBG Plume are expected to travel at the same speed as groundwater. Groundwater beneath the off-site residential areas is approximately 80 feet deep. Contaminated groundwater (above the NR 140 ES) in the off-site portion of the PBG Plume has been identified within the sand aquifer at depths between 80 and 210 feet. The sand aquifer extends down to 210 feet. The three residential wells located within the areal extent of the PBG Plume range in depth from 240 to 534 feet and are screened in the bedrock. These bedrock residential wells draw their groundwater from beneath the contaminated portion of the PBG Plume. The residential wells located outside the areal extent of the PBG Plume range in depth from 122 to 310 feet and average 250 feet deep. Over half the residential wells located outside of the PBG Plume are screened in the bedrock. Both DNT and VOCs have been detected in monitoring wells located in the PBG Plume. Sections 4.4 and 4.5 provide additional details on groundwater properties and groundwater contaminants.

The Army has performed various soil remediation activities at the PBG source areas (bioremediation, soil excavation, and soil vapor extraction). The Army performed groundwater remediation using a groundwater pump and treat system from 1990 to 2015 at the source areas and downgradient of the source areas. The pump and treat system influenced the groundwater flow in the PBG Plume by drawing groundwater downward within the area of pumping influence. The pump and treat system also reduced the off-site migration of the PBG Plume when it was operational. Since pumping was stopped in 2015, the portion of the PBG Plume near the BAAP boundary has shifted eastward toward residential wells. Over the past 24 months, the groundwater table beneath the PBG source areas has risen six feet. This rise in groundwater has resulted in an increase of DNT concentrations directly downgradient (south) of the source areas. Sections 4.4 and 4.5 provide additional details on groundwater properties and groundwater contaminants.

A graphical depiction of the PBG Plume in relationship to the local geology, monitoring wells, residential wells, site features, and groundwater plume boundaries is shown on the subsequent page. The groundwater flow direction is from the upper left (north) towards the right (southeast). The groundwater contaminant plume is shown below the water table and migrates into the Wisconsin River. The groundwater contaminant plume is shown to have traveled past the BAAP property and beneath a residential area.



A groundwater conceptual site model (CSM) for the PBG Plume is provided in Appendix H. The CSM shows the relationship between the sources of contamination, how the contamination is transported, type of media exposure, the route of exposure, and who may be exposed. The contaminants infiltrated through the soil (leaching) below the waste disposal areas until they reached the groundwater. The contaminants within the groundwater have been transported with the directional flow of groundwater into the Wisconsin River to the south-southeast. Contaminated groundwater has the potential to reach residential wells which may be used for domestic or potable purposes. Residential well users can be exposed to contaminated groundwater through ingestion or drinking of water, inhalation of vapor during bathing or dishwashing, and dermal contact while bathing.

The exposure routes associated with domestic use of water, as shown on the CSM (Appendix H), include ingestion, inhalation, and dermal exposure. The off-site residential use pathways are potentially complete under current land use conditions and warrant further evaluation. The on-site hypothetical future residential use pathways are incomplete or considered insignificant under current land use conditions but potentially complete under hypothetical future on-site groundwater usage.

The exposure route associated with vapor intrusion, as shown on the CSM, includes only inhalation of indoor vapors. Both the off-site and on-site residential use vapor intrusion exposure pathways are incomplete or considered insignificant based on past vapor intrusion investigations and so no further evaluation is warranted. Section 5.4.2 discusses the potential vapor intrusion exposure into buildings.

### 5.4.2 Exposure Quantification – Vapor Intrusion Pathway Analysis

An evaluation was conducted to determine whether vapors from PBG Plume of groundwater contamination pose a current or hypothetical future risk to human health. Vapor intrusion occurs when there is a migration of vapor-forming chemicals from a subsurface source (i.e., contaminated groundwater) into an overlying building. The exposure route evaluated was the inhalation of contaminants from indoor air.

The subsurface contaminants that have the greatest potential to pose a health concern via vapor intrusion, based upon their volatility, and potential hazards is provided in the USEPA's OSWER Technical Guide for Assessing and Mitigating the Vapor Intrusion Pathway from Subsurface Vapor Sources to Indoor Air [OSWER Publication 9200.2-154, June 2015]. The USEPA's OSWER Technical Guide specifies that a chemical generally is "volatile" if: 1) vapor pressure is greater than 1 millimeter of mercury (mm Hg), or 2) Henry's law constant (ratio of a chemical's vapor pressure in air to its solubility in water) is greater than 10<sup>-5</sup> atmosphere-meter cubed per mole (atm m<sup>3</sup> mol<sup>-1</sup>). Common vapor-forming chemicals are VOCs, such as carbon tetrachloride, gasoline compounds, and trichloroethene. Other compounds such as the six DNT isomers (2,3-, 2,4-, 2,5-, 2,6-, 3,4-, and 3,5-) are not as volatile and are semi-volatile organic compounds (SVOCs).

The USEPA's OSWER specifications were compared to the chemical properties of the six DNT isomers. The vapor pressures for all six DNT isomers are well below 1 mm Hg. Also, Henry's

law constants for all DNT isomers are well below 10<sup>-5</sup> atm m<sup>3</sup> mol<sup>-1</sup>. Therefore DNT is not likely to volatilize from the groundwater into soil and therefore does not pose a vapor pathway risk. Based on this information, DNT does not contribute to vapor intrusion risk related to human health. In addition to DNT, the PBG Plume contains VOCs that could be considered a vapor intrusion risk; therefore, further evaluation of VOC vapor intrusion was conducted.

During 2012, the Army conducted two vapor intrusion pathway analysis investigations associated with the PBG Plume. Copies of these investigation reports are enclosed in Appendix F. The goal of the vapor intrusion pathway analysis was to evaluate if VOCs in the groundwater could vertically migrate through the subsurface and into buildings. Vapor sampling was conducted at eight locations south of BAAP, within/near the PBG Plume. The off-site locations were also positioned near current residential properties. Vapor samples were collected using the post-run tubing vapor sampling technique in accordance with WDNR vapor intrusion guidance, Addressing Vapor Intrusion at Remediation & Redevelopment Sites in Wisconsin, PUB-RR-800, December 2010. The vapor samples were collected through soil borings drilled approximately 40 feet below ground surface. The groundwater depth for the sampling locations was approximately 80 feet below ground surface. The vapor samples were laboratory analyzed for chloroform, carbon tetrachloride, and trichloroethene by the Wisconsin State Laboratory of Hygiene. Based on groundwater samples collected from monitoring wells located within the offsite portion of the PBG Plume, only chloroform, carbon tetrachloride, and trichloroethene were detected. Evaluation or laboratory analysis of other VOCs was not warranted during the 2012 vapor intrusion pathway analysis investigations.

Analytical results of soil gas samples collected off-site did not exceed the 2011 WDNR Vapor Regional Screening Levels (RSLs) for Deep Soil Gas. The 2012 vapor intrusion pathway analysis reports concluded that the PBG Plume does not present a risk to human health via vapor intrusion off-site of BAAP. Because the vapor sample analysis results for chloroform, carbon tetrachloride, and trichloroethene did not exceed the RSLs, additional investigation (e.g., subslab, indoor air) of the vapor pathway was not warranted.

Based on the vapor intrusion pathway analysis investigations conducted during 2012, inhalation exposure due to soil gas vapor intrusion from the PBG Plume does not pose a current or potential future risk to area residents.

### 5.4.3 Exposure Quantification – Groundwater Pathway Analysis

### 5.4.3.1 Current and Potential Future Uses of Groundwater

Groundwater located in the PBG Plume within the boundary of BAAP is not used for human consumption. The land that was transferred from the Army to other property owners includes a deed restriction on the use of groundwater and so restricts the potential exposure to groundwater within the boundary of BAAP. These groundwater access restrictions state that the property owner "shall not access or use groundwater underlying the property for any purpose without the prior written approval of the Army and the WDNR".
Residential wells located outside of BAAP use groundwater for potable water and domestic purposes. The potential future use of groundwater adjacent to and downgradient of BAAP is expected to be for potable water and domestic purposes.

Residential well users can be exposed to contaminated groundwater through ingestion or drinking of water, inhalation of vapor during showering or dishwashing, and dermal contact while bathing. Groundwater contaminants from BAAP impacted three residential wells located in the PBG Plume. The Army replaced these three residential wells due to impacts from VOCs (carbon tetrachloride and/or chloroform). Section 4.6 provides additional details on the residential well replacements conducted by the Army.

## 5.4.4 Risk Evaluation Summary

## 5.4.4.1 Hypothetical Future On-Site Groundwater Risks

The cumulative cancer risk, non-cancer HI and contaminants of concern related to hypothetical future on-site risks associated with the PBG Plume are summarized below. Contaminants of concern are analytes found to significantly contribute to the cumulative risk in an area where risk was estimated to be above the risk management criteria (cumulative cancer risk >  $1 \times 10^{-4}$  or HI > 1).

Location	Location Cumulative Cancer Risk		Contaminants of Concern	
On-Site (Hypothetical Future Risk)	6x10 <sup>-3</sup>	53	2,6-DNT Ethyl Ether Trichloroethene	

#### Summary of Hypothetical Future Risk – Propellant Burning Ground Plume (On-Site Monitoring Well Data)

Risks calculated using the simple scaling method for a hypothetical future residential scenario, along with the maximum observed concentration of each COPC, yielded cumulative cancer risk estimates above the risk management criterion for the on-site portion of the PBG Plume. The cumulative cancer risk ( $6x10^{-3}$ ) for the PBG Plume was above the risk management criterion ( $1x10^{-4}$ ). The contaminant of concern that contributed to the cumulative cancer risk for the PBG Plume was 2,6-DNT.

The calculated non-cancer HI of 53 was above the risk management criterion (HI > 1) in the onsite portion of the PBG Plume. The contaminants of concern that contributed to the HI > 1 in the PBG Plume were 2,6-DNT, ethyl ether and trichloroethene.

Based on the maximum risk scenario, the on-site portion of the PBG Plume represents an area that, if groundwater migrated off-site would be associated with cumulative groundwater risks above the risk management criteria (cumulative cancer risk above  $1 \times 10^{-4}$  and non-cancer HI

above 1). The on-site portion of the PBG Plume has the potential to migrate off-site, thus impacting downgradient residential wells. Further evaluation of the risks is provided in Section 8.0 Remedial Alternative Development Process.

# 5.4.4.2 *Current Off-Site Groundwater Risks*

The cumulative cancer risk, non-cancer HI and contaminants of concern related to current offsite groundwater risks associated with the PBG Plume are summarized below. Contaminants of concern are analytes found to significantly contribute to the cumulative risk in an area where risk was estimated to be above the risk management criteria (cumulative cancer risk >  $1x10^{-6}$  or HI > 1).

Location	Cumulative Cancer Risk	Non-cancer Hazard Index (HI)	Contaminants of Concern
Off-Site (Current Risk)	1x10 <sup>-4</sup>	5	2,6-DNT Carbon Tetrachloride Chloroform Trichloroethene

# Summary of Current Risk – Propellant Burning Ground Plume (Residential Well and Off-Site Monitoring Well Data)

Risks calculated using the simple scaling method for a current residential scenario, along with the maximum observed concentration of COPC, yielded cumulative cancer risk estimates above the risk management criterion for the off-site portion of the PBG Plume. The cumulative cancer risk  $(1x10^{-4})$  for the PBG Plume was above the risk management criterion  $(1x10^{-6})$ . The contaminants of concern that contributed to the cumulative cancer risk for the PBG Plume were 2,6-DNT, carbon tetrachloride, chloroform, and trichloroethene.

The calculated non-cancer HI of 5 was above the risk management criterion (HI > 1) in the offsite portion of the PBG Plume. The contaminant of concern that contributed to the HI > 1 in the PBG Plume was trichloroethene.

Based on the maximum risk scenario, the off-site portion of the PBG Plume is associated with cumulative groundwater risks above the risk management criteria (cumulative cancer risk above  $1 \times 10^{-6}$  and non-cancer HI above 1). Further evaluation of the risks is provided in Section 8.0 Remedial Alternative Development Process.

# 5.5 Deterrent Burning Ground Plume

# 5.5.1 Characterization of Exposure Settings

The sources of the DBG Plume are in the northeastern portion of BAAP, see Figure 1. The DBG sources are comprised of the DBG (waste pits), Landfill #3, and Landfill #5. From the 1940s to the 1970s, liquid deterrent, comprised mostly of DNT, is known to have been burned and

disposed of at the DBG. During the same period, coal ash from the power plant, construction debris, trash, and burned garbage were disposed of in Landfill #3. In 2003, a geosynthetic clay and geomembrane barrier were installed above the DBG and Landfill #3 as one contiguous cap. In addition, an enhanced biodegradation system was operated at this site from 2003 to 2008. From 1979 to 1988, solid waste including office and laboratory waste, demolition debris, and coal ash were disposed of in Landfill #5. In 1988, Landfill #5 was closed with a clay barrier cap. The Army has covered each of the DBG source areas with an engineered cap to inhibit the movement of contaminants in the soil to the groundwater. The Army has performed various soil remediation activities at the DBG (bioremediation and soil excavation). The Army has not performed any groundwater remediation in the DBG Plume. Section 3.2 provides additional details on remediation activities of the source areas.

Contamination from these disposal and open burning activities migrated through the soil and into the groundwater. Groundwater beneath the DBG source areas is approximately 130 feet deep. The contaminated groundwater in the DBG Plume has migrated southeast (off-site) towards the Wisconsin River (Weigand's Bay), see Figure 1. As the DBG Plume migrates away from the source area, it sinks lower into the sand aquifer. The off-site portion of the DBG Plume sinks below the groundwater surface and deeper into the sand aquifer. Groundwater in the DBG Plume travels approximately 109 feet per year. Contaminants in the DBG Plume are expected to travel at the same speed as groundwater. Groundwater beneath the off-site residential areas is approximately 25 feet deep. Contaminated groundwater (above the NR 140 ES) in the off-site portion of the DBG Plume has been identified within the sand aquifer at depths between 50 and 180 feet. The sand aquifer extends down to 216 feet. The residential wells located outside the areal extent of the DBG Plume range in depth from 20 to 260 feet and average 100 feet deep. Most of the residential wells located outside the areal extent of the DBG Plume are screened in the sand. Total DNT has been detected in monitoring wells located both on-site and off-site in the DBG Plume. Over the past three years, the total DNT concentrations in off-site monitoring wells (ELN-1003B and ELN-1003C) have been increasing. These increases indicate that the DBG Plume is migrating off-site (southeast) towards residential wells located near Weigand's Bay. During April 2019, total DNT was detected in a residential well above the NR 140 ES. During July 2019, the Army replaced one residential well associated with the DBG Plume that was impacted by total DNT. Sections 4.4 and 4.5 provide additional details on groundwater properties and groundwater contaminants. Section 4.6 provides additional details on the residential well replacement conducted by the Army.

A graphical depiction of the DBG Plume in relationship to the local geology, monitoring wells, site features, and groundwater plume boundaries is shown on the subsequent page. The groundwater contaminant plume is shown below the water table and migrating towards the Wisconsin River. The groundwater contaminant plume is shown to have traveled past the BAAP property.



A groundwater CSM for the DBG Plume is provided in Appendix H. The CSM shows the relationship between the sources of contamination, how the contamination is transported, type of media exposure, the route of exposure, and who may be exposed. The contaminants infiltrated through the soil (leaching) below the waste disposal areas until they reached the groundwater. The contaminants within the groundwater have been transported with the directional flow of groundwater towards the Wisconsin River (Weigand's Bay) to the southeast. Contaminated groundwater has the potential to reach residential wells which may be used for domestic or potable purposes. Residential well users can be exposed to contaminated groundwater through ingestion or drinking of water, inhalation of vapor during bathing or dishwashing, and dermal contact while bathing.

The exposure routes associated with domestic use of water, as shown on the CSM (Appendix H), include ingestion, inhalation, and dermal exposure. The off-site residential use pathways are potentially complete under current land use conditions and warrant further evaluation. The on-site hypothetical future residential use pathways are incomplete or considered insignificant under current land use conditions but potentially complete under hypothetical future on-site groundwater usage.

The exposure route associated with vapor intrusion, as shown on the CSM, includes only inhalation of indoor vapors. Both the off-site and on-site residential use vapor intrusion exposure pathways are incomplete or considered insignificant based on past vapor intrusion investigations and so no further evaluation is warranted. Section 5.5.2 discusses the potential vapor intrusion exposure into buildings.

# 5.5.2 Exposure Quantification – Vapor Intrusion Pathway Analysis

An evaluation was conducted to determine whether vapors from the DBG Plume of groundwater contamination pose a current or hypothetical future risk to human health. Vapor intrusion occurs when there is a migration of vapor-forming chemicals from a subsurface source (i.e., contaminated groundwater) into an overlying building. The exposure route evaluated was the inhalation of contaminants from indoor air.

The Army did not conduct a vapor intrusion pathway analysis investigation specifically in the DBG Plume. Section 5.4.2 discussed the 2012 vapor intrusion pathway analysis investigations conducted by the Army in the PBG Plume. The PBG Plume represents the worst-case scenario for volatile-forming chemicals present in the groundwater and thus provides a conservative representation of vapor conditions associated with the DBG Plume. The 2012 vapor intrusion pathway analysis reports concluded that VOCs in the PBG Plume do not present a risk to human health via vapor intrusion.

Based on the information in Section 5.4.2, inhalation exposure due to soil gas vapor intrusion from the DBG Plume does not pose a current or potential future risk to area residents.

## 5.5.3 Exposure Quantification - Groundwater Pathway Analysis

## 5.5.3.1 *Current and Potential Future Uses of Groundwater*

Groundwater located in the DBG Plume within the boundary of BAAP is not used for human consumption. The land that was transferred from the Army to other property owners includes a deed restriction on the use of groundwater and so restricts the potential exposure to groundwater within the boundary of BAAP. These groundwater access restrictions state that the property owner "shall not access or use groundwater underlying the property for any purpose without the prior written approval of the Army and the WDNR".

Currently, residential wells located outside of BAAP use groundwater for potable water and domestic purposes. The potential future use of groundwater adjacent to and downgradient of BAAP is expected to be for potable water and domestic purposes.

Residential well users can be exposed to contaminated groundwater through ingestion or drinking of water, inhalation of vapor during showering or dishwashing, and dermal contact while bathing. Groundwater contaminants from BAAP have resulted in groundwater impacts in one residential well located in the DBG Plume. The Army replaced one residential well associated with the DBG Plume that has been impacted by total DNT. Section 4.6 provides additional details on the residential well replacement conducted by the Army.

# 5.5.4 Risk Assessment Summary

## 5.5.4.1 Hypothetical Future On-Site Groundwater Risks

The cumulative cancer risk, non-cancer HI and contaminants of concern related to hypothetical future on-site risks associated with the DBG Plume are summarized below. Contaminants of concern are analytes found to significantly contribute to the cumulative risk in an area where risk was estimated to be above the risk management criteria (cumulative cancer risk >  $1x10^{-4}$  or HI > 1).

#### Summary of Hypothetical Future Risks – Deterrent Burning Ground Plume (On-Site Monitoring Well Data)

Location	Cumulative Cancer Risk	Non-cancer Hazard Index (HI)	Contaminants of Concern
On-Site (Hypothetical Future Risk)	9x10 <sup>-5</sup>	3	1,1,2-Trichloroethane

Risks calculated using the simple scaling method for a hypothetical future residential scenario, along with the maximum observed concentration of each COPC, yielded cumulative cancer risk estimates below the risk management criterion  $(1x10^{-4})$  for the on-site portion of the DBG Plume.

The calculated non-cancer HI of 3 was above the risk management criterion (HI > 1) in the onsite portion of the DBG Plume. The contaminant of concern that contributed to the HI > 1 in the DBG Plume was 1,1,2-trichloroethane.

Based on the maximum risk scenario, the on-site portion of the DBG Plume represents an area that, if future residential development occurred, would be associated with cumulative non-cancer risk above the risk management criterion (HI above 1). The on-site portion of the DBG Plume has the potential to migrate off-site, thus impacting downgradient residential wells. Further evaluation of the risks is provided in Section 8.0 Remedial Alternative Development Process.

# 5.5.4.2 *Current Off-Site Groundwater Risks*

The cumulative cancer risk, non-cancer HI and contaminants of concern related to current offsite groundwater risks associated with the DBG Plume are summarized below. Contaminants of concern are analytes found to significantly contribute to the cumulative risk in an area where risk was estimated to be above the risk management criteria (cumulative cancer risk >  $1x10^{-6}$  or HI > 1).

#### Summary of Current Risk – Deterrent Burning Ground Plume (Residential Well and Off-Site Monitoring Well Data)

Location	Cumulative Cancer Risk	Non-cancer Hazard Index (HI)	Contaminants of Concern
Off-Site (Current Risk)	2x10 <sup>-5</sup>	2	Chloroform Total DNT Trichloroethene

Risks calculated using the simple scaling method for a current residential scenario, along with the maximum observed concentration of COPC, yielded cumulative cancer risk estimates above the risk management criterion for the off-site portion of the DBG Plume. The cumulative cancer risk  $(2x10^{-5})$  for the DBG Plume area was above the risk management criterion  $(1x10^{-6})$ . The contaminants of concern that contributed to the cumulative cancer risk for the DBG Plume were chloroform, total DNT, and trichloroethene.

The calculated non-cancer HI of 2 was above the risk management criterion (HI > 1) in the offsite portion of the DBG Plume. The contaminant of concern that contributed to the HI > 1 in the DBG Plume was trichloroethene.

Based on the maximum risk scenario, the off-site portion of the DBG Plume represents an area that would be associated with cumulative groundwater risks above the risk management criteria (cumulative cancer risk above  $1 \times 10^{-6}$  and non-cancer HI above 1). Further evaluation of the risks is provided in Section 8.0 Remedial Alternative Development Process.

# 5.6 Central Plume

# 5.6.1 Characterization of Exposure Settings

The source of the Central Plume is in the north-central portion of BAAP where nitroglycerin, rocket paste, and rocket propellant were produced, see Figure 1. Within the production area, containers of production chemicals, which contained DNT, were transported by rail to each Pre-Mix House from the Bag Loading House. Nitrocellulose and nitroglycerin were added to the chemical mixture in each Pre-Mix House. The resulting slurry was then pumped to the Final Mix Houses. The Rocket Paste production area was not connected to the main industrial sewer

network, so production related wash waters were discharged to open ditches. It is believed that the broad production area may have caused the DNT impacted groundwater. The primary source of contaminated soil originated in former production areas. The Army has performed numerous soil excavations in ditches and ponds, sewer pipe removals, and building demolition throughout the Central Plume source area. The Army has not performed any groundwater remediation in the Central Plume.

The contaminated groundwater in the Central Plume has migrated south and off-site towards the Wisconsin River (Gruber's Grove Bay), see Figure 1. Based on current groundwater monitoring data, there is no evidence to suggest that the Central Plume is discharging into the Wisconsin River. As the Central Plume migrates south, it vertically sinks into the sand aquifer. The thickness of Central Plume narrows as it moves off-site and towards the Wisconsin River. Groundwater in the Central Plume travels approximately 143 feet per year. Contaminants in the Central Plume source area is approximately 105 feet deep. Contaminated groundwater (above the NR 140 ES) in the Central Plume has only been identified within the sand aquifer at depths between 85 and 130 feet. Groundwater beneath the off-site residential areas is approximately 20 feet deep. Section 3.3 provides additional details on source investigation and remediation. Sections 4.4 and 4.5 provide additional details on groundwater properties and groundwater contaminants.

The seven residential wells located within the areal extent of the Central Plume range in depth from 80 to 324 feet. Two of those seven residential wells are screened in the bedrock and draw their groundwater from beneath the contaminated portion of the Central Plume. Three of those seven residential wells are screened in the sand but draw their groundwater from beneath the contaminated portion of those seven residential wells are screened in the sand but draw their groundwater from beneath the contaminated portion of the Central Plume. Two of those seven residential wells are screened in the sand and at the same depth as the contaminated portion of the Central Plume. DNT has been detected in these two residential wells below the NR 140 ES.

The residential wells located outside the areal extent of the Central Plume range in depth from 80 to 575 feet. Most of the residential wells located outside of the Central Plume are screened in the sand and average 120 feet deep. DNT has been detected in monitoring wells located both on-site and off-site in the Central Plume. The Army has replaced three residential wells, screened in the sand, located in the southern extent of the Central Plume. Section 4.6 provides additional details on the residential well replacements conducted by the Army.

A graphical depiction of the Central Plume in relationship to the local geology, monitoring wells, residential wells, site features, and groundwater plume boundaries is shown below. The groundwater flow direction is from the left (north) towards the right (south). The groundwater contaminant plume is shown below the water table and migrating towards Gruber's Grove Bay. The groundwater contaminant plume is shown to have traveled past the BAAP property and beneath a residential area.



A groundwater CSM for the Central Plume is provided in Appendix H. The CSM shows the relationship between the sources of contamination, how the contamination is transported, type of media exposure, the route of exposure, and who may be exposed. The contaminants infiltrated through the soil (leaching) beneath production areas (i.e., buildings, ditches, ponds or sewers) until they reached the groundwater. The contaminants within the groundwater have been transported with the directional flow of groundwater towards the Wisconsin River (Gruber's Grove Bay) to the south. Contaminated groundwater has the potential to reach residential wells which may be used for domestic or potable purposes. Residential well users can be exposed to contaminated groundwater through ingestion or drinking of water, inhalation of vapor during bathing or dishwashing, and dermal contact while bathing.

The exposure routes associated with domestic use of water, as shown on the CSM (Appendix H), include ingestion, inhalation, and dermal exposure. The off-site residential use pathways are potentially complete under current land use conditions and warrant further evaluation. The on-site hypothetical future residential use pathways are incomplete or considered insignificant under current land use conditions but potentially complete under hypothetical future on-site groundwater usage.

The exposure route associated with vapor intrusion, as shown on the CSM, includes only inhalation of indoor vapors. Both the off-site and on-site residential use vapor intrusion exposure pathways are incomplete or considered insignificant based on past vapor intrusion investigations and so no further evaluation is warranted. Section 5.6.2 discusses the potential vapor intrusion exposure into buildings.

#### 5.6.2 Exposure Quantification – Vapor Intrusion Pathway Analysis

An evaluation was conducted to determine whether vapors from the Central Plume of groundwater contamination pose a current or hypothetical future risk to human health. Vapor intrusion occurs when there is a migration of vapor-forming chemicals from a subsurface source (i.e., contaminated groundwater) into an overlying building. The exposure route evaluated was the inhalation of contaminants from indoor air.

The Army did not conduct a vapor intrusion pathway analysis investigation specifically in the Central Plume. Section 5.2.3 discussed the 2012 vapor intrusion pathway analysis investigations conducted by the Army in the PBG Plume. The PBG Plume represents the worst-case scenario for volatile-forming chemicals present in the groundwater and thus provides a conservative representation of vapor conditions associated with the Central Plume. The 2012 vapor intrusion pathway analysis reports concluded that VOCs in the PBG Plume do not present a risk to human health via vapor intrusion.

Based on the above information in Section 5.2.3, inhalation exposure due to soil gas vapor intrusion from the Central Plume does not pose a current or potential future risk to area residents.

#### 5.6.3 Exposure Quantification - Groundwater Pathway Analysis

#### 5.6.3.1 Current and Potential Future Uses of Groundwater

Groundwater located in the Central Plume found within the boundary of BAAP is not used for human consumption. The land that was transferred from the Army to other property owners includes a deed restriction on the use of groundwater and so restricts the potential exposure to groundwater within the boundary of BAAP. These groundwater access restrictions state that the property owner "shall not access or use groundwater underlying the property for any purpose without the prior written approval of the Army and the WDNR".

Currently, residential wells located outside of BAAP use groundwater for potable water and domestic purposes. The potential future use of groundwater adjacent to and downgradient of BAAP is expected to be for potable water and domestic purposes.

Residential well users can be exposed to contaminated groundwater through ingestion or drinking of water, inhalation of vapor during showering or dishwashing, and dermal contact while bathing. Groundwater contaminants from BAAP have resulted in groundwater impacts in three residential wells located in the Central Plume. The Army has replaced three residential

wells due to impacts from DNT. Section 4.6 provides additional details on the residential well replacements conducted by the Army.

#### 5.6.4 Risk Assessment Summary

#### 5.6.4.1 Hypothetical Future On-Site Groundwater Risks

The cumulative cancer risk, non-cancer HI and contaminants of concern related to hypothetical future on-site risks associated with the Central Plume are summarized below. Contaminants of concern are analytes found to significantly contribute to the cumulative risk in an area where risk was estimated to be above the risk management criteria (cumulative cancer risk >  $1x10^{-4}$  or HI > 1).

#### Summary of Hypothetical Future Risks – Central Plume (Monitoring Well Data)

Location	Cumulative CancerNon-cancerRiskHazardIndex (HI)		Contaminants of Concern	
On-Site (Hypothetical Future Risk)	3x10 <sup>-6</sup>	0.02	None	

Risks calculated using the simple scaling method for a hypothetical future residential scenario, along with the maximum observed concentration of each COPC, yielded cumulative cancer risk estimates below the risk management criterion  $(1x10^{-4})$  for the on-site portion of the Central Plume.

The non-cancer HI risk calculations were below the risk management criterion (HI  $\leq$  1) in the on-site portion of the Central Plume.

Based on the maximum risk scenario, the on-site portion of the Central Plume represents an area where cumulative risk estimates are below the risk management criteria, and so no contaminants of concern were identified. Further evaluation of the risks is provided in Section 8.0 Remedial Alternative Development Process.

## 5.6.4.2 *Current Off-Site Groundwater Risks*

The cumulative cancer risk, non-cancer HI and contaminants of concern related to current offsite groundwater risks associated with the Central Plume are summarized below. Contaminants of concern are analytes found to significantly contribute to the cumulative risk in an area where risk was estimated to be above the risk management criteria (cumulative cancer risk >  $1x10^{-6}$  or HI > 1).

#### Summary of Current Risk – Central Plume (Residential Well and Off-Site Monitoring Well Data)

Location	Cumulative Cancer Risk	Non-cancer Hazard Index (HI)	Contaminants of Concern
Off-Site (Current Risk)	4x10 <sup>-5</sup>	0.4	1,2-Dichloroethane 2,6-DNT Benzene Chloroform

Risks calculated using the simple scaling method for a current residential scenario, along with the maximum observed concentration of COPC, yielded cumulative cancer risk estimates above the risk management criterion for the off-site portion of the Central Plume. The cumulative cancer risk  $(4x10^{-5})$  for the Central Plume was above the risk management criterion  $(1x10^{-6})$ . The contaminants of concern that contributed to the cumulative cancer risk for the Central Plume were 1,2-dichloroethane, 2,6-DNT, benzene, and chloroform.

The non-cancer HI risk calculations were below the risk management criterion (HI  $\leq$  1) in the off-site portion of the Central Plume.

Based on the maximum risk scenario, the off-site portion of the Central Plume represents an area that would be associated with a cumulative cancer risk above the risk management criterion (above  $1 \times 10^{-6}$ ). Further evaluation of the risks is provided in Section 8.0 Remedial Alternative Development Process.

## 5.7 Nitrocellulose Production Area Plume

# 5.7.1 Characterization of Exposure Settings

The source of the NC Area Plume is in the northwestern portion of BAAP where nitrocellulose was produced (see Figure 1). Completed nitrocellulose was used to manufacture single-base propellants such as smokeless powder or double-base propellants such as rocket grains or Ball Powder. DNT was added to the manufacturing process in various production buildings. The broad production area contained numerous production buildings and process water disposal sewer piping that caused the DNT impacted groundwater. The sources of the DNT contamination have been removed. The Army has performed numerous soil excavations, sewer pipe removals, and building demolition throughout the NC Area Plume.

As the NC Area Plume migrates south, it remains near the groundwater surface and doesn't sink vertically. Groundwater in the NC Area travels approximately 132 feet per year. Contaminants in the NC Area Plume are expected to travel at the same speed as groundwater. Groundwater beneath the NC Area Plume is approximately 100 feet deep. Contaminated groundwater (above the NR 140 ES) in the NC Area Plume has only been identified within the sand aquifer at depths between 100 and 120 feet. The contaminated groundwater in the NC Area Plume has migrated south but remains on-site, see Figure 1. The Army has not performed any groundwater

remediation in the NC Area Plume. Based on the direction of groundwater flow, the NC Area Plume is migrating towards the PBG Plume. In the future, the NC Area Plume could comingle with the PBG Plume while on BAAP property. There are no residential wells located within 2 miles downgradient (south) of the NC Area Plume. Sections 4.4 and 4.5 provide additional details on groundwater properties and groundwater contaminants.

A graphical depiction of the NC Area Plume in relationship to the local geology, monitoring wells, site features, and groundwater plume boundaries is shown below. The groundwater flow direction is from the upper left (north) towards the lower right (south). The groundwater contaminant plume is shown below the water table. The groundwater contaminant plume is contained on the BAAP property.



A groundwater CSM for the NC Area Plume is provided in Appendix H. The CSM shows the relationship between the sources of contamination, how the contamination is transported, type of media exposure, the route of exposure, and who may be exposed. The contaminants infiltrated through the soil (leaching) beneath production areas (i.e., buildings or sewers) until they reached the groundwater. The contaminants within the groundwater have been transported with the directional flow of groundwater but have remained on-site.

The exposure routes associated with domestic use of water, as shown on the CSM (Appendix H), include ingestion, inhalation, and dermal exposure. The off-site residential use pathways are considered incomplete since the NC Area Plume has not migrated off-site. The on-site hypothetical future residential use pathways are incomplete or considered insignificant under current land use conditions but potentially complete under hypothetical future on-site groundwater usage.

The exposure route associated with vapor intrusion, as shown on the CSM, includes only inhalation of indoor vapors. Both the off-site and on-site residential use vapor intrusion exposure pathways are incomplete or considered insignificant based on past vapor intrusion investigations and so no further evaluation is warranted. Section 5.7.2 discusses the potential vapor intrusion exposure into buildings.

# 5.7.2 Exposure Quantification - Vapor Intrusion Pathway Analysis

Contaminated groundwater from the NC Area Plume has not migrated off-site. There are no onsite buildings located over the NC Area Plume. Based on these factors, there is no vapor intrusion exposure pathway from groundwater associated with the NC Area Plume and there is no current or potential future risk to area residents.

## 5.7.3 Exposure Quantification - Groundwater Pathway Analysis

## 5.7.3.1 Current and Potential Future Uses of Groundwater

Groundwater located in the NC Area Plume is only found within the boundary of BAAP and is not used for human consumption. The land that was transferred from the Army to other property owners includes a deed restriction on the use of groundwater and so restricts the potential exposure to groundwater within the boundary of BAAP. These groundwater access restrictions state that the property owner "shall not access or use groundwater underlying the property for any purpose without the prior written approval of the Army and the WDNR". It should be noted that there are no residential wells located within 2 miles downgradient (south) of the NC Area Plume. In addition, there are no off-site monitoring wells associated with the NC Area Plume.

### 5.7.4 Risk Assessment Summary

#### 5.7.4.1 Hypothetical Future On-Site Groundwater Risks

The cumulative cancer risk, non-cancer HI and contaminants of concern related to hypothetical future on-site groundwater risks associated with the NC Area Plume are summarized below. Contaminants of concern are analytes found to significantly contribute to the cumulative risk in an area where risk was estimated to be above the risk management criteria (cumulative cancer risk >  $1 \times 10^{-4}$  or HI > 1).

Location	Cumulative Cancer Risk	Non-cancer Hazard Index (HI)	Contaminants of Concern	
On-Site (Hypothetical Future Risk)	4x10 <sup>-6</sup>	0.04	None	

#### Summary of Hypothetical Future Well Risks – NC Area Plume (On-Site Monitoring Well Data)

Risks calculated using the simple scaling method for a hypothetical future residential scenario, along with the maximum observed concentration of each COPC, yielded cumulative cancer risk estimates below the risk management criterion  $(1x10^{-4})$  for the NC Area Plume.

The non-cancer HI risk calculations were below the risk management criterion (HI  $\leq$  1) in the NC Area Plume.

Based on the maximum risk scenario, the NC Area Plume represents an area where cumulative risk estimates are below the risk management criteria, and so no contaminants of concern were identified.

## 5.7.4.2 *Current Off-Site Groundwater Risks*

There are no off-site monitoring wells associated with the NC Area Plume. In addition, there are no residential wells located within 2 miles downgradient (south) of the NC Area Plume; therefore, current groundwater risks were not evaluated.

# 6.0 **REGULATORY REQUIREMENTS**

CERCLA requires that on-site remedial actions attain or waive federal environmental applicable and relevant and appropriate requirements (ARARs), or more stringent state environmental ARARs, upon completion of the remedial action. The USEPAs 1994 National Oil and Hazardous Substances Pollution Contingency Plan (NCP) also requires compliance with ARARs during removal and remedial actions to the extent practicable. These ARARs, in conjunction with the overall protection to human health and the environment criterion, help form the criteria to evaluate remedial alternatives. Under CERCLA, remedial actions must be protective of human health and the environment. Additionally, CERCLA remedial actions must meet a level and standard of control that attains standards, requirements, limitations, or criteria that are "applicable or relevant and appropriate" under the circumstances of the release. Information that is "to be considered" (TBC) federal and state criteria, advisories, and guidance may also be considered/evaluated along with ARARs as a part of a risk assessment conducted at a CERCLA site to help set clean-up level targets.

Applicable requirements are those cleanup standards, standards of control, and other substantive environmental protection requirements, criteria, or limitations promulgated under federal environmental or state environmental or facility siting laws that specifically address a hazardous substance, pollutant, contaminant, remedial action, location or other circumstance at a CERCLA site. In other words, an applicable requirement is one with which a private party would have to comply by law if the same action was being undertaken apart from CERCLA authority.

If a requirement is not applicable, it still may be relevant and appropriate. Relevant and appropriate requirements are those cleanup standards, standards of control, and other substantive requirements, criteria or limitations promulgated under federal environmental or state environmental or facility siting laws that, while not "applicable" to a hazardous substance, pollutant, contaminant, remedial action, location or other circumstance at a CERCLA site, address problems or situations sufficiently similar to those encountered at the CERCLA site that their use is well suited to the particular site. Only those state standards that are identified in a timely manner and are more stringent than federal requirements may be relevant and appropriate.

"Applicability" is a legal and jurisdictional determination, while the determination of "relevant and appropriate" relies on professional judgment, considering environmental and technical factors at the site.

USEPA identifies three basic types of ARARs:

- Chemical-specific ARARs are generally health- or risk-based values which, when applied to site-specific conditions, result in numerical values. These values establish the acceptable concentration of a chemical that may be found in, or discharged to, the ambient environment.
- Location-specific ARARs are restrictions placed upon removal activities of hazardous substances solely because they are occurring in a particular place.

• Action-specific ARARs are general technology or activity-based requirements on actions taken with respect to hazardous substances. These requirements are triggered by the particular activities that are selected to accomplish a remedy. Thus, action-specific requirements do not in themselves determine the removal alternative; rather, they indicate how the selected alternative must be achieved.

TBCs are non-promulgated advisories or guidance issued by federal or state governments that are not legally binding and do not have the status of potential ARARs; however, TBCs may be considered along with ARARs as part of the site risk assessment and may be used in determining the necessary level of clean-up for protection of health and the environment.

Potential State and Federal ARARs and TBCs to be used in the groundwater remedial alternatives evaluation are presented in Table 14.

# Table 14Potential State and Federal Applicable or Relevant and Appropriate Requirements<br/>Remedial Investigation/Feasibility Study<br/>Badger Army Ammunition Plant

Standard, Requirement, Criteria, or Limitation	Citation	Description	Applicable or Relevant and Appropriate Requirement (ARAR)					
Chemical-Specific ARARs and TBC	Chemical-Specific ARARs and TBCs							
Groundwater Quality	Wisconsin Administrative Code, Chapter NR 140.26	Chemical-specific groundwater Enforcement Standard (ES).	<u>ARAR</u> – Establishes applicable groundwater quality standards.					
National Primary Drinking Water Regulations: Maximum Contaminant Levels (MCLs) and Maximum Residual Disinfectant Levels	40 CFR Part 141 Subpart G	Chemical-specific drinking water quality standards.	<u>ARAR</u> – Establishes relevant and appropriate groundwater quality standards.					
National Secondary Drinking Water Regulations	40 CFR Part 143	Chemical-specific drinking water quality standards related to aesthetics.	<u>TBC</u> – Recommended drinking water quality guidelines.					
Regional Screening Level (RSL) Resident Tapwater Table	USEPA – November 2017	Screening level guidance for human health risk from exposure to groundwater.	<u>TBC</u> – Recommended groundwater quality screening levels.					
Location-Specific ARARs								
No Location-Specific ARARs were ide	ntified.							
Action-Specific ARARs	Action-Specific ARARs							
Well Construction and Pump Installation	Wisconsin Administrative Code, Chapter NR 812 excluding subsections 812.05, 25, 38, 39, 40, 42, 43, 44 and 45.	Establishes requirements for installing water supply wells and extracting groundwater.	<u>ARAR</u> – Applicable to alternatives that would replace a contaminated residential well or active remediation activities that pump groundwater.					

Note: Table 16 lists the groundwater cleanup level & regulatory concentration for each contaminant of concern.

# 7.0 CONTAMINANTS OF CONCERN

This section presents the contaminants of concern (COC) for the groundwater contamination associated with four groundwater contamination plumes at BAAP. The COCs are based on the results of the HHRA detailed in Section 5.0. Table 15 summarizes the groundwater COCs for the BAAP. Table 15 provides a breakdown of which risk-based COCs were identified as having a cancer risk and/or non-cancer risk. Table 15 also shows which risk-based COC was identified as an on-site or off-site risk above the risk management criteria.

Table 16 provides the groundwater cleanup levels for each risk-based COC related to the PBG Plume, DBG Plume, Central Plume, and NC Area Plume. The groundwater cleanup level for each risk-based COC is based on the lower of either the MCLs (National Primary Drinking Water Regulations per 40 CFR Part 141) or the Wisconsin NR 140 ES.

#### 7.1 **Propellant Burning Ground Plume**

The risk-based COCs identified in the PBG Plume were chloroform, CTET, ethyl ether, TCE, and 2,6-DNT.

- Chloroform was identified as having an off-site cancer risk above the risk management criteria. Based on the groundwater monitoring data from 2015 to 2018, chloroform concentrations were not identified above the groundwater cleanup level listed in Table 16. Therefore, remedial alternatives are not being considered for chloroform.
- CTET was identified as having an off-site cancer risk above the risk management criteria. Based on the groundwater monitoring data from 2015 to 2018, CTET concentrations were identified above the groundwater cleanup level listed in Table 16.
- Ethyl ether was identified as having an on-site non-cancer risk above the risk management criteria. Based on the groundwater monitoring data from 2015 to 2018, ethyl ether concentrations were identified above the groundwater cleanup level listed in Table 16.
- TCE was identified as having both an off-site cancer risk and an on-site and off-site noncancer risk above the risk management criteria. Based on the groundwater monitoring data from 2015 to 2018, TCE concentrations were identified above the groundwater cleanup level listed in Table 16.
- 2,6-DNT was identified as having an on-site and off-site cancer risk plus an on-site noncancer risk above the risk management criteria. Based on the groundwater monitoring data from 2015 to 2018, 2,6-DNT concentrations were identified above the groundwater cleanup level listed in Table 16.

Based on the above information, CTET, ethyl ether, TCE, and 2,6-DNT will be the COCs considered in the FS for the development of remedial alternatives in the PBG Plume.

# Table 15Groundwater Contaminants of ConcernRemedial Investigation/Feasibility StudyBadger Army Ammunition Plant

	Contaminant of Concern (COC) - HHRA						
Groundwater Plume	Cancer Risk <sup>(1)</sup>			Non-Cancer Risk <sup>(2)</sup>			
	сос	On-Site	Off-Site	coc	On-Site	Off-Site	
	Carbon Tetrachloride		Х				
	Chloroform		X				
Propellant Burning Ground				Ethyl Ether	X		
	Trichloroethene		X	Trichloroethene	Х	X	
	2,6-Dinitrotoluene	Х	Х	2,6-Dinitrotoluene	X		
	Chloroform		Х				
Deterrent Burning Cround				1,1,2-Trichloroethane	X		
Deterrent burning Ground	Trichloroethene (3)		X	Trichloroethene (3)		Х	
	Dinitrotoluene, Total *		Х				
	Benzene <sup>(4)</sup>		Х				
Control	Chloroform		Х	none			
Central	1,2-Dichloroethane		Х				
	2,6-Dinitrotoluene		X				
Nitrocellulose Production Area	none			none			

Notes:

Based on analytical lab results from residential and groundwater monitoring well samples for 2015, 2016, 2017, and 2018.

HHRA - Human Health Risk Assessment

<sup>(1)</sup> Contaminants found to contribute to a cumulative human cancer risk above the risk management criteria.

<sup>(2)</sup> Contaminants found to contribute to a cumulative human non-cancer risk above the risk management criteria.

<sup>(3)</sup> Trichloroethene (TCE) is not considered a COC in the Detererent Burning Ground Plume.

The source of TCE is not attributable to the Army and has been found in residential well jet pumps.

<sup>(4)</sup> Benzene is not considered a COC in the Central Plume. The source of benzene is not attributable to the Army.

\* Total Dinitrotoluene (DNT) Isomers (2,3-DNT; 2,4-DNT; 2,5-DNT; 2,6-DNT; 3,4-DNT; 3,5-DNT) - NR 140.10

Table 16 lists the groundwater cleanup level & regulatory concentration for each contaminant of concern.

# Table 16Groundwater Cleanup LevelsRemedial Investigation/Feasibility StudyBadger Army Ammunition Plant

	Contaminants of Concern	State	Federal	Groundwater
Groundwater Plume	(COC)	NR 140 ES	40 CFR Part 141 MCLs	Cleanup Level <sup>(1)</sup>
	Carbon Tetrachloride	5	5	5
	Chloroform	6	80 (2)	6
Propellant Burning Ground	2,6-Dinitrotoluene	0.05	none	0.05
	Ethyl Ether	1000	none	1000
	Trichloroethene	5	5	5
	Chloroform	6	80 (2)	6
	Total Dinitrotoluene	0.05	none	0.05
Deterrent Burning Ground	1,1,2-Trichlorethane	5	5	5
	Trichloroethene (3)	5	5	5
	Benzene <sup>(4)</sup>	5	5	5
Control	Chloroform	6	80 (2)	6
Central	1,2-Dichloroethane	5	5	5
	2,6-Dinitrotoluene	0.05	none	0.05
Nitrocellulose Production Area	none	none	none	none

Notes:

<sup>(1)</sup> Cleanup Level is the lowest value of either the NR 140 ES or Federal MCL.

<sup>(2)</sup> The Chloroform MCL is for Total Trihalomethanes (sum of bromodichloromethane, bromoform, dibromochloromethane, & chloroform)

<sup>(3)</sup> Trichloroethene is not considered a COC in the Detererent Burning Ground Plume. The source of trichloroethene is not attributable to the Army. Trichloroethene has been found in residential well jet pumps.

<sup>(4)</sup> Benzene is not considered a COC in the Central Plume. The source of benzene is not attributable to the Army.

All concentration values are expresed in micrograms-per-liter ( $\mu g/l$ )

ES = Enforcement Standard

40 CFR Part 141 - National Primary Drinking Water Regulations: Maximum Contaminant Levels (MCLs)

Total Dinitrotoluene (DNT) consists of isomers (2,3-DNT; 2,4-DNT; 2,5-DNT; 2,6-DNT; 3,4-DNT; 3,5-DNT)

### 7.2 Deterrent Burning Ground Plume

The risk-based COCs identified in the DBG Plume were chloroform, 1,1,2-TCA, TCE, and total DNT.

- Chloroform was identified as having an off-site cancer risk above the risk management criteria. Based on the groundwater monitoring data from 2015 to 2018, chloroform concentrations were not identified above the groundwater cleanup level listed in Table 16. Therefore, remedial alternatives are not being considered for chloroform.
- 1,1,2-TCA was identified as having an on-site non-cancer risk above the risk management criteria. Based on the groundwater monitoring data from 2015 to 2018, 1,1,2-TCA concentrations were not identified above the groundwater cleanup level listed in Table 16. Therefore, remedial alternatives are not being considered for 1,1,2-TCA.
- TCE was identified as having both an off-site cancer and non-cancer risk above the risk management criteria. TCE has not been detected in monitoring wells nor is there a known source associated with the DBG Plume. The source of TCE is not attributable to the Army and has been found in residential well jet pumps. Therefore, remedial alternatives are not being considered for TCE.
- Total DNT was identified as having an off-site cancer risk above the risk management criteria. Based on the groundwater monitoring data from 2015 to 2018, total DNT concentrations were identified above the groundwater cleanup level listed in Table 16.

Based on the above information, total DNT will be the only COC considered in the FS for the development of remedial alternatives in the DBG Plume.

## 7.3 Central Plume

The risk-based COCs identified in the Central Plume were benzene, chloroform, 1,2-dichloroethane, and 2,6-DNT.

- Benzene was identified as having an off-site cancer risk above the risk management criteria. Benzene has not been detected in monitoring wells located on-site (upgradient) nor is there a known source associated with the Central Plume. The source of benzene is not attributable to the Army. Benzene was also not detected in any monitoring wells or residential wells that were sampled during 2018. Therefore, remedial alternatives are not being considered for benzene.
- Chloroform was identified as having an off-site cancer risk above the risk management criteria. Based on the groundwater monitoring data from 2015 to 2018, chloroform concentrations were not identified above the groundwater cleanup level listed in Table 16. Therefore, remedial alternatives are not being considered for chloroform.
- 1,2-Dichloroethane was identified as having an off-site cancer risk above the risk management criteria. Based on the groundwater monitoring data from 2015 to 2018, 1,2-dichloroethane concentrations were not identified above the groundwater cleanup level

listed in Table 16. Therefore, remedial alternatives are not being considered for 1,2dichloroethane.

• 2,6-DNT was identified as having an off-site cancer risk above the risk management criteria. Based on the groundwater monitoring data from 2015 to 2018, 2,6-DNT concentrations were identified above the groundwater cleanup level listed in Table 16.

Based on the above information, 2,6-DNT will be the only COC considered in the FS for the development of remedial alternatives in the Central Plume.

## 7.4 Nitrocellulose Production Area Plume

The HHRA did not identify any human health risk related COCs for the NC Area Plume; therefore, no remedial alternatives will be developed for the NC Area Plume in the FS.

# 8.0 REMEDIAL ALTERNATIVE DEVELOPMENT PROCESS

As described in Section 5.0, a HHRA was completed as it relates to current and hypothetical future risks for the groundwater contaminant plumes as appropriate. Based on groundwater monitoring results for 2015, 2016, 2017 and 2018, the HHRA found unacceptable risk related to groundwater from BAAP and identified a completed exposure pathway for the PBG, DBG, and Central Plumes. Based on these factors (cleanup level exceedances, risk identified above the risk management criteria and completed exposure pathway identification), the Army is evaluating the feasibility of groundwater remedial actions to reduce, control, or mitigate exposure to be protective of human health and the environment for the PBG, DBG, and Central Plumes. Section 7.0 identifies the COCs for each plume and the groundwater cleanup levels for each COC.

The HHRA did not identify risk above the risk management criteria for the NC Area Plume. Therefore, groundwater remedial alternatives are not being considered by the Army for the NC Area Plume and groundwater sampling of the monitoring wells is not part of the CERCLA remedy for the NC Area Plume.

For ease of review, clarity and appropriateness, the remedial alternative development process was completed for each individual plume. As each contaminant plume has a specific set of circumstances including but not limited to size, location, geology, hydrogeology and contaminants of concern, plume-specific alternatives were developed. This process allows plume-specific alternatives to be tailored to the circumstances associated with each individual plume.

## 8.1 Previous Soil Remedial Activities

Soil remedial activities have been conducted at the source areas of the four groundwater contaminant plumes, PBG Plume, DBG Plume, Central Plume, and NC Area Plume. These soil remedial activities are summarized in Sections 3.1, 3.2, 3.3, and 3.4. Source areas with contaminated soil have been addressed at BAAP either with removal, in-situ treatment, vapor extraction, soil covers, or engineered barriers. These remedial activities have minimized the potential exposure to contaminated soil at BAAP. The Army has received site closure from the WDNR on all soil related investigations and remedial actions at BAAP.

## 8.2 Previous Groundwater Remedial Activities

Groundwater remedial activities were first conducted at the PBG Plume starting in 1990 with the construction and operation of the IRM. The IRM ultimately consisted of two source control wells, three boundary control wells, a treatment process building and a discharge pipeline to the Wisconsin River. These wells extracted and treated approximately 310 gpm of groundwater until the IRM's operational termination in 2012. The IRM was augmented by the construction of the MIRM in 1996. This system ultimately consisted of five extraction wells, a treatment process building and discharge pipeline to the Wisconsin River. These wells extracted and treated approximately 2,400 gpm of groundwater until the MIRM's operational termination in 2015. Biochemical treatment of groundwater at the PBG Waste Pits began in 2001 and was operational until 2005. These groundwater remedial activities are summarized in Section 3.1.

Groundwater remedial activities have not been conducted at the Central Plume, DBG Plume, or NC Area Plume.

## 8.3 Groundwater Remedial Action Objectives

Groundwater remedial action objectives (RAOs) provide a general description of what the cleanup will accomplish, serve as the basis for evaluating each remedial alternative, and provide an understanding of how the unacceptable risks will be addressed by each remedial alternative. Groundwater RAOs require the remedy to protect human health by preventing exposure to contaminated groundwater, to restore groundwater to the extent practicable, and minimize the impact of the contaminant plumes on the environment. Specifically, the RAOs for any individual plume are achieved when the risk-based groundwater COCs listed in Table 15 are below the groundwater cleanup levels provided in Table 16. The groundwater cleanup levels shown in Table 16 are based on either the NR 140 ES or federal MCL.

## 8.4 General Response Actions

The General Response Actions (GRAs) are general actions that would satisfy the RAOs. The potential applicability of GRAs and associated technologies were evaluated based on site specific constraints. The applicable GRAs and a brief description for the BAAP groundwater are listed below.

- Land Use Controls Administrative actions such as land use restrictions to protect public health and the environment.
- **Development of New Water Resources** Provision of bottled water well replacement and alternate water supply systems.
- **Groundwater Treatment** Removal, treatment and disposal of contaminated groundwater.
- Groundwater Containment Isolation of groundwater using subsurface barriers.

# 8.5 Identification and Screening of Potentially Applicable Technologies

This section identifies the appropriate plume specific remedial technologies and process options for each GRA for groundwater at BAAP. Process options refer to a specific process within each technology type. For example, the vertical barrier technology category could include process options such as a slurry wall, sheet pile wall or deep soil mixing. For each GRA, several broad technology types may be identified and within each remedial technology, several process options may be applicable.

During this screening step, process options and entire technology types are eliminated from further consideration based on technical implementability. This is completed by using readily available information from site conditions, contaminant types and concentrations and sitespecific circumstances. Based on this evaluation, some remedial technologies and process options were eliminated from further consideration. The technology screening process and subsequent process option evaluation for each plume meeting the qualifying criteria (cleanup level exceedances, risk identified above the risk management criteria and completed exposure pathway identification) is shown in Table 17.

## 8.6 Process Option Screening Criteria

This section contains the description of process options screening criteria for each technology which provides the basis for developing remedial alternatives. For technologies with more than one process option, each option was evaluated. Each process option is evaluated according to the following criteria:

- **Effectiveness** which includes evaluating the following:
  - Potential effectiveness in handling the estimated area or volumes of media.
  - Potential in meeting the RAOs.
  - Potential impacts to human health and the environment during the construction and implementation phase.
  - Demonstrated reliability of the process with respect to the contaminants and site conditions.
- **Implementability** which includes technical and administrative feasibility of implementing a process option:
  - Technologies passing the initial screen of applicability are screened based on technical feasibility. This criterion means feasibility under site specific conditions. This evaluation may indicate that although a technology may be generally applicable for the COCs, the specific technology may be limited due to site-specific conditions.
  - Institutional feasibility emphasizing the institutional aspects of implementability such as the ability to obtain necessary permits for off-site actions.
- **Cost** Plays a limited role in the screening process and is used only when two alternatives are found to be equally protective. Cost analyses are based on engineering judgement and evaluated as to whether costs are high, moderate or low in relation to other process options.

Following the selection of the most appropriate process options for each technology type, the process options are combined to form remedial alternatives. Remedial alternatives are discussed in Sections 9.0 for the PBG Plume, 10.0 for the DBG Plume, and 11.0 for the Central Plume.

## 8.7 Evaluation and Selection of Representative Process Options

This section evaluates the process options using the criteria listed in Section 8.6: effectiveness, implementability and cost. Only the most applicable process options, as identified in Table 17, were carried forward and are included in the development of remedial alternatives.

# Table 17 Technology Screening Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

General Response Action	Remedial Technology	Process Option	Description/Comments	Retained for Further Consideration
Land Use Controls	Access Restrictions	Deed Restrictions - On-site	Deed would restrict on-site water use only.	Yes
		Provision of Bottled Water	Provide bottled water to residential well owners with impacts above the Enforcement Standards.	Yes
<b>N N N</b>	Altomoto Woton	Impacted Well Replacement	Replacement of residential wells impacted above the Enforcement Standards.	Yes
Resources	Supply	Well Replacement - Plume Areas	Replacement of residential wells within plume boundaries.	Yes
		Municipal Water Supply - New System	Construct new municipal water system for residential well owners south and east of BAAP. Army does not have authority.	No
	Course broaten	Extraction Wells	Series of wells to extract contaminated groundwater.	Yes
	Removal	Subsurface Drains	Perforated pipe in trenches backfilled with porous media to collect contaminated groundwater. Not feasible due to depth of necessary trenches.	No
		Biochemical Injection	Injection of treatment agent into groundwater.	Yes
	In-Situ Treatment	Permeable Reactive Barrier	Reactive barrier allows contaminated groundwater to pass through with passive treatment. Not feasible due to depth of contamination.	No
		Monitored Natural Attenuation	Allowing natural processes (dilution, dispersion and sorption) to slowly degradation contamination.	Yes
Groundwater Treatment	Ex-Situ - On-site Treatment	Mobile Treatment Facility	Utilize mobile treatment units to treat contaminated water.	Yes
		On-site Treatment Facility	Construct on-site facility to treat contaminated water. Army no longer owns property.	No
		Bluffview Sanitary District (BSD)	Utilize BSD wastewater treatment plant. Not feasible due to flow limitations.	No
	Ex-Situ - Off-site	Publicly Owned Treatment Works (POTW)	Haul extracted groundwater to POTW. Not feasible due to anticipated volume.	No
		On-site Discharge - Injection	Treated water discharged to deep well injection system. Not feasible due to anticipated volume.	No
	Groundwater Disposal	On-site Discharge - Infiltration Gallery	Treated water discharged to an on-site infiltration gallery. Not feasible due to anticipate volume.	No
		Pipeline to Wisconsin River	Treated water discharged into the Wisconsin River.	Yes
		Slurry Wall	Trench around impacted area is filled with a soil/cement/bentonite mix. Not feasible due to depth of contamination.	No
Groundwater Containment	Vertical Barriers	Vertical Barriers         Sheet Pile Wall         Sheet pile wall around impacted areas. Not feasible of depth of contamination.           Deep Soil Mixing         Mixing of bentonite in soil through augers. Not feasil due to depth of contamination.		No
				No

# 8.7.1 Land Use Controls

Access Restrictions - On-site Deed Restrictions - Groundwater access restrictions for the BAAP property are already in place and restricts property owners from accessing groundwater as part of the property transfer agreement. Specifically, the Groundwater Restrictions state, "The Grantee, its successors and assigns, shall not access or use groundwater underlying the Property for any purpose without the prior written approval of the Army and the WDNR. For the purpose of this restriction, "groundwater" shall have the same meaning as Section 101(12) of CERCLA."

- **Effectiveness** Access restrictions are effective in controlling human activities such as potable well construction on the BAAP property.
- **Implementability** These deed restrictions are currently implemented as a result of parcel transfer agreements.
- Cost Low

Land Use Controls are carried forward as a process option which can be combined with other process options to meet the RAO.

# 8.7.2 Development of New Water Resources

Alternate Water Supply - Provision of Bottled Water and Well Replacement - For areas impacted by groundwater contamination off the BAAP property, the Army currently has an environmental monitoring and health protection program in place that is protective of the residential water well users. If a Chapter NR 140 ES is exceeded in a residential well once, bottled water is made available to the occupant. If the exceedance occurs a second, consecutive time, well replacement is offered to the owner. Bottled water would be made available to the occupant until the well is replaced, operational and water quality verified (typically 3 months and based on driller availability). If the NR 140 ES exceedance is not detected for two consecutive rounds after the first NR 140 ES exceedance detection, bottled water would be discontinued. To date, the Army has replaced seven shallow residential wells with deeper aquifer residential wells.

- **Effectiveness** The alternate water supply has been effective in conjunction with groundwater monitoring to replace residential wells.
- Implementability These options can be readily implemented.
- $\mathbf{Cost} \mathbf{Low}$

Alternate Water Supply – Provision of Bottled Water and Well Replacement is carried forward as a process option which can be combined with other process options to meet the RAO.

Alternate Water Supply – Well Replacement within the Plume Areas - This process option would involve replacing individual residential shallow wells with a deeper aquifer well for existing residents. If sampling results indicate an increasing trend for a plume's COC in three consecutive rounds and that the plume is migrating toward a residential well, the Army will evaluate if well replacement is necessary. This process option would provide a safe, clean and reliable water source for potentially affected residential well owners downgradient of BAAP. The Army currently monitors 54 residential wells.

- **Effectiveness** This process option would eliminate receptors from potential exposure to groundwater contamination within the plume areas by proactively providing a deeper aquifer well. This process option would rely on natural processes such as dilution, dispersion and sorption to degrade the contaminant plume over time.
- **Implementability** This process option could be implemented by replacing individual shallow wells, meeting the criteria, with an individual deeper aquifer well.
- **Cost** Low to Moderate depending upon replacement frequency.

Alternate Water Supply – Well Replacement within the Plume Area is carried forward as a process option which can be combined with other process options to meet the RAO.

Alternate Water Supply – Municipal Water System - This process option would involve construction of a new municipal water system servicing residents located east and south of the BAAP with the potential of being impacted by the contaminant plumes. In 2011, the Army submitted a Revised Alternative Feasibility Study, Groundwater Remedial Strategy report to the WDNR. The selected groundwater remedy was Monitored Natural Attenuation (MNA). Due to the relatively long remedial timeframe for the MNA remedy to achieve the proposed cleanup levels, the proposed remedy included construction and operation of a municipal drinking water system that would provide residents in the communities surrounding the BAAP with drinking water while groundwater contamination continued to diminish over time. During an evaluation by the Army's Office of General Counsel it was determined the Army did not have the legal or funding authority to procure and operate a municipal water system as identified in the 2011 Revised Alternative Feasibility Study, so this option was not carried forward in the Feasibility Study.

While a draft Decision Document (DD) for Site-Wide Groundwater was being prepared in 2012, the Army identified several areas where the draft DD did not meet both legal and policy requirements. Specifically, a human health risk assessment was not prepared, incorrect legal standards were identified for the selected groundwater remedy and key components of the proposed response action were outside the Army's authority. For these reasons, this process option was not carried forward.

## 8.7.3 Groundwater Treatment

**Groundwater Removal - Extraction Wells** - Vertical extraction wells are installed to collect and extract contaminated groundwater to reduce concentrations and/or contain a contaminant plume.

- **Effectiveness** This process option is commonly used as an effective groundwater removal technology. Proper well location is necessary for effective source reduction and plume control. This process has been used at BAAP and based on previous experience, additional study and design may be needed to maximize source reduction and plume control.
- **Implementability** This process option has been used at BAAP and is commonly used in the industry to remove groundwater. This process option would require utilities to be

extended to the site of the extraction well network. Groundwater extraction wells are relatively easy to install. This process option would also require coordination from existing property owners on- and off-site as the land in which the extraction wells would be located is owned and/or managed by other entities.

• **Cost** – Moderate cost due to site infrastructure improvements necessary for site specific conditions.

Groundwater Removal – Extraction Wells is carried forward as a process option which can be combined with other process options to meet the RAO.

**Groundwater Removal – Subsurface Drains** - This process option utilizes horizontal interceptor trenches filled with porous media to convey impacted water to extraction points. This application is typically used in shallow applications. Based on the depth of the contaminant plumes, this process option was not carried forward.

**Groundwater Treatment - In-situ Biochemical Injection** - Vertical injection points are installed within the contaminant plume, in areas where COCs exceed groundwater cleanup levels, and injected with a biochemical selected for the ability to degrade specific chemicals into harmless by-products through anaerobic biodegradation.

- **Effectiveness** The procedures and applications of biochemical injection are applicable to numerous anaerobically biodegradable contaminants including but not limited to chlorinated solvents, energetics, and nitrates.
- **Implementability** Equipment and expertise would be readily available; however, a field-scale pilot test would be necessary. This process option would also require coordination from existing property owners on- and off-site as the land in which the injection points would be located is owned and/or managed by other entities.
- **Cost** Moderate to high cost depending upon the amount and corresponding cost of biochemical necessary to treat the plume.

Groundwater Treatment - In-situ Biochemical Injection is carried forward as a process option which can be combined with other process options to meet the RAO.

**Groundwater Treatment - Permeable Reactive Barrier** - This process option utilizes reactive media constructed across the path of a contaminant plume to treat groundwater. A permeable reactive barrier is generally limited to shallow applications and its effectiveness is a concern based on the longevity of the reactive media. Due to the depth of the contaminant plumes and concerns about the lifespan of the reactive media, this process option was not carried forward.

**Groundwater Treatment - Monitored Natural Attenuation** - MNA is a passive remedial process that utilizes groundwater sampling results to monitor the reduction in groundwater contaminants. Natural processes such as dilution, dispersion and sorption would be monitored over time to confirm contaminant reduction.

These natural attenuation processes include a variety of physical, chemical, or biological processes that act without human intervention to reduce the mass, toxicity, mobility, volume, or concentration of contaminants in groundwater. These in-situ processes include biodegradation,

dispersion, dilution, sorption, volatilization, and chemical or biological stabilization, transformation, or destruction of contaminants. Natural attenuation processes may reduce the potential risk posed by site contaminants in three ways: (1) transformation of contaminant(s) to a less toxic form through destructive processes such as biodegradation or abiotic transformations; (2) reduction of contaminant concentrations whereby potential exposure levels may be reduced; and (3) reduction of contaminant mobility and bioavailability through sorption onto the soil or rock matrix.

Under CERCLA, MNA is considered to be a remedy like any other remedy. According to the USEPA (Office of Solid Waste and Emergency Response (OSWER) Directive 9200.4-17P), MNA can be an alternative means of achieving the RAO that may be appropriate for specific site circumstances where its use meets the applicable statutory and regulatory requirements. MNA can be used in conjunction with other remedies as a follow-up measure that will be monitored and compared with expectations. The USEPA expects that MNA will be most appropriate when used in conjunction with other remedial methods (e.g., source control, groundwater extraction), or as a follow-up to active remedial methods that have already been implemented. Both the USEPA and WDNR recognize MNA may be an appropriate remedial method for contaminated groundwater under certain circumstances.

- **Effectiveness** This process option is an effective long-term solution as groundwater concentrations are expected to decrease as the chemicals would continue to undergo a slow degradation process.
- **Implementability** This process option is easily implemented as monitoring well and residential well sampling and analytical testing are currently being conducted in accordance with the most recent regulatory approval.
- **Cost** Low

Groundwater Treatment - Monitored Natural Attenuation is carried forward as a process option which can be combined with other process options to meet the RAO.

**Groundwater Treatment – Mobile Treatment** - Pressurized, mobile, tractor-trailer mounted treatment tanks utilizing activated carbon to treat extracted groundwater water in areas where COCs exceed groundwater cleanup levels.

- **Effectiveness** Activated carbon has been previously used at BAAP to successfully treat DNT.
- **Implementability** These units are capable of supporting treatment at flow rates up to 500 gpm. A separate mobile treatment unit would be required to support each well. This process option would require utilities to be extended to the site of the mobile treatment facility. This process option would also require coordination from existing property owners on- and off-site as the land on which the treatment units would be located is owned and/or managed by other entities. These mobile treatment units could be used in cold weather months with appropriate heating and insulation provisions.
- **Cost** Moderate

Groundwater Treatment – Mobile Treatment is carried forward as a process option which can be combined with other process options to meet the RAO.

**Groundwater Treatment - On-site Treatment Facility** - A treatment facility for the extracted groundwater in areas where COCs exceed groundwater cleanup levels would be located on the BAAP property. It is anticipated that the treatment system would require a structure (treatment facility) equipped with supporting utilities including gas, electric, water, sewer and communication. Utilities would need to be extended to the site in addition to other site improvements. At the treatment facility, activated carbon is the treatment media expected to be utilized to treat the impacted groundwater. This process option has been utilized at the BAAP previously. The Army no longer owns the land in or around the contaminant plumes, for which they would require, for facility construction. For this reason, this process option was not carried forward.

**Groundwater Treatment – Bluffview Sanitary District** - This process option would involve pumping extracted groundwater to the Bluffview Sanitary District Wastewater Treatment Plant. The maximum daily capacity of this facility is 45,000 gallons per day which will not accommodate the flow rates anticipated for a pump and treat system. Each extraction well is expected to pump 720,000 gallons per day. For this reason, this process option was not carried forward.

**Groundwater Treatment at Publicly Owned Treatment Works** - This process option would involve pumping extracted groundwater in areas where COCs exceed groundwater cleanup levels to a holding tank and utilizing tanker trucks to transport the extracted groundwater to a publicly owned treatment works. Based on the anticipated flow rates needed for source removal and plume control (720,000 gallons per day per extraction well), the number of tanker trucks necessary to transport the impacted water would be not practicable. For this reason, this process option was not carried forward.

**Groundwater Disposal - On-site Discharge** - This process option would discharge treated groundwater on-site to either groundwater injection points or to an infiltration gallery. The areas near the contaminant plumes are on property owned and/or managed by other entities. Based on the anticipated flow rates needed for source removal and plume control, the size of the area necessary to facilitate injection or infiltration would be not practicable. Base on the anticipated discharge rates (720,000 gallons per day per extraction well) and subsequent size of the injection or infiltration area necessary for disposal, on-site discharge was not carried forward as a process option or remediation technology.

**Off-site Discharge Pipeline to the Wisconsin River** - This process option would discharge treated groundwater into the Wisconsin River. This would require pumping and a piping network to convey treated groundwater to the surface water discharge point.

• **Effectiveness** – This process option is an effective method for discharge water disposal provided that permit requirements could be met. This process option has been previously utilized at the BAAP.

- **Implementability** This process option can be implemented as equipment and materials required for construction are readily available. This process option would require additional studies to design the discharge system to meet site specific requirements and constraints. This process option would also require coordination from existing property owners as the land in which the discharge piping would be located is owned and/or managed by other entities.
- **Cost** Moderate cost depending upon discharge system design.

Off-site Discharge Disposal to the Wisconsin River is carried forward as a process option which can be combined with other process options to meet the RAO.

#### 8.7.4 Groundwater Containment

**Vertical Barriers** - Vertical barriers including slurry and sheet pile walls and deep soil mixing would be installed around the contaminant plumes to provide horizontal containment. These walls are typically "keyed" into a relatively impervious formation, providing horizontal and vertical containment. However, there are some of these walls that are constructed to "hang" when the contaminant plume is at shallow elevations effectively stagnating the plume. Based on site geology and depth of the contaminant plumes, vertical barriers were not carried forward as a process option or remediation technology.

#### 8.7.5 Summary of Process Options for Groundwater

The following process options remain after screening:

- Land Use Controls including on-site groundwater access restrictions
- Development of New Water Resources including provision of bottled water and residential well replacement within the plume areas
- Groundwater Treatment including removal through extraction wells, treatment through biochemical injection, monitored natural attenuation, and mobile treatment units and discharge through pipeline to the Wisconsin River

#### 8.8 Alternatives Analysis Process

The NCP (40 CFR 300.430) states that the primary objective of the FS is to "ensure that appropriate remedial alternatives are developed and evaluated," and that "the number and type of alternatives to be analyzed shall be determined at each site, considering the scope characteristics and complexity of the site problem that is being addressed."

Nine evaluation criteria have been developed to serve as the basis for conducting a detailed analysis of the remedial alternatives. The evaluation criteria with the associated statutory considerations are:

- 1. Overall Protection of Human Health and the Environment
- 2. Compliance with the ARARs
- 3. Long-Term Effectiveness and Permanence
- 4. Reduction in Toxicity, Mobility and Volume through Treatment

- 5. Short-Term Effectiveness
- 6. Implementability
- 7. Cost
- 8. State Acceptance
- 9. Community Acceptance

A process to evaluate remedial alternatives has been developed based on statutory requirements. The nine criteria are categorized into three groups and include threshold criteria, primary balancing criteria and modifying criteria.

Evaluation against two criteria relate directly to statutory findings that must ultimately be made in the remedy. Therefore, these are categorized as threshold criteria in that each alternative must meet them. These two criteria are briefly described below:

- **Overall Protection of Human Health and the Environment** The assessment against this criterion describes how the alternative achieves and maintains protection of human health and the environment.
- **Compliance with ARARs** The assessment against this criterion describes how the alternative complies with ARARs. The assessment also addresses other information from advisories, criteria and guidance.

The five criteria listed below represent the primary balancing criteria upon which the analysis is based.

- Long-Term Effectiveness and Permanence The assessment of alternatives against this criterion evaluates the long-term effectiveness of alternatives in maintaining protection of human health and the environment after response objectives have been met.
- **Reduction of Toxicity Mobility and Volume through Treatment** The assessment against this criterion evaluates the anticipated performance of the specific treatment technologies an alternate may employ.
- **Short-Term Effectiveness** The assessment against this criterion examines the effectiveness of alternatives in protecting human health and the environment during the construction and implementation of a remedy until response objectives have been met.
- **Implementability** This assessment evaluates the technical and administrative feasibility of alternatives and the availability of require goods and services.
- **Cost** This assessment evaluates the capital and operation and maintenance cost of each alternative.

The final two modifying criteria are briefly described below.

• State Acceptance – This assessment reflects that State's apparent preferences among or concerns about the remedy. State acceptance of an alternative will be evaluated in the Proposed Plan issued for public comment. Therefore, this criterion is not considered in this FS.

• **Community Acceptance** – This assessment reflects the community's apparent preferences among or concerns about alternatives. Community acceptance of each alternative will be evaluated after a Proposed Plan is issued for public comment. Therefore, this criterion is not considered in this FS.

The sections below present the detailed analysis of alternatives based on criteria 1 through 7 from the NCP (40 CFR 300.4309(e)(9)), as listed above.

# 9.0 REMEDIAL ALTERNATIVES – PBG PLUME

As identified in Section 7.1, CTET, ethyl ether, TCE, and 2,6-DNT were the only risk-related COCs considered for the development of remedial alternatives in the PBG Plume. The RAO for the PBG Plume requires the remedy to protect human health by preventing exposure to contaminated groundwater, to minimize the impact of the contaminants on the environment, and to restore groundwater to the extent practicable. The RAO for the PBG Plume will be achieved when groundwater concentrations of CTET, ethyl ether, TCE, and 2,6-DNT are below the groundwater cleanup level listed in Table 16.

Based on site conditions and the screening of process options, six remedial alternatives were developed to address the presence of CTET, ethyl ether, TCE, and 2,6-DNT in the PBG Plume. Monitored Natural Attenuation (MNA) is expected to reduce the concentrations of the following VOCs by natural processes: CTET, chloroform, ethyl ether, and TCE. Active remedial alternatives were developed specifically for elevated concentrations of 2,6-DNT for the PBG Plume. Alternative 1 - No Action, provides a baseline to evaluate the other alternatives.

#### Alternative 1: No Action

The No Action Alternative is a mandatory evaluation that provides a baseline to evaluate the other alternatives. This alternative would have no impact on the contaminant plume and would not require groundwater monitoring of residential wells or monitoring wells. This alternative would include on-site groundwater access restrictions.

#### Alternative 2: Monitored Natural Attenuation and Alternate Water Supply

The Monitored Natural Attenuation and Alternate Water Supply Alternative would continue the current remedial action approach and include the following components:

- Continued groundwater monitoring of residential and monitoring wells
- On-site groundwater access restrictions
- Provision for an alternate water supply condition including bottled water and well replacement

#### Alternative 3: Active Groundwater Remediation – Pump and Treat

The Active Groundwater Remediation – Pump and Treat Alternative would target removing and treating impacted groundwater with elevated 2,6-DNT concentrations and include the following components:

- Continued groundwater monitoring of residential and monitoring wells
- On-site groundwater access restrictions
- Provision for an alternate water supply condition including bottled water and well replacement
- Groundwater removal through the installation of four groundwater extraction wells
- Groundwater treatment through the use of four mobile treatment units
- Groundwater disposal through the construction of piping leading to the Wisconsin River
### Alternative 4: Active Groundwater Remediation – Anaerobic Bioremediation

The Active Groundwater Remediation – Anaerobic Bioremediation Alternative would target treating impacted groundwater with elevated 2,6-DNT concentrations and include the following components:

- Continued groundwater monitoring of residential and monitoring wells
- On-site groundwater access restrictions
- Provision for an alternate water supply condition including bottled water and well replacement
- Groundwater treatment through in-situ biochemical injection at nine permanent injection well locations directly downgradient of the source area
- Groundwater treatment through in-situ biochemical injection at 150 temporary locations (on-site and off-site)

# Alternative 5: Well Replacement – Plume Area

The Well Replacement – Plume Area Alternative would involve replacing shallow aquifer wells (meeting qualifying criteria) within the PBG Plume area with deeper aquifer wells and include the following components:

- Continued groundwater monitoring of residential and monitoring wells
- On-site groundwater access restrictions
- Replacement of as many as 47 existing residential wells

### Alternative 6: Source Area Treatment

The Source Area Treatment Alternative would target treating impacted groundwater with elevated 2,6-DNT concentrations directly downgradient of the source area and include the following components:

- Continued groundwater monitoring of residential and monitoring wells
- On-site groundwater access restrictions
- Provision for an alternate water supply condition including bottled water and well replacement
- Groundwater treatment through in-situ biochemical injection at nine permanent injection well locations directly downgradient of the source area

# 9.1 Alternative 1 – No Action

The No Action Alternative would have no impact on any of the contaminant plumes and would not require groundwater monitoring of residential wells or monitoring wells. There would be no contaminant removal, treatment, containment or monitoring related to this alternative. As a condition of the Army's property transfer, groundwater access restrictions would continue for areas within the BAAP boundary.

### **Overall Protection of Human Health and the Environment**

Groundwater access is restricted within the BAAP boundary based on conditions of property transfer documentation. The groundwater access restrictions would require Army and WDNR

authorization prior to well installation within the BAAP boundary; however, there are no groundwater access restrictions outside the BAAP boundary. This alternative would not provide any protection of human health or the environment beyond the groundwater access restrictions within the BAAP boundary. This alternative would result in the Army terminating the residential and monitoring well sampling program.

### **Compliance with ARARs**

The residential and monitoring well sampling program is being conducted in accordance with the most recent regulatory approval. This alternative would result in the Army terminating the residential and monitoring well sampling program. This alternative would not comply with ARARs.

### Long-Term Effectiveness and Permanence

This alternative would not provide an effective or permanent long-term solution. In this alternative, groundwater concentrations are expected to decrease as the chemicals would continue to undergo a slow degradation process (dilution, dispersion, and sorption). This alternative would result in the Army terminating the residential and monitoring well sampling program. Consequently, the degradation process would not be evaluated under this alternative.

### Reduction of Toxicity, Mobility, and Volume through Treatment

Limited reductions in toxicity, mobility, and volume would occur through natural degradation processes only. This alternative would discontinue the residential and monitoring well sampling program. Consequently, the degradation process would not be evaluated.

### **Short-Term Effectiveness**

There would be no action taken for this alternative. Since groundwater monitoring would be discontinued, any groundwater exceedances would go unidentified. Therefore, this alternative has no short-term effects.

### Implementability

This alternative is inherently implementable as no remedial action would be taken.

#### Cost

There is no cost associated with the No Action Alternative.

### 9.2 Alternative 2 – Monitored Natural Attenuation and Alternate Water Supply

The Monitored Natural Attenuation and Alternate Water Supply Alternative would include MNA for the PBG Plume, on-site groundwater access restrictions and a provision for an alternate water supply condition for residential wells. This alternative would also continue residential and

monitoring well sampling of the PBG Plume as previously specified in Section 4.2 and Appendix D.

MNA relies on natural attenuation processes to achieve the RAO within a time frame that is reasonable compared to that offered by other more active remedial methods. MNA is expected to reduce the concentrations of the COCs that were carried forward in the development of remedial alternatives (see Section 7.1), which include CTET, ethyl ether, TCE, and 2,6-DNT. These natural attenuation processes include a variety of physical, chemical, or biological processes that act without human intervention to reduce the mass, toxicity, mobility, volume, or concentration of contaminants in groundwater. These in-situ processes include biodegradation, dispersion, dilution, sorption, volatilization, and chemical or biological stabilization, transformation, or destruction of contaminants.

The Army currently has an environmental monitoring and health protection program in place that is protective of the residential water well users, "alternate water supply". If a Chapter NR 140 ES is exceeded in a residential well once, bottled water is made available to the occupant. If the exceedance occurs a second, consecutive time, well replacement is offered to the owner. Bottled water would be made available to the occupant until the well is replaced, operational and water quality verified (typically 3 months and based on driller availability). If the NR 140 ES exceedance detection, bottled water would be discontinued. To date, the Army has replaced three shallow residential wells within the PBG Plume.

### **Overall Protection of Human Health and the Environment**

This alternative would provide protection of human health and the environment due to groundwater access restrictions within the BAAP boundary and the provision of an alternate water supply condition for residential wells. The groundwater sampling program would monitor the groundwater concentrations for compliance and contaminant reduction.

### **Compliance with ARARs**

The residential and monitoring well sampling program is being conducted in accordance with the most recent regulatory approval. This alternative would continue the residential and groundwater monitoring program and comply with ARARs over time though natural degradation processes only.

### Long-Term Effectiveness and Permanence

This alternative offers a long-term solution as groundwater concentrations are expected to decrease as the chemicals would continue to undergo a slow degradation process (dilution, dispersion, and sorption). This alternative would continue to restrict groundwater access within the BAAP and the provision of an alternate water supply condition would address concerns associated with residential well impacts. Groundwater impacts are expected to remain and the groundwater monitoring program is expected to continue for at least 30 years.

### Reduction of Toxicity, Mobility, and Volume through Treatment

Limited reductions in toxicity, mobility, and volume is expected to occur through natural degradation processes only. This reduction would be verified through the monitoring program.

### **Short-Term Effectiveness**

This alternative offers a short-term solution as it is currently being applied and no additional work associated with implementation would be required. This alternative would continue to restrict groundwater access within the BAAP and the provision of an alternate water supply condition would address concerns associated with residential wells. If the alternate water supply provision is necessary, state licensed well drillers would be utilized for well replacement. The well drillers would be appropriately trained and would maintain applicable certifications to install any replacement well necessary.

#### Implementability

This alternative would be easily implementable as this action is currently being applied to the site. No remedial activities other than sampling under the MNA program would be performed. Groundwater access restrictions are already in place within BAAP.

#### Cost

The estimated total cumulative costs for Alternative 2 are shown below. See Appendix I for a summary of the costs for Alternative 2.

### Alternative 2 – Monitored Natural Attenuation and Alternate Water Supply

Direct Capital Cost:	\$ 0
Indirect Capital Cost:	\$ 0
30 Years of Annual O&M:	\$ 4,913,113
Total Present Worth:	\$ 4,913,113

\* Total costs use current rates and do not include inflation

### 9.3 Alternative 3 – Active Groundwater Remediation – Pump and Treat

The Active Groundwater Remediation – Pump and Treat Alternative would include groundwater extraction and treatment with mobile treatment units and continued groundwater monitoring of residential and monitoring wells. This alternative would also include on-site groundwater access restrictions and a provision for an alternate water supply condition.

As identified in Section 9.0, active remedial alternatives are only being developed for 2,6-DNT concentrations above the groundwater cleanup level listed in Table 16. Consequently, the extraction wells would be strategically located to target elevated 2,6-DNT concentrations. This technology is expected to also reduce the concentrations of chlorinated solvents that coexist within the targeted treatment areas for 2,6-DNT.

It is anticipated that four extraction wells and four mobile treatment units (one treatment unit per extraction well) would be necessary for source area reduction and plume migration control. Four extraction wells were selected based on previous performance (capture zone) of extraction wells in this area. Generally, three of the extraction wells would be located on-site, south of the source area. The northern-most well would be located directly downgradient of the source area for source area reduction. The other two on-site extraction wells would be located in the southern portion of the on-site plume for migration control. One additional well would be located off-site, just south of the BAAP southern boundary, for migration control. Proposed pumping well locations and target pumping capture zones are shown on Drawing PBG-ALT 3 in Appendix J. The remainder of the PBG Plume located by Highway 78 would be allowed to degrade through natural processes, as no at-risk residential wells have been identified in this area.

Each extraction well is expected to pump at approximately 500 gpm. Similarly, each mobile treatment unit would be designed to treat 500 gpm. Based on previous experience with pump and treat systems in this area, groundwater flow velocities of 306 ft/yr (see Table 8) and assuming no additional source area contribution, the individual extraction wells and mobile treatment units are expected to operate continuously for various durations. The two extraction wells located in the southern on-site portion of the plume are expected to operate for at least 8 years. The extraction well located off-site is expected to operate for at least 6 years. The extraction well located closest to the source area is expected to operate for at least 2 years. The mobile treatment units are expected to use activated carbon as the primary treatment media as activated carbon has successfully treated DNT at BAAP. Site improvements including mobile treatment trailer staging area construction, electrical utility provision and site security would be necessary at each one of the extraction well/mobile treatment trailer areas.

A network of piping and appurtenances would be necessary to route extracted water from the extraction wells to the mobile treatment units and treated water from the mobile treatment units to a discharge location. Treated groundwater would ultimately discharge to the Wisconsin River. It is anticipated that the pump and treat system would require the services of an environmental technician to monitor and maintain the extraction wells and mobile treatment units.

# **Overall Protection of Human Health and the Environment**

This alternative would be designed to control and limit the migration of and treat the groundwater with elevated 2,6-DNT concentrations. The provision of the alternate water supply condition would address concerns associated with residential well impacts.

# **Compliance with ARARs**

This alternative would be designed to comply with ARARs. The provision of the alternate water supply condition would address concerns associated with residential well impacts.

### Long-Term Effectiveness and Permanence

This alternative would be designed to reduce contaminant concentrations to comply with regulatory standards in groundwater through recovery and treatment of the portion of the PBG Plume with elevated 2,6-DNT concentrations. This alternative would continue to restrict groundwater access within the BAAP and the provision of an alternate water supply condition for residential wells. The previous pump and treat effort (MIRM) at the PBG showed effective DNT concentration.

Based on previous experience, the groundwater pump and treat system's individual extraction wells and mobile treatment units are expected to operate continuously for various durations for up to eight years. The groundwater monitoring program is expected to continue for at least 30 years.

### Reduction of Toxicity, Mobility, and Volume through Treatment

This alternative is expected to result in reductions in toxicity, mobility, and volume through treatment of the PBG Plume with elevated 2,6-DNT concentrations. It is assumed that there would be no additional contribution of 2,6-DNT from the source areas into the groundwater. The previous pump and treat effort (MIRM) at the PBG showed effective 2,6-DNT concentration reduction. The groundwater contamination would also continue to decrease due to natural attenuation processes.

### **Short-Term Effectiveness**

For this alternative, there would be some short-term effects to workers, residents and the environment during implementation. As described above, this alternative would require three on-site and one off-site extraction wells coupled with a mobile treatment unit for each extraction well. These locations would require construction of a staging area for the well and mobile treatment unit, security and electricity for the site for operations and lighting.

It is anticipated that from each extraction well to the mobile treatment unit and from the mobile treatment unit to a discharge location, a discharge pipe would be constructed. Treated water is expected to be discharged to the Wisconsin River.

There is some risk associated with the operation of heavy equipment for site preparation, well drilling, excavation, piping installation and backfilling; however, proper training and equipment would be required to mitigate these risks. Utility crossing, near public road working conditions and work on private land would also be items that would need planning, coordination, and health and safety training.

To maximize contaminant reduction and plume migration control, it is anticipated that additional investigation, sampling and testing would need to be completed. This effort is expected to take approximately one year. Construction and implementation of this alternative including well installation, piping construction, treatment area preparation and utility extension is expected to be completed in approximately one year.

### Implementability

Equipment and materials required for construction of this alternative are readily available. However, extraction well and mobile treatment unit locations would have to be coordinated carefully and with input from existing land owners as they are responsible for the ownership and/or management of the area around the PBG Plume. In addition, utilities to support the extraction wells and mobile treatment facility would need to be extended to the site, since none currently exist. The discharge line location would need to be determined and appropriate piping and appurtenance construction competed. The previous pump and treat discharge location to the Wisconsin River was identified during winter months with a high-visibility buoy system. This identified open water as a safety precaution to those who utilize the Wisconsin River in the winter for recreational activities such as ice fishing and snowmobiling. It is expected that a similar buoy system would be installed during the winter months and subsequently removed in the spring. This process of installation and decommissioning the buoy system would need to be repeated each winter and spring, respectively, as long as the system continued operation.

### Cost

The estimated total cumulative costs for Alternative 3 are shown below. See Appendix I for a summary of the costs for Alternative 3.

#### Alternative 3 – Active Groundwater Remediation – Pump and Treat

Direct Capital Cost:	\$ 3,633,573
Indirect Capital Cost:	\$ 1,635,108
30 Years of Annual O&M:	\$ 7,433,131
Total Cost:	\$ 12,701,812

\* Total costs use current rates and do not include inflation

### 9.4 Alternative 4 – Active Groundwater Remediation – Anaerobic Bioremediation

The Active Groundwater Remediation – Anaerobic Bioremediation Alternative would include in-situ anaerobic biodegradation targeting elevated concentrations of 2,6-DNT in groundwater and continued groundwater monitoring of residential and monitoring wells. This alternative would also include on-site groundwater access restrictions and a provision for an alternate water supply condition.

As identified in Section 9.0, active remedial alternatives are only being developed for 2,6-DNT concentrations above the groundwater cleanup level listed in Table 16. Consequently, the in-situ biochemical injection locations would be strategically located to target elevated 2,6-DNT concentrations. This technology is expected to also reduce the concentrations of chlorinated solvents that coexist within the targeted treatment areas for 2,6-DNT.

For this alternative, a nutrient-enriched emulsified vegetable oil (EVO) is being proposed as the injection product. EVO has been used to stimulate in-situ anaerobic biodegradation of groundwater contaminants at commercial, industrial, and military sites. The procedures and

applications of EVO are applicable to numerous anaerobically biodegradable contaminants including but not limited to chlorinated solvents, energetics, and nitrates.

The primary objective of injecting EVO into the groundwater is to stimulate the anaerobic biodegradation of the target contaminants. Groundwater aquifers are complex ecosystems populated by a broad and diverse array of microbial communities. The composition and activity of these microbial communities' changes continuously as their environment changes. Alterations in aquifer geochemistry and the availability of substrates and nutrients that can be used to generate energy and support growth and reproduction significantly affect microbial activity.

EVO would be distributed in the aquifer as an oil-in-water emulsion (mixture). In this approach, an oil-in-water emulsion would be first prepared using a food-grade oil, food-grade surfactants, and water. The emulsion would have small uniform droplets to allow transport in the aquifer. The emulsion would be injected into the aquifer (through injection wells or DPT) with additional water to distribute the oil droplets. The oil droplets would be distributed through the aquifer pore spaces and adhere to soil particles. The soil particle surfaces would gradually become coated with a thin layer of oil droplets that provide a carbon source for long-term anaerobic biodegradation. The oil droplets remain in the aquifer as a viable carbon source for approximately two years. Soluble substrates and nutrients (e.g., lactate, yeast extract, vitamins) can be added to the mixture prior to injection to stimulate rapid growth of desired bacteria. When the contaminated groundwater naturally flows toward and through the distributed EVO, the groundwater contaminants interact with the carbon source and break-down into less harmful byproducts.

It is anticipated that 159 injection points (both on-site and off-site and at varying stratigraphic depths) would be required to treat the plume. These injection points would be arranged in a series of eight treatment lines and consist of both permanent injection wells (nine) and temporary injection points (150). The nine permanent wells would be arranged in one treatment line located just downgradient of the source area. It is assumed that the source area would no longer contribute to the groundwater contamination. However, should this occur, the permanent wells could be utilized for additional injections. The other seven treatment lines consisting of temporary injection points would be located both on-site and off-site within the plume. Anticipated treatment line locations are shown on Drawing PBG-ALT 4 in Appendix J.

The spacing of the treatment lines is based on a groundwater flow velocity of 306 ft/yr (see Table 8) and the viability of the carbon source remaining in the aquifer for approximately two years. The distance between each treatment line is based on two years of treatment. Based on the geology and hydrogeology associated with the plume, a 25-foot radius of influence is anticipated to provide sufficient distribution of the EVO within the aquifer. The radius of influence is measured from the injection well or point location radially, out to the maximum extent of EVO product distribution. Each treatment line would be designed to fully capture contaminated groundwater migrating downgradient.

Though EVO is a proven technology to effectively treat chlorinated solvents and energetics, a field-scale pilot test would be necessary to determine the site specific constraints and a design to

be developed to target 2,6-DNT within the plume at BAAP. Upon successful completion of a field-scale pilot test, the remedial design could be finalized.

### **Overall Protection of Human Health and the Environment**

This alternative would be designed to meet the requirements of the RAO as it would effectively degrade the contaminants in the PBG Plume. The provision of the alternate water supply condition would address concerns associated with residential well impacts. Groundwater access restrictions would continue for areas within the BAAP.

### **Compliance with ARARs**

Concentrations of contaminants in the treated area are expected to comply with ARARs relatively quickly (approximately two years). The provision of the alternate water supply condition would address concerns associated with residential well impacts.

### Long-Term Effectiveness and Permanence

This alternative would be designed to reduce the concentration in groundwater to comply with regulatory standards for 2,6-DNT. This alternative would continue to restrict groundwater access within the BAAP and the provision of an alternate water supply condition for residential wells. It is anticipated that a single round of injections of the biochemical product would be sufficient to treat the plume. Based on a groundwater flow velocity and the viability of the carbon source, treatment is expected to take approximately two years. However, depending upon groundwater monitoring results, it is possible that this technology may require supplemental post-treatment applications. The proposed biochemical product for use with this technology has shown successful contaminant reduction with explosives and chlorinated solvents; however, it has not been applied at full scale for 2,6-DNT treatment. Lastly, potential increases in groundwater table elevation may have the ability to mobilize residual contamination remaining in the vadose zone. The groundwater monitoring program is expected to continue for at least 30 years.

### Reduction of Toxicity, Mobility, and Volume through Treatment

This alternative is expected to reduce the toxicity, mobility, and volume of 2,6-DNT and chlorinated solvents in the treated areas more quickly than natural processes alone. It is assumed that there would be no additional contribution of 2,6-DNT from the source areas into the groundwater. The groundwater contamination would also continue to decrease due to natural attenuation processes.

### **Short-Term Effectiveness**

There would be some short-term effects to workers, residents and the environment during implementation. As described above, this alternative would require both on-site and off-site injection points.

There is some risk associated with heavy equipment necessary for permanent injection well installation, temporary injection point installation and injection. Proper training and equipment would be required to mitigate these risks. The bioremediation is expected to occur over the course of two years and no additional worker safety issues have been identified. Near public road working conditions and work on private land would also be items that would need planning, coordination, and health and safety training.

To maximize contaminant reduction, it is anticipated that additional investigation, sampling and testing would need to be completed. This effort is expected to take approximately two years. Construction and implementation of this alternative including permanent well installation, temporary injection point installation, biochemical injection and injection point abandonment is expected to be completed in approximately one year.

# Implementability

The installation of permanent injection wells and temporary injection points may be challenging at certain locations based on the stratigraphy. The area has been studied extensively and previous investigations have identified glacial outwash that may contain larger boulders. The potential stratigraphic obstructions may result in the need to change the location of permanent injection wells or temporary injection points.

Equipment and materials required for construction are readily available. However, permanent injection wells and temporary injection point locations would have to be coordinated carefully and with input from existing land owners as they are responsible for the ownership and/or management of the area around the PBG Plume.

The biochemical product has been demonstrated to be effective in treating explosives and chlorinated solvents. Depending upon groundwater monitoring results, it is possible that this technology may require supplemental post-treatment applications.

### Cost

The estimated total cumulative costs for Alternative 4 are shown below. See Appendix I for a summary of the costs for Alternative 4.

Alternative 4 – Active Groundwater Remediation – Anaerobic Bioremediation
---

30 Years of Annual O&M: Total Cost:	\$ 4,913,113 <b>\$ 9.632.470</b>
Indirect Capital Cost:	\$ 1,464,628
Direct Capital Cost:	\$ 3,254,729

\* Total costs use current rates and do not include inflation

# 9.5 Alternative 5 – Well Replacement – Plume Area

The Well Replacement – Plume Area Alternative would involve replacing shallow aquifer wells, meeting replacement criteria, within the PBG Plume area with deeper aquifer wells. This alternative would also include continued groundwater monitoring of residential and monitoring wells and on-site groundwater access restrictions.

A reasonable worst-case scenario was developed considering potential plume migration which resulted in the potential for 47 existing wells being impacted. If sampling results indicate an increasing trend for a plume's COC in three consecutive rounds and that the plume is migrating toward a residential well, the Army will evaluate if well replacement is necessary.

Based on deeper aquifer well information in the area, replacement wells would be drilled to approximately 500 feet below the existing ground surface and into the Mt. Simon Sandstone Formation. This formation is isolated from the shallow impacted groundwater by a confining shale layer. The 500-foot depth is necessary to satisfy water quality and production criteria. Wells would be installed by a state licensed well driller and would be cased to isolate the shallow aquifer from the deeper bedrock aquifer. Connections from the well to the dwelling would be completed. Well replacement would be completed with abandonment of the shallow well and restoration of disturbed areas.

# **Overall Protection of Human Health and the Environment**

This alternative would be protective of human health as potential receptors would be provided potable water from a deeper aquifer. Effectively, there would be no route of entry through groundwater consumption, eliminating the risk of exposure through groundwater. Groundwater access is restricted within the BAAP boundary based on conditions of property transfer documentation. The groundwater access restrictions would require Army and WDNR authorization prior to well installation within the BAAP boundary.

# **Compliance with ARARs**

Groundwater monitoring would continue in monitoring and residential wells to monitor groundwater quality. Since the deep aquifer has been unimpacted by BAAP production or disposal activities, compliance with ARARs is expected. The contaminants within the plume are expected to comply with ARARs over time though natural degradation processes only.

### Long-Term Effectiveness and Permanence

This alternative would be an effective long-term and permanent solution. These wells are expected to provide receptors with long-term access to potable water that has been unimpacted by BAAP production or disposal activities. This alternative would also continue to restrict groundwater access within the BAAP property. Groundwater contamination within the plume is expected to decrease over time due to natural degradation processes only. The groundwater monitoring program is expected to continue for at least 30 years.

### Reduction of Toxicity, Mobility, and Volume through Treatment

The well replacement alternative would eliminate the groundwater exposure pathway by providing potential receptors access to potable water from a deep aquifer. Limited reductions in toxicity, mobility, and volume of contaminants within the plume is expected to occur through natural degradation processes only. This reduction would be verified through the monitoring program.

### **Short-Term Effectiveness**

For this alternative, there would be some short-term effects to workers, residents and the environment during implementation. The alternative would require off-site well installation on private property. Wells would be installed by a state licensed well driller and would be cased to isolate the shallow aquifer from the deeper bedrock aquifer. There is some risk associated with heavy equipment necessary for well installation. Proper training and equipment would be required to mitigate these risks. Near public road working conditions and work on private land would also be items that would need planning, coordination, and health and safety training.

Implementation and construction of this alternative is expected to be completed in approximately three months once qualifying criteria have been established for a residential well. Additional well replacements would be addressed as necessary upon establishment of qualifying criteria.

#### Implementability

Implementation of this alternative would involve well installation and residential connections on private property. Equipment and materials required for construction are readily available and wells would be installed by a state licensed well driller. However, well replacement would have to be coordinated with private land owners.

#### Cost

The estimated total cumulative costs for Alternative 5 are shown below. See Appendix I for a summary of the costs for Alternative 5.

#### Alternative 5 – Well Replacement – Plume Area

Total Cost:	\$ 7,919,246
30 Years of Annual O&M:	\$ 4,511,746
Indirect Capital Cost:	\$ 1,057,500
Direct Capital Cost:	\$ 2,350,000

\* Total costs use current rates and do not include inflation

# 9.6 Alternative 6 – Source Area Treatment

The Source Area Treatment Alternative would involve in-situ biochemical injection to treat elevated 2,6-DNT concentrations directly downgradient of the source area. This alternative would also include continued groundwater monitoring of residential and monitoring wells, on-site groundwater access restrictions and a provision for an alternate water supply condition.

As identified in Section 9.0, active remedial alternatives are only being developed for 2,6-DNT concentrations above the groundwater cleanup level listed in Table 16. Consequently, the in-situ biochemical injection locations would be strategically located to target elevated 2,6-DNT concentrations directly downgradient of the source area. This technology is expected to also reduce the concentrations of chlorinated solvents that coexist within the targeted treatment areas for 2,6-DNT.

For this alternative, a nutrient-enriched emulsified vegetable oil (EVO) is being proposed as the injection product. EVO has been used to stimulate in-situ anaerobic biodegradation of groundwater contaminants at commercial, industrial, and military sites. The procedures and applications of EVO are applicable to numerous anaerobically biodegradable contaminants including but not limited to chlorinated solvents, energetics, and nitrates.

The primary objective of injecting EVO into the groundwater is to stimulate the anaerobic biodegradation of the target contaminants. Groundwater aquifers are complex ecosystems populated by a broad and diverse array of microbial communities. The composition and activity of these microbial communities' changes continuously as their environment changes. Alterations in aquifer geochemistry and the availability of substrates and nutrients that can be used to generate energy and support growth and reproduction significantly affect microbial activity.

EVO would be distributed in the aquifer as an oil-in-water emulsion (mixture). In this approach, an oil-in-water emulsion would be first prepared using a food-grade oil, food-grade surfactants, and water. The emulsion would have small uniform droplets to allow transport in the aquifer. The emulsion would be injected into the aquifer (through injection wells or direct-push-technology) with additional water to distribute the oil droplets. The oil droplets would be distributed through the aquifer pore spaces and adhere to soil particles. The soil particle surfaces would gradually become coated with a thin layer of oil droplets that provide a carbon source for long-term anaerobic biodegradation. The oil droplets remain in the aquifer as a viable carbon source for approximately two years. Soluble substrates and nutrients (e.g., lactate, yeast extract, vitamins) can be added to the mixture prior to injection to stimulate rapid growth of desired bacteria. When the contaminated groundwater naturally flows toward and through the distributed EVO, the groundwater contaminants interact with the carbon source and break-down into less harmful byproducts.

It is anticipated that nine permanent injection wells would be installed and arranged in one treatment line located just downgradient of the source area. It is assumed that the source area would no longer contribute to the groundwater contamination. However, should this occur, the permanent wells could be utilized for additional injections. Anticipated treatment line locations are shown on Drawing PBG-ALT 6 in Appendix J.

The spacing of the treatment lines is based on a groundwater flow velocity of 306 ft/yr (see Table 8) and the viability of the carbon source remaining in the aquifer for approximately two years. Based on the geology and hydrogeology associated with the plume, a 25-foot radius of influence is anticipated to provide sufficient distribution of the EVO within the aquifer. The radius of influence is measured from the injection well or point location radially, out to the maximum extent of EVO product distribution. Each treatment line would be designed to fully capture contaminated groundwater migrating downgradient.

Though EVO is a proven technology to effectively treat chlorinated solvents and energetics, a field-scale pilot test would be necessary to determine the site specific constraints and a design to be developed to target 2,6-DNT within the PBG Plume. Upon successful completion of a field-scale pilot test, the remedial design could be finalized.

# **Overall Protection of Human Health and the Environment**

This alternative would be protective of human health and the environment. This alternative would be designed to treat the highest concentrations of 2,6-DNT in the PBG Plume directly downgradient of the source area. The remainder of the plume would degrade over time through natural processes only; however, the provision of the alternate water supply condition would address concerns associated with residential well impacts. Groundwater access restrictions would continue for areas within the BAAP.

# **Compliance with ARARs**

Concentrations of contaminants in the treated area are expected to comply with ARARs relatively quickly (approximately two years). The remainder of the plume would degrade over time through natural processes only; however, the provision of the alternate water supply condition would address concerns associated with residential well impacts.

# Long-Term Effectiveness and Permanence

It is anticipated that this alternative would be effective in the long term as the highest concentrations of 2,6-DNT would be treated directly downgradient of the source area. It is anticipated that a single round of injections of the biochemical product would be sufficient to treat the highest concentrations in the plume directly downgradient of the source. Based on a groundwater flow velocity and the viability of the carbon source, treatment is expected to take at approximately two years. However, depending upon groundwater monitoring results, it is possible that this technology may require supplemental post-treatment applications. The proposed biochemical product for use with this technology has shown successful contaminant reduction with explosives and chlorinated solvents; however, it has not been applied at full scale for 2,6-DNT treatment. Lastly, potential increases in groundwater table elevation may have the ability to mobilize residual contamination remaining in the vadose zone. This alternative would continue to restrict groundwater access within the BAAP and the provision of an alternate water supply condition for residential wells. The groundwater monitoring program is expected to continue at least 30 years.

### Reduction of Toxicity, Mobility, and Volume through Treatment

This alternative would reduce toxicity, mobility and volume of 2,6-DNT and chlorinated solvents in the treated areas more quickly than natural processes alone. It is assumed that there would be no additional contribution of 2,6-DNT from the source areas into the groundwater. Parts of the plume untreated are expected to decrease in concentration due to natural degradation processes. This reduction would be verified through the monitoring program.

### **Short-Term Effectiveness**

There would be minimal short-term effects to workers, residents and the environment during implementation as most of the work would be completed on-site. Generally, there is some risk associated with heavy equipment necessary for well installation and injection. Proper training and equipment would be required to mitigate these risks.

To maximize contaminant reduction, it is anticipated that additional investigation, sampling and testing would need to be completed. This effort is expected to take approximately two years. Construction and implementation of this alternative including permanent injection well installation, temporary injection point installation, biochemical injection and injection point abandonment is expected to be complete in approximately one year.

### Implementability

The installation of the permanent injection wells may be challenging at certain locations based on the stratigraphy. The area has been studied extensively and previous investigations have identified glacial outwash that may contain larger boulders. The potential stratigraphic obstructions may result in the need to change the location of the permanent injection wells.

Equipment and materials required for construction are readily available. However, permanent injection well locations would have to be coordinated carefully and with input from existing land owners as they are responsible for the ownership and/or management of the area around the PBG Plume.

The biochemical product has been demonstrated to be effective in treating explosives and chlorinated solvents. Depending upon groundwater monitoring results, it is possible that this technology may require supplemental post-treatment applications.

### Cost

The estimated total cumulative costs for Alternative 6 are shown below. See Appendix I for a summary of the costs for Alternative 6.

# Alternative 6 – Source Area Treatment

\$ 5,205,190
\$ 4,913,113
\$ 90,645
\$ 201,433
\$ \$ <b>\$</b>

\* Total costs use current rates and do not include inflation

### 9.7 PBG Plume Remedial Alternative Summary

A summary of the cleanup timeframe, treatment duration, groundwater monitoring duration, and cost for each of the six proposed remedial alternatives for the PBG Plume is presented below.

Alternative	Time to Achieve Cleanup	Active Treatment Duration	Groundwater Monitoring Duration	Total Cost
Alternative 1 – No Action	NA	NA	NA	\$0
Alternative 2 – MNA and Alternate Water Supply	30 Years	NA	30 Years	\$4.9
Alternative 3 – Pump and Treat	30 Years	8 Years	30 Years	\$12.7
Alternative 4 – Anaerobic Bioremediation	30 Years	2 Years	30 Years	\$9.6
Alternative 5 – Well Replacement	30 Years	NA	30 Years	\$7.9
Alternative 6 – Source Area Treatment	30 Years	2 Years	30 Years	\$5.2

### Propellant Burning Ground Plume Remedial Alternative Summary

Notes: Total cost in millions of dollars & includes direct capital, indirect capital and annual operation and maintenance costs. Total cost is based on current rates and does not include inflation.

An evaluation criteria summary of the proposed remedial alternatives for the PBG Plume is presented below. Each of the six proposed alternatives are listed in the left column. As described in Section 8.8, nine evaluation criteria have been developed to serve as the basis for conducting a detailed analysis of the remedial alternatives. The nine criteria include threshold, primary balancing and modifying criteria are listed below in the top row. The two modifying criteria (State Acceptance and Community Acceptance) are incorporated during the remedy selection stage and presented in the Proposed Plan.

An objective and qualitative evaluation was completed to compare the six proposed remedial alternatives. A designation of "H" represents a high confidence of the alternative meeting the criteria. Similarly, a designation of "L" represents a low and "M" represents a moderate

confidence of the alternative meeting the criteria. A designation of "N" represents no confidence of the alternative meeting the criteria and a designation of "TBD" represents to be determined. The designations are supported by their respective preceding text section for each remedial alternative and were made in relation to other alternatives. Approximate total costs for each remedial alternative are shown in the right column.

### Propellant Burning Ground Plume Evaluation Criteria Summary

				Evalua	ation	Crite	ria		
Alternative	Overall Protection to Human Health & Environment	Compliance with ARARs	Long-Term Effectiveness & Permanence	Reduction in Toxicity, Mobility & Volume Through Treatment	Short-Term Effectiveness	Implementability	State Acceptance	Community Acceptance	Total Cost (1)(2)
Alternative 1 – No Action • Groundwater access restrictions	L	N	Ν	Ν	N	Н	TBD	TBD	\$0
Alternative 2 – MNA and Alternate Water Supply Groundwater access restrictions Groundwater monitoring Alternate water supply	М	L	М	L	М	Н	TBD	TBD	\$4.9
<ul> <li>Alternative 3 – Pump and Treat <sup>(3)</sup></li> <li>Extraction wells (four)</li> <li>Mobile treatment units (four)</li> </ul>	Н	Н	Н	Н	М	М	TBD	TBD	\$12.7
<ul> <li>Alternative 4 – Anaerobic Bioremediation <sup>(3)</sup></li> <li>Permanent injection wells (nine)</li> <li>Temporary injection points (150)</li> </ul>	Н	Н	М	Н	М	М	TBD	TBD	\$9.6
<ul> <li>Alternative 5 – Well Replacement <sup>(4)</sup></li> <li>Replacement of residential wells (47)</li> </ul>	М	М	Н	L	М	М	TBD	TBD	\$7.9
<ul> <li>Alternative 6 – Source Area Treatment <sup>(3)</sup></li> <li>Permanent injection wells (nine)</li> </ul>	М	М	М	М	Н	М	TBD	TBD	\$5.2

Notes: H - High, L - Low, M - Moderate, N - None, TBD - To Be Determined.

(1) Cost in millions of dollars & includes direct capital, indirect capital and annual operation and maintenance costs.

- (2) Based on current rates and does not include inflation.
- (3) Alternative includes groundwater access restrictions, groundwater monitoring and alternate water supply.
- (4) Alternative includes groundwater access restrictions and groundwater monitoring.

# **10.0 REMEDIAL ALTERNATIVES – DBG PLUME**

As identified in Section 7.2, total DNT was the only risk-related COC considered for the development of remedial alternatives in the DBG Plume. The RAO for the DBG Plume requires the remedy to protect human health by preventing exposure to contaminated groundwater, to minimize the impact of the contaminants on the environment, and to restore groundwater to the extent practicable. The RAO for the DBG Plume will be achieved when groundwater concentrations of total DNT are below the groundwater cleanup level listed in Table 16.

Based on site conditions and the screening of process options, six remedial alternatives were developed to address the presence of total DNT in the DBG Plume. Alternative 1 - No Action, provides a baseline to evaluate the other alternatives.

### Alternative 1: No Action

The No Action Alternative is a mandatory evaluation that provides a baseline to evaluate the other alternatives. This alternative would have no impact on the contaminant plume and would not require groundwater monitoring of residential wells or monitoring wells. This alternative would include on-site groundwater access restrictions.

### Alternative 2: Monitored Natural Attenuation and Alternate Water Supply

The Monitored Natural Attenuation and Alternate Water Supply Alternative would continue the current remedial action approach and include the following components:

- Continued groundwater monitoring of residential and monitoring wells
- On-site groundwater access restrictions
- Provision for an alternate water supply condition including bottled water and well replacement

### Alternative 3: Active Groundwater Remediation – Pump and Treat

The Active Groundwater Remediation – Pump and Treat Alternative would target removing and treating impacted groundwater with elevated total DNT concentrations and include the following components:

- Continued groundwater monitoring of residential and monitoring wells
- On-site groundwater access restrictions
- Provision for an alternate water supply condition including bottled water and well replacement
- Groundwater removal through the installation of three groundwater extraction wells
- Groundwater treatment through the use of three mobile treatment units
- Groundwater disposal through the construction of piping leading to the Wisconsin River

### Alternative 4: Active Groundwater Remediation – Anaerobic Bioremediation

The Active Groundwater Remediation – Anaerobic Bioremediation Alternative would target treating impacted groundwater with elevated total DNT concentrations and include the following components:

- Continued groundwater monitoring of residential and monitoring wells
- On-site groundwater access restrictions
- Provision for an alternate water supply condition including bottled water and well replacement
- Groundwater treatment through in-situ biochemical injection at 406 temporary locations (on-site and off-site)

### Alternative 5: Well Replacement – Plume Area

The Well Replacement – Plume Area Alternative would involve replacing all shallow aquifer wells (meeting qualifying criteria) within the DBG Plume area with deeper aquifer wells and include the following components:

- Continued groundwater monitoring of residential and monitoring wells
- On-site groundwater access restrictions
- Replacement of as many as 57 existing residential wells

### Alternative 6: Source Area Treatment

The Source Area Treatment Alternative would target treating impacted groundwater with elevated total DNT concentrations directly downgradient of the source area and include the following components:

- Continued groundwater monitoring of residential and monitoring wells
- On-site groundwater access restrictions
- Provision for an alternate water supply condition including bottled water and well replacement
- Groundwater treatment through in-situ biochemical injection at 56 temporary on-site locations

### **10.1** Alternative 1 – No Action

The No Action Alternative would have no impact on any of the contaminant plumes and would not require groundwater monitoring of residential wells or monitoring wells. There would be no contaminant removal, treatment, containment or monitoring related to this alternative. As a condition of the Army's property transfer, groundwater access restrictions would continue for areas within the BAAP boundary.

### **Overall Protection of Human Health and the Environment**

Groundwater access is restricted within the BAAP boundary based on conditions of property transfer documentation. The groundwater access restrictions would require Army and WDNR authorization prior to well installation within the BAAP boundary; however, there are no groundwater access restrictions outside the BAAP boundary. This alternative would not provide any protection of human health or the environment beyond the groundwater access restrictions within the BAAP boundary. This alternative multiple access restrictions within the BAAP boundary. This alternative would result in the Army terminating the residential and monitoring well sampling program.

### **Compliance with ARARs**

The residential and monitoring well sampling program is being conducted in accordance with the most recent regulatory approval. This alternative would result in the Army terminating the residential and monitoring well sampling program. This alternative would not comply with ARARs.

### Long-Term Effectiveness and Permanence

This alternative would not provide an effective or permanent long-term solution. In this alternative, groundwater concentrations are expected to decrease as the chemicals would continue to undergo a slow degradation process (dilution, dispersion, and sorption). This alternative would result in the Army terminating the residential and monitoring well sampling program. Consequently, the degradation process would not be evaluated under this alternative.

### Reduction of Toxicity, Mobility, and Volume through Treatment

Limited reductions in toxicity, mobility, and volume would occur through natural degradation processes only. This alternative would discontinue the sampling of residential and groundwater monitoring wells. Consequently, the degradation process would not be evaluated.

### **Short-Term Effectiveness**

There would be no action taken for this alternative. Since groundwater monitoring would be discontinued, any groundwater exceedances would go unidentified. Therefore, this alternative has no short-term effects.

#### Implementability

This alternative is inherently implementable as no remedial action would be taken.

#### Cost

There is no cost associated with the No Action Alternative.

### 10.2 Alternative 2 – Monitored Natural Attenuation and Alternate Water Supply

The Monitored Natural Attenuation and Alternate Water Supply Alternative would include MNA for the DBG Plume, on-site groundwater access restrictions and a provision for an alternate water supply condition for residential wells. This alternative would also continue residential and monitoring well sampling of the DBG Plume as previously specified in Section 4.2 and Appendix D.

MNA relies on natural attenuation processes to achieve the RAO within a time frame that is reasonable compared to that offered by other more active remedial methods. MNA is expected to reduce the concentrations of the COCs that were carried forward in the development of

remedial alternatives (see Section 7.2), which includes only total DNT. These natural attenuation processes include a variety of physical, chemical, or biological processes that act without human intervention to reduce the mass, toxicity, mobility, volume, or concentration of contaminants in groundwater. These in-situ processes include biodegradation, dispersion, dilution, sorption, volatilization, and chemical or biological stabilization, transformation, or destruction of contaminants.

The Army currently has an environmental monitoring and health protection program in place that is protective of the residential water well users, "alternate water supply". If a Chapter NR 140 ES is exceeded in a residential well once, bottled water is made available to the occupant. If the exceedance occurs a second, consecutive time, well replacement is offered to the owner. Bottled water would be made available to the occupant until the well is replaced, operational and water quality verified (typically 3 months and based on driller availability). If the NR 140 ES exceedance detection, bottled water would be discontinued. The Army has replaced one residential well associated with the DBG Plume that has been impacted by total DNT.

# **Overall Protection of Human Health and the Environment**

This alternative would provide protection of human health and the environment due to groundwater access restrictions within the BAAP boundary and the provision of an alternate water supply condition for residential wells. The groundwater sampling program would monitor the groundwater concentrations for compliance and contaminant reduction.

# **Compliance with ARARs**

The residential and monitoring well sampling program is being conducted in accordance with the most recent regulatory approval. This alternative would continue the residential and groundwater monitoring program and comply with ARARs over time through natural degradation processes only.

# Long-Term Effectiveness and Permanence

This alternative offers a long-term solution as groundwater concentrations are expected to decrease as the chemicals would continue to undergo a slow degradation process (dilution, dispersion, and sorption). The alternative would continue to restrict groundwater access within the BAAP and the provision of an alternate water supply condition would address concerns associated with residential well impacts. Groundwater impacts are expected to remain and the groundwater monitoring program is expected to continue for at least 30 years.

# Reduction of Toxicity, Mobility, and Volume through Treatment

Limited reductions in toxicity, mobility, and volume is expected to occur through natural degradation processes only. This reduction would be verified through the monitoring program.

# **Short-Term Effectiveness**

This alternative offers a short-term solution as it is currently being applied and no additional work associated with implementation would be required. This alternative would continue to restrict groundwater access within the BAAP and the provision of an alternate water supply condition would address concerns associated with residential wells. If the alternate water supply provision is necessary, state licensed well drillers would be utilized for well replacement. The well drillers would be appropriately trained and would maintain applicable certifications to install any replacement well necessary.

### Implementability

This alternative would be easily implementable as this action is currently being applied to the site. No remedial activities other than sampling under the MNA program would be performed. Groundwater access restrictions are already in place within BAAP.

#### Cost

The estimated total cumulative costs for Alternative 2 are shown below. See Appendix I for a summary of the costs for Alternative 2.

### Alternative 2 – Monitored Natural Attenuation

Direct Capital Cost:	\$ 0
Indirect Capital Cost:	\$ 0
30 Years of Annual O&M:	\$ 4,240,490
<b>Total Present Worth:</b>	\$ 4,240,490

\* Total costs use current rates and do not include inflation

### 10.3 Alternative 3 – Active Groundwater Remediation – Pump and Treat

The Active Groundwater Remediation – Pump and Treat Alternative would include groundwater extraction and treatment with mobile treatment units and continued groundwater monitoring of residential and monitoring wells. This alternative would also include on-site groundwater access restrictions and a provision for an alternate water supply condition.

As identified in Section 10.0, active remedial alternatives are only being developed for total DNT concentrations above the groundwater cleanup level listed in Table 16. Consequently, the extraction wells would be strategically located to target elevated total DNT concentrations. This technology is expected to also reduce the concentrations of chlorinated solvents that coexist within the targeted treatment areas for total DNT.

It is anticipated that three extraction wells and three mobile treatment units (one treatment unit per extraction well) would be necessary to provide source control and minimize off site migration of the plume. Three extraction wells were selected based on previous performance (capture zone) of extraction wells located in the PBG Plume area. One extraction well would be

located directly downgradient of the source area (on-site), along the long axis of the plume, and within the highest total DNT concentration for source control. The other on-site extraction well would be located at the BAAP boundary to minimize off-site plume migration. One additional extraction well would be located off-site toward the southeastern end of the DBG Plume. Proposed pumping well locations and target pumping capture zones are shown on Drawing DBG-ALT 3 in Appendix J.

Each extraction well is expected to pump at approximately 500 gpm. Similarly, each mobile treatment unit would be designed to treat 500 gpm. Based on previous experience with pump and treat systems at BAAP (MIRM), groundwater flow velocities of 109 ft/yr (see Table 8) and assuming no additional source area contribution, the individual extraction wells and mobile treatment units are expected to operate continuously for various durations. The extraction well located closest to the source area is expected to operate for at least 10 years. The two other extraction wells are expected to operate for at least 22 years. The mobile treatment units are expected to use activated carbon as the primary treatment media as activated carbon has successfully treated DNT at BAAP. Site improvements including mobile treatment trailer staging area construction, electrical utility provision and site security would be necessary at each one of the extraction well/mobile treatment trailer areas.

A network of piping and appurtenances would be necessary to route extracted water from the extraction wells to the mobile treatment units and treated water from the mobile treatment units to a discharge pipeline leading to the Wisconsin River. It is anticipated that the pump and treat system would require the services of an environmental technician to monitor and maintain the extraction wells and mobile treatment units.

# **Overall Protection of Human Health and the Environment**

This alternative would be designed to control and limit the migration of and treat the groundwater with elevated total DNT concentrations. The provision of the alternate water supply condition would address concerns associated with residential well impacts.

### **Compliance with ARARs**

This alternative would be designed to comply with ARARs. The provision of the alternate water supply condition would address concerns associated with residential well impacts.

### Long-Term Effectiveness and Permanence

This alternative would be designed to reduce contaminant concentrations to comply with regulatory standards in groundwater through recovery and treatment of the portion of the DBG Plume with elevated total DNT concentrations. This alternative would continue to restrict groundwater access within the BAAP and the provision of an alternate water supply condition for residential wells. A similar pump and treat system was operated at BAAP (MIRM) and showed effective DNT concentration.

Based on previous experience, the groundwater pump and treat system's individual extraction wells and mobile treatment units are expected to operate continuously for various durations for up to 22 years. The groundwater monitoring program is expected to continue for at least 24 years.

### Reduction of Toxicity, Mobility, and Volume through Treatment

This alternative is expected to result in reductions in toxicity, mobility, and volume through treatment of the DBG Plume with elevated total DNT concentrations. It is assumed that there would be no additional contribution of total DNT from the source areas into the groundwater. The previous performance of this technology at the BAAP (MIRM), showed effective DNT concentration reduction. The groundwater contamination would also continue to decrease due to natural attenuation processes.

### **Short-Term Effectiveness**

For this alternative there would be some short-term effects to workers, residents and the environment during implementation. As described above, the technology would require two on-site and one off-site extraction wells coupled with a mobile treatment units for each extraction well. These locations would require construction of a staging area for the well and mobile treatment trailer, security and electricity for the site for operations and lighting.

It is anticipated that a new discharge pipeline would need to be constructed for the mobile treatment unit's discharge. From each extraction well and mobile treatment unit staging areas a discharge pipe would be constructed to transport treated water to the discharge piping leading to the Wisconsin River.

There is some risk associated with the operation of heavy equipment for site preparation, well drilling, excavation, piping installation and backfilling; however, proper training and equipment would be required to mitigate these risks. Utility crossing, near public road working conditions and work on private land would also be items that would need planning, coordination, and health and safety training.

To maximize contaminant reduction and plume migration control, it is anticipated that additional investigation, sampling and testing would need to be completed. This effort is expected to take approximately one year. Construction and implementation of this alternative including well installation, piping construction, treatment area preparation and utility extension is expected to be completed in approximately one year.

### Implementability

Equipment and materials required for construction of this alternative are readily available. However, extraction well and mobile treatment unit locations would have to be coordinated carefully and with input from the existing land owners as they are responsible for the ownership and/or management of the area around the DBG Plume. In addition, utilities to support the extraction wells and mobile treatment facility would need to be extended to the site, since none **Total Cost:** 

currently exist. The discharge line location would need to be determined and appropriate piping and appurtenance construction competed. The discharge location to the Wisconsin River would need to be identified during winter months with a high-visibility buoy system. This would identify open water as a safety precaution to those who utilize the Wisconsin River in the winter for recreational activities such as ice fishing and snowmobiling. It is expected that this buoy system would be installed during the winter months and subsequently removed in the spring. This process of installation and decommissioning the buoy system would need to be repeated each winter and spring, respectively, as long as the system continued operation.

### Cost

The estimated total cumulative costs for Alternative 3 are shown below. See Appendix I for a summary of the costs for Alternative 3.

Alternative 3 – Active Groundwa	iter Remedi	<u>ation – Pump a</u>	and Treat
Direct Capital Cost:	\$	2,776,030	
Indirect Capital Cost:	\$	1,249,214	
24 Years of Annual O&M:	\$	8,522,395	

\* Total costs use current rates and do not include inflation

### 10.4 Alternative 4 – Active Groundwater Remediation – Anaerobic Bioremediation

The Active Groundwater Remediation – Anaerobic Bioremediation Alternative would include in-situ anaerobic biodegradation targeting elevated concentrations of total DNT in groundwater and continued groundwater monitoring of residential and monitoring wells. This alternative would also include on-site groundwater access restrictions and a provision for an alternate water supply condition.

\$ 12,547,639

As identified in Section 10.0, active remedial alternatives are only being developed for total DNT concentrations above the groundwater cleanup level listed in Table 16. Consequently, the in-situ biochemical injection locations would be strategically located to target elevated total DNT concentrations. This technology is expected to also reduce the concentrations of chlorinated solvents that coexist within the targeted treatment areas for total DNT.

For this alternative, a nutrient-enriched emulsified vegetable oil (EVO) is being proposed as the injection product. EVO has been used to stimulate in-situ anaerobic biodegradation of groundwater contaminants at commercial, industrial, and military sites. The procedures and applications of EVO are applicable to numerous anaerobically biodegradable contaminants including but not limited to chlorinated solvents, energetics, and nitrates.

The primary objective of injecting EVO into the groundwater is to stimulate the anaerobic biodegradation of the target contaminants. Groundwater aquifers are complex ecosystems populated by a broad and diverse array of microbial communities. The composition and activity of these microbial communities' changes continuously as their environment changes. Alterations

in aquifer geochemistry and the availability of substrates and nutrients that can be used to generate energy and support growth and reproduction significantly affect microbial activity.

EVO would be distributed in the aquifer as an oil-in-water emulsion (mixture). In this approach, an oil-in-water emulsion would be first prepared using a food-grade oil, food-grade surfactants, and water. The emulsion would have small uniform droplets to allow transport in the aquifer. The emulsion would be injected into the aquifer (through injection wells or DPT) with additional water to distribute the oil droplets. The oil droplets would be distributed through the aquifer pore spaces and adhere to soil particles. The soil particle surfaces would gradually become coated with a thin layer of oil droplets that provide a carbon source for long-term anaerobic biodegradation. The oil droplets remain in the aquifer as a viable carbon source for approximately two years. Soluble substrates and nutrients (e.g., lactate, yeast extract, vitamins) can be added to the mixture prior to injection to stimulate rapid growth of desired bacteria. When the contaminated groundwater naturally flows toward and through the distributed EVO, the groundwater contaminants interact with the carbon source and break-down into less harmful byproducts.

It is anticipated that 406 injection points (both on-site and off-site and at varying stratigraphic depths) would be required to treat the plume. These injection points would be arranged in a series of 29 treatment lines and consist of temporary injection points. It is assumed that the source area would no longer contribute to the groundwater contamination. Anticipated treatment line locations are shown on Drawing DBG-ALT 4 in Appendix J.

The spacing of the treatment lines is based on a groundwater flow velocity of 109 ft/yr (see Table 8) and the viability of the carbon source remaining in the aquifer for approximately two years. The distance between each treatment line is based on two years of treatment. Based on the geology and hydrogeology associated with the plume, a 25-foot radius of influence is anticipated to provide sufficient distribution of the EVO within the aquifer. The radius of influence is measured from the injection well or point location radially, out to the maximum extent of EVO product distribution. Each treatment line would be designed to fully capture contaminated groundwater migrating downgradient.

Though EVO is a proven technology to effectively treat chlorinated solvents and energetics, a field-scale pilot test would be necessary to determine the site specific constraints and a design to be developed to target total DNT within the plume at BAAP. Upon successful completion of a field-scale pilot test, the remedial design could be finalized.

# **Overall Protection of Human Health and the Environment**

This alternative would be designed to meet the requirements of the RAO as it would effectively degrade the contaminants in the DBG Plume. The provision of the alternate water supply condition would address concerns associated with residential well impacts. Groundwater access restrictions would continue for areas within the BAAP.

### **Compliance with ARARs**

Concentrations of contaminants in the treated area are expected to comply with ARARs relatively quickly (approximately two years). The provision of the alternate water supply condition would address concerns associated with residential well impacts.

### Long-Term Effectiveness and Permanence

This alternative would be designed to reduce the concentration in groundwater to comply with regulatory standards for total DNT. This alternative would continue to restrict groundwater access within the BAAP and the provision of an alternate water supply condition for residential wells. Several issues have been identified regarding the alternative's long-term effectiveness and permanence. It is anticipated that a single round of injections of the biochemical product would be sufficient to treat the plume. Based on a groundwater flow velocity and the viability of the carbon source, treatment is expected to take approximately two years. However, depending upon groundwater monitoring results, it is possible that this technology may require supplemental post-treatment applications. The proposed biochemical product for use with this technology has shown successful contaminant reduction with chlorinated solvents; however, it has not been applied at full scale for total DNT treatment. The groundwater monitoring program is expected to continue for at least four years.

### Reduction of Toxicity, Mobility, and Volume through Treatment

This alternative is expected to reduce the toxicity, mobility, and volume of total DNT and chlorinated solvents in the treated areas more quickly than natural processes alone. It is assumed that there would be no additional contribution of total DNT from the source areas into the groundwater. The groundwater contamination would also continue to decrease due to natural attenuation processes.

### **Short-Term Effectiveness**

There would be some short-term effects to workers, residents and the environment during implementation. As described above, this alternative would require both on-site and off-site injection points.

There is some risk associated with heavy equipment necessary for temporary injection point installation and injection. Proper training and equipment would be required to mitigate these risks. The bioremediation is expected to occur over the course of two years and no additional worker safety issues have been identified. Near public road working conditions and work on private land would also be items that would need planning, coordination, and health and safety training.

To maximize contaminant reduction, it is anticipated that additional investigation, sampling and testing would need to be completed. This effort is expected to take approximately two years. Construction and implementation of this alternative including temporary injection point

installation, biochemical injection and injection point abandonment is expected to be completed in approximately one year.

### Implementability

The installation of temporary injection points may be challenging at certain locations based on the stratigraphy. The area has been studied extensively and previous investigations have identified glacial outwash that may contain larger boulders. The potential stratigraphic obstructions may result in the need to change locations of temporary injection points.

Equipment and materials required for construction are readily available. However, temporary injection point locations would have to be coordinated carefully and with input from existing land owners as they are responsible for the ownership and/or management of the area around the DBG Plume.

The biochemical product has been demonstrated to be effective in treating explosives and chlorinated solvents. Depending upon groundwater monitoring results, it is possible that this technology may require supplemental post-treatment applications.

#### Cost

The estimated total cumulative costs for Alternative 4 are shown below. See Appendix I for a summary of the costs for Alternative 4.

	Alternative 4 – Active Grou	ndwater Remediation -	- Anaerobic Bioremediation
--	-----------------------------	-----------------------	----------------------------

4 Years of Annual O&M: Total Cost:	\$ 706,748 <b>\$ 12,463,156</b>
Indirect Capital Cost:	\$ 3,648,540
Direct Capital Cost:	\$ 8,107,868

\* Total costs use current rates and do not include inflation

### **10.5** Alternative 5 – Well Replacement – Plume Area

The Well Replacement – Plume Area Alternative would involve replacing shallow aquifer wells, meeting replacement criteria, within the DBG Plume area with deeper aquifer wells. This alternative would also include continued groundwater monitoring of residential and monitoring wells and on-site groundwater access restrictions.

A reasonable worst-case scenario was developed considering potential plume migration which resulted in the potential for 57 existing wells being impacted. If sampling results indicate an increasing trend for a plume's COC in three consecutive rounds and that the plume is migrating toward a residential well, the Army will evaluate if well replacement is necessary.

Based on deeper aquifer well information in the area, replacement wells would be drilled to approximately 400 feet below the existing ground surface and into the Mt. Simon Sandstone

Formation. This sandstone formation is isolated from the shallow impacted groundwater by a confining layer of dolomite, shale, and siltstone. The 400-foot depth is necessary to satisfy water quality and production criteria. Wells would be installed by a state licensed well driller and would be cased to isolate the shallow aquifer from the deeper bedrock aquifer. Connections from the well to the dwelling would be completed. Well replacement would be completed with abandonment of the shallow well and restoration of disturbed areas.

# **Overall Protection of Human Health and the Environment**

This alternative would be protective of human health as potential receptors would be provided potable water from a deeper aquifer. Effectively, there would be no route of entry through groundwater consumption, eliminating the risk of exposure through groundwater. Groundwater access is restricted within the BAAP boundary based on conditions of property transfer documentation. The groundwater access restrictions would require Army and WDNR authorization prior to well installation within the BAAP boundary.

# **Compliance with ARARs**

Groundwater monitoring would continue in monitoring and residential wells to monitor groundwater quality. Since the deep aquifer has been unimpacted by BAAP production or disposal activities, compliance with ARARs is expected. The contaminants within the plume are expected to comply with ARARs over time through natural degradation processes only.

### Long-Term Effectiveness and Permanence

This alternative would be an effective long-term and permanent solution. These wells are expected to provide receptors with long-term access to potable water that has been unimpacted by BAAP production or disposal activities. This alternative would also continue to restrict groundwater access within the BAAP property. Groundwater contamination within the plume is expected to decrease over time due to natural degradation processes only. The groundwater monitoring program is expected to continue for at least 30 years.

# Reduction of Toxicity, Mobility, and Volume through Treatment

This alternative would eliminate the groundwater exposure pathway by providing potential receptors access to potable water from a deep aquifer. Limited reductions in toxicity, mobility, and volume of contaminants within the plume is expected to occur through natural degradation processes only. This reduction would be verified through the monitoring program.

### **Short-Term Effectiveness**

For this alternative, there would be some short-term effects to workers, residents and the environment during implementation. The alternative would require off-site well installation on private property. Wells would be installed by a state licensed well driller and would be cased to isolate the shallow aquifer from the deeper bedrock aquifer. There is some risk associated with heavy equipment necessary for well installation. Proper training and equipment would be

required to mitigate these risks. Near public road working conditions and work on private land would also be items that would need planning, coordination, and health and safety training.

Implementation and construction of this alternative is expected to be completed in approximately three months once qualifying criteria have been established for a residential well. Additional well replacements would be addressed as necessary upon establishment of qualifying criteria.

### Implementability

Implementation of this alternative would involve well installation and residential connections on private property. Equipment and materials required for construction are readily available and wells would be installed by a state licensed well driller. However, well replacement would have to be coordinated with private land owners.

### Cost

The estimated total cumulative costs for Alternative 5 are shown below. See Appendix I for a summary of the costs for Alternative 5.

### Alternative 5 – Well Replacement – Plume Area

Total Cost:	\$ 7,145,123
30 Years of Annual O&M:	\$ 3,839,123
Indirect Capital Cost:	\$ 1,026,000
Direct Capital Cost:	\$ 2,280,000

\* Total costs use current rates and do not include inflation

### 10.6 Alternative 6 – Source Area Treatment

The Source Area Treatment Alternative would involve in-situ biochemical injection to treat elevated total DNT concentrations directly downgradient of the source area. This alternative would also include continued groundwater monitoring of residential and monitoring wells, onsite groundwater access restrictions and a provision for an alternate water supply condition.

As identified in Section 10.0, active remedial alternatives are only being developed for total DNT concentrations above the groundwater cleanup level listed in Table 16. Consequently, the in-situ biochemical injection locations would be strategically located to target elevated total DNT concentrations directly downgradient of the source area. This technology is expected to also reduce the concentrations of chlorinated solvents that coexist within the targeted treatment areas for total DNT.

For this alternative, a nutrient-enriched emulsified vegetable oil (EVO) is being proposed as the injection product. EVO has been used to stimulate in-situ anaerobic biodegradation of groundwater contaminants at commercial, industrial, and military sites. The procedures and

applications of EVO are applicable to numerous anaerobically biodegradable contaminants including but not limited to chlorinated solvents, energetics, and nitrates.

The primary objective of injecting EVO into the groundwater is to stimulate the anaerobic biodegradation of the target contaminants. Groundwater aquifers are complex ecosystems populated by a broad and diverse array of microbial communities. The composition and activity of these microbial communities' changes continuously as their environment changes. Alterations in aquifer geochemistry and the availability of substrates and nutrients that can be used to generate energy and support growth and reproduction significantly affect microbial activity.

EVO would be distributed in the aquifer as an oil-in-water emulsion (mixture). In this approach, an oil-in-water emulsion would be first prepared using a food-grade oil, food-grade surfactants, and water. The emulsion would have small uniform droplets to allow transport in the aquifer. The emulsion would be injected into the aquifer (through injection wells or direct-push-technology) with additional water to distribute the oil droplets. The oil droplets would be distributed through the aquifer pore spaces and adhere to soil particles. The soil particle surfaces would gradually become coated with a thin layer of oil droplets that provide a carbon source for long-term anaerobic biodegradation. The oil droplets remain in the aquifer as a viable carbon source for approximately two years. Soluble substrates and nutrients (e.g., lactate, yeast extract, vitamins) can be added to the mixture prior to injection to stimulate rapid growth of desired bacteria. When the contaminated groundwater naturally flows toward and through the distributed EVO, the groundwater contaminants interact with the carbon source and break-down into less harmful byproducts.

It is anticipated that 56 temporary injection points would be installed on-site. These injection points would be arranged in a series of four treatment lines located just downgradient of the source area. It is assumed that the source area would no longer contribute to the groundwater contamination. Anticipated treatment line locations are shown on Drawing DBG-ALT 6 in Appendix J.

The spacing of the treatment lines is based on a groundwater flow velocity of 109 ft/yr (see Table 8) and the viability of the carbon source remaining in the aquifer for approximately two years. Based on the geology and hydrogeology associated with the plume, a 25-foot radius of influence is anticipated to provide sufficient distribution of the EVO within the aquifer. The radius of influence is measured from the injection well or point location radially, out to the maximum extent of EVO product distribution. Each treatment line would be designed to fully capture contaminated groundwater migrating downgradient.

Though EVO is a proven technology to effectively treat chlorinated solvents and energetics, a field-scale pilot test would be necessary to determine the site specific constraints and a design to be developed to target total DNT within the DBG Plume. Upon successful completion of a field-scale pilot test, the remedial design could be finalized.

### **Overall Protection of Human Health and the Environment**

This alternative would be protective of human health and the environment. This alternative would be designed to treat the highest concentrations of total DNT in the DBG Plume. The provision of the alternate water supply condition would address concerns associated with residential well impacts. Groundwater access restrictions would continue for areas within the BAAP.

### **Compliance with ARARs**

Concentrations of contaminants in the treated area are expected to comply with ARARs relatively quickly (approximately two years). The remainder of the plume's contamination would decrease over time through natural processes only; however, the provision of the alternate water supply condition would address concerns associated with residential well impacts.

#### Long-Term Effectiveness and Permanence

It is anticipated that this alternative would be effective in the long term as the highest concentrations of total DNT would be treated directly downgradient of the source area. It is anticipated that a single round of injections of the biochemical product would be sufficient to treat the highest concentrations in the plume directly downgradient of the source area. Based on a groundwater flow velocity and the viability of the carbon source, treatment is expected to take at approximately two years. However, depending upon groundwater monitoring results, it is possible that this technology may require supplemental post-treatment applications. The proposed biochemical product for use with this technology has shown successful contaminant reduction with explosives and chlorinated solvents; however, it has not been applied at full scale for total DNT treatment. This alternative would continue to restrict groundwater access within the BAAP and the provision of an alternate water supply condition for residential wells. The groundwater monitoring program is expected to continue for at least 30 years.

### Reduction of Toxicity, Mobility, and Volume through Treatment

This alternative would reduce toxicity, mobility and volume of total DNT and chlorinated solvents in the treated areas more quickly than natural processes alone. It is assumed that there would be no additional contribution of total DNT from the source areas into the groundwater. Portions of the plume untreated are expected to decrease in concentration due to natural attenuation processes. This reduction would be verified through the monitoring program.

#### **Short-Term Effectiveness**

There would be minimal short-term effects to workers, residents and the environment during implementation as most of the work would be completed on-site. Generally, there is some risk associated with heavy equipment necessary for temporary injection point installation, and injection. Proper training and equipment would be required to mitigate these risks.

To maximize contaminant reduction, it is anticipated that additional investigation, sampling and testing would need to be completed. This effort is expected to take approximately two years. Construction and implementation of this alternative including temporary injection point installation, biochemical injection and injection point abandonment is expected to be complete in approximately one year.

### Implementability

The installation of temporary injection points may be challenging at certain locations based on the stratigraphy. The area has been studied extensively and previous investigations have identified glacial outwash that may contain larger boulders. The potential stratigraphic obstructions may result in the need to change locations of temporary injection points.

Equipment and materials required for construction are readily available. However, temporary injection point locations would have to be coordinated carefully and with input from existing land owners as they are responsible for the ownership and/or management of the area around the DBG Plume.

The biochemical product has been demonstrated to be effective in treating explosives and chlorinated solvents. Depending upon groundwater monitoring results, it is possible that this technology may require supplemental post-treatment applications.

#### Cost

The estimated total cumulative costs for Alternative 6 are shown below. See Appendix I for a summary of the costs for Alternative 6.

### <u> Alternative 6 – Source Area Treatment</u>

Total Cost:	\$ 5,176,654
30 Years of Annual O&M:	\$ 4,240,490
Indirect Capital Cost:	\$ 290,534
Direct Capital Cost:	\$ 645,631

\* Total costs use current rates and do not include inflation

### **10.7 DBG Plume Remedial Alternative Summary**

A summary of the cleanup timeframe, treatment duration, groundwater monitoring duration, and cost for each of the six proposed remedial alternatives for the DBG Plume is presented below.

Alternative	Time to Achieve Cleanup	Active Treatment Duration	Groundwater Monitoring Duration	Total Cost
Alternative 1 – No Action	NA	NA	NA	\$0
Alternative 2 – MNA and Alternate Water Supply	30 Years	NA	30 Years	\$4.2
Alternative 3 – Pump and Treat	24 Years	22 Years	24 Years	\$12.5
Alternative 4 – Anaerobic Bioremediation	4 Years	2 Years	4 Years	\$12.5
Alternative 5 – Well Replacement	30 Years	NA	30 Years	\$7.1
Alternative 6 – Source Area Treatment	30 Years	2 Years	30 Years	\$5.2

# Deterrent Burning Ground Plume Remedial Alternative Summary

Notes: Total cost in millions of dollars & includes direct capital, indirect capital and annual operation and maintenance costs. Total cost is based on current rates and does not include inflation.

An evaluation criteria summary of the proposed remedial alternatives for the DBG Plume is presented below. Each of the six proposed alternatives are listed in the left column. As described in Section 8.8, nine evaluation criteria have been developed to serve as the basis for conducting a detailed analysis of the remedial alternatives. The nine criteria include threshold, primary balancing and modifying criteria are listed below in the top row. The two modifying criteria (State Acceptance and Community Acceptance) are incorporated during the remedy selection stage and presented in the Proposed Plan.

An objective and qualitative evaluation was completed to compare the six proposed remedial alternatives. A designation of "H" represents a high confidence of the alternative meeting the criteria. Similarly, a designation of "L" represents a low and "M" represents a moderate confidence of the alternative meeting the criteria. A designation of "N" represents no confidence of the alternative meeting the criteria and a designation of "TBD" represents to be determined. The designations are supported by their respective preceding text section for each remedial alternative and were made in relation to other alternatives. Approximate total costs for each remedial alternative are shown in the right column.

	Evaluation Criteria								
Alternative	Overall Protection to Human Health & Environment	Compliance with ARARs	Long-Term Effectiveness & Permanence	Reduction in Toxicity, Mobility & Volume Through Treatment	Short-Term Effectiveness	Implementability	State Acceptance	Community Acceptance	Total Cost <sup>(1)(2)</sup>
Alternative 1 – No Action • Groundwater access restrictions	L	N	Ν	N	N	Н	TBD	TBD	\$0
Alternative 2 – MNA and Alternate Water Supply Groundwater access restrictions Groundwater monitoring Alternate water supply	М	L	М	L	М	Н	TBD	TBD	\$4.2
<ul> <li>Alternative 3 – Pump and Treat <sup>(3)</sup></li> <li>Extraction wells (three)</li> <li>Mobile treatment units (three)</li> </ul>	Н	Н	Н	Н	М	М	TBD	TBD	\$12.5
<ul> <li>Alternative 4 – Anaerobic Bioremediation <sup>(3)</sup></li> <li>Biochemical injection points (406)</li> </ul>	Н	Н	М	Н	Μ	М	TBD	TBD	\$12.5
<ul> <li>Alternative 5 – Well Replacement <sup>(4)</sup></li> <li>Replacement of residential wells (57)</li> </ul>	М	М	Н	L	Μ	М	TBD	TBD	\$7.1
<ul> <li>Alternative 6 – Source Area Treatment <sup>(3)</sup></li> <li>Temporary injection points (56)</li> </ul>	М	М	М	М	Н	М	TBD	TBD	\$5.2

# Deterrent Burning Ground Plume Evaluation Criteria Summary

Notes: H - High, L - Low, M - Moderate, N - None, TBD - To Be Determined.

(1) Cost in millions of dollars & includes direct capital, indirect capital and annual operation and maintenance costs.

(2) Based on current rates and does not include inflation.

(3) Alternative includes groundwater access restrictions, groundwater monitoring and alternate water supply.

(4) Alternative includes groundwater access restrictions and groundwater monitoring.

# **11.0 REMEDIAL ALTERNATIVES – CENTRAL PLUME**

As identified in Section 7.3, 2,6-DNT was the only risk-related COC considered for the development of remedial alternatives in the Central Plume. The RAO for the Central Plume requires the remedy to protect human health by preventing exposure to contaminated groundwater, to minimize the impact of the contaminants on the environment, and to restore groundwater to the extent practicable. The RAO for the Central Plume will be achieved when groundwater concentrations of 2,6-DNT are below the groundwater cleanup level listed in Table 16.

Based on site conditions and the screening of process options, five remedial alternatives were developed to address the presence of 2,6-DNT in the Central Plume. A source area alternative was not developed for the Central Plume due to no known source areas remaining. Alternative 1 - No Action, provides a baseline to evaluate the other alternatives.

### Alternative 1: No Action

The No Action Alternative is a mandatory evaluation that provides a baseline to evaluate the other alternatives. This alternative would have no impact on the contaminant plume and would not require groundwater monitoring of residential wells or monitoring wells. This alternative would include on-site groundwater access restrictions.

### Alternative 2: Monitored Natural Attenuation and Alternate Water Supply

The Monitored Natural Attenuation and Alternate Water Supply Alternative would continue the current remedial action approach and include the following components:

- Continued groundwater monitoring of residential and monitoring wells
- On-site groundwater access restrictions
- Provision for an alternate water supply condition including bottled water and well replacement

### Alternative 3: Active Groundwater Remediation – Pump and Treat

The Active Groundwater Remediation – Pump and Treat Alternative would target removing and treating impacted groundwater with elevated 2,6-DNT concentrations and include the following components:

- Continued groundwater monitoring of residential and monitoring wells
- On-site groundwater access restrictions
- Provision for an alternate water supply condition including bottled water and well replacement
- Groundwater removal through the installation of eight groundwater extraction wells
- Groundwater treatment through the use of eight mobile treatment units
- Groundwater disposal through the construction of piping leading to the Wisconsin River
#### Alternative 4: Active Groundwater Remediation – Anaerobic Bioremediation

The Active Groundwater Remediation – Anaerobic Bioremediation Alternative would target treating impacted groundwater with elevated 2,6-DNT concentrations and include the following components:

- Continued groundwater monitoring of residential and monitoring wells
- On-site groundwater access restrictions
- Provision for an alternate water supply condition including bottled water and well replacement
- Groundwater treatment through in-situ biochemical injection at 988 temporary locations (on-site and off-site)

#### Alternative 5: Well Replacement – Plume Area

The Well Replacement – Plume Area Alternative would involve replacing shallow aquifer wells (meeting qualifying criteria) within the Central Plume area with deeper aquifer wells and include the following:

- Continued groundwater monitoring of residential and monitoring wells
- On-site groundwater access restrictions
- Replacement of as many as 23 existing residential wells

#### **11.1** Alternative 1 – No Action

The No Action Alternative would have no impact on any of the contaminant plumes and would not require groundwater monitoring of residential wells or monitoring wells. There would be no contaminant removal, treatment, containment or monitoring related to this alternative. As a condition of the Army's property transfer, groundwater access restrictions would continue for areas within the BAAP boundary.

#### **Overall Protection of Human Health and the Environment**

Groundwater access is restricted within the BAAP boundary based on conditions of property transfer documentation. The groundwater access restrictions would require Army and WDNR authorization prior to well installation within the BAAP boundary; however, there are no groundwater access restrictions outside the BAAP boundary. This alternative would not provide any protection of human health or the environment beyond the groundwater access restrictions within the BAAP boundary. This alternative multiple access restrictions within the BAAP boundary. This alternative would result in the Army terminating the residential and monitoring well sampling program.

#### **Compliance with ARARs**

The residential and monitoring well sampling program is being conducted in accordance with the most recent regulatory approval. This alternative would result in the Army terminating the residential and monitoring well sampling program. This alternative would not comply with ARARs.

#### Long-Term Effectiveness and Permanence

This alternative would not provide an effective or permanent long-term solution. In this alternative, groundwater concentrations are expected to decrease as the chemicals would continue to undergo a slow degradation process (dilution, dispersion, and sorption). This alternative would result in the Army terminating the residential and monitoring well sampling program. Consequently, the degradation process would not be evaluated under this alternative.

#### Reduction of Toxicity, Mobility, and Volume through Treatment

Limited reductions in toxicity, mobility, and volume would occur through natural degradation processes only. This alternative would discontinue the sampling of residential and groundwater monitoring wells. Consequently, the degradation process would not be evaluated.

#### **Short-Term Effectiveness**

There would be no action taken for this alternative. Since groundwater monitoring would be discontinued, any groundwater exceedances would go unidentified. Therefore, this alternative has no short-term effects.

#### Implementability

This alternative is inherently implementable as no remedial action would be taken.

#### Cost

There is no cost associated with the No Action Alternative.

#### **11.2** Alternative 2 – Monitored Natural Attenuation and Alternate Water Supply

The Monitored Natural Attenuation and Alternate Water Supply Alternative would include MNA for the Central Plume, on-site groundwater access restrictions and a provision for an alternate water supply condition for residential wells. This alternative would also continue residential and monitoring well sampling of the Central Plume as previously specified in Section 4.2 and Appendix D.

MNA relies on natural attenuation processes to achieve the RAO within a time frame that is reasonable compared to that offered by other more active remedial methods. MNA is expected to reduce the concentrations of the COCs that were carried forward in the development of remedial alternatives (see Section 7.3), which includes only 2,6-DNT. These natural attenuation processes include a variety of physical, chemical, or biological processes that act without human intervention to reduce the mass, toxicity, mobility, volume, or concentration of contaminants in groundwater. These in-situ processes include biodegradation, dispersion, dilution, sorption, volatilization, and chemical or biological stabilization, transformation, or destruction of contaminants.

The Army currently has an environmental monitoring and health protection program in place that is protective of the residential water well users, "alternate water supply". If a Chapter NR 140 ES is exceeded in a residential well once, bottled water is made available to the occupant. If the exceedance occurs a second, consecutive time, well replacement is offered to the owner. Bottled water would be made available to the occupant until the well is replaced, operational and water quality verified (typically 3 months and based on driller availability). If the NR 140 ES exceedance detection, bottled water would be discontinued. To date, the Army has replaced three shallow residential wells within the Central Plume.

### **Overall Protection of Human Health and the Environment**

This alternative would provide protection of human health and the environment due to groundwater access restrictions within the BAAP boundary and the provision of an alternate water supply condition for residential wells. The MNA program would monitor the groundwater concentrations for compliance and contaminant reduction.

### **Compliance with ARARs**

The residential and monitoring well sampling program is being conducted in accordance with the most recent regulatory approval. This alternative would continue the residential and groundwater monitoring program and comply with ARARs over time through natural degradation processes only.

### Long-Term Effectiveness and Permanence

This alternative offers long-term solution as groundwater concentrations are expected to decrease as the chemicals would continue to undergo a slow degradation process (dilution, dispersion, and sorption). The alternative would continue to restrict groundwater access within the BAAP and the provision of an alternate water supply condition would address concerns associated with residential well impacts. Groundwater impacts are expected to remain and the groundwater monitoring program is expected to continue for at least 30 years.

### Reduction of Toxicity, Mobility, and Volume through Treatment

Limited reductions in toxicity, mobility, and volume is expected to occur through natural degradation processes only. This reduction would be verified through the monitoring program.

#### **Short-Term Effectiveness**

This alternative offers a short-term solution as it is currently being applied and no additional work associated with implementation would be required. This alternative would continue to restrict groundwater access within the BAAP and the provision of an alternate water supply condition would address concerns associated with residential wells. If the alternate water supply provision is necessary, state licensed well drillers would be utilized for well replacement. The

well drillers would be appropriately trained and would maintain applicable certifications to install any replacement well necessary.

#### Implementability

This alternative would be easily implementable as this action is currently being applied to the site. No remedial activities other than sampling under the MNA program would be performed. Groundwater access restrictions are already in place within BAAP.

#### Cost

The estimated total cumulative costs for Alternative 2 are shown below. See Appendix I for a summary of the costs for Alternative 2.

	Alternative 2 – Monitored	Natural Attenuation	and Alternate	Water Supply
--	---------------------------	---------------------	---------------	--------------

Total Cost:	\$ 2,398,538
30 Years of Annual O&M:	\$ 2,398,538
Indirect Capital Cost:	\$ 0
Direct Capital Cost:	\$ 0

\* Total costs use current rates and do not include inflation

#### 11.3 Alternative 3 - Active Groundwater Remediation – Pump and Treat

The Active Groundwater Remediation – Pump and Treat Alternative would include groundwater extraction and treatment with mobile treatment units and continued groundwater monitoring of residential and monitoring wells. This alternative would also include on-site groundwater access restrictions and a provision for an alternate water supply condition.

As identified in Section 11.0, active remedial alternatives are only being developed for 2,6-DNT concentrations above the groundwater cleanup level listed in Table 16. Consequently, the extraction wells would be strategically located to target elevated 2,6-DNT concentrations. This technology is expected to also reduce the concentrations of chlorinated solvents that coexist within the targeted treatment areas for 2,6-DNT.

It is anticipated that eight extraction wells and eight mobile treatment units (one treatment unit per extraction well) would be necessary for source are reduction and plume migration control. Eight extraction wells were selected based on previous performance (capture zone) of extraction wells at the BAAP. Spatially, the wells would be located along the long axis of the plume and equidistant from one another and the plume's upgradient and downgradient extents. Proposed pumping well locations and target pumping capture zones are shown on Drawing Central-ALT 3 in Appendix J.

Each extraction well is expected to pump at approximately 500 gpm. Similarly, each mobile treatment unit would be designed to treat 500 gpm. Based on previous experience with pump and treat systems at BAAP, groundwater flow velocities of 143 ft/yr (see Table 8) and assuming

no additional source area contribution, the individual extraction wells and mobile treatment units are expected to operate continuously for at least 10 years. The mobile treatment units are expected to use activated carbon as the primary treatment media as activated carbon has successfully treated DNT at BAAP. Site improvements including mobile treatment trailer staging area construction, electrical utility provision and site security would be necessary at each one of the extraction well/mobile treatment trailer areas.

A network of piping and appurtenances would be necessary to route extracted water from the extraction wells to the mobile treatment units and treated water from the mobile treatment units to a discharge location. Treated groundwater would ultimately discharge to the Wisconsin River.

It is anticipated that the pump and treat system would require the services of an environmental technician to monitor and maintain the extraction wells and mobile treatment units.

#### **Overall Protection of Human Health and the Environment**

This alternative would be designed to control and limit the migration of and treat the groundwater with elevated 2,6-DNT concentrations. The provision of the alternate water supply condition would address concerns associated with residential well impacts.

#### **Compliance with ARARs**

This alternative would be designed to comply with ARARs. The provision of the alternate water supply condition would address concerns associated with residential well impacts.

### Long-Term Effectiveness and Permanence

This alternative would be designed to reduce contaminant concentrations to comply with regulatory standards in groundwater through recovery and treatment of the portion of the Central Plume with total elevated 2,6-DNT concentrations. This alternative would continue to restrict groundwater access within the BAAP and the provision of an alternate water supply condition for residential wells. A similar pump and treat system (MIRM) showed effective DNT concentration.

Based on previous experience, the groundwater pump and treat system is expected to operate continuously for 10 years. The groundwater monitoring program is expected to continue for at least 12 years.

#### Reduction of Toxicity, Mobility, and Volume through Treatment

This alternative is expected to result in reductions in toxicity, mobility, and volume through treatment of the Central Plume with elevated 2,6-DNT concentrations. It is assumed that there would be no additional contribution of 2,6-DNT from the source areas into the groundwater. Based on performance of this technology at the BAAP (MIRM), the pump and treat system showed effective DNT concentration reduction. The groundwater contamination would also continue to decrease due to natural attenuation processes.

#### **Short-Term Effectiveness**

For this alternative there would be some short-term effects to workers, residents and the environment during implementation. As described above, the alternative would require eight extraction wells coupled with a mobile treatment units for each extraction well. These locations would require construction of a staging area for the well and mobile treatment unit, security and electricity for the site for operations and lighting.

It is anticipated that from each extraction well to the and mobile treatment and from the mobile treatment unit to a discharge location, a discharge pipe would be constructed. Treated water is expected to be discharged to the Wisconsin River.

There is some risk associated with the operation of heavy equipment for site preparation, well drilling, excavation, piping installation and backfilling; however, proper training and equipment would be required to mitigate these risks. Utility crossing, near public road working conditions and work on private land would also be items that would need planning, coordination, and health and safety training.

To maximize contaminant reduction and plume migration control, it is anticipated that additional investigation, sampling and testing would need to be completed. This effort is expected to take approximately one year. Construction and implementation of this alternative including well installation, piping construction, treatment area preparation and utility extension is expected to be completed in approximately one year.

#### Implementability

Equipment and materials required for construction of this alternative are readily available. However, extraction well, mobile treatment unit locations and piping alignment would have to be coordinated carefully and with input from existing land owners as they are responsible for the ownership and/or management of the area around the Central Plume. In addition, utilities to support the extraction wells and mobile treatment facility would need to be extended to the site, since none currently exist. The discharge line location would need to be determined and appropriate piping and appurtenance construction competed. The discharge location to the Wisconsin River would need to be identified during winter months with a high-visibility buoy system. This would identify open water as a safety precaution to those who utilize the Wisconsin River in the winter for recreational activities such as ice fishing and snowmobiling. It is expected that this buoy system would be installed during the winter months and subsequently removed in the spring. This process of installation and decommissioning the buoy system would need to be repeated each winter and spring, respectively, as long as the system continued operation.

#### Cost

The estimated total cumulative costs for Alternative 3 are shown below. See Appendix I for a summary of the costs for Alternative 3.

#### <u>Alternative 3 – Active Groundwater Remediation – Pump and Treat</u>

¢	19 015 617
\$	7,953,709
\$	3,122,661
\$	6,939,247
	\$ \$ \$

\* Total costs use current rates and do not include inflation

#### 11.4 Alternative 4 – Active Groundwater Remediation – Anaerobic Bioremediation

The Active Groundwater Remediation – Anaerobic Bioremediation Alternative would include in-situ anaerobic biodegradation targeting elevated concentrations of 2,6-DNT in groundwater and continued groundwater monitoring of residential and monitoring wells. This alternative would also include on-site groundwater access restrictions and a provision for an alternate water supply condition.

As identified in Section 11.0, active remedial alternatives are only being developed for 2,6-DNT concentrations above the groundwater cleanup level listed in Table 16. Consequently, the in-situ biochemical injection locations would be strategically located to target elevated 2,6-DNT concentrations. This technology is expected to also reduce the concentrations of chlorinated solvents that coexist within the targeted treatment areas for 2,6-DNT.

For this alternative, a nutrient-enriched emulsified vegetable oil (EVO) is being proposed as the injection product. EVO has been used to stimulate in-situ anaerobic biodegradation of groundwater contaminants at commercial, industrial, and military sites. The procedures and applications of EVO are applicable to numerous anaerobically biodegradable contaminants including but not limited to chlorinated solvents, energetics, and nitrates.

The primary objective of injecting EVO into the groundwater is to stimulate the anaerobic biodegradation of the target contaminants. Groundwater aquifers are complex ecosystems populated by a broad and diverse array of microbial communities. The composition and activity of these microbial communities' changes continuously as their environment changes. Alterations in aquifer geochemistry and the availability of substrates and nutrients that can be used to generate energy and support growth and reproduction significantly affect microbial activity.

EVO would be distributed in the aquifer as an oil-in-water emulsion (mixture). In this approach, an oil-in-water emulsion would be first prepared using a food-grade oil, food-grade surfactants, and water. The emulsion would have small uniform droplets to allow transport in the aquifer. The emulsion would be injected into the aquifer (through injection wells or DPT) with additional water to distribute the oil droplets. The oil droplets would be distributed through the aquifer pore spaces and adhere to soil particles. The soil particle surfaces would gradually become coated with a thin layer of oil droplets that provide a carbon source for long-term anaerobic biodegradation. The oil droplets remain in the aquifer as a viable carbon source for approximately two years. Soluble substrates and nutrients (e.g., lactate, yeast extract, vitamins) can be added to the mixture prior to injection to stimulate rapid growth of desired bacteria. When the contaminated groundwater naturally flows toward and through the distributed EVO,

the groundwater contaminants interact with the carbon source and break-down into less harmful byproducts.

It is anticipated that 988 injection points (both on-site and off-site and at varying stratigraphic depths) would be required to treat the plume. These injection points would be arranged in a series of 38 treatment lines and consist of temporary injection points. It is assumed that the source area would no longer contribute to the groundwater contamination. Anticipated treatment line locations are shown on Drawing Central-ALT 4 in Appendix J.

The spacing of the treatment lines is based on a groundwater flow velocity of 143 ft/yr (see Table 8) and the viability of the carbon source remaining in the aquifer for approximately two years. The distance between each treatment line is based on two years of treatment. Based on the geology and hydrogeology associated with the plume, a 25-foot radius of influence is anticipated to provide sufficient distribution of the EVO within the aquifer. The radius of influence is measured from the injection well or point location radially, out to the maximum extent of EVO product distribution. Each treatment line would be designed to fully capture contaminated groundwater migrating downgradient.

Though EVO is a proven technology to effectively treat chlorinated solvents and energetics, a field-scale pilot test would be necessary to determine the site specific constraints and a design to be developed to target 2,6-DNT within the plume at BAAP. Upon successful completion of a field-scale pilot test, the remedial design could be finalized.

#### **Overall Protection of Human Health and the Environment**

This alternative would be designed to meet the requirements of the RAO as it would effectively degrade the contaminants in the Central Plume. The provision of the alternate water supply condition would address concerns associated with residential well impacts. Groundwater access restrictions would continue for areas within the BAAP.

#### **Compliance with ARARs**

Concentrations of contaminants in the treated area are expected to comply with ARARs relatively quickly (approximately two years). The provision of the alternate water supply condition would address concerns associated with residential well impacts.

#### Long-Term Effectiveness and Permanence

This alternative would be designed to reduce the concentration in groundwater to comply with regulatory standards for 2,6-DNT. This alternative would continue to restrict groundwater access within the BAAP and the provision of an alternate water supply condition for residential wells. It is anticipated that a single round of injections of the biochemical product would be sufficient to treat the plume. Based on a groundwater flow velocity and the viability of the carbon source, treatment is expected to take approximately two years. However, depending upon groundwater monitoring results, it is possible that this technology may require supplemental post-treatment applications. The proposed biochemical product for use with this technology has

shown successful contaminant reduction with explosives and chlorinated solvents; however, it has not been applied at full scale for 2,6-DNT treatment. The groundwater monitoring program is expected to continue for at least 4 years.

#### Reduction of Toxicity, Mobility, and Volume through Treatment

This alternative is expected to reduce the toxicity, mobility, and volume of 2,6-DNT and chlorinated solvents in the treated areas more quickly than natural processes alone. It is assumed that there would be no additional contribution of 2,6-DNT from the source areas into the groundwater. The groundwater contamination would also continue to decrease due to natural attenuation processes.

#### **Short-Term Effectiveness**

There would be some short-term effects to workers, residents and the environment during implementation. As described above, this alternative would require both on-site and off-site injection points.

There is some risk associated with heavy equipment necessary for temporary injection point installation and injection. Proper training and equipment would be required to mitigate these risks. The bioremediation is expected to occur over the course of two years and no additional worker safety issues have been identified. Near public road working conditions and work on private land would also be items that would need planning, coordination, and health and safety training.

To maximize contaminant reduction, it is anticipated that additional investigation, sampling and testing would need to be completed. This effort is expected to take approximately two years. Construction and implementation of this alternative including temporary injection point installation, biochemical injection and injection point abandonment is expected to be completed in approximately one year.

#### Implementability

The installation of temporary injection points may be challenging at certain locations based on the stratigraphy. The area has been studied extensively and previous investigations have identified glacial outwash that may contain larger boulders. The potential stratigraphic obstructions may result in the need to change the location of temporary injection points.

Equipment and materials required for construction are readily available. However, temporary injection point locations would have to be coordinated carefully and with input from existing land owners as they are responsible for the ownership and/or management of the area around the Central Plume.

The biochemical product has been demonstrated to be effective in treating explosives and chlorinated solvents. Depending upon groundwater monitoring results, it is possible that this technology may require supplemental post-treatment applications.

#### Cost

The estimated total cumulative costs for Alternative 4 are shown below. See Appendix I for a summary of the costs for Alternative 4.

Total Cost:	\$ 23,719,733
4 Years of Annual O&M:	\$ 399,756
Indirect Capital Cost:	\$ 7,237,234
Direct Capital Cost:	\$ 16,082,742
$\mathbf{D}^{\prime}$ , $\mathbf{C}^{\prime}$ , $1$	¢ 16 00 <b>0 7</b>

\* Total costs use current rates and do not include inflation

#### 11.5 Alternative 5 – Well Replacement – Plume Area

The Well Replacement – Plume Area Alternative would involve replacing shallow aquifer wells, meeting replacement criteria, within the Central Plume area with deeper aquifer wells. This alternative would also include continued groundwater monitoring of residential and monitoring wells and on-site groundwater access restrictions.

A reasonable worst-case scenario was developed considering potential plume migration which resulted in the potential for 23 existing wells being impacted. If sampling results indicate an increasing trend for a plume's COC in three consecutive rounds and that the plume is migrating toward a residential well, the Army will evaluate if well replacement is necessary.

Based on deeper aquifer well information in the area, replacement wells would be drilled to approximately 400 feet below the existing ground surface and into the Mt. Simon Sandstone Formation. This formation is isolated from the shallow impacted groundwater by a confining shale layer. The 400-foot depth is necessary to satisfy water quality and production criteria. Wells would be installed by a state licensed well driller and would be cased to isolate the shallow aquifer from the deeper bedrock aquifer. Connections from the well to the dwelling would be completed. Well replacement would be completed with abandonment of the shallow well and restoration of disturbed areas.

#### **Overall Protection of Human Health and the Environment**

This alternative would be protective of human health as potential receptors would be provided potable water from a deeper aquifer. Effectively, there would be no route of entry through groundwater consumption, eliminating the risk of exposure through groundwater. Groundwater access is restricted within the BAAP boundary based on conditions of property transfer documentation. The groundwater access restrictions would require Army and WDNR authorization prior to well installation within the BAAP boundary.

#### **Compliance with ARARs**

Groundwater monitoring would continue in monitoring and residential wells to monitor groundwater quality. Since the deep aquifer has been unimpacted by BAAP production or disposal activities, compliance with ARARs is expected. The contaminants within the plume are expected to comply with ARARs over time through natural degradation processes only.

#### Long-Term Effectiveness and Permanence

This alternative would be an effective long-term and permanent solution. These wells are expected to provide receptors with long-term access to potable water that has been unimpacted by BAAP production or disposal activities. This alternative would also continue to restrict groundwater access within the BAAP property. Groundwater contamination within the plume is expected to decrease over time due to natural degradation processes only. The groundwater monitoring program is expected to continue for at least 30 years.

#### **Reduction of Toxicity, Mobility, and Volume through Treatment**

This alternative would eliminate the groundwater exposure pathway by providing potential receptors access to potable water from a deep aquifer. Limited reductions in toxicity, mobility, and volume of contaminants within the plume is expected to occur through natural degradation processes only. This reduction would be verified through the monitoring program.

#### **Short-Term Effectiveness**

For this alternative, there would be some short-term effects to workers, residents and the environment during implementation. The alternative would require off-site well installation on private property. Wells would be installed by a state licensed well driller and would be cased to isolate the shallow aquifer from the deeper bedrock aquifer. There is some risk associated with heavy equipment necessary for well installation. Proper training and equipment would be required to mitigate these risks. Near public road working conditions and work on private land would also be items that would need planning, coordination, and health and safety training.

Implementation and construction of this alternative is expected to be completed in approximately three months once qualifying criteria have been established for a residential well. Additional well replacements would be addressed as necessary upon establishment of qualifying criteria.

#### Implementability

Equipment and materials required for construction of this alternative are readily available and wells would be installed by a state licensed well driller. However, well replacement would have to be coordinated with private land owners.

#### Cost

The estimated total cumulative costs for Alternative 5 are shown below. See Appendix I for a summary of the costs for Alternative 5.

#### <u>Alternative 5 – Well Replacement – Plume Area</u>

Total Cost:	\$ 3,331,172
30 Years of Annual O&M:	\$ 1,997,172
Indirect Capital Cost:	\$ 414,000
Direct Capital Cost:	\$ 920,000

\* Total costs use current rates and do not include inflation

#### 11.6 Central Plume Remedial Alternative Summary

A summary of the cleanup timeframe, treatment duration, groundwater monitoring duration, and cost for each of the five proposed remedial alternatives for the Central Plume is presented below.

Alternative	Time to Achieve Cleanup	Active Treatment Duration	Groundwater Monitoring Duration	Total Cost
Alternative 1 – No Action	NA	NA	NA	\$0
Alternative 2 – MNA and Alternate Water Supply	30 Years	NA	30 Years	\$2.4
Alternative 3 – Pump and Treat	12 Years	10 Years	12 Years	\$18.0
Alternative 4 – Anaerobic Bioremediation	4 Years	2 Years	4 Years	\$23.7
Alternative 5 – Well Replacement	30 Years	NA	30 Years	\$3.3

Central Plume Remedial Alternative Summary

Notes: Total cost in millions of dollars & includes direct capital, indirect capital and annual operation and maintenance costs. Total cost is based on current rates and does not include inflation.

An evaluation criteria summary of the proposed remedial alternatives related to the Central Plume is presented below. Each of the five proposed alternatives are listed in the left column. As described in Section 8.8, nine evaluation criteria have been developed to serve as the basis for conducting a detailed analysis of the remedial alternatives. The nine criteria include threshold, primary balancing and modifying criteria are listed below in the top row. The two modifying criteria (State Acceptance and Community Acceptance) are incorporated during the remedy selection stage and presented in the Proposed Plan.

An objective and qualitative evaluation was completed to compare the six proposed remedial alternatives. A designation of "H" represents a high confidence of the alternative meeting the criteria. Similarly, a designation of "L" represents a low and "M" represents a moderate confidence of the alternative meeting the criteria. A designation of "N" represents no confidence of the alternative meeting the criteria and a designation of "TBD" represents to be determined. The designations are supported by their respective preceding text section for each remedial alternative and were made in relation to other alternatives. Approximate total costs for each remedial alternative are shown in the right column.

Evaluation Criteria									
Alternative	Overall Protection to Human Health & Environment	Compliance with ARARs	Long-Term Effectiveness & Permanence	Reduction in Toxicity, Mobility & Volume Through Treatment	Short-Term Effectiveness	Implementability	State Acceptance	Community Acceptance	Total Cost (1)(2)
Alternative 1 – No Action • Groundwater access restrictions	L	N	Ν	Ν	N	Н	TBD	TBD	\$0
Alternative 2 – MNA and Alternate Water Supply Groundwater access restrictions Groundwater monitoring Alternate water supply	М	L	М	L	М	Н	TBD	TBD	\$2.4
<ul> <li>Alternative 3 – Pump and Treat <sup>(3)</sup></li> <li>Extraction wells (eight)</li> <li>Mobile treatment units (eight)</li> </ul>	Н	Н	Н	Н	М	М	TBD	TBD	\$18.0
<ul> <li>Alternative 4 – Anaerobic Bioremediation <sup>(3)</sup></li> <li>Biochemical injection points (988)</li> </ul>	Н	Н	М	Н	М	М	TBD	TBD	\$23.7
<ul> <li>Alternative 5 – Well Replacement <sup>(4)</sup></li> <li>Replacement of residential wells (23)</li> </ul>	М	М	Н	L	М	М	TBD	TBD	\$3.3

#### **Central Plume Evaluation Criteria Summary**

Notes: H – High, L – Low, M – Moderate, N – None, TBD – To Be Determined.

(1) Cost in millions of dollars & includes direct capital, indirect capital and annual operation and maintenance costs.

- (2) Based on current rates and does not include inflation.
- (3) Alternative includes groundwater access restrictions, groundwater monitoring and alternate water supply.
- (4) Alternative includes groundwater access restrictions and groundwater monitoring.

## **12.0 REMEDY SELECTION**

The Army's preferred alternative or remedy will be presented in the Proposed Plan; the remedy will be based on the results of this RI/FS. The Proposed Plan will briefly summarize the remedial investigation and the remedial alternatives evaluated in this RI/FS, highlighting the key factors that led to identifying the preferred alternative. The Army will submit the Proposed Plan to the regulatory agencies and then the public for review. After this review, the Army will release a Decision Document that documents the selected remedy, certifies that the remedy selection process was carried out in accordance with CERCLA, and addresses public comments on the Proposed Plan.

### **13.0 REFERENCES**

ABB Environmental Services, Inc., 1993. *Final Remedial Investigation Report, Badger Army Ammunition Plant*. Prepared for United States Army Toxic and Hazardous Materials Agency, Aberdeen Proving Ground, Maryland. April.

ABB, 1994. *Final Feasibility Study, Badger Army Ammunition Plant*. Prepared for United States Army Environmental Center, Aberdeen Proving Ground, Maryland. August.

Agency for Toxic Substances and Disease Registry (ATSDR), 1999. *Public Health Assessment for U.S. Army Badger Army Ammunition Plant, Baraboo, Sauk County, Wisconsin.* CERLIS No. WI9210020054. U.S. Department of Health and Human Services. May 28.

ATSDR, 2006. Health Consultation for Dinitrotoluene in Private Wells. U.S. Army Badger Army Ammunition Plant, Baraboo, Sauk County, Wisconsin. EPA Facility ID: WI9210020054. U.S. Department of Health and Human Services. September 30.

Attig, J.W., 1990. *Geology of the Badger Army Ammunition Plant Area*. Wisconsin Geological and Natural History Survey.

Badger Technical Services, LLC (BTS), 2012. Interim Remedial Measures Shutdown Plan. October.

BTS, 2012. Modified Interim Remedial Measures Shutdown Plan. January.

BTS, 2012. Letter, Subject: Vapor Intrusion Pathway Analysis Report. April 19.

BTS, 2012. Letter, Subject: Supplemental Vapor Intrusion Pathway Analysis Report. July 16.

BTS, 2013. Surface Waters Impact Investigation, Badger Army Ammunition Plant. November.

Clayton, L. and Attig, J.W., 1990. *Geology of Sauk County, Wisconsin*. Wisconsin Geological and Natural History Survey, Information Circular 67.

Department of the Army, 2011. *Letter, Subject: Deterrent Burning Ground – Final Remedy, Enhanced Biodegradation System Notification of Discontinuance, Badger Army Ammunition Plant.* November 17.

Department of the Army, 2012. Letter, Subject: Vapor Intrusion Pathway Analysis Reports. August 15.

Department of the Army, 2012. Letter, Subject: Surface Waters Impact Investigation Workplan. October 31.

Department of the Army, 2014. Letter, Subject: Request to Dismantle the Interim Remedial Measures System. June 17.

Department of the Army, 2016. *Letter, Subject: Submittal of May 2016 MIRM Phase 2 Groundwater Data.* Request to Dismantle Modified Interim Remedial Measures (MIRM) System. June 27.

Department of the Army, 2017. *Letter, Subject: Badger Army Ammunition Plant – Public Drinking Water System*. July 25.

Department of Defense (DoD), 2012. DoD Manual 4715.2. Defense Environmental Restoration Program (DERP) Management, March.

Gotkowitz, M.B. and Zeiler, K.K., 2005. *Water-Table Elevation Map of Sauk County, Wisconsin.* Wisconsin Geological and Natural History Survey, Miscellaneous Map 54-DI.

Gotkowitz, M.B., Zeiler, K.K., Dunning, C.P., Thomas, J.C., and Lin, Y.F., 2005. *Hydrogeology and Simulation of Groundwater Flow in Sauk County, Wisconsin*. Wisconsin Geological and Natural History Survey, Bulletin 102.

Hindall, S.M. and Borman, R.G., 1974. *Water Resources of Wisconsin Lower Wisconsin River Basin*. Wisconsin Geological and Natural History Survey, Atlas HA-479.

Hanson, G.F., 1970. *Geology of the Baraboo District, Wisconsin.* Wisconsin Geological and Natural History Survey, Informational Circular Number 14. Madison, Wisconsin.

Olin Corporation (Olin), 1996. Final Documentation Report For Soil Cover Construction, Racetrack And Thermal Treatment Unit Closure, Badger Army Ammunition Plant. October.

Olin, 1998. Landfill #1 Final Cap Construction Report, Badger Army Ammunition Plant. January.

Olin, 1999. 1949 Pit Phase One Cap, Final Construction Report, Badger Army Ammunition Plant. January.

Shaw Environmental, Inc. (Shaw), 2003. *Technical Memorandum - Data Report Biologically Enhanced Subsurface Treatment System – Soil Boring and Air Sparge Well Installation – Propellant Burning Ground, Badger Army Ammunition Plant*. Prepared for United States Army Corps of Engineers. July 16.

Shaw, 2004. Draft Preliminary Groundwater Investigation Report, Southern Boundary Groundwater, Badger Army Ammunition Plant. Prepared for United States Army Corps of Engineers. April 26.

Shaw, 2005. January 2005 Field Activities Technical Memorandum – Propellant Burning Ground, Badger Army Ammunition Plant. Prepared for United States Army Corps of Engineers. June 6.

Shaw, 2005. Draft Groundwater and Soil Investigation Report, Water's Edge Development, Badger Army Ammunition Plant. Prepared for United States Army Corps of Engineers. August 5.

Shaw, 2005. *Draft Southern Boundary Groundwater Phase II Investigation Report, Badger Army Ammunition Plant.* Prepared for United States Army Corps of Engineers. March 28.

Shaw, 2005. Draft Technical Memorandum Performance Assessment and Recommended Disposition of the Biologically Enhanced Subsurface Treatment System, Propellant Burning

*Ground, Badger Army Ammunition Plant.* Prepared for United States Army Corps of Engineers. November 14.

Shaw, 2006. Draft Alternative Feasibility Study, Propellant Burning Ground Waste Pits Subsurface Soil, Badger Army Ammunition Plant. Prepared for United States Army Corps of Engineers. April 6.

Shaw, 2006. Draft Corrective Measures Implementation Report, MIRM Extraction Well Realignment Project, Badger Army Ammunition Plant. Prepared for United States Army Corps of Engineers. April 10.

SpecPro, Inc., 2008. *Dinitrotoluene Sources Remedial Investigation Report, Badger Army Ammunition Plant*. March.

SpecPro, Inc., 2009. Construction Documentation Report, Propellant Burning Ground Phase 2 Cap Construction, Badger Army Ammunition Plant. February.

SpecPro, Inc., 2009. Northeast Boundary Groundwater Investigation Report, Badger Army Ammunition Plant. December.

SpecPro, Inc., 2010. Feasibility Report Contiguous Addition to Landfill 3646, Badger Army Ammunition Plant. October.

SpecPro, Inc. and Badger Technical Services, LLC, 2011. *Revised Alternative Feasibility Study, Groundwater Remedial Strategy, Badger Army Ammunition Plant.* December.

Stone and Webster, Inc. 1999. *Draft Technical Report – Natural Attenuation Screening Study for the Propellant Burning Ground, Badger Army Ammunition Plant*. Prepared for United States Army Corps of Engineers. August 25.

Stone and Webster, Inc. 2002. Draft Alternative Feasibility Study – Deterrent Burning Ground Waste Pits Subsurface Soil – Revision 2, Badger Army Ammunition Plant. Prepared for United States Army Corps of Engineers. April 24.

Stone and Webster, Inc., 2003. *Draft Groundwater Investigation Report, Deterrent Burning Ground, Badger Army Ammunition Plant*. Prepared for United States Army Corps of Engineers. January 3.

Sverdrup Environmental, Inc., 1997. *Final Documentation of Construction and Completion for Modified Interim Remedial Measures System at the Badger Army Ammunition Plant*. Prepared for United States Army Corps of Engineers. May.

Tsai, S.Y., Benioff, P.A., Chiu, S.Y., and Quinn, J.J., 1988. *Master Environmental Plan for the Badger Army Ammunition Plant*. Prepared for the U.S. Army Toxic and Hazardous Materials Agency. Argonne National Laboratory.

United States Army Corps of Engineers (USACE), Omaha District, 2013. *Five-Year Review Report Deterrent Burning Ground, Badger Army Ammunition Plant.* Prepared for Base Realignment and Closure Division, United States Army Environmental Command. June 26.

USACE, Omaha District, 2013. *Five-Year Review Report Propellant Burning Ground, Badger Army Ammunition Plant*. Prepared for Base Realignment and Closure Division, United States Army Environmental Command. June 26.

United States Army Toxic and Hazardous Materials Agency, 1989. *Decision Memorandum for an Interim Remedial Measures Plan at Badger Army Ammunition Plant Propellant Burning Grounds*. February.

U.S. Environmental Protection Agency (USEPA), 1987. *Interim Guidance on Compliance with Applicable or Relevant and Appropriate Requirements*, Office of Solid Waste and Emergency Response, (OSWER) Directive 9234.0-05, July 9, 1987.

USEPA, 1988, *Guidance for Conducting Remedial Investigations and Feasibility Studies Under CERCLA*. Interim Final. USEPA 540/G-89/004. October.

USEPA, 1989. *Risk Assessment Guidance for Superfund - Volume I: Human Health Evaluation Manual (Part A)*. Interim Final. EPA/540/1-89/002. December.

USEPA, 1994. *National Oil and Hazardous Substances Pollution Contingency Plan*. Office of Solid Waste and Emergency Response. Final Rule. September.

USEPA, 1999. Use of Monitored Natural Attenuation at Superfund, RCRA Corrective Action, and Underground Storage Tank Sites, Office of Solid Waste and Emergency Response, (OSWER) Directive 9200.4-17P, April.

USEPA, 2009. National Primary Drinking Water Regulations, EPA 816-F-09-004. May.

USEPA, 2009. 2009 Edition of the Drinking Water Standards and Health Advisories, Office of Water, EPA 822-R-09-011. October.

USEPA, 2011. Code of Federal Regulations (CFR), Remedial Investigation/Feasibility Study and Selection of Remedy. 40 CFR 300.430. July.

USEPA, 2015. *Technical Guide for Assessing and Mitigating the Vapor Intrusion Pathway from Subsurface Vapor Sources to Indoor Air*, Office of Solid Waste and Economy Response, (OSWER) Publication 9200.2-154, June.

USEPA, 2017. Regional Screening Levels (RSLs) - Generic Tables (November 2017). November.

Wisconsin Department of Natural Resources (WDNR), 1987. Letter, Subject: Final Review of In-Field Conditions Report at the Badger Army Ammunition Plant (BAAP), Sauk County. September 14.

WDNR, 1989. Letter, Subject: Closure Documentation Approval, Badger Army Ammunition Plant. September 20.

WDNR, 1995. Letter, Subject: Plan Modification of the September 14, 1987, In-Field Conditions Report Approval: Approval of Corrective Measures Selected in the Final Feasibility Study Report/Corrective Measures Study Report for the Badger Army Ammunition Plant. June 1. WDNR, 1997. Letter, Subject: Final Remedial Investigation Report, Racetrack Area Waste Pile Delineation/Characterization, Badger Army Ammunition Plant. December 2.

WDNR, 2002. Letter, Subject: Final Determination of Remedy for the Deterrent Burning Ground. October 14.

WDNR, 2003. Understanding Chlorinated Hydrocarbon Behavior in Groundwater: Investigation, Assessment and Limitations of Monitored Natural Attenuation, PUB-RR-699. April.

WDNR, 2007. WPDES Permit No. WI-0043975-05-0, Badger Army Ammunition Plant. (Effective July 1, 2007, through June 30, 2012). June 29.

WDNR, 2008. Letter, Subject: Final Determination of Remedy for Propellant Burning Ground Waste Pits Subsurface Soil. March 17.

WDNR, 2009. Letter, Subject: WDNR Review of Propellant Burning Ground Phase 2 Cap Construction Documentation Report. March 25.

WDNR, 2010. Addressing Vapor Intrusion at Remediation & Redevelopment Sites in Wisconsin, PUB-RR-800. December 16.

WDNR, 2012. Letter, Subject: Final Determination of Feasibility for an Alternative Groundwater Remedial Strategy at and near the Badger Army Ammunition Plant. June 28.

WDNR, 2012. Letter, Subject: WDNR Review of IRM Shutdown Proposal Report. December 11.

WDNR, 2013. Letter, Subject: WDNR Review of the Surface Water Investigation Proposal Report. January 4.

WDNR, 2014. Email Communication, Subject: Surface Water Investigation. January 16.

WDNR, 2014. Letter, Subject: Department Approval to Dismantle the Interim Remedial Measures Pump and Treat System. August 4.

WDNR, 2014. Letter, Subject: Department Approval of the Modified Interim Remedial Measures (MIRM) Shutdown Plan. August 4.

WDNR, 2014. Letter, Subject: Liability Clarification and Current Environmental Conditions at Badger Army Ammunition Plant, Landfill #1. August 27.

WDNR, 2016. Letter, Subject: Department Approval to Dismantle the Modified Interim Remedial Measures (MIRM) System. July 15.

WDNR, 2017. Chapter NR 140, Groundwater Quality, Wisconsin Administrative Code Register. February.

WDNR, 2017. Groundwater and Environmental Monitoring System (GEMS). Public access to groundwater monitoring data. http://dnr.wi.gov/topic/Landfills/gems.html

Wisconsin State Climatology Office, 2010. http://www.aos.wisc.edu/~sco/

Figures



# Explanation





Base from U.S. Geological Survey County Map Series (Topographic), 1986.

Cartography by D.L. Patterson

Subsummit terraces. Probably cut into Baraboo quartzite by wave erosion during the Ordovician.

# Geology of Sauk County, Wisconsin

# John W. Attig and Lee Clayton, 1990





Major geologic features of Sauk County during the maximum late Wisconsin extent of the Green Bay Lobe. The map shows the location of the western edge of the Green Bay Lobe (arrows show direction of ice flow), outwash plains (red), and lakes (blue). The Baraboo Hills and the remainder of the Driftless Area in Sauk County are shown in shades of gray.





Anticline.

Old open-pit iron mine.

\*

Fault (or narrow monocline).







# Badger Army Ammunition Plant Boundary Bedrock Surface Elevation Contour Contour Interval = 50 feet Wisconsin River

0

Road

Bedrock Surface Map RI/FS Badger Army Ammunition Plant





Feet

12,500

1 inch = 3,750 feet

6,250









## WELL DESIGNATION; SUFFIXES REFER TO WATER TABLE WELL (A) AND PIEZOMETER (B,C,D,E,F)






































\_\_\_\_







• Wells Used To Construct Isoconcentrations

Carbon Tetrachloride Concentration (µg/l)

2018 Groundwater Data































SPECPRO PROFESSIONAL SERVICES











• Wells Used To Construct Isoconcentrations

Trichloroethene Concentration (µg/I)

2018 Groundwater Data



















1

## LEGEND

• Wells Used To Construct Isoconcentrations

## Total Dinitrotoluene Concentration (µg/l)

# 2018 Groundwater Data











- NOTES:

- NOTES: 1. SEE FIGURE 6 FOR LOCATION AND ORIENTATION OF PROFILES 2. PROFILES ARE BASED ON AN INTERRETATION OF AVAILABLE SUBSILFACE DATA AND THE APRIL 1933 REMEDIAL INVESTIGATION REPORT PREPARED BY ABB ENVIRONMENTAL SERVICES, INC. 3. MSL MEAN SEAL LEVEL 4. TOTAL DINTROTOLUENE CONCENTRATIONS SHOWN IN RED ARE BASED ON ANALYTICAL DATA COLLECTED DURING 9/18. 5. GROUNDWATER FLOAL DATA COLLECTED DURING 9/18. 5. GROUNDWATER FLOAD US HORIZONTAL FROM NORTH TO SOUTH. 6. JULI- MICROGRAMS PER LITER 7. GROUNDWATER ANALYTICAL DATA WAS NOT AVAILABLE FOR ALL WELLS: NOT EVERY WELL IS ROUNTINELY SAMPLED. 8. WELLS COMMONT TO INTERRECTING CROSS SECTIONS ARE HIGHLIGHTED IN VELLOW.

# WELL DESIGNATION; SUFFIXES REFER TO WATER TABLE WELL (A) AND PIEZOMETER (B,C,D,E,F)





















# WELL DESIGNATION; SUFFIXES REFER TO WATER TABLE WELL (A) AND PIEZOMETER (B,C,D,E)

- GROUND SURFACE
- WATER TABLE ELEVATION

SCREENED INTERVAL TOTAL DINITROTOLUENE CONCENTRATION µg/I BOTTOM OF EXPLORATION TOTAL DINITROTOLUENE ISOCONCENTRATION LINE (µg/I)

### GEOLOGIC DESCRIPTIONS:



SILT AND CLAY MIXTURE, LOESS

SAND, MIXTURE OF VARYING GRAIN SIZES, GLACIAL TILL

SAND, GRAVEL, AND COBBLES, GLACIAL OUTWASH

SW-SP-SM SAND, MIXTURE OF VARYING GRAIN SIZES, GLACIAL OUTWASH

SILTY SAND, GLACIAL OUTWASH

LAYERED CLAY AND SILT, GLACIAL OUTWASH

GRAVEL, GLACIAL OUTWASH

BEDROCK, EAU CLAIRE FORMATION (SHALE, INTERBEDDED SANDSTONE)

### NOTES:

- NOTES:
  1. SEE FIGURE 6 FOR LOCATION AND ORIENTATION OF PROFILES
  2. PROFILES ARE BASED ON AN INTERPRETATION OF AVAILABLE SUBSURFACE DATA AND THE APRIL 1993 REMEDIAL INVESTIGATION REPORT PREPARED BY ABB ENVIRONMENTAL SERVICES, INC.
  3. MSL MEAN SEA LEVEL
  4. TOTAL DINITROTOLUENE CONCENTRATIONS SHOWN IN RED ARE BASED ON ANALYTICAL DATA COLLECTED DURING 9/18. ESTIMATED BOUNDARIES ARE DASHED.
  5. GROUNDWATER FLOW IS APPROXIMATELY PERPENDICULAR TO CROSS SECTION.
  9. µg/1- MICROGRAMS PER LITER
  7. GROUNDWATER ANALYTICAL DATA WAS NOT AVAILABLE FOR ALL WELLS: NOT EVERY WELL IS ROUNTINELY SAMPLED.
  8. WELLS COMMON TO INTERSECTING CROSS SECTIONS ARE HIGHLIGHTED IN YELLOW.

FIGURE 45

TOTAL DINITROTOLUENE **ISOCONCENTRATION SECTION F-F'** DETERRENT BURNING GROUND RI/FS BADGER ARMY AMMUNITION PLANT

















Appendix A

Summary of WDNR Conditions of Approval

# **Summary of WDNR Conditions of Approval**

The Wisconsin Department of Natural Resources (WDNR) June 28, 2012 Final Determination of Feasibility for an Alternative Groundwater Remedial Strategy approval letter established specific requirements (conditions) to be satisfied by the Army. The Army has addressed these conditions and provided appropriate documentation to the WDNR. The following summarizes the conditions of approval (*italics*) and the Army's subsequent activities (**bold**) to address these conditions.

# Conditions of Approval

- Condition 1: The Army shall continue operation of the Interim Remedial Measures (IRM) and Modified Interim Remedial Measures (MIRM) systems, as currently required, to collect and treat contaminated groundwater until modifications are approved by the Department in writing.
- Army Action: The Army submitted an Interim Remedial Measures (IRM) Shutdown Plan in October 2012 that outlined a systematic approach to restoring natural groundwater conditions so that PBG Plume dynamics and attenuation could be evaluated. Based on the WDNR's December 11, 2012 approval letter, the IRM was shut down on December 17, 2012. The Army's June 17, 2014 letter to the WDNR summarized the monitoring activities conducted during 2013 and 2014 and requested that the IRM system be dismantled. The WDNR's August 4, 2014 letter approved the dismantling of the IRM system. During 2014, the IRM extraction wells were abandoned and the IRM treatment building was demolished.

The Army submitted a Modified Interim Remedial Measures Shutdown Plan (MIRM) in January 2014 that outlined a systematic approach to restoring natural groundwater conditions so that PBG Plume dynamics and attenuation could be evaluated. Based on the WDNR's August 4, 2014 approval letter, the MIRM was completely shut down on August 31, 2015. The Army's June 27, 2016 letter to the WDNR summarized the monitoring activities conducted between 2014 and 2016 and requested that the MIRM system be dismantled. The WDNR's July 15, 2016 letter approved the dismantling of the MIRM system. During 2016, the MIRM extraction wells were abandoned. Ownership of the MIRM treatment building was transferred from the Army to the Bluffview Sanitary District in July 2016.

Condition 2: Prior to requesting modification or termination of the operation of the IRM or MIRM systems, the Army shall propose to the Department a process by which the effects of the requested changes can be predicted and evaluated. This proposal will be reviewed by the Department. Written Department approval will be required before any modifications are implemented.
- Army Action: See response to Condition 1. Both the IRM and MIRM Shutdown Plans provided a summary of the groundwater treatment operations, contaminant mass removal data, hydrogeologic conditions, contaminant trends in monitoring wells, and proposed shutdown activities.
- Condition 3: As part of the workplan for the phased shutdown of the IRM/MIRM, the Army shall prepare a comprehensive written report evaluating the effectiveness of the MIRM and IRM systems in preventing offsite groundwater contaminant migration. The evaluation shall include, but not be limited to, delineation of hydraulic capture zones, contaminant trends in select on- and off-site monitoring wells, calculations of contaminant travel times, and a concise statement concerning the effectiveness of the remedial systems in preventing offsite contaminant movement. Also included in the report shall be a description of the maintenance activities taken to keep the MIRM and IRM systems operational such as well chlorination and pump replacements.

#### Army Action: See response to Conditions 1 and 2.

Condition 4: The IRM and MIRM systems and all associated appurtenances shall be maintained in operational condition until such time that the Army obtains written Department approval to abandon or dismantle either or both systems.

### Army Action: Routine maintenance of the IRM and MIRM systems was performed prior to obtaining WDNR approval to dismantle each system.

- Condition 5: The Army shall propose an investigation and monitoring program to define the degree and extent to which contaminated groundwater is entering the Wisconsin River, Lake Wisconsin and/or other surface waters. This shall include, but not be limited to, installation of additional groundwater monitoring wells in addition to those already being used to monitor the plume(s) and may include the sampling of surface water.
- Army Action: On October 31, 2012, the Army provided to the WDNR the Surface Waters Impact Investigation workplan. The WDNR provided verbal approval of the workplan on January 4, 2013. The field investigation was conducted in June 2013 and the results were summarized in the Surface Waters Impact Investigation Report. The report was provided to the WDNR on November 21, 2013. The investigation involved collecting groundwater samples from temporary wells located near Weigand's Bay for the DBG Plume and near the Wisconsin River for the PBG Plume.
- *Condition 6: The BAAP groundwater monitoring program currently implemented by the Army shall continue until modifications are approved by the Department.*

### Army Action: The Army has continued their groundwater monitoring efforts consistent will all plan approvals and subsequent modifications.

- Condition 7: By July 1, 2013, the Army shall propose modifications to the groundwater monitoring program with the goal of providing data on the long-term effectiveness of natural attenuation as a remedial alternative. The requested changes in the monitoring program to evaluate natural attenuation shall encompass all three known groundwater contaminant plumes (propellant burning ground (PBG), deterrent burning ground (DBG) and central plumes). The proposal shall include (but not be limited to) a map or maps showing the names and locations of all monitoring wells associated with the property investigation and those that will be included in the groundwater monitored natural attenuation (MNA) network. The Army shall identify any locations where new wells will be installed to address any gaps in data collection to support monitored natural attenuation. The modification proposal shall include cross-sections, a table or tables providing information about the wells (by name or number identifier and/or associated license number) in the network, which *plume(s) the wells are associated with, the parameters for which groundwater* will be monitored, test methods used, and the frequency of sampling. The Army shall obtain written Department approval prior to implementing the modification to the groundwater monitoring network.
- Army Action: The Army, under separate covers, made three requests related to the groundwater monitoring modifications. The requested modifications related to data validation on April 4, 2013, private well sampling reduction dated May 14, 2013 and monitoring well sampling schedule on June 24, 2013. The WDNR approved these requests in a letter dated September 4, 2013.
- Condition 8: A groundwater narrative summary report similar in format to those submitted in the past several years shall be submitted to the Department annually for each calendar year by May 1 of the following year. Contents of the report shall focus on the results of work performed to evaluate monitored natural attenuation.

### Army Action: Groundwater narrative summary reports are completed annually and submitted to the WDNR.

- Condition 9: The Army shall conduct adequate saturated and unsaturated soil sampling, for all appropriate parameters, within the PBG, DBG and Central plumes to determine the nature and extent of site contaminants adsorbed onto the soil. Because back-diffusion of adsorbed waste constituents appears to be a major contributor to the groundwater plumes' stability, fully characterizing the adsorbed waste mass is necessary to evaluate natural attenuation as a possible remedial alternative.
- Army Action: Sampling within the final cap area associated with the PBG and DBG would compromise the barrier component of these systems. The barrier components of the PBG final cap consist of clay and geomembrane. The barrier components of the DBG final cap consist of geosynthetic clay and

geomembrane. These barrier components would need to be penetrated for samples to be obtained. In addition, the final cap layers above the barrier layer including the sand drainage layer, geotextile filter fabric, frost protection layer, and topsoil would need to be excavated at an angle for adequate access and subsequent cover repair/replacement.

Both the PBG and DBG have conditions of approval that strictly prohibit activities that would impact the integrity of the final cap system. Because of these factors and the conditions of approval for these facilities, the Army has not pursued sampling within the final cap area of the PBG or DBG.

The Army has conducted numerous soil investigations and remedial actions in the potential source areas of the Central Plume. These soil investigations and remedial actions are summarized in multiple reports submitted to the WDNR. The WDNR has provided the Army with multiple site closures related to the source areas for the Central Plume.

Condition 10: In conducting the required descriptions of plume configuration and behavior as well as the impact of past or future remedial efforts (as required in conditions 3, 5, 7 and 9, above) the Army shall consider that the plumes at Badger are three dimensional entities. All investigations and analyses described in this section shall be implemented to fully characterize in three dimensions the characteristics of all flow systems.

### Army Action: Investigations and analyses have been completed to characterize plume configurations and tendencies.

- Condition 11: Within 30 days of the date of approval, the Army shall provide a proposed schedule of events regarding the efforts to obtain local approval of the municipal water supply system. Inform the Department's Remediation and Redevelopment and Drinking Water and Groundwater programs, in writing, of any unforeseen delays in obtaining local approvals. Updates shall be submitted at least every 60 days.
- Army Action: The Army submitted the first proposed schedule of events on July 25, 2012. The Army's final submission was on November 8, 2016. These submissions were ceased in 2017 when the Army choose to not install a municipal water supply system.
- Condition 12: The source control action taken at the PBG and DBG shall be maintained as required in past approvals from the Department. Specifically, these approvals are Condition 6 of the October 14, 2002 approval for the PBG and Condition 5 of the March 17, 2008 approval for the DBG.

### Army Action: The Army completes annual cap inspections for the PBG and DBG and addresses corrective action as necessary. In addition, the Army controls

vegetation at these waste containment facilities through annual mowing. An annual cap and cover report containing inspection results, corrective actions and annual maintenance is completed and provided to the WDNR in January of each year for the previous year's activities.

Condition 13: If approved by local units of government, the Army shall obtain the Department's approval of plans for the municipal water supply system prior to commencing construction. Please be aware that you will also need approvals from the Public Service Commission of Wisconsin. Please contact them directly.

Army Action: The Army has determined that they neither have the legal nor funding authority to authorize a municipal water system. The Army's Supplemental Remedial Investigation/Feasibility Study is being provided within this document to outline remedial alternatives for groundwater contamination.

Condition 14: If the chosen remedy is not effective, the Department has the authority to require the Army to take additional actions to address contamination at the site.

Army Action: No action required.

### Appendix B

**PBG Waste Pits Soil Investigation Information (2005)** 



D:\BADGER\843333\F-0150\REV

### TABLE 1WASTE PIT 1 SAMPLE NUMBERS AND ANALYSES

	~ -			Ana	lyses				
Sample Number	Sample Interval (ft bgs)	VOC Headspace	VOCs	SVOCs	тос	Select Metals	Select Anions	pН	Comment
									Duplicate of PBB
PBB 0501 010	20 - 30	Х			Х	Х	Х	Х	0501 030
PBB 0501 022	21 - 22		Х	Х					
PBB 0501 026	25 - 26		Х	Х					
PBB 0501 030	20 - 30	Х			Х	Х	Х	Х	
PBB 0501 031	30 - 31		Х	Х					
PBB 0501 040	30 - 40	X			Х	Х	Х	Х	
PBB 0501 041	40 - 41		Х	Х					
PBB 0501 050	40 - 50	X			Х	Х	Х	Х	
PBB 0501 051	50 - 51		Х	Х					
PBB 0501 060	50 - 60	X			Х	Х	Х	Х	
PBB 0501 061	60 - 61		Х	Х					
PBB 0501 070	60 - 70	X			Х	Х	Х	Х	
PBB 0501 071	70 - 71		Х	Х					
PBB 0501 080	70 - 80	Х			Х	Х	Х	Х	
									Duplicate of PBB 0501 091 (VOCs &
PBB 0501 080	90 - 91		Х	X					SVOCs only)
PBB 0501 090	80 - 90	X			Х	Х	Х	Х	
PBB 0501 091	90 - 91		Х	Х					
PBB 0501 100	90 - 100	Х			Х	Х	Х	Х	

Notes:

ft bgs = feet below ground surface

SVOC = semivolatile organic compound

TOC = total organic carbon

VOC = volatile organic compound

## TABLE 2WASTE PIT 2 SAMPLE NUMBERS AND ANALYSES

				Ana	lyses				
Sample Number	Sample Interval (ft bgs)	VOC Headspace	VOCs	SVOCs	тос	Select Metals	Select Anions	pН	Comment
									Duplicate of PBB
PBB 0502 010	104 - 105				X	Х	X	Х	0502 105
PBB 0502 023	22 - 23		Х	Х					
PBB 0502 029	28 - 29		Х	Х					
PBB 0502 030	20 - 30	Х			Х	Х	Х	Х	
PBB 0502 035	34 - 35		Х	Х					
PBB 0502 040	30 - 40	Х			Х	Х	Х	Х	
PBB 0502 050	40 - 50	Х			Х	Х	Х	Х	
PBB 0502 053	52 - 53		Х	Х					
PBB 0502 060	50 - 60	Х			Х	Х	Х	Х	
PBB 0502 070	60 - 70	Х	Х	Х	Х	Х	Х	Х	
PBB 0502 080	70 - 80	Х			Х	Х	Х	Х	
PBB 0502 080	80 - 90		Х						Duplicate of PBB 0502 090 (VOCs only)
PBB 0502 090	80 - 90	Х	Х	Х	Х	Х	X	Χ	
PBB 0502 100	90 - 100	Х			Х	Х	X	Х	
PBB 0502 105	104 - 105		Х	Х	Х	Х	X	Х	

Notes:

ft bgs = feet below ground surface

SVOC = semivolatile organic compound

TOC = total organic carbon

VOC = volatile organic compound

## TABLE 3WASTE PIT 3 SAMPLE NUMBERS AND ANALYSES

				Ana	lyses				
Sample Number	Sample Interval (ft bgs)	VOC Headspace	VOCs	SVOCs	тос	Select Metals	Select Anions	pН	Comment
									Duplicate of PBB
PBB-0503 010	60 - 70				Х	Х	Х	Х	0503 070
PBB 0503 013	12 - 13		Х	Х					
PBB 0503 020	10 - 20	Х	Х	Х	Х	Х	Х	Х	
PBB 0503 030	20 - 30	Х	Х	Х	Х	Х	Х	Х	
PBB 0503 040	30 - 40	Х			Х	Х	Х	Х	
PBB 0503 050	40 - 50	Х			Х	Х	Х	Х	
PBB 0503 055	54 - 55		Х	Х					
PBB 0503 060	50 - 60	Х			Х	Х	Х	Х	
PBB 0503 070	60 -70	X	Х	Х	Х	Х	Х	Х	
									Duplicate of PBB
PBB 0503 080	70 - 80	Х	Х	Х	Х	Х	Х	Х	0503 090
PBB 0503 090	80 - 90	X	Х	X	Х	Х	Х	Х	
PBB 0503 100	90 - 100	X			Х	Х	Х	Х	
PBB 0503 105	100 - 105	X			Х	X	X	Х	

Notes:

ft bgs = feet below ground surface

SVOC = semivolatile organic compound

TOC = total organic carbon

VOC = volatile organic compound

TABLE 4
WASTE PIT 1 SOIL RESULTS COMPARISON

	Oct-91			Feb-97	Aug-02				Dec-03			Jan-05					
Depth (ft)	2,3-DNT (mg/kg)	2,4-DNT (mg/kg)	2,6-DNT (mg/kg)	2,4- & 2,6-DNT (mg/kg)	2,4- & 2,6-DNT (mg/kg)	2,3-DNT (mg/kg)	2,4-DNT (mg/kg)	2,6-DNT (mg/kg)	2,4- & 2,6-DNT (mg/kg)	2,3-DNT (mg/kg)	2,4-DNT (mg/kg)	2,6-DNT (mg/kg)	2,4- & 2,6-DNT (mg/kg)	2,3-DNT (mg/kg)	2,4-DNT (mg/kg)	2,6-DNT (mg/kg)	2,4- & 2,6-DNT (mg/kg)
20	NA	NA	NA		19,000	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
22	NA	NA	NA			NA	33,000	260	33,260	<340	47,000	<380	47,380	<120	13,000	<140	13,140
26	NA	56,000	ND	56000		NA	14,000	160	14,160	120	4,300	290	4,590	<31	5,000	<35	5,035
31	NA	8,200	ND	8200		NA	2,200	380	2,580	120	5,900	240	6,140	<30	5,100	<33	5,133
41	NA	5,700	1,000	6700		NA	15,000	1,500	16,500	<15	1,500	110	1,610	< 0.61	48	4.7	53
51	NA	4,700	1,000	5700		NA	39,000	1,400	40,400	<61	8,300	360	8,660	4.4	53	36	89
61	NA	8,500	1,000	9500		NA	8,200	200	8,400	350	9,200	2,000	11,200	13	8.2	41	49
71	NA	1,900	830	2730		NA	3	28	31	140	4,100	1,100	5,200	2.6	5.2	29	34
91	NA	58	12	70		NA	1	20	21	6	130	48	178	0.7	2.2	1.1	3
Total				88,900					115,352				84,958				23,536

Notes:

DNT = dinitrotoluene

ft = feet

mg/kg = milligram(s) per kilogram

NA = not analyzed

ND = not detected

-- = no data

	Feb-97 2,4- & 2,6-DNT (mg/kg)					Dec-03			Jan-05			
Depth (ft)				2,3-DNT (mg/kg)	2,4-DNT (mg/kg)	2,6-DNT (mg/kg)	2,4- & 2,6-DNT (mg/kg)	2,3-DNT (mg/kg)	2,4-DNT (mg/kg)	2,6-DNT (mg/kg)	2,4- & 2,6- DNT	
23			11,000	1.20	35	< 0.36	35	<16	2,800	<18	2,818	
25	25,000			NA	NA	NA	NA	NA	NA	NA	NA	
29			20,000	< 0.79	130	< 0.87	131	52	3,200	1,200	4,400	
30		3,500		NA	NA	NA	NA	NA	NA	NA	NA	
35				<6.6	1,200	<7.3	1,207	30	3,700	320	4,020	
53			11,000	<120	10,000	890	10,890	1.90	5	31	36	
70			5,300	15	640	100	740	9	570	160	730	
90			290	3	5.60	33	39	0.23	0.13	0.36	0.5	
105			3	0.23	0.17	0.16	0.33	0.045	0.087	< 0.032	0.12	
Total		76,093					13,042				12,005	

# TABLE 5WASTE PIT 2 SOIL RESULTS COMPARISON

Notes:

DNT = dinitrotoluene ft = feet mg/kg = milligram(s) per kilogram NA = not analyzed -- = no data

	Feb-97 2,4- & 2,6-DNT				Ι	Dec-03		Jan-05			
Depth				2,3-DNT	2,4-DNT	2,6-DNT	2,4- & 2,6-DNT	2,3-DNT	2,4-DNT	2,6-DNT	2,4- & 2,6-
(ft)	(mg/kg)			(mg/kg)	(mg/kg)	(mg/kg)	(mg/kg)	(mg/kg)	(mg/kg)	(mg/kg)	DNT
13			64,000	<15	2,800	<17	2,817	<160	16,000	<180	16,180
20	5,300	16,000	39,000	<6.2	1,100	<6.8	1,107	<120	16,000	<130	16,130
30			6,900	120	3,400	470	3,870	0.17	12	0.74	13
55			7,200	74	2,500	440	2,940	9.7	4.7	67	72
70			14,000	14	340	66	406	6.1	<1.1	160	161
80				0.75	17	3.6	20.6	0.29	0.19	1.9	2
90			2	0.17	0.35	0.18	0.53	0.23	0.24	16	16
Total	131,102						11,161				32,574

# TABLE 6WASTE PIT 3 SOIL RESULTS COMPARISON

Notes:

DNT = dinitrotoluene

ft = feet

mg/kg = milligram(s) per kilogram

-- = no data

Appendix C

**Groundwater Quality Regulations** 

# National Primary Drinking Water Regulations



Contaminant	MCL or TT <sup>1</sup> (mg/L) <sup>2</sup>	Potential health effects from long-term <sup>3</sup> exposure above the MCL	Common sources of contaminant in drinking water	Public Health Goal (mg/L) <sup>2</sup>
Acrylamide	TT <sup>4</sup>	Nervous system or blood problems; increased risk of cancer	Added to water during sewage/ wastewater treatment	zero
Alachlor	0.002	Eye, liver, kidney, or spleen problems; anemia; increased risk of cancer	Runoff from herbicide used on row crops	zero
Alpha/photon emitters	15 picocuries per Liter (pCi/L)	Increased risk of cancer	Erosion of natural deposits of certain minerals that are radioactive and may emit a form of radiation known as alpha radiation	zero
Reference Antimony	0.006	Increase in blood cholesterol; decrease in blood sugar	Discharge from petroleum refineries; fire retardants; ceramics; electronics; solder	0.006
ဆို Arsenic	0.010	Skin damage or problems with circulatory systems, and may have increased risk of getting cancer	Erosion of natural deposits; runoff from orchards; runoff from glass & electronics production wastes	0
Asbestos (fibers >10 micrometers)	7 million fibers per Liter (MFL)	Increased risk of developing benign intestinal polyps	Decay of asbestos cement in water mains; erosion of natural deposits	7 MFL
Atrazine	0.003	Cardiovascular system or reproductive problems	Runoff from herbicide used on row crops	0.003
ခဲ့တို့ Barium	2	Increase in blood pressure	Discharge of drilling wastes; discharge from metal refineries; erosion of natural deposits	2
Benzene	0.005	Anemia; decrease in blood platelets; increased risk of cancer	Discharge from factories; leaching from gas storage tanks and landfills	zero
Benzo(a)pyrene (PAHs)	0.0002	Reproductive difficulties; increased risk of cancer	Leaching from linings of water storage tanks and distribution lines	zero
ဆို Beryllium	0.004	Intestinal lesions	Discharge from metal refineries and coal-burning factories; discharge from electrical, aerospace, and defense industries	0.004
Beta photon emitters	4 millirems per year	Increased risk of cancer	Decay of natural and man-made deposits of certain minerals that are radioactive and may emit forms of radiation known as photons and beta radiation	zero
Bromate	0.010	Increased risk of cancer	Byproduct of drinking water disinfection	zero
ဆို Cadmium	0.005	Kidney damage	Corrosion of galvanized pipes; erosion of natural deposits; discharge from metal refineries; runoff from waste batteries and paints	0.005
Carbofuran	0.04	Problems with blood, nervous system, or reproductive system	Leaching of soil fumigant used on rice and alfalfa	0.04



DISINFECTANT





MICROORGANISM





#### National Primary Drinking Water Regulations

Contaminant	MCL or TT <sup>1</sup> (mg/L) <sup>2</sup>	Potential health effects from long-term³ exposure above the MCL	Common sources of contaminant in drinking water	Public Health Goal (mg/L)²
Carbon tetrachloride	0.005	Liver problems; increased risk of cancer	Discharge from chemical plants and other industrial activities	zero
Chloramines (as Cl <sub>2</sub> )	MRDL=4.0 <sup>1</sup>	Eye/nose irritation; stomach discomfort; anemia	Water additive used to control microbes	MRDLG=4 <sup>1</sup>
Chlordane	0.002	Liver or nervous system problems; increased risk of cancer	Residue of banned termiticide	zero
Chlorine (as Cl <sub>2</sub> )	MRDL=4.0 <sup>1</sup>	Eye/nose irritation; stomach discomfort	Water additive used to control microbes	MRDLG=4 <sup>1</sup>
Chlorine dioxide (as CIO <sub>2</sub> )	MRDL=0.8 <sup>1</sup>	Anemia; infants, young children, and fetuses of pregnant women: nervous system effects	Water additive used to control microbes	MRDLG=0.8 <sup>1</sup>
	1.0	Anemia; infants, young children, and fetuses of pregnant women: nervous system effects	Byproduct of drinking water disinfection	0.8
Chlorobenzene	0.1	Liver or kidney problems	Discharge from chemical and agricultural chemical factories	0.1
ဆို Chromium (total)	0.1	Allergic dermatitis	Discharge from steel and pulp mills; erosion of natural deposits	0.1
လို Copper	TT⁵; Action Level=1.3	Short-term exposure: Gastrointestinal distress. Long- term exposure: Liver or kidney damage. People with Wilson's Disease should consult their personal doctor if the amount of copper in their water exceeds the action level	Corrosion of household plumbing systems; erosion of natural deposits	1.3
Cryptosporidium	TT7	Short-term exposure: Gastrointestinal illness (e.g., diarrhea, vomiting, cramps)	Human and animal fecal waste	zero
Cyanide (as free cyanide)	0.2	Nerve damage or thyroid problems	Discharge from steel/metal factories; discharge from plastic and fertilizer factories	0.2
() 2,4-D	0.07	Kidney, liver, or adrenal gland problems	Runoff from herbicide used on row crops	0.07
Dalapon	0.2	Minor kidney changes	Runoff from herbicide used on rights of way	0.2
1,2-Dibromo-3- chloropropane (DBCP)	0.0002	Reproductive difficulties; increased risk of cancer	Runoff/leaching from soil fumigant used on soybeans, cotton, pineapples, and orchards	zero
o-Dichlorobenzene	0.6	Liver, kidney, or circulatory system problems	Discharge from industrial chemical factories	0.6
p-Dichlorobenzene	0.075	Anemia; liver, kidney, or spleen damage; changes in blood	Discharge from industrial chemical factories	0.075
1,2-Dichloroethane	0.005	Increased risk of cancer	Discharge from industrial chemical factories	zero

LEGEND

L+ DISINFECTANT





MICROORGANISM





#### National Primary Drinking Water Regulations

Contaminant	MCL or TT <sup>1</sup> (mg/L) <sup>2</sup>	Potential health effects from long-term³ exposure above the MCL	Common sources of contaminant in drinking water	Public Health Goal (mg/L) <sup>2</sup>
1,1-Dichloroethylene	0.007	Liver problems	Discharge from industrial chemical factories	0.007
cis-1,2- Dichloroethylene	0.07	Liver problems	Discharge from industrial chemical factories	0.07
trans-1,2, Dichloroethylene	0.1	Liver problems	Discharge from industrial chemical factories	0.1
Dichloromethane	0.005	Liver problems; increased risk of cancer	Discharge from industrial chemical factories	zero
1,2-Dichloropropane	0.005	Increased risk of cancer	Discharge from industrial chemical factories	zero
Di(2-ethylhexyl) adipate	0.4	Weight loss, liver problems, or possible reproductive difficulties	Discharge from chemical factories	0.4
Di(2-ethylhexyl) phthalate	0.006	Reproductive difficulties; liver problems; increased risk of cancer	Discharge from rubber and chemical factories	zero
Dinoseb	0.007	Reproductive difficulties	Runoff from herbicide used on soybeans and vegetables	0.007
Dioxin (2,3,7,8-TCDD)	0.00000003	Reproductive difficulties; increased risk of cancer	Emissions from waste incineration and other combustion; discharge from chemical factories	zero
Diquat	0.02	Cataracts	Runoff from herbicide use	0.02
Endothall	0.1	Stomach and intestinal problems	Runoff from herbicide use	0.1
Endrin	0.002	Liver problems	Residue of banned insecticide	0.002
Epichlorohydrin	TT <sup>4</sup>	Increased cancer risk; stomach problems	Discharge from industrial chemical factories; an impurity of some water treatment chemicals	zero
Ethylbenzene	0.7	Liver or kidney problems	Discharge from petroleum refineries	0.7
Ethylene dibromide	0.00005	Problems with liver, stomach, reproductive system, or kidneys; increased risk of cancer	Discharge from petroleum refineries	zero
Fecal coliform and <i>E. coli</i>	MCL <sup>6</sup>	Fecal coliforms and <i>E. coli</i> are bacteria whose presence indicates that the water may be contaminated with human or animal wastes. Microbes in these wastes may cause short term effects, such as diarrhea, cramps, nausea, headaches, or other symptoms. They may pose a special health risk for infants, young children, and people with severely compromised immune systems.	Human and animal fecal waste	zero <sup>6</sup>

LEGEND

+ DISINFECTANT





MICROORGANISM





	Contaminant	MCL or TT <sup>1</sup> (mg/L) <sup>2</sup>	Potential health effects from long-term <sup>3</sup> exposure above the MCL	Common sources of contaminant in drinking water	Public Health Goal (mg/L) <sup>2</sup>
ංදිං	Fluoride	4.0	Bone disease (pain and tenderness of the bones); children may get mottled teeth	Water additive which promotes strong teeth; erosion of natural deposits; discharge from fertilizer and aluminum factories	4.0
	Ciardia lamblia	TT7	Short-term exposure: Gastrointestinal illness (e.g., diarrhea, vomiting, cramps)	Human and animal fecal waste	zero
$\bigcirc$	Glyphosate	0.7	Kidney problems; reproductive difficulties	Runoff from herbicide use	0.7
A	Haloacetic acids (HAA5)	0.060	Increased risk of cancer	Byproduct of drinking water disinfection	n/aº
$\bigcirc$	Heptachlor	0.0004	Liver damage; increased risk of cancer	Residue of banned termiticide	zero
$\bigcirc$	Heptachlor epoxide	0.0002	Liver damage; increased risk of cancer	Breakdown of heptachlor	zero
	Heterotrophic plate count (HPC)	TT7	HPC has no health effects; it is an analytic method used to measure the variety of bacteria that are common in water. The lower the concentration of bacteria in drinking water, the better maintained the water system is.	HPC measures a range of bacteria that are naturally present in the environment	n/a
$\bigcirc$	Hexachlorobenzene	0.001	Liver or kidney problems; reproductive difficulties; increased risk of cancer	Discharge from metal refineries and agricultural chemical factories	zero
$\bigcirc$	Hexachloro- cyclopentadiene	0.05	Kidney or stomach problems	Discharge from chemical factories	0.05
ංදිං	Lead	TT⁵; Action Level=0.015	Infants and children: Delays in physical or mental development; children could show slight deficits in attention span and learning abilities; Adults: Kidney problems; high blood pressure	Corrosion of household plumbing systems; erosion of natural deposits	zero
	Legionella	TT7	Legionnaire's Disease, a type of pneumonia	Found naturally in water; multiplies in heating systems	zero
$\bigcirc$	Lindane	0.0002	Liver or kidney problems	Runoff/leaching from insecticide used on cattle, lumber, and gardens	0.0002
ංදිං	Mercury (inorganic)	0.002	Kidney damage	Erosion of natural deposits; discharge from refineries and factories; runoff from landfills and croplands	0.002
$\bigcirc$	Methoxychlor	0.04	Reproductive difficulties	Runoff/leaching from insecticide used on fruits, vegetables, alfalfa, and livestock	0.04
ංරිං	Nitrate (measured as Nitrogen)	10	Infants below the age of six months who drink water containing nitrate in excess of the MCL could become seriously ill and, if untreated, may die. Symptoms include shortness of breath and blue-baby syndrome.	Runoff from fertilizer use; leaching from septic tanks, sewage; erosion of natural deposits	10



DISINFECTANT

ঙ্গ

÷





MICROORGANISM



RADIONUCLIDES

#### National Primary Drinking Water Regulations

Contaminant	MCL or TT <sup>1</sup> (mg/L) <sup>2</sup>	Potential health effects from long-term³ exposure above the MCL	Common sources of contaminant in drinking water	Public Health Goal (mg/L)²
Nitrite (measured as Nitrogen)	1	Infants below the age of six months who drink water containing nitrite in excess of the MCL could become seriously ill and, if untreated, may die. Symptoms include shortness of breath and blue-baby syndrome.	Runoff from fertilizer use; leaching from septic tanks, sewage; erosion of natural deposits	1
Oxamyl (Vydate)	0.2	Slight nervous system effects	Runoff/leaching from insecticide used on apples, potatoes, and tomatoes	0.2
Pentachlorophenol	0.001	Liver or kidney problems; increased cancer risk	Discharge from wood-preserving factories	zero
Picloram	0.5	Liver problems	Herbicide runoff	0.5
Polychlorinated biphenyls (PCBs)	0.0005	Skin changes; thymus gland problems; immune deficiencies; reproductive or nervous system difficulties; increased risk of cancer	Runoff from landfills; discharge of waste chemicals	zero
Radium 226 and Radium 228 (combined)	5 pCi/L	Increased risk of cancer	Erosion of natural deposits	zero
Selenium	0.05	Hair or fingernail loss; numbness in fingers or toes; circulatory problems	Discharge from petroleum and metal refineries; erosion of natural deposits; discharge from mines	0.05
Simazine	0.004	Problems with blood	Herbicide runoff	0.004
Styrene	0.1	Liver, kidney, or circulatory system problems	Discharge from rubber and plastic factories; leaching from landfills	0.1
Tetrachloroethylene	0.005	Liver problems; increased risk of cancer	Discharge from factories and dry cleaners	zero
ဆို Thallium	0.002	Hair loss; changes in blood; kidney, intestine, or liver problems	Leaching from ore-processing sites; discharge from electronics, glass, and drug factories	0.0005
Toluene	1	Nervous system, kidney, or liver problems	Discharge from petroleum factories	1
Total Coliforms	5.0 percent <sup>8</sup>	Coliforms are bacteria that indicate that other, potentially harmful bacteria may be present. See fecal coliforms and <i>E. coli</i>	Naturally present in the environment	zero
Total Trihalomethanes (TTHMs)	0.080	Liver, kidney, or central nervous system problems; increased risk of cancer	Byproduct of drinking water disinfection	n/aº
Toxaphene	0.003	Kidney, liver, or thyroid problems; increased risk of cancer	Runoff/leaching from insecticide used on cotton and cattle	zero
() 2,4,5-TP (Silvex)	0.05	Liver problems	Residue of banned herbicide	0.05
1,2,4- Trichlorobenzene	0.07	Changes in adrenal glands	Discharge from textile finishing factories	0.07

LEGEND

+ DISINFECTANT





MICROORGANISM



RADIONUCLIDES

National Primary Drinking Water Regulations

EPA 816-F-09-004 | MAY 2009

Contaminant	MCL or TT <sup>1</sup> (mg/L) <sup>2</sup>	Potential health effects from long-term <sup>3</sup> exposure above the MCL	Common sources of contaminant in drinking water	Public Health Goal (mg/L) <sup>2</sup>
1,1,1- Trichloroethane	0.2	Liver, nervous system, or circulatory problems	Discharge from metal degreasing sites and other factories	0.2
1,1,2- Trichloroethane	0.005	Liver, kidney, or immune system problems	Discharge from industrial chemical factories	0.003
Trichloroethylene	0.005	Liver problems; increased risk of cancer	Discharge from metal degreasing sites and other factories	zero
Turbidity	Π7	Turbidity is a measure of the cloudiness of water. It is used to indicate water quality and filtration effectiveness (e.g., whether disease- causing organisms are present). Higher turbidity levels are often associated with higher levels of disease-causing microorganisms such as viruses, parasites, and some bacteria. These organisms can cause short term symptoms such as nausea, cramps, diarrhea, and associated headaches.	Soil runoff	n/a
Uranium	30µg/L	Increased risk of cancer, kidney toxicity	Erosion of natural deposits	zero
Vinyl chloride	0.002	Increased risk of cancer	Leaching from PVC pipes; discharge from plastic factories	zero
Viruses (enteric)	Π7	Short-term exposure: Castrointestinal illness (e.g., diarrhea, vomiting, cramps)	Human and animal fecal waste	zero
Xylenes (total)	10	Nervous system damage	Discharge from petroleum factories; discharge from chemical factories	10
LEGEND DISINFECTANT DISINFECTION DISINFECTION BYPRODUCT INORGANIC CHEMICAL MICROORGANISM ORGANIC CHEMICAL ORGANIC				

#### NOTES

1 Definitions

- Maximum Contaminant Level Goal (MCLG): The level of a contaminant in drinking water below which there is no known or expected risk to health. MCLCs allow for a margin of safety and are non-enforceable public health goals.
- Maximum Contaminant Level (MCL): The highest level of a contaminant that is allowed in drinking water. MCLs are set as close to MCLGs as feasible using the best available treatment technology and taking cost into consideration. MCLs are enforceable standards.
- Maximum Residual Disinfectant Level Goal (MRDLG): The level of a drinking water disinfectant below which there is no known or expected risk to health. MRDLGs do not reflect the benefits of the use of disinfectants to control microbial contaminants.
- Maximum Residual Disinfectant Level (MRDL): The highest level of a disinfectant allowed in drinking water. There is convincing evidence that addition of a disinfectant is necessary for control of microbial contaminants.
- Treatment Technique (TT): A required process intended to reduce the level of a contaminant in drinking water.

2 Units are in milligrams per liter (mg/L) unless otherwise noted. Milligrams per liter are equivalent to parts per million (ppm).

- 3 Health effects are from long-term exposure unless specified as short-term exposure.
- 4 Each water system must certify annually, in writing, to the state (using third-party or manufacturers certification) that when it uses acrylamide and/or epichlorohydrin to treat water, the combination (or product) of dose and monomer level does not exceed the levels specified, as follows: Acrylamide = 0.05 percent dosed at 1 mg/L (or equivalent); Epichlorohydrin = 0.01 percent dosed at 20 mg/L (or equivalent).
- 5 Lead and copper are regulated by a Treatment Technique that requires systems to control the corrosiveness of their water. If more than 10 percent of tap water samples exceed the action level, water systems must take additional steps. For copper, the action level is 1.3 mg/L, and for lead is 0.015 mg/L.
- 6 A routine sample that is fecal coliform-positive or E. coli-positive triggers repeat samplesif any repeat sample is total coliform-positive, the system has an acute MCL violation. A routine sample that is total coliform-positive and fecal coliform-negative or E. colinegative triggers repeat samples--if any repeat sample is fecal coliform-positive or E. coli-positive, the system has an acute MCL violation. See also Total Coliforms.

7 EPA's surface water treatment rules require systems using surface water or ground water under the direct influence of surface water to (1) disinfect their water, and (2) filter their water or meet criteria for avoiding filtration so that the following contaminants are controlled at the following levels:

Cryptosporidium: 99 percent removal for systems that filter. Unfiltered systems are required to include Cryptosporidium in their existing watershed control provisions.

- Ciardia lamblia: 99.9 percent removal/inactivation
- Viruses: 99.9 percent removal/inactivation
- Legionella: No limit, but EPA believes that if Giardia and viruses are removed/ inactivated, according to the treatment techniques in the surface water treatment rule, Legionella will also be controlled.
- Turbidity: For systems that use conventional or direct filtration, at no time can turbidity (cloudiness of water) go higher than 1 nephelometric turbidity unit (NTU), and samples for turbidity must be less than or equal to 0.3 NTU in at least 95 percent of the samples in any month. Systems that use filtration other than the conventional or direct filtration must follow state limits, which must include turbidity at no time exceeding 5 NTU.
   HPC: No more than 500 bacterial colonies per milliliter
- Long Term 1 Enhanced Surface Water Treatment: Surface water systems or ground water systems under the direct influence of surface water serving fewer than 10,000 people must comply with the applicable Long Term 1 Enhanced Surface Water Treatment Rule provisions (e.g. turbidity standards, individual filter monitoring, *Cryptosporidium* removal requirements, updated watershed control requirements for unfiltered systems).
- Long Term 2 Enhanced Surface Water Treatment: This rule applies to all surface water systems or ground water systems under the direct influence of surface water. The rule targets additional *Cryptosporidium* treatment requirements for higher risk systems and includes provisions to reduce risks from uncovered finished water storages facilities and to ensure that the systems maintain microbial protection as they take steps to reduce the formation of disinfection byproducts. (Monitoring start dates are staggered by system size. The largest systems (serving at least 100,000 people) will begin monitoring in October 2006 and the smallest systems (serving fewer than 10,000 people) will not begin monitoring until October 2008. After completing monitoring and determining their treatment bin, systems generally have three years to comply with any additional treatment requirements.)
- Filter Backwash Recycling: The Filter Backwash Recycling Rule requires systems that recycle to return specific recycle flows through all processes of the system's existing conventional or direct filtration system or at an alternate location approved by the state
- 8 No more than 5.0 percent samples total coliform-positive in a month. (For water systems that collect fewer than 40 routine samples per month, no more than one sample can be total coliform-positive per month.) Every sample that has total coliform must be analyzed for either fecal coliforms or E. coli. If two consecutive TC-positive samples, and one is also positive for E. coli or fecal coliforms, system has an acute MCL violation.
- 9 Although there is no collective MCLG for this contaminant group, there are individual MCLGs for some of the individual contaminants:
   Haloacetic acids: dichloroacetic acid (zero); trichloroacetic acid (0.3 mg/L)
  - Haloacetic acids: dichloroacetic acid (zero); trichloroacetic acid (0.3 mg// Trihalomethanes: bromodichloromethane (zero); bromoform (zero); dibromochloromethane (0.06 mg/L)

#### NATIONAL SECONDARY DRINKING WATER REGULATION

National Secondary Drinking Water Regulations are non-enforceable guidelines regarding contaminants that may cause cosmetic effects (such as skin or tooth discoloration) or aesthetic effects (such as taste, odor, or color) in drinking water. EPA recommends secondary standards to water systems but does not require systems to comply. However, some states may choose to adopt them as enforceable standards.

Contaminant	Secondary Maximum Contaminant Level
Aluminum	0.05 to 0.2 mg/L
Chloride	250 mg/L
Color	15 (color units)
Copper	1.0 mg/L
Corrosivity	Noncorrosive
Fluoride	2.0 mg/L
Foaming Agents	0.5 mg/L
Iron	0.3 mg/L
Manganese	0.05 mg/L
Odor	3 threshold odor number
рН	6.5-8.5
Silver	0.10 mg/L
Sulfate	250 mg/L
Total Dissolved Solids	500 mg/L
Zinc	5 mg/L

FOR MORE INFORMATION ON EPA'S SAFE DRINKING WATER:



visit: epa.gov/safewater



call: (800) 426-4791

#### **ADDITIONAL INFORMATION:**

To order additional posters or other ground water and drinking water publications, please contact the National Service Center for Environmental Publications at: **(800) 490-9198**, or email: **nscep@bps-Imit.com**.



DEPARTMENT OF NATURAL RESOURCES

NR 140.03

### Chapter NR 140 GROUNDWATER QUALITY

Subchapter I - NR 140.01 NR 140.02 NR 140.03 NR 140.05 Subchapter II NR 140.10 NR 140.12	<ul> <li>General</li> <li>Purpose.</li> <li>Regulatory framework.</li> <li>Applicability.</li> <li>Definitions.</li> <li>Groundwater Quality Standards</li> <li>Public health related groundwater standards.</li> <li>Public welfare related groundwater standards.</li> </ul>	NR 140.16 Subchapter II NR 140.20 NR 140.22 NR 140.24 NR 140.26 NR 140.27	Monitoring and laboratory data requirements. II — Evaluation and Response Procedures Indicator parameter groundwater standards. Point of standards application for design and compliance. Responses when a preventive action limit is attained or exceeded. Responses when an enforcement standard is attained or exceeded at a location other than a point of standards application
NR 140.10 NR 140.12 NR 140.14	Public welfare related groundwater standards. Statistical procedures.	NR 140.27	a location other than a point of standards application. Exemptions.

Subchapter I — General

**NR 140.01 Purpose.** The purpose of this chapter is to establish groundwater quality standards for substances detected in or having a reasonable probability of entering the groundwater resources of the state; to specify scientifically valid procedures for determining if a numerical standard has been attained or exceeded; to specify procedures for establishing points of standards application, and for evaluating groundwater monitoring data; to establish ranges of responses the department may require if a groundwater standard is attained or exceeded; and to provide for exemptions for facilities, practices and activities regulated by the department.

History: Cr. Register, September, 1985, No. 357, eff. 10-1-85.

**NR 140.02 Regulatory framework. (1)** This chapter supplements the regulatory authority elsewhere in the statutes and administrative rules. The department will continue to exercise the powers and duties in those regulatory programs, consistent with the enforcement standards and preventive action limits for substances in groundwater under this chapter. This chapter provides guidelines and procedures for the exercise of regulatory authority which is established elsewhere in the statutes and administrative rules, and does not create independent regulatory authority.

(2) The department may adopt regulations which establish specific design and management criteria for regulated facilities or activities, if the regulations will ensure that the regulated facilities and activities will not cause the concentration of a substance in groundwater affected by the facilities or activities to exceed the enforcement standards and preventive action limits under this chapter at a point of standards application. The department may adopt more stringent regulations under authority elsewhere in the statutes based on the best currently available technology for regulated activities and practices which ensure a greater degree of groundwater protection or when necessary to comply with state or federal laws.

(3) Preventive action limits serve to inform the department of potential groundwater contamination problems, establish the level of groundwater contamination at which the department is required to commence efforts to control the contamination and provide a basis for design and management practice criteria in administrative rules. Preventive action limits are applicable both to controlling new releases of contamination as well as to restoring groundwater quality contaminated by past releases of contaminants. Although a preventive action limit is not intended to always require remedial action, activities affecting groundwater must be regulated to minimize the level of substances to the extent technically and economically feasible, and to maintain compliance with the preventive action limits unless compliance with the preventive action limits unless compliance with the preventive action limits and economically feasible.

(4) The department may take any actions within the context of regulatory programs established in statutes or rules outside of this

chapter, if those actions are necessary to protect public health and welfare or prevent a significant damaging effect on groundwater or surface water quality for present or future consumptive or nonconsumptive uses, whether or not an enforcement standard and preventive action limit for a substance have been adopted under this chapter. Nothing in this chapter authorizes an impact on groundwater quality which would cause surface water quality standards contained in chs. NR 102 to 105 to be attained or exceeded.

History: Cr. Register, January, 1992, No. 433, eff. 2–1–92; reprinted to restore dropped copy, Register, March, 1992, No. 435.

**NR 140.03** Applicability. This subchapter and subch. II apply to all facilities, practices, and activities which may affect groundwater quality and which are regulated under chs. 85, 93, 94, 101, 145, 281, 283, 287, 289, 291, and 292, Stats., by the department of agriculture, trade and consumer protection, the department of safety and professional services, the department of transportation, or the department of natural resources, as well as to facilities, practices, and activities which may affect groundwater quality which are regulated by other regulatory agencies. Health-related enforcement standards adopted in s. NR 140.10 also apply to bottled drinking water manufactured, bottled, sold, or distributed in this state as required by s. 97.34 (2) (b), Stats., and to determining eligibility for the well compensation program under s. 281.75, Stats. Subchapter III applies to all facilities, practices, and activities which may affect groundwater quality and which are regulated by the department under ch. 281, 283, 287, 289, 291, 292, 295, or 299, Stats. This chapter applies to ferrous metallic mining operations and mining sites, including mining waste sites, as defined in s. 295.41 (31), Stats., but only to the extent that it does not conflict with subch. III of ch. 295, Stats. Groundwater quality standards, consisting of enforcement standards and preventive action limits contained in ss. NR 140.10 and 140.12, and preventive action limits for indicator parameters identified under s. NR 140.20 (2), apply to ferrous metallic mining operations and mining sites, as defined in s. 295.41 (31), Stats., including mining waste sites, regulated under subch. III of ch. 295, Stats. This chapter does not apply to any facilities, practices, or activities on a nonferrous metallic mining prospecting site or mining site regulated under ch. 293, Stats., because those facilities, practices, and activities are subject to the groundwater quality requirements of chs. NR 131, 132, and 182. The department may promulgate new rules or amend rules governing facilities, practices or activities regulated under ch. 293, Stats., if the department determines that the amendment or promulgation of rules is necessary to protect public health, safety, or welfare. The requirements of this chapter are in addition to the requirements of any other statutes and rules, except as provided in s. 295.645 (9), Stats.

Note: The groundwater standards in this chapter do not replace the maximum contaminant levels applicable to public water systems contained in ch. NR 809. Drinking water maximum contaminant levels and health advisory levels may take into account such factors as treatment costs and feasibility for public water systems.

**History:** Cr. Register, September, 1985, No. 357, eff. 10–1–85; am. Register, December, 1998, No. 516, eff. 1–1–99; correction made under s. 13.93 (2m) (b) 7.,

 Published under s. 35.93, Stats. Updated on the first day of each month. Entire code is always current. The Register date on each page

 is the date the chapter was last published.

 Register February 2017 No. 734

Stats., Register, March, 2000, No. 531; correction made under s. 13.92 (4) (b) 6., Stats., Register January 2012 No. 673; CR 13–057: am. Register July 2015 No. 715, eff. 8–1–15.

**NR 140.05 Definitions. (1)** "Accuracy" means the closeness of a measured value to its generally accepted value or its value based upon an accepted reference standard.

(1m) "Alternative concentration limit" means the concentration of a substance in groundwater established by the department for a site to replace a preventive action limit or enforcement standard or both, from Table 1 or 2, when an exemption is granted in accordance with s. NR 140.28.

(1s) "Approval" means written acceptance by the department of a plan, report or other document that has been submitted to the department for review.

(1u) "Aquifer storage recovery" or "ASR" means placement of treated drinking water underground through a well for the purpose of storing and later recovering the water through the same well for potable use.

**Note:** Underground placement of water for the purpose of restoring an aquifer is not included in the definition of "aquifer storage recovery" or "ASR".

(1w) "ASR displacement zone" means the 3-dimensional subsurface region surrounding an aquifer storage recovery well into which treated drinking water is placed for storage and later recovery.

(1y) "ASR system" means all of the ASR wells, ASR monitoring wells and related appurtenances within a municipal well system and any interconnected public water system served by the municipal water system.

(2) "Attain or exceed" means that the concentration of a substance is determined to be equal to or greater than the preventive action limit or enforcement standard for that substance.

(3) "Background water quality" or "background concentration" means groundwater quality at or near a facility, practice or activity which has not been affected by that facility, practice or activity.

(4) "Certified laboratory" means a laboratory which performs tests for hire in connection with a covered program and which receives certification under s. 299.11 (7), Stats., or receives reciprocal recognition under s. 299.11 (5), Stats.

(5) "Department" means the department of natural resources.

(6) "Design management zone" means a 3-dimensional boundary surrounding each regulated facility, practice or activity established under s. NR 140.22 (3).

(7) "Enforcement standard" means a numerical value expressing the concentration of a substance in groundwater which is adopted under s. 160.07, Stats., and s. NR 140.10 or s. 160.09, Stats., and s. NR 140.12.

(8) "Facility, practice or activity" means any source or potential source of a substance which is detected in or has a reasonable probability of entering the groundwater resources of the state.

(9) "Groundwater" means any of the waters of the state, as defined in s. 281.01 (18), Stats., occurring in a saturated subsurface geological formation of rock or soil.

(10) "Indicator parameter" means a substance for which a preventive action limit has been established under s. NR 140.20, which is used to indicate the potential for a preventive action limit established under s. NR 140.10 or 140.12 to be attained or exceeded and for which an enforcement standard has not been established under s. NR 140.10 or 140.12.

(10e) "Infiltration" means the underground emplacement of substances or remedial material, or both, into an excavation that is wider than deep so as to percolate or move through unsaturated material to groundwater.

(10s) "Injection" means the underground emplacement of substances or remedial material, or both, into a borehole or other excavation that is deeper than wide so as to percolate or move through unsaturated material to groundwater or to enter groundwater directly.

(11) "Land disposal system" means a facility for disposing of liquid wastes consisting of:

(a) An absorption or seepage pond system,

(b) A ridge and furrow system;

(c) A spray irrigation system,

(d) An overland flow system,

(e) A subsurface field absorption system,

(f) A land spreading system, or

(g) Any other land area receiving liquid waste discharges.

(12) "Limit of detection" means the lowest concentration level that can be determined to be statistically different from a blank.

(13) "Limit of quantitation" means the level above which quantitative results may be obtained with a specified degree of confidence.

Note: The limit of quantitation is 10/3 or 3.333 times the limit of detection.

(14) "Monitoring" means all procedures used to collect data on groundwater, surface water or soils.

(14m) "Natural attenuation" means the reduction in the concentration and mass of a substance and its breakdown products in groundwater, due to naturally occurring physical, chemical, and biological processes without human intervention or enhancement. These processes include, but are not limited to, dispersion, diffusion, sorption and retardation, and degradation processes such as biodegradation, abiotic degradation and radioactive decay.

(15) "Point of standards application" means the specific location, depth or distance from a facility, activity or practice at which the concentration of a substance in groundwater is measured for purposes of determining whether a preventive action limit or an enforcement standard has been attained or exceeded.

(16) "Precision" means the closeness of repeated measurements of the same parameter within a sample.

(17) "Preventive action limit" means a numerical value expressing the concentration of a substance in groundwater which is adopted under s. 160.15, Stats., and s. NR 140.10, 140.12 or 140.20.

(18) "Property boundary" means the boundary of the total contiguous parcel of land owned or leased by a common owner or lessor, regardless of whether public or private roads run through the parcel.

(19) "Registered laboratory" means a laboratory which is registered under s. 299.11 (8), Stats., or receives reciprocal recognition under s. 299.11 (5), Stats.

(20) "Regulatory agency" means the department of agriculture, trade and consumer protection, the department of safety and professional services, the department of transportation, the department of natural resources and other state agencies which regulate activities, facilities or practices which are related to substances which have been detected in or have reasonable probability of entering the groundwater resources of the state.

(20h) "Remedial action" means a response which is taken to achieve compliance with groundwater quality standards established under this chapter. This term includes, but is not limited to, actions designed to prevent or minimize the further discharge or release of substances to groundwater and actions designed to renovate or restore groundwater quality.

(20k) "Remedial material" means any solid, liquid, semisolid or gaseous material, either naturally occurring or manmade, in its original form or as a metabolite or degradation product, or naturally occurring non-pathogenic biological organisms which have not undergone human induced genetic alteration, which enhances the restoration of soil or groundwater quality, or both.

Published under s. 35.93, Stats. Updated on the first day of each month. Entire code is always current. The Register date on each page y 2017 No. 734 is the date the chapter was last published. (20m) "Response" means any action taken to respond to an attainment or exceedance of a preventive action limit or enforcement standard as required by s. NR 140.24 or 140.26.

Note: A response may include a remedial action.

(20s) "Specified substance" means one of the following: chloroform, bromodichloromethane, dibromochloromethane or bromoform.

(21) "Substance" means any solid, liquid, semisolid, dissolved solid or gaseous material, naturally occurring or manmade chemical, parameter for measurement of water quality or biological organism which, in its original form, or as a metabolite or a degradation or waste product, may decrease the quality of groundwater.

(22) "Wastewater and sludge storage or treatment lagoon" means a natural or man-made containment structure, constructed primarily of earthen materials for the treatment or storage of wastewater or sludge, which is not a land disposal system. Wastewater of studge, which is not a fand disposal system. **History:** Cr. Register, September, 1985, No. 357, eff. 10–1–85; cr. (1m), am. (7), (17) and (18), Register, October, 1988, No. 394, eff. 11–1–88; am. (6), cr. (20h) and (20m), Register, March, 1994, No. 459, eff. 4–1–94; cr. (1s), (10e), (10s), (20k), r. and recr. (12), (13), Register, August, 1995, No. 476, eff. 9–1–95; cr. (14m), Register, October, 1996, No. 490, eff. 11–1–96; am. (20), Register, December, 1998, No. 516, eff. 1–1–99; correction in (9) made unders. 13.93 (2m) (b) 7., Stats., Register, April, 2001, No. 544; CR 02–134; cr. (1u), (1w), (1y) and (20s) Register June 2003 No. 570, eff. 7–1–03; accreation in (20) meda unders. 13.02 (4) (b) 6. Stats. Register Representation (10) for the second seco

#### Subchapter II — Groundwater Quality Standards

NR 140.10 Public health related groundwater standards. The groundwater quality standards for substances of public health concern are listed in Table 1.

Note: For all substances that have carcinogenic, mutagenic or teratogenic properties or interactive effects, the preventive action limit is 10% of the enforcement standard. The preventive action limit is 20% of the enforcement standard for all other substances that are of public health concern. Enforcement standards and preventive action limits for additional substances will be added to Table I as recommendations are developed pursuant to ss. 160.07, 160.13 and 160.15, Stats. eff. 7–1–03; correction in (20) made under s. 13.92 (4) (b) 6., Stats., Register January 2012 No. 673.

Table 1
Public Health Groundwater Quality Standards

Substance <sup>1</sup>	Enforcement Standard (micrograms per liter – except as noted)	Preventive Action Limit (micrograms per liter – except as noted)
Acetochlor	7	0.7
Acetochlor ethane sulfonic acid + oxanilic acid (Acetochlor – ESA + OXA)	230	46
Acetone	9 mg/1	1.8 mg/1
Alachlor	2	0.2
Alachlor ethane sulfonic acid (Alachlor – ESA)	20	4
Aldicarb	10	2
Aluminum	200	40
Ammonia (as N)	9.7 mg/l	0.97 mg/l
Antimony	6	1.2
Anthracene	3000	600
Arsenic	10	1
Asbestos	7 million fibers per liter (MFL)	0.7 MFL
Atrazine, total chlorinated residues	3 <sup>2</sup>	$0.3^{2}$
Bacteria, Total Coliform	$0^{3}$	$0^{3}$
Barium	2 milligrams/liter (mg/l)	0.4 mg/l
Bentazon	300	60
Benzene	5	0.5
Benzo(b)fluoranthene	0.2	0.02
Benzo(a)pyrene	0.2	0.02
Beryllium	4	0.4
Boron	1000	200
Bromodichloromethane	0.6	0.06
Bromoform	4.4	0.44
Bromomethane	10	1
Butylate	400	80
Cadmium	5	0.5
Carbaryl	40	4
Carbofuran	40	8
Carbon disulfide	1000	200
Carbon tetrachloride	5	0.5
Chloramben	150	30
Chlordane	2	0.2
Chlorodifluoromethane	7 mg/l	0.7 mg/l
Chloroethane	400	80
Chloroform	6	0.6
Chlorpyrifos	2	0.4
Chloromethane	30	3
Chromium (total)	100	10
Chrysene	0.2	0.02

Published under s. 35.93, Stats. Updated on the first day of each month. Entire code is always current. The Register date on each page

#### DEPARTMENT OF NATURAL RESOURCES

NR 140.10

Enforcement Standard (micrograms Preventive Action Limit (micrograms			
Substance <sup>1</sup>	per liter – except as noted)	per liter – except as noted)	
Cobalt	40	8	
Copper	1300	130	
Cyanazine	1	0.1	
Cyanide, free <sup>4</sup>	200	40	
Dacthal	70	14	
1,2-Dibromoethane (EDB)	0.05	0.005	
Dibromochloromethane	60	6	
1,2-Dibromo-3-chloropropane (DBCP)	0.2	0.02	
Dibutyl phthalate	1000	100	
Dicamba	300	60	
1,2-Dichlorobenzene	600	60	
1,3-Dichlorobenzene	600	120	
1,4–Dichlorobenzene	75	15	
Dichlorodifluoromethane	1000	200	
1.1–Dichloroethane	850	85	
1 2-Dichloroethane	5	0.5	
1 1–Dichloroethylene	7	0.7	
1.2-Dichloroethylene (cis)	70	7	
1.2 Dichloroethylene (trans)	100	20	
2 4-Dichlorophenoxyacetic Acid (2 4-D)	70	7	
1.2-Dichloropropage	5	,	
1.2 Dichloropropane (sis/trans)	0.4	0.5	
Di (2. sthulhauul) #thalata	6.4	0.64	
Di (2-etitymexyl) philalate	50	5	
Dimethenta	30	5	
2.4. Diritartaluare	2	0.4	
2,4-Dinitrotoluene	0.05	0.005	
2,6-Dinitrotoluene	0.05	0.005	
Dinitrotoluene, Iotal Residues <sup>5</sup>	0.05	0.005	
	7	1.4	
1,4-Dioxane	3	0.3	
Dioxin (2, 3, 7, 8–1CDD)	0.00003	0.000003	
Endrin	2	0.4	
EPIC	250	50	
Ethylbenzene	700	140	
Ethyl ether	1000	100	
Ethylene glycol	14 mg/l	2.8 mg/l	
Fluoranthene	400	80	
Fluorene	400	80	
Fluoride	4 mg/l	0.8 mg/l	
Fluorotrichloromethane	3490	698	
Formaldehyde	1000	100	
Heptachlor	0.4	0.04	
Heptachlor epoxide	0.2	0.02	
Hexachlorobenzene	1	0.1	
<i>N</i> -Hexane	600	120	
Hydrogen sulfide	30	6	
Lead	15	1.5	
Lindane	0.2	0.02	
Manganese	300	60	
Mercury	2	0.2	

#### Table 1 – Continued Public Health Groundwater Quality Standards

 Published under s. 35.93, Stats. Updated on the first day of each month. Entire code is always current. The Register date on each page

 is the date the chapter was last published.

 Register February 2017 No. 734

NR 140.10

Public Health Groundwater Quality Standards			
Substance <sup>1</sup>	Enforcement Standard (micrograms per liter – except as noted)	Preventive Action Limit (micrograms per liter – except as noted)	
Methanol	5000	1000	
Methoxychlor	40	4	
Methylene chloride	5	0.5	
Methyl ethyl ketone (MEK)	4 mg/l	0.8 mg/l	
Methyl isobutyl ketone (MIBK)	500	50	
Methyl tert-butyl ether (MTBE)	60	12	
Metolachlor/s-Metolachlor	100	10	
Metolachlor ethane sulfonic acid + oxanilic acid (Metolachlor – ESA + OXA)	1.3 mg/l	0.26 mg/l	
Metribuzin	70	14	
Molybdenum	40	8	
Monochlorobenzene	100	20	
Naphthalene	100	10	
Nickel	100	20	
Nitrate (as N)	10 mg/l	2 mg/l	
Nitrate + Nitrite (as N)	10 mg/l	2 mg/l	
Nitrite (as N)	1  mg/1	0.2 mg/l	
<i>N</i> –Nitrosodiphenylamine	7	0.7	
Pentachlorophenol (PCP)	1	0.1	
Perchlorate	1	0.1	
Phenol	2 mg/l	0.4 mg/l	
Picloram	500	100	
Polychlorinated biphenyls (PCBs)	0.03	0.003	
Prometon	100	20	
Propazine	10	2	
Pyrene	250	50	
Pyridine	10	2	
Selenium	50	10	
Silver	50	10	
Simazine	4	0.4	
Styrene	100	10	
Tertiary Butyl Alcohol (TBA)	12	12	
1 1 1 2–Tetrachloroethane	70	7	
1 1 2 2–Tetrachloroethane	0.2	0.02	
Tetrachloroethylene	5	0.5	
Tetrahydrofuran	50	10	
Thallium	2	0.4	
Toluene	800	160	
Toyaphene	3	0.3	
1.2.4-Trichlorobenzene	70	14	
1,2,4 Themorobenzene	200	40	
1,1,2-Trichloroethane	200	40	
Trichloroethylene (TCE)	5	0.5	
2,4,5–Trichlorophenoxy–propionic acid (2,4,5–TP)	50	5	
1 2 3-Trichloropropage	60	12	
Trifluralin	7.5	0.75	
Trimethylbenzenes	480	96	
(12.4 - and 12.5 - combined)	100	20	
(1,2,7) and $(1,3,3)$ combined)	30	6	
V (11) (N 11) (11)			

Table 1 – Continued Public Health Groundwater Quality Standard

Public Health Groundwater Quality Standards		
Substance <sup>1</sup>	Enforcement Standard (micrograms per liter – except as noted)	Preventive Action Limit (micrograms per liter – except as noted)
Vinyl chloride	0.2	0.02
Xylene <sup>6</sup>	2 mg/l	0.4 mg/l

Table 1 - Continued

<sup>1</sup> Appendix I contains Chemical Abstract Service (CAS) registry numbers, common synonyms and trade names for most substances listed in Table 1.

<sup>2</sup> Total chlorinated atrazine residues includes parent compound and the following metabolites of health concern: 2-chloro-4-amino-6-isopropylamino-s-triazine (formerly deethylatrazine), 2-chloro-4-amino-6-ethylamino-s-triazine (formerly deisopropylatrazine) and 2-chloro-4,6-diamino-s-triazine (formerly diaminoatrazine).

<sup>3</sup> Total coliform bacteria may not be present in any 100 ml sample using either the membrane filter (MF) technique, the presence–absence (P–A) coliform test, the minimal medium ONPG–MUG (MMO–MUG) test or not present in any 10 ml portion of the 10-tube multiple tube fermentation (MTF) technique.

<sup>4</sup> "Cyanide, free" refers to the simple cyanides (HCN, CN<sup>-</sup>) and /or readily dissociable metal-cyanide complexes. Free cyanide is regulatorily equivalent to cyanide quantified by approved analytical methods for "amenable cyanide" or "available cyanide".

<sup>5</sup> Dinitrotoluene, Total Residues includes the dinitrotoluene (DNT) isomers: 2,3–DNT, 2,4–DNT, 2,5–DNT, 2,6–DNT, 3,4–DNT and 3,5–DNT.

<sup>6</sup> Xylene includes meta-, ortho-, and para-xylene combined.

**History:** Cr. Register, September, 1985, No. 357, eff. 10–1–85; am. table 1, Register, October, 1988, No. 394, eff. 11–1–88; am. table 1, Register, September, 1990, No. 417, eff. 10–1–90; am. Register, January, 1992, No. 433, eff. 2–1–92; am. Table 1, Register, March, 1994, No. 459, eff. 4–1–94; am. Table 1, Register, August, 1995, No. 476, eff. 9–1–95; am. Table 1, Register, December, 1998, No. 516, eff. 1–1–99; am. Table 1, boron, Register, December, 1998, No. 516, eff. 12–31–99; am. Table 1, Register, Bernard, 2000, No. 531, eff. 4–1–00; CR 03–063; am Table 1, Register February 2004 No. 578, eff. 3–1–04; CR 02–095; am. Table 1, Register November 2006 No. 611, eff. 12–1–06; reprinted to correct errors in Table 1, Register January 2007 No. 613; CR 07–034; am. Table 1 Register January 2008 No. 625, eff. 2–1–08; CR 09–102; am. Table 1 Register December 2010 No. 660, eff. 1–1–11.

**NR 140.12 Public welfare related groundwater standards.** The groundwater quality standards for substances of public welfare concern are listed in Table 2.

Note: For each substance of public welfare concern, the preventive action limit is 50% of the established enforcement standard.

Table 2
Public Welfare Groundwater Quality Standards

Substance	Enforcement Standard (milligrams	Preventive Action Limit (milligrams
Chloride	250	125
Color	15 color units	7.5 color units
Foaming agents MBAS (Methylene–Blue Active Substances)	0.5	0.25
Iron	0.3	0.15
Manganese	0.05	0.025
Odor	3	1.5
	(Threshold Odor No.)	(Threshold Odor No.)
Sulfate	250	125
Zinc	5	2.5

History: Cr. Register, September, 1985, No. 357, eff. 10–1–85; am. table 2, Register, October, 1990, No. 418, eff. 11–1–90; am. Table 2, Register, March, 1994, No. 459, eff. 4–1–94.

**NR 140.14 Statistical procedures. (1)** If a preventive action limit or an enforcement standard for a substance listed in Table 1 or 2, an alternative concentration limit issued in accordance with s. NR 140.28 or a preventive action limit for an indicator parameter established according to s. NR 140.20 (2) is attained or exceeded at a point of standards application:

(a) The owner or operator of the facility, practice or activity at which a standard is attained or exceeded shall notify the appropriate regulatory agency that a standard has been attained or exceeded; and

(b) The regulatory agency shall require a response in accordance with the rules promulgated under s. 160.21, Stats. No response shall be required if it is demonstrated to the satisfaction of the appropriate regulatory agency that a scientifically valid determination cannot be made that the preventive action limit or enforcement standard for a substance in Table 1 or 2 has been attained or exceeded based on consideration of sampling procedures or laboratory precision and accuracy, at a significance level of 0.05.

(2) The regulatory agency shall use one or more valid statistical procedures to determine if a change in the concentration of a substance has occurred. A significance level of 0.05 shall be used for all tests.

(3) In addition to sub. (2), the following applies when a preventive action limit or enforcement standard is equal to or less than the limit of quantitation:

(a) If a substance is not detected in a sample, the regulatory agency may not consider the preventive action limit or enforcement standard to have been attained or exceeded.

(b) If the preventive action limit or enforcement standard is less than the limit of detection, and the concentration of a substance is reported between the limit of detection and the limit of quantitation, the regulatory agency shall consider the preventive action limit or enforcement standard to be attained or exceeded only if:

1. The substance has been analytically confirmed to be present in the same sample using an equivalently sensitive analytical method or the same analytical method, and

2. The substance has been statistically confirmed to be present above the preventive action limit or enforcement standard, determined by an appropriate statistical test with sufficient samples at a significance level of 0.05.

(c) If the preventive action limit or enforcement standard is between the limit of detection and the limit of quantitation, the regulatory agency shall consider the preventive action limit or

Published under s. 35.93, Stats. Updated on the first day of each month. Entire code is always current. The Register date on each page is the date the chapter was last published.
Register February 2017 No. 734

enforcement standard to be attained or exceeded if the concentration of a substance is reported at or above the limit of quantitation.

History: Cr. Register, September, 1985, No. 357, eff. 10-1-85; am. (1) (intro.) and (b), r. and recr. (2), Register, October, 1988, No. 394, eff. 11-1-88; am. (1) (b), (2) and (3) (b), Register, September, 1990, No. 417, eff. 10-1-90; am. (1) (b), Register, March, 1994, No. 459, eff. 4-1-94; r. and recr. (3) (intro.), (a), (b), renum. (3) (c) to be 140.16 (5) and am., Register, August, 1995, No. 476, eff. 9-1-95.

NR 140.16 Monitoring and laboratory data require**ments.** (1) (a) All groundwater quality samples collected to determine compliance with ch. 160, Stats., shall comply with this section except as noted.

(b) Groundwater sampling requirements. All groundwater quality samples shall be collected and handled in accordance with procedures specified by the applicable regulatory agency or, where no sampling procedures are specified by that agency, in accordance with the sampling procedures referenced in par. (c). The sampling procedures specified by a regulatory agency may include requirements for field filtration.

(c) Department groundwater sampling procedures. 1. If sampling procedures are not specified by the applicable regulatory agency pursuant to par. (b), all groundwater quality samples shall be collected and handled in accordance with the sampling procedures contained in the following publications:

Groundwater Sampling Desk Reference. Wisconsin Department of Natural Resources, PUBL-DG-037-96, September. 1996.

b. Groundwater Sampling Field Manual. Wisconsin Department of Natural Resources, PUBL-DG-038-96, September, 1996

Note: Copies of these publications may be purchased from:

Wisconsin Department of Administration

Document Sales Unit

4622 University Avenue Madison, WI 53705-2156

These publications are available for inspection at the offices of the department, the secretary of state and the legislative reference bureau.

2. Where no procedure for collecting a particular groundwater quality sample is specified by the appropriate regulatory agency or in the publications referenced in subd. 1., other published scientifically valid groundwater sampling procedures may be used.

(d) Laboratory requirements. All groundwater quality samples, except samples collected for total coliform bacteria analysis and field analyses for pH, specific conductance and temperature, shall be analyzed in accordance with provisions of ch. NR 149 by a laboratory certified or registered under ch. NR 149. Samples for total coliform bacteria analysis shall be analyzed by the state laboratory of hygiene or at a laboratory approved or certified by the department of agriculture, trade and consumer protection.

Note: Refer to s. NR 149.46 for sample preservation procedures and holding times

(e) Data submittal. The results of the analysis of groundwater quality samples shall be submitted to the department and any applicable regulatory agency. Except as provided in s. NR 205.07 (3) (c) for wastewater permittees, this section does not require the submission of groundwater monitoring data which is collected voluntarily and is not required to be collected to determine compliance with this chapter or another rule or statute.

(2) The laboratory shall select the analytical methodology which:

(a) Is specified in rules or approved by the regulatory agency, and

(b) Is appropriate for the concentration of the sample, and

(c) Is one of the following:

1. Has a limit of detection and limit of quantitation below the preventive action limit, or

2. Produces the lowest available limit of detection and limit of quantitation if the limit of detection and limit of quantitation are above the preventive action limit.

(3) If the owner or operator of a facility, practice or activity believes that a sample result does not represent groundwater quality in the vicinity of the facility, practice or activity, the owner or operator shall resample the appropriate well or wells to obtain a representative sample at the earliest possible time. All sample results shall be submitted to the department and the appropriate regulatory agency with an explanation of why the owner or operator believes that all or some of the results are invalid.

(4) The department may reject groundwater quality data that does not meet the requirements of the approved or designated analytical methods.

(5) The owner or operator of the facility, practice or activity shall report the limit of detection and the limit of quantitation with the sample results. If a substance is detected below the limit of quantitation, the owner or operator shall report the detected value with the appropriate qualifier to the regulatory agency.

History: Cr. Register, September, 1985, No. 357, eff. 10-1-85; am. (1), Register, September, 1990, No. 417, eff. 10–1–90; am. (1), r. and recr. (2), Register, March, 1994, No. 459, eff. 4–1–94; (5) renum. from NR 140.14 (3) (c), cr. (4), Register, August, 1995, No. 476, eff. 9-1-95; r. and recr. (1), Register, December, 1998, No. 516, eff. 1-1-99.

#### Subchapter III — Evaluation and Response **Procedures**

NR 140.20 Indicator parameter groundwater standards. (1) Establishing background water quality. Background water quality at a facility, practice or activity at which monitoring is required shall be established by sampling one or more monitoring points at locations and depths sufficient to yield groundwater samples that are representative of background water quality at or near the facility, practice or activity. Background water quality shall be determined for indicator parameters specified by the department. Background water quality for indicator parameters shall be established by averaging a minimum of 8 sample results from each well. The department may exclude any sample result which is nonrepresentative of background water quality. In making the calculations required in this section, the department may use as many representative sample points as are available.

(2) ESTABLISHING PREVENTIVE ACTION LIMITS FOR INDICATOR PARAMETERS. For each indicator parameter for which groundwater monitoring is required by the department, the preventive action limit shall be established based upon a change of water quality with respect to background water quality according to the methodology specified in pars. (a) to (c) and in Table 3.

(a) For field pH, the preventive action limit shall be one pH unit above or below the pH of the background water quality.

(b) For field temperature, the preventive action limit shall be 3 standard deviations or 10°F (5.6°C), whichever is greater, above or below the temperature of the background water quality.

(c) For all other indicator parameters, the preventive action limit shall be the background water quality for that parameter plus 3 standard deviations or the background water quality plus the increase of that parameter listed in Table 3, whichever is greater.

Note: The standard deviation for a group of samples is equal to the square root of: the value of the sum of the squares of the difference between each sample in the sample group and the mean for that sample group divided by the number of samples in the sample group where the sample group has 30 or more samples and by one less than the number of samples in the sample group where the sample group has less than 30 samples.

#### Table 3

#### Methodology for Establishing Preventive Action Limit for **Indicator Parameters**

Published under s. 35.93, Stats. Updated on the first day of each month. Entire code is always current. The Register date on each page is the date the chapter was last published.

DEPARTMENT OF NATURAL RESOURCES

	Minimum Increase (mal
Parameter	l)
Alkalinity	100
Biochemical oxygen demand	25
$(BOD_5)$	
Calcium	25
Chemical oxygen demand (COD)	25
Magnesium	25
Nitrogen series	
Ammonia nitrogen	2
Organic nitrogen	2
Total nitrogen	5
Potassium	5
Sodium	10
Field specific conductance	200 microSiemens/cm
Total dissolved solids (TDS)	200
Total hardness	100
Total organic carbon (TOC)	1
Total organic halogen (TOX)	0.25

**History:** Cr. Register, September, 1985, No. 357, eff. 10–1–85; am. table 3, Register, October, 1990, No. 418, eff. 11–1–90; am. Table 3, Register, December, 1998, No. 516, eff. 1–1–99; CR 09–102: am. Table 3 Register December 2010 No. 660, eff. 1–1–11.

**NR 140.22** Point of standards application for design and compliance. (1) DESIGN. Except as specified in sub. (1m), facilities, practices or activities regulated by the department, including remedial actions, shall be designed to minimize the level of substances in groundwater and to comply with the preventive action limits to the extent technically and economically feasible at all the following locations:

(a) Any point of present groundwater use.

(b) Any point beyond the boundary of the property on which the facility, practice or activity is located.

(c) Any point within the property boundaries beyond the 3-dimensional design management zone if one is established by the department at each facility, practice or activity under sub. (3).

(d) Every point at which groundwater is monitored to determine if a preventive action limit or enforcement standard has been attained or exceeded for sites identified under s. NR 140.22 (2) (c).

(1m) DESIGN OF ASR SYSTEMS; SPECIFIED SUBSTANCES. The point of standards application to determine if the design of an aquifer storage recovery system, regulated under ch. 280 or 281, Stats., complies with the preventive action limits for a specified substance is 1,200 feet from an aquifer storage and recovery well and at any other well that is not part of the ASR system and that is within 1,200 feet of an aquifer storage recovery well.

(2) COMPLIANCE. (a) Except as specified in par. (d), the point of standards application to determine if a preventive action limit has been attained or exceeded is any point at which groundwater is monitored.

(b) Except as specified in par. (d), the point of standards application to determine whether an enforcement standard has been attained or exceeded shall be the following locations:

1. Any point of present groundwater use;

2. Any point beyond the boundary of the property on which the facility, practice or activity is located;

3. Any point within the property boundaries beyond the 3 dimensional design management zone if one is established by the department at each facility, practice or activity under sub. (3).

**Note:** The boundary beyond which the enforcement standards apply is the closer of the property boundary or the design management zone boundary to the waste boundary for the facility, practice or activity.

(c) For discharges, releases, sites or facilities regulated under s. 292.11, 291.29 or 291.37, Stats., or s. NR 600.07, for which a design management zone has not been established in sub. (3), Table 4, the point of standards application shall be every point at which groundwater is monitored to determine if a preventive action limit or enforcement standard has been attained or exceeded.

Note: Section NR 600.07 no longer exists.

(d) The point of standards application to determine if a preventive action limit or enforcement standard for a specified substance has been attained or exceeded at an aquifer storage recovery well, regulated under ch. 280 or 281, Stats., is 1,200 feet from the aquifer storage and recovery well and at any other well that is not part of the ASR system and that is within 1,200 feet of the aquifer storage recovery well.

(3) DESIGN MANAGEMENT ZONE. (a) The design management zone for facilities, practices or activities subject to regulation by the department shall be an area enclosed by vertical boundaries which extend from the land surface downward through all saturated geological formations. The design management zone shall extend horizontally beyond the waste boundary or ASR displacement zone to the distance indicated in Table 4 for the specific type of facility, practice or activity. The waste boundary shall be the outermost limit at which waste from a facility, practice or activity has been stored, applied or disposed of, or permitted or approved for storage, application or disposal. For hazardous waste facilities regulated under ch. 291, Stats., the waste boundary shall include the horizontal space taken up by any liner, dike or other barrier to contain waste.

(b) In issuing or reissuing a permit, license or approval, the department may consider an expansion or reduction of the design management zone at a regulated or proposed facility, practice or activity by a horizontal distance not to exceed 50% of the distance listed in Table 4.

(c) The department shall consider the following factors in determining whether to expand or reduce the design management zone:

1. Nature, thickness and permeability of unconsolidated materials, including topography;

2. Nature and permeability of bedrock;

3. Groundwater depth, flow direction and velocity;

 Waste volume, waste type and characteristics, including waste loading;

5. Contaminant mobility:

6. Distances to property boundary and surface waters;

7. Engineering design of the facility, practice or activity;

8. Life span of the facility, practice or activity;

9. Present and anticipated uses of land and groundwater; and

10. Potential abatement options if an enforcement standard is exceeded.

(d) The design management zone may not be expanded or reduced unless it has been demonstrated to the satisfaction of the department that the preventive action limits and enforcement standards will be met at the adjusted design management zone.

331

The design management zone may not be expanded unless it has been demonstrated to the satisfaction of the department that the preventive action limits and enforcement standards cannot be met at the design management zone specified in Table 4.

Table 4

	Horizontal
	Distances for the
	Design
Type of Facility, Practice or Activity	Management Zone
Land disposal systems regulated under ch. 283, Stats.	250 feet
Wastewater and sludge storage or treatment lagoons regulated under ch. 281 or 283, Stats.	100 feet
Solid waste disposal facilities regu- lated under ch. 289, Stats., which have feasibility reports approved after October 1, 1985.	150 feet
All other solid waste disposal facilities regulated under ch. 289, Stats.	300 feet
Hazardous waste disposal facilities, waste piles, landfills and surface impoundments subject to regulation under ss. NR 665.0090 to 665.0094	300 feet
Hazardous waste disposal facilities, waste piles, landfills and surface impoundments subject to regulation under ss. NR 664.0090 to 664.0100.	0 feet
Aquifer storage recovery systems reg- ulated under ch. 280 or 281, Stats.	0 feet

**History:** Cr. Register, September, 1985, No. 357, eff. 10–1–85; am. (1) (b), Register, October, 1988, No. 394, eff. 11–1–88; am. (4) and table 4, Register, January, 1992, No. 433, eff. 2–1–92; am. (1), cr. (1) (d), renum. (2) to (5) to be (2) (a), (b), (c) and (3) and am. (2) (b) 3, Register, March, 1994, No. 459, eff. 4–1–94; CR 02–134; am. (1) (intro.), (2) (a), (b) (intro.), (3) (a) and Table 4, cr. (1m) and (2) (d) Register June 2003 No. 570, eff. 7–1–03; correction in Table 4 made under s. 13.93 (2m) (b) 7., Stats., Register November 2006 No. 611.

**NR 140.24 Responses when a preventive action limit is attained or exceeded. (1)** NOTIFICATION AND ASSESS-MENT. If the concentration of a substance, including indicator parameters, in groundwater attains or exceeds a preventive action limit at a point of standards application as described in s. NR 140.22 (2):

(a) The owner or operator of the facility, practice or activity shall notify the department in writing when monitoring data is submitted that a preventive action limit has been attained or exceeded in accordance with any deadlines in applicable statutes, rules, permits or plan approvals. Where no deadlines are imposed, the owner or operator shall notify the department as soon as practical after the results are received. When the results of any private well sampling attain or exceed a preventive action limit, the owner or operator of the facility, practice or activity shall notify the department within 10 days after the results are received. The notification shall provide a preliminary analysis of the cause and significance of the concentration.

**Note:** Section 292.11 (2) (a), Stats., requires that the department be notified immediately of hazardous substance discharges.

Note: See s. NR 140.27.

(b) Upon receipt of the notice under par. (a), the department shall evaluate the information and, if further information is required to make the assessment under par. (c), direct the owner or operator to prepare and submit a report by a specified deadline. The report shall assess the cause and significance of the increased concentration based on a consideration of the factors identified in par. (c) and shall propose a response to meet the objectives of sub. (2). (c) The department shall assess the cause and significance of the concentration of the substance in determining the appropriate response to meet the objectives of sub. (2). In addition to all other relevant information, the department shall consider the information submitted under par. (b) and the following factors where applicable:

1. Background water quality. a. The department shall compare background water quality data and monitoring data from wells downgradient of the facility, practice or activity to determine if downgradient water quality is adversely affected. If the background water quality at a facility, practice or activity is not known or is inadequately defined, the department may require additional sampling of existing wells, or installation and sampling of additional wells, or both.

b. Except for substances which are carcinogenic, teratogenic or mutagenic in humans, before requiring a response at a site where the background concentration of a substance is determined to be equal to or greater than the preventive action limit, the department shall determine that the proposed remedial action will protect or substantially improve groundwater quality notwithstanding the background concentrations of naturally occurring substances.

2. Reliability of sampling data. As part of its review of the quality of the sampling data, the department shall evaluate the sampling procedures, precision and accuracy of the analytical test, size of the data set, and the quality control and quality assurance procedures used. If there is insufficient information to evaluate the reliability of the sampling data, the department may require additional samples or other changes in the monitoring program at the facility, practice or activity.

3. Public health, welfare and environmental effects of the substance. The department shall consider the public health, welfare and environmental effects of the substance, including but not limited to its mobility in the subsurface, environmental fate, the risks considered when the standard was adopted and whether it is carcinogenic, mutagenic, teratogenic or has interactive effects with other substances.

4. Probability that a preventive action limit or an enforcement standard may be attained or exceeded outside the design management zone. In evaluating the probability that a preventive action limit or an enforcement standard may be attained or exceeded outside the design management zone, the department shall consider, at a minimum, geologic conditions, groundwater flow rate and direction, contaminant mobility in the subsurface and environmental fate.

5. Performance of the facility, practice or activity. The department shall consider whether the facility, practice or activity is performing as designed in accordance with the design requirements in s. NR 140.22 (1). The department shall consider the type, age and size of the facility, practice or activity; the type of design, if applicable; the operational history; and other factors related to performance of the facility, practice or activity as appropriate.

6. Location of the monitoring point. The department shall consider the location of the monitoring point in relation to the facility, practice or activity and the design management zone in assessing the appropriate response.

7. Other known or suspected sources of the substance in the area. If other known or suspected sources are present in the vicinity of a facility, practice or activity of concern, the department shall evaluate the probability of contributions from other sources of the substance. The department shall consider, at a minimum, the number, size, type and age of nearby sources; the groundwater flow patterns; and the substances involved.

8. Hydrogeologic conditions. The department shall consider the geologic and groundwater conditions, including but not limited to the nature, thickness and permeability of the unconsolidated materials; the nature and permeability of bedrock; the depth

Published under s. 35.93, Stats. Updated on the first day of each month. Entire code is always current. The Register date on each page is the date the chapter was last published. to the water table; groundwater flow gradients, both vertical and horizontal; the position of the facility, practice or activity within the groundwater flow system; and the present and potential groundwater use in the vicinity of the facility, practice or activity at which an exceedance occurs. If there is insufficient hydrogeologic information, the department may require additional information.

9. Extent of groundwater contamination. The department shall consider the current and anticipated future extent of groundwater contamination in 3 dimensions. If water supplies are affected or threatened, the department shall evaluate the existing effects and potential risks of the substance on the potable water supplies. If the extent of contamination is not known, the department may require further documentation of the extent of contamination.

10. Alternate responses. The department shall evaluate alternate responses, including consideration of the technical and economic feasibility of alternate responses from Table 5 or 6 or both, the practicality of stopping the further release of the substance and the risks and benefits of continued operation of the facility, practice or activity and the ability of a response to meet other applicable environmental protection laws.

(2) RESPONSE OBJECTIVES. Based on its evaluation of the report required under sub. (1), and the assessment criteria of sub. (1) (c), the department shall specify the responses to be implemented by the owner or operator of the facility, practice or activity designed to the extent technically and economically feasible to prevent any new releases of the substance from traveling beyond the design management zone or other applicable points of standards application described in s. NR 140.22 and restore contaminated groundwater within a reasonable period of time, considering the criteria specified in s. NR 722.07. Both the source control and the groundwater restoration components of the response shall be designed and implemented to:

(a) Minimize the concentration of the substance in groundwater at the point of standards application where technically and economically feasible;

(b) Regain and maintain compliance with the preventive action limit. If the department determines that compliance with the preventive action limit is either not technically or economically feasible, the owner or operator shall achieve compliance with the lowest possible concentration which is technically and economically feasible; and

(c) Ensure that the enforcement standard is not attained or exceeded at the point of standards application.

(3) RANGE OF RESPONSES FOR INDICATOR PARAMETERS. Except as otherwise provided in this subsection, the range of responses which the department may take or may require if a preventive action limit for an indicator parameter identified in Table 3 has been attained or exceeded, is one or more of the responses in items 1 to 4 in Table 5. The range of responses is one or more of the responses in items 1 to 6 of Table 5 in the event the department determines that:

(a) There is a threat to public health or welfare as a result of a preventive action limit for an indicator parameter being attained or exceeded; or

(b) The results demonstrate a significant design flaw or failure of the facility to contain substances, such that the facility can be expected to emit one or more of the substances on Table 1 or 2 in excess of a preventive action limit at a point of standards application.

(4) RANGE OF RESPONSES FOR SUBSTANCES OF PUBLIC HEALTH OR WELFARE CONCERN. The range of responses which the department may take or may require the owner or operator of a facility, practice or activity to take if a preventive action limit for a substance of health or welfare concern has been attained or exceeded are listed in Table 5. More than one response may be taken or required by the department.

#### Table 5

Range of Responses for Exceedances of a Preventive Action Limit for Indicator Parameters and Substances of Health or Welfare Concern

- 1. No action pursuant to s. NR 140.24 (5) and consistent with s. 160.23, Stats.
- 2. Require the installation and sampling of groundwater monitoring wells.
- Require a change in the monitoring program, including increased monitoring.
- 4. Require an investigation of the extent of groundwater contamination.
- 5. Require a revision of the operational procedures at the facility, practice or activity.
- Require a change in the design or construction of the facility, practice or activity.
- Require an alternate method of waste treatment or disposal.
- 8. Require prohibition or closure and abandonment of a facility, practice or activity in accordance with sub. (6).
- Require remedial action to renovate or restore groundwater quality.
- Require remedial action to prevent or minimize the further discharge or release of the substance to groundwater.
- Revise rules or criteria on facility design, location or management practices.
- 12. Require the collection and evaluation of data to determine whether natural attenuation can be effective to restore groundwater quality within a reasonable period of time, considering applicable criteria specified in ss. NR 140.24, 722.07 and 722.09 or 722.11, and require monitoring to determine whether or not natural attenuation is occurring in compliance with the response objectives in s. NR 140.24 (2).

(5) NO ACTION RESPONSE CRITERIA. For facilities, practices and activities with a design management zone specified in s. NR 140.22 (3) Table 4, the department may determine that no response is necessary and that an exemption under s. NR 140.28 is not required when either of the following conditions is met:

(a) The concentration of a substance within a design management zone is detected above the preventive action limit, the enforcement standard has not been attained or exceeded within the design management zone, and the department determines that there is no indication that the preventive action limit will be attained or exceeded at any point outside the design management zone, or

(b) The background concentration of a substance is greater than the preventive action limit, the anticipated or detected incremental increase in the concentration of a substance which results from a specific facility, practice or activity is not greater than the preventive action limit, and the anticipated or detected concentration is not greater than the enforcement standard either within or outside of the design management zone.

(6) PROHIBITION AND CLOSURE CRITERIA. The department may not impose a prohibition on a practice or activity or require closure of a facility which produces the substance unless the department:

(a) Bases its decision upon reliable test data;

(b) Determines, to a reasonable certainty, by the greater weight of the credible evidence, that no other remedial action would prevent the violation of the enforcement standard at the point of standards application;

(c) Establishes the basis for the boundary and duration of the prohibition; and

Published under s. 35.93, Stats. Updated on the first day of each month. Entire code is always current. The Register date on each page is the date the chapter was last published.
Register February 2017 No. 734

(d) Ensures that any prohibition imposed shall be reasonably related in time and scope to maintaining compliance with the enforcement standard at the point of standards application.

History: Cr. Register, September, 1985, No. 357, eff. 10–1–85; am. (5) (intro.) and (6) (intro.), Register, October, 1988, No. 394, eff. 11–1–88; am. (1) (intro.), (a), (b), (c) (intro.), 5. and 10. (2) (intro.), and (5) (intro.), renum. (7) to be NR 104.02 (4), Register, January, 1992, No. 433, eff. 2–1–92; am. (1) (intro.), (c) (intro.), (3) (intro.) and Table 5, Register, March, 1994, No. 459, eff. 4–1–94; am. (1) (a), (5) (intro.), Register, August, 1995, No. 476, eff. 9–1–95; am. (2) (intro.), (4) and Table 5, Register, Ctober, 1996, No. 490, eff. 11–1–96; am. (1) (a), Register, December, 1998, No. 516, eff. 1–1–99.

**NR 140.26 Responses when an enforcement standard is attained or exceeded. (1)** NOTIFICATION AND ASSESS-MENT. If the concentration of a substance in groundwater attains or exceeds an enforcement standard at a point of standards application as described in s. NR 140.22 (2):

(a) The owner or operator of the facility, practice or activity shall notify the department in writing when monitoring data is submitted that an enforcement standard has been attained or exceeded in accordance with any deadlines in applicable statutes, rules, permits or plan approvals. Where no deadlines are imposed, the owner or operator shall notify the department as soon as practical after the results are received. When the results of any private well sampling attain or exceed an enforcement standard or preventive action limit, the owner or operator of the facility, practice or activity shall notify the department within 10 days after the results are received. The notification shall provide a preliminary analysis of the cause and significance of the concentration.

**Note:** Section 292.11 (2) (a), Stats., requires that the department be notified immediately of hazardous substance discharges.

Note: See s. NR 140.27.

(b) Upon receipt of the notice under par. (a), the department shall evaluate the information and, if further information is required to make the assessment under par. (c), direct the owner or operator to prepare and submit a report by a specified deadline. The report shall assess the cause and significance of the increased concentration based on a consideration of the factors identified in s. NR 140.24 (1) (c) and shall propose a response to achieve compliance with the enforcement standard at the point of standards application and to comply with sub. (4).

(c) The department shall assess the cause and significance of the concentration of the substance in determining the appropriate response measures to achieve compliance with the enforcement standard at the point of standards application and to comply with sub. (4). In addition to all other relevant information, the department shall consider the information submitted under sub. (1) and the factors listed in s. NR 140.24 (1) (c), where applicable.

(2) REGULATORY RESPONSES. (a) If a facility, activity or practice is regulated under subch. IV of ch. 283, Stats., ch. 289, 291, or 292, Stats., the department shall require responses as necessary, based on the evaluation of the increased concentration as outlined in sub. (1), to prevent any new releases of the substance from traveling beyond the design management zone or other applicable point of standards application described in s. NR 140.22 and restore contaminated groundwater within a reasonable period of time, considering the criteria specified in s. NR 722.07. Both the source control and the groundwater restoration components of the response shall be designed to achieve compliance with the enforcement standard at the point of standards application and to achieve compliance with the preventive action limit at the point of standards application unless compliance with the preventive action limit is not technically and economically feasible. The range of responses which the department may take or may require the owner or operator of a facility, practice or activity to take if an enforcement standard for a substance of public health or welfare concern has been attained or exceeded at a point of standards application is listed in Table 6. More than one response listed in Table 6 may be required by the department. In addition, the department may take or may require the owner or operator of a

facility, practice or activity to take one or more responses from Table 5, except response number one.

#### Table 6

Range of Responses for Exceedance of Enforcement Standards for Substances of Health or Welfare Concern

- 1. Require a revision of the operational procedures at a facility, practice or activity.
- 2. Require a change in the design or construction of the facility, practice or activity.
- Require an alternate method of waste treatment or disposal.
- Require prohibition or closure and abandonment of a facility, practice or activity.
- 5. Require remedial action to renovate or restore groundwater quality.
- 6. Require remedial action to prevent or minimize the further release of the substance to groundwater.
- 7. Revise rules or criteria on facility design, location or management practices.
- Require the collection and evaluation of data to determine whether natural attenuation can be effective to restore groundwater quality within a reasonable period of time, considering applicable criteria specified in ss. NR 140.24, 722.07 and 722.09 or 722.11, and require monitoring to determine whether or not natural attenuation is occurring in compliance with the requirements of s. NR 140.26 (2) (a).

(b) If an activity or practice is not subject to regulation under subch. IV of ch. 283, Stats., ch. 289, 291, or 292, Stats., and if the concentration of a substance in groundwater attains or exceeds an enforcement standard at a point of standards application, the department shall take the following responses unless it can be shown to the department that, to a reasonable certainty, by the greater weight of the credible evidence, an alternative response will achieve compliance with the enforcement standard at the point of standards application:

1. Prohibit the activity or practice which uses or produces the substance; and

2. Require remedial actions with respect to the specific site in accordance with this chapter.

(3) RESPONSES FOR NITRATE AND SUBSTANCES OF PUBLIC WEL-FARE CONCERN. If nitrates or any substance of welfare concern only attains or exceeds an enforcement standard, the department is not required to impose a prohibition or close a facility if it determines that:

(a) The enforcement standard was attained or exceeded, in whole or in part, because of high background concentrations of the substance; and

(b) The additional concentration does not represent a public welfare concern.

(4) COMPLIANCE WITH PREVENTIVE ACTION LIMITS. When compliance with the enforcement standard is achieved at the point of standards application, s. NR 140.24 applies.

**History:** Cr. Register, September, 1985, No. 357, eff. 10–1–85; am. (1) (intro.), (a), (b), (2), r. (6), Register, January, 1992, No. 433, eff. 2–1–92; am. (1) (intro.) and Table 6, renum. (2) to (5) to be (2) (a), (b), (3) and (4), Register, March, 1994, No. 459, eff. 4–1–94; am. (1) (a), Register, August, 1995, No. 476, eff. 9–1–95; correction in (1) (b) and (c) made under s. 13.93 (2m) (b) 7., Stats., Register, August, 1995, No. 476; am. (2) (a) and Table 6, Register, October, 1996, No. 490, eff. 11–1–96; am. (1) (a), Register, December, 1998, No. 516, eff. 1–1–99; correction in (2) (a), (b) (intro.) made under s. 13.92 (4) (b) 7., Stats., Register February 2017 No. 734.

Published under s. 35.93, Stats. Updated on the first day of each month. Entire code is always current. The Register date on each page 2017 No. 734 is the date the chapter was last published. NR 140.27 Responses when an enforcement standard is attained or exceeded at a location other than a point of standards application. If the concentration of a substance in groundwater attains or exceeds an enforcement standard at a location other than a point of standards application for an enforcement standard, s. NR 140.24 shall apply.

History: Cr. Register, October, 1988, No. 394, eff. 11-1-88.

**NR 140.28 Exemptions. (1)** APPLICABILITY. (a) The department may not approve a proposed facility, practice or activity at a location where a preventive action limit or enforcement standard adopted under s. NR 140.10 or 140.12 has been attained or exceeded unless an exemption has been granted under this section.

(b) For an existing facility, practice or activity, a response is required under s. NR 140.24 (2) or 140.26 (2) when a preventive action limit or an enforcement standard has been attained or exceeded at a point of standards application unless an exemption has been granted under this section or the criteria of s. NR 140.24 (5) (a) or (b) are met.

(c) For an existing facility, practice or activity that has taken or is taking a response under s. NR 140.24 (2) or 140.26 (2), a continued response is required unless a substance no longer attains or exceeds a preventive action limit or an exemption has been granted under this section.

(d) If a substance or remedial material is to be infiltrated or injected into groundwater at a concentration which attains or exceeds a preventive action limit, or at any concentration for a substance or remedial material for which a groundwater quality standard has not been established under this chapter, a temporary exemption is required under sub. (5).

(2) CRITERIA FOR GRANTING EXEMPTIONS WHERE THE BACK-GROUND CONCENTRATION IS BELOW THE PREVENTIVE ACTION LIMIT. (a) The department may grant an exemption under this section to a facility, practice or activity which is regulated by the department in an area where the background concentration of nitrate or a substance of public welfare concern is below the preventive action limit if the facility, practice or activity is designed and implemented to achieve the lowest possible concentration for that substance which is technically and economically feasible and the existing or anticipated increase in the concentration of that substance does not present a threat to public health or welfare.

(b) The department may grant an exemption under this section to a facility, practice or activity which is regulated by the department in an area where the background concentration of a substance of public health concern, other than nitrate, is below the preventive action limit for that substance if all of the following occur:

1. The measured or anticipated increase in the concentration of the substance will be minimized to the extent technically and economically feasible.

2. Compliance with the preventive action limit is either not technically or economically feasible.

3. The enforcement standard for that substance will not be attained or exceeded at the point of standards application.

4. Any existing or projected increase in the concentration of the substance above the background concentration does not present a threat to public health or welfare.

**Note:** An exemption may be considered under this subsection even if monitoring data indicates no detectable background concentration of the substance.

(3) CRITERIA FOR GRANTING EXEMPTIONS WHERE THE BACK-GROUND CONCENTRATION IS ABOVE A PREVENTIVE ACTION LIMIT. (a) The department may grant an exemption under this section to a facility, practice or activity which is regulated by the department in an area where the background concentration of nitrate or a substance of public welfare concern attains or exceeds the preventive action limit if the facility, practice or activity is designed to achieve the lowest possible concentration for that substance which is technically and economically feasible and the existing or anticipated increase in the concentration of the substance does not present a threat to public health or welfare.

(b) The department may grant an exemption under this section to a facility, practice or activity which is regulated by the department in an area where the background concentration of a substance of public health concern, other than nitrate, attains or exceeds a preventive action limit for that substance:

1. If the facility, practice or activity has not caused and will not cause the further release of that substance into the environment; or

2. If the background concentration of the substance does not exceed the enforcement standard for that substance, the facility, practice or activity has not caused and will not cause the concentration of the substance to exceed the enforcement standard for that substance at a point of standards application and the facility, practice or activity is designed to achieve the lowest possible concentration of that substance which is technically and economically feasible.

(4) CRITERIA FOR GRANTING EXEMPTIONS WHERE THE BACK-GROUND CONCENTRATION IS ABOVE AN ENFORCEMENT STANDARD. (a) The department may grant an exemption under this section to a facility, practice or activity which is regulated by the department in an area where the background concentration of nitrate or a substance of public welfare concern attains or exceeds an enforcement standard if the facility, practice or activity is designed to achieve the lowest possible concentration for that substance which is technically and economically feasible and the existing or anticipated increase in the concentration of the substance does not present a threat to public health or welfare.

(b) The department may grant an exemption under this section to a facility, practice or activity which is regulated by the department in an area where the background concentration of a substance of public health concern, other than nitrate, attains or exceeds the enforcement standard for that substance if:

1. The facility has not caused and will not cause the further release of that substance into the environment; or

2. a. The facility is designed to achieve the lowest possible concentration of that substance which is technically and economically feasible; and

b. The existing or anticipated increase in the concentration of the substance has not caused or will not cause an increased threat to public health or welfare; and

c. The existing or anticipated incremental increase in the concentration of the substance by itself, has not exceeded or will not exceed the preventive action limit.

(c) The department shall take action under s. NR 140.26 if it determines that the increase in the concentration of the substance causes an increased threat to public health or welfare or it determines that the incremental increase in the concentration of the substance, by itself, exceeds the preventive action limit.

(5) CRITERIA FOR GRANTING A TEMPORARY EXEMPTION WHERE INFILTRATION OR INJECTION IS UTILIZED FOR A REMEDIAL ACTION. (a) *General.* In lieu of an exemption granted in compliance with the criteria in subs. (2) to (4), the department may grant a temporary exemption if the criteria in this subsection are complied with. This exemption applies to the owner or operator of a facility, practice or activity that is undertaking a remedial action that: includes the infiltration or injection of contaminated groundwater or remedial material, has been approved by the department, and will comply with the applicable response objectives under s. NR 140.24 or 140.26 within a reasonable period of time. The owner or operator of the facility, practice or activity may submit a temporary exemption request to the department at the same time or after the department has approved the remedial action.

(b) *Exemption request.* The owner or operator of the facility, practice or activity shall submit a request for a temporary exemption to the department. As part of the request, the applicant shall indicate how the exemption prerequisites under par. (c) and appli-

Published under s. 35.93, Stats. Updated on the first day of each month. Entire code is always current. The Register date on each page is the date the chapter was last published.
Register February 2017 No. 734

cable remedial design, operational and monitoring criteria under par. (d) will be met.

**Note:** For most remedial actions, a microcosm or treatability study, or other bench scale or pilot scale study will be required by the department prior to consideration of an exemption for the full–scale remedial action under this section. If a pilot scale study is deemed necessary before an exemption for a full–scale remedial action can be granted, a separate temporary exemption issued under this section is required before the pilot scale study can begin.

(c) *Exemption prerequisites*. As part of the temporary exemption request, the owner or operator shall demonstrate to the satisfaction of the department that all of the following requirements will be met:

1. The remedial action for restoring contaminated soil or groundwater, and any infiltrated or injected contaminated water and remedial material, shall achieve the applicable response objectives required by s. NR 140.24 (2) or 140.26 (2) within a reasonable period of time.

2. The type, concentration and volume of substances or remedial material to be infiltrated or injected shall be minimized to the extent that is necessary for restoration of the contaminated soil or groundwater and be approved by the department prior to use.

3. Any infiltration or injection of contaminated water or remedial material into soil or groundwater will not significantly increase the threat to public health or welfare.

4. No uncontaminated or contaminated water, substance or remedial material will be infiltrated or injected into an area where a floating non-aqueous phase liquid is present in the contaminated soil or groundwater.

5. There will be no expansion of soil or groundwater contamination, or migration of any infiltrated or injected contaminated water or remedial material, beyond the edges of previously contaminated areas, except that infiltration or injection into previously uncontaminated areas may be allowed if the department determines that expansion into adjacent, previously uncontaminated areas is necessary for the restoration of the contaminated soil or groundwater, and the requirements of subd. 1. will be met.

6. All necessary federal, state and local licenses, permits and other approvals are obtained and all applicable environmental protection requirements will be complied with.

**Note:** The issuance of a wastewater discharge permit by the department is required prior to the infiltration or injection of substances or remedial material into unsaturated soil or groundwater for discharges, as defined by s. 283.01 (4), Stats. A wastewater discharge permit establishes the effluent or injection limits for substances or remedial material which may be infiltrated or injected into unsaturated soil or groundwater. A temporary exemption granted under this subsection applies to substances or remedial material which may enter groundwater or may be detected at a point of standards applications; it does not apply to substances or remedial material infiltrated or injected into unsaturated soil.

(d) Remedial action design, operation and monitoring criteria. In addition to providing information on how the requirements under par. (c) will be met, the application shall specify the following information where applicable.

1. The remedial action design, operation and soil and groundwater monitoring procedures to insure compliance with the requirements under par. (c) and applicable criteria under this paragraph. The level of pre-treatment for contaminated groundwater prior to reinfiltration or reinjection.

3. The types and concentrations of substances or remedial material being proposed for infiltration or injection.

4. The volume and rate of infiltration or injection of contaminated groundwater or remedial material.

5. The location where the contaminated groundwater or remedial material will be infiltrated or injected.

(e) *Granting an exemption*. The department may only grant a temporary exemption under this subsection at the same time or after the department has approved the remedial action. When the department grants an exemption under this subsection, it shall follow the exemption procedures included in sub. (6) and shall require the owner or operator of the facility, practice or activity to comply with the requirements and criteria in pars. (c) and (d). The temporary exemption shall also include:

1. The expiration date of the temporary exemption. The expiration date shall be selected to achieve the applicable response objectives required by s. NR 140.24 (2) or 140.26 (2) within a reasonable period of time, not to exceed 5 years from the effective date of the exemption. The temporary exemption may be reissued following a department review of information documenting the performance of the remedial action and a successful demonstration that reissuance of the exemption is necessary to achieve the response objectives required by s. NR 140.24 (2) or 140.26 (2).necessary relating to the temporary exemption.

(f) *Responses to exemption violations*. If the department determines that the conditions or requirements specified in the temporary exemption are not being met, the department may:

1. Require that the owner or operator of the facility, practice or activity revise the remedial action design, operation or monitoring procedures in accordance with par. (d). All revisions shall comply with the requirements established under pars. (c) and (e) and may require approval from the department prior to implementation.

2. Revoke the exemption and require implementation of an alternate remedial action to restore soil or groundwater quality.

(6) EXEMPTION PROCEDURES. If the department grants an exemption under this section for a substance or a remedial material, it shall specify:

(a) The substance or remedial material to which the exemption applies;

(b) The terms and conditions of the exemption, which may include an alternative concentration limit, under which the department may seek a response under s. NR 140.24 or 140.26 relating to the substance or remedial material; and

(c) Any other conditions relating to the exemption.

(c) Arly other conditions relating to the exemption. **History:** Cr. Register, September, 1985, No. 357, eff. 10–1–85; am. (1) (a) and (b), (3) (a), (b) (intro.) and 2., (4) (a) and (b) 1. and (5) (b), Register, October, 1988, No. 394, eff. 11–1–88; am. (1) (b), Register, January, 1992, No. 433, eff. 2–1–92; correction in (4) (b) made under s. 13.93 (2m) (b) 1., Stats., Register, January, 1992, No. 433; am. (1) (b) and (5) (b), Register, March, 1994, No. 459, eff. 4–1–94; renum. (5) to be (6), cr. (5), Register, August, 1995, No. 476, eff. 9–1–95; cr. (1) (c), (d), am. (2) (intro.), (5) (a), (6) (intro.), (a) and (b), Register, December, 1998, No. 516, eff. 1–1–99; r. and recr. (2), Register, March, 2000, No. 531, eff. 4–1–00. DEPARTMENT OF NATURAL RESOURCES

NR 140.28

Substance	CAS RN <sup>1</sup>	Common synonyms/ <i>Tradename</i> <sup>2</sup>
Acetochlor	34256-82-1	Cadence, Degree, Harness, Keystone, Over- time, Volley
Acetochlor ethane sulfonic acid +	187022-11-3 (ESA)	Acetochlor – ESA + OXA
oxanilic acid	184992-44-4 (OXA)	
Acetone	67-64-1	Propanone
Alachlor	15972-60-8	Lasso
Alachlor ethane sulfonic acid	142363-53-9	Alachlor–ESA, Alachlor Ethane Sulfonate, MON 5775
Aldicarb	116-06-3	Temik
Aluminum	7429-90-5	
Ammonia	7664-41-7	
Anthracene	120-12-7	Para-naphthalene
Asbestos	1332-21-4	
Bentazon	25057-89-0	Basagran
Benzene	71-43-2	
Benzo(b)fluoranthene	205-99-2	B(b)F,3,4–Benzofluoranthene
Benzo(a)pyrene	50-32-8	BaP, B(a)P
Boron	7440-42-8	
Bromodichloromethane	75-27-4	Dichlorobromomethane, BDCM
Bromoform	75-25-2	Tribromomethane
Bromomethane	74-83-9	Methyl bromide
Butylate	2008-41-5	S-ethyl di-isobutylthiocarbamate, Sutan+
Carbaryl	63-25-2	Sevin
Carbofuran	1563-66-2	Furadan
Carbon disulfide	75-15-0	Carbon bisulfide
Carbon tetrachloride	56-23-5	Tetrachloromethane, Perchloroethane
Chloramben	133-90-4	
Chlordane	57-74-9	
Chlorodifluoromethane	75-45-6	HCFC-22, Freon 22
Chloroethane	75-00-3	Ethyl chloride, Monochloroethane
Chloroform	67-66-3	Trichloromethane
Chlorpyrifos	2921-88-2	Dursban, Lorsban, Warhawk, Hatchet, Yuma, Whirlwind, Eraser
Chloromethane	74-87-3	Methyl chloride
Chromium (total)	7440-47-3	
Chrysene	218-01-9	1,2-Benzphenanthrene
Cobalt	7440-48-4	
Cyanazine	21725-46-2	<i>Bladex</i> , 2-chloro-4-ethylamino-6- nitriloisopropylamino-s-triazine
Cyanide, free	57-12-5	
Dacthal	1861-32-1	DPCA, Chlorothal, <i>Dacthalor</i> , 1,4-benzene- dicarboxylic acid
Dibromochloromethane	124-48-1	Chlorodibromomethane, DBCM
1,2-Dibromo-3-chloropropane	96-12-8	DBCP, Dibromochloropropane
1,2–Dibromoethane	106-93-4	EDB, Ethylene dibromide, Dibromoethane
Dibutyl phthalate	84-74-2	DP, Di- <i>n</i> -butyl phthalate, <i>n</i> -Butyl phthalate
Dicamba	1918-00-9	Banvel
1,2-Dichlorobenzene	95-50-1	o-Dichlorobenzene, o-DCB
1,3-Dichlorobenzene	541-73-1	m-Dichlorobenzene, m-DCB
1,4-Dichlorobenzene	106-46-7	p-Dichlorobenzene, p-DCB
Dichlorodifluoromethane	75-71-8	Freon 12
1,1,-Dichloroethane	75-34-3	Ethylidine chloride
1,2–Dichloroethane	107-06-2	1,2–DCA, Ethylene dichloride

#### CHAPTER <mark>NR 140</mark> APPENDIX I TO TABLE 1 UBLIC HEALTH GROUNDWATER QUALITY STANDARDS

 Published under s. 35.93, Stats. Updated on the first day of each month. Entire code is always current. The Register date on each page

 is the date the chapter was last published.

 Register February 2017 No. 734

NR 140.28

#### WISCONSIN ADMINISTRATIVE CODE

Substance	CAS RN <sup>1</sup>	Common synonyms/ <i>Tradename</i> <sup>2</sup>
1,1-Dichloroethylene	75-35-4	1,1–DCE, 1,1–Dichloroethene, Vinylidene chloride
1,2-Dichloroethylene (cis)	156-59-2	cis–Dichloroethylene, 1,2–Dichloroethene
1 2–Dichloroethylene (trans)	156-60-5	trans-1 2-Dichloroethylene
2 4–Dichlorophenoxyacetic acid	94-75-7	2 4–D
1 2–Dichloropropane	78-87-5	Propylene dichloride
1.3-Dichloropropene	542-75-6	Telone DCP Dichloropropylene
(cis/trans) <sup>3</sup>	512 15 0	reione, Der, Diemoropropylene
Di(2-ethylhexyl) phthalate	117-81-7	DEHP, Bis(2–ethylhexyl) phthalate, 1,2–Benzenedicarboxylic acid, Bis (2–ethyl- hexyl)ester
Dimethenamid/Dimethinamid-P	87674-68-8	Frontier, Outlook, Propel, Establish, Sortie,
	163515-14-8 (-P)	Tower
Dimethoate	60-51-5	
2,4-Dinitrotoluene	121-14-2	2,4–DNT, 1–methyl–2,4–dinitrobenzene
2,6-Dinitrotoluene	606-20-2	2,6–DNT, 2–methyl–1,3–dinitrobenzene
Dinitrotoluene, Total Residues	25321-14-6	Dinitrotoluene, DNT
Dinoseb	88-85-7	2-(1-methylpropyl)-4,6-dinitrophenol
1,4–Dioxane	123-91-1	<i>p</i> –Dioxane
Dioxin	1746-01-6	2,3,7,8-TCDD,2,3,7,8-Tetrachlorodibenzo- p-dioxin
Endrin	72-20-8	
EPTC	759-94-4	Eptam, Eradicane
Ethylbenzene	100-41-4	Phenylethane, EB
Ethyl ether	60-29-7	Diethyl Ether
Ethylene glycol	107-21-1	
Fluoranthene	206-44-0	Benzo(jk)fluorene
Fluorene	86-73-7	2,3-Benzidine, Diphenylenemethane
Fluoride	7681-49-4	
Fluorotrichloromethane	75-69-4	Freon11, Trichlorofluoromethane
Formaldehyde	50-00-0	
Heptachlor	76-44-8	Velsicol
Heptachlor epoxide	1024-57-3	
Hexachlorobenzene	118-74-1	Perchlorobenzene, Granox
<i>N</i> –Hexane	110-54-3	Hexane, Skellysolve B
Hydrogen sulfide	7783-06-4	Dihydrogen sulfide
Lindane	58-89-9	
Manganese	7439–96–5	
Mercury	7439–97–6	
Methanol	67-56-1	Methyl alcohol, Wood alcohol
Methoxychlor	72-43-5	
Methylene chloride	75-09-2	Dichloromethane, Methylene dichloride
Methyl ethyl ketone	78-93-3	MEK, 2–Butanone
Methyl isobutyl ketone	108-10-1	MIBK, 4–Methyl–2–pentanone, Isopropyla- cetone, <i>Hexone</i>
Methyl tert-butyl ether	1634-04-4	MTBE, 2–Methoxy–2–methyl–propane, tert–Butyl methyl ether
Metolachlor/s-Metolachlor	51218-45-2 87392-12-9 (s-)	Dual, Bicep, Milocep, Stalwart, Parallel, Prefix, Charger, Brawl, Cinch, Dual Mag- num, Boundary
Metolachlor ethane sulfonic acid	171118-09-5 (ESA)	Metolachlor – ESA + OXA
+ oxanilic acid	152019-73-3 (OXA)	
Metribuzin	21087-64-9	Sencor, Lexone
Molybdenum	7439-98-7	·
Monochlorobenzene	108-90-7	Chlorobenzene
Naphthalene	91-20-3	

Published under s. 35.93, Stats. Updated on the first day of each month. Entire code is always current. The Register date on each page 2017 No. 734 is the date the chapter was last published.
336-3

#### DEPARTMENT OF NATURAL RESOURCES

NR 140.28

Substance	CAS RN <sup>1</sup>	Common synonyms/ <i>Tradename</i> <sup>2</sup>
<i>N</i> –Nitrosodiphenylamine	86-30-6	NDPA
Pentachlorophenol	87-86-5	PCP, Pentachlorohydroxybenzene
Perchlorate	14797-73-0	Perchlorate and perchlorate salts, Perchlo- rate ion
Phenol	108-95-2	
Picloram	1918-02-1	<i>Tordon</i> , 4–amino–3,5,6–trichloropicolinic acid
Polychlorinated biphenyls <sup>4</sup>		PCBs
Prometon	1610-18-0	Pramitol, Prometone
Pyrene	129-00-0	Benzo(def)phenanthrene
Pyridine	110-86-1	Azabenzene
Simazine	122-34-9	<i>Princep</i> , 2-chloro-4,6-diethylamino- s-tri- azine
Styrene	100-42-5	Ethenylbenzene, Vinylbenzene
Tertiary Butyl Alcohol	75-65-0	TBA
1,1,1,2–Tetrachlorethane	630-20-6	1,1,1,2–TCA, 1,1,1,2–PCA
1,1,2,2,-Tetrachloroethane	79-34-5	1,1,2,2–TCA, 1,1,2,2–PCA
Tetrachloroethylene	127-18-4	Perchloroethylene, PERC, Tetrachloroethene
Tetrahydrofuran	109-99-9	THF
Toluene	108-88-3	Methylbenzene
Toxaphene	8001-35-2	
1,2,4–Trichlorobenzene	120-82-1	
1,1,1–Trichloroethane	71-55-6	Methyl chloroform, 1,1,1–TCA
1,1,2-Trichloroethane	79-00-5	1,1,2–TCA, Vinyl trichloride
Trichloroethylene	79-01-6	TCE, Chloroethene
2,4,5–Trichlorophenoxy– propionic acid	93-72-1	2,4,5–TP, <i>Silvex</i>
1,2,3–Trichloropropane	96-18-4	1,2,3–TCP, Glycerol trichlorohyrin
Trifluralin	1582-09-8	Treflan
1,2,4-Trimethylbenzene	95-63-6	
1,3,5-Trimethylbenzene	108-67-8	
Vanadium	7440-62-2	
Vinyl chloride	75-01-4	VC, Chloroethene
Xylene <sup>5</sup>		

<sup>1</sup>Chemical Abstracts Service (CAS) registry numbers are unique numbers assigned to a chemical substance. The CAS registry numbers were published by the U.S. Environ-mental Protection Agency in 40 CFR Part 264, Appendix IV <sup>2</sup>Common synonyms include those widely used in government regulations, scientific publications, commerce and the general public. A trade name, also known as the propri-etary name, is the specific, registered name given by a manufacturer to a product. Trade names are listed in *italics*. Common synonyms and trade names should be cross-refer-enced with CAS registry number to ensure the correct substance is identified.

<sup>3</sup>This is a combined chemical substance which includes cis 1,3–Dichloropropene (CAS RN 10061–01–5) and trans 1,3–Dichloropropene (CAS RN 10061–02–6).

<sup>4</sup>Polychlorinated biphenyls (CAS RN 1336–36–3); this category contains congener chemicals (same molecular composition, different molecular structure and formula), including constituents of Aroclor–1016 (CAS RN 12674–11–2), Aroclor–1221 (CAS RN 1104–28–2), Aroclor–1232 (CAS RN 11141–16–5), Aroclor–1242 (CAS RN 53469–21–9), Aroclor–1248 (CAS RN 12672–29–6), Aroclor–1254 (CAS RN 11097–69–1), and Aroclor–1260 (CAS RN 11096–82–5).
<sup>5</sup>Xylene (CAS RN 1330–20–7) refers to a mixture of three isomers, meta–xylene (CAS RN 108–38–3), ortho–xylene (CAS RN 95–47–6), and para–xylene (CAS RN 1097–69–1).

106-42-3)

Key: I = IRIS	S; P = PPRTV; D	= DWSHA; (	D = OPP; A =	ATSDR; (	C = Cal EP/	A; X = A			TV SCREEN (See FAQ #29); H = HEAST; F = See FAQ; E = see user guide Section rs: $p SL < 1000$ c SL : ** = where $p SL < 100$ c SL : SSL values are based on DAE=1:	2.3.5; W = see	user guide Se	ction 2.3.6;	L = see user gu	ide on lead; M = n	nutagen; S = s	ee user guide	e Section 5; V	= volatile; R = RBA a	pplied (See
	Т	oxicity and C	hemical-spec	ific Inform	ation		cancer,	= wrie	Contaminant	m = concentrati	Ca	rcinoaenic T	arget Risk (TR	= 1E-06	alion may exi	Noncance	r CHILD Haza	;) ird Index (HI) = 0.1	
	тт т			ПТ									T T		Ingestion SL	Dermal SL	Inhalation SL	Noncarcinogenic SL	
050	k k		k pro	k v							Ingestion SL	Dermal SL	Inhalation SL	Carcinogenic SL	Child	Child	Child	Child	
SFO	e IUR e	RfD,	e RICi	e o mut	a-	014.00			And to	010 11	TR=1E-06	TR=1E-06	TR=1E-06	TR=1E-06	THQ=0.1	THQ=0.1	THQ=0.1	THI=0.1	MCL
(mg/kg-day)	y (ug/m°) y	(mg/kg-day)	y (mg/m°)	y I ger	n LOGP	GIABS	FAI	In EPD?	Analyte	CAS No.	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)
	0.05.00	1.2E-03	0		-0.85	1	1	Yes	Acephate	30560-19-1			0.05.00	0.05.00	2.4E+00	3.4E+03	1.05.00	2.4E+00	
	2.2E-06 I	2 0E-02	9.0E-03	IV	-0.34	1	00	Yes	Acetacher	24256-82-1			2.6E+00	2.6E+00	4.05+01	2 0 =+02	1.9E+00	1.9E+00 3.5E+01	
		2.0E-02	1 3 1E±01	A V	-0.24	1	0.5	Voc	Acetope	67-64-1					4.0E+01	2.3L+02	6 4E±03	1 4E+03	
		3.02-01	2 0E-03	x	-0.24	1	1	Yes	Acetone Cvanobydrin	75-86-5					1.02+03	4.42+03	0.42+03	1.42+03	
			6.0E-02	ÎV	-0.34	1	1	Yes	Acetonitrile	75-05-8							1.3E+01	1.3E+01	
		1.0E-01	1	V	1.58	1	1	Yes	Acetophenone	98-86-2					2.0E+02	4.6E+03		1.9E+02	
3.8E+00	C 1.3E-03 C				3.12	1	1	Yes	Acetylaminofluorene, 2-	53-96-3	2.1E-02	6.7E-02		1.6E-02					
		5.0E-04	I 2.0E-05	IV	-0.01	1	1	Yes	Acrolein	107-02-8					1.0E+00	1.7E+02	4.2E-03	4.2E-03	
5.0E-01	I 1.0E-04 I	2.0E-03	I 6.0E-03	I M	-0.67	1	1	Yes	Acrylamide	79-06-1	5.0E-02	2.3E+01		5.0E-02	4.0E+00	2.1E+03		4.0E+00	
		5.0E-01	I 1.0E-03	IV	0.35	1	1	Yes	Acrylic Acid	79-10-7					1.0E+03	1.1E+05	2.1E-01	2.1E-01	
5.4E-01	I 6.8E-05 I	4.0E-02	A 2.0E-03	IV	0.25	1	1	Yes	Acrylonitrile	107-13-1	1.4E-01	1.4E+01	8.3E-02	5.2E-02	8.0E+01	8.9E+03	4.2E-01	4.1E-01	
	_		6.0E-03	Р	-0.32	1	1	Yes	Adiponitrile	111-69-3									
5.6E-02	С	1.0E-02	1		3.52	1	0.9	Yes	Alachlor	15972-60-8	1.4E+00	4.4E+00		1.1E+00	2.0E+01	6.9E+01		1.6E+01	2.0E+00
		1.0E-03	-		1.13	1	1	Yes	Aldicarb	116-06-3					2.0E+00	1.4E+02		2.0E+00	3.0E+00
		1.0E-03	1		-0.57	1	1	Yes	Aldicarb Sulfone	1646-88-4					2.0E+00	2.4E+03		2.0E+00	2.0E+00
1 7E+01	1 4 9E-03 1	3 0E-05		V	-0.78	1	1	Yes	Aldicard sulfoxide	1646-87-3	4.6E-03		1 15-03	9.2E-04	6.0E-02			6 0E-02	4.0E+00
1.7 2401	1 4.32-03 1	5.0E-03	1 1 0E-04	¥ V	0.3	1	1	Voc		107-18-6	4.02-03		1.12-03	3.22-04	1.0E±01	1 3E±03	2 1E-02	2.1E-02	
2 1E-02	C 60E-06 C	3.0L-03	1 0E-04	I V	1 93	1	1	Yes		107-05-1	3.7E±00	3.5E±01	9.4E-01	7 3E-01	1.02+01	1.52+05	2.1E-02 2.1E-01	2.1E-02 2.1E-01	
2.12 02	0 0.02 00 0	1.0E+00	P 5.0E-03	P	1.55	1	1	Yes	Aluminum	7429-90-5	0.7 2100	0.02101	0.42 01	1.52 01	2.0E+03	4.6E+05	2.12 01	2.0E+03	
		4.0E-04	1			1	1	Yes	Aluminum Phosphide	20859-73-8					8.0E-01	1.8E+02		8.0E-01	
		9.0E-03	i		2.98	1	1	Yes	Ametryn	834-12-8					1.8E+01	9.8E+01		1.5E+01	
2.1E+01	C 6.0E-03 C				2.86	1	1	Yes	Aminobiphenyl, 4-	92-67-1	3.7E-03	1.5E-02		3.0E-03					
		8.0E-02	Р		0.21	1	1	Yes	Aminophenol, m-	591-27-5					1.6E+02	2.8E+04		1.6E+02	
		4.0E-03	х		0.62	1	1	Yes	Aminophenol, o-	95-55-6					8.0E+00	7.5E+02		7.9E+00	
		2.0E-02	Р		0.04	1	1	Yes	Aminophenol, p	123-30-8					4.0E+01	9.1E+03		4.0E+01	
		2.5E-03	1		5.5	1	0.9	Yes	Amitraz	33089-61-1					5.0E+00	9.8E-01		8.2E-01	
			5.0E-01	IV	0.23	1	1		Ammonia	7664-41-7									
		2.0E-01	I			1	1	Yes	Ammonium Sulfamate () 🔥 👘 👘 👘	7773-06-0					4.0E+02	9.1E+04		4.0E+02	
			3.0E-03	X V	0.89	1	1	Yes	Amyl Alcohol, tert-	75-85-4	=				=		6.3E-01	6.3E-01	
5.7E-03	I 1.6E-06 C	7.0E-03	P 1.0E-03	1	0.9	1	1	Yes	Aniline	62-53-3	1.4E+01	6.9E+02		1.3E+01	1.4E+01	7.7E+02		1.4E+01	
4.0E-02	F	2.0E-03	<u>^</u>		3.39	0.15	0.9	Vec	Antimaquinone, 9,10-	7440.26.0	1.9E+00	3.1E+00		1.4E+00	4.0E+00	2.7E+01		3.0E+00	6 0E 100
		4.0E-04	і Ц			0.15	1	Yes	Antimony (metallic)	1314-60-0					8.0E-01	2.7E+01		0.7E-01	6.0E+00
		4.0E-04	н			0.15	1	Yes	Antimony Tetroxide	1332-81-6					8.0E-01	2.7E+01		7.8E-01	
			2.0E-04	1		0.15	1	Yes		1309-64-4									
1.5E+00	4.3E-03	3.0E-04	I 1.5E-05	c		1	1	Yes	Arsenic, Inorganic	7440-38-2	5.2E-02	9.7E+00		5.2E-02	6.0E-01	1.4E+02		6.0E-01	1.0E+01
		3.5E-06	C 5.0E-05	i -		1	1	Yes	Arsine	7784-42-1					7.0E-03	1.6E+00		7.0E-03	
		3.6E-02	0		-0.27	1	1	Yes	Asulam	3337-71-1					7.2E+01	5.8E+04		7.2E+01	
2.3E-01	С	3.5E-02	1		2.61	1	1	Yes	Atrazine	1912-24-9	3.4E-01	2.8E+00		3.0E-01	7.0E+01	6.2E+02		6.3E+01	3.0E+00
8.8E-01	C 2.5E-04 C				2.98	1	0.9	Yes	Auramine	492-80-8	8.9E-02	2.7E-01		6.7E-02					
		4.0E-04			4.48	1	1	No	Avermectin B1	65195-55-3					8.0E-01			8.0E-01	
4.45.01	1 245 05 1	3.0E-03	A 1.0E-02	A	2.75	1	1	Yes	Azinphos-methyl	86-50-0	745.01	7 05 04	4.05.01	4.05.04	6.0E+00	8.3E+01		5.6E+00	
1.1E-01	1 3.1E-05 I	1.05.00	D 7 0E 00	V	3.82	1	1	Yes	Azouenzene	103-33-3	7.1E-01	7.3E-01	1.6E-01	1.2E-01	2.0E+02	6 9E 100		2.05.02	
		2 0E-01	1 5 0E-04	н	-1.7	0.07	1	Yes	Barium	7440,30-3					2.0E+03 4.0E+02	6.4E+03		2.02+03 3.8E±02	2 0E+03
		5.0E-01	0	v	5.29	1	0.8	Yes	Benfluralin	1861-40-1					1.0E+02	4.0E+00		2.8E+02	2.00+03
		5.0E-02	1		2 12	1	1	Yes	Benomyl	17804-35-2					1.0E+02	3.0E+03		9.7E+01	
		2.0E-01	i		2.18	1	1	Yes	Bensulfuron-methyl	83055-99-6					4.0E+02	2.4E+04		3.9E+02	
		3.0E-02	1		2.34	1	1	Yes	Bentazon	25057-89-0					6.0E+01	9.4E+02		5.7E+01	
4.0E-03	Р	1.0E-01	1	V	1.48	1	1	Yes	Benzaldehyde	100-52-7	1.9E+01	4.4E+02		1.9E+01	2.0E+02	4.9E+03		1.9E+02	
5.5E-02	I 7.8E-06 I	4.0E-03	I 3.0E-02	IV	2.13	1	1	Yes	Benzene	71-43-2	1.4E+00	9.8E+00	7.2E-01	4.6E-01	8.0E+00	6.1E+01	6.3E+00	3.3E+00	5.0E+00
1.0E-01	Х	3.0E-04	Х		-3.7267	1	1	No	Benzenediamine-2-methyl sulfate, 1,4-	6369-59-1	7.8E-01			7.8E-01	6.0E-01			6.0E-01	
		1.0E-03	P	V	2.52	1	1	Yes	Benzenethiol	108-98-5					2.0E+00	1.0E+01		1.7E+00	
2.3E+02	I 6.7E-02 I	3.0E-03	1	M	1.34	1	1	Yes	Benzidine	92-87-5	1.1E-04	5.0E-03		1.1E-04	6.0E+00	3.0E+02		5.9E+00	
4.05.04		4.0E+00	1	M	1.87	1	1	Yes	Denzolic Acid	65-85-0	0.05.00			2.05.00	8.0E+03	1.2E+05		7.5E+03	
1.3E+01	1	1.05.04	D	v	3.9	1	1	Yes	Benzolinchionae	98-07-7	6.0E-03	6.0E-03		3.0E-03	2.05.02	8 0E - 00		2.05.02	
1.7E-01	1 4.9E-05 C	2.0E-01	P 1.0E-03	ΡV	2.3	1	1	Yes	Benzyl Chloride	100-51-6	4.6E-01	3.4E+00	1.1E-01	8.9E-02	2.0E+02 4.0E+00	3.2E+03	2.1E-01	2.0E+02 2.0E-01	
	2.4E-03	2.0E-03	1 2.0E-05	1	2.0	0.007	1	Yes	Bervilium and compounds	7440-41-7		0.12.00		0.02 02	4.0E+00	6.4E+00		2.5E+00	4.0E+00
	2.42 00 1	9.0E-03	P	•	4.48	1	0.9	Yes	Bifenox	42576-02-3					1.8E+01	2.3E+01		1.0E+01	1.02100
		1.5E-02	1		6	1	0	Yes	Biphenthrin	82657-04-3					3.0E+01			3.0E+01	
8.0E-03	1	5.0E-01	I 4.0E-04	ΧV	4.01	1	1	Yes	Biphenyl, 1,1'-	92-52-4	9.7E+00	6.5E+00		3.9E+00	1.0E+03	7.3E+02	8.3E-02	8.3E-02	
		4.0E-02	1	V	2.48	1	1	Yes	Bis(2-chloro-1-methylethyl) ether	108-60-1					8.0E+01	6.5E+02		7.1E+01	
		3.0E-03	Р		1.3	1	1	Yes	Bis(2-chloroethoxy)methane	111-91-1					6.0E+00	3.0E+02		5.9E+00	
1.1E+00	1 3.3E-04 I			V	1.29	1	1	Yes	Bis(2-chloroethyl)ether	111-44-4	7.1E-02	2.7E+00	1.7E-02	1.4E-02					

Key: I = IRIS	s; P = PPRTV; D	= DWSHA; C	) = OPP; A = 4	ATSDR; C	= Cal EPA	A; X = Af	PPEND		TV SCREEN (See FAQ #29); H = HEAST; F = See FAQ; E = see user guide Section	2.3.5; W = see	user guide Se	ction 2.3.6; l	_ = see user gu	ide on lead; M = n	nutagen; S = s	ee user guid	e Section 5; \	/ = volatile; R = RBA a	pplied (See
		User Guid oxicity and C	tor Arsenic r	notice) ; c = fic Informa	= cancer; r tion	1 = nonc	ancer;	" = whei	e: n SL < 100X c SL; ** = where n SL < 10X c SL; SSL values are based on DAF=1; r Contaminant	m = Concentrati	on may excee Ca	ed ceiling iim rcinogenic Ta	arget Risk (TR)	= 1E-06	ration may exi	Noncance	ee User Guide er CHILD Haza	e) ard Index (HI) = 0.1	
			ПТТ	П								I			Ingestion SL	Dermal SL	Inhalation SL	Noncarcinogenic SL	
SEO	k k	RfD	k RfCi	k v							Ingestion SL	Dermal SL	Inhalation SL	Carcinogenic SL	Child	Child	Child	Child	MCI
(mg/kg-day) <sup>-1</sup>	y (ug/m <sup>3</sup> ) <sup>-1</sup> y	(mg/kg-day)	y $(mq/m^3)$	y I gen	LOGP	GIABS	FA I	n EPD?	Analyte	CAS No.	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)
2.2E+02	1 6.2E-02 I	( 5 5 . ))	, ( 5 ,	V	0.57	1	1	Yes	Bis(chloromethyl)ether	542-88-1	3.5E-04	3.4E-02	9.1E-05	7.2E-05	(5)	(137)	(10)	(13)	(.57)
		5.0E-02	1		3.32	1	1	Yes	Bisphenol A	80-05-7					1.0E+02	3.2E+02		7.7E+01	
		2.0E-01	I 2.0E-02	Н		1	1	Yes	Boron And Borates Only	7440-42-8					4.0E+02	9.1E+04		4.0E+02	
		2.0E+00	P 2.0E-02	PV	1.16	1	1	Yes	Boron Trichloride	10294-34-5					4.0E+03	9.1E+05	4.2E+00	4.2E+00	
7.0E-01	1	4.0E-02	1 I.3E-02	C V	0.22	1	1	Ves	Bromate	15541-45-4	1 1E-01	2 1E+01		1 1E-01	8.0E+01	1.8E±03	2.7E+00	2.0E+00 8.0E+00	1.0E±01
2.0E+00	X 6.0E-04 X	4.02-03	'	v	1.92	1	1	Yes	Bromo-2-chloroethane, 1-	107-04-0	3.9E-02	5.7E-01	9.4E-03	7.4E-03	0.02+00	1.02+03		0.02+00	1.02+01
		3.0E-04	х	V	2.92	1	1	Yes	Bromo-3-fluorobenzene, 1-	1073-06-9					6.0E-01	2.6E+00		4.9E-01	
		3.0E-04	Х	V	3.08	1	1	Yes	Bromo-4-fluorobenzene, 1-	460-00-4					6.0E-01	2.0E+00		4.6E-01	
		8.0E-03	I 6.0E-02		2.99	1	1	Yes	Bromobenzene	108-86-1					1.6E+01	5.4E+01	1.3E+01	6.2E+00	
6 2E 02	1 3 7E 0F C	2.0E.02	4.0E-02	X V	1.41	1	1	Yes	Bromochioromethane	74-97-5	1.25,00	1.05.01	1 55 01	1 25 01	4.05+01	6 EE 102	0.3E+00	8.3E+00	9 0E 101/E)
7.9E-03	1 1.1E-06 L	2.0E-02 2.0E-02	1	v	2.4	1	1	Yes	Bromoform	75-25-2	9.9E+00	1.4E+02	5.1E+00	3.3E+00	4.0E+01	6.2E+02		3.8E+01	8.0E+01(F)
		1.4E-03	I 5.0E-03	I V	1.19	1	1	Yes	Bromomethane	74-83-9					2.8E+00	1.0E+02	1.0E+00	7.5E-01	
		5.0E-03	Н	V	5.21	1	0.8	Yes	Bromophos	2104-96-3					1.0E+01	5.5E+00		3.5E+00	
			1.0E-01	ΑV	2.1	1	1	Yes	Bromopropane, 1-	106-94-5							2.1E+01	2.1E+01	
1.0E-01	0	1.5E-02	0	V	2.8	1	0.9	Yes	Bromoxynil Bromoxynil Octoposto	1689-84-5	7.6E-01	3.1E+00		6.1E-01	3.0E+01	1.3E+02		2.5E+01	
3.4E+00	C 3.0E-05	1.56-02	2.0E-03	I V	1.99	1	0.8	Yes	Bromoxym Octanoate Butadiene, 1.3-	106-99-0	2.3E-02	1.6E-01	1.9E-01	1.8E-02	3.02+01	1.02+01	4.2E-01	4.2E-01	
0.4E100	0 0.02 00 1	3.0E-02	0		3.53	1	0.9	Yes	Butanoic acid, 4-(2,4-dichlorophenoxy)-	94-82-6	2.02 02	1.02 01	1.52 01	1.02 02	6.0E+01	1.8E+02	4.22 01	4.5E+01	
		1.0E-01	I	V	0.88	1	1	Yes	Butanol, N-	71-36-3					2.0E+02	1.0E+04		2.0E+02	
		2.0E+00	P 3.0E+01	ΡV	0.61	1	1	Yes	Butyl alcohol, sec-	78-92-2					4.0E+03	3.0E+05	6.3E+03	2.4E+03	
0.05.04	0.575.00.0	5.0E-02		V	4.15	1	1	Yes	Butylate	2008-41-5	0.05.00	0.55.00		4.55.00	1.0E+02	8.5E+01		4.6E+01	
2.0E-04	C 5.7E-08 C	3.0E-01	D		3.5	1	0.8	Yes	Butylated hydroxytaluona	25013-16-5	3.9E+02	2.5E+02		1.5E+02 3.4E+00	6 0E±02	1 25+02		1.05+02	
3.0L-03		5.0E-01	P	V	4.38	1	1	No	Butylbenzene, n-	104-51-8	2.26401	4.02+00		3.42+00	1.0E+02	1.22702		1.0E+02	
		1.0E-01	Х	V	4.57	1	1	No	Butylbenzene, sec-	135-98-8					2.0E+02			2.0E+02	
		1.0E-01	Х	V	4.11	1	1	Yes	Butylbenzene, tert-	98-06-6					2.0E+02	1.1E+02		6.9E+01	
		2.0E-02	A		0.36	1	1	Yes	Cacodylic Apid	75-60-5					4.0E+01	6.7E+03		4.0E+01	
	1.8E-03 I	1.0E-03	I 1.0E-05	A ^		0.025	1	Voc	Cadmium (Diet)	7440-43-9					1.0E+00	1 15+01		0.2E-01	5 0E+00
	1.62-03 1	5.0E-04 5.0E-01	I 2.2E-03	C C	-0.19	1	1	Yes		105-60-2					1.0E+00 1.0E+03	9.0E+04		9.9E+02	5.0E+00
1.5E-01	C 4.3E-05 C	2.0E-03	1	-	3.8	1	0.9	Yes	Captafol U U V cellas Lanas V3/ V3/	2425-06-1	5.2E-01	1.8E+00		4.0E-01	4.0E+00	1.5E+01		3.2E+00	
2.3E-03	C 6.6E-07 C	1.3E-01	1		2.8	1	1	Yes	Captan	133-06-2	3.4E+01	3.6E+02		3.1E+01	2.6E+02	3.0E+03		2.4E+02	
		1.0E-01	-		2.36	1	1	Yes	Carbaryl	63-25-2					2.0E+02	2.4E+03		1.8E+02	
		5.0E-03		1.17	2.32	1	1	Yes	Carbon Disultate Control Contr	1563-66-2					1.0E+01	1.4E+02	1 55 . 02	9.4E+00	4.0E+01
7.0E-02	I 6.0E-06 I	4.0E-01	I 1.0E-01	IV	2.83	1	1	Yes	Carbon Tetrachloride	56-23-5	1.1E+00	4.3E+00	9.4E-01	4.6E-01	8.0E+02	2.0E+03 3.4E+01	2.1E+02	4.9E+00	5.0E+00
			1.0E-01	ΡV	-1.33	1	1	Yes	Carbonyl Sulfide	463-58-1							2.1E+01	2.1E+01	
		1.0E-02	1		5.57	1	0.8	Yes	Carbosulfan U U U VSK VS/ 0 cd.b	55285-14-8					2.0E+01	6.9E+00		5.1E+00	
		1.0E-01			2.14	1	1	Yes	Carboxin	5234-68-4					2.0E+02	4.1E+03		1.9E+02	
		1 0E 01	9.0E-04	1	0.00	1	1	Yes	Ceric oxide	1306-38-3					2.05.02	1 55 . 04		2.05.02	
		1.5E-01	i	v	1.9	1	1	Yes	Chloramben	133-90-4					3.0E+02	7.4E+02		2.9E+02 2.9E+01	
4.0E-01	Н				2.22	1	1	Yes	Chloranil	118-75-2	1.9E-01	3.5E+00		1.8E-01					
3.5E-01	I 1.0E-04 I	5.0E-04	I 7.0E-04	IV	6.16	1	0.7	Yes	Chlordane	12789-03-6	2.2E-01	3.6E-02	5.6E-02	2.0E-02	1.0E+00	1.8E-01	1.5E-01	7.4E-02	2.0E+00
1.0E+01	I 4.6E-03 C	3.0E-04	1		5.41	1	0.8	Yes	Chlordecone (Kepone)	143-50-0	7.8E-03	6.5E-03		3.5E-03	6.0E-01	5.4E-01		2.9E-01	
		7.0E-04 9.0E-02	0		2.5	1	0.9	res Yes	Chlorimuron, Ethyl-	470-90-6 90982-32-4					1.4E+00 1.8E+02	5.6E+00 6.8E+03		1.1E+00 1.8E+02	
		1.0E-01	I 1.5E-04	ΑV	0.85	1	1	Yes	Chlorine	7782-50-5					2.0E+02	4.6E+04	3.0E-02	3.0E-02	
		3.0E-02	I 2.0E-04	IV		1	1	Yes	Chlorine Dioxide	10049-04-4					6.0E+01	1.4E+04	4.2E-02	4.2E-02	
		3.0E-02	1			1	1	Yes	Chlorite (Sodium Salt)	7758-19-2					6.0E+01	1.4E+04		6.0E+01	1.0E+03
	2.05.04	2.05.02	5.0E+01		2.05	1	1	Yes	Chloro-1,1-difluoroethane, 1-	/5-68-3			4.05.00	1.05.00	4.05.04	4.05.00	1.0E+04	1.0E+04	
4 6E-01	3.0E-04 I	2.0E-02	H 2.0E-02	I V	2.53	1	1	Yes	Chloro-1,3-butadiene, 2- Chloro-2-methylaniline HCL 4-	3165-93-3	1 7E-01	5 1E+02	1.9E-02	1.9E-02 1.7E-01	4.0E+01	1.6E+02	4.2E+00	3.7E+00	
1.0E-01	P 7.7E-05 C	3.0E-03	х		2.27	1	1	Yes	Chloro-2-methylaniline, 4-	95-69-2	7.8E-01	6.6E+00		7.0E-01	6.0E+00	5.6E+01		5.4E+00	
2.7E-01	Х			V	0.09	1	1	Yes	Chloroacetaldehyde, 2-	107-20-0	2.9E-01	4.6E+01		2.9E-01					
			0.05.65		0.22	1	1	Yes	Chloroacetic Acid	79-11-8									6.0E+01
0.05.04	D	4.05.00	3.0E-05	1	1.93	1	1	Yes	Chloroacetophenone, 2-	532-27-4	2.05.04	5.05.00		0.75.04	0.05.00	4.05.00		7.05 - 00	
2.0E-01	٢	4.0E-03 2.0E-02	1 5 0E-02	ΡV	2.84	1	1	res Yes	Chlorobenzene	106-47-8	3.9E-01	5.9E+00		3.7E-01	8.0E+00 4.0E+01	1.3E+02 1.3E+02	1.0E+01	7.6E+00 7.8E+00	1.0E+02
		1.0E-01	X X		-0.52	1	1	Yes	Chlorobenzene sulfonic acid, p-	98-66-8					2.0E+02	1.8E+05		2.0E+02	1.02.02
1.1E-01	C 3.1E-05 C	2.0E-02	1		4.74	1	0.8	Yes	Chlorobenzilate	510-15-6	7.1E-01	5.6E-01		3.1E-01	4.0E+01	3.5E+01		1.9E+01	
		3.0E-02	X	D 1/	2.65	1	1	Yes	Chlorobenzoic Acid, p-	74-11-3					6.0E+01	3.4E+02	0.05.01	5.1E+01	
		3.0E-03	P 3.0E-01	r v V	3.6	1	1	Yes	Chlorobutano, 1-	98-56-6					6.0E+00	9.3E+00	6.3E+01	3.5E+00	
		4.02-02	5.0E+01	ĪV	1.08	1	1	Yes	Chlorodifluoromethane	75-45-6					0.02+01	3.12+02	1.0E+04	1.0E+04	

Key: I = IRIS	S; P = PPRTV; D	= DWSHA; C	) = OPP; A = AT	SDR; C =	Cal EPA	; X = API	PENDIX PPF	RTV SCREEN (See FAQ #29); H = HEAST; F = See FAQ; E = see user guide Section	2.3.5; W = see	user guide Se	ction 2.3.6; L	. = see user gu	ide on lead; M = r	nutagen; S = s	see user guid	le Section 5; V	= volatile; R = RBA a	pplied (See
	т	User Guid	e for Arsenic not hemical-specific	tice) ; c = (	cancer; n	= noncal	ncer; * = whe	rre: n SL < 100X c SL; ** = where n SL < 10X c SL; SSL values are based on DAF=1; r Contaminant	n = Concentrati	on may excee	d ceiling limi	t (See User Gi arget Risk (TR)	uide); s = Concent = 1E-06	ration may ex	Noncanci	See User Guide er CHILD Haza	) rd Index (HI) = 0.1	
	TT TÏ	oniony and o						Containmaint		ou.			- 12 00	Ingestion SL	Dermal SL	Inhalation SL	Noncarcinogenic SL	
SEO	k k	DID	k k	v						Ingestion SL	Dermal SL	Inhalation SL	Carcinogenic SL	Child	Child	Child	Child	
5FU (mg/kg-day) <sup>-1</sup>	e IUR e	(ma/ka_day)	e (mg/m <sup>3</sup> ) v	o muta-	LOGP	GIABS		Analyte	CAS No	TR=1E-06	TR=1E-06	TR=1E-06	TR=1E-06	THQ=0.1	THQ=0.1	THQ=0.1	THI=0.1	MCL (ug/L)
(IIIg/kg-uay)	y (ug/iii ) y	2 0E-02	P (IIIg/III ) y	v gen	0.03	1	1 Yes	Chloroethanol 2-	107-07-3	(ug/L)	(ug/L)	(ug/L)	(ug/L)	4 0E+01	7 7E+03	(ug/L)	4 0E+01	(ug/L)
3.1E-02	C 2.3E-05 1	1.0E-02	1 9.8E-02 A	v	1.97	1	1 Yes	Chloroform	67-66-3	2.5E+00	2.9E+01	2.4E-01	2.2E-01	2.0E+01	2.5E+02	2.0E+01	9.7E+00	8.0E+01(F)
0.12 02	0 2.02 00 1	1.02 02	9.0E-02 I	v	0.91	1	1 Yes	Chloromethane	74-87-3	2.02100	2.02.001	2.12.01	2.22 01	2.02101	2.02.02	1.9E+01	1.9E+01	0.02101(1)
2.4E+00	C 6.9E-04 C			V	0.32	1	1 Yes	Chloromethyl Methyl Ether	107-30-2	3.2E-02	3.7E+00	8.1E-03	6.5E-03					
3.0E-01	Р	3.0E-03	P 1.0E-05 X		2.24	1	1 Yes	Chloronitrobenzene, o-	88-73-3	2.6E-01	2.6E+00		2.4E-01	6.0E+00	6.4E+01		5.5E+00	
6.0E-02	Р	7.0E-04	P 2.0E-03 P	V	2.39	1	1 Yes	Chloronitrobenzene, p-	100-00-5	1.3E+00	1.0E+01		1.2E+00	1.4E+00	1.2E+01		1.3E+00	
		5.0E-03	1 4.0E-04_C	V	2.15	1	1 Yes	Chloropicrip	90-07-8					1.0E+01	1.0E+02	8 3E-02	9.1E+00 8.3E-02	
3.1E-03	C 8.9E-07 C	1.5E-02	4.02-04 C	v	3.05	1 (	).9 Yes	Chlorothalonil	1897-45-6	2.5E+01	1.6E+02		2.2E+01	3.0E+01	2.1E+02	0.52-02	2.6E+01	
0.12 00	0 0.02 07 0	2.0E-02	i	v	3.42	1	1 Yes	Chlorotoluene, o-	95-49-8	2.02101			2.22.01	4.0E+01	5.8E+01		2.4E+01	
		2.0E-02	Х	V	3.33	1	1 Yes	Chlorotoluene, p-	106-43-4					4.0E+01	6.6E+01		2.5E+01	
2.4E+02	C 6.9E-02 C				-1.02	1	1 Yes	Chlorozotocin	54749-90-5	3.2E-04	1.0E+00		3.2E-04					
		5.0E-02	0		3.51	1 (	).9 Yes	Chlorpropham	101-21-3					1.0E+02	2.5E+02		7.1E+01	
		1.0E-03	A		4.96	1 (	).8 Yes	Chlorpyritos	2921-88-2					2.0E+00	1.5E+00		8.4E-01	
		2.0E-02	0		4.31	1	1 Yes	Chlorsulfuron	64902-72-3					4.0E+01	2.3E+01		3.9E+01	
		1.0E-02	1		4.28	1 (	).9 Yes	Chlorthal-dimethyl	1861-32-1					2.0E+01	3.3E+01		1.2E+01	
		8.0E-04	н		5.8	1 (	0.8 Yes	Chlorthiophos	60238-56-4					1.6E+00	3.4E-01		2.8E-01	
		1.5E+00	1			0.013	1 Yes	Chromium(III), Insoluble Salts	16065-83-1					3.0E+03	8.9E+03		2.2E+03	
5.0E-01	C 8.4E-02 S	3.0E-03	I 1.0E-04 I	М		0.025	1 Yes	Chromium(VI)	18540-29-9	5.0E-02	1.2E-01		3.5E-02	6.0E+00	1.7E+01		4.4E+00	
		1 3E-02			3.1	0.013	1 Yes	Chromium, Total	7440-47-3					2 6E+01	2 1E+02		2 3E+01	1.0E+02
	9.0E-03 P	3.0E-02	P 6 0E-06 P		5.1	1	1 Yes		74113-24-3					6.0E-01	3.4E±02		6.0E-01	
	6.2E-04 I	0.02 04	1 0.02 00 1	V М		1	0	Coke Oven Emissions	8007-45-2					0.02 01	0.42102		0.02 01	
		4.0E-02	н			1	1 Yes	Copper	7440-50-8					8.0E+01	1.8E+04		8.0E+01	1.3E+03
		5.0E-02	I 6.0E-01 C		1.96	1	1 Yes	Cresol, m-	108-39-4					1.0E+02	1.2E+03		9.3E+01	
		5.0E-02	I 6.0E-01 C		1.95	1	1 Yes	Cresol, o-	95-48-7					1.0E+02	1.2E+03		9.3E+01	
		1.0E-01	A 6.0E-01 C		1.94	1	1 Yes	Cresol, p-	106-44-5					2.0E+02	2.5E+03		1.9E+02	
		1.0E-01 1.0E-01	A 6.0E-01 C		3.1	1	1 Yes	Cresols	59-50-7					2.0E+02 2.0E+02	5.2E+02 6.7E+02		1.4E+02 1.5E+02	
1.9E+00	н	1.0E-03	P	V	0.6	1	1 Yes	Crotonaldehyde, trahş-	123-73-9	4.1E-02	2.7E+00		4.0E-02	2.0E+02	1.5E+02		2.0E+00	
		1.0E-01	I 4.0E-01 I	V	3.66	1	1 Yes	Cumene	98-82-8					2.0E+02	1.9E+02	8.3E+01	4.5E+01	
2.2E-01	C 6.3E-05 C				-1.73	1	1 Yes	Cupferron	135-20-6	3.5E-01	1.3E+04		3.5E-01					
8.4E-01	Н	2.0E-03	н		2.22	1	1 Yes	Cyanazine	21725-46-2	9.3E-02	1.6E+00		8.8E-02	4.0E+00	7.6E+01		3.8E+00	L
		1 0E-03				1	1 Voc		502-01-8					2 0E+00	4 6E+02		2 0E+00	
		5.0E-03	i			1	1 Yes	~Copper Cyanide	544-92-3					1.0E+01	2.3E+03		1.0E+01	
		6.0E-04	I 8.0E-04 S	V		1	1 Yes	~Cyanide (CN-)	57-12-5					1.2E+00	2.7E+02	1.7E-01	1.5E-01	2.0E+02
		1.0E-03	1	V	0.07	1	1 Yes	-Cyanogen U U U Control Contro	460-19-5					2.0E+00	5.1E+02		2.0E+00	
		9.0E-02	1	V		1	1 Yes	~Cyanogen Bromide	506-68-3					1.8E+02	1.6E+05		1.8E+02	
		5.0E-02		V	0.25	1	1 Yes	~Cyanogen Chloride	506-77-4					1.0E+02	5.8E+04	1 75 01	1.0E+02	
		2.0E-04	1 0.0E-04 1	v	-0.25	1	1 Yes	~Potassium Cvanide	151-50-8					4.0E+00	4.6E+02	1.72-01	4.0E+00	
		5.0E-03	1			0.04	1 Yes	~Potassium Silver Cyanide	506-61-6					1.0E+01	4.6E+01		8.2E+00	
		1.0E-01	1			0.04	1 Yes	~Silver Cyanide	506-64-9					2.0E+02	1.8E+03		1.8E+02	
		1.0E-03	1			1	1 Yes	~Sodium Cyanide	143-33-9					2.0E+00	4.6E+02		2.0E+00	2.0E+02
		2.0E-04	P	V	0.50	1	0 Yes	~Thiocyanates	E1790664					4.0E-01	9.1E+01		4.0E-01	
		2.0E-04 5.0E-02	X	V	0.58	1	1 Yes 1 Yes	~Thiocyanic Acid	463-56-9					4.0E-01 1.0E±02	9.1E+01 3.8E+04		4.0E-01 1.0E+02	
		0.02 02	6.0E+00	V	3.44	1	1 Yes	Cyclohexane	110-82-7					1.02102	0.02101	1.3E+03	1.3E+03	
2.0E-02	х	2.0E-02	х		4.72	1 (	).9 Yes	Cyclohexane, 1,2,3,4,5-pentabromo-6-chloro-	87-84-3	3.9E+00	9.6E+00		2.8E+00	4.0E+01	1.1E+02		2.9E+01	
		5.0E+00	I 7.0E-01 P	V	0.81	1	1 Yes	Cyclohexanone	108-94-1					1.0E+04	6.5E+05	1.5E+02	1.4E+02	
		5.0E-03	P 1.0E+00 X	V	2.86	1	1 Yes	Cyclohexene	110-83-8					1.0E+01	2.5E+01	2.1E+02	7.0E+00	
		2.0E-01	1	V	1.49	1	1 Yes	Cyclohexylamine Cyfluthrin	108-91-8					4.0E+02	9.3E+03		3.8E+02	
		1.0E-02	0		6.9	1 (	).5 No	Cyhalothrin	68085-85-8					2.0E+00	1.02701		2 0E+00	
		6.0E-02	õ		6.6	1 0	).7 No	Cypermethrin	52315-07-8					1.2E+02			1.2E+02	
		1.5E-02	0		-0.061	1	1 Yes	Cyromazine	66215-27-8					3.0E+01	2.4E+03		3.0E+01	
2.4E-01	I 6.9E-05 C	3.0E-05	х		6.02	1 (	).8 Yes	DDD, p,p`- (DDD)	72-54-8	3.2E-01	3.5E-02		3.2E-02	6.0E-02	7.1E-03		6.3E-03	
3.4E-01	I 9.7E-05 C	3.0E-04	x	V	6.51	1 (	0.8 No	DDE, p,p'-	72-55-9	2.3E-01		5.8E-02	4.6E-02	6.0E-01			6.0E-01	
3.4E-01	1 9.7E-05 I	3.0E-02	1		0.91	1 (	1 Voc	Dalapon	75-99-0	2.3E-01			2.3E-01	6.0E+00	5.5E±02		6.0E+00	2.0E+02
1.8E-02	C 5.1E-06 C	1.5E-02	i		-1.5	1	1 Yes	Daminozide	1596-84-5	4.3E+00	1.3E+04		4.3E+00	3.0E+02	1.0E+06		3.0E+02	2.02+02
7.0E-04	1	7.0E-03	I		12.11	1	0 No	Decabromodiphenyl ether, 2,2',3,3',4,4',5,5',6,6'- (BDE-209)	1163-19-5	1.1E+02			1.1E+02	1.4E+01			1.4E+01	
		4.0E-05	I		3.21	1 (	).8 Yes	Demeton	8065-48-3					8.0E-02	8.8E-02		4.2E-02	
1.2E-03	1	6.0E-01	I		6.11	1	0 Yes	Di(2-ethylhexyl)adipate	103-23-1	6.5E+01			6.5E+01	1.2E+03			1.2E+03	4.0E+02
6.1E-02	Н				4.49	1 (	J.9 Yes	Diallate	2303-16-4	1.3E+00	9.2E-01		5.4E-01					

Key: I = IRI	S; P = PPRTV; D	D = DWSHA;	O = OPP; A = 4	ATSDR; (	C = Cal EPA	A; X = AP	PENDIX F	PPRT	V SCREEN (See FAQ #29); H = HEAST; F = See FAQ; E = see user guide Section	2.3.5; W = see	user guide Se	ction 2.3.6; L	= see user gu	ide on lead; M = n	nutagen; S = s	ee user guid	le Section 5; V	= volatile; R = RBA a	pplied (See
		Toxicity and C	chemical-specif	fic Inform	ation	1 = nonca	ncer; = v	where	Contaminant	m = Concentrati	on may excee Car	cinogenic Ta	raet Risk (TR)	= 1E-06	ration may exi	Noncance	ee User Guide er CHILD Haza	) rd Index (HI) = 0.1	
			TT T	T											Ingestion SL	Dermal SL	Inhalation SL	Noncarcinogenic SL	
SEO	k k	DfD	k RfC	k v							Ingestion SL	Dermal SL	Inhalation SL	Carcinogenic SL	Child	Child	Child	Child	
(mg/kg-day)	e IOK e	(ma/ka-dav)	e (mg/m <sup>3</sup> )	e o mut	a-	GIABS		PD2	Analyte	CAS No	IR=1E-06	IR=1E-06	IR=1E-06	IR=1E-06 (ug/L)	IHQ=0.1	1HQ=0.1	1 HQ=0.1	I HI=0.1	MCL (ug/L)
(IIIg/kg-uay)	y (ug/iii ) y	(IIIg/Kg-uay)	, y (iiig/iii )	y i gei	2.01					222 41 E	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	2.05.00	(ug/L)	1.05+00	(ug/L)
		1.0E-04	x	v	4.38	1	0.9 Te 1 Ye		Dipenzothionhene	132-65-0					2.0E+01	9.6E+00		6.5E+00	
8.0E-01	P 6.0E-03 P	2.0E-04	P 2.0E-04	ιν Μ	2.96	1	1 Ye	es D	Dibromo-3-chloropropane, 1,2-	96-12-8	3.1E-02	1.7E-01	3.4E-04	3.3E-04	4.0E-01	2.4E+00	4.2E-02	3.7E-02	2.0E-01
		4.0E-04	Х	V	3.75	1	0.9 Ye	es D	Dibromobenzene, 1,3-	108-36-1					8.0E-01	1.6E+00		5.3E-01	
		1.0E-02	1	V	3.79	1	0.9 Ye	es D	Dibromobenzene, 1,4-	106-37-6					2.0E+01	3.7E+01		1.3E+01	
8.4E-02	I	2.0E-02	I	V	2.16	1	1 Ye	es D	Dibromochloromethane	124-48-1	9.3E-01	1.4E+01		8.7E-01	4.0E+01	6.7E+02		3.8E+01	8.0E+01(F)
2.0E+00	I 6.0E-04 I	9.0E-03	I 9.0E-03	IV	1.96	1	1 Ye	es D	Dibromoethane, 1,2-	106-93-4	3.9E-02	7.1E-01	9.4E-03	7.5E-03	1.8E+01	3.6E+02	1.9E+00	1.7E+00	5.0E-02
		2.05.04	4.0E-03	хv	1.7	1	1 Ye	es D	Dibromomethane (Methylene Bromide)	74-95-3					0.05.04		8.3E-01	8.3E-01	
		3.0E-04	F		2.21	1	1 Vo		Neombo	1018 00 0					6.0E+01	1.05.02		6.0E-01	
	4.2E-03 P	5.0L-02		v	2.6	1	1 Ye	as D	Dichloro-2-butene, 1.4-	764-41-0			1.3E-03	1.3E-03	0.02+01	1.02+03		3.72+01	
	4.2E-03 P			v	2.6	1	1 Ye	es D	Dichloro-2-butene, cis-1,4-	1476-11-5			1.3E-03	1.3E-03					
	4.2E-03 P	1		V	2.6	1	1 Ye	es D	Dichloro-2-butene, trans-1,4-	110-57-6			1.3E-03	1.3E-03					
5.0E-02	1 I	4.0E-03	1		0.92	1	1 Ye	es D	Dichloroacetic Acid	79-43-6	1.6E+00	9.6E+01		1.5E+00	8.0E+00	5.4E+02		7.9E+00	6.0E+01
		9.0E-02	I 2.0E-01	ΗV	3.43	1	1 Ye	es D	Dichlorobenzene, 1,2-	95-50-1					1.8E+02	2.9E+02	4.2E+01	3.0E+01	6.0E+02
5.4E-03	C 1.1E-05 C	7.0E-02	A 8.0E-01	IV	3.44	1	1 Ye	es D	Dichlorobenzene, 1,4-	106-46-7	1.4E+01	2.1E+01	5.1E-01	4.8E-01	1.4E+02	2.2E+02	1.7E+02	5.7E+01	7.5E+01
4.5E-01	1 3.4E-04 C	9.0E-03	x		3.51	1	ng ye	es L	Jichlorobenzialne, 3,3 - Dichlorobenzophenone, 4,4'-	91-94-1	1.7E-01	4.5E-01		1.3E-01	1.8E±01	1.4E±01		7 8E±00	
		2.0E-01	L 1.0E-01	x v	2.16	1	1 Ye	es D	Dichlorodifluoromethane	75-71-8					4 0E+02	3.8E+03	2 1E+01	2.0E+01	
5.7E-03	C 1.6E-06 C	2.0E-01	P	v	1.79	1	1 Ye	es D	Dichloroethane, 1.1-	75-34-3	1.4E+01	1.8E+02	3.5E+00	2.8E+00	4.0E+02	5.8E+03	2.12101	3.8E+02	
9.1E-02	I 2.6E-05 I	6.0E-03	X 7.0E-03	ΡV	1.48	1	1 Ye	es D	Dichloroethane, 1,2-	107-06-2	8.6E-01	1.8E+01	2.2E-01	1.7E-01	1.2E+01	2.8E+02	1.5E+00	1.3E+00	5.0E+00
		5.0E-02	I 2.0E-01	IV	2.13	1	1 Ye	es D	Dichloroethylene, 1,1-	75-35-4					1.0E+02	8.5E+02	4.2E+01	2.8E+01	7.0E+00
		2.0E-03	1	V	1.86	1	1 Ye	es D	Dichloroethylene, 1,2-cis-	156-59-2					4.0E+00	3.6E+01		3.6E+00	7.0E+01
		2.0E-02		V	2.09	1	1 Ye	es D	Dichloroethylene, 1,2-trans-	156-60-5					4.0E+01	3.6E+02		3.6E+01	1.0E+02
		3.0E-03			3.06	1	1 Ye	es D	Dichlorophenol, 2,4-	120-83-2					6.0E+00	1.9E+01		4.6E+00	7.05.04
3 7E-02	P 37E-06 P	1.0E-02	P 4 0E-03	I V	2.81	1	1 Te		Dichloropropage 1.2-	94-75-7 78-87-5	2 1E+00	2 3E±01	1 5E+00	8 5E-01	2.0E+01 8.0E+01	0.6E±02	8 3E-01	8.2E-01	7.0E+01
0.72 02	1 0.72 00 1	2.0E-02	P	V	2	1	1 Ye	es D	Dichloropropane: 1:35 (TTTS)	142-28-9	2.12100	2.02101	1.52100	0.02 01	4.0E+01	4.6E+02	0.02 01	3.7E+01	0.02100
		3.0E-03	I	•	0.78	1	1 Ye	es D	Dichloropropanol, 2,3-	616-23-9					6.0E+00	5.0E+02		5.9E+00	
1.0E-01	I 4.0E-06 I	3.0E-02	I 2.0E-02	IV	2.04	1	1 Ye	es D	Dichloropropene, 1,3-	542-75-6	7.8E-01	7.8E+00	1.4E+00	4.7E-01	6.0E+01	6.6E+02	4.2E+00	3.9E+00	
2.9E-01	I 8.3E-05 C	5.0E-04	I 5.0E-04	1	1.43	1	1 Ye	es D	Dichlorvos	62-73-7	2.7E-01	1.4E+01		2.6E-01	1.0E+00	5.6E+01		9.9E-01	
		7.0E-05	0		0	1	1 Ye	es D	Dicrotophos	141-66-2					1.4E-01	7.7E+01		1.4E-01	
4.05.04		8.0E-02	P 3.0E-04	ΧV	3.16	1	1 Ye	es L	Dicyclopentadiene	//-/3-6	1.05.02	0.75.00		4.05.00	1.6E+02	3.5E+02	6.3E-02	6.3E-02	
1.6E+01	1 4.6E-03 1 3.0E-04 C	5.0E-05	5 0E-03		5.4	1	0.6 10	25 L	Diesel Engine Exhaust	60-57-1 E17136615	4.9E-03	2.7E-03		1.6E-03	1.0E-01	6.1E-02		3.8E-02	
	3.0L-04 C	2.0E-03	P 2.0E-04	P	-1.43	1	1 Ye	es D		111-42-2					4.0E+00	8.4E+03		4.0E+00	
		3.0E-02	P 1.0E-04	P	0.56	1	1 Ye	es D	Diethylene Glycol Monobutyl Ether	112-34-5					6.0E+01	8.7E+03		6.0E+01	
		6.0E-02	P 3.0E-04	Р	-0.54	1	1 Ye	es D	Diethylene Glycol Monoethyl Ether	111-90-0					1.2E+02	7.8E+04		1.2E+02	
		1.0E-03	Р	V	0.05	1	1 Ye	es D	Diethylformamidel U U NEXX NEX O ctillio	617-84-5					2.0E+00	4.3E+02		2.0E+00	
3.5E+02	C 1.0E-01 C				5.07	1 1	0.9 Ye	es D	Diethylstilbestrol	56-53-1	2.2E-04	6.6E-05		5.1E-05					
		8.3E-02 2.0E-02	1		0.65	1	1 Ye	es D	Jitenzoquat Diflubenzuron	43222-48-6					1.7E+02 4.0E+01	7.5E+04 1.0E+02		1.7E+02 2.9E+01	
		2.02-02	4 0E±01	LV	0.00	1	1 Vo		Difluoroethane 11-	75-37-6					4.02401	1.02+02	8.3E±03	8.3E+03	
			3.0E+01	x v	2.29	1	1 Ye	es D	Difluoropropane, 2,2-	420-45-1							6.3E+03	6.3E+03	
4.4E-02	C 1.3E-05 C	:		V	3.58	1	1 Ye	es D	Dihydrosafrole	94-58-6	1.8E+00	2.3E+00	4.3E-01	3.0E-01					
			7.0E-01	ΡV	1.52	1	1 Ye	es D	Diisopropyl Ether	108-20-3							1.5E+02	1.5E+02	
		8.0E-02	1	V	1.03	1	1 Ye	es D	Diisopropyl Methylphosphonate	1445-75-6					1.6E+02	1.3E+04		1.6E+02	
		2.2E-02	0		-0.17	1	1 Ye	es D	vimetnipin Vimetheate	55290-64-7					4.4E+01	2.6E+04		4.4E+01	
1 6E+00	P	2.2E-03	0		0.78	1	1 Ye	25 L	Dimethoxybenzidine 3.3'-	119-00-4	4 95-02	1.6E+00		4 7E-02	4.4E+00	7.0E+02		4.4E+00	
1.7E-03	P	6.0E-02	Р		-0.61	1	1 Ye	es D	Dimethyl methylphosphonate	756-79-6	4.6E+01	2.8E+04		4.6E+01	1.2E+02	8.1E+04		1.2E+02	
4.6E+00	C 1.3E-03 C	:			4.58	1	1 Ye	es D	Dimethylamino azobenzene [p-]	60-11-7	1.7E-02	7.2E-03		5.0E-03					
5.8E-01	н				2.17	1	1 Ye	es D	Dimethylaniline HCl, 2,4-	21436-96-4	1.3E-01	5.2E+02		1.3E-01					
2.0E-01	Р	2.0E-03	Х		1.68	1	1 Ye	es D	Dimethylaniline, 2,4-	95-68-1	3.9E-01	7.1E+00		3.7E-01	4.0E+00	8.0E+01		3.8E+00	
2.7E-02	P	2.0E-03	1	V	2.31	1	1 Ye	es D	Dimethylaniline, N,N-	121-69-7	2.9E+00	2.0E+01		2.5E+00	4.0E+00	3.1E+01		3.5E+00	
1.1E+01	Р	1.05.04	D 305 03	LV	2.34	1	1 Ye	es D	Jimethylbenzidine, 3,3'-	119-93-7	7.1E-03	8.5E-02		6.5E-03	2 0E : 02	1.85.05	6 3E .00	6 1E .00	
		1.0E-04	X 2 0E-06	XV	-1.01	1	1 Vo		Dimethylbydrazine 11-	57-14-7					2.02+02	3.5E+03	4.2E-04	4.2E-04	
5.5E+02	C 1.6E-01 C	1.02-04	X 2.0L-00	v	-0.54	1	1 Ye	es D	Dimethylhydrazine, 1,2-	540-73-8	1.4E-04	5.0E-02	3.5E-05	2.8E-05	2.02-01	0.0L+02	4.22-04	4.22-04	
		2.0E-02	I		2.3	1	1 Ye	es D	Dimethylphenol, 2,4-	105-67-9					4.0E+01	3.1E+02		3.6E+01	
		6.0E-04	I		2.36	1	1 Ye	es D	Dimethylphenol, 2,6-	576-26-1					1.2E+00	8.5E+00		1.1E+00	
4.55.44	0.465.55	1.0E-03	I		2.23	1	1 Ye	es D	Dimethylphenol, 3,4-	95-65-8	4.75.44	0.55	105.01	0.05.01	2.0E+00	1.7E+01		1.8E+00	
4.5E-02	C 1.3E-05 C	0.05.65	V	V	2.58	1	1 Ye	es D	Dimethylvinylchloride	513-37-1	1./E+00	6.5E+00	4.3E-01	3.3E-01	1.05.01	0.05.00		1.55.01	
		8.0E-05	X		2.13	1	I Ye	es D	Dinitro-o-cresol, 4,6-	534-52-1 131-89-5					1.6E-01 4.0E+00	2.6E+00		1.5E-01 2.3E+00	
		1.0E-04	P		1.69	1	1 Ye	es D	Dinitrobenzene. 1.2-	528-29-0					2.0E-01	5.3E+00		1.9E-01	
		1.0E-04	1		1.49	1	1 Ye	es D	Dinitrobenzene, 1,3-	99-65-0					2.0E-01	7.3E+00		2.0E-01	

Key: I = IRIS;	; P = PPRTV; D	= DWSHA; C	= OPP; A = A	ATSDR; C	= Cal EP/	A; X = AP	PEND	IX PPR	TV SCREEN (See FAQ #29); H = HEAST; F = See FAQ; E = see user guide Section $p_1 = 2 \sum_{i=1}^{N} (100 \times 2 \sum_{i=1}^{N} p_i) = p_i p_i p_i p_i p_i p_i p_i p_i p_i p_i$	2.3.5; W = see	user guide Se	ction 2.3.6; L	. = see user gu	ide on lead; M = n	nutagen; S = s	ee user guid	e Section 5; V	= volatile; R = RBA a	pplied (See
	Т	oxicity and Ch	e for Arsenic r nemical-specif	fic Informa	tion	I = HUHCA	ncer,	= wrier	Contaminant	n = concentrati	Car	cinogenic Ta	arget Risk (TR)	= 1E-06	ration may ext	Noncance	er CHILD Hazar	rd Index (HI) = 0.1	
													ľ í		Ingestion SL	Dermal SL	Inhalation SL	Noncarcinogenic SL	
SEO	k k	RfD	k RfC	k v							Ingestion SL	Dermal SL	Inhalation SL	Carcinogenic SL	Child	Child	Child	Child	MCI
(mg/kg-dav) <sup>-1</sup>	v (ug/m <sup>3</sup> ) <sup>-1</sup> v	(mg/kg-dav)	$v (ma/m^3)$	v I gen	LOGP	GIABS	FA In	EPD?	Analyte	CAS No.	(ug/L)	(ua/L)	(ua/L)	(ua/L)	(ug/L)	(ua/L)	(ua/L)	(ua/L)	(ua/L)
( 3 3,/	, (, 5, 7, 7, 7,	1.0E-04	P		1.46	1	1	Yes	Dinitrobenzene, 1.4-	100-25-4	(5)	(5)	(5)	(137)	2.0E-01	7.6E+00	(13)	2.0E-01	(13)
		2.0E-03	1		1.67	1	1	Yes	Dinitrophenol, 2,4-	51-28-5					4.0E+00	1.2E+02		3.9E+00	
6.8E-01	I				2.18	1	1	Yes	Dinitrotoluene Mixture, 2,4/2,6-	E1615210	1.1E-01	1.5E+00		1.1E-01					
3.1E-01	C 8.9E-05 C	2.0E-03	1		1.98	1	1	Yes	Dinitrotoluene, 2,4-	121-14-2	2.5E-01	4.3E+00		2.4E-01	4.0E+00	7.5E+01		3.8E+00	
1.5E+00	Р	3.0E-04	X		2.1	1	1	Yes	Dinitrotoluene, 2,6-	606-20-2	5.2E-02	7.4E-01		4.9E-02	6.0E-01	9.3E+00		5.7E-01	
		2.0E-03	S		1.84	1	1	Yes	Dinitrotoluene, 2-Amino-4,6-	35572-78-2					4.0E+00	1.0E+02		3.9E+00	
4 5E-01	x	2.0E-03 9.0E-04	S X		2.18	1 1	0.8	Yes	Dinitrotoluene, 4-Amino-2,6- Dinitrotoluene, Technical grade	25321-14-6	1 7E-01	2.6E-01		1.0E-01	4.0E+00	3.0E+02		3.9E+00 1 1E+00	
	~	1.0E-03	1		3.56	1	0.9	Yes	Dinoseb	88-85-7		2.02 01			2.0E+00	5.4E+00		1.5E+00	7.0E+00
1.0E-01	I 5.0E-06 I	3.0E-02	I 3.0E-02	ΙV	-0.27	1	1	Yes	Dioxane, 1,4-	123-91-1	7.8E-01	2.3E+02	1.1E+00	4.6E-01	6.0E+01	1.9E+04	6.3E+00	5.7E+00	
									Dioxins										
6.2E+03	I 1.3E+00 I				8.21	1	0	No	~Hexachlorodibenzo-p-dioxin, Mixture		1.3E-05			1.3E-05					
1.3E+05	C 3.8E+01 C	7.0E-10	1 4.0E-08 0	CV	6.8	1 1	0.5	No	~TCDD, 2,3,7,8-	1746-01-6	6.0E-07		1.5E-07	1.2E-07	1.4E-06	4.05.00	8.3E-06	1.2E-06	3.0E-05
		3.0E-02	1 4 05 04 3	× \/	2.17	1	1	res	Diphenamia Disharad Ethan	957-51-7					6.0E+01	4.2E+02	0.05.00	5.3E+01	
		8 0E-04	4.0E-04 /	~ v	4.21	1	1	Yes	Diphenyl Sulfone	101-64-6					1.6E±00	2.0E±01	6.3E-02	0.3E-02 1.5E±00	
		1.0E-01	õ		3.5	1	1	Yes	Diphenylamine	122-39-4					2.0E+02	3.4E+02		1.3E+02	
8.0E-01	I 2.2E-04 I				2.94	1	1	Yes	Diphenylhydrazine, 1,2-	122-66-7	9.7E-02	3.9E-01		7.8E-02					
		2.2E-03	I		-4.6	1	1	No	Diquat	85-00-7					4.4E+00			4.4E+00	2.0E+01
7.1E+00	C 1.4E-01 C				4.9	1	1	No	Direct Black 38	1937-37-7	1.1E-02			1.1E-02					
7.4E+00	C 1.4E-01 C				2.6	1	1	No	Direct Blue 6	2602-46-2	1.1E-02			1.1E-02					
6.7E+00	C 1.4E-01 C	4.0E-05			-6.53	1	1	N0 Voc	Direct Brown 95	16071-86-6	1.2E-02			1.2E-02	8 0E-02	1.35-01		5.0E-02	
		4.0E-03	1	V	4.02	1	1	Ves	Dithiane 1 4-	505-29-3					2.0E+01	1.5E+03		2.0E+01	
		2.0E-03	i	•	2.68	1	1	Yes	Diuron	330-54-1					4.0E+00	3.6E+01		3.6E+00	
		2.0E-02	Ó		1.15	1	1	Yes		2439-10-3					4.0E+01	5.3E+03		4.0E+01	
		5.0E-02	0	V	3.21	1	1	Yes	EPTC	759-94-4					1.0E+02	3.0E+02		7.5E+01	
		6.0E-03	1	V	3.83	1	0.9	Yes	Endosulfan	115-29-7					1.2E+01	6.3E+01		1.0E+01	
		2.0E-02	1		1.91	1	1	Yes	Endothall U U V and the United VS/ VS/	145-73-3					4.0E+01	8.5E+02		3.8E+01	1.0E+02
0.05.00		3.0E-04		1. 17	5.2	1 1	0.8	Yes	Endrin	72-20-8	7.05.00	7.05.00	4.75.00	2.05.00	6.0E-01	3.7E-01	0.45.04	2.3E-01	2.0E+00
9.9E-03	1 1.2E-06 I	6.0E-03	2.0E-02		0.45	1	1	Yes	Epichioronyann Epoxybutane, 1,2-	106-89-8	7.9E+00	7.9E+02	4.7E+00	2.9E+00	1.2E+01	1.3E+03	4.2E+00	2.0E-01 4.2E+00	
		4.0E-02	P		-1.18	1	1	Yes	Ethanol. 2-(2-methoxyethoxy)-0	111-77-3					8.0E+01	3.9E+04		8.0E+01	
		5.0E-03	1		-0.22	1	1	Yes	Ethephon 91	16672-87-0					1.0E+01	4.2E+03		1.0E+01	
		5.0E-04	1		5.07	1	0.8	Yes	Ethion	563-12-2					1.0E+00	7.7E-01		4.3E-01	
		1.0E-01	P 6.0E-02 F	ΡV	0.59	1	1	Yes	Ethoxyethanol Acetate, 2	111-15-9					2.0E+02	2.3E+04	1.3E+01	1.2E+01	
		9.0E-02	P 2.0E-01		-0.32	1	1	Yes	Ethoxyethanol, 2-	110-80-5					1.8E+02	6.3E+04	4.2E+01	3.4E+01	
		9.0E-01	P 8 0E-02		1.32	1	1	Voc	Ethyl Acedate	141-78-0					1.0E+03	3.0E±03	1.3E+01	1.4E+01	
		J.0L-03	1.0E+01	IV	1.43	1	1	Yes	Ethyl Chloride (Chloroethane)	75-00-3					1.02+01	3.0L+02	2.1E+03	2.1E+03	
		2.0E-01	1	V	0.89	1	1	Yes	Ethyl Ether	60-29-7					4.0E+02	2.0E+04		3.9E+02	
			3.0E-01	ΡV	1.94	1	1	Yes	Ethyl Methacrylate	97-63-2							6.3E+01	6.3E+01	
		1.0E-05	1		4.78	1	0.8	Yes	Ethyl-p-nitrophenyl Phosphonate	2104-64-5					2.0E-02	1.6E-02		8.9E-03	
1.1E-02	C 2.5E-06 C	1.0E-01	1 1.0E+00	IV	3.15	1	1	Yes	Etnyidenzene	100-41-4	7.1E+00	1.2E+01	2.2E+00	1.5E+00	2.0E+02	3.8E+02	2.1E+02	8.1E+01	7.0E+02
		7.0E-02	P	V	-0.94	1	1	T ES	Ethylene Diamine	109-78-4					1.4E+02	1.1E+05		1.4E+02 1.8E+02	
		2.0E+00	I 4.0E-01 (	с	-1.36	1	1	Yes	Ethylene Glycol	107-21-1					4.0E+02	5.7E+06		4.0E+02	
		1.0E-01	I 1.6E+00	1	0.83	1	1	Yes	Ethylene Glycol Monobutyl Ether	111-76-2					2.0E+02	1.4E+04		2.0E+02	
3.1E-01	C 3.0E-03 I		3.0E-02 0	CV M	-0.3	1	1	Yes	Ethylene Oxide	75-21-8	8.1E-02	1.7E+01	6.8E-04	6.7E-04			6.3E+00	6.3E+00	
4.5E-02	C 1.3E-05 C	8.0E-05			-0.66	1	1	Yes	Ethylene Thiourea	96-45-7	1.7E+00	1.0E+03		1.7E+00	1.6E-01	1.0E+02		1.6E-01	
6.5E+01	C 1.9E-02 C	0.05.00		V	-0.28	1	1	Yes	Ethyleneimine	151-56-4	1.2E-03	2.5E-01	3.0E-04	2.4E-04	0.05.05	4.55.05		5.05.00	
		3.0E+00 2.5E-04	1		2.19	1	1 0 0	Yes	Ethylphthalyl Ethyl Glycolate	84-72-0					6.0E+03 5.0E-01	1.5E+05 3.4E±00		5.8E+03 4.4E-01	
		2.5E-02	1		5.7	1 1	0.8	Yes	Fenoronathrin	39515-41-8					5.0E+01	7.3E+00		6.4E+00	
		2.5E-02	1		6.2	1	0.7	No	Fenvalerate	51630-58-1					5.0E+01			5.0E+01	
		1.3E-02	1		2.42	1	1	Yes	Fluometuron	2164-17-2					2.6E+01	3.4E+02		2.4E+01	
		4.0E-02	C 1.3E-02 (	С		1	1	Yes	Fluoride	16984-48-8					8.0E+01	1.8E+04		8.0E+01	
		6.0E-02	I 1.3E-02 (	С	0.40	1	1	Yes	Fluorine (Soluble Fluoride)	7782-41-4					1.2E+02	2.7E+04		1.2E+02	4.0E+03
		8.0E-02	0		3.16	1	0.9	res	Flumaone	59756-60-4					1.6E+02	1.4E+03		1.4E+02	
		2.0E-02	0		3.34	1	0.9	Yes	Flusilazole	85509-19-0					3.0E+01 4.0E+00	1.0E+02 1.4E+01		2.0E+01 3.1E+00	
		5.0E-01	õ		3.7	1	0.9	Yes	Flutolanil	66332-96-5					1.0E+03	3.7E+03		7.9E+02	
		1.0E-02	1		6.81	1	0.6	No	Fluvalinate	69409-94-5					2.0E+01			2.0E+01	
		9.0E-02	0		2.85	1	1	Yes	Folpet	133-07-3					1.8E+02	1.9E+03		1.6E+02	
		2.5E-03	0		2.9	1	1	Yes	Fomesafen	72178-02-0					5.0E+00	1.2E+02		4.8E+00	
	1 25 05	2.0E-03		A \/	3.94	1	0.9	Yes	Fonoros	944-22-9			4 2E 04	4.25.01	4.0E+00	6.3E+00	2.05.00	2.4E+00	
	1.3E-00	2.0E-01	1 9.0E-U3 /	m v	0.30		1	res	Formaldenyde	50-00-0			4.3E-01	4.3E-01	4.0E+02	3.2E+04	2.0E+00	2.0E+00	

Key: I = IRIS	; P = PPRTV; D	= DWSHA; C	) = OPP; A = A	ATSDR; C	= Cal EPA	; X = AP	PENDIX	PPR1	TV SCREEN (See FAQ #29); H = HEAST; F = See FAQ; E = see user guide Section	2.3.5; W = see	user guide Se	ction 2.3.6; l	= see user gu	ide on lead; M = n	nutagen; S = s	see user guid	le Section 5; V	= volatile; R = RBA ar	pplied (See
	т	User Guid	e for Arsenic r hemical-specif	ic Informa	= cancer; n tion	= nonca	ncer; " =	wner	e: n SL < 100X c SL; = where n SL < 10X c SL; SSL values are based on DAF=1; r Contaminant	m = Concentrati	on may excee	ed ceiling ilm rcinogenic Ta	it (See User G arget Risk (TR)	uide); s = Concent	ration may ex	Noncanc	ee User Guide er CHILD Haza	i rd Index (HI) = 0.1	
	TT TÏ	skielty and of							Containinging		- Cu		ingot rubit ( rrt)	- 12 00	Ingestion SL	Dermal SL	Inhalation SL	Noncarcinogenic SL	
050	k k		k PFC	k v							Ingestion SL	Dermal SL	Inhalation SL	Carcinogenic SL	Child	Child	Child	Child	
SFO	e IUR e	RfD <sub>o</sub>	e RIC <sub>i</sub> (	e o muta-		CIADO		002	Arabia		TR=1E-06	TR=1E-06	TR=1E-06	TR=1E-06	THQ=0.1	THQ=0.1	THQ=0.1	THI=0.1	MCL
(mg/kg-day)	y (ug/m <sup>*</sup> ) y	(mg/kg-day)	y (mg/m)	y i gen	LUGP	GIABS		PD?	Analyte	CAS NO.	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)
		9.0E-01	P 3.0E-04 /	x v	-0.54	1	1 16	35	Formic Acid	20149 24 9					1.0E+03	0.4E+05	0.3E-02	6.3E-02	
		2.3E+00	0		-2.4		1 11	0	Fusetyi-AL	39140-24-0					5.0E+03			5.0E+03	
		1.0E-03	х	V	4.12	1	1 Ye	es	~Dibenzofuran	132-64-9					2.0E+00	1.3E+00		7.9E-01	
		1.0E-03	1	V	1.34	1	1 Ye	es	~Furan	110-00-9					2.0E+00	4.8E+01		1.9E+00	
		9.0E-01	I 2.0E+00	IV	0.46	1	1 Ye	es	~Tetrahydrofuran	109-99-9					1.8E+03	1.7E+05	4.2E+02	3.4E+02	
3.8E+00	н	2.05.02		1.1/	-0.04	1	1 Ye	es	Furazolidone	67-45-8	2.1E-02	1.0E+01		2.0E-02	0.05.00	745.00	4.05.04	2.05.00	
1 5E+00	C 43E-04 C	3.0E-03	1 5.0E-02 F	ч v	1.8	1	1 TE 1 Ya	35	Fununai	531_82_8	5.2E-02	1 9E±00		5 1E-02	0.0E+00	7.1E+02	1.0E+01	3.8E+00	
3.0E-02	I 8.6E-06 C				4.38	1	).9 Ye	es	Furmecyclox	60568-05-0	2.6E+00	2.0E+00		1.1E+00					
		6.0E-03	0		-4.81	1	1 N	0	Glufosinate, Ammonium	77182-82-2					1.2E+01			1.2E+01	
		1.0E-01	A 8.0E-05 0	0	-0.33	1	1 Ye	es	Glutaraldehyde	111-30-8					2.0E+02	6.0E+04		2.0E+02	
		4.0E-04	I 1.0E-03 H	ΗV	-0.12	1	1 Ye	es	Glycidyl	765-34-4					8.0E-01	1.8E+02	2.1E-01	1.7E-01	
		1.0E-01	I V	V	-3.4	1	1 N	0	Giyphosate	1071-83-6					2.0E+02	4.25.04		2.0E+02	7.0E+02
		2.0E-02	P	v	-1.63	1	1 TE	0	Guanidine Guanidine Chloride	50-01-1					2.0E+01 4.0E+01	4.2E+04		2.0E+01 4.0E+01	
		3.0E-02	X		-8.35	1	1 N	0	Guanidine Nitrate	506-93-4					6.0E+01			6.0E+01	
		5.0E-05	1		4.07	1	).9 Ye	es	Haloxyfop, Methyl	69806-40-2					1.0E-01	3.1E-01		7.6E-02	
4.5E+00	I 1.3E-03 I	5.0E-04	1	V	6.1	1	).8 Ye	es	Heptachlor	76-44-8	1.7E-02	2.3E-03	4.3E-03	1.4E-03	1.0E+00	1.5E-01		1.3E-01	4.0E-01
9.1E+00	I 2.6E-03 I	1.3E-05	1	V	4.98	1 1	).8 Ye	es	Heptachlor Epoxide	1024-57-3	8.6E-03	7.1E-03	2.2E-03	1.4E-03	2.6E-02	2.4E-02		1.2E-02	2.0E-01
		0.05.04	3.0E-03	XV	2.29	1	1 Ye	es	Heptanal, n-	111-71-7					0.05.04		6.3E-01	6.3E-01	
		3.0E-04	X 4.0E-01 I	P V	4.66	1	1 N	0	Heptane, N-	142-82-5					6.0E-01		8.3E+01	6.0E-01	
		2.0E-03 2.0E-04	1	v	6.07	1	0.7 N	0	Hexabromodinbenvl ether-2 2' 4 4' 5 5'- (BDE-153)	87-82-1					4.0E+00 4.0E-01			4.0E+00 4.0E-01	
1.6E+00	4.6E-04	8.0E-04	i	V	5.73	1	0.9 N	0	Hexachlorobenzene	118-74-1	4.9E-02		1.2E-02	9.8E-03	1.6E+00			1.6E+00	1.0E+00
7.8E-02	1 2.2E-05 I	1.0E-03	Р	V	4.78	1	).9 Ye	es	Hexachlorobutadiene	87-68-3	1.0E+00	4.4E-01	2.6E-01	1.4E-01	2.0E+00	9.5E-01		6.5E-01	
6.3E+00	I 1.8E-03 I	8.0E-03	A		3.8	1	).9 Ye	es	Hexachlorocyclohexane Alpha-	319-84-6	1.2E-02	1.8E-02		7.2E-03	1.6E+01	2.5E+01		9.7E+00	
1.8E+00	I 5.3E-04 I				3.78	1	).9 Ye	es	Hexachlorocyclohexane, Beta	319-85-7	4.3E-02	6.1E-02		2.5E-02					
1.1E+00	C 3.1E-04 C	3.0E-04	1		3.72	1	).9 Ye	es	Hexachlorocyclohexane, Gamma- (Lindane)	58-89-9	7.1E-02	1.0E-01		4.2E-02	6.0E-01	9.3E-01		3.6E-01	2.0E-01
1.8E+00	I 5.1E-04 I	6 0E-03	L 2.0E-04	L V	4.14	1	).9 Ye	es	Hexachlorocyclohexane, lechnical	608-73-1 77-47-4	4.3E-02	6.1E-02		2.5E-02	1.25+01	4 2E+00	4 2E-02	4 1E-02	5 0E+01
4.0E-02	L 1 1E-05 C	7.0E-04	1 2.0E-04		4 14	1	1 Ya	55 26	Hexachloroethane	67-72-1	1.9E±00	1 7E±00	5 1E-01	3 3E-01	1.2L+01	4.2E+00	6.3E±00	4.1E-02 6.2E-01	3.0L+01
4.02 02	1 1.12 00 0	3.0E-04	1 0.02 02		7.54	1	0 N	0	Hexachlorophene // // ······	70-30-4	1.52100	1.7 2100	0.1E 01	0.02 01	6.0E-01	1.42100	0.02100	6.0E-01	
1.1E-01	1	3.0E-03	i		0.87	1	1 Ye	es	Hexahydro-1,3,5-trinitro-1,3,5-triazine (RDX)	121-82-4	7.1E-01	8.6E+01		7.0E-01	6.0E+00	8.0E+02		6.0E+00	
			1.0E-05	IV	3.2	1	1 Ye	es	HexamethyleneDiisoclyanatel-1,6-	822-06-0							2.1E-03	2.1E-03	
		4.0E-04	P		0.28	1	1 Ye	es	Hexamethylphosphoramide	680-31-9					8.0E-01	2.0E+02	4 55 .00	8.0E-01	
		0.05.00	7.0E-01	IV	3.9	1	1 Ye	es	Hexane, N-	110-54-3					4.05.00	1.15.00	1.5E+02	1.5E+02	
		2.0E+00 5.0E-03	P 1 3 0E-02	I V	0.08	1	1 Ye 1 Ye	es	Hexanedioic Acid	124-04-9					4.0E+03	1.1E+06 2.8E+02	6 3E±00	4.0E+03 3.8E+00	
		3.3E-02	1 0.02 02		1.85	1	1 Ye	es	Hexazinone	51235-04-2					6.6E+01	2.4E+03	0.02100	6.4E+01	
		2.5E-02	1		5.57	1	).8 Ye	es	Hexythiazox	78587-05-0					5.0E+01	1.4E+01		1.1E+01	
		1.7E-02	0		2.31	1	1 Ye	es	Hydramethylnon	67485-29-4					3.4E+01	2.9E+03		3.4E+01	
3.0E+00	I 4.9E-03 I		3.0E-05 F	ΡV	-2.07	1	1 Ye	es	Hydrazine	302-01-2	2.6E-02	1.1E+02	1.1E-03	1.1E-03			6.3E-03	6.3E-03	
3.0E+00	I 4.9E-03 I		2.05.02			1	1 Ye	es	Hydrazine Sulfate	10034-93-2	2.6E-02	4.9E+00		2.6E-02			4.25.00	4.95.00	
		4.0E-02	C 1.4E-02	C V	0.23	1	1 Ye	25 25	Hydrogen Chloride	7664-39-3					8.0E+01	1.8E+04	4.2E+00 2.9E+00	4.2E+00 2.8E+00	
			2.0E-03	IV	0.23	1	1 Ye	es	Hydrogen Sulfide	7783-06-4					0.02101	1.02104	4.2E-01	4.2E-01	
6.0E-02	Р	4.0E-02	P		0.59	1	1 Ye	es	Hydroquinone	123-31-9	1.3E+00	1.2E+02		1.3E+00	8.0E+01	7.9E+03		7.9E+01	
6.1E-02	0	2.5E-03	0		3.82	1	).9 Ye	es	Imazalil	35554-44-0	1.3E+00	3.1E+00		9.0E-01	5.0E+00	1.3E+01		3.6E+00	
		2.5E-01	1		1.86	1	1 Ye	es	Imazaquin	81335-37-7					5.0E+02	2.6E+04		4.9E+02	
		2.5E+00	0		1.49	1	1 Ye	es	Imazethapyr	81335-77-5					5.0E+03	7.2E+04		4.7E+03	
		4.0E-02	A		2.49	1 1		25 26	Incodione	36734-19-7					2.0E+01 8.0E+01	9.1E±02		7.4E+01	
		7.0E-01	P		0	1	1 Ye	25	Iron	7439-89-6					1.4E+03	3.2E+05		1.4E+03	
		3.0E-01	1	V	0.76	1	1 Ye	es	Isobutyl Alcohol	78-83-1					6.0E+02	3.6E+04		5.9E+02	
9.5E-04	I	2.0E-01	I 2.0E+00 0	0	1.7	1	1 Ye	es	Isophorone	78-59-1	8.2E+01	1.6E+03		7.8E+01	4.0E+02	8.6E+03		3.8E+02	
		1.5E-02		V	5.8	1	).8 Ye	es	Isopropalin	33820-53-0					3.0E+01	4.6E+00	105.00	4.0E+00	
		2.0E+00	P 2.0E-01 I	v	0.05	1	1 Ye	35	Isopropario	07-03-0					4.0E+03	0.5E+05	4.2E+01	4.1E+01	
		1.0E-01			0.27	1	1 Ye	#S	Isopropyi ivietnyi Phosphonic Acia	1832-54-8					2.0E+02	3.9E+04		2.0E+02	
		3.0E-02	3.0E-01	ΑV	8	1	0 N	0	JP-7	E1737665					1.0E+02	2.7 E+02	6.3E+01	6.3E+01	
		8.0E-03	0		4.81	1	).9 Ye	es	Lactofen	77501-63-4					1.6E+01	2.7E+01		1.0E+01	
		2.0E-04	Х		-0.94	1	1 Ye	es	Lactonitrile	78-97-7					4.0E-01	3.2E+02		4.0E-01	
									Lead Compounds										
8.5E-03	C 1.2E-05 C				0.00	1	).8 Ye	es	~Lead Phosphate	7446-27-7	9.2E+00	1.7E+03		9.1E+00					
8.5E-03	C 1.2E-05 C				-0.08	1	1 Ye	#S	~Lead adelate	301-04-2	9.2E+00	9.1E+03		9.2E+00				1.5E+01	1.5E+01
							. 16		Loud and compounds	1400.0211								1.0LTUI	1.02701

Key: I = IRIS; P = PPRTV; I	D = DWSHA; (	O = OPP; A = A	TSDR; C	= Cal EPA	A; X = A			TV SCREEN (See FAQ #29); H = HEAST; F = See FAQ; E = see user guide Section $p_1 = 2 \sum_{i=1}^{N} (100 \times 2 \sum_{i=1}^{N} p_i) = where p_i \sum_{i=1}^{N} (100 \times 2 \sum_{i=1}^{N} p_i)$	2.3.5; W = see	user guide Se	ction 2.3.6; L	. = see user gu	ide on lead; M = n	nutagen; S = s	ee user guide	e Section 5; V	= volatile; R = RBA a	pplied (See
	Toxicity and C	Chemical-specif	ic Informat	tion		ancer,	= wrie	Contaminant	m = concentiati	Car	cinogenic Ta	arget Risk (TR)	= 1E-06	auon may ext	Noncance	r CHILD Hazar	rd Index (HI) = 0.1	
											L			Ingestion SL	Dermal SL	Inhalation SL	Noncarcinogenic SL	
SEO O IUR O	RfD	k RfCi k	<pre>&lt; v </pre>							Ingestion SL	Dermal SL	Inhalation SL	Carcinogenic SL	Child	Child	Child	Child	MCI
(mg/kg-day) <sup>-1</sup> y (ug/m <sup>3</sup> ) <sup>-1</sup> y	(mg/kg-day)	y (mg/m <sup>3</sup> )	/ I gen	LOGP	GIABS	FA I	n EPD?	Analyte	CAS No.	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)
8.5E-03 C 1.2E-05 C	;	, , ,		-4	1	1	No	~Lead subacetate	1335-32-6	9.2E+00	(0)	,	9.2E+00	,	(0)	(0)	,	
	1.0E-07	1	V	4.15	1	0.9	Yes	~Tetraethyl Lead	78-00-2					2.0E-04	3.8E-04		1.3E-04	
	5.0E-06	Р	V	2.56	1	1	Yes	Lewisite	541-25-3					1.0E-02	9.1E-02		9.0E-03	
	7.7E-03	0		3.2	1	0.9	Yes	Linuron	330-55-2					1.5E+01	7.6E+01		1.3E+01	
	2.0E-03 5.0E-04	- I		3.25	1	1	Yes	MCPA	94-74-6					4.0E+00 1.0E+00	3.0E+02		4.0E+00 7.5E-01	
	4.4E-03	0		2.79	1	0.9	Yes	МСРВ	94-81-5					8.8E+00	2.4E+01		6.5E+00	
	1.0E-03	1		3.13	1	1	Yes	MCPP	93-65-2					2.0E+00	7.1E+00		1.6E+00	
	2.0E-02	I		2.36	1	1	Yes	Malathion	121-75-5					4.0E+01	1.1E+03		3.9E+01	
	1.0E-01	I 7.0E-04 0	)	1.62	1	1	Yes	Maleic Anhydride	108-31-6					2.0E+02	3.8E+03		1.9E+02	
	5.0E-01 1.0E-04	P		-0.84	1	1	Yes	Maleic Hydrazide Malononitrile	123-33-1					1.0E+03 2.0E-01	8.9E+05 9.2E±01		1.0E+03 2.0E-01	
	3.0E-02	Н		1.33	1	0.9	Yes	Mancozeb	8018-01-7					6.0E+01	4.9E+02		5.4E+01	
	5.0E-03	I		0.62	1	1	Yes	Maneb	12427-38-2					1.0E+01	3.6E+02		9.8E+00	
	1.4E-01	I 5.0E-05			1	1		Manganese (Diet)	7439-96-5									
	2.4E-02	S 5.0E-05			0.04	1	Yes	Manganese (Non-diet)	7439-96-5					4.8E+01	4.4E+02		4.3E+01	
	9.0E-05	н		1.04	1	1	Yes	Mephosfolan Meniguat Chlorida	950-10-7					1.8E-01 6.0E+01	2.5E+01		1.8E-01 6.0E+01	
1 1E-02 P	4.0E-02	P		2.02	1	1	Yes	Mercaptobenzothiazole 2-	149-30-4	7 1E+00	5.6E+01		6.3E+00	8.0E+00	6.9E+01		7.2E+00	
1.12 02 1	4.02 00			2.72			105	Mercury Compounds	140 00 4	7.12100	0.02101		0.02100	0.02100	0.52101		7.22100	
	3.0E-04	I 3.0E-04 \$	S	-0.22	0.07	1	Yes	~Mercuric Chloride (and other Mercury salts)	7487-94-7					6.0E-01	9.6E+00		5.7E-01	2.0E+00
		3.0E-04	V	0.62	1	1	Yes	~Mercury (elemental)	7439-97-6							6.3E-02	6.3E-02	2.0E+00
	1.0E-04	1		0.74	1	1	Yes	~Methyl Mercury	22967-92-6					2.0E-01	4.6E+01		2.0E-01	
	8.0E-05		V	7.67	1	0.2	res	APrenymercunc Acetate	62-38-4					1.0E-01	5.7E+01		1.6E-01	
	3.0E-03 1.0E-04	0	v	5.7	1	0.9	Yes	Merphos Oxide ()) (1) (()) (()	78-48-8					2.0E-02	3.3E-02		2.8E-02	
	6.0E-02	Ī		1.65	1	1	Yes	Metalaxyl	57837-19-1					1.2E+02	6.4E+03		1.2E+02	
	1.0E-04	I 3.0E-02 F	٧	0.68	1	1	Yes	Methacrylonitrile	126-98-7					2.0E-01	1.3E+01	6.3E+00	1.9E-01	
	5.0E-05	1		-0.8	1	1	Yes	Methamidophos	10265-92-6					1.0E-01	1.0E+02		1.0E-01	
	2.0E+00	I 2.0E+01	V	-0.77	1	1	Yes	Methanol	67-56-1					4.0E+03	1.8E+06	4.2E+03	2.0E+03	
	1.5E-03	0		2.2	1	1	Yes	Methodathion	950-37-8 16752-77-5					3.0E+00 5.0E+01	8.7E+01		2.9E+00	
4.9E-02 C 1.4E-05 C	2.51-02	1		1.47	1	1	Yes	Methoxy-5-nitroaniline 2-	99-59-2	1.6E+00	5.4E+01		1.5E+00	3.0L+01	0.02+03		3.02+01	
	5.0E-03	I		5.08	1	0.8	Yes	Methoxychlor	72-43-5					1.0E+01	5.9E+00		3.7E+00	4.0E+01
	8.0E-03	P 1.0E-03 F	٧	0.1	1	1	Yes	Methoxyethanol Acetate 2-	110-49-6					1.6E+01	3.5E+03	2.1E-01	2.1E-01	
	5.0E-03	P 2.0E-02	V	-0.77	1	1	Yes	Methoxyethanol; 2-	109-86-4					1.0E+01	6.3E+03	4.2E+00	2.9E+00	
	1.0E+00	X	V	0.18	1	1	Yes	Methyl Acetate	79-20-9					2.0E+03	2.9E+05	4.05.00	2.0E+03	
	6.0E-01	1 5.0E+00	- v I V	0.29	1	1	Yes	Methyl Ethyl Ketone (2-Butanone)	78-93-3					1.2E+03	1.5E+05	4.2E+00 1.0E+03	4.2E+00 5.6E+02	
1.0E-03 X	( 1.0E-03	P 2.0E-05 >	< V	-1.05	1	1	Yes	Methyl Hydrazine	60-34-4			5.6E-03	5.6E-03	2.0E+00	1.5E+03	4.2E-03	4.2E-03	
		3.0E+00	V	1.31	1	1	Yes	Methyl Isobutyl Ketone (4-methyl-2-pentanone)	108-10-1							6.3E+02	6.3E+02	
		1.0E-03 (	C V	0.79	1	1	Yes	Methyl Isocyanate	624-83-9							2.1E-01	2.1E-01	
	1.4E+00	I 7.0E-01	V	1.38	1	1	Yes	Methyl Methacrylate	80-62-6					2.8E+03	7.7E+04	1.5E+02	1.4E+02	
	2.5E-04 6.0E-02	X		-0.7	1	1	Yes	Methyl Paramion	298-00-0 993-13-5					5.0E-01 1.2E+02	4.1E+00 1.2E+05		4.5E-01 1.2E+02	
	6.0E-03	H 4.0E-02 H	ΗV	3.44	1	0.8	Yes	Methyl Styrene (Mixed Isomers)	25013-15-4					1.2E+01	4.3E+00	8.3E+00	2.3E+00	
9.9E-02 C 2.8E-05 C	;			-0.66	1	1	Yes	Methyl methanesulfonate	66-27-3	7.9E-01	4.8E+02		7.9E-01					
1.8E-03 C 2.6E-07 C	;	3.0E+00	V	0.94	1	1	Yes	Methyl tert-Butyl Ether (MTBE)	1634-04-4	4.3E+01	2.0E+03	2.2E+01	1.4E+01			6.3E+02	6.3E+02	
	3.0E-04	X		-2.06	1	1	Yes	Methyl-1,4-benzenediamine dihydrochloride, 2-	615-45-2					6.0E-01	5.9E+03	0.05.00	6.0E-01	
9.0E-03 P	2 0E-02	3.0E+00 X	(V	1.43 1.87	1	1	Yes	Metnyl-2-Pentanol, 4- Methyl-5-Nitroaniline 2-	108-11-2	8 7E+00	1 4E+02		8 2E+00	4 0E+01	7.3E+02	6.3E+02	6.3E+02 3.8E+01	
8.3E+00 C 2.4E-03 C	2.02.02	~		-0.92	1	1	Yes	Methyl-N-nitro-N-nitrosoguanidine N-	70-25-7	9.4E-03	1 1E+01		9.4E-03		1.02102		0.02101	
1.3E-01 C 3.7E-05 C	;			1.62	1	1	Yes	Methylaniline Hydrochloride, 2-	636-21-5	6.0E-01	3.9E+03		6.0E-01					
	1.0E-02	А		-1.18	1	1	Yes	Methylarsonic acid	124-58-3					2.0E+01	3.6E+04		2.0E+01	
	2.0E-04	X		_	1	0	No	Methylbenzene,1-4-diamine monohydrochloride, 2-	74612-12-7					4.0E-01			4.0E-01	
1.0E-01 X 2.2E+01 C 6.3E-02 C	3.0E-04	Х	м	6.42	1	0	No	Methylbenzene-1,4-diamine sulfate, 2- Methylcholanthrene, 3-	615-50-9	7.8E-01			7.8E-01	6.0E-01			6.0Ë-01	
2.0E-03   1.0E-08	6.0E-03	L 6.0E-01	V M	1.25	1	1	Yes	Methylene Chloride	75-09-2	1.3E+01	3.5E+02	2.0E+02	1 1E+01	1 2E+01	3.7E+02	1.3E+02	1 1E+01	5.0E+00
1.0E-01 P 4.3E-04 C	2.0E-03	P	M	3.91	1	0.9	Yes	Methylene-bis(2-chloroaniline), 4,4'-	101-14-4	2.5E-01	4.3E-01	2.02702	1.6E-01	4.0E+00	7.5E+00	1.02 +02	2.6E+00	0.02+00
4.6E-02 I 1.3E-05 C	;			4.37	1	1	Yes	Methylene-bis(N,N-dimethyl) Aniline, 4,4'-	101-61-1	1.7E+00	6.7E-01		4.8E-01					
1.6E+00 C 4.6E-04 C	;	2.0E-02 C		1.59	1	1	Yes	Methylenebisbenzenamine, 4,4'-	101-77-9	4.9E-02	1.7E+00		4.7E-02					
	7 0E 02	6.0E-04	V	5.22	1	0.9	Yes	Methylenediphenyl Diisocyanate	101-68-8					1.45.02	1 7E : 02		7 85 .01	
	1.0E-02	1	v	3.40	1	1	Vec	Metolachlor	51218-45-2					3.0E+02	2.6E+02		2 7E+02	
	2.5E-02	1		1.7	1	1	Yes	Metribuzin	21087-64-9					5.0E+02	1.8E+03		4.9E+01	
	2.5E-01	1		2.2	1	1	Yes	Metsulfuron-methyl	74223-64-6					5.0E+02	2.4E+04		4.9E+02	
	3.0E+00	Р	V	6.1	1	1	No	Mineral oils	8012-95-1					6.0E+03			6.0E+03	

Key: I = IRIS	; P = PPRTV; D	= DWSHA; O	= OPP; A = A1	SDR; C	= Cal EPA	; X = AP	PEND	X PPR	TV SCREEN (See FAQ #29); H = HEAST; F = See FAQ; E = see user guide Section	2.3.5; W = see	user guide Se	ction 2.3.6; I	_ = see user gu	ide on lead; M = r	mutagen; S = s	ee user guid	le Section 5; V	′ = volatile; R = RBA a	pplied (See
	т	User Guide	e for Arsenic no	tice) ; c =	= cancer; n	i = nonca	incer; *	= whe	re: n SL < 100X c SL; ** = where n SL < 10X c SL; SSL values are based on DAF=1;	m = Concentrati	on may excee	d ceiling lim	it (See User G	uide); s = Concent	tration may ex	ceed Csat (S	ee User Guide	e) ard Index (HI) = 0.1	
	rr ri	UNICITY AND CI	iennical-specific	inioma					Contaninant		Ca	cinogenic n	arget Nisk (TR)	- 12-00	Ingestion SL	Dermal SL	Inhalation SL	Noncarcinogenic SL	
050	k k		k k	v							Ingestion SL	Dermal SL	Inhalation SL	Carcinogenic SL	. Child	Child	Child	Child	
SFO	e IUR e	RfD <sub>o</sub>	e <sup>RIC</sup> i e	o muta-		CIADO			Arabia		TR=1E-06	TR=1E-06	TR=1E-06	TR=1E-06	THQ=0.1	THQ=0.1	THQ=0.1	THI=0.1	MCL
(mg/kg-day)	y (ug/m <sup>-</sup> ) y	(mg/kg-day)	y (mg/m <sup>-</sup> ) y	i gen	LUGP	GIABS		EPD?	Analyte	CAS NO.	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)
1.8E+01	C 5.1E-03 C	2.0E-04 2.0E-03	1	v	6.89 3.21	1	0.5	NO Yes	Mirex Molinate	2385-85-5	4.3E-03		1.1E-03	8.8E-04	4.0E-01 4.0E+00	1 2E±01		4.0E-01 3.0E+00	
		5.0E-03			0.21	1	1	Yes	Molybdenum	7439-98-7					1.0E+01	2.3E+03		1.0E+01	
		1.0E-01	Ì			1	1	Yes	Monochloramine	10599-90-3					2.0E+02	4.6E+04		2.0E+02	4.0E+03
		2.0E-03	Р		1.66	1	1	Yes	Monomethylaniline	100-61-8					4.0E+00	7.5E+01		3.8E+00	
		2.5E-02	1		2.94	1	1	Yes	Myclobutanil	88671-89-0					5.0E+01	4.7E+02		4.5E+01	
		3.0E-04	x		4.04	1	0.9	Yes	N,N'-Diphenyl-1,4-benzenediamine	74-31-7					6.0E-01	8.9E-01		3.6E-01	
		2.0E-03	1	V	1.38	1	1	Yes	Naled	300-76-5					4.0E+00	6.8E+02		4.0E+00	
1.95.00		3.0E-02	X 1.0E-01 P	v	2.20	1	0	No	Naphtha, High Flash Aromatic (HFAN)	64/42-95-6	4 25 02	2 6E 01		2 0E 02	6.0E+01		2.1E+01	1.5E+01	
1.6E+00	C 0.0E+00 C	1.2E-01	0		3.36	1	0.9	Yes	Napropamide	15299-99-7	4.3E-02	3.00-01		3.9E-02	2.4E+02	1.1E+03		2.0E+02	
	2.6E-04 C	1.1E-02	C 1.4E-05 C		-1.38	1	1	Yes	Nickel Acetate	373-02-4					2.2E+01	6.8E+04		2.2E+01	
	2.6E-04 C	1.1E-02	C 1.4E-05 C		-2.12	1	1	Yes	Nickel Carbonate	3333-67-3					2.2E+01	1.4E+05		2.2E+01	
	2.6E-04 C	1.1E-02	C 1.4E-05 C	V		1	0	Yes	Nickel Carbonyl	13463-39-3			2.2E-02	2.2E-02	2.2E+01		2.9E-03	2.9E-03	
	2.6E-04 C	1.1E-02	C 1.4E-05 C			0.04	1	Yes	Nickel Hydroxide	12054-48-7					2.2E+01	2.0E+02		2.0E+01	
	2.6E-04 C	1.1E-02	C 2.0E-05 C			0.04	1	Yes	Nickel Oxide	1313-99-1					2.2E+01	2.0E+02		2.0E+01	
	2.4E-04 T	1.1E-02	C 1.4E-05 C			0.04	0	Yes	Nickel Refinery Dust	E715532					2.2E+01	1.0E+03		2.2E+01	
1 7E±00	C 4 8E-04 L	2.0E-02 1 1E-02	C 14E-05 C			0.04	1	Yes	Nickel Subsulfide	12035-72-2	4.6E-02	1 7E+00		4 5E-02	2.2E+01	1.0E+03		2.9E+01	
1.7 2700	2.6E-04 C	1.1E-02	C 1.4E-05 C			1	0	Yes	Nickelocene	1271-28-9	4.0L-02	1.7 2+00		4.02-02	2.2E+01	1.02+03		2.2E+01	
		1.6E+00	1			1	1	Yes	Nitrate	14797-55-8					3.2E+03	7.3E+05		3.2E+03	1.0E+04
						1	0	Yes	Nitrate + Nitrite (as N)	E701177									1.0E+04
		1.0E-01	1			1	1	Yes	Nitrite The A CP A A A A A A A A A A A A A A A A A	14797-65-0					2.0E+02	4.6E+04		2.0E+02	1.0E+03
		1.0E-02	X 5.0E-05 X		1.85	1	1	Yes	Nitroaniline, 2-	88-74-4					2.0E+01	3.4E+02		1.9E+01	
2.0E-02	P	4.0E-03	P 6.0E-03 P	V	1.39	1	1	Yes	Nitroaniline, 4-	100-01-6	3.9E+00	1.2E+02	1 45 01	3.8E+00	8.0E+00	2.8E+02	1.05.00	7.8E+00	
	4.0E-05 I	2.0E-03	1 9.0E-03 1	v	1.65	1	1	res		98-95-3			1.4E-01	1.4E-01	4.0E+00	0.2E+01	1.9E+00	1.3E+00	
		3.0E+03 7.0E-02	P H		-4.50	1	1	Yes	Nitrofurantoin	9004-70-0 67-20-9					0.0E+06	1.6E+05		0.0E+06 1 4E+02	
1.3E+00	C 3.7E-04 C	1.02 02			0.23	1	1	Yes	Nitrofurazone	59-87-0	6.0E-02	1.7E+01		6.0E-02	1.42102	1.02100		1.42102	
1.7E-02	Р	1.0E-04	Р		1.62	1	1	Yes	Nitroglycerin	55-63-0	4.6E+00	1.8E+02		4.5E+00	2.0E-01	8.7E+00		2.0E-01	
		1.0E-01	I.		-0.89	1	1	Yes	Nitroguanidine	556-88-7					2.0E+02	1.8E+05		2.0E+02	
	8.8E-06 P		5.0E-03 P	V	-0.35	1	1	Yes	Nitromethane	75-52-5			6.4E-01	6.4E-01			1.0E+00	1.0E+00	
0.75.04	2.7E-03 H		2.0E-02 I	V	0.93	1	1	Yes	Nitropropane, 2	79-46-9	0.05.04	4 55 04	2.1E-03	2.1E-03			4.2E+00	4.2E+00	
2.7E+01 1.2E+02	C 34E-02 C			M	-0.03	1	1	Yes	Nitroso-N-methylurea N-	759-73-9 684-93-5	9.3E-04 2.1E-04	4.6E-02		9.2E-04 2.1E-04					
5.4E+00	1 1.6E-03 L			V	2.63	1	1	Yes	Nitroso-di-N-butylancia, N-	924-16-3	1.4E-02	7.9E-02	3.5E-03	2.7E-03	1				
7.0E+00	I 2.0E-03 C				1.36	1	1	Yes	Nitroso-di-N-propylamine, N-	621-64-7	1.1E-02	3.5E-01		1.1E-02					
2.8E+00	I 8.0E-04 C				-1.28	1	1	Yes	Nitrosodiethanolamine, N-	1116-54-7	2.8E-02	8.1E+01		2.8E-02					
1.5E+02	I 4.3E-02 I			М	0.48	1	1	Yes	Nitrosodiethylamine, N-	55-18-5	1.7E-04	1.7E-02		1.7E-04					
5.1E+01	I 1.4E-02 I	8.0E-06	P 4.0E-05 X	V M	-0.57	1	1	Yes	Nitrosodimethylamine, N-	62-75-9	4.9E-04	2.0E-01	1.4E-04	1.1E-04	1.6E-02	7.4E+00	8.3E-03	5.5E-03	
4.9E-03	1 2.6E-06 C			V	3.13	1	1	Yes	Nitrosomethylathine, N-	10505 05 6	1.0E+01	5.2E+01	8 0E 04	7.1E.04	<u> </u>				
6.7E±00	C 19E-03 C			v	-0.44	1	1	Yes	Nitrosomorpholine [N-]	59-89-2	1.2E-03	5.3E±00	0.9E-04	1.1E-04					
9.4E+00	C 2.7E-03 C				0.36	1	1	Yes	Nitrosopiperidine [N-]	100-75-4	8.3E-03	1.1E+00		8.2E-03					
2.1E+00	I 6.1E-04 I				-0.19	1	1	Yes	Nitrosopyrrolidine, N-	930-55-2	3.7E-02	1.0E+01		3.7E-02					
		1.0E-04	Х		2.45	1	1	Yes	Nitrotoluene, m-	99-08-1					2.0E-01	1.4E+00		1.7E-01	
2.2E-01	Ч	9.0E-04	۲	V	2.3	1	1	Yes	Nitrotoluene, 0-	88-72-2	3.5E-01	2.8E+00		3.1E-01	1.8E+00	1.5E+01		1.6E+00	
1.6E-02	Р	4.0E-03	Y 20E02 D	V	2.37	1	1	Yes	Nitrotoluene, p-	99-99-0	4.9E+00	3.4E+01		4.3E+00	8.0E+00	6.2E+01	4 2E : 00	7.1E+00	
		1.5E-04	0	v	2.3	1	1	Yes	Norflurazon	27314-13-2					3.0E+01	7.5E+02	4.2E+00	2.9E+01	
		3.0E-03	1		8.71	1	0.3	No	Octabromodiphenyl Ether	32536-52-0					6.0E+00			6.0E+00	
		5.0E-02	1		0.16	1	1	Yes	Octahydro-1,3,5,7-tetranitro-1,3,5,7-tetrazocine (HMX)	2691-41-0					1.0E+02	6.3E+04		1.0E+02	
		2.0E-03	Н		-1.01	1	1	Yes	Octamethylpyrophosphoramide	152-16-9					4.0E+00	1.4E+04		4.0E+00	
7.8E-03	0	1.4E-01	0		3.73	1	0.9	Yes	Oryzalin	19044-88-3	1.0E+01	3.8E+01		7.9E+00	2.8E+02	1.2E+03		2.3E+02	
		5.0E-03			4.8	1	0.8	Yes	Oxadiazon	19666-30-9					1.0E+01	9.0E+00		4.7E+00	2.05.02
7 3E-02	0	2.5E-02 3.0E-02	0		-0.47	1	0.8	Vos	Oxalliyi	42874-03-2	1 1E+00	1 1E+00		5.4E-01	5.0E+01	5.1E+04		3.0E+01	2.0E+02
1.56-02	0	1.3E-02	1		32	1	0.9	Yes	Paclobutrazol	76738-62-0	1.12+00	1.12+00		3.4E-01	2.6E+01	1.7E+02		2.3E+01	
		4.5E-03	i		-4.5	1	1	No	Paraquat Dichloride	1910-42-5					9.0E+00			9.0E+00	
		6.0E-03	Н		3.83	1	0.9	Yes	Parathion	56-38-2					1.2E+01	3.0E+01		8.6E+00	
		5.0E-02	н	V	3.83	1	1	Yes	Pebulate	1114-71-2					1.0E+02	1.3E+02		5.6E+01	
		3.0E-02	0		5.2	1	0.9	Yes	Pendimethalin	40487-42-1					6.0E+01	1.8E+01		1.4E+01	
		2.0E-03		V	6.84	1	0.6	No	Pentabromodiphenyl Ether	32534-81-9					4.0E+00			4.0E+00	
		1.0E-04 8.0E-04		V	7.66	1	0.6	N0 Yes	Pentablomodiprienyi etner, 2,2,4,4,5- (BDE-99) Pentachlorobenzene	608-93-5					2.0E-01 1.6E+00	3.9E-01		2.0E-01 3.2E-01	
9.0E-02	Р	0.0L 04		v	3.22	1	1	Yes	Pentachloroethane	76-01-7	8.7E-01	2.5E+00		6.5E-01	1.02100	0.02 01		0.22 01	
2.6E-01	н	3.0E-03	L	V	4.64	1	0.9	Yes	Pentachloronitrobenzene	82-68-8	3.0E-01	2.0E-01		1.2E-01	6.0E+00	4.4E+00		2.6E+00	

Key: I = IRIS; P = PP	PRTV; D = DWSHA; (	D = OPP; A = A	TSDR; C = Cal I	EPA; X =	APPEN	DIX PPR	TV SCREEN (See FAQ #29); H = HEAST; F = See FAQ; E = see user guide Section 2	2.3.5; W = see	user guide Se	ection 2.3.6;	L = see user gu	ide on lead; M = r	nutagen; S =	see user guid	e Section 5; V =	= volatile; R = RBA a	pplied (See
	Toxicity and C	hemical-specific	c Information	r; n = no	ncancer	; = when	e: n SL < 100X c SL; = where n SL < 10X c SL; SSL values are based on DAF=1; n Contaminant	n = Concentrati	Ca	rcinoaenic T	arget Risk (TR	1 = 1E-06	ration may ex	Noncance	ee Oser Guide) er CHILD Hazar	d Index (HI) = 0.1	
													Ingestion SL	Dermal SL	Inhalation SL	Noncarcinogenic SL	
k K	k DfD	k RfC. k	v						Ingestion SL	Dermal SL	Inhalation SL	Carcinogenic SL	Child	Child	Child	Child	
(ma/ka-day) <sup>-1</sup> y (ua/r	(m <sup>3</sup> ) <sup>-1</sup> v (mg/kg-dav)	$e (mq/m^3) v$	o muta-		S FA	In EPD2	Analyte	CAS No	IR=1E-06 (ug/L)	IR=1E-06	IR=1E-06	IR=1E-06 (ug/L)	1HQ=0.1	1HQ=0.1	1HQ=0.1	I HI=0.1	MCL (ug/L)
4.0E-01 J 5.1E	E-06 C 5.0E-03	y (ing/in / y	5 13	2 1	09	Yes	Pentachlorophenol	87-86-5	1 9E-01	5 2E-02	(49/2)	4 1E-02	1 0E+01	2 9E±00	(09/2)	2 3E±00	1 0E±00
4.0E-01 1 3.1E	2 0E-03	P	2.32	2 1	0.3	Yes	Pentaen/thritol_tetranitrate (PETN)	78-11-5	1.9E±01	4 3E±02		1.9E±01	4.0E+01	9.6E±01		3.9E±00	1.02+00
4.02 00 X	2.02 00	1.0E+00 P	V 3.39	9 1	1	Yes	Pentane, n-	109-66-0	1.52101	4.02102		1.52101	4.02100	5.0L101	2.1E+02	2.1E+02	
							Perchlorates										
	7.0E-04	I		1	1	Yes	~Ammonium Perchlorate	7790-98-9					1.4E+00	3.2E+02		1.4E+00	
	7.0E-04	1		1	1	Yes	~Lithium Perchlorate	7791-03-9					1.4E+00	3.2E+02		1.4E+00	
	7.0E-04	1		1	1	Yes	~Perchlorate and Perchlorate Salts	14/9/-/3-0					1.4E+00	3.2E+02		1.4E+00	1.5E+01(F)
	7.0E-04 7.0E-04	1		1	1	Yes	~Polassium Perchlorate	7601-89-0					1.4E+00	3.2E±02		1.4E+00	
	2.0E-02	P		1	ò	Yes	Perfluorobutane sulfonic acid (PFBS)	375-73-5					4.0E+01	0.22102		4.0E+01	
	2.0E-02	Р		1	0	Yes	Perfluorobutanesulfonate	45187-15-3					4.0E+01			4.0E+01	
	5.0E-02	1	6.5	1	0.6	No	Permethrin	52645-53-1					1.0E+02			1.0E+02	
2.2E-03 C 6.3E	8E-07 C	-	1.58	3 1	1	Yes	Phenacetin	62-44-2	3.5E+01	1.1E+03		3.4E+01					
	2.4E-01	0	3.59	91	0.9	Yes	Phenmedipham	13684-63-4					4.8E+02	1.8E+03		3.8E+02	
	3.0E-01 4.0E-03	1 2.0E-01 C	1.40	2 I	1	Yes	Phenol 2-(1-methylethoxy)- methylcarbamate	108-95-2					8.0E+02	3.6E+02		5.8E+02 7.8E+00	
	5.0E-04	X	4.15	5 1	1	Yes	Phenothiazine	92-84-2					1.0E+00	7.6E-01		4.3E-01	
	2.0E-04	x	V 3.28	3 1	1	Yes	Phenyl Isothiocyanate	103-72-0					4.0E-01	7.6E-01		2.6E-01	
	6.0E-03	1	-0.3	31	1	Yes	Phenylenediamine, m-	108-45-2					1.2E+01	4.8E+03		1.2E+01	
1.2E-01 P	4.0E-03	P	0.15	5 1	1	Yes	Phenylenediamine, o-	95-54-5	6.5E-01	1.1E+02		6.5E-01	8.0E+00	1.5E+03		8.0E+00	
105.02	1.0E-03	х	-0.3	3 1 2 1	1	Yes	Phenylenediamine, p-	106-50-3	4.05.01	1 25 .02		2.05.01	2.0E+00	7.6E+02		2.0E+00	
1.32-03 11	2 0E-04	Ц	3.56	3 1	0.0	Voc	Phonete Colored Co	208-02-2	4.02401	1.22702		3.02+01	4.0E-01	1 2E±00		3.0E-01	
	2.02-04	3.0E-04 I	V -0.7	1 1	1	163		75-44-5					4.02-01	1.22400		3.02-01	
	2.0E-02	1	2.78	31	1	Yes	Phosmet 0 0 0 0 cmmb cmmb Cmmb X2 X22	732-11-6					4.0E+01	5.3E+02		3.7E+01	
							Phosphates, Inorganic										
	4.9E+01	P		1	1	Yes	~Aluminum metaphosphate	13776-88-0					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	P		1	0	Yes	~Ammonium polyphosphate	68333-79-9					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01 4.9E+01	P		1	1	Yes	-Calcium pyrophosphate	7783-28-0					9.7E+04 9.7E+04	2.2E+07		9.7E+04 9.7E+04	
	4.9E+01	P		1	1	Yes	~Dicalcium phosphate	7757-93-9					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	Р		1	1	Yes	~Dimagnesium phosphate	7782-75-4					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	Р		1	1	Yes	~Dipotassium phosphate	7758-11-4					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	P		1	1	Yes	~Disodium phosphate	7558-79-4					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	P		1	1	Yes	~Monoaluminum phosphate	13530-50-2					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01 4.9E+01	P		1	1	Yes	~Monocalcium phosphate	7758-23-8					9.7E+04 9.7E+04	2.2E+07 2.2E+07		9.7E+04 9.7E+04	
	4.9E+01	P		1	1	Yes	~Monomagnesium phosphate	7757-86-0					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	Р		1	1	Yes	~Monopotassium phosphate	7778-77-0					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	Р		1	1	Yes	~Monosodium phosphate	7558-80-7					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	P		1	1	Yes	~Polyphosphoric acid	8017-16-1					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	P		1	0.9	Yes	~Potassium tripolyphosphate	13845-36-8					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	P		1	1	Yes	-Sodium aluminum phosphate (acidic)	7785-88-8					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	P		1	0	Yes	~Sodium aluminum phosphate (anhydrous)	10279-59-1					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	Р		1	0.8	Yes	~Sodium aluminum phosphate (tetrahydrate)	10305-76-7					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	Р		1	0.9	Yes	~Sodium hexametaphosphate	10124-56-8					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	P		1	1	Yes	~Sodium polyphosphate	68915-31-1					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	P		1	1	Yes	~Sodium trimetaphosphate	7759 00 4					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01 4.9E+01	P		1	1	Yes	~Sodium tripolyphosphate	7758-29-4					9.7E+04 9.7E+04	2.2E+07		9.7E+04 9.7E+04	
	4.9E+01	P		1	1	Yes	~Tetrasodium pyrophosphate	7722-88-5					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	Р		1	0.8	Yes	~Trialuminum sodium tetra decahydrogenoctaorthophosphate (dihydrate)	15136-87-5					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	Р		1	1	Yes	~Tricalcium phosphate	7758-87-4					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	Р		1	1	Yes	~Trimagnesium phosphate	7757-87-1					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01	P		1	1	Yes	~Tripotassium phosphate	7778-53-2					9.7E+04	2.2E+07		9.7E+04	
	4.9E+01 3.0E-04	3.0E-04	V -0.2	7 1	1	Yes	~ msodrum phosphate Phosphine	7803-51-2					9.7E+04 6.0E-01	2.2E+07 1.4E+02	6.3E-02	5.7E+04	
	4.9E+01	P 1.0E-02 I	5.2	1	1	Yes	Phosphoric Acid	7664-38-2					9.7E+04	2.2E+07		9.7E+04	
	2.0E-05	1	V 3.08	3 1	1	Yes	Phosphorus, White	7723-14-0					4.0E-02	9.1E+00		4.0E-02	
							Phthalates										
1.4E-02 I 2.4E	E-06 C 2.0E-02	-	7.6	1	0.8	No	~Bis(2-ethylhexyl)phthalate	117-81-7	5.6E+00	0.75.00		5.6E+00	4.0E+01	0.05.00		4.0E+01	6.0E+00
1.9E-03 P	2.0E-01		4.73	3 1 5 1	0.9	Yes	~Butylobthalyl Butylolycolate	85-68-7 85-70-1	4.1E+01	2.7E+01		1.6E+01	4.0E+02 2.0E+03	2.9E+02 4 1E+03		1.7E+02 1.3E+03	
	1 0E-01		4.10	1	0.9	Yes	~Dibutyl Phthalate	84-74-2					2.0E+02	1.6E+02		9.0E+01	
	8.0E-01	i	2.42	2 1	1	Yes	~Diethyl Phthalate	84-66-2					1.6E+03	2.0E+04		1.5E+03	
	1.0E-01	1	V 2.25	5 1	1	Yes	~Dimethylterephthalate	120-61-6					2.0E+02	2.7E+03		1.9E+02	

Key: I = IRIS	; P = PPRTV; D	= DWSHA; (	D = OPP; A = A	TSDR; C	= Cal EPA	A; X = AF	PPEND	IX PPR	TV SCREEN (See FAQ #29); H = HEAST; F = See FAQ; E = see user guide Section : re: $n SL < 100 X c SL ** = where n SL < 10 X c SL * SSL values are based on DAE=1 r$	2.3.5; W = see	user guide Se	ction 2.3.6; L d ceiling limi	. = see user gu t (See User Gi	ide on lead; M = n uide): s = Concent	nutagen; S = s	ee user guide	e Section 5; V	/ = volatile; R = RBA a	pplied (See
	Т	oxicity and C	hemical-specif	ic Informa	ation		anoor,	- 1110	Contaminant	in = oonoonnaa	Car	cinogenic Ta	rget Risk (TR)	= 1E-06		Noncance	r CHILD Haza	ard Index (HI) = 0.1	
															Ingestion SL	Dermal SL	Inhalation SL	Noncarcinogenic SL	
SEO	k k	PfD	k RfC	v							Ingestion SL	Dermal SL	Inhalation SL	Carcinogenic SL	Child	Child	Child	Child	MOL
(ma/ka-day)-1	e 1010 e	(ma/ka-dav)	$e^{(m\alpha/m^3)}$	e o muta-		GIABS	FA Ir	EPD2	Analyte	CAS No	1R=1E-06	1R=1E-06 (ug/L)	IR=1E-06 (ug/L)	1R=1E-06 (ug/L)	HQ=0.1	(ug/L)	(ug/L)	1 HI=0.1	(ug/L)
(mg/ng day)	y (ug/iii / y	1 05 02	у (mg/m / :	i gen	0.1	1	0	No.	Ostul Bhthalata, di N	117.94.0	(ug/L)	(ug/L)	(ug/L)	(09/2)	2.0E+01	(ug/L)	(09/2)	2.05.01	(ug/L)
		1.0E+02	P L		0.1	1	1	Voc	~Octyl Phinalale, dr-N-	100-21-0					2.0E+01	3 35+04		2.0E+01 1.0E+03	
		2.0E+00	1 2.0E-02 (	:	1.6	i	1	Yes	~Phthalic Anhydride	85-44-9					4.0E+03	1.1E+05		3.9E+03	
		7 0E-02	1		1.9	1	1	Yes	Picloram	1918-02-1					1 4E+02	4.3E+03		1 4E+02	5.0E+02
		1.0E-04	x		0.93	1	1	Yes	Picramic Acid (2-Amino-4,6-dinitrophenol)	96-91-3					2.0E-01	2.1E+01		2.0E-01	
		9.0E-04	х		1.44	1	1	Yes	Picric Acid (2,4,6-Trinitrophenol)	88-89-1					1.8E+00	1.2E+02		1.8E+00	
		6.7E-05	0		4.2	1	0.9	Yes	Pirimiphos, Methyl	29232-93-7					1.3E-01	2.1E-01		8.1E-02	
3.0E+01	C 8.6E-03 C	7.0E-06	н			1	0	No	Polybrominated Biphenyls	59536-65-1	2.6E-03			2.6E-03	1.4E-02			1.4E-02	
									Polychlorinated Biphenyls (PCBs)										
7.0E-02	S 2.0E-05 S	7.0E-05	1	V	5.69	1	0	No	~Aroclor 1016	12674-11-2	1.1E+00		2.8E-01	2.2E-01	1.4E-01			1.4E-01	
2.0E+00	S 5.7E-04 S			V	4.65	1	1	Yes	~Aroclor 1221	11104-28-2	3.9E-02	1.2E-02	9.8E-03	4.7E-03					
2.0E+00	S 5.7E-04 S			V	6.24	1	0.7	No	Aradior 1242	F2460.21.0	3.9E-02	1.2E-02	9.02-03	4.7E-03					
2.0E+00	S 57E-04 S			v	6.2	1	0.7	No	~Alociol 1242	12672-20-6	3.9E-02		9.0E-03	7.62-03					
2.0E+00	S 5.7E-04 S	2.0E-05	1	v	6.5	i	0.5	No	~Aroclor 1248	11097-69-1	3.9E-02		9.8E-03	7.8E-03	4.0E-02			4.0E-02	
2.0E+00	S 5.7E-04 S			V	7,55	1	0	No	~Aroclor 1260	11096-82-5	3.9E-02		9.8E-03	7.8E-03					
		6.0E-04	Х	V	6.34	1	0.7	No	~Aroclor 5460	11126-42-4					1.2E+00			1.2E+00	
3.9E+00	E 1.1E-03 E	2.3E-05	E 1.3E-03 E	V	8.27	1	0	No	~Heptachlorobiphenyl, 2,3,3',4,4',5,5'- (PCB 189)	39635-31-9	2.0E-02		4.9E-03	4.0E-03	4.7E-02		2.8E-01	4.0E-02	
3.9E+00	E 1.1E-03 E	2.3E-05	E 1.3E-03 E	V	7.5	1	0	No	~Hexachlorobiphenyl, 2,3',4,4',5,5'- (PCB 167)	52663-72-6	2.0E-02		4.9E-03	4.0E-03	4.7E-02		2.8E-01	4.0E-02	
3.9E+00	E 1.1E-03 E	2.3E-05	E 1.3E-03 E	V	7.6	1	0	No	~Hexachlorobiphenyl, 2,3,3',4,4',5'- (PCB 157)	69782-90-7	2.0E-02		4.9E-03	4.0E-03	4.7E-02		2.8E-01	4.0E-02	
3.9E+00	E 1.1E-03 E	2.3E-05	E 1.3E-03 E	V	7.6	1	0	No	~Hexachlorobiphenyl, 2,3,3',4,4',5- (PCB 156)-	38380-08-4	2.0E-02		4.9E-03	4.0E-03	4.7E-02		2.8E-01	4.0E-02	
3.9E+03	E 1.1E+00 E	2.3E-08	E 1.3E-06 E	V	7.41	1	0.1	No	~Hexachlorobiphenyl, 3 3,4,4,5,5'- (PCB 169)	32774-16-6	2.0E-05		4.9E-06	4.0E-06	4.7E-05		2.8E-04	4.0E-05	
3.9E+00	E 1.1E-03 E	2.3E-05	E 1.3E-03 E	V	6.98	1	0.4	No	~Pentachlorobiphenyl, 2',3,4,4',5- (PCB 123)	65510-44-3	2.0E-02		4.9E-03	4.0E-03	4.7E-02		2.8E-01	4.0E-02	
3.9E+00	E 1.1E-03 E	2.3E-05	E 1.3E-03 E	: V	7.12	1	0.3	NO	~Pentachiorobiphenyi, 2,3,4,4,5-(PGB-118)	31508-00-6	2.0E-02		4.9E-03	4.0E-03	4.7E-02		2.8E-01	4.0E-02	
3.9E+00	E 1.1E-03 E	2.3E-05	E 1.3E-03 E	= V	6.79	1	0.5	No	~Pentachlorobiphenyl, 2,3,3',4,4'- (PCB 105)	32598-14-4	2.0E-02		4.9E-03	4.0E-03	4.7E-02		2.8E-01	4.0E-02	
3.9E+00	E 3.8E±00 E	Z.3E-03	E 1.3E-03 E	v V	6.98	1	0.4	No	~Pentachlorobiphenyl, 3,3',4,4',5- (PCB 114)	57465-28-8	2.0E-02 6.0E-06		4.9E-03	4.0E-03	4.7E-02		2.0E-01 8.3E-05	4.0E-02 1.2E-05	
2.0E+00	1 5 7E-04 1	1.0L 00	2 4.02 07 1	V	7 1	1	0.7	140	~Polychlorinated-Biphenyls (high risk)	1336-36-3	0.02 00		1.52 00	1.22 00	1.42 00		0.02 00	1.22 00	5.0E-01
4.0E-01	1 1.0E-04 1			v	7.1	1	0.7	No	~Polychlorinated Biphenvis (low risk)	1336-36-3	1.9E-01		5.6E-02	4.4E-02					5.0E-01
7.0E-02	I 2.0E-05 I			V	7.1	1	0.7	-	~Polychlorinated Biphenyls (lowest risk)	1336-36-3									5.0E-01
1.3E+01	E 3.8E-03 E	7.0E-06	E 4.0E-04 E		6.63	1	0.6	No	~Tetrachlorobiphenyl, 3(3,4,4+ (PCB.77) //	32598-13-3	6.0E-03			6.0E-03	1.4E-02			1.4E-02	
3.9E+01	E 1.1E-02 E	2.3E-06	E 1.3E-04 E	V	6.34	1	0.7	No	~Tetrachlorobiphenyl, 3,4,4',5-'(PCB'81)	70362-50-4	2.0E-03		4.9E-04	4.0E-04	4.7E-03		2.8E-02	4.0E-03	
			6.0E-04		10.46	1	0	No	Polymeric Methylene Diphenyl Diisocyanate (PMDI)	9016-87-9									
		0.05.00			0.00				Polynuclear Aromatic Hydrocarbons (PAHs)						1.05.00	0.05.04		5.05.04	
		6.0E-02 3.0E-01	-	v	3.92	1	1	Yes	~Acenaphthene	83-32-9					1.2E+02 6.0E+02	9.6E+01 2.5E±02		5.3E+01 1.8E±02	
1.0E-01	E 6.0E-05 E	0.02 01	•	V M	5.76	1	1	No	~Benz[a]anthracene	56-55-3	2.5E-01		3 4E-02	3.0E-02	0.02102	2.02102		1.02102	
1.2E+00	C 1.1E-04 C				6.11	1	0.9	No	~Benzo(j)fluoranthene	205-82-3	6.5E-02			6.5E-02					
1.0E+00	I 6.0E-04 I	3.0E-04	I 2.0E-06	М	6.13	1	1	No	~Benzo[a]pyrene	50-32-8	2.5E-02			2.5E-02	6.0E-01			6.0E-01	2.0E-01
1.0E-01	E 6.0E-05 E			М	5.78	1	1	No	~Benzo[b]fluoranthene	205-99-2	2.5E-01			2.5E-01					
1.0E-02	E 6.0E-06 E	0.05.00		М	6.11	1	0.9	No	~Benzo[k]fluoranthene	207-08-9	2.5E+00			2.5E+00	4.05.00	4 45 .00		7.55.04	
1.05.02		8.0E-02	1	V	3.9	1	1	res	~Chioronaphthalene, Beta-	91-56-7	0.55.04			2.55.04	1.6E+02	1.4E+02		7.5E+01	
1.0E-03	E 6.0E-07 E			M	6.75	1	0.6	No	~Dibenzía hlanthracene	53-70-3	2.5E+01 2.5E-02			2.5E+01					
1.2E+01	C 1.1E-03 C			171	7.71	1	0.3	No	~Dibenzo(a,e)pyrene	192-65-4	6.5E-03			6.5E-03					
2.5E+02	C 7.1E-02 C			М	5.8	1	0.9	No	~Dimethylbenz(a)anthracene, 7,12-	57-97-6	1.0E-04			1.0E-04					
		4.0E-02	1		5.16	1	1	No	~Fluoranthene	206-44-0					8.0E+01			8.0E+01	
		4.0E-02	1	V	4.18	1	1	Yes	~Fluorene	86-73-7					8.0E+01	4.6E+01		2.9E+01	
1.0E-01	E 6.0E-05 E			М	6.7	1	0.6	No	~Indeno[1,2,3-cd]pyrene	193-39-5	2.5E-01			2.5E-01					
2.9E-02	Р	7.0E-02	A	V	3.87	1	1	Yes	-Methylnaphthalene, 1-	90-12-0	2.7E+00	2.0E+00		1.1E+00	1.4E+02	1.1E+02		6.2E+01	
	2.45.05.0	4.0E-03	1 2.05.02	V	3.86	1	4	Tes	~wewymaphthalene, 2-	91-57-6			4.75.04	4 75 04	8.0E+00	0.52+00	0.05.04	3.0E+00	
1 2E±00	3.4E-05 C	2.0E-02	1 3.0E-03	v	3.3 4.75	1	0.0	Yes	~Naphthalene	91-20-3 57835-92-4	6.5E-02	2 7E-02	1.7E-01	1.7E-01 1.9E-02	4.0E+01	7.0E+01	6.3E-01	0.1E-01	
1.22100	0 1.12 04 0	3.0E-02	1	V	4.88	1	1	Yes	~Pvrene	129-00-0	0.02 02	2.7 2 02		1.52 02	6.0E+01	1.5E+01		1.2E+01	
		2.0E-02	Р			1	0	Yes	Potassium Perfluorobutane Sulfonate	29420-49-3					4.0E+01			4.0E+01	
1.5E-01	1	9.0E-03	1		4.1	1	0.9	Yes	Prochloraz	67747-09-5	5.2E-01	1.4E+00		3.8E-01	1.8E+01	5.1E+01		1.3E+01	
		6.0E-03	Н	V	5.58	1	0.8	Yes	Profluralin	26399-36-0					1.2E+01	3.3E+00		2.6E+00	
		1.5E-02	1		2.99	1	1	Yes	Prometon	1610-18-0					3.0E+01	1.6E+02		2.5E+01	
		4.0E-02	0		3.51	1	0.9	Yes	Prometryn	7287-19-6					8.0E+01	2.3E+02		6.0E+01	
		5.0E-02	-		2.10	1	1	Ves	Propanil	700-09-9					2.0E+01	4.32+02 4.4E+01		2.02+01 8.2E+00	
3.3E-02	0	4.0E-03	0		5	1	0.8	Yes	Proparaite	2312-35-8	2.4E+00	1.5E+00		9.2E-01	8.0E+01	5.5E+01		3.3E+01	
0.02 02	J	2.0E-02	ĩ	V	-0.38	1	1	Yes	Propargyl Alcohol	107-19-7	2.42100	1.02100		0.2L 01	4.0E+00	1.2E+03		4.0E+00	
		2.0E-02	1		2.93	1	1	Yes	Propazine	139-40-2					4.0E+01	2.4E+02		3.4E+01	
		2.0E-02	1		2.6	1	1	Yes	Propham	122-42-9					4.0E+01	2.8E+02		3.5E+01	
		1.0E-01	0		3.72	1	0.9	Yes	Propiconazole	60207-90-1					2.0E+02	8.2E+02		1.6E+02	
			8.0E-03	V	0.59	1	1	Yes	Propionaldehyde	123-38-6							1.7E+00	1.7E+00	

Key: I = IRI	S; P = PPRTV; D	= DWSHA; C	O = OPP; A	= ATSDF	R; C = C	al EPA;	X = AP	PEND		TV SCREEN (See FAQ #29); H = HEAST; F = See FAQ; E = see user guide Section	2.3.5; W = see	user guide Se	ction 2.3.6; I	_ = see user gu	ide on lead; M = n	nutagen; S = s	ee user guid	e Section 5; V =	= volatile; R = RBA a	pplied (See
	Т	oxicity and C	hemical-spe	c notice)	; c = ca	ncer, n	= nonca	ancer;	= whe	e: n SL < 100X c SL; = where n SL < 10X c SL; SSL values are based on DAF=1, 1 Contaminant	n = Concentrati	Car	cinogenic T	arget Risk (TR)	= 1E-06	ration may exc	Noncance	ee Oser Guide) er CHILD Hazar	/ rd Index (HI) = 0.1	
			T	TIT										<b>.</b>		Ingestion SL	Dermal SL	Inhalation SL	Noncarcinogenic SL	
SEO	k k	DID	k RfC	k v								Ingestion SL	Dermal SL	Inhalation SL	Carcinogenic SL	Child	Child	Child	Child	
SFU (ma/ka-day)	e IUR e	(mg/kg-day)		e o m	nuta-	OGP		FA Ir	EPD2	Analyte	CAS No	TR=1E-06	TR=1E-06	TR=1E-06	TR=1E-06	THQ=0.1	THQ=0.1	THQ=0.1	THI=0.1	MCL (ug/L)
(IIIg/kg-day)	y (ug/iii ) y	(IIIg/Kg-uay)	Y 1.0E+0	) y ' S	gen L		1	1	Voo.	Analyte Dropul henzono	102 65 1	(ug/L)	(ug/L)	(ug/L)	(Ug/L)	(Ug/L)	(Ug/L)	(dg/L)	(UG/L)	(ug/L)
		1.0E-01	3 0E+0			3.09 1 77	1	1	Yes	Propyl benzene Propylene	103-05-1					2.0E+02	1.6E+02	2.1E+02 6.3E+02	6.0E+01	
		2.0E+01	P		-	0.92	1	1	Yes	Propylene Glycol	57-55-6					4.0E+04	3.2E+07	0.02102	4.0E+04	
			2.7E-04	4 A		1.83	1	1	Yes	Propylene Glycol Dinitrate	6423-43-4									
		7.0E-01	H 2.0E+0	0 I V	-	0.49	1	1	Yes	Propylene Glycol Monomethyl Ether	107-98-2					1.4E+03	3.9E+05	4.2E+02	3.2E+02	
2.4E-01	I 3.7E-06 I		3.0E-02	2 I V	(	0.03	1	1	Yes	Propylene Oxide	75-56-9	3.2E-01	4.7E+01	1.5E+00	2.7E-01			6.3E+00	6.3E+00	
		7.5E-02	1		3	3.43	1	0.9	Yes	Propyzamide	23950-58-5					1.5E+02	5.5E+02		1.2E+02	
		1.0E-03		V	(	).65	1	1	Yes	Pyridine	110-86-1					2.0E+00	1.5E+02		2.0E+00	
0.05.00		5.0E-04	I		4	4.44	1	0.9	Yes	Quinalphos	13593-03-8	0.05.00	0.05.04		0.45.00	1.0E+00	1.0E+00		5.1E-01	
3.0E+00	I	9.0E-03			-	2.03	1	1	Yes	Quinoiine Quizalofon-ethyl	91-22-5 76578-14-8	2.6E-02	2.9E-01		2.4E-02	1.8E+01	3.8E±01		1 2E+01	
-		3.0L-03	3 0E-01	2 1		+.20	1	0.5	Voc	Refractory Coramic Eibers	E715557					1.02+01	3.0L+01		1.22401	
		3.0E-02	0.0Ľ-02	- ^	e	5.14	1	0.7	Yes	Resmethrin	10453-86-8					6.0E+01	7.6E+00		6.7E+00	
		5.0E-02	Ĥ	V	4	4.88	1	0.8	Yes	Ronnel	299-84-3					1.0E+02	6.8E+01		4.1E+01	
		4.0E-03	1			4.1	1	0.9	Yes	Rotenone	83-79-4					8.0E+00	2.6E+01		6.1E+00	
2.2E-01	C 6.3E-05 C				M 3	3.45	1	1	Yes	Safrole	94-59-7	1.1E-01	6.0E-01		9.6E-02					
		5.0E-03	1				1	1	Yes	Selenious Acid	7783-00-8					1.0E+01	2.3E+03		1.0E+01	
		5.0E-03	1 2.0E-02	2 C			1	1	Yes	Selenium	7782-49-2					1.0E+01	2.3E+03		1.0E+01	5.0E+01
		5.0E-03	C 2.0E-02	2 C		4.00	1	1	Yes	Selenium Sulfide	7446-34-6					1.0E+01	2.3E+03		1.0E+01	
		1.4E-01	0		4	4.38	1	0.9	res		74051-80-2		_			2.8E+02	3.8E+02		1.6E+02	
		5.0E-02	3.0E-03	50			1	1	Yes	Silver	7631-86-9					1.05+01	1.55+02		9.4E+00	
1.2E-01	н	5.0E-03				2.18	1	1	Yes		122-34-9	6.5E-01	9.3E+00		6.1E-01	1.0E+01	1.6E+02		9.4E+00	4.0E+00
1.22 01		1.3E-02	1			1.37	1	1	Yes	Sodium Abifluorfen	62476-59-9	0.02 01	3.0L100		0.12 01	2.6E+01	2 1E+04		2.6E+01	4.02100
		4.0E-03	i			5.01	1	1	Yes	Sodium Azide	26628-22-8					8.0E+00	1.8E+03		8.0E+00	
2.7E-01	н	3.0E-02	i		-	1.43	1	1	Yes	Sodium Diethyldithiocarbamate	148-18-5	2.9E-01	8.5E+02		2.9E-01	6.0E+01	1.9E+05		6.0E+01	
		5.0E-02	A 1.3E-02	2 C			1	1	Yes	Sodium Fluoride	7681-49-4					1.0E+02	2.3E+04		1.0E+02	
		2.0E-05	1		-	3.78	1	1	No	Sodium Fluoroagetate	62-74-8					4.0E-02			4.0E-02	
		1.0E-03	Н				1	1	Yes	Sodium Metavanadate //	13718-26-8					2.0E+00	4.6E+02		2.0E+00	
		8.0E-04	P				1	1	Yes	Sodium Tungstate	13472-45-2					1.6E+00	3.6E+02		1.6E+00	
0.45.00		8.0E-04	P				1	1	Yes	Sodium Tungstate Dihydrate	10213-10-2	2.05.00	4.05.04		0.05.00	1.6E+00	3.6E+02		1.6E+00	
2.4E-02	п	3.0E-02	-			3.33	1	0.9	res	Survive Order	961-11-5	3.2E+00	1.9E+01		2.8E+00	6.0E+01	3.8E+02		5.2E+01	
		6.0E-01	-			1 03	1	1	Yes	Strontium, Stable	7440-24-6					1.2E+03 6.0E-01	2.7E+05 3.2E+01		1.2E+03	
		2.0E-04	1 1 0E+0	0 I V		2 95	1	1	Yes	Stryennie	100-42-5					4 0E+02	1.0E+03	2 1E+02	1 2E+02	1.0E+02
		3.0E-03	P			3.1	1	1	Yes	Styrene-Acrylonitrile (SAN) Trimer						6.0E+00	2.4E+01		4.8E+00	
		1.0E-03	P 2.0E-03	зх	-	0.77	1	1	Yes	Sulfolane	126-33-0					2.0E+00	1.7E+03		2.0E+00	
		8.0E-04	Р			3.9	1	0.9	Yes	Sulfonylbis(4-chlorobenzene), 1,1'-	80-07-9					1.6E+00	3.5E+00		1.1E+00	
			1.0E-03	зсv			1	1	Yes	Sulfur Trioxide	7446-11-9							2.1E-01	2.1E-01	
			1.0E-03	зС			1	1	Yes	Sulfuric Acid	7664-93-9									
2.5E-02	I 7.1E-06 I	5.0E-02	Н		4	4.82	1	0.8	Yes	Sulturous acid, 2-chloroethyl 2-[4-(1,1-dimethylethyl)phenoxy]-1-methylethyl ester	140-57-8	3.1E+00	2.3E+00		1.3E+00	1.0E+02	8.2E+01		4.5E+01	
		3.0E-02	Н			3.3	1	0.9	Yes	TCMIB	21564-17-0					6.0E+01	2.4E+02		4.8E+01	
		2.0E-02	Н			1.79 5.96	1	1	res	Temenhos	34014-18-1					1.4E+02 4.0E+01	4.7E+03		1.4E+02 4.0E±01	
		1.3E-02				1.89	1	1	Yes	Terhacil	5902-51-2					2.6E±01	7 0E±02		2.5E+01	
		2.5E-05	Н	V	4	4.48	1	0.9	Yes	Terbufos	13071-79-9					5.0E-02	4.5E-02		2.4E-02	
		1.0E-03	I	-	3	3.74	1	0.9	Yes	Terbutryn	886-50-0					2.0E+00	4.1E+00		1.3E+00	
		1.0E-04	I		6	6.77	1	0.6	No	Tetrabromodiphenyl ether, 2,2',4,4'- (BDE-47)	5436-43-1					2.0E-01			2.0E-01	
		3.0E-04	I	V	4	4.64	1	1	Yes	Tetrachlorobenzene, 1,2,4,5-	95-94-3					6.0E-01	2.4E-01		1.7E-01	
2.6E-02	I 7.4E-06 I	3.0E-02	1	V	2	2.93	1	1	Yes	Tetrachloroethane, 1,1,1,2-	630-20-6	3.0E+00	1.1E+01	7.6E-01	5.7E-01	6.0E+01	2.4E+02		4.8E+01	
2.0E-01	I 5.8E-05 C	2.0E-02		V	2	2.39	1	1	Yes	Tetrachloroethane, 1,1,2,2-	79-34-5	3.9E-01	3.3E+00	9.7E-02	7.6E-02	4.0E+01	3.6E+02		3.6E+01	
2.1E-03	I 2.6E-07 I	6.0E-03	1 4.0E-02	210		3.4	1	1	Yes	I etrachloroethylene	127-18-4	3.7E+01	6.5E+01	2.2E+01	1.1E+01	1.2E+01	2.3E+01	8.3E+00	4.1E+00	5.0E+00
0.05.01		3.0E-02	I		4	4.45	1	0.9	res	Tetrachiorophenoi, 2,3,4,6-	58-90-2	0.05.00	0.05.00		4.05.00	6.0E+01	3.9E+01		2.4E+01	
2.0E+01	n	5.0E-04		v	4	+.54	1	0.9	Tes	Tetrachiorototuene, p- alpha, alpha, alpha- Tetraethyl Dithiopyrophosphate	3689-24-5	3.9E-03	2.0E-03		1.3E-03	1.0E±00	2 4E+00		7 1E-01	
		0.0L-04	8.0E+0	1 I V		1.68	1	1	Yes	Tetrafluoroethane, 1.1.1.2-	811-97-2					1.02+00	2.46700	1.7E+04	1.7E+04	
		2.0E-03	P			1.64	1	1	Yes	Tetryl (Trinitrophenylmethylnitramine)	479-45-8					4.0E+00	2.5E+02		3.9E+00	
		2.0E-05	S				1	0.9	Yes	Thallic Oxide	1314-32-5					4.0E-02	9.1E+00		4.0E-02	
		1.0E-05	Х				1	1	Yes	Thallium (I) Nitrate	10102-45-1					2.0E-02	4.6E+00		2.0E-02	
		1.0E-05	Х				1	1	Yes	Thallium (Soluble Salts)	7440-28-0					2.0E-02	4.6E+00		2.0E-02	2.0E+00
		1.0E-05	X	V	-	0.17	1	1	Yes	Thallium Acetate	563-68-8					2.0E-02	1.7E+01		2.0E-02	
		2.0E-05	X	V	-	0.86	1	1	Yes	I hallium Carbonate	6533-73-9					4.0E-02	9.3E+03		4.0E-02	
		1.0E-05	X				1	1	Yes	I hallium Chloride	7791-12-0					2.0E-02	4.6E+00		2.0E-02	
		1.0E-05 2.0E-05	S X				1	00	Yes	Thallium Sulfate	7446-18-6					2.0E-02 4.0E-02	4.6E+00 9.1E+00		2.0E-02 4.0E-02	
		4.3E-02	0			1.56	1	1	Yes	Thifensulfuron-methyl	79277-27-3					8.6E+01	1.2E+04		8.6E+01	
		1.0E-02	I			3.4	1	0.9	Yes	Thiobencarb	28249-77-6					2.0E+01	7.7E+01		1.6E+01	

Key: I = IRIS	; P = PPRTV; D	= DWSHA; C	D = OPP; A = AT	SDR; C = Cal El	PA; X = .	APPEN	NDIX PPR	TV SCREEN (See FAQ #29); H = HEAST; F = See FAQ; E = see user guide Section 2	2.3.5; W = see	user guide Se	ction 2.3.6;	_ = see user gu	iide on lead; M = r	nutagen; S = s	see user guid	le Section 5; V	= volatile; R = RBA a	pplied (See
	т	User Guid	le for Arsenic not	tice) ; c = cancer	; n = nor	ncance	r; * = whei	re: n SL < 100X c SL; ** = where n SL < 10X c SL; SSL values are based on DAF=1; n	n = Concentrati	on may excee	d ceiling lim	it (See User G	uide); s = Concent	tration may ex	ceed Csat (S	ee User Guide	) rd Index (HI) = 0.1	
			nernical-specific	monnation	1			Contaminant		Ca	chogenic i		= 12-00	Ingestion SL	Dermal SL	Inhalation SL	Noncarcinogenic SL	
	k k		k k	v						Ingestion SL	Dermal SL	Inhalation SL	Carcinogenic SL	Child	Child	Child	Child	
SFO	e IUR e	RfD <b>。</b>	e <sup>RfC</sup> i e	o muta-						TR=1E-06	TR=1E-06	TR=1E-06	TR=1E-06	THQ=0.1	THQ=0.1	THQ=0.1	THI=0.1	MCL
(mg/kg-day) <sup>-1</sup>	y (ug/m <sup>3</sup> ) <sup>-1</sup> y	(mg/kg-day)	y (mg/m <sup>3</sup> ) y	I gen LOGP	GIAB	S FA	In EPD?	Analyte	CAS No.	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)	(ug/L)
		7.0E-02	Х	-0.63	1	1	Yes	Thiodiglycol	111-48-8					1.4E+02	9.7E+04		1.4E+02	
		3.0E-04	Н	2.16	1	1	Yes	Thiofanox	39196-18-4					6.0E-01	4.4E+00		5.3E-01	
1.2E-02	0	2.7E-02	0	1.4	1	1	Yes	Thiophanate, Methyl	23564-05-8	6.7E+00	7.9E+02		6.7E+00	5.4E+01	6.8E+03		5.3E+01	
		1.5E-02	0	1.73	1	1	Yes	Thiram	137-26-8					3.0E+01	1.2E+03		2.9E+01	
		6.0E-01	Н		1	1	Yes	Tin	7440-31-5					1.2E+03	2.7E+05		1.2E+03	
		8 0E-02	1.0E-04 A	V V 273	1	1	Yes	Taluana	108-88-3					1.65+02	5 3E+02	2.1E-02 1.0E+03	2.1E-02 1.1E+02	1.0E+03
	1 1E-05 C	0.02-02	8 0E-06 C	V 2.75	1	1	Voc	Toluono-2 4-diisocuanate	584-84-9			5 1E-01	5 1E-01	1.02+02	J.JL+02	1.75-03	1.7E-03	1.02+03
1.8E-01	X 1.12 00 0	2 0E-04	X 0.02 00 0	0.16	1	1	Yes	Toluene-2.5-diamine	95-70-5	4 3E-01	8 2E+01	0.12 01	4.3E-01	4 0E-01	8.3E+01	1.72 00	4.0E-01	
	1.1E-05 C	2.02 01	8.0E-06 C	V 3.74	1	1	Yes	Toluene-2,6-diisocyanate	91-08-7		0.22101	5.1E-01	5.1E-01		0.02101	1.7E-03	1.7E-03	
		5.0E-03	Р	2.27	1	1	Yes	Toluic Acid, p-	99-94-5					1.0E+01	8.9E+01		9.0E+00	
1.6E-02	P 5.1E-05 C			1.32	1	1	Yes	Toluidine, o- (Methylaniline, 2-)	95-53-4	4.9E+00	1.4E+02		4.7E+00					
3.0E-02	Р	4.0E-03	Х	1.39	1	1	Yes	Toluidine, p-	106-49-0	2.6E+00	6.8E+01		2.5E+00	8.0E+00	2.3E+02		7.7E+00	
		3.0E+00	P	V 6.1	1	1	No	Total Petroleum Hydrocarbons (Aliphatic High)	E1790670					6.0E+03			6.0E+03	
			6.0E-01 P	V 3.9	1	1	Yes	Total Petroleum Hydrocarbons (Aliphatic Low)	E1790666							1.3E+02	1.3E+02	
		1.0E-02	X 1.0E-01 P	V 5.65	1	1	No	Total Petroleum Hydrocarbons (Aliphatic Medium)	E1790668					2.0E+01		2.1E+01	1.0E+01	
		4.0E-02	P	5.16	1	1	No	Total Petroleum Hydrocarbons (Aromatic High)	E1790676					8.0E+01			8.0E+01	
		4.0E-03	P 3.0E-02 P	V 2.13	1	1	Yes	Total Petroleum Hydrocarbons (Aromatic Low)	E1790672					8.0E+00	6.1E+01	6.3E+00	3.3E+00	
4.45.00	1 2 25 04	4.0E-03	P 3.0E-03 P	v 3.58	1	1	Yes	Total Petroleum Hydrocarbons (Aromatic Medium)	E1790674	745.00			7 45 00	8.0E+00	9.0E+00	6.3E-01	5.5E-01	2.05.02
1.1E+00	1 3.2E-04 I	7 55 00		5.9	1	0.8	NO No		8001-35-2	7.1E-02			7.1E-02	4.55.04			4.55.04	3.0E+00
		7.5E-03 3.0E-04	Δ ,	7.50 V 4.1	1	0.5	Ves		688-73-3					6.0E-01	9 9E-01		3.7E-01	
		8.0E±01	X	0.25	1	1	Voc	Triacetin	102-76-1					1.6E±05	5.3E±07		1.6E+05	
		3.4E-02	ô	2.77	1	1	Yes	Triadimefon	43121-43-3					6.8E+01	7.8E+02		6.3E+01	
7.2E-02	0	2.5E-02	0 '	V 4.6	1	0.9	Yes	Triallate	2303-17-5	1.1E+00	8.3E-01		4.7E-01	5.0E+01	4.2E+01		2.3E+01	
		1.0E-02	1	1.1	1	1	Yes		82097-50-5					2.0E+01	6.0E+03		2.0E+01	
		8.0E-03	1	0.78	1	1	Yes	Tribenuron-methyl	101200-48-0					1.6E+01	5.0E+02		1.6E+01	
		5.0E-03	1	V 4.66	1	0.9	Yes	Tribromobenzene, 1,2,4-	615-54-3					1.0E+01	8.1E+00		4.5E+00	
		9.0E-03	Х	4.13	1	0.9	Yes	Tribromophenol, 2,4,6-	118-79-6					1.8E+01	3.7E+01		1.2E+01	
9.0E-03	Р	1.0E-02	P	4	1	0.9	Yes	Tributyl Phosphate	126-73-8	8.7E+00	1.3E+01		5.2E+00	2.0E+01	3.3E+01		1.2E+01	
		3.0E-04	Р		1	0	No	Tributyltin Compounds	E1790678					6.0E-01			6.0E-01	
		3.0E-04		4.05	1	1	Yes	Tributyitin Oxide	56-35-9					6.0E-01	9.5E+00	4.05.00	5.7E-01	
7.0E-02	1	2.0E+01	1 5.0E+00 P	V 3.10 1.33	1	1	Yes	Trichloro-1,2,2-trilluoroethane, 1,1,2-	76-13-1	1 1E+00	4 6E±01		1 1E+00	6.0E+04	1.9E+05	1.0E+03	1.0E+03 3.9E+01	6.0E+01
2.9E-02	H	2.02 02	•	-0.67	1	1	Yes	Trichloroaniline HCL 2.4.6-	33663-50-2	2.7E+00	3.7E+03		2 7E+00		1.02100		0.02101	0.02101
7.0E-03	x	3.0E-05	х	3.52	1	1	Yes	Trichloroaniline, 2,4,6-	634-93-5	1.1E+01	2.0E+01		7.1E+00	6.0E-02	1.2E-01		4.0E-02	
		8.0E-04	Х	V 4.05	1	1	Yes	Trichlorobenzene, 1,2,3-	87-61-6					1.6E+00	1.3E+00		7.0E-01	
2.9E-02	Р	1.0E-02	I 2.0E-03 P	V 4.02	1	1	Yes	Trichlorobenzene, 1,2,4-	120-82-1	2.7E+00	2.0E+00		1.2E+00	2.0E+01	1.6E+01	4.2E-01	4.0E-01	7.0E+01
		2.0E+00	I 5.0E+00 I	V 2.49	1	1	Yes	Trichloroethane, 1,1,1-	71-55-6					4.0E+03	2.5E+04	1.0E+03	8.0E+02	2.0E+02
5.7E-02	I 1.6E-05 I	4.0E-03	I 2.0E-04 X	V 1.89	1	1	Yes	Trichloroethane, 1,1,2-	79-00-5	1.4E+00	2.0E+01	3.5E-01	2.8E-01	8.0E+00	1.3E+02	4.2E-02	4.1E-02	5.0E+00
4.6E-02	I 4.1E-06 I	5.0E-04	1 2.0E-03 1	V M 2.42	1	1	Yes	I richloroethylene	79-01-6	1.2E+00	7.4E+00	9.6E-01	4.9E-01	1.0E+00	6.9E+00	4.2E-01	2.8E-01	5.0E+00
		3.0E-01		V 2.53	1	1	Yes	I richlorofluoromethane	75-69-4					6.0E+02	3.6E+03		5.2E+02	
1 15 02	1 2 1 5 0 6	1.0E-01	D D	3.12	1	1	Vee	Trichlerenhandl 2.4.6	99.06.2	7 15,00	0.95.00		4.15+00	2.00+02	2.00		1.20+02	
1.12-02	1 3.12-00 T	1.0E-03		3.09	1	00	Yes	Trichlorophenoxyacetic Acid. 2.4.5-	93-76-5	7.12+00	5.0E+00		4.12+00	2.0E+00	8.7E+00		1.6E+01	
		8.0E-03	I	3.8	1	0.9	Yes	Trichlorophenoxypropionic acid, -2,4,5	93-72-1					1.6E+01	3.6E+01		1.1E+01	5.0E+01
		5.0E-03	1	V 2.43	1	1	Yes	Trichloropropane, 1,1,2-	598-77-6					1.0E+01	7.5E+01		8.8E+00	
3.0E+01	1	4.0E-03	I 3.0E-04 I	V M 2.27	1	1	Yes	Trichloropropane, 1,2,3-	96-18-4	8.4E-04	7.3E-03		7.5E-04	8.0E+00	7.7E+01	6.3E-02	6.2E-02	
		3.0E-03	X 3.0E-04 P	V 2.78	1	1	Yes	Trichloropropene, 1,2,3-	96-19-5					6.0E+00	2.6E+01	6.3E-02	6.2E-02	
		2.0E-02	A	5.11	1	0.8	Yes	Tricresyl Phosphate (TCP)	1330-78-5					4.0E+01	2.6E+01		1.6E+01	
		3.0E-03	1	5.18	1	0.8	Yes	Tridiphane	58138-08-2					6.0E+00	2.6E+00		1.8E+00	
			7.0E-03 I	V 1.45	1	1	Yes	Triethylamine	121-44-8							1.5E+00	1.5E+00	
		2.0E+00	P	-1.75	1	1	Yes	Triethylene Glycol	112-27-6					4.0E+03	1.8E+07		4.0E+03	
7 75 02		7 55 02	2.0E+01 P	V 1.74	1	1	Yes	Influoroethane, 1,1,1-	420-46-2	1.05.01	2 45.00		2.65.00	1.55.01	E EE 100	4.2E+03	4.2E+03	
2.0E.02	I D	1.0E-03	D D	V 5.54	1	0.0	Vee	Trimotanii Trimothyl Bhoophoto	F10 F6 1	2.0E+00	3.4L+00		2.02+00	1.5E+01	1.65.04		4.0E+00	
2.0E-02	r -	1.0E-02	F	-0.05 V 3.66	1	1	Yes	Trimethylhenzene 123-	526-73-8	3.9E+00	2.0E+03		3.9E+00	2.0E+01	1.0E+04	1 3E±01	2.0E+01	
		1.0E-02	I 6.0E-02 I	V 3.63	1	1	Yes	Trimethylbenzene, 1,2,4-	95-63-6					2.0E+01	2.0E+01	1.3E+01	5.6E+00	
		1.0E-02	1 6.0E-02 1	V 3.42	1	1	Yes	Trimethylbenzene, 1,3,5-	108-67-8					2.0E+01	2.8E+01	1.3E+01	6.0E+00	
		1.0E-02	Х	V 4.08	1	1	Yes	Trimethylpentene, 2,4,4-	25167-70-8					2.0E+01	9.6E+00		6.5E+00	
		3.0E-02	1	1.18	1	1	Yes	Trinitrobenzene, 1,3,5-	99-35-4					6.0E+01	4.7E+03		5.9E+01	
3.0E-02	1	5.0E-04	1	1.6	1	1	Yes	Trinitrotoluene, 2,4,6-	118-96-7	2.6E+00	1.1E+02		2.5E+00	1.0E+00	4.5E+01		9.8E-01	
		2.0E-02	P	2.83	1	1	Yes	Triphenylphosphine Oxide	791-28-6					4.0E+01	3.8E+02		3.6E+01	
		2.0E-02	A	3.65	1	0.9	Yes	Tris(1,3-Dichloro-2-propyl) Phosphate	13674-87-8					4.0E+01	3.2E+02		3.6E+01	
2.25.00	C 6 65 04 C	1.0E-02	X	2.59	1	1	Yes	Tris(1-cnioro-2-propyl)phosphate	13674-84-5	2 4E 02		9 EE 00	6 9E 02	2.0E+01	3.8E+02		1.9E+01	
2.3E+00 2.0E-02	P 0.0E-04 C	7 0E-03	Р	v 4.29	1	1	Yes	Tris(2,5-dibronethyl)phosphate	120-72-7	3.4E-02 3.9E+00	3.0E+02	6.5E-03	3.8E+00	1 4E+01	1 2E+03		1 4E±01	
2.02-02	•	1.02-03		1.44			100	not construction optimic	. 10 00-0	0.02700	U.ULTUZ		0.02700	1.46401	1.22703		1.42701	

Key: I = IRIS	ey: I = IRIS; P = PPRTV; D = DWSHA; O = OPP; A = ATSDR; C = Cal EPA; X = APPENDIX PPRTV SCREEN (See FAQ #29); H = HEAST; F = See user guide Section 2.3.5; W = see user guide Section 2.3.6; L = see user guide on lead; M = mutagen; S = see user guide Section 5; V = volatile; R = RBA applied (See User Guide for Arsenic notice) : c = cancer; n = noncancer; * = where: n SL < 10X c SL; ** = where n SL < 10X c SL; * = where n SL < 10X c SL; * = see user guide Section DAF=1; m = Concentration may exceed ceiling limit (See User Guide): s = Concentration may exceed Csat (See User Guide)																			
	Т	oxicity and C	Chemical-sp	ecific Inform	nation					Contaminant		Ca	rcinogenic Ta	rget Risk (TR)	= 1E-06		Noncance	er CHILD Haze	rd Index (HI) = 0.1	
SFO (mg/kg-day) <sup>-1</sup>	k k e IUR e y (ug/m <sup>3)-1</sup> y	RfD <b>。</b> (mg/kg-day)	k e RfC <sub>i</sub> y (mg/m	kv eomut )ylge	ta- n LOG	€P GI	IABS F	A In I	EPD?	Analyte	CAS No.	Ingestion SL TR=1E-06 (ug/L)	Dermal SL TR=1E-06 (ug/L)	Inhalation SL TR=1E-06 (ug/L)	Carcinogenic SL TR=1E-06 (ug/L)	Ingestion SL Child THQ=0.1 (ug/L)	Dermal SL Child THQ=0.1 (ug/L)	Inhalation SL Child THQ=0.1 (ug/L)	Noncarcinogenic SL Child THI=0.1 (ug/L)	MCL (ug/L)
3.2E-03	Р	1.0E-01	Р		9.4	9	1	0	No	Tris(2-ethylhexyl)phosphate	78-42-2	2.4E+01			2.4E+01	2.0E+02			2.0E+02	
		8.0E-04	Р				1	1 Y	'es	Tungsten	7440-33-7					1.6E+00	3.6E+02		1.6E+00	
		2.0E-04	A 4.0E-0	5 A			1	1 ۱	'es	Uranium (Soluble Salts)	E715565					4.0E-01	9.1E+01		4.0E-01	3.0E+01
1.0E+00	C 2.9E-04 C			M	1 -0.1	5	1	1 ١	'es	Urethane	51-79-6	2.5E-02	6.1E+00		2.5E-02	1 1				
	8.3E-03 P	9.0E-03	I 7.0E-0	5 P		0.	.026	1)	es	Vanadium Pentoxide	1314-62-1					1.8E+01	1.1E+02		1.5E+01	
		5.0E-03	S 1.0E-0	4 A		0.	.026	1	'es	Vanadium and Compounds	7440-62-2					1.0E+01	6.0E+01		8.6E+00	
		1.0E-03	1	V	3.8	4	1	1	es	Vernolate (m) (m)	1929-77-7					2.0E+00	2.5E+00		1.1E+00	
		1.2E-03	0		3.1	L I	1 0	.9 \	es	Vinclozolin Lev comp Less III I (see	50471-44-8					2.4E+00	1.8E+01		2.1E+00	
		1.0E+00	H 2.0E-0	1 I V	0.7	3	1	1 1	es	Vinyl Acetate CCCC CCCC	108-05-4					2.0E+03	1.4E+05	4.2E+01	4.1E+01	
	3.2E-05 H		3.0E-0	3 I V	1.5	7	1	1)	es	Vinyl Bromide (J. (J. V.) crifting United View View	593-60-2			1.8E-01	1.8E-01	l		6.3E-01	6.3E-01	
7.2E-01	I 4.4E-06 I	3.0E-03	I 1.0E-0		1 1.3	5	1	1 1	es	Vinyl Chloride	75-01-4	2.1E-02	2.8E-01	3.4E-01	1.9E-02	6.0E+00	8.9E+01	2.1E+01	4.4E+00	2.0E+00
		3.0E-04			2.1		1	1 1	es	warrarin	81-81-2					6.0E-01	8.4E+00		5.6E-01	
		2.0E-01	S 1.0E-0	1 S V	3.1	ō	1	1	es	Xylene, P-	106-42-3					4.0E+02	7.6E+02	2.1E+01	1.9E+01	
		2.0E-01	S 1.0E-0	1 S V	3.2	<u> </u>	1	1	es	Xylene, m- company and a company	108-38-3					4.0E+02	7.1E+02	2.1E+01	1.9E+01	
		2.0E-01	S 1.0E-0	150	3.1	2	1	ו 1	'es	Xylene, o-	95-47-6					4.0E+02	8.0E+02	2.1E+01	1.9E+01	
2.0E-01   1.0E-01   V 3.16 1 1 Yes Xylenes						Xylenes	1330-20-7					4.0E+02	7.5E+02	2.1E+01	1.9E+01	1.0E+04				
3.0E-04 1 1 Yes Zinc Phosphide						1314-84-7					6.0E-01	2.3E+02		6.0E-01						
3.0E-01 I 1 Yes Zinc and Compounds of Proceedings of Statement					/440-66-6					6.0E+02	2.3E+05		6.0E+02							
5.0E-02 I 1.3 1 1 Yes Zineb						12122-67-7					1.0E+02	9.7E+03		9.9E+01						
	8.0E-05 X 1 1 Yes Zirconium					7440-67-7					1.6E-01	3.6E+01		1.6E-01						

Appendix D

**Groundwater Sampling Schedules (2019-2020)** 

# April 2019 Groundwater Sampling Schedule

Well Name	<u>Well ID</u>	<u>Plume Area</u>	<u>Nitrates</u>	<u>Sulfate</u>	VOC	DNT	<u>Well Type</u>	Sample Frequency
WE-ZE512	437	Central			1	1	Residential Well	Quarterly
Purcell-D	163	DBG			1	1	Residential Well	Quarterly
ELN-8203A	210	DBG		1	1	1	Monitoring Well	Semi-Annual
ELN-8203B	211	DBG		1	1	1	Monitoring Well	Semi-Annual
ELN-8203C	212	DBG		1	1	1	Monitoring Well	Semi-Annual
ELM-8901	216	DBG		1	1	1	Monitoring Well	Semi-Annual
ELM-8907	220	DBG		1	1	1	Monitoring Well	Semi-Annual
ELM-8908	221	DBG		1	1	1	Monitoring Well	Semi-Annual
ELM-8909	222	DBG		1	1	1	Monitoring Well	Semi-Annual
ELN-8902B	224	DBG		1	1	1	Monitoring Well	Semi-Annual
ELN-9107A	227	DBG		1	1	1	Monitoring Well	Semi-Annual
ELN-9107B	228	DBG		1	1	1	Monitoring Well	Semi-Annual
ELN-9402AR	231	DBG		1	1	1	Monitoring Well	Semi-Annual
ELM-9501	234	DBG			1	1	Monitoring Well	Semi-Annual
S1134R	236	DBG		1	1	1	Monitoring Well	Semi-Annual
DBM-8201	301	DBG		1	1	1	Monitoring Well	Semi-Annual
DBM-8202	302	DBG		1	1	1	Monitoring Well	Semi-Annual
DBM-8903	306	DBG		-	-	-	Monitoring Well	Semi-Annual
DBN-9501A	314	DBG			-	-	Monitoring Well	Semi-Annual
DBN-9501B	315	DBG			1	1	Monitoring Well	Semi-Annual
DBN-9501C	316	DBG			1	1	Monitoring Well	Semi-Annual
DBN-95016	317	DBG			1	1	Monitoring Well	Semi-Annual
FLN-0801B	455	DBG			1	1	Monitoring Well	Semi-Annual
ELN-0801C	456	DBG			1	1	Monitoring Well	Semi-Annual
ELN-0801C	457	DBG			1	1	Monitoring Well	Semi-Annual
ELN-1001B	460	DBG			1	1	Monitoring Well	Semi-Annual
ELN-1001C	461	DBG			1	1	Monitoring Well	Semi-Annual
ELN-1001C	401	DBG			1	1 1	Monitoring Well	Semi-Annual
	402				1	1	Monitoring Well	Somi Annual
	405				1	1	Monitoring Well	Somi Annual
	404				1	1	Monitoring Well	Somi Annual
	405	DBG			1	1	Monitoring Well	Semi-Annual
	400				1	1	Monitoring Well	Quartarly
ELIN-1003A	407				1	1	Monitoring Well	Quarterly
ELIN-1003B	408	DBG			1	1	Monitoring Well	Quarterly
ELN-1003C	409	DBG			1	1	Monitoring Well	Quarterly
ELN-1003E	470	DBG			1	1	Monitoring Well	Quarteriy
DBN-1001B	472	DBG			1	1	Monitoring Well	Semi-Annual
DBN-1001C	4/3	DBG			1	1	Monitoring Well	Semi-Annual
DBN-1001E	4/4	DBG			1	1	Monitoring Well	Semi-Annual
DBN-1002C	476	DBG		1	1	1	Monitoring Well	Semi-Annual
DBN-1002E	4//	DBG		1	1	1	Monitoring Well	Semi-Annual
S1121	/55	DBG			1	1	Monitoring Well	Semi-Annual
ELN-1502A	533	DBG			1	1	Monitoring Well	Semi-Annual
ELN-1502C	534	DBG			1	1	Monitoring Well	Semi-Annual
ELN-1503A	535	DBG			1	1	Monitoring Well	Semi-Annual
ELN-1503C	536	DBG			1	1	Monitoring Well	Semi-Annual
ELN-1504B	537	DBG			1	1	Monitoring Well	Quarterly
RIM-0705	442	NC Area				1	Monitoring Well	Semi-Annual
RIM-1002	478	NC Area				1	Monitoring Well	Semi-Annual
RIN-1001A	480	NC Area				1	Monitoring Well	Semi-Annual
S1125	504	NC Area				1	Monitoring Well	Semi-Annual
PBM-0001	367	PBG	1		1	1	Monitoring Well	Semi-Annual
PBM-0002	368	PBG	1		1	1	Monitoring Well	Semi-Annual

# April 2019 Groundwater Sampling Schedule

<u>Well Name</u>	Well ID	<u>Plume Area</u>	<u>Nitrates</u>	<u>Sulfate</u>	<u>VOC</u>	<u>DNT</u>	Well Type	Sample Frequency
PBM-0006	372	PBG	1		1	1	Monitoring Well	Semi-Annual
PBM-0008	374	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9101C	561	PBG			1	1	Monitoring Well	Semi-Annual
SWN-9103B	571	PBG			1	1	Monitoring Well	Semi-Annual
SWN-9103C	572	PBG			1	1	Monitoring Well	Semi-Annual
SWN-9103D	573	PBG			1	1	Monitoring Well	Semi-Annual
SWN-9103E	574	PBG			1	1	Monitoring Well	Semi-Annual
SWN-9104C	575	PBG			1	1	Monitoring Well	Semi-Annual
SWN-9104D	576	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1001C	595	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8202A	613	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8202B	614	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8202C	615	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8205A	622	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8205B	623	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8205C	624	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8502A	632	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8503A	633	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8902C	645	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8903B	646	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8903C	647	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8912A	654	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8912B	655	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9112C	665	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9112D	666	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9301B	668	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9301C	669	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9303B	673	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9303C	674	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9303D	675	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9304D	687	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9902D	691	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9903A	692	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9903B	693	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9903C	694	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9903D	695	PBG			1	1	Monitoring Well	Semi-Annual
S1147	709	PBG			1	1	Monitoring Well	Semi-Annual
S1148	710	PBG			1	1	Monitoring Well	Semi-Annual
SPN-8903B	718	PBG			1	1	Monitoring Well	Semi-Annual
SPN-8903C	719	PBG			1	1	Monitoring Well	Semi-Annual
SPN-8904B	720	PBG			1	1	Monitoring Well	Semi-Annual
SPN-8904C	721	PBG			1	1	Monitoring Well	Semi-Annual
SPN-9103D	725	PBG			1	1	Monitoring Well	Semi-Annual
SPN-9104D	726	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1302A	770	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1302B	771	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1302C	772	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1302D	773	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1303A	774	PBG			- 1	- 1	Monitoring Well	Semi-Annual
PBN-1303B	775	PBG			-	-	Monitoring Well	Semi-Annual
PBN-1303C	776	PBG			- 1	- 1	Monitoring Well	Semi-Annual
PBN-1303D	777	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1304A	778	PBG			1	1	Monitoring Well	Semi-Annual
							-	

# April 2019 Groundwater Sampling Schedule

Well Name	<u>Well ID</u>	<u>Plume Area</u>	<u>Nitrates</u>	<u>Sulfate</u>	<u>VOC</u>	<u>DNT</u>	<u>Well Type</u>	Sample Frequency
PBN-1304B	779	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1304C	780	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1304D	781	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1401A	782	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1401B	783	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1401C	784	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1404B	791	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1404C	792	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1404D	793	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8902BR	795	PBG			1	1	Monitoring Well	Semi-Annual
PBM-9001D	981	PBG			1	1	Monitoring Well	Semi-Annual
Totals			3	16	113	117		

#### June 2019 Groundwater Sampling Schedule

<u>Well Name</u>	Well ID	<u>Plume Area</u>	VOC	DNT	Well Type	Sample Frequency
Purcell-D	163	DBG	1	1	Residential Well	Quarterly
ELN-1003A	467	DBG		1	Monitoring Well	Quarterly
ELN-1003B	468	DBG		1	Monitoring Well	Quarterly
ELN-1003C	469	DBG		1	Monitoring Well	Quarterly
ELN-1003E	470	DBG		1	Monitoring Well	Quarterly
ELN-1504B	537	DBG		1	Monitoring Well	Quarterly
WE-ZE512	437	Central	1	1	<b>Residential Well</b>	Quarterly
SEN-0501A	580	Central	1	1	Monitoring Well	Semi-Annual
SEN-0501B	581	Central	1	1	Monitoring Well	Semi-Annual
SEN-0501D	582	Central	1	1	Monitoring Well	Semi-Annual
SEN-0502A	583	Central	1	1	Monitoring Well	Semi-Annual
SEN-0502D	584	Central	1	1	Monitoring Well	Semi-Annual
SEN-0503A	585	Central	1	1	Monitoring Well	Semi-Annual
SEN-0503B	586	Central	1	1	Monitoring Well	Semi-Annual
SEN-0503D	587	Central	1	1	Monitoring Well	Semi-Annual
NLN-1001A	331	Central		1	Monitoring Well	Annual
NLN-1001C	332	Central		1	Monitoring Well	Annual
NLN-8203A	258	Central		1	Monitoring Well	Annual
NLN-8203B	259	Central		1	Monitoring Well	Annual
NLN-8203C	260	Central		1	Monitoring Well	Annual
NPM-8901	506	Central		1	Monitoring Well	Annual
RIM-1003	491	Central		1	Monitoring Well	Annual
RIM-1004	494	Central		1	Monitoring Well	Annual
RIN-0701C	443	Central		1	Monitoring Well	Annual
RIN-0702C	444	Central		1	Monitoring Well	Annual
RIN-0703C	445	Central		1	Monitoring Well	Annual
RIN-1002A	492	Central		1	Monitoring Well	Annual
RIN-1002C	493	Central		1	Monitoring Well	Annual
RIN-1003A	495	Central		1	Monitoring Well	Annual
RIN-1004B	498	Central		1	Monitoring Well	Semi-Annual
RIN-1005A	496	Central		1	Monitoring Well	Annual
RIN-1005C	497	Central		1	Monitoring Well	Annual
RPM-8901	507	Central		1	Monitoring Well	Annual
S1111	751	Central		1	Monitoring Well	Annual
RIN-1501B	538	Central		1	Monitoring Well	Annual
RIN-1501C	539	Central		1	Monitoring Well	Annual
RIN-1501D	540	Central		1	Monitoring Well	Annual
RIN-1502B	541	Central		1	Monitoring Well	Annual
RIN-1502C	542	Central		1	Monitoring Well	Annual
RIN-1502D	543	Central		1	Monitoring Well	Annual

10

40

# August 2019 Groundwater Sampling Schedule

Well Name	Well ID	Plume Area	VOC	DNT	Well Type	Sample Frequency
Purcell-D	163	DBG	1	1	Residential Well	Quarterly
Anderson-R	411	DBG	1	1	<b>Residential Well</b>	Annual
Curto	412	DBG	1	1	Residential Well	Annual
Wenger	414	DBG	1	1	Residential Well	Annual
Grosse	415	DBG	1	1	Residential Well	Annual
Gruber-D	417	DBG	1	1	Residential Well	Annual
Hendershot	418	DBG	1	1	Residential Well	Annual
Howery	419	DBG	1	1	Residential Well	Annual
Österland	422	DBG	1	1	Residential Well	Annual
Melum	423	DBG	1	1	Residential Well	Annual
Raschein	424	DBG	1	1	Residential Well	Annual
Revers	425	DBG	1	1	Residential Well	Annual
Roll	426	DBG	1	1	Residential Well	Annual
Reif	427	DBG	1	1	Residential Well	Annual
Schumann	428	DBG	1	1	Residential Well	Annual
Spear	803	DBG	1	1	Residential Well	Annual
Brev	817	DBG	1	1	Residential Well	Annual
Gibbs	839	DBG	1	1	Residential Well	Annual
Groth	842	DBG	1	1	Residential Well	Annual
Lukens	860	DBG	1	1	Residential Well	Annual
Kopras	874	DBG	1	1	Residential Well	Annual
Nowotarski	891	DBG	1	1	Residential Well	Annual
Olah	904	DBG	1	1	Residential Well	Annual
Purcell-G	916	DBG	1	1	Residential Well	Annual
Zurbachen-A	967	DBG	1	1	Residential Well	Annual
USDA 3	126	Central		1	Residential Well	Annual
USDA 6	128	Central		1	Residential Well	Annual
USDA 1	828	Central		1	Residential Well	Annual
USDA 2	829	Central		1	Residential Well	Annual
WE-TM599	129	Central		1	Residential Well	Annual
WE-RM383	153	Central		1	Residential Well	Annual
WE-RR542	156	Central		1	Residential Well	Annual
WE-QR441	157	Central	1	1	Residential Well	Annual
WE-QN039	158	Central	1	1	Residential Well	Annual
WE-RD430	159	Central		1	Residential Well	Annual
WE-SQ017	164	Central	1	1	Residential Well	Annual
WE-SQ001	165	Central	1	1	Residential Well	Annual
WE-RR598	169	Central		1	Residential Well	Annual
WE-SQ002	170	Central		1	Residential Well	Annual
WE-TF023	174	Central		1	Residential Well	Annual
WE-UK125	431	Central		1	Residential Well	Annual
WE-7E512	437	Central	1	1	Residential Well	Quarterly
WE-UA297	433	Central	-	-	Residential Well	Annual
WE-XD828	434	Central		1	Residential Well	Annual
WE-XK342	435	Central		-	Residential Well	Annual
WE-YW972	436	Central	1	-	Residential Well	Annual
Delanev	152	PBG	-	- 1	Residential Well	Annual
Mittenzwei	800	PBG	1	1	Residential Well	Annual
Judd	862	PBG	-	- 1	Residential Well	Annual
Krumenauer	875	PBG	1	1	Residential Well	Annual

## August 2019 Groundwater Sampling Schedule

Well Name	Well ID	Plume Area	VOC	DNT	Well Type	Sample Frequency
PDS-3	911	PBG	1	1	Residential Well	Annual
Ramaker-J	917	PBG	1	1	<b>Residential Well</b>	Annual
Schlender	931	PBG	1	1	<b>Residential Well</b>	Annual
Apel	998	PBG	1	1	Residential Well	Annual
Totals			39	54		

## September 2019 Groundwater Sampling Schedule

<u>Well Name</u>	<u>Well ID</u>	<u>Plume Area</u>	<u>Nitrates</u>	<u>VOC</u>	DNT	<u>Well Type</u>	Sample Frequency
ELN-8203A	210	DBG			1	Monitoring Well	Semi-Annual
ELN-8203B	211	DBG			1	Monitoring Well	Semi-Annual
ELN-8203C	212	DBG			1	Monitoring Well	Semi-Annual
ELM-8901	216	DBG			1	Monitoring Well	Semi-Annual
ELM-8907	220	DBG			1	Monitoring Well	Semi-Annual
ELM-8908	221	DBG			1	Monitoring Well	Semi-Annual
ELM-8909	222	DBG			1	Monitoring Well	Semi-Annual
ELN-8902B	224	DBG			1	Monitoring Well	Semi-Annual
ELN-9107A	227	DBG			1	Monitoring Well	Semi-Annual
ELN-9107B	228	DBG			1	Monitoring Well	Semi-Annual
ELN-9402AR	231	DBG			1	Monitoring Well	Semi-Annual
ELM-9501	234	DBG			1	Monitoring Well	Semi-Annual
S1134R	236	DBG			1	Monitoring Well	Semi-Annual
DBM-8201	301	DBG			1	Monitoring Well	Semi-Annual
DBM-8202	302	DBG			1	Monitoring Well	Semi-Annual
DBM-8903	306	DBG			1	Monitoring Well	Semi-Annual
DBN-9501A	314	DBG			1	Monitoring Well	Semi-Annual
DBN-9501B	315	DBG			1	Monitoring Well	Semi-Annual
DBN-9501C	316	DBG			1	Monitoring Well	Semi-Annual
DBN-9501E	317	DBG			1	Monitoring Well	Semi-Annual
ELN-0801B	455	DBG			1	Monitoring Well	Semi-Annual
ELN-0801C	456	DBG			1	Monitoring Well	Semi-Annual
ELN-0801E	457	DBG			1	Monitoring Well	Semi-Annual
ELN-1001B	460	DBG			1	Monitoring Well	Semi-Annual
ELN-1001C	461	DBG			1	Monitoring Well	Semi-Annual
ELN-1001E	462	DBG			1	Monitoring Well	Semi-Annual
ELN-1002A	463	DBG			1	Monitoring Well	Semi-Annual
ELN-1002B	464	DBG			1	Monitoring Well	Semi-Annual
ELN-1002C	465	DBG			1	Monitoring Well	Semi-Annual
ELN-1002E	466	DBG			1	Monitoring Well	Semi-Annual
ELN-1003A	467	DBG			1	Monitoring Well	Quarterly
ELN-1003B	468	DBG			1	Monitoring Well	Quarterly
ELN-1003C	469	DBG			1	Monitoring Well	Quarterly
ELN-1003E	470	DBG			1	Monitoring Well	Quarterly
DBN-1001B	472	DBG			1	Monitoring Well	Semi-Annual
DBN-1001C	473	DBG			1	Monitoring Well	Semi-Annual
DBN-1001E	474	DBG			1	Monitoring Well	Semi-Annual
DBN-1002C	476	DBG			1	Monitoring Well	Semi-Annual
DBN-1002E	477	DBG			1	Monitoring Well	Semi-Annual
S1121	755	DBG			1	Monitoring Well	Semi-Annual
ELN-1502A	533	DBG			1	Monitoring Well	Semi-Annual
ELN-1502C	534	DBG			1	Monitoring Well	Semi-Annual
ELN-1503A	535	DBG			1	Monitoring Well	Semi-Annual
ELN-1503C	536	DBG			1	Monitoring Well	Semi-Annual
ELN-1504B	537	DBG			1	Monitoring Well	Quarterly
RIM-0703	440	NC Area			1	Monitoring Well	Annual
RIM-0705	442	NC Area			1	Monitoring Well	Semi-Annual
RIM-1002	478	NC Area			1	Monitoring Well	Semi-Annual
RIN-1007C	479	NC Area			1	Monitoring Well	Annual
RIN-1001A	480	NC Area			1	Monitoring Well	Semi-Annual
RIN-1001C	481	NC Area			1	Monitoring Well	Annual

## September 2019 Groundwater Sampling Schedule

<u>Well Name</u>	<u>Well ID</u>	<u>Plume Area</u>	<u>Nitrates</u>	VOC	<u>DNT</u>	Well Type	Sample Frequency
S1125	504	NC Area			1	Monitoring Well	Semi-Annual
PBM-9801	360	PBG		1	1	Monitoring Well	Annual
PBM-0001	367	PBG	1	1	1	Monitoring Well	Semi-Annual
PBM-0002	368	PBG	1	1	1	Monitoring Well	Semi-Annual
PBM-0006	372	PBG	1	1	1	Monitoring Well	Semi-Annual
PBM-0008	374	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9101C	561	PBG		1	1	Monitoring Well	Semi-Annual
SWN-9102C	569	PBG		1	1	Monitoring Well	Annual
SWN-9102D	570	PBG		1	1	Monitoring Well	Annual
SWN-9103B	571	PBG		1	1	Monitoring Well	Semi-Annual
SWN-9103C	572	PBG		1	1	Monitoring Well	Semi-Annual
SWN-9103D	573	PBG		1	1	Monitoring Well	Semi-Annual
SWN-9103E	574	PBG		1	1	Monitoring Well	Semi-Annual
SWN-9104C	575	PBG		1	1	Monitoring Well	Semi-Annual
SWN-9104D	576	PBG		1	1	Monitoring Well	Semi-Annual
SWN-9105B	577	PBG		1	1	Monitoring Well	Annual
SWN-9105C	578	PBG		1	1	Monitoring Well	Annual
SWN-9105D	579	PBG		1	1	Monitoring Well	Annual
PBN-1003C	592	PBG		1	1	Monitoring Well	Annual
PBN-1001C	595	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8202A	613	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8202B	614	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8202C	615	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8205A	622	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8205B	623	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8205C	624	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8502A	632	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8503A	633	PBG		1	1	Monitoring Well	Semi-Annual
PBM-8907	637	PBG		1	1	Monitoring Well	Annual
PBN-8902C	645	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8903B	646	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8903C	647	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8912A	654	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8912B	655	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9112C	665	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9112D	666	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9301B	668	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9301C	669	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9303B	673	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9303C	674	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9303D	675	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9304D	687	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9902D	691	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9903A	692	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9903B	693	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9903C	694	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9903D	695	PBG		1	1	Monitoring Well	Semi-Annual
S1147	709	PBG		1	1	Monitoring Well	Semi-Annual
S1148	710	PBG		1	1	Monitoring Well	Semi-Annual
SPN-8903B	718	PBG		1	1	Monitoring Well	Semi-Annual
SPN-8903C	719	PBG		1	1	Monitoring Well	Semi-Annual

## September 2019 Groundwater Sampling Schedule

Well Name	<u>Well ID</u>	<u>Plume Area</u>	<u>Nitrates</u>	<u>VOC</u>	DNT	<u>Well Type</u>	Sample Frequency
SPN-8904B	720	PBG		1	1	Monitoring Well	Semi-Annual
SPN-8904C	721	PBG		1	1	Monitoring Well	Semi-Annual
SPN-9103D	725	PBG		1	1	Monitoring Well	Semi-Annual
SPN-9104D	726	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1302A	770	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1302B	771	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1302C	772	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1302D	773	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1303A	774	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1303B	775	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1303C	776	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1303D	777	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1304A	778	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1304B	779	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1304C	780	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1304D	781	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1401A	782	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1401B	783	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1401C	784	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1404B	791	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1404C	792	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1404D	793	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8902BR	795	PBG		1	1	Monitoring Well	Semi-Annual
PBM-9001D	981	PBG		1	1	Monitoring Well	Semi-Annual
Totals			3	74	126		

3

74

Page 9 of 20

## November 2019 Groundwater Sampling Schedule

<u>Well Name</u>	Well ID	<u>Plume Area</u>	VOC	DNT	<u>Well Type</u>	Sample Frequency
SEN-0501A	580	Central	1	1	Monitoring Well	Semi-Annual
SEN-0501B	581	Central	1	1	Monitoring Well	Semi-Annual
SEN-0501D	582	Central	1	1	Monitoring Well	Semi-Annual
SEN-0502A	583	Central	1	1	Monitoring Well	Semi-Annual
SEN-0502D	584	Central	1	1	Monitoring Well	Semi-Annual
SEN-0503A	585	Central	1	1	Monitoring Well	Semi-Annual
SEN-0503B	586	Central	1	1	Monitoring Well	Semi-Annual
SEN-0503D	587	Central	1	1	Monitoring Well	Semi-Annual
RIN-1004B	498	Central		1	Monitoring Well	Semi-Annual
WE-ZE512	437	Central	1	1	<b>Residential Well</b>	Quarterly
Purcell-D	163	DBG	1	1	<b>Residential Well</b>	Quarterly
ELN-1003A	467	DBG		1	Monitoring Well	Quarterly
ELN-1003B	468	DBG		1	Monitoring Well	Quarterly
ELN-1003C	469	DBG		1	Monitoring Well	Quarterly
ELN-1003E	470	DBG		1	Monitoring Well	Quarterly
ELN-1504B	537	DBG		1	Monitoring Well	Quarterly

Totals

10

16

# April 2020 Groundwater Sampling Schedule

Well Name	Well ID	<u>Plume Area</u>	<u>Nitrates</u>	<u>Sulfate</u>	VOC	DNT	<u>Well Type</u>	Sample Frequency
WE-ZE512	437	Central			1	1	<b>Residential Well</b>	Quarterly
Purcell-D	163	DBG			1	1	Residential Well	Quarterly
ELN-8203A	210	DBG		1	1	1	Monitoring Well	Semi-Annual
ELN-8203B	211	DBG		1	1	1	Monitoring Well	Semi-Annual
ELN-8203C	212	DBG		1	1	1	Monitoring Well	Semi-Annual
ELM-8901	216	DBG		1	1	1	Monitoring Well	Semi-Annual
ELM-8907	220	DBG		1	1	1	Monitoring Well	Semi-Annual
ELM-8908	221	DBG		1	1	1	Monitoring Well	Semi-Annual
ELM-8909	222	DBG		1	1	1	Monitoring Well	Semi-Annual
ELN-8902B	224	DBG		1	1	1	Monitoring Well	Semi-Annual
ELN-9107A	227	DBG		1	1	1	Monitoring Well	Semi-Annual
ELN-9107B	228	DBG		1	1	1	Monitoring Well	Semi-Annual
ELN-9402AR	231	DBG		1	1	1	Monitoring Well	Semi-Annual
ELM-9501	234	DBG			1	1	Monitoring Well	Semi-Annual
S1134R	236	DBG		1	1	1	Monitoring Well	Semi-Annual
DBM-8201	301	DBG		1	1	1	Monitoring Well	Semi-Annual
DBM-8202	302	DBG		1	-	1	Monitoring Well	Semi-Annual
DBM-8903	306	DBG		-	1	1	Monitoring Well	Semi-Annual
DBN-9501A	314	DBG			1	1	Monitoring Well	Semi-Annual
DBN-9501R	315	DBG			1	-	Monitoring Well	Semi-Annual
DBN-9501C	316	DBG			1	1	Monitoring Well	Semi-Annual
DBN-9501E	317	DBG			1	1	Monitoring Well	Semi-Annual
FLN-0801B	455	DBG			1	1	Monitoring Well	Semi-Annual
ELN-0801C	456	DBG			1	1	Monitoring Well	Semi-Annual
ELN-0801C	457	DBG			1	1	Monitoring Well	Semi-Annual
	457	DBG			1	1	Monitoring Well	Semi-Annual
	400	DBG			1	1	Monitoring Well	Semi-Annual
	401	DBG			1	1	Monitoring Well	Somi Annual
	402	DBG			1	1	Monitoring Well	Semi-Annual
	403	DBG			1	1	Monitoring Well	Semi-Annual
	404				1	1	Monitoring Well	Semi-Annual
ELN-1002C	405				1	1	Monitoring Well	Semi-Annual
	400				1	1	Monitoring Well	Quartarly
ELN-1003A	407	DBG			1	1	Monitoring Well	Quarterly
ELN-1003B	468	DBG			1	1	Monitoring Well	Quarterly
ELN-1003C	469	DBG			1	1	Monitoring Well	Quarterly
ELN-1003E	470	DBG			1	1	Monitoring Well	Quarteriy
DBN-10016	472	DBG			1	1	Monitoring Well	Semi-Annual
DBN-1001C	473	DBG			1	1	Monitoring Well	Semi-Annual
DBN-1001E	474	DBG		4	1	1	Monitoring Well	Semi-Annual
DBN-1002C	476	DBG		1	1	1	Monitoring Well	Semi-Annual
DBN-1002E	4//	DBG		T	1	T	Monitoring Well	Semi-Annual
51121	/55	DBG			1	1	Monitoring Well	Semi-Annual
ELN-1502A	533	DBG			1	1	Monitoring Well	Semi-Annual
ELN-1502C	534	DBG			1	1	Monitoring Well	Semi-Annual
ELN-1503A	535	DBG			1	1	Monitoring Well	Semi-Annual
ELN-1503C	536	DBG			1	1	Monitoring Well	Semi-Annual
ELN-1504B	537	DBG			1	1	Monitoring Well	Quarterly
RIM-0705	442	NC Area				1	Monitoring Well	Semi-Annual
RIM-1002	478	NC Area				1	Monitoring Well	Semi-Annual
RIN-1001A	480	NC Area				1	Monitoring Well	Semi-Annual
S1125	504	NC Area				1	Monitoring Well	Semi-Annual
PBM-0001	367	PBG	1		1	1	Monitoring Well	Semi-Annual
PBM-0002	368	PBG	1		1	1	Monitoring Well	Semi-Annual
PBM-0006	372	PBG	1		1	1	Monitoring Well	Semi-Annual

# April 2020 Groundwater Sampling Schedule

<u>Well Name</u>	<u>Well ID</u>	<u>Plume Area</u>	<u>Nitrates</u>	<u>Sulfate</u>	<u>VOC</u>	DNT	<u>Well Type</u>	Sample Frequency
PBM-0008	374	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9101C	561	PBG			1	1	Monitoring Well	Semi-Annual
SWN-9103B	571	PBG			1	1	Monitoring Well	Semi-Annual
SWN-9103C	572	PBG			1	1	Monitoring Well	Semi-Annual
SWN-9103D	573	PBG			1	1	Monitoring Well	Semi-Annual
SWN-9103E	574	PBG			1	1	Monitoring Well	Semi-Annual
SWN-9104C	575	PBG			1	1	Monitoring Well	Semi-Annual
SWN-9104D	576	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1001C	595	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8202A	613	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8202B	614	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8202C	615	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8205A	622	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8205B	623	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8205C	624	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8502A	632	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8503A	633	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8902C	645	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8903B	646	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8903C	647	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8912A	654	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8912B	655	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9112C	665	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9112D	666	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9301B	668	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9301C	669	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9303B	673	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9303C	674	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9303D	675	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9304D	687	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9902D	691	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9903A	692	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9903B	693	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9903C	694	PBG			1	1	Monitoring Well	Semi-Annual
PBN-9903D	695	PBG			1	1	Monitoring Well	Semi-Annual
S1147	709	PBG			1	1	Monitoring Well	Semi-Annual
S1148	710	PBG			1	1	Monitoring Well	Semi-Annual
SPN-8903B	718	PBG			1	1	Monitoring Well	Semi-Annual
SPN-8903C	719	PBG			1	1	Monitoring Well	Semi-Annual
SPN-8904B	720	PBG			1	1	Monitoring Well	Semi-Annual
SPN-8904C	721	PBG			1	1	Monitoring Well	Semi-Annual
SPN-9103D	725	PBG			1	1	Monitoring Well	Semi-Annual
SPN-9104D	726	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1302A	770	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1302B	771	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1302C	772	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1302D	773	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1303A	774	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1303B	775	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1303C	776	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1303D	777	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1304A	778	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1304B	779	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1304C	780	PBG			1	1	Monitoring Well	Semi-Annual

# April 2020 Groundwater Sampling Schedule

Well Name	Well ID	<u>Plume Area</u>	<u>Nitrates</u>	<u>Sulfate</u>	<u>VOC</u>	DNT	Well Type	Sample Frequency
PBN-1304D	781	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1401A	782	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1401B	783	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1401C	784	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1404B	791	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1404C	792	PBG			1	1	Monitoring Well	Semi-Annual
PBN-1404D	793	PBG			1	1	Monitoring Well	Semi-Annual
PBN-8902BR	795	PBG			1	1	Monitoring Well	Semi-Annual
PBM-9001D	981	PBG			1	1	Monitoring Well	Semi-Annual
Totals			3	16	113	117		

#### June 2020 Groundwater Sampling Schedule

<u>Well Name</u>	Well ID	<u>Plume Area</u>	VOC	DNT	Well Type	Sample Frequency
Purcell-D	163	DBG	1	1	Residential Well	Quarterly
ELN-1003A	467	DBG		1	Monitoring Well	Quarterly
ELN-1003B	468	DBG		1	Monitoring Well	Quarterly
ELN-1003C	469	DBG		1	Monitoring Well	Quarterly
ELN-1003E	470	DBG		1	Monitoring Well	Quarterly
ELN-1504B	537	DBG		1	Monitoring Well	Quarterly
WE-ZE512	437	Central	1	1	Residential Well	Quarterly
SEN-0501A	580	Central	1	1	Monitoring Well	Semi-Annual
SEN-0501B	581	Central	1	1	Monitoring Well	Semi-Annual
SEN-0501D	582	Central	1	1	Monitoring Well	Semi-Annual
SEN-0502A	583	Central	1	1	Monitoring Well	Semi-Annual
SEN-0502D	584	Central	1	1	Monitoring Well	Semi-Annual
SEN-0503A	585	Central	1	1	Monitoring Well	Semi-Annual
SEN-0503B	586	Central	1	1	Monitoring Well	Semi-Annual
SEN-0503D	587	Central	1	1	Monitoring Well	Semi-Annual
NLN-1001A	331	Central		1	Monitoring Well	Annual
NLN-1001C	332	Central		1	Monitoring Well	Annual
NLN-8203A	258	Central		1	Monitoring Well	Annual
NLN-8203B	259	Central		1	Monitoring Well	Annual
NLN-8203C	260	Central		1	Monitoring Well	Annual
NPM-8901	506	Central		1	Monitoring Well	Annual
RIM-1003	491	Central		1	Monitoring Well	Annual
RIM-1004	494	Central		1	Monitoring Well	Annual
RIN-0701C	443	Central		1	Monitoring Well	Annual
RIN-0702C	444	Central		1	Monitoring Well	Annual
RIN-0703C	445	Central		1	Monitoring Well	Annual
RIN-1002A	492	Central		1	Monitoring Well	Annual
RIN-1002C	493	Central		1	Monitoring Well	Annual
RIN-1003A	495	Central		1	Monitoring Well	Annual
RIN-1004B	498	Central		1	Monitoring Well	Semi-Annual
RIN-1005A	496	Central		1	Monitoring Well	Annual
RIN-1005C	497	Central		1	Monitoring Well	Annual
RPM-8901	507	Central		1	Monitoring Well	Annual
S1111	751	Central		1	Monitoring Well	Annual
RIN-1501B	538	Central		1	Monitoring Well	Annual
RIN-1501C	539	Central		1	Monitoring Well	Annual
RIN-1501D	540	Central		1	Monitoring Well	Annual
RIN-1502B	541	Central		1	Monitoring Well	Annual
RIN-1502C	542	Central		1	Monitoring Well	Annual
RIN-1502D	543	Central		1	Monitoring Well	Annual

10

40

## August 2020 Groundwater Sampling Schedule

Well Name	Well ID	Plume Area	VOC	DNT	Well Type	Sample Frequency
Purcell-D	163	DBG	1	1	Residential Well	Quarterly
Anderson-R	411	DBG	1	1	<b>Residential Well</b>	Annual
Curto	412	DBG	1	1	<b>Residential Well</b>	Annual
Wenger	414	DBG	1	1	<b>Residential Well</b>	Annual
Grosse	415	DBG	1	1	<b>Residential Well</b>	Annual
Gruber-D	417	DBG	1	1	<b>Residential Well</b>	Annual
Hendershot	418	DBG	1	1	<b>Residential Well</b>	Annual
Howery	419	DBG	1	1	<b>Residential Well</b>	Annual
Osterland	422	DBG	1	1	<b>Residential Well</b>	Annual
Melum	423	DBG	1	1	<b>Residential Well</b>	Annual
Raschein	424	DBG	1	1	<b>Residential Well</b>	Annual
Revers	425	DBG	1	1	<b>Residential Well</b>	Annual
Roll	426	DBG	1	1	<b>Residential Well</b>	Annual
Reif	427	DBG	1	1	<b>Residential Well</b>	Annual
Schumann	428	DBG	1	1	<b>Residential Well</b>	Annual
Spear	803	DBG	1	1	<b>Residential Well</b>	Annual
Brey	817	DBG	1	1	<b>Residential Well</b>	Annual
Gibbs	839	DBG	1	1	<b>Residential Well</b>	Annual
Groth	842	DBG	1	1	<b>Residential Well</b>	Annual
Lukens	860	DBG	1	1	<b>Residential Well</b>	Annual
Kopras	874	DBG	1	1	<b>Residential Well</b>	Annual
Nowotarski	891	DBG	1	1	<b>Residential Well</b>	Annual
Olah	904	DBG	1	1	<b>Residential Well</b>	Annual
Purcell-G	916	DBG	1	1	<b>Residential Well</b>	Annual
Zurbachen-A	967	DBG	1	1	<b>Residential Well</b>	Annual
USDA 3	126	Central		1	<b>Residential Well</b>	Annual
USDA 6	128	Central		1	<b>Residential Well</b>	Annual
USDA 1	828	Central		1	<b>Residential Well</b>	Annual
USDA 2	829	Central		1	<b>Residential Well</b>	Annual
WE-TM599	129	Central		1	<b>Residential Well</b>	Annual
WE-RM383	153	Central		1	<b>Residential Well</b>	Annual
WE-RR542	156	Central		1	<b>Residential Well</b>	Annual
WE-QR441	157	Central	1	1	<b>Residential Well</b>	Annual
WE-QN039	158	Central	1	1	<b>Residential Well</b>	Annual
WE-RD430	159	Central		1	<b>Residential Well</b>	Annual
WE-SQ017	164	Central	1	1	<b>Residential Well</b>	Annual
WE-SQ001	165	Central	1	1	<b>Residential Well</b>	Annual
WE-RR598	169	Central		1	<b>Residential Well</b>	Annual
WE-SQ002	170	Central		1	Residential Well	Annual
WE-TF023	174	Central		1	<b>Residential Well</b>	Annual
WE-UK125	431	Central		1	<b>Residential Well</b>	Annual
WE-ZE512	437	Central	1	1	Residential Well	Quarterly
WE-UA297	433	Central		1	<b>Residential Well</b>	Annual
WE-XD828	434	Central		1	Residential Well	Annual
WE-XK342	435	Central		1	Residential Well	Annual
WE-YW972	436	Central	1	1	Residential Well	Annual
Delaney	152	PBG	1	1	Residential Well	Annual
Mittenzwei	800	PBG	1	1	Residential Well	Annual
Judd	862	PBG	1	1	Residential Well	Annual
Krumenauer	875	PBG	1	1	<b>Residential Well</b>	Annual

## August 2020 Groundwater Sampling Schedule

Well Name	Well ID	Plume Area	VOC	DNT	Well Type	Sample Frequency
PDS-3	911	PBG	1	1	Residential Well	Annual
Ramaker-J	917	PBG	1	1	<b>Residential Well</b>	Annual
Schlender	931	PBG	1	1	<b>Residential Well</b>	Annual
Apel	998	PBG	1	1	Residential Well	Annual
Totals			39	54		

# September 2020 Groundwater Sampling Schedule

ELN-8203A         210         DBG         1         Monitoring Weil         Semi-Annual           ELN-8203B         211         DBG         1         Monitoring Weil         Semi-Annual           ELN-8203C         212         DBG         1         Monitoring Weil         Semi-Annual           ELM-8901         216         DBG         1         Monitoring Weil         Semi-Annual           ELM-8908         221         DBG         1         Monitoring Weil         Semi-Annual           ELN-8907A         227         DBG         1         Monitoring Weil         Semi-Annual           ELN-9107A         227         DBG         1         Monitoring Weil         Semi-Annual           ELN-9107A         227         DBG         1         Monitoring Weil         Semi-Annual           ELN-9107B         228         DBG         1         Monitoring Weil         Semi-Annual           ELN-9107B         230         DBG         1         Monitoring Weil         Semi-Annual           ELN-9107B         230         DBG         1         Monitoring Weil         Semi-Annual           DBM-8201         301         DBG         1         Monitoring Weil         Semi-Annual <t< th=""><th>Well Name</th><th>Well ID</th><th><u>Plume Area</u></th><th><u>Nitrates</u></th><th>VOC</th><th>DNT</th><th>Well Type</th><th>Sample Frequency</th></t<>	Well Name	Well ID	<u>Plume Area</u>	<u>Nitrates</u>	VOC	DNT	Well Type	Sample Frequency
ELN-82028         211         DBG         1         Monitoring Well         Semi-Annual           ELN-8203C         212         DBG         1         Monitoring Well         Semi-Annual           ELM-8901         216         DBG         1         Monitoring Well         Semi-Annual           ELM-8902         221         DBG         1         Monitoring Well         Semi-Annual           ELM-8902         224         DBG         1         Monitoring Well         Semi-Annual           ELN-8107A         227         DBG         1         Monitoring Well         Semi-Annual           ELN-9107B         228         DBG         1         Monitoring Well         Semi-Annual           ELN-9107A         227         DBG         1         Monitoring Well         Semi-Annual           ELN-9107A         226         DBG         1         Monitoring Well         Semi-Annual           ELN-9107A         230         DBG         1         Monitoring Well         Semi-Annual           DBM-8201         301         DBG         1         Monitoring Well         Semi-Annual           DBM-8202         302         DBG         1         Monitoring Well         Semi-Annual	ELN-8203A	210	DBG			1	Monitoring Well	Semi-Annual
ELN-8203C         212         DBG         1         Monitoring Well         Semi-Annual           ELM-8907         220         DBG         1         Monitoring Well         Semi-Annual           ELM-8908         221         DBG         1         Monitoring Well         Semi-Annual           ELM-8008         221         DBG         1         Monitoring Well         Semi-Annual           ELN-8007         227         DBG         1         Monitoring Well         Semi-Annual           ELN-9107A         227         DBG         1         Monitoring Well         Semi-Annual           ELN-9107B         228         DBG         1         Monitoring Well         Semi-Annual           ELN-9107B         228         DBG         1         Monitoring Well         Semi-Annual           DBM-8201         301         DBG         1         Monitoring Well         Semi-Annual           DBM-8202         302         DBG         1         Monitoring Well         Semi-Annual           DBM-8501A         314         DBG         1         Monitoring Well         Semi-Annual           DBN-9501C         316         DBG         1         Monitoring Well         Semi-Annual           D	ELN-8203B	211	DBG			1	Monitoring Well	Semi-Annual
ELM-8901         216         DBG         1         Monitoring Well         Semi-Annual           ELM-8907         220         DBG         1         Monitoring Well         Semi-Annual           ELM-8908         221         DBG         1         Monitoring Well         Semi-Annual           ELM-8028         224         DBG         1         Monitoring Well         Semi-Annual           ELN-9107A         227         DBG         1         Monitoring Well         Semi-Annual           ELN-9107A         228         DBG         1         Monitoring Well         Semi-Annual           ELN-9107A         224         DBG         1         Monitoring Well         Semi-Annual           ELN-9402AR         231         DBG         1         Monitoring Well         Semi-Annual           DBM-8201         301         DBG         1         Monitoring Well         Semi-Annual           DBM-8201         301         DBG         1         Monitoring Well         Semi-Annual           DBM-9501A         314         DBG         1         Monitoring Well         Semi-Annual           DBN-9501C         315         DBG         1         Monitoring Well         Semi-Annual	ELN-8203C	212	DBG			1	Monitoring Well	Semi-Annual
ELM-8907         220         DBG         1         Monitoring Weil         Semi-Annual           ELM-8908         221         DBG         1         Monitoring Weil         Semi-Annual           ELM-8907         222         DBG         1         Monitoring Weil         Semi-Annual           ELM-8007         227         DBG         1         Monitoring Weil         Semi-Annual           ELM-9107B         228         DBG         1         Monitoring Weil         Semi-Annual           ELM-9501         234         DBG         1         Monitoring Weil         Semi-Annual           S1134A         236         DBG         1         Monitoring Weil         Semi-Annual           DBM-8201         301         DBG         1         Monitoring Weil         Semi-Annual           DBM-8202         302         DBG         1         Monitoring Weil         Semi-Annual           DBM-9501A         314         DBG         1         Monitoring Weil         Semi-Annual           DBM-9501B         315         DBG         1         Monitoring Weil         Semi-Annual           DBN-9501B         317         DBG         1         Monitoring Weil         Semi-Annual           DLN-0	ELM-8901	216	DBG			1	Monitoring Well	Semi-Annual
ELM-8908         221         DBG         1         Monitoring Well         Semi-Annual           ELM-89028         224         DBG         1         Monitoring Well         Semi-Annual           ELN-8107A         227         DBG         1         Monitoring Well         Semi-Annual           ELN-9107A         227         DBG         1         Monitoring Well         Semi-Annual           ELN-9107A         227         DBG         1         Monitoring Well         Semi-Annual           ELN-9402AR         231         DBG         1         Monitoring Well         Semi-Annual           Sti134A         236         DBG         1         Monitoring Well         Semi-Annual           DBM-8201         301         DBG         1         Monitoring Well         Semi-Annual           DBM-8303         306         DBG         1         Monitoring Well         Semi-Annual           DBN-95018         315         DBG         1         Monitoring Well         Semi-Annual           DBN-95016         317         DBG         1         Monitoring Well         Semi-Annual           DBN-95016         317         DBG         1         Monitoring Well         Semi-Annual <td< td=""><td>ELM-8907</td><td>220</td><td>DBG</td><td></td><td></td><td>1</td><td>Monitoring Well</td><td>Semi-Annual</td></td<>	ELM-8907	220	DBG			1	Monitoring Well	Semi-Annual
ELM-8909         222         DBG         1         Monitoring Well         Semi-Annual           ELN-9107A         227         DBG         1         Monitoring Well         Semi-Annual           ELN-9107B         228         DBG         1         Monitoring Well         Semi-Annual           ELN-9107B         228         DBG         1         Monitoring Well         Semi-Annual           ELM-9501         234         DBG         1         Monitoring Well         Semi-Annual           DBM-8201         301         DBG         1         Monitoring Well         Semi-Annual           DBM-8202         302         DBG         1         Monitoring Well         Semi-Annual           DBM-9501A         314         DBG         1         Monitoring Well         Semi-Annual           DBN-9501B         315         DBG         1         Monitoring Well         Semi-Annual           DBN-9501C         316         DBG         1         Monitoring Well         Semi-Annual           ELN-0801C         455         DBG         1         Monitoring Well         Semi-Annual           ELN-0801C         455         DBG         1         Monitoring Well         Semi-Annual <td< td=""><td>ELM-8908</td><td>221</td><td>DBG</td><td></td><td></td><td>1</td><td>Monitoring Well</td><td>Semi-Annual</td></td<>	ELM-8908	221	DBG			1	Monitoring Well	Semi-Annual
EUN-89028         224         DBG         1         Monitoring Well         Semi-Annual           EUN-9107A         227         DBG         1         Monitoring Well         Semi-Annual           EUN-9107A         228         DBG         1         Monitoring Well         Semi-Annual           EUN-9402AR         231         DBG         1         Monitoring Well         Semi-Annual           St134A         236         DBG         1         Monitoring Well         Semi-Annual           DBM-8201         301         DBG         1         Monitoring Well         Semi-Annual           DBM-8303         306         DBG         1         Monitoring Well         Semi-Annual           DBN-9501A         314         DBG         1         Monitoring Well         Semi-Annual           DBN-9501B         315         DBG         1         Monitoring Well         Semi-Annual           DBN-9501C         316         DBG         1         Monitoring Well         Semi-Annual           ELN-801E         457         DBG         1         Monitoring Well         Semi-Annual           ELN-801C         468         DBG         1         Monitoring Well         Semi-Annual           E	ELM-8909	222	DBG			1	Monitoring Well	Semi-Annual
EUN-9107A         227         DBG         1         Monitoring Well         Semi-Annual           EUN-9107B         228         DBG         1         Monitoring Well         Semi-Annual           EUN-9501         234         DBG         1         Monitoring Well         Semi-Annual           Stil34R         236         DBG         1         Monitoring Well         Semi-Annual           DBM-8301         301         DBG         1         Monitoring Well         Semi-Annual           DBM-8303         306         DBG         1         Monitoring Well         Semi-Annual           DBN-9501A         314         DBG         1         Monitoring Well         Semi-Annual           DBN-9501C         316         DBG         1         Monitoring Well         Semi-Annual           DBN-9501C         316         DBG         1         Monitoring Well         Semi-Annual           EUN-0801C         456         DBG         1         Monitoring Well         Semi-Annual           EUN-0801C         457         DBG         1         Monitoring Well         Semi-Annual           EUN-0801C         456         DBG         1         Monitoring Well         Semi-Annual	ELN-8902B	224	DBG			1	Monitoring Well	Semi-Annual
ELN-9107B         228         DBG         1         Monitoring Well         Semi-Annual           ELN-9402AR         231         DBG         1         Monitoring Well         Semi-Annual           S1134R         236         DBG         1         Monitoring Well         Semi-Annual           S1134R         236         DBG         1         Monitoring Well         Semi-Annual           DBM-8201         302         DBG         1         Monitoring Well         Semi-Annual           DBM-9501A         314         DBG         1         Monitoring Well         Semi-Annual           DBN-9501B         315         DBG         1         Monitoring Well         Semi-Annual           DBN-9501B         317         DBG         1         Monitoring Well         Semi-Annual           DBN-9501E         317         DBG         1         Monitoring Well         Semi-Annual           ELN-0801B         455         DBG         1         Monitoring Well         Semi-Annual           ELN-0801E         457         DBG         1         Monitoring Well         Semi-Annual           ELN-0802C         458         DBG         1         Monitoring Well         Semi-Annual           E	ELN-9107A	227	DBG			1	Monitoring Well	Semi-Annual
ELN-9402AR         231         DBG         1         Monitoring Well         Semi-Annual           ELM-9501         234         DBG         1         Monitoring Well         Semi-Annual           DBM-8202         301         DBG         1         Monitoring Well         Semi-Annual           DBM-8201         301         DBG         1         Monitoring Well         Semi-Annual           DBM-8202         302         DBG         1         Monitoring Well         Semi-Annual           DBM-8203         306         DBG         1         Monitoring Well         Semi-Annual           DBN-9501A         314         DBG         1         Monitoring Well         Semi-Annual           DBN-9501E         317         DBG         1         Monitoring Well         Semi-Annual           ELN-0801C         456         DBG         1         Monitoring Well         Semi-Annual           ELN-0802C         459         DBG         1         Monitoring Well         Semi-Annual           ELN-1001E         461         DBG         1         Monitoring Well         Semi-Annual           ELN-1002E         465         DBG         1         Monitoring Well         Semi-Annual <td< td=""><td>ELN-9107B</td><td>228</td><td>DBG</td><td></td><td></td><td>1</td><td>Monitoring Well</td><td>Semi-Annual</td></td<>	ELN-9107B	228	DBG			1	Monitoring Well	Semi-Annual
ELM-5901234DBG1Monitoring WeilSemi-AnnualS1134R236DBG1Monitoring WeilSemi-AnnualDBM-8201301DBG1Monitoring WeilSemi-AnnualDBM-8202302DBG1Monitoring WeilSemi-AnnualDBM-8203306DBG1Monitoring WeilSemi-AnnualDBN-9501A314DBG1Monitoring WeilSemi-AnnualDBN-9501B315DBG1Monitoring WeilSemi-AnnualDBN-9501E317DBG1Monitoring WeilSemi-AnnualDBN-9501E317DBG1Monitoring WeilSemi-AnnualELN-0801E455DBG1Monitoring WeilSemi-AnnualELN-0801E457DBG1Monitoring WeilSemi-AnnualELN-0801E457DBG1Monitoring WeilSemi-AnnualELN-1001E461DBG1Monitoring WeilSemi-AnnualELN-1001E462DBG1Monitoring WeilSemi-AnnualELN-1002E464DBG1Monitoring WeilSemi-AnnualELN-1002E465DBG1Monitoring WeilSemi-AnnualELN-1002E466DBG1Monitoring WeilSemi-AnnualELN-1002E466DBG1Monitoring WeilSemi-AnnualELN-1003E470DBG1Monitoring WeilSemi-AnnualELN-1003E470DBG	ELN-9402AR	231	DBG			1	Monitoring Well	Semi-Annual
S1134R         236         DBG         1         Monitoring Well         Semi-Annual           DBM-8201         301         DBG         1         Monitoring Well         Semi-Annual           DBM-8203         306         DBG         1         Monitoring Well         Semi-Annual           DBM-9301         316         DBG         1         Monitoring Well         Semi-Annual           DBN-9501B         315         DBG         1         Monitoring Well         Semi-Annual           DBN-9501C         316         DBG         1         Monitoring Well         Semi-Annual           DBN-9501C         316         DBG         1         Monitoring Well         Semi-Annual           ELN-0801B         455         DBG         1         Monitoring Well         Semi-Annual           ELN-0802A         458         DBG         1         1         Monitoring Well         Semi-Annual           ELN-0802A         458         DBG         1         1         Monitoring Well         Semi-Annual           ELN-0802A         450         DBG         1         Monitoring Well         Semi-Annual           ELN-0802A         450         DBG         1         Monitoring Well         Semi-Annual	ELM-9501	234	DBG			1	Monitoring Well	Semi-Annual
DBM-8201301DBG1Monitoring WellSemi-AnnualDBM-8303306DBG1Monitoring WellSemi-AnnualDBM-9501A314DBG1Monitoring WellSemi-AnnualDBN-9501B315DBG1Monitoring WellSemi-AnnualDBN-9501C316DBG1Monitoring WellSemi-AnnualDBN-9501E317DBG1Monitoring WellSemi-AnnualDBN-9501E317DBG1Monitoring WellSemi-AnnualELN-0801B455DBG1Monitoring WellSemi-AnnualELN-0801C456DBG1Monitoring WellSemi-AnnualELN-0802C459DBG11Monitoring WellSemi-AnnualELN-1001B460DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1001B460DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002A464DBG1Monitoring WellSemi-AnnualELN-1002A467DBG1Monitoring WellSemi-AnnualELN-1003A467<	S1134R	236	DBG			1	Monitoring Well	Semi-Annual
DBM-8202302DBG1Monitoring WellSemi-AnnualDBM-9501A314DBG1Monitoring WellSemi-AnnualDBN-9501B315DBG1Monitoring WellSemi-AnnualDBN-9501E316DBG1Monitoring WellSemi-AnnualDBN-9501E317DBG1Monitoring WellSemi-AnnualELN-0801B455DBG1Monitoring WellSemi-AnnualELN-0801C456DBG1Monitoring WellSemi-AnnualELN-0802A458DBG11Monitoring WellSemi-AnnualELN-0802A458DBG11Monitoring WellSemi-AnnualELN-1001E460DBG1Monitoring WellSemi-AnnualELN-1001C461DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1003A467DBG1Monitoring WellSemi-AnnualELN-1003E470DBG1Monitoring WellSemi-AnnualELN-1003E470DBG1Monitoring WellSemi-AnnualELN-1003E470DBG1Monitoring WellSemi-AnnualELN-1003E470DBG1Monitoring WellSemi-AnnualELN-1003E <t< td=""><td>DBM-8201</td><td>301</td><td>DBG</td><td></td><td></td><td>1</td><td>Monitoring Well</td><td>Semi-Annual</td></t<>	DBM-8201	301	DBG			1	Monitoring Well	Semi-Annual
DBM-9903306DBG1Monitoring WellSemi-AnnualDBN-9501A314DBG1Monitoring WellSemi-AnnualDBN-9501C316DBG1Monitoring WellSemi-AnnualDBN-9501E317DBG1Monitoring WellSemi-AnnualELN-0801B455DBG1Monitoring WellSemi-AnnualELN-0801C456DBG1Monitoring WellSemi-AnnualELN-0801C457DBG1Monitoring WellSemi-AnnualELN-0802C459DBG11Monitoring WellBiennialELN-1001B460DBG1Monitoring WellSemi-AnnualELN-1001C461DBG1Monitoring WellSemi-AnnualELN-1002C463DBG1Monitoring WellSemi-AnnualELN-1002C464DBG1Monitoring WellSemi-AnnualELN-1002C465DBG1Monitoring WellSemi-AnnualELN-1002C466DBG1Monitoring WellSemi-AnnualELN-1003A467DBG1Monitoring WellSemi-AnnualELN-1003B468DBG1Monitoring WellSemi-AnnualELN-1003E470DBG1Monitoring WellSemi-AnnualELN-1003E470DBG1Monitoring WellSemi-AnnualELN-1003E470DBG1Monitoring WellSemi-AnnualELN-1003E470 <td< td=""><td>DBM-8202</td><td>302</td><td>DBG</td><td></td><td></td><td>1</td><td>Monitoring Well</td><td>Semi-Annual</td></td<>	DBM-8202	302	DBG			1	Monitoring Well	Semi-Annual
DBN-9501A314DBG1Monitoring WellSemi-AnnualDBN-9501C315DBG1Monitoring WellSemi-AnnualDBN-9501E317DBG1Monitoring WellSemi-AnnualDEN-9501E317DBG1Monitoring WellSemi-AnnualELN-0801B455DBG1Monitoring WellSemi-AnnualELN-0801E457DBG1Monitoring WellSemi-AnnualELN-0802A458DBG11Monitoring WellBiennialELN-1001B460DBG1Monitoring WellSemi-AnnualELN-1001C461DBG1Monitoring WellSemi-AnnualELN-1002B462DBG1Monitoring WellSemi-AnnualELN-1002B463DBG1Monitoring WellSemi-AnnualELN-1002B464DBG1Monitoring WellSemi-AnnualELN-1002B464DBG1Monitoring WellSemi-AnnualELN-1002B466DBG1Monitoring WellSemi-AnnualELN-1002B466DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1003A467DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellSemi-AnnualELN-1003E470DBG1Monitoring WellSemi-AnnualELN-1003E470	DBM-8903	306	DBG			1	Monitoring Well	Semi-Annual
DBN-9501B315DBG1Monitoring WellSemi-AnnualDBN-9501C316DBG1Monitoring WellSemi-AnnualDBN-9501C317DBG1Monitoring WellSemi-AnnualELN-0801B455DBG1Monitoring WellSemi-AnnualELN-0801C456DBG1Monitoring WellSemi-AnnualELN-0802C457DBG1Monitoring WellBiennialELN-0802C459DBG11Monitoring WellBiennialELN-1001B460DBG1Monitoring WellSemi-AnnualELN-1001C461DBG1Monitoring WellSemi-AnnualELN-1001E462DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1003C467DBG1Monitoring WellSemi-AnnualELN-1003E468DBG1Monitoring WellQuarterlyELN-1003E469DBG1Monitoring WellGuarterlyELN-1003E469DBG1Monitoring WellGuarterlyELN-1003E469DBG1Monitoring WellGuarterlyELN-1003E470DBG1Monitoring WellSemi-AnnualDBN-1001E474DBG<	DBN-9501A	314	DBG			1	Monitoring Well	Semi-Annual
DBN-9501C316DBG1Monitoring WellSemi-AnnualDBN-9501E317DBG1Monitoring WellSemi-AnnualELN-0801C455DBG1Monitoring WellSemi-AnnualELN-0801E457DBG1Monitoring WellSemi-AnnualELN-0801E457DBG1Monitoring WellBiennialELN-0802C459DBG11Monitoring WellBiennialELN-0101B460DBG1Monitoring WellSemi-AnnualELN-1001E461DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002B464DBG1Monitoring WellQuarterlyELN-1002C465DBG1Monitoring WellQuarterlyELN-1003C469DBG1Monitoring WellQuarterlyELN-1003C469DBG1Monitoring WellSemi-AnnualELN-1003C469DBG1Monitoring WellSemi-AnnualELN-1003C470DBG <td>DBN-9501B</td> <td>315</td> <td>DBG</td> <td></td> <td></td> <td>1</td> <td>Monitoring Well</td> <td>Semi-Annual</td>	DBN-9501B	315	DBG			1	Monitoring Well	Semi-Annual
DBN-9501E317DBG1Monitoring WellSemi-AnnualELN-0801B455DBG1Monitoring WellSemi-AnnualELN-0801C456DBG1Monitoring WellSemi-AnnualELN-0801E457DBG1Monitoring WellBiennialELN-0802A458DBG11Monitoring WellBiennialELN-0802A459DBG11Monitoring WellSemi-AnnualELN-1001B460DBG1Monitoring WellSemi-AnnualELN-1001C461DBG1Monitoring WellSemi-AnnualELN-1001E462DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1003A467DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellSemi-AnnualDBN-1001C473DBG1Monitoring WellSemi-AnnualDBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualELN-1003C533DBG1Monitoring WellSemi-AnnualELN-1003C534 <td>DBN-9501C</td> <td>316</td> <td>DBG</td> <td></td> <td></td> <td>1</td> <td>Monitoring Well</td> <td>Semi-Annual</td>	DBN-9501C	316	DBG			1	Monitoring Well	Semi-Annual
ELN-0801B455DBG1Monitoring WellSemi-AnnualELN-0801C456DBG1Monitoring WellSemi-AnnualELN-0801E457DBG1Monitoring WellSemi-AnnualELN-0802C459DBG11Monitoring WellBiennialELN-0802C459DBG11Monitoring WellSemi-AnnualELN-1001B460DBG1Monitoring WellSemi-AnnualELN-1001C461DBG1Monitoring WellSemi-AnnualELN-1001E462DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002E464DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellQuarterlyELN-1003A467DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellSemi-AnnualDBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1001E477DBG1Monitoring WellSemi-AnnualDBN-1001E476DBG1Monitoring WellSemi-AnnualELN-1502A533 <td>DBN-9501E</td> <td>317</td> <td>DBG</td> <td></td> <td></td> <td>1</td> <td>Monitoring Well</td> <td>Semi-Annual</td>	DBN-9501E	317	DBG			1	Monitoring Well	Semi-Annual
ELN-0801C456DBG1Monitoring WellSemi-AnnualELN-0801E457DBG1Monitoring WellSemi-AnnualELN-0802A458DBG11Monitoring WellBiennialELN-0802C459DBG1Monitoring WellBiennialELN-1001B460DBG1Monitoring WellSemi-AnnualELN-1001C461DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002B464DBG1Monitoring WellSemi-AnnualELN-1002C465DBG1Monitoring WellSemi-AnnualELN-1002C466DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellQuarterlyELN-1003A467DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyELN-1003E476DBG1Monitoring WellSemi-AnnualDBN-1001C473DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1502A534DBG<	ELN-0801B	455	DBG			1	Monitoring Well	Semi-Annual
ELN-0801E457DBG1Monitoring WellSemi-AnnualELN-0802A458DBG11Monitoring WellBiennialELN-0802C459DBG11Monitoring WellSemi-AnnualELN-1001E460DBG1Monitoring WellSemi-AnnualELN-1001C461DBG1Monitoring WellSemi-AnnualELN-1001E462DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002C465DBG1Monitoring WellSemi-AnnualELN-1002C466DBG1Monitoring WellSemi-AnnualELN-1002C466DBG1Monitoring WellQuarterlyELN-1003C469DBG1Monitoring WellQuarterlyELN-1003C469DBG1Monitoring WellQuarterlyELN-1003C469DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellSemi-AnnualDBN-1001C473DBG1Monitoring WellSemi-AnnualDBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1001E477DBG1Monitoring WellSemi-AnnualDBN-1001C473DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1502A535	ELN-0801C	456	DBG			1	Monitoring Well	Semi-Annual
ELN-0802A458DBG11Monitoring WellBiennialELN-0802C459DBG11Monitoring WellBiennialELN-1001B460DBG1Monitoring WellSemi-AnnualELN-1001C461DBG1Monitoring WellSemi-AnnualELN-1001E462DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002A465DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellQuarterlyELN-1003A467DBG1Monitoring WellQuarterlyELN-1003E469DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellSemi-AnnualDBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1001E476DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1001E477DBG1Monitoring WellSemi-AnnualDBN-1002E476DBG1Monitoring WellSemi-AnnualELN-1503A533DBG1Monitoring WellSemi-AnnualELN-1503C536 <td< td=""><td>ELN-0801E</td><td>457</td><td>DBG</td><td></td><td></td><td>1</td><td>Monitoring Well</td><td>Semi-Annual</td></td<>	ELN-0801E	457	DBG			1	Monitoring Well	Semi-Annual
ELN-0802C459DBG11Monitoring WellBiennialELN-1001B460DBG1Monitoring WellSemi-AnnualELN-1001C461DBG1Monitoring WellSemi-AnnualELN-1001C463DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002B464DBG1Monitoring WellSemi-AnnualELN-1002C465DBG1Monitoring WellSemi-AnnualELN-1002B466DBG1Monitoring WellQuarterlyELN-1003A467DBG1Monitoring WellQuarterlyELN-1003B468DBG1Monitoring WellQuarterlyELN-1003C469DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyDBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002C477DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002C534DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503A536DBG	ELN-0802A	458	DBG		1	1	Monitoring Well	Biennial
ELN-1001B460DBG1Monitoring WellSemi-AnnualELN-1001C461DBG1Monitoring WellSemi-AnnualELN-1001E462DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002B464DBG1Monitoring WellSemi-AnnualELN-1002C465DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1003B467DBG1Monitoring WellQuarterlyELN-1003B468DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyDBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1001E476DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1502A534DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1	ELN-0802C	459	DBG		1	1	Monitoring Well	Biennial
ELN-1001C461DBG1Monitoring WellSemi-AnnualELN-1001E462DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002B464DBG1Monitoring WellSemi-AnnualELN-1002C465DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1003A467DBG1Monitoring WellQuarterlyELN-1003A467DBG1Monitoring WellQuarterlyELN-1003C469DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyDBN-1001B472DBG1Monitoring WellSemi-AnnualDBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1	ELN-1001B	460	DBG			1	Monitoring Well	Semi-Annual
ELN-1001E462DBG1Monitoring WellSemi-AnnualELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002B464DBG1Monitoring WellSemi-AnnualELN-1002C465DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1003A467DBG1Monitoring WellQuarterlyELN-1003B468DBG1Monitoring WellQuarterlyELN-1003C469DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyDBN-1001B472DBG1Monitoring WellSemi-AnnualDBN-1001E473DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualELN-1502C534DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503A536DBG1Monitoring WellSemi-AnnualELN-1503A536DBG1Monitoring WellSemi-AnnualELN-1503A536DBG1Monitoring WellSemi-AnnualELN-1503A537DBG1	ELN-1001C	461	DBG			1	Monitoring Well	Semi-Annual
ELN-1002A463DBG1Monitoring WellSemi-AnnualELN-1002B464DBG1Monitoring WellSemi-AnnualELN-1002C465DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1003A467DBG1Monitoring WellQuarterlyELN-1003B468DBG1Monitoring WellQuarterlyELN-1003E469DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellSemi-AnnualDBN-1001B472DBG1Monitoring WellSemi-AnnualDBN-1001C473DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1	ELN-1001E	462	DBG			1	Monitoring Well	Semi-Annual
ELN-1002B464DBG1Monitoring WellSemi-AnnualELN-1002C465DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1003A467DBG1Monitoring WellQuarterlyELN-1003B468DBG1Monitoring WellQuarterlyELN-1003C469DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellSemi-AnnualDBN-1001B472DBG1Monitoring WellSemi-AnnualDBN-1001C473DBG1Monitoring WellSemi-AnnualDBN-1001C474DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1503C535DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1503C537DBG1	ELN-1002A	463	DBG			1	Monitoring Well	Semi-Annual
ELN-1002C465DBG1Monitoring WellSemi-AnnualELN-1002E466DBG1Monitoring WellQuarterlyELN-1003A467DBG1Monitoring WellQuarterlyELN-1003B468DBG1Monitoring WellQuarterlyELN-1003E469DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyELN-1001B472DBG1Monitoring WellSemi-AnnualDBN-1001C473DBG1Monitoring WellSemi-AnnualDBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualDBN-1002E476DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1502A535DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503A536DBG1Monitoring WellSemi-AnnualELN-1503A537DBG1 </td <td>ELN-1002B</td> <td>464</td> <td>DBG</td> <td></td> <td></td> <td>1</td> <td>Monitoring Well</td> <td>Semi-Annual</td>	ELN-1002B	464	DBG			1	Monitoring Well	Semi-Annual
ELN-1002E466DBG1Monitoring WellSemi-AnnualELN-1003A467DBG1Monitoring WellQuarterlyELN-1003B468DBG1Monitoring WellQuarterlyELN-1003C469DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyDBN-1001B472DBG1Monitoring WellSemi-AnnualDBN-1001C473DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualS1121755DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503A537DBG1Monitoring WellSemi-AnnualELN-1503A537DBG1Monitoring WellSemi-AnnualELN-1503A537DBG1Monitoring WellSemi-AnnualELN-1503A537DBG1Monitoring WellSemi-AnnualELN-1503A537DBG1 <td>ELN-1002C</td> <td>465</td> <td>DBG</td> <td></td> <td></td> <td>1</td> <td>Monitoring Well</td> <td>Semi-Annual</td>	ELN-1002C	465	DBG			1	Monitoring Well	Semi-Annual
ELN-1003A467DBG1Monitoring WellQuarterlyELN-1003B468DBG1Monitoring WellQuarterlyELN-1003C469DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyDBN-1001B472DBG1Monitoring WellSemi-AnnualDBN-1001C473DBG1Monitoring WellSemi-AnnualDBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualSI121755DBG1Monitoring WellSemi-AnnualSI121755DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1503C440NC Area1Monitoring WellSemi-AnnualRIM-0703440NC Area1Monitoring WellSemi-AnnualRIM-0705478NC Area1Monitoring WellSemi-AnnualRIN-1002478NC Area1Monitoring WellSemi-AnnualRIN-1001A480NC Area<	ELN-1002E	466	DBG			1	Monitoring Well	Semi-Annual
ELN-1003B468DBG1Monitoring WellQuarterlyELN-1003C469DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyDBN-1001B472DBG1Monitoring WellSemi-AnnualDBN-1001C473DBG1Monitoring WellSemi-AnnualDBN-1001C474DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellAnnualRIM-0703440NC Area1Monitoring WellSemi-AnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellSemi-AnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC	ELN-1003A	467	DBG			1	Monitoring Well	Quarterly
ELN-1003C469DBG1Monitoring WellQuarterlyELN-1003E470DBG1Monitoring WellQuarterlyDBN-1001B472DBG1Monitoring WellSemi-AnnualDBN-1001C473DBG1Monitoring WellSemi-AnnualDBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualS1121755DBG1Monitoring WellSemi-AnnualS1121755DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1502C534DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellGuarterlyRIM-0703440NC Area1Monitoring WellSemi-AnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellSemi-AnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Ar	ELN-1003B	468	DBG			1	Monitoring Well	Quarterly
ELN-1003E470DBG1Monitoring WellQuarterlyDBN-1001B472DBG1Monitoring WellSemi-AnnualDBN-1001C473DBG1Monitoring WellSemi-AnnualDBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualS1121755DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellSemi-AnnualRIM-0705442NC Area1Monitoring WellSemi-AnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellSemi-AnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellSemi-Annual	ELN-1003C	469	DBG			1	Monitoring Well	Quarterly
DBN-1001B472DBG1Monitoring WellSemi-AnnualDBN-1001C473DBG1Monitoring WellSemi-AnnualDBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualS1121755DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1502C534DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellSemi-AnnualRIM-0703440NC Area1Monitoring WellSemi-AnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellSemi-AnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellSemi-Annual	ELN-1003E	470	DBG			1	Monitoring Well	Quarterly
DBN-1001C473DBG1Monitoring WellSemi-AnnualDBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualS1121755DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1502C534DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1503C537DBG1Monitoring WellSemi-AnnualELN-1503C440NC Area1Monitoring WellSemi-AnnualRIM-0703440NC Area1Monitoring WellSemi-AnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellSemi-AnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellSemi-Annual	DBN-1001B	472	DBG			1	Monitoring Well	Semi-Annual
DBN-1001E474DBG1Monitoring WellSemi-AnnualDBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualS1121755DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1502C534DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellSemi-AnnualRIM-0705442NC Area1Monitoring WellSemi-AnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellSemi-AnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellSemi-Annual	DBN-1001C	473	DBG			1	Monitoring Well	Semi-Annual
DBN-1002C476DBG1Monitoring WellSemi-AnnualDBN-1002E477DBG1Monitoring WellSemi-AnnualS1121755DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1502C534DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellAnnualRIM-0703440NC Area1Monitoring WellAnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellAnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellSemi-Annual	DBN-1001E	474	DBG			1	Monitoring Well	Semi-Annual
DBN-1002E477DBG1Monitoring WellSemi-AnnualS1121755DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1502C534DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellQuarterlyRIM-0703440NC Area1Monitoring WellSemi-AnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellSemi-AnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellSemi-Annual	DBN-1002C	476	DBG			1	Monitoring Well	Semi-Annual
S1121755DBG1Monitoring WellSemi-AnnualELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1502C534DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellQuarterlyRIM-0703440NC Area1Monitoring WellQuarterlyRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellSemi-AnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellSemi-Annual	DBN-1002E	477	DBG			1	Monitoring Well	Semi-Annual
ELN-1502A533DBG1Monitoring WellSemi-AnnualELN-1502C534DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellQuarterlyRIM-0703440NC Area1Monitoring WellQuarterlyRIM-0705442NC Area1Monitoring WellSemi-AnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellAnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellAnnual	S1121	755	DBG			1	Monitoring Well	Semi-Annual
ELN-1502C534DBG1Monitoring WellSemi-AnnualELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellQuarterlyRIM-0703440NC Area1Monitoring WellAnnualRIM-0705442NC Area1Monitoring WellSemi-AnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellAnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellAnnual	ELN-1502A	533	DBG			1	Monitoring Well	Semi-Annual
ELN-1503A535DBG1Monitoring WellSemi-AnnualELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellQuarterlyRIM-0703440NC Area1Monitoring WellAnnualRIM-0705442NC Area1Monitoring WellSemi-AnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellAnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellAnnual	ELN-1502C	534	DBG			1	Monitoring Well	Semi-Annual
ELN-1503C536DBG1Monitoring WellSemi-AnnualELN-1504B537DBG1Monitoring WellQuarterlyRIM-0703440NC Area1Monitoring WellAnnualRIM-0705442NC Area1Monitoring WellSemi-AnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellAnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellAnnual	ELN-1503A	535	DBG			1	Monitoring Well	Semi-Annual
ELN-1504B537DBG1Monitoring WellQuarterlyRIM-0703440NC Area1Monitoring WellAnnualRIM-0705442NC Area1Monitoring WellSemi-AnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellAnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellAnnual	ELN-1503C	536	DBG			1	Monitoring Well	Semi-Annual
RIM-0703440NC Area1Monitoring WellAnnualRIM-0705442NC Area1Monitoring WellSemi-AnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellAnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellAnnual	ELN-1504B	537	DBG			1	Monitoring Well	Quarterly
RIM-0705442NC Area1Monitoring WellSemi-AnnualRIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellAnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellAnnual	RIM-0703	440	NC Area			1	Monitoring Well	, Annual
RIM-1002478NC Area1Monitoring WellSemi-AnnualRIN-1007C479NC Area1Monitoring WellAnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellAnnual	RIM-0705	442	NC Area			1	Monitoring Well	Semi-Annual
RIN-1007C479NC Area1Monitoring WellAnnualRIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellAnnual	RIM-1002	478	NC Area			1	Monitoring Well	Semi-Annual
RIN-1001A480NC Area1Monitoring WellSemi-AnnualRIN-1001C481NC Area1Monitoring WellAnnual	RIN-1007C	479	NC Area			1	Monitoring Well	Annual
RIN-1001C 481 NC Area 1 Monitoring Well Annual	RIN-1001A	480	NC Area			1	Monitoring Well	Semi-Annual
	RIN-1001C	481	NC Area			1	Monitoring Well	Annual

## September 2020 Groundwater Sampling Schedule

Well Name	<u>Well ID</u>	Plume Area	<u>Nitrates</u>	<u>VOC</u>	DNT	Well Type	Sample Frequency
S1125	504	NC Area			1	Monitoring Well	Semi-Annual
PBM-9801	360	PBG		1	1	Monitoring Well	Annual
PBM-0001	367	PBG	1	1	1	Monitoring Well	Semi-Annual
PBM-0002	368	PBG	1	1	1	Monitoring Well	Semi-Annual
PBM-0006	372	PBG	1	1	1	Monitoring Well	Semi-Annual
PBM-0008	374	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9101C	561	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9102B	562	PBG		1	1	Monitoring Well	Biennial
PBN-9102C	563	PBG		1	1	Monitoring Well	Biennial
SWN-9102C	569	PBG		1	1	Monitoring Well	Annual
SWN-9102D	570	PBG		1	1	Monitoring Well	Annual
SWN-9103B	571	PBG		1	1	Monitoring Well	Semi-Annual
SWN-9103C	572	PBG		1	1	Monitoring Well	Semi-Annual
SWN-9103D	573	PBG		1	1	Monitoring Well	Semi-Annual
SWN-9103E	574	PBG		1	1	Monitoring Well	Semi-Annual
SWN-9104C	575	PBG		1	1	Monitoring Well	Semi-Annual
SWN-9104D	576	PBG		1	1	Monitoring Well	Semi-Annual
SWN-9105B	577	PBG		1	1	Monitoring Well	Annual
SWN-9105C	578	PBG		1	1	Monitoring Well	Annual
SWN-9105D	579	PBG		1	1	Monitoring Well	Annual
PBN-1003C	592	PBG		1	1	Monitoring Well	Annual
PBN-1001C	595	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8202A	613	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8202B	614	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8202C	615	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8205A	622	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8205B	623	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8205C	624	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8502A	632	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8503A	633	PBG		1	1	Monitoring Well	Semi-Annual
PBM-8907	637	PBG		1	1	Monitoring Well	Annual
PBM-8909	639	PBG		1	1	Monitoring Well	Biennial
PBN-8902C	645	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8903B	646	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8903C	647	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8912A	654	PBG		1	1	Monitoring Well	Semi-Annual
PBN-8912B	655	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9112C	665	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9112D	666	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9301B	668	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9301C	669	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9303B	673	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9303C	674	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9303D	675	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9304D	687	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9902D	691	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9903A	692	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9903B	693	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9903C	694	PBG		1	1	Monitoring Well	Semi-Annual
PBN-9903D	695	PBG		1	1	Monitoring Well	Semi-Annual
S1147	709	PBG		1	1	Monitoring Well	Semi-Annual
S1148	710	PBG		1	1	Monitoring Well	Semi-Annual
SPN-8903B	718	PBG		1	1	Monitoring Well	Semi-Annual

# September 2020 Groundwater Sampling Schedule

Well Name	Well ID	<u>Plume Area</u>	<u>Nitrates</u>	VOC	<u>DNT</u>	Well Type	Sample Frequency
SPN-8903C	719	PBG		1	1	Monitoring Well	Semi-Annual
SPN-8904B	720	PBG		1	1	Monitoring Well	Semi-Annual
SPN-8904C	721	PBG		1	1	Monitoring Well	Semi-Annual
SPN-9103D	725	PBG		1	1	Monitoring Well	Semi-Annual
SPN-9104D	726	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1302A	770	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1302B	771	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1302C	772	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1302D	773	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1303A	774	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1303B	775	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1303C	776	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1303D	777	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1304A	778	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1304B	779	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1304C	780	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1304D	781	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1401A	782	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1401B	783	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1401C	784	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1404B	791	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1404C	792	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1404D	793	PBG		1	1	Monitoring Well	Semi-Annual
PBN-1405F	794	PBG		1	1	Monitoring Well	Biennial
PBN-8902BR	795	PBG		1	1	Monitoring Well	Semi-Annual
PBM-9001D	981	PBG		1	1	Monitoring Well	Semi-Annual
PBM-9002D	982	PBG		1	1	Monitoring Well	Biennial
Totals			3	81	133		

Totals

81

## November 2020 Groundwater Sampling Schedule

<u>Well Name</u>	Well ID	<u>Plume Area</u>	VOC	DNT	<u>Well Type</u>	Sample Frequency
SEN-0501A	580	Central	1	1	Monitoring Well	Semi-Annual
SEN-0501B	581	Central	1	1	Monitoring Well	Semi-Annual
SEN-0501D	582	Central	1	1	Monitoring Well	Semi-Annual
SEN-0502A	583	Central	1	1	Monitoring Well	Semi-Annual
SEN-0502D	584	Central	1	1	Monitoring Well	Semi-Annual
SEN-0503A	585	Central	1	1	Monitoring Well	Semi-Annual
SEN-0503B	586	Central	1	1	Monitoring Well	Semi-Annual
SEN-0503D	587	Central	1	1	Monitoring Well	Semi-Annual
RIN-1004B	498	Central		1	Monitoring Well	Semi-Annual
WE-ZE512	437	Central	1	1	<b>Residential Well</b>	Quarterly
Purcell-D	163	DBG	1	1	<b>Residential Well</b>	Quarterly
ELN-1003A	467	DBG		1	Monitoring Well	Quarterly
ELN-1003B	468	DBG		1	Monitoring Well	Quarterly
ELN-1003C	469	DBG		1	Monitoring Well	Quarterly
ELN-1003E	470	DBG		1	Monitoring Well	Quarterly
ELN-1504B	537	DBG		1	Monitoring Well	Quarterly

Totals

10

16

Appendix E

Plume Concentration Over Time Graphs
## **Concentration Graphs Propellant Burning Ground Plume**

Source Area Wells	<u>Compound</u>	Year Range	<u>Page</u>
PBM-0002, PBN-8202A, B, C	DNT	1992 - 2018	1
PBM-0002, PBN-8202A, B, C	DNT	2004 - 2018	2
PBM-0002, PBN-8202A, B, C	CTET	1988 - 2018	3
PBM-0002, PBN-8202A, B, C	Chloroform	1988 - 2018	4
PBM-0002, PBN-8202A, B, C	TCE	1988 - 2018	5
PBM-0008	DNT	2000 - 2018	6
PBM-0008	DNT	2010 - 2018	7
PBN-8205A, B, C	DNT	1989 - 2018	8
PBN-8205A, B, C	CTET	1982 - 2018	9
PBN-8205A, B, C	Chloroform	1983 - 2018	10
PBN-8205A, B, C	TCE	1982 - 2018	11

On-Site Downgradient Wells	<u>Compound</u>	<u>Year Range</u>	<u>Page</u>
PBN-8502A, 8902BR, 8902C	DNT	1989 - 2018	12
PBN-8502A, 8902BR, 8902C	CTET	1988 - 2018	13
PBN-8502A, 8902BR, 8902C	Chloroform	1988 - 2018	14
PBN-8502A, 8902BR, 8902C	TCE	1988 - 2018	15
PBN-8912A, B, 9112C, D	DNT	1989 - 2018	16
PBN-8912A, B, 9112C, D	CTET	1989 - 2018	17
PBN-8912A, B, 9112C, D	Chloroform	1989 - 2018	18
PBN-8912A, B, 9112C, D	TCE	1989 - 2018	19
S1147, SPN-8903B, C, 9103D	CTET	1988 - 2018	20
S1147, SPN-8903B, C, 9103D	Chloroform	1988 - 2018	21
S1147, SPN-8903B, C, 9103D	TCE	1988 - 2018	22
S1148, SPN-8904B, C, 9104D	DNT	2010 - 2018	23
S1148, SPN-8904B, C, 9104D	CTET	1988 - 2018	24
S1148, SPN-8904B, C, 9104D	Chloroform	1988 - 2018	25
S1148, SPN-8904B, C, 9104D	TCE	1988 - 2018	26
PBN-1001C	Ethyl Ether	2010 - 2018	27
PBN-9304D	Ethyl Ether	2013 - 2018	28

Off-Site Downgradient Wells	<u>Compound</u>	<u>Year Range</u>	<u>Page</u>
PBN-9903D	Ethyl Ether	2005 - 2018	29
PBN-9903A, B, C, D	DNT	2000 - 2018	30
PBN-9903A, B, C, D	CTET	2000 - 2018	31
PBN-9903A, B, C, D	Chloroform	2000 - 2018	32
PBN-9903A, B, C, D	TCE	2000 - 2018	33
SWN-9102C, D	DNT	1991 - 2018	34
SWN-9102C, D	CTET	1991 - 2018	35
SWN-9102C, D	Chloroform	1991 - 2018	36
SWN-9102C, D	TCE	1991 - 2018	37
SWN-9103B, C, D, E	DNT	1991 - 2018	38
SWN-9103B, C, D, E	CTET	1991 - 2018	39
SWN-9103B, C, D, E	Chloroform	1991 - 2018	40
SWN-9103B, C, D, E	TCE	1991 - 2018	41
SWN-9104C, D	DNT	1991 - 2018	42
SWN-9104C, D	CTET	1991 - 2018	43
SWN-9104C, D	Chloroform	1991 - 2018	44
SWN-9104C, D	TCE	1991 - 2018	45
PBN-9101C, PBM-9001D	DNT	1991 - 2018	46
PBN-9101C, PBM-9001D	CTET	1991 - 2018	47
PBN-9101C, PBM-9001D	Chloroform	1991 - 2018	48
PBN-9101C, PBM-9001D	TCE	1991 - 2018	49
PBN-9102B, C, PBM-9002D	DNT	1991 - 2018	50
PBN-9102B, C, PBM-9002D	CTET	1991 - 2018	51
PBN-9102B, C, PBM-9002D	Chloroform	1991 - 2018	52
PBN-9102B, C, PBM-9002D	TCE	1991 - 2018	53










































































































## **Concentration Graphs Deterrent Burning Ground Plume**

Plume Trend Wells	Compound	Year Range	Page
DBM-8201, 8202, ELM-8907, 8908	DNI	1998 - 2018	1
Source Area Wells	<u>Compound</u>	<u>Year Range</u>	<u>Page</u>
DBM-8201	DNT	1998 - 2018	2
DBM-8202	DNT	1998 - 2018	3
DBN-1001B, C, E	DNT	2010 - 2018	4
On-Site Downgradient Wells	<u>Compound</u>	<u>Year Range</u>	<u>Page</u>
On-Site Downgradient Wells ELM-8901	<u>Compound</u> DNT	<u>Year Range</u> 2009 - 2018	<u>Page</u> 5
On-Site Downgradient Wells ELM-8901 ELM-8907, 8908	<u>Compound</u> DNT DNT	<u>Year Range</u> 2009 - 2018 1998 - 2018	<u>Page</u> 5 6
<u>On-Site Downgradient Wells</u> ELM-8901 ELM-8907, 8908 ELN-1502A, C	Compound DNT DNT DNT	<u>Year Range</u> 2009 - 2018 1998 - 2018 2015 - 2018	<u>Page</u> 5 6 7
<u>On-Site Downgradient Wells</u> ELM-8901 ELM-8907, 8908 ELN-1502A, C ELM-9501, ELN-0801B, C, E	Compound DNT DNT DNT DNT	<u>Year Range</u> 2009 - 2018 1998 - 2018 2015 - 2018 2002 - 2018	<u>Page</u> 5 6 7 8
<u>On-Site Downgradient Wells</u> ELM-8901 ELM-8907, 8908 ELN-1502A, C ELM-9501, ELN-0801B, C, E	<u>Compound</u> DNT DNT DNT DNT	<u>Year Range</u> 2009 - 2018 1998 - 2018 2015 - 2018 2002 - 2018	<u>Page</u> 5 6 7 8
On-Site Downgradient Wells ELM-8901 ELM-8907, 8908 ELN-1502A, C ELM-9501, ELN-0801B, C, E Off-Site Downgradient Wells	Compound DNT DNT DNT DNT Compound	<u>Year Range</u> 2009 - 2018 1998 - 2018 2015 - 2018 2002 - 2018 <u>Year Range</u>	Page 5 6 7 8 Page
On-Site Downgradient Wells ELM-8901 ELM-8907, 8908 ELN-1502A, C ELM-9501, ELN-0801B, C, E <u>Off-Site Downgradient Wells</u> ELN-1003A, B, C, E	Compound DNT DNT DNT DNT Compound DNT	<u>Year Range</u> 2009 - 2018 1998 - 2018 2015 - 2018 2002 - 2018 <u>Year Range</u> 2010 - 2018	Page 5 6 7 8 <u>Page</u> 9

**BAAP Groundwater Data** 



















## Concentration Graphs Central Plume

<u>Compound</u>	Year Range	<u>Page</u>
DNT	1998 - 2018	1
DNT	1998 - 2018	2
DNT	2010 - 2018	3
DNT	2010 - 2018	4
DNT	2010 - 2018	5
DNT	2006 - 2018	6
DNT	2010 - 2018	7
	Compound DNT DNT DNT DNT DNT DNT DNT	CompoundYear RangeDNT1998 - 2018DNT1998 - 2018DNT2010 - 2018DNT2010 - 2018DNT2010 - 2018DNT2006 - 2018DNT2010 - 2018

Off-Site Downgradient Wells	<u>Compound</u>	Year Range	<u>Page</u>
SEN-0501A, B, D	DNT	2005 - 2018	8
SEN-0502A, D	DNT	2005 - 2018	9
SEN-0503A, B, D	DNT	2005 - 2018	10




















## **Concentration Graphs Nitrocellulose Production Area Plume**

Source Area Wells	<u>Compound</u>	<u>Year Range</u>	<u>Page</u>
RIM-1002	DNT	2010 - 2018	1
RIM-0705, RIN-1007C	DNT	2007 - 2018	2
On-Site Downgradient Wells	<u>Compound</u>	<u>Year Range</u>	<u>Page</u>
RIN-1001A, C	DNT	2010 - 2018	3







Appendix F

Vapor Intrusion Investigation Reports (2012)



DEPARTMENT OF THE ARMY BADGER ARMY AMMUNITION PLANT 2 BADGER ROAD BARABOO, WISCONSIN 53913-5000

August 15, 2012

SUBJECT: Vapor Intrusion Pathway Analysis Reports Badger Army Ammunition Plant

Mr. Will Myers Hydrogeologist Program Coordinator Wisconsin Department of Natural Resources South Central Region 3911 Fish Hatchery Road Fitchburg, WI 53711-5397

Dear Mr. Myers:

The Wisconsin Department of Natural Resources (WDNR) issued a letter dated September 9, 2011 reminding responsible parties that a vapor intrusion investigation should be conducted at all sites where volatile organic compounds (VOC) are present in the soil and groundwater. The Propellant Burning Ground (PBG) Plume contains the following VOCs: carbon tetrachloride, chloroform, and trichloroethylene. The Deterrent Burning Ground and Central Plumes contain primarily dinitrotoluene, which does not pose a vapor pathway risk.

Based on the WDNR letter and the Alternative Feasibility Study – Groundwater Remedial Strategy, Badger Technical Services, LLC (BTS) conducted a vapor intrusion pathway analysis for the PBG Plume.

In February 2012, BTS personnel conducted ten vapor intrusion pathway borings in accordance with WDNR vapor intrusion guidance, *Addressing Vapor Intrusion at Remediation & Redevelopment Sites in Wisconsin*, PUB-RR-800, December 2010 using the post-run tubing vapor sampling technique. Due to inconclusive results for half the samples, due to ineffectively sealed fittings, BTS recommended conducting further investigation of this pathway at five locations.

On April 19, 2012, BTS submitted the Vapor Intrusion Pathway Analysis Report to the Department of the Army (Enclosure 1). A complete evaluation of the vapor pathway was not possible based on the initial information because several of the soil gas samples reported elevated levels of the leak detection tracer gas. Therefore, BTS conducted a supplemental investigation in June and July 2012, as reported in the BTS July 16, 2012, Supplemental Vapor Intrusion Pathway Analysis Report (Enclosure 2), at locations where the soil gas analytical data were deemed invalid be re-sampled using helium as the leak-detection tracer gas using an alternate leak detection methodology.

Based on the results of the activities presented within these reports, the data indicate the PBG Plume does not present a significant risk to human health via vapor intrusion off-site. Analytical results of soil gas samples collected off-site do not exceed the WDNR Vapor Risk Screening Levels for Deep Soil Gas.

Please do not hesitate to contact me at 608-643-3361 if you have any questions.

Sincerely,

Joan M. Kenney

Joan M. Kenney Commander's Representative

Attachments

Enclosures

,

Copy furn: Hank Kuehling, WDNR SCR R&R Program LTE Hydrogeologist Michelle Mullin, U.S. Environmental Protection Agency Ryan Wozniak, Wisconsin Department of Health and Family Services Ralph Jesse, U.S. Department of Agriculture Badger Technical Services, LLC Attachments

Soil Boring Log Information

State Depai	of Wisco tment of	onsin Matura	al Resource	es		_	_			SOII	BOR	ING LO	G INFC For	ORMA' m 440	TION 0-122	
		<u>Rout</u>	i <u>e To:</u> R	Watershed/Wastewater Remediation/Redevelopment	Waste Manager Other	nent [						Pa	ge 1 of	1		
Facility	/Project	Name			License/Perm	nit/Mor	itorin	ig Numł	ber	Boring	y Num	ber	50 1 01			
Badg	er Arm	y Am	munition	n Plant / Vapor Intrusion				8		VIP-	1					
Pathv	vay An	alysis	6			~						~				
Boring First N	Drilled ame:	by: Na Dav	me of crew	chief (first, last) and Firm Last Name: Paulson	Date Drilling	Started		Date Dr. 2/27/1	ıllıng ( 7	Complet	ted	Drilling Direct-	Methoo Push	1		
Firm:	S	Soil Es	ssentials	LTD	2/2//12		4	2/2//1	2			Direct	1 4511			
WI Un	ique We	ll No	DNR W	Vell ID No. Well Name	Final Static V	Vater MSL		Surface	Elevat Feet I	ion MSL		Boi 2.	ehole I 125	Diamet	<sup>er</sup> hes	
Local C	Grid Orig	gin 5/	(estin	mated: ) or Boring Location	Local Grid Loc	cation		м					Б			
NE	of SV	V of	Section	14 T 10 N R 6 E	1	Feet		N S			Fee	•t	E W			
Facili	tv ID			County	County Co	de		Civi	il Tov	vn/Cit	$\frac{1}{v/or}$	Village				
15705	53930			Sauk	57			Том	vn of	Sumn	, ter	0				
10700				Suuk	57		1	100			nei					
San	ple		face)								So	il Proper	ties			
_	&	s	et d sur	SOIL BOCK DESCRIPTIO	N		50	m		e		t	dex		nents	
r and	Att. red (	ount	n Fe	SOIL ROCK DESCRIPTIC			c Log	iagra	0	ssiv h	e -	Limi	ty In		umo	
mbe oe	ngth cove	W C	pth i low g			CS	ihhid	II D	)/FII	mpre engtl	istu nten	biu	stici	00	D/C	
Nu Tyj	Lei Rei	Blc	De] (Be			SN	<u>G</u>	We	IId	Co: Str	Co	Liq	Pla	P 2	RQ	
			0-45'	No soil samples were collected. drilled to install HDPE tubing for vapor collection. B.T. @ 45' bgs.	Boring or soil											
1 here	by cert	tify th	at the in	tormation on this form is true and	t correct to t	he bes	st of	my kn	lowle	dge.						
Signati	ure: Jef	f Lark	kin		Fi	rm: E	Badg	er Tec	hnica	l Serv	rices,	LLC.				
This for in forfe this for	rm is au eiture of m is not	thorize betwee intend	d by Chapt n \$10 and ed to be us	ters 281, 283, 289, 291, 292, 293, 295, an \$25,000, or imprisonment for up to one y ed for any other purpose. NOTE: See ins	d 299, Wis. Sta year, depending structions for m	ts. Con on the ore info	mpleti progr	ion of th am and ion, incl	nis forr condu luding	n is mai ct invol where t	ndator ved. F he con	y. Failur ersonally	e to file identi orm sho	this fo fiable i ould be	orm may nformati sent.	result on on

State Depai	of Wisco tment of	onsin Matura <u>Rout</u>	al Resource <u>te To:</u>	es Watershed/Wastewater	Waste Manage	ment	]			SOII	BOR	ING LO	G INFO Foi	ORMA' rm 440	TION 0-122	
			k	Remediation/Redevelopment	Other							Pa	ge 1 of	1	_	
Facility Badg Pathy	y/Project er Arm vav An	Name y Am alvsis	munition	n Plant / Vapor Intrusion	License/Perm	nit/Moni	itorin	g Numl	ber	Boring VIP-2	g Numl 2	ber			<u> </u>	
Boring First N Firm:	Drilled ame:	by: Na Dav Soil Es	me of crew e ssentials	v chief (first, last) and Firm Last Name: Paulson LTD	Date Drilling 2/27/12	g Started	1 2	Date Dr 2/27/1	illing ( 2	Complet	ed	Drilling Direct-	Metho Push	d		
WI Un	ique We	ll No	DNR W	Vell ID No. Well Name	Final Static	Water	5	Surface	Elevat	ion		Boi	rehole I	Diamet	er	
Local ( State	Grid Orig Plane	gin 54	(estin 40372	mated: ) or Boring Location N, 317498 E	Local Grid Lo	cation		N		VISL		<u></u> .	<u>Е</u>		lies	
SW	of NI	E of	Section	14 T 10 N R 6 E		Feet		S			Fee	et	W			
Facili	ty ID			County	County Co	ode		Civi	l Tov	vn/Cit	y/ or	Village	•			
1570	53930			Sauk	57	· · · · ·		Tov	vn of	Sump	ter				1	
San	ple		face)								So	il Proper	ties	Γ		
Number and Type	Length Att. & Recovered (in)	Blow Counts	Depth in Feet (Below ground sur	SOIL ROCK DESCRIPTIO	DN	USCS	Graphic Log	Well Diagram	PID/FID	Compressive Strength	Moisture Content	Liquid Limit	Plasticity Index	P 200	RQD/Comments	
			0-23	No soil samples were collected. drilled to install HDPE tubing f vapor collection. B.T. @ 25' bgs.	or soil											
1 nere	by cert	f I or	at the in	iormation on this form is true and	u correct to t	ne bes	i of :	my kn	owle	uge.	inca					-
Signat	are: Jef	1 Lark	d by Charl	tore 201 202 200 201 202 202 205	F	irm: B	adg	er Tec	nnica	u Serv	ices,	LLC.	a to £1	this f		au1+
in forfe this for	eiture of m is not	betwee intend	en \$10 and ed to be us	\$25,000, or imprisonment for up to one sed for any other purpose. NOTE: See in	year, depending structions for m	ns. Con on the port of the por	ipieti progra rmati	am and	condu condu uding	ct invol where t	ved. P	y. Failur ersonally opleted f	e to file y identi orm sho	fiable i	nformation sent.	on

State Depai	of Wisco rtment of	onsin Matur <u>Rout</u>	al Resourc t <u>e To:</u> F	es Watershed/Wastewater Remediation/Redevelopment	Waste Manager Other	ment [				SOII	BOR	NG LO	G INFO Foi	ORMA' rm 440	TION 0-122
				1								Pa	ige 1 of	1	-
Facility Badg Pathy	y/Project er Arm vav An	Name y Am alvsis	munition	n Plant / Vapor Intrusion	License/Pern	nit/Mon	itorir	ng Numb	ber	Boring VIP	g Numb 3	oer			
Boring First N Firm:	Drilled	by: Na Dav	me of crew e	v chief (first, last) and Firm Last Name: Paulson	Date Drilling 2/27/12	g Started	1	Date Dri 2/27/12	illing ( 2	Complet	ed	Drilling Direct	Metho Push	d	
WI Un	ique We	ll No	DNR W	Vell ID No. Well Name	Final Static V	Water MSL		Surface	Elevat Feet l	ion MSL		Bor 2.	rehole I 125	Diamet	er hes
Local ( State )	Grid Orig Plane	gin 54	(estin 40322	mated: ) or Boring Location N, 315818 E	Local Grid Lo	cation		N			Б		E		
SE Essili		v ol	Section		County Co	Feet		S	1 Tar		Fee	t Village	W		
				County		bde			1 1 0 \	vn/Cit	y/ or	v mage			
1570	53930		r	Sauk	57	1		Tow	vn of	Prair	ie du	Sac			
San	nple		face)								Soi	l Proper	ties		
Number and Type	Length Att. & Recovered (in)	Blow Counts	Depth in Feet (Below ground sur	SOIL ROCK DESCRIPTIC	DN	USCS	Graphic Log	Well Diagram	PID/FID	Compressive Strength	Moisture Content	Liquid Limit	Plasticity Index	P 200	RQD/Comments
				drilled to install HDPE tubing for vapor collection. B.T. @ 45' bgs.	or soil										
1 nere	by cer	ing th	at the in	iormation on this form is true and	a correct to t	ne bes	st of	my kn	owle	age.	•				
Signat	ure: Jef	t Lark	cin		Fi	irm: E	3adg	er Tec	hnica	I Serv	ices,	LLC.			
This for in forfe this for	orm is au eiture of rm is not	thorize betwee intend	d by Chapt on \$10 and ed to be us	ters 281, 283, 289, 291, 292, 293, 295, an \$25,000, or imprisonment for up to one y sed for any other purpose. NOTE: See ins	nd 299, Wis. Sta year, depending structions for m	on the	nplet progi ormat	ion of th am and ion, incl	nis forr condu luding	n is mai ct invol where t	ndatory ved. Po he com	r. Failur ersonally pleted f	e to file y identi orm sho	e this fo fiable i ould be	orm may result nformation on sent.

State	of Wisco	onsin f Notur	al Pacouro	-05									SOII	BORI	NG LO	G INFC	ORMA'	TION 0 122	
Depai		Rout	te To:	Waters	shed/Wastewa	ter		Waste Manage	ment							1.01		0-122	
			R	Cemediation	Redevelopme	ent 🗵		Other							Pa	ge 1 of	1	_	
Facility	y/Project	Name						License/Perm	nit/Mon	itorir	ng Numb	ber	Boring	g Numb	er				
Badge	er Arm	y Am alveis	munition	n Plant / V	Vapor Intru	sion							VIP-	4					
Boring First N	Drilled ame:	by: Na Dav	, me of crew e	v chief (first Last Na	, last) and Firr ame: Pau	n lson		Date Drilling 2/27/12	s Started	1	Date Dri 2/27/11	illing ( 2	Complet	ied ]	Drilling Direct-	Methoo Push	d		
Firm:	S	oil E	ssentials	LTD															
WI Un	ique We	ll No	DNR W	Vell ID No.	Well Na	ame		Final Static V Feet	Vater MSL		Surface ]	Elevat Feet l	ion MSL		Boi 2.	ehole I 125	Diamet incl	<sup>er</sup> hes	
Local C State I	Grid Orig Plane	gin 54	(estii 40766	mated: ) N,	or Boring Lo 315809	cation E		Local Grid Lo	cation		N					Е			
SW	of NI	E of	f Section	26 T	10 N	R	6 E		Feet		S			Fee	t	W			
Facili	ty ID			County	T			County Co	ode		Civi	l Tov	vn/Cit	y/ or `	/illage	•			
15705	53930			Sauk				57	1	1	Tow	vn of	Prair	ie du	Sac				
Sam	nple		ace)											Soi	l Proper	ties			
Number and Spe	length Att. & Recovered (in)	low Counts	Depth in Feet Below ground surf		SOIL ROCK	K DESCI	RIPTIC	N	JSCS	Jraphic Log	Vell Diagram	D/FID	Compressive strength	Aoisture Content	iquid Limit	lasticity Index	200	tQD/Comments	
			0-45'	No soil s drilled to vapor co	samples we o install HE ollection. 45' bgs.	re colla	ected. bing fo	Boring or soil											
I here	by cert	tity th	at the in	tormation	on this for	m is tr	ue and	l correct to t	he bes	st of	my kn	$\frac{1}{1}$	dge.						
Signatu	are: Jef	t Lark	cin					F	irm: E	3adg	er Tec	hnica	I Serv	ices, l	LLC.				
This fo in forfe this for	orm 1s au eiture of rm is not	thorize betwee intend	d by Chapt n \$10 and ed to be us	ters 281, 283 \$25,000, or ed for any c	5, 289, 291, 29 imprisonmen other purpose.	92, 293, t for up t NOTE:	295, an to one y See ins	d 299, Wis. Sta ear, depending tructions for m	ts. Cor on the ore info	nplet progi ormat	ion of th ram and ion, incl	us forr condu uding	n 1s mai ct invol where t	ndatory ved. Pe he com	. Failur rsonally pleted f	e to file / identi orm sho	e this fo fiable i ould be	orm may 1 nformatic sent.	esult on on

State Depai	of Wisco tment of	onsin f Natur	al Resourc	es								SOII	BORI	NG LO	G INFO Foi	ORMA' rm 440	TION 0-122	
1		<u>Rout</u>	<u>te To:</u> F	Watershed/V Remediation/Rede	Vastewat velopme	er nt X	Waste Manage Other	ment [	]					Do	an 1 of	. 1		
Facilit	/Project	Name					License/Pern	nit/Mor	itorir	ng Numb	ber	Boring	Numb	er	ge i oi			
Badg	er Arm	y Am	munition	n Plant / Vapo	r Intrus	sion				-8		VIP-	5					
Pathy Boring First N	vay An Drilled	alysis by: Na	me of crew	v chief (first, last) Last Name:	and Firm	n son	Date Drilling	g Started	1	Date Dri	illing (	Complet	ed I	Drilling	Metho	d		
Firm:	s S	Soil Es	c ssentials	LTD	1 au	15011	2/2//12			2/2//1	2		1	Jitet	-i usii			
WI Un	ique We	ll No	DNR W	Vell ID No.	Well Na	ime	Final Static V	Water MSL		Surface	Elevat Feet I	ion MSL		Boi 2.	rehole I 125	Diamet	<sup>er</sup> hes	
Local ( State )	Grid Orig Plane	gin 54	(estin 41097	mated: ) or Be N, 3158	oring Lo	cation E	Local Grid Lo	cation		N			-		Е			
SW	of Ni	e of	f Section	26 T 10	Ν	R 6 E	Country Co	Feet		S	1 Tar		Fee	t 7:110.00	W			
1570	1y ID			County				bde		Tor	1 1 0 v	Dere int	y/ or v	mage	;			
15/0:	53930	1	1	Sauk			57	1	1	10%	n or	Prair	le du s	Sac			1	
San	nple		ace)										Soil	l Proper	ties		-	
Number and Type	Length Att. & Recovered (in)	Blow Counts	Depth in Feet (Below ground surf	SOI	L ROCK	C DESCRIPTIO	ON	USCS	Graphic Log	Well Diagram	PID/FID	Compressive Strength	Moisture Content	Liquid Limit	Plasticity Index	P 200	RQD/Comments	
			0-45'	No soil samp drilled to insi vapor collect B.T. @ 45' b	oles we tall HD ion.	re collected. PE tubing f	. Boring for soil											
I here	by cert	tify th	at the in	formation on t	his for	m is true and	d correct to t	he bes	st of	`my kn	owle	dge.						
Signat	ure: Jef	f Larl	kin				F	irm: E	Badg	ger Tec	hnica	l Serv	ices, I	LLC.				
This for in forfe this for	orm is au eiture of m is not	thorize betwee intend	d by Chapt en \$10 and led to be us	ters 281, 283, 289 \$25,000, or impri sed for any other p	, 291, 29 sonment ourpose.	2, 293, 295, and for up to one y NOTE: See in	nd 299, Wis. Sta year, depending structions for m	nts. Con on the ore info	nplet progi ormat	tion of th ram and tion, incl	iis forr condu uding	n is mai ct invol where t	ndatory. ved. Pe he com	Failur rsonally	e to file y identi orm she	e this fo fiable i ould be	orm may r nformatio sent.	esult n on

State Depai	of Wisco tment of	onsin f Natur <u>Rout</u>	al Resource t <u>e To:</u>	Watershed/Wastewater	Waste Manage	ement				SOII	BOR	NG LO	G INFO For	ORMA rm 440	TION 0-122
			F	Kemediation/Kedevelopment	Other							Pa	ige 1 of	1	_
Facility Badg	y/Project er Arm	Name y Am	munition	n Plant / Vapor Intrusion	License/Per	mit/Mon	itorir	ng Numb	ber	Boring VIP-0	g Numb 5	er			
Boring First N Firm:	Drilled ame:	by: Na Dav	, me of crew e scentials	v chief (first, last) and Firm Last Name: Paulson I TD	Date Drillin 2/27/12	g Started	1	Date Dri 2/27/12	illing ( 2	Complet	ed	Drilling Direct	Metho Push	d	
WI Un	ique We	ll No	DNR W	Vell ID No. Well Name	Final Static Fee	Water t MSL		Surface	Elevat Feet 1	ion MSL		Bor 2.	rehole I 125	Diamet inc	<sup>er</sup> hes
Local ( State	Grid Orig Plane	gin 54	(estii 41455	mated: ) or Boring Location N, 315689 E	Local Grid Lo	ocation		N					Е		
NW	of SV	V of	Section	25 T 10 N R 6 E		Feet		S	<u>1 m</u>	10:4	Fee	t	W		
Facili	ty ID			County	County C	ode		Civi	1100	vn/Cit	y/ or	Village	e		
1570	53930			Sauk	57			Том	n of	Prairi	ie du	Sac			
San	ple		ace)								Soi	l Proper	ties		-
Number and Type	Length Att. & Recovered (in)	Blow Counts	Depth in Feet (Below ground sur	SOIL ROCK DESCRIPTI	ON	USCS	Graphic Log	Well Diagram	PID/FID	Compressive Strength	Moisture Content	Liquid Limit	Plasticity Index	P 200	RQD/Comments
				drilled to install HDPE tubing vapor collection. B.T. @ 45' bgs.	for soil										
I here	by cer	tify th	at the in	formation on this form is true an	d correct to	the bes	st of	my kn	owle	dge.					
Signat	ure: Jef	f Lark	kin		F	Firm: E	Badg	er Tec	hnica	l Serv	ices, I	LLC.			
This for in forfe this for	orm is au eiture of m is not	thorize betwee intend	d by Chapt n \$10 and ed to be us	ters 281, 283, 289, 291, 292, 293, 295, a \$25,000, or imprisonment for up to one sed for any other purpose. NOTE: See in	nd 299, Wis. St year, depending astructions for n	ats. Cor g on the p nore info	nplet progi ormat	ion of th am and ion, incl	iis forr condu uding	n is mai ct invol <sup>i</sup> where t	ndatory ved. Po he com	r. Failur ersonally pleted f	e to file y identi orm sho	e this fo fiable i ould be	orm may result information on e sent.

State Depai	of Wisco tment of	onsin Matur Rout	al Resource te To:	es Watershed/Wastewater □	Waste Manager	nent [	7			SOII	BOR	ING LO	G INFO For	ORMA' rm 440	ГІОN 0-122	
			R	Remediation/Redevelopment X	Other							Pa	ge 1 of	1		
Facility Badge Pathv	//Project er Arm vay An	Name y Am alysis	munition	n Plant / Vapor Intrusion	License/Perm	iit/Mon	itorin	g Numł	ber	Boring VIP-'	g Numt 7	ber				
Boring First N Firm:	Drilled ame:	by: Na Dav Soil Es	me of crew e ssentials	v chief (first, last) and Firm Last Name: Paulson LTD	Date Drilling 2/28/12	Started	1 2	Date Dri 2/28/11	illing ( 2	Complet	ed	Drilling Direct-	Methoo Push	đ		
WI Un	ique We	ll No	DNR W	Vell ID No. Well Name	Final Static V Feet	Vater MSL	S	Surface	Elevat Feet I	ion MSL		Boi 2.	ehole I 125	Diamet inc	<sup>er</sup> hes	
Local C State	Grid Orig Plane	gin 54	(estii 41541	nated: ) or Boring Location N, 314767 E	Local Grid Loo	cation		N					Е			
NW	of NV	V of	fSection	36 T 10 N R 6 E		Feet		S	1 T		Fee	et	W			
Facili 15704	ty ID			County	County Co	de			1 1 OV	vn/Cit	y/ or	v illage Soc	,			
15703	53930			Sauk	57			100	10 11	r rairi	le au	Sac				
Sam	ple		face)								So	il Proper	ties	[		
Number and Type	Length Att. & Recovered (in)	Blow Counts	Depth in Feet (Below ground sur	SOIL ROCK DESCRIPTIC	N	USCS	Graphic Log	Well Diagram	PID/FID	Compressive Strength	Moisture Content	Liquid Limit	Plasticity Index	P 200	RQD/Comment	
			0-45'	No soil samples were collected. drilled to install HDPE tubing for vapor collection. B.T. @ 45' bgs.	Boring or soil											
I here	by cert	tify th	at the in	formation on this form is true and	l correct to t	he bes	t of	my kn	owle	dge.	•	~				-
Signati	ire: Jef	f Lark	cin		Fi	rm: B	adge	er Tec	hnica	l Serv	ices,	LLC.				
This for in forfe	rm is au eiture of	thorize betwee	d by Chapt on \$10 and	ters 281, 283, 289, 291, 292, 293, 295, an \$25,000, or imprisonment for up to one y ad for any other purpose. NOTE: See in	d 299, Wis. Sta year, depending	ts. Con on the p	npleti progra	on of th am and	us forr condu	n is mai ct invol <sup>1</sup> where t	ndatory ved. P	v. Failur ersonally	e to file / identi	e this fo fiable i	orm may resund nformation of sent	ılt on

State Depai	of Wisco rtment of	onsin Matura Rout	al Resource te To:	es Watershed/Wastewater □	Waste Manage	ment [	7			SOII	. BORI	NG LO	G INFO Foi	ORMA' rm 440	TION 0-122
			F	Remediation/Redevelopment X	Other							Pa	oelof	2 1	
Facility Badg	y/Project er Arm	Name y Am	munition	n Plant / Vapor Intrusion	License/Pern	nit/Mon	itoriı	ng Numb	ber	Boring VIP-8	g Numb 8	er	50101		
Boring First N Firm:	Drilled ame:	by: Na Dav Soil Es	, me of crew e ssentials	v chief (first, last) and Firm Last Name: Paulson LTD	Date Drilling 2/28/12	g Started	1	Date Dri 2/28/12	illing ( 2	Complet	ied ]	Drilling Direct-	Metho Push	d	
WI Un	ique We	ll No	DNR W	Vell ID No. Well Name	Final Static V	Water		Surface	Elevat Foot I	ion		Boi 2	ehole I	Diamet	er bes
Local C State	Grid Orig Plane	gin 54	(estin 41463	nated: ) or Boring Location N, 315099 E	Local Grid Lo	cation		I		VISL		<u></u> .	<u>Е</u>		
SW	of SV	V of	fSection	25 T 10 N R 6 E		Feet		S	1 T	<u> </u>	Fee	t	W		
Facili 1570	ty ID			County	County Co	bae			1 1 0 V m of		y/ or v	v illage Soo	;		
1570:	55950			Sauk	57	1	1	10%	10 11	Frair	le du	Sac			
San	nple		rface)								Soi	l Proper	ties		ø
Number and Type	Length Att. & Recovered (in)	Blow Counts	Depth in Feet (Below ground su	SOIL ROCK DESCRIPTIC	DN	USCS	Graphic Log	Well Diagram	PID/FID	Compressive Strength	Moisture Content	Liquid Limit	Plasticity Index	P 200	RQD/Comment
			0-45'	No soil samples were collected. drilled to install HDPE tubing for vapor collection. B.T. @ 45' bgs.	Boring or soil										
I here	by cert	tity th	at the in	formation on this form is true and	a correct to t	the bes	st of	my kn	owle	dge.					
Signat	ure: Jef	t Lark	tin	201 202 200 201 202 202 205	Final Action Barrier	irm: E	sadg	ger Tec	hnica	I Serv	ices, l	LLC.	- 4- 61	4.1.1.0	
in forfe this for	orm is au eiture of rm is not	tnorize betwee intend	a by Chapt n \$10 and ed to be us	set for any other purpose. NOTE: See ins	a 299, W1s. Sta lear, depending structions for m	tts. Cor on the port of the core of the co	nplet progi ormat	tion of th ram and tion, incl	us forr condu uding	n 1s mai ct invol where t	ndatory ved. Pe he com	. Failur ersonally pleted f	e to file / identi orm sho	e this fo fiable i ould be	nformation on sent.

State Depa	of Wisco rtment of	onsin Natura <u>Rout</u>	al Resourc <u>te To:</u>	Watershed/Wastewater	Waste Manage	ment [	]			SOII	BOR	NG LO	G INFO Foi	ORMA' rm 440	TION 0-122	
			F	Remediation/Redevelopment X	Other							Pa	ge 1 of	1		
Facilit Badg Pathy	y/Project er Arm vav An	Name y Am alvsis	munitio	n Plant / Vapor Intrusion	License/Pern	nit/Monit	torinį	g Numb	oer	Boring VIP-9	g Numb 9	er			•	
Boring First N Firm:	Drilled ame:	by: Na Dav oil Es	me of crew e ssentials	v chief (first, last) and Firm Last Name: Paulson LTD	Date Drilling 2/28/12	g Started	Г 2	Date Dri 2/28/12	illing ( 2	Complet	ted	Drilling Direct-	Metho Push	d		
WI Un	ique We	ll No	DNR W	Vell ID No. Well Name	Final Static V Feet	Water t MSL	S	urface I	Elevat Feet I	ion MSL	•	Boi 2.	ehole I 125	Diamet incl	<sup>er</sup> hes	
Local Contracts	Grid Orig Plane	gin 54	(estin 41465	mated: ) or Boring Location N, 315363 E	Local Grid Lo	cation		N					Е			
SW	of SV	V of	Section	25 T 10 N R 6 E		Feet		S			Fee	t	W			
Facili	ity ID			County	County Co	ode		Civi	l Tov	vn/Cit	y/ or `	Village	;			
1570	53930			Sauk	57			Том	n of	Prair	ie du	Sac				
San	nple		ace)								Soi	l Proper	ties	T		
Number and Type	Length Att. & Recovered (in)	Blow Counts	Depth in Feet (Below ground surf	SOIL ROCK DESCRIPTIC	DN	USCS	Graphic Log	Well Diagram	PID/FID	Compressive Strength	Moisture Content	Liquid Limit	Plasticity Index	P 200	RQD/Comments	
				drilled to install HDPE tubing for vapor collection. B.T. @ 45' bgs.	or soil											
I here	eby cert	ify th	at the in	formation on this form is true and	d correct to t	he best	ofı	my kn	owle	dge.						
Signat	ure: Jef	f Lark	kin		F	irm: Ba	adge	er Tec	hnica	l Serv	rices, l	LLC.				
This for in forfe this for	orm is au eiture of rm is not	thorize betwee intend	d by Chapt n \$10 and ed to be us	ters 281, 283, 289, 291, 292, 293, 295, an \$25,000, or imprisonment for up to one y sed for any other purpose. NOTE: See ins	nd 299, Wis. Sta year, depending structions for m	ats. Com on the p ore infor	pleti rogra mati	on of th am and on, incl	is forr condu uding	n is mai et invol where t	ndatory ved. Po he com	Failur Failur Frsonally	e to file / identi orm sho	e this fo fiable i ould be	orm may nformati sent.	result on on

State of Wisconsin

State Depai	of Wisco	onsin Matura	al Resourc	es								SOII	BORI	NG LO	G INFO	ORMA' m 440	TION 0-122	
		Rout	t <u>e To:</u> F	Watersh Remediation/	ned/Wastewa Redevelopme	ter ent X	Waste Mana Other	igement										
														Pa	ge 1 of	1		
Facility Badge	/Project er Arm	Name y Am	munition	n Plant / V	apor Intru	sion	License/P	ermit/Mc	nitori	ng Num	ber	Boring VIP-	g Numb 10	er				
Boring First N	Drilled ame:	by: Na Dav	, me of crew e	v chief (first, Last Nai	last) and Firr me: Pau	n lson	Date Drill 2/28/12	ing Start	ed	Date Dr 2/28/1	rilling ( 2	Complet	ted I	Drilling Direct-	Metho Push	d		
Firm: WI Un	S ique We	boil Es 11 No	ssentials DNR W	LTD Vell ID No.	Well Na	ame	Final Stat	ic Water	-	Surface	Elevat	ion		Bo	rehole I	Diamet	er	
Local C State	Grid Orig Plane	gin 54	(estin 41565	mated: ) o N.	or Boring Lo 316021	cation E	Local Grid	Location	-	N	Feet	MSL		<u></u>	125 F	1nc	nes	
SW	of NV	V of	f Section	26 T	10 N	R 6 1	Е	Feet		S			Fee	t	W			
Facili	ty ID			County			County	Code		Civ	il Tov	vn/Cit	y/ or V	Village	<b>;</b>			
15705	53930		Γ	Sauk			57	1		Tov	vn of	Prair	ie du S	Sac			1	
Sam	ple		ace)										Soil	l Proper	ties	1	-	
Number and Type	Length Att. & Recovered (in)	Blow Counts	Depth in Feet (Below ground surf		SOIL ROCK	X DESCRIPT	ION	USCS	Graphic Log	Well Diagram	PID/FID	Compressive Strength	Moisture Content	Liquid Limit	Plasticity Index	P 200	RQD/Comments	
			0-45'	No soil sa drilled to vapor col B.T. @ 4	amples we install HE llection.	re collected DPE tubing	d. Boring for soil											
I here	by cert	tify th	at the in	formation	on this for	m is true a	nd correct t	o the be	est of	my kr	nowle	dge.						
Signati	are: Jef	f Lark	kin					Firm:	Badg	ger Teo	chnica	l Serv	ices, I	LLC.				
This for in forfe this for	rm is au eiture of m is not	thorize betwee intend	d by Chapt n \$10 and ed to be us	ters 281, 283 \$25,000, or i sed for any ot	, 289, 291, 29 imprisonment her purpose.	92, 293, 295, 5 t for up to one NOTE: See i	and 299, Wis. e year, depend nstructions for	Stats. Co ing on the r more in	omple e prog forma	tion of t ram and tion, inc	his fori condu luding	n is mai ct invol where t	ndatory. ved. Pe he com	. Failur ersonally pleted f	e to file y identi orm sho	e this fo fiable i ould be	orm may nformations sent.	result on on

State of Wisconsin Department of Natural Resources		Weste Mensee		_			SOII	BOR	ING LO	G INFC For	0RMA] m 4400	ГІОN )-122	
<u>Route To:</u> Watersned/Waster Remediation/Redevelop	oment X	Other							Do	and of	1		
Facility/Project Name		License/Perm	uit/Mon	itorin	a Numł	ver	Boring	Numl	Fa Der	ge 1 01			
Badger Army Ammunition Plant / Vapor Int	rusion	License/1 enin	11/1/1011	norm	g Nullic		VIP-	lA					
Pathway Analysis													
Boring Drilled by: Name of crew chief (first, last) and	Firm	Date Drilling	Started	I I	Date Dr	illing (	Complet	ed	Drilling	Method	l		
First Name: Dave Last Name: P	aulson	//3/12			//3/12				Direct-	Push			
WI Unique Well No DNR Well ID No. Wel	Name	Final Static V	Vater	5	Surface	Elevat	ion		Bo	ehole I	Diamete	er	
Local Grid Origin (estimated: ) or Boring	Location	Local Grid Loc	MSL cation			Feet I	MSL		<u>2</u> .	125	_1ncl	nes	
State Plane         540043         N,         319020           NE         of         SW         of Section         14         T         10         N	E IR 6 E	1	Feet		N S			Fee	•t	E W			
Facility ID County		County Co	de		Civi	1 Tov	vn/Cit	v/or	Village	;			
157053930 Sauk		57			Том	n of	Sumn	fer	0				
Jan Sauk		57		1	100	II UI	Sump						
Sample								So	il Proper	ties			
					в		0			lex		lents	
and SOIL KC	OCK DESCRIPTIO	VIN		Log	agra	~	ssive	n)	imi	y Inc		uuu	
e e overingth, w Co over			S	phic	1 Di	/FIL	npre ngth	stur	I biu	ticit	00	D/C	
Nur Typ Blo Dep Bel-			NSC	Gra	Wel	PID	Con Stre	Con Con	Liqu	Plas	P 2(	RQI	
Mumber       No soil samples were collected. Boring drilled to install HDPE tubing for soil vapor collection.       No soil samples were collected. Boring drilled to install HDPE tubing for soil       No       No <td></td> <td></td>													
I hereby certify that the information on this Signature: Brenda Boyce This form is authorized by Chapters 281, 283, 289, 291	form is true and , 292, 293, 295, an	d correct to tl Fi d 299, Wis. Sta	he bes rm: B ts. Cor	tt of Badg	my kn er Tec	owlee hnica	dge. 1 Serv	ices,	LLC. 7. Failur	e to file	this fo	rm may	result
in forfeiture of between \$10 and \$25,000, or imprisonn this form is not intended to be used for any other purpo	nent for up to one y se. NOTE: See ins	rear, depending	on the j	progr	am and	condue uding	t invol where f	ved. P	ersonally	v identii	fiable in ould be	nformati sent	on on

State of Wisconsin

Depar	rtment of	f Natur Rout	al Resourc	es Waterched/Wastewater	Waste Manage	ment [	-						For	m 440	0-122	
		Rout	<u>e 10.</u> R	temediation/Redevelopment	Other							<b>D</b> -	1 -6	1		
Facility	/Project	Name			License/Pern	nit/Mor	itorin	9 Numł	ber	Boring	, Numb	Pa er	ge I of		-	
Badg	er Arm	y Am	munition	n Plant / Vapor Intrusion				6		VIP-4	4A	••				
Pathw Boring	vay An Drilled	alysis bv: Na	me of crew	chief (first, last) and Firm	Date Drilling	y Started	1 1	Date Dr	illing (	Complet	ed I	Drilling	Metho	1		_
First N	ame:	Dan		Last Name: Bendorf	6/27/12	,	(	5/27/1	2		1	Direct-	Push			
Firm: WI Un	P ique We	robe ll No	Technolo DNR W	ogies, Inc. /ell ID No. Well Name	Final Static V	Water	5	Surface	Elevat	ion		Bor	ehole I	Diamet	er	_
<u> </u>	1				Feet	t MSL	_	]	Feet I	MSL <u>2.125</u> inches						
State 1	Plane	gin 54	(estii 40766	N, 315809 E	Local Grid Lo	cation		Ν		Е						
SW	of NI	E of	Section	26 T 10 N R 6 E		Feet		S	1 7	10.1	Fee	t	W			_
Facili	ty ID			County	County Co	County Code     Civil Town/City/ or Village       57     Town of Prairie du Sac										
15703	53950			Sauk	57	1		100		r rair	ie au s	Sac				
Sam	nple		tce)								Soil	l Proper	ties			
	& in)	s	ət d surfa	SOIL BOCK DESCRIPTIO	NI		50	ш		a		t t	dex		ients	
er and	Att. ered (	Count	in Fe groun	SOIL ROCK DESCRIPTIC	/1		c Log	iagra	D	essiv th	re lt	Limi	ity In		Jomn	
lumbe ype	ength	low (	epth 3elow			ISCS	iraphi	Vell D	ID/FI	ompr treng	foistu	iquid	lastic	200	QD/(Q	
ΖF	ЦК	щ	1 = 0.37	No soil samples were collected	Boring		0	Λ	Р	s C	N O		Р	P	24	
			0-37	drilled to install HDPE tubing for	or soil											
				vapor collection.												
				B.T. @ 37' bgs.												
				2.11 (6.07) (5.0												
I here	by cert	tify th	at the int	formation on this form is true and	l correct to t	he bes	stof	my kn	owle	dge.						
Signatu	ure: Bre	nda Bo	yce		F	irm: E	Badg	er Tec	hnica	l Serv	ices, I	LLC.				
This fo	orm is au	thorize	d by Chapt	ters 281, 283, 289, 291, 292, 293, 295, an \$25,000, or imprisonment for up to one x	d 299, Wis. Sta	ats. Con	npleti	on of th	is form	n is mai	ndatory	Failur	e to file	this fo	orm may result	t
this for	m is not	intend	ed to be us	ed for any other purpose. NOTE: See ins	structions for m	ore info	ormati	on, incl	uding	where t	he com	pleted for	orm sho	ould be	sent.	

SOIL BORING LOG INFORMATION Form 4400-122

State Depai	State of Wisconsin SOIL BORING LOG INFORMATION Department of Natural Resources Form 4400-122															
		Rout	<u>e To:</u> R	Watershed/Wastewater	Waste Manager Other	ment										
			1		other							Pa	ge 1 of	1	-	
Facility	y/Project	Name	munitio	n Plant / Vapor Intrusion	License/Perm	nit/Mon	itori	ng Numł	ber	Boring	g Numl 5 A	ber				
Pathy	vay An	alysis	iniunitioi ;	n Flant / Vapor Inclusion						V 1F	JA					
Boring	Drilled	by: Na	me of crew	v chief (first, last) and Firm	Date Drilling	Started	1	Date Dri	illing (	Complet	ed	Drilling	Metho	1		
Firm:	anne. P	Dan Probe	Technolo	ogies. Inc.	0/2//12			0/2//1	Ζ			Direct	-Pusn			
WI Un	ique We	ll No	DNR W	Vell ID No. Well Name	Final Static V Feet	Vater MSL		Surface	Elevat Feet I	ion MSL		Boi 2	rehole I 125	Diamet	er hes	
Local (	Grid Orig	gin _	(estin	mated: ) or Boring Location	Local Grid Lo	cation							lies			
State SW	Plane of NI	54 E of	1097 Section	N, 315811 E 26 T 10 N R 6 E	,	Feet		N S	N E S Feet W							
Facili	ty ID			County	County Co	ode		Civi	l Tov	vn/Cit	y/or	Village	;			
15705	53930			Sauk	57			Tow	n of	Prair	ie du	Sac				
Sam	nple		(c)								So	il Proper	ties			
			surfac					_					×		nts	
and	Att. & ed (ir	ounts	Feet ound	SOIL ROCK DESCRIPTIO	N		Log	ıgran	_	ssive		imit	/ Inde		mme	
mber oe	ngth /	w C	pth in low g			cs	uphic	II Dia	/FID	mpre: ength	isture ntent	uid I	sticit	00	D/Cc	
Nu Tyj	Ler Rec	Blc	De <sub>l</sub> (Be			SN	Gra	We	ЫI	Co1 Stre	Mo Coi	Liq	Pla	P 2	RQ	
			0-40'	No soil samples were collected.	Boring or soil											
				vapor collection.	5011											
				B.T. @ 40' bgs.												
I here	by cert	tify th	at the int	formation on this form is true and	l correct to t	he bes	st of	my kn	owle	dge.						
Signati	ure: Bre	nda Bo	усе		Fi	rm: E	Badg	ger Tec	hnica	l Serv	ices,	LLC.				
This fo	orm is au	thorize	d by Chapt in \$10 and	ters 281, 283, 289, 291, 292, 293, 295, and \$25,000, or imprisonment for up to one v	d 299, Wis. Sta ear. denending	ts. Cor	nple	tion of th ram and	is form	n is mai ct invol	ndatory	y. Failur ersonally	e to file	this fo	orm may nformati	result
this for	m is not	intend	ed to be us	sed for any other purpose. NOTE: See ins	tructions for m	ore info	orma	tion, incl	uding	where t	he con	npleted f	orm sho	ould be	sent.	

Route To: Watershed/Wastewater	Waste Managen Other	nent									
		P;							ge 1 of	1	
Facility/Project Name Badger Army Ammunition Plant / Vapor Intrusion Pathway Analysis	License/Permi	it/Monit	oring	g Numbo	er	Boring VIP-7	g Numl 7A	ber			
Boring Drilled by: Name of crew chief (first, last) and Firm First Name: Dan Last Name: Bendorf Firm: Probe Technologies Inc	Date Drilling 6/27/12	Started	D 6	ate Dril /27/12	lling C	Complet	ed	Drilling Direct-	Methoo Push	1	
WI Unique Well No DNR Well ID No. Well Name	Final Static W	ater	S	urface E	Elevati	on		Bo	ehole I	Diamete	er
Local Grid Origin (estimated: ) or Boring Location	Local Grid Loca	MSL ation		F	eet N	ASL		<u>2</u> .	125	incl	nes
State Plane 541541 N, 314767 E NW of NW of Section 36 T 10 N R 6 F	Б	N E									
Facility ID County	County Cod	de		Civil	Том	/n/Cit	y/or	n Village	••• ;		
157053930 Sauk	57 Town of Prairie du Sac										
Sample 3							So	il Proper	ties		
Number and Type and Length Att. & Recovered (in) Depth in Feet (in) Blow Counts (Below ground surfa	N	USCS Graddio Lore	Utaptile Log	Well Diagram	PID/FID	Compressive Strength	Moisture Content	Liquid Limit	Plasticity Index	P 200	RQD/Comments
Joannader mader     Characteristic problem     No soil samples were collected. Boring drilled to install HDPE tubing for soil vapor collection.     SSI 1     I     I     I     I     I       B.T. @ 37' bgs.     B.T. @ 37' bgs.     I     I     I     I     I     I     I     I											
I hereby certify that the information on this form is true and	l correct to th	ne best	ofr	ny kno	owled	lge.	•				
Signature: Brenda Boyce	Fir	m: Ba	idge	r Tech	nnica	I Serv	ices,	LLC.	a ta El-	this f-	m may samlt
in forfeiture of between \$10 and \$25,000, or imprisonment for up to one y this form is not intended to be used for any other purpose. NOTE: See ins	ear, depending of tructions for mo	on the prover on the province information of the province of t	ogra matic	m and c	s torm conduc iding v	t involvent	ved. P	ersonally	identi orm sho	fiable in fiable in	nformation on sent.

SOIL BORING LOG INFORMATION Form 4400-122

State Depar	of Wisco tment of	onsin f Natur <u>Rout</u>	al Resource te To:	Watershed/Wastewater	Waste Manage	ment				SOII	BOR	ING LO	G INFO For	ORMA' m 440	ГІОN 0-122	
			ľ	Remediation/Redevelopment	Other	Ш_						Pa	ge 1 of	1		
Facility Badge Pathw	//Project er Arm vay An	Name Name Ny Am alysis	munition	n Plant / Vapor Intrusion	License/Perr	nit/Monite	oring	g Numb	ber	Boring VIP-9	g Numb 9A	ber				
Boring First N Firm:	Drilled ame:	by: Na Dan Probe	me of crew Technolo	v chief (first, last) and Firm Last Name: Bendorf ogies	Date Drilling 6/27/12	g Started	D 6.	ate Dri /27/12	illing ( 2	Complet	ied	Drilling Direct-	Methoo Push	1		
WI Un	ique We	ll No	DNR W	Vell ID No. Well Name	Final Static Fee	Water t MSL	S	urface I	Elevat Feet l	ion MSL		Boi 2.	ehole I 125	Diameto incl	er hes	
Local C State I	Grid Orig Plane	gin 54	(estii 41465	mated: ) or Boring Location N, 315363 E	Local Grid Lo	ocation		N					Е			
SW	of SV	V of	Section	25 T 10 N R 6 E		Feet		S			Fee	et	W			
Facili	ty ID			County	County Co	ode		Civil Town/City/ or Village								
15705	53930			Sauk	57			Tow	Fown of Prairie du Sac							
Sam	ple		ace)								Soi	il Proper	ties			
Number and Type	Length Att. & Recovered (in)	Blow Counts	Depth in Feet (Below ground surf	SOIL ROCK DESCRIPTI	ON	USCS Granhie Lor	Source tog	Well Diagram	PID/FID	Compressive Strength	Moisture Content	Liquid Limit	Plasticity Index	P 200	RQD/Comments	
			0-39'	No soil samples were collected drilled to install HDPE tubing vapor collection. B.T. @ 39' bgs.	l. Boring for soil											
I here	by cer	tify th	at the in	formation on this form is true ar	nd correct to t	the best	ofn	ny kn	owle	dge.						
Signatu	ire: Bre	nda Bo	yce		F	irm: Ba	dge	r Tec	hnica	l Serv	ices,	LLC.				
This fo in forfe this for	rm is au eiture of m is not	thorize betwee intend	d by Chapt n \$10 and ed to be us	ters 281, 283, 289, 291, 292, 293, 295, a \$25,000, or imprisonment for up to one sed for any other purpose. NOTE: See in	nd 299, Wis. Sta year, depending astructions for m	ats. Comp g on the pr nore inform	oletic ogra natic	on of th m and on, incl	is forr condu uding	n is maı ct invol where t	ndatory ved. P he com	7. Failur ersonally pleted f	e to file / identi orm sho	e this fo fiable i ould be	orm may ro nformatio sent.	esult n on

State Depar	State of Wisconsin     SOIL BORING LOG INFORMATION       Department of Natural Resources     Form 4400-122															
		<u>Rout</u>	<u>e To:</u> R	Watershed/Wastewater	Waste Manager Other	ment						Pa	ge 1 of	· 1		
Facility	/Project	Name			License/Perm	nit/Mor	itorii	ng Numl	ber	Boring	y Numl	ber	-			
Badg	er Årm	y Am	munition	n Plant / Vapor Intrusion				0		VIP-	10A					
Pathw	vay An	alysis					<u> </u>									
Boring	Drilled	by: Nai	me of crew	v chief (first, last) and Firm	Date Drilling	; Started	1	Date Dr	illing (	Complet	ed	Drilling	Metho	d		
First N	ame:	Dav	e	Last Name: Paulson	7/3/12			//3/12				Direct-	Push			
WI Un	ique We	011 ES	DNR W	LID Vell ID No. Well Name	Final Static V	Vater		Surface	Flevat	ion		Bot	ehole I	Diamet	er	
WI OII	ique ire		DIAR	ven ib ivo. Wen ivane	Feet	MSL		]	Feet I	MSL		<u>2.125</u> inches				
Local (	Grid Orig	gin -	(estin	mated: ) or Boring Location	Local Grid Loc	cation										
State I	Plane	54 V of	F1565 Section	N, 316021 E		<b>F</b> 4		N		E E						
Facili		v 01	Section		County Co	reet		S Feet W								
15706				County		Jue			1 100		y/ 01	v mage	, ,			
15/03	53930			Sauk	5/	1	1	100	n or	Prair	le du	Sac			1	
Sam	ple		ace)								So	il Proper	ties	I		
	ж п)		:t I surf					н					lex		ents	
and	Att. e ed (i	ounts	Fee	SOIL ROCK DESCRIPTIC	)N		Log	ıgraı	_	ssive	0	imit	/ Inc		uuu	
aber e	gth ∕ over	v Cc	th in w gr			Ņ	ohic	Dia	FID	ipres ngth	sture tent	id L	ticity	0	0/Cc	
Nun Type	Leng Reco	Blov	Depi (Beld			USC	Grap	Wel	PID	Con Stre	Moi Con	Liqu	Plas	P 20	RQI	
			0-45'	No soil samples were collected	Boring											
			0 45	drilled to install HDPE tubing for	or soil											
				vapor collection.												
				-												
				B T @ 45' bos												
				D.1. @ 10 0gs.												
I here	by cert	ify th	at the int	formation on this form is true and	d correct to t	he bes	st of	`my kn	owle	dge.						
Signat	ire: Bre	enda F	Bovce		Fi	irm F	- Bado	per Tec	hnica	l Serv	ices	LLC				
This fo	m is ar	thorize	d by Chart	tere 281 283 280 201 202 202 205	1 200 Win Sta	te Ca	mpla	tion of $t^{1}$	in form		idatar	LLC.	e to file	this fo	mm mott	regult
in forfe	iture of	betwee	n \$10 and	\$25,000, or imprisonment for up to one y	ear, depending	on the	prog	ram and	condu	ct invol	ved. P	ersonally	v identi	fiable i	nformati	on on
this for	m is not	intend	ed to be us	ed for any other purpose. NOTE: See ins	structions for m	ore info	orma	tion, incl	uding	where t	he con	pleted for	orm sho	ould be	sent.	

Well / Drillhole / Borehole Abandonment Forms

Douto to:

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

I. General Information       2. Facility / Owner Information         WI Unique Well No.       DNR Well ID No.       County       Facility Name         Sauk       Badger Army Ammunition Plant       Badger Army Ammunition Plant         Common Well Name       Gov't Lot # (if applicable)       Facility ID       License/Permit/Monitoring No.         VIP-1       Ya / Ya       Section       Township       Range       E       Street Address of Well         NE / SW       14       10       N       6       W       Street Address of Well
WI Unique Well No.       DNR Well ID No.       County Sauk       Facility Name Badger Army Ammunition Plant         Common Well Name VIP-1       Gov't Lot # (if applicable)       Facility ID 157053930       License/Permit/Monitoring No.         ½ / ¼       ¼       Section 14       Township 10       Range 6       E       Street Address of Well 2 Badger Road
Sauk     Badger Army Ammunition Plant       Common Well Name     Gov't Lot # (if applicable)     Facility ID     License/Permit/Monitoring No.       VIP-1     14     Township     Range     E     Street Address of Well       NE / SW     14     10     N     6     W     Street Address of Well
Common Well Name     Gov't Lot # (if applicable)     Facility ID     License/Permit/Monitoring No.       VIP-1     VIP-1     157053930     157053930       V// ¼     ¼     Section     Township     Range     E       NE / SW     14     10     N     6     W
VIP-1         157053930           ½ / ¼         ¼         Section         Township         Range         E         Street Address of Well           NE / SW         14         10         N         6         W         Street Address of Well
1/4     1/4     Section     Township     Range     Image
NE / SW 14 10 N 6 2 Badger Road
Well Location ft. / M (Local Grid ) Datum
540043 NV S 319020 EV W NAD 83 Present Well Owner Original Well Owner
WTM-
Local Grid Origin ft. / M Datum 2 Badger Road
N,E / WCity State ZIP Code
WIM- UIM- Latitude/Longitude- State Plane- STC N Attubing Woll New York Wall New York
Reason For Abandonment Wi Unique Well No. of Replacement Well
3. Well / Drillinole / Borenole Information
Monitoring Well
Water Well
Borehole / Drillhole please attach.
Construction Type:
Did material settle after 24 hours?
If bentonite chips were used, were they hydrated
Other (specify): <u>Geophone</u> with water from a known safe source? □Yes □No □N
Formation Type: Required Method of Placing Sealing Material
Unconsolidated Formation
Total Well Depth From Groundsurface (ft.) Casing Diameter (in.)
45 Sealing Materials
Lower Drillhole Diameter (in.) Casing Depth (ft.)
2.125 Sand-Cement (Concrete) Grout Bentonite-Sand Slurry "
Was well annular space grouted?
If yes, to what depth (feet)? Depth to Water (feet) Bentonite Chips Bentonite - Cement Grout
Granular Bentonite
5. Material Used To Fill Well / Drillhole From (ft.) To (ft.) No. Yards, Sacks Sealant Mix Ratio or
Chipped Bentonite Surface 45 40 lbs
6. Comments

7. Supervision of Work					DNR Use Only				
Name of Person or Firm Doing Sealing Work			Date of Abar	ndonment	Date Received	Noted By			
Dave Paulson			2/27/12						
Street or Route			Telephone N	lumber	Comments				
W6306 State Road 39			(608) 527	-2355					
City	State	ZIP Code	Э	Signature of Perso	on Doing Work		Date Signed		
New Glarus	WI	53574					4/19/12		

Douto to:

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Drinking Water Watershe	ed/Wastewater	ste Management	🖌 Remedi	ation/Redeve	lopment	Other:		
1. General Information			2. Facility	/ Owner In	formation			
WI Unique Well No. DNR We	ell ID No. County		Facility Nam	е				
	Sauk		Badger Ar	my Ammunitio	n Plant			
Common Well Name	Gov't Lot # (if a	pplicable)	Facility ID	_	Lice	nse/Permit/Mon	itoring No.	
VIP-2		_	15705393	0				
1/4 / 1/4 Sect	ion Township I	Range I E	Street Addre	ess of Well				
SW / NE 14	10 N	6 🗌 W	Z Badger Ros	ad or Town				
Well Location ft. / M (Local C	Grid 🔲 )	Datum	Barahoo	OF TOWN				
540372 N S 3	17498 <b>(E)</b>	W NAD 83	Present Wel	l Owner		Original Well C	)wner	
			U.S. Army					
	ongitudeState Plane		Street Addre	ess or Route	of Present O	wner		
Eccar Ghu Ongin ft. / M		Datum	2 Badger R	oad				
N,	E	/W	City			State	ZIP Code	
WTM- UTM- Latitude/Lo	ongitude- State Plane		Baraboo			WI	53913	
Reason For Abandonment	WI Unique Well No. of	f Replacement We	4. Pump, I	iner, Scree	en, Casing a	& Sealing Mat	erial	
Temp. Borehole-Vapor Collection			Pump and	d piping remo	oved?	Г	] <sub>Yes</sub> □ <sub>No</sub>	Ŀ N/A
3. Well / Drillhole / Borehole I	Information		Liner(s) re	emoved?			] <sub>Yes</sub> □ <sub>No</sub>	۲ II N/A
	Driginal Construction Dat	e	Screen re	moved?			] <sub>Yes</sub> □ <sub>No</sub>	. 🗹 <sub>N/A</sub>
Monitoring Well	2/27/12		Casing le	ft in place?			] <sub>Yes</sub> D <sub>No</sub>	⊡ <sub>N/A</sub>
Water Well	If a Well Construction Re	port is available,	Was casi	ng cut off belo	ow surface?		J <sub>Yes</sub> □ <sub>No</sub>	. ⊡ <sub>N/A</sub>
✓ Borehole / Drillhole	please attach.		Did sealir	g material ris	se to surface	?	′] <sub>Yes</sub> □ <sub>No</sub>	. 🗆 <sub>N/A</sub>
Construction Type:	_		Did mater	ial settle afte	r 24 hours?	Г	] <sub>Yes</sub> ⊡ <sub>No</sub>	□ □ <sub>N/A</sub>
Drilled Driven (S	Sandpoint)	ug	If yes	, was hole ret	topped?		] <sub>Yes</sub> □ <sub>No</sub>	⊡ <sub>N/A</sub>
✓ Other (specify): Geoprobe			If bentonit with water	e chips were from a know	used, were th n safe source	hey hydrated		
Formation Type:			Required Me	thod of Placir	ng Sealing Ma	aterial		
	Bedrock		Condu	ctor Pipe-Gra	vity 🔲 Cor	ductor Pipe-Pun	nped	
Total Well Depth From Groundsurf	face (ft.) Casing Diame	ter (in.)	- C Screer	ed & Poured	C Oth	er (Explain): <u>Pou</u>	red / chipped bent	onite
25		( )	Sealing Mate	erials				
Lower Drillhole Diameter (in.)	Casing Depth	(ft.)	Neat C	ement Grout		🗌 Clay-Sa	and Slurry (11 l	b./gal. wt.)
2.125			Sand-C	Cement (Conc	crete) Grout	Bentoni	te-Sand Slurry	
Was well appular space grouted?	Ves VN			ete		🖌 Bentoni	te Chips	
If yes, to what depth (feet)?	Depth to Water (fee	t)	For Monitorii	ng Wells and I	Monitoring W	'ell Boreholes Or I	nly:	
in yes, to what depth (leet):		()		ite Chips		Bentonite - Cer	ment Grout	
			Granul	ar Bentonite		Bentonite - Sar		tio or
5. Material Used To Fill Well / D	rillhole		From (ft.)	To (ft.)	or Volun	ne (circle one)	Mud W	eight
Chipped Bentonite			Surface	25	25 lbs			
6. Comments								

7. Supervision of Work					DNR Use Only				
Name of Person or Firm Doing Sealing Work			Date of Abar	ndonment	Date Received	Noted By			
Dave Paulson			2/27/12						
Street or Route			Telephone N	lumber	Comments				
W6306 State Road 39			(608) 527	-2355					
City	State	ZIP Code	Э	Signature of Perso	on Doing Work		Date Signed		
New Glarus	WI	53574					4/19/12		

Douto to:

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Drinking Water Watershe	ed/Wastewater	aste Management	🖌 Remedi	ation/Redeve		Other:		
1. General Information			2. Facility	/ Owner In	formation			
WI Unique Well No. DNR We	ell ID No. County		Facility Nam	е				
	Sauk		Badger Ar	my Ammunitio	n Plant			
Common Well Name	Gov't Lot # (if a	applicable)	Facility ID	_	Lice	nse/Permit/Mon	itoring No.	
VIP-3		L	15705393	0				
<sup>1</sup> / <sub>4</sub> / <sup>1</sup> / <sub>4</sub> Sect	ion lownship	Range 🖌 E	Street Addre	ess of Well				
SE / NW 26	10 N	6 🗌 W	Z Badger Ro	ad or Town				
Well Location ft. / M (Local C	Grid 🔲 )	Datum	Barahoo					
540322 N S 3	15818 <b>E</b>	W NAD 83	Present Wel	l Owner		Original Well C	Owner	
		Zone	U.S. Army					
	ongitudeState Plar		Street Addre	ess or Route	of Present O	wner		
Local Glid Oligin [ft.]/M		Datum	2 Badger R	oad				
N,	E	/W	City			State	ZIP Code	
WTM- UTM- Latitude/Lo	ongitude- State Plan		Baraboo			WI	53913	
Reason For Abandonment	WI Unique Well No. d	of Replacement W	4. Pump, I	_iner, Scree	en, Casing	& Sealing Mat	terial	
Temp. Borehole-Vapor Collection		·	Pump and	d piping remo	oved?	Ľ	]Yes □Nc	, <b>∠</b> <sub>N/A</sub>
3. Well / Drillhole / Borehole I	Information		Liner(s) r	emoved?		Ľ	∃ <sub>Yes</sub> □ <sub>Nc</sub>	,
	Driginal Construction Da	ate	Screen re	moved?			] <sub>Yes</sub> □ <sub>Nc</sub>	, <b>∠</b> <sub>N/A</sub>
Monitoring Well	2/27/12		Casing le	ft in place?			<u>]<sub>Yes</sub> D<sub>Nc</sub></u>	<u>, ⊡<sub>N/A</sub></u>
Water Well	If a Well Construction R	eport is available,	Was casi	ng cut off belo	ow surface?			. ⊡ <sub>N/A</sub>
✓ Borehole / Drillhole	please attach.		Did sealir	ng material ris	se to surface	? •	⊴ <sub>Yes</sub> □ <sub>Nc</sub>	, □ <sub>N/A</sub>
Construction Type:	_		Did mater	ial settle afte	r 24 hours?	Ľ	∃ <sub>Yes</sub> ∠ <sub>Nc</sub>	, □ <sub>N/A</sub>
Drilled Driven (S	Sandpoint)	Dug	If yes	, was hole ret	topped?	E	∃ <sub>Yes</sub> □ <sub>Nc</sub>	, ビ <sub>N/A</sub>
✓ Other (specify): Geoprobe			If bentonit with water	e chips were from a know	used, were the source	hey hydrated		,
Formation Type:			Required Me	thod of Placir	ng Sealing M	aterial		
	Bedrock		Condu	ctor Pipe-Gra	vity 🔲 Cor	nductor Pipe-Pur	mped	
Total Well Depth From Groundsurf	face (ft.) Casing Diame	eter (in.)	- Screer	ned & Poured	🖸 <sub>Oth</sub>	er (Explain): <u>Pou</u>	ured / chipped bent	ionite
25		( )	Sealing Mate	erials				
Lower Drillhole Diameter (in.)	Casing Depth	ı (ft.)	Neat C	ement Grout		Clay-Sa	and Slurry (11 I	b./gal. wt.)
2.125			Sand-(	Cement (Conc	crete) Grout	Bentoni	ite-Sand Slurry	,
Was well annular space grouted?				ete		<ul> <li>Bentoni</li> </ul>	ite Chips	
If ves, to what denth (feet)?	Depth to Water (fee	et)	For Monitorii	ng Wells and I	Monitoring W	/ell Boreholes Oı T	nly:	
				nite Chips		Bentonite - Ce	ment Grout	
			Granul	ar Bentonite	No Varde	Bentonite - Sai	nd Slurry	tio or
5. Material Used To Fill Well / D	rillhole		From (ft.)	To (ft.)	or Volur	ne (circle one)	Mud W	/eight
Chipped Bentonite			Surface	25	25 lbs		_	
6. Comments								

7. Supervision of Work	DNR Use Only						
Name of Person or Firm Doing Sealing Work			Date of Abandonment		Date Received	Noted By	
Dave Paulson			2/27/12				
Street or Route			Telephone N	lumber	Comments		
W6306 State Road 39			(608) 527	-2355			
City	State	ZIP Code	Э	Signature of Perso	on Doing Work		Date Signed
New Glarus	WI	53574					4/19/12

Douto to:

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Drinking Water Watershed/	Wastewater 🗌 Waste Managemer	t 🖌 Remed	iation/Redeve	lopment Ot	:her:	
1. General Information		2. Facility	/ / Owner In	formation		
WI Unique Well No. DNR Well II	D No. County	Facility Nan	ne			
	Sauk	Badger A	rmy Ammunitio	n Plant		
Common Well Name	Gov't Lot # (if applicable)	Facility ID		License/I	Permit/Monitori	ing No.
VIP-4		15705393	30			
1/4 / 1/4 Section	Township Range	E Street Addr	ess of Well			
SW / NE 26	10 N 6	W 2 Badger Ro	ad Taxaa			
Well Location ft. / M (Local Grid	Datum	City, Village	or Iown			
540766 (N) (S) 3158	09 (E) W NAD 83	Baraboo Present We		Ori	ninal Well Own	or
	Zone				Jinar Weil Own	
WTM-	itude– State Plane– 🖌 (S) C	N Street Addr	ess or Route	of Present Owner		
Local Grid Origin ft. / M	Datum	2 Badger F	load			
N,	E / W	City			State Z	ZIP Code
		Baraboo			WI	53913
	ItudeState PlaneSC	A Pump,	Liner, Scree	en, Casing & Se	aling Materia	al
Temp Borebole-Vapor Collection		Pump an	d piping rome			
2 Well / Drillhole / Perchala Inf			omovod?	veu:		
	anal Construction Date		emoved?			
Monitoring Well	7/12	Casing le	oft in place?			
Water Well	Well Construction Report is available					
Borehole / Drillhole ple	ase attach.	, was cas	ng cut on bei	ow surface?		
Construction Type:		Did seall	ng material ris	se to surface?		
Drilled Driven (San	dpoint)	Did male If yes	was hole rei	topped?		
Cthor (specify): Geoprobe		If bentoni	te chips were	used, were they h	ydrated	
		<ul> <li>with wate</li> </ul>	r from a know	n safe source?		es 니No ビN/A
Formation Type:	—		ettor Dine Cro			
Unconsolidated Formation	Bedrock		ned & Poured		or Pipe-Pumpe	;U / chinned bentonite
Total Well Depth From Groundsurface	e (ft.) Casing Diameter (in.)	(Bento	nite Chips)	U Other (E	xplain): <u>routeu</u>	
		Sealing Mat	erials			
Lower Drillhole Diameter (in.)	Casing Depth (ft.)		Cement Grout		Clay-Sand	Slurry (11 lb./gal. wt.
2.120			Cement (Cond	crete) Grout	Bentonite-S	Sand Slurry
Was well annular space grouted?	Yes 🗹 No 📙 Unkno	wn For Monitori	ete Ing Wells and	Monitoring Well B	Coreboles Only:	Jubs
If yes, to what depth (feet)?	Depth to Water (feet)		nite Chips		ntonite - Cemer	nt Grout
		Granu	lar Bentonite		ntonite - Sand S	Slurry
5. Material Used To Fill Well / Drill	hole	From (ft.)	To (ft.)	No. Yards, Sac	ks Sealant	Mix Ratio or Mud Weight
Chipped Bentonite		Surface	25	25 lbs		
			1		†	
6. Comments						

7. Supervision of Work	DNR Use Only						
Name of Person or Firm Doing Sealing Work			Date of Abandonment		Date Received	Noted By	
Dave Paulson			2/27/12				
Street or Route			Telephone N	lumber	Comments		
W6306 State Road 39			(608) 527	-2355			
City	State	ZIP Code	Э	Signature of Perso	on Doing Work		Date Signed
New Glarus	WI	53574					4/19/12

Douto to:

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Drinking Water Watershed	/Wastewater 🗌 Was	ste Management	🖌 Remedi	ation/Redeve		Other:		
1. General Information			2. Facility	/ Owner In	formation			
WI Unique Well No. DNR Well	ID No. County		Facility Nam	е				
	Sauk		Badger Ar	my Ammunitio	n Plant			
Common Well Name	Gov't Lot # (if a	pplicable)	Facility ID		Lice	ense/Permit/Mon	itoring No.	
VIP-5			15705393	0				
1/4 / 1/4 Section	n Township I	Range 🗹 E	Street Addre	ess of Well				
SE / NE 26	10 N	6 🗌 W	2 Badger Ros	ad				
Well Location ft. / M (Local Gr	id 🔲 )	Datum	City, Village	or Iown				
541097 (N) (S) 315	811	W NAD 83	Baraboo Present Wel				Jwner	
		Zone	U.S. Army	Owner		Oliginal Well C	JWIIEI	
WTM-	gitude- State Plane		Street Addre	ess or Route (	of Present O	l )wner		
Local Grid Origin ft. / M		Datum	2 Badger R	oad				
N,	E	/ W	City			State	ZIP Code	
			Baraboo			WI	53913	
	gitude- State Plan		4. Pump, I	_iner, Scree	en, Casing	& Sealing Mat	terial	
Temp Borebole-Vapor Collection		Replacement we	Bump and	d pipipa romo		Γ		
2 Well / Drillhole / Borobala Int				piping remo	weu:	Ē		
	iginal Construction Dat	0		moved?		Ē		
Monitoring Well	27/12		Casing le	ft in place?		Ē	$\exists res \square nc$ $\exists v_{os} \square nc$	
Water Well	a Wall Construction Pa	nort is susilable				с Г		
Borehole / Drillhole	lease attach.	port is available,	vvas casi	ng cut on bei	ow surface?	ے ۱		
Construction Type:			Did seallr	ig material ris		r L		
Drilled Driven (Sat	ndpoint)	uq	If ves	was hole ref	topped?	Ē		
Cthor (specify): Geoprobe		- 5	If bentonit	e chips were	used, were th	hey hydrated		
			with water	from a know	n safe source			)ビN/A
Formation Type:	-			etor Dino Cro		alenai aduatar Dina Dur	mood	
Unconsolidated Formation	Bedrock			led & Poured			IIPEU	tonite
Total Well Depth From Groundsurfac	ce (ft.) Casing Diame	ter (in.)	(Bento	nite Chips)	L Oth	ier (Explain): 100		
	Or size a Dan th	151.)	Sealing Mate	erials				
Lower Drillhole Diameter (In.)	Casing Depth	(π.)		ement Grout			and Slurry (11 I	b./gal. wt.)
2.120				Jement (Cond	crete) Grout		ite-Sand Slurry	
Was well annular space grouted?	∐Yes ĽN	o 📙 Unknown	For Monitori	na Wells and I	Monitorina M		ne Chips nlv	
If yes, to what depth (feet)?	Depth to Water (fee	t)		ite Chips		Bentonite - Ce	ment Grout	
			Granul	ar Bentonite	Ē	Bentonite - Sa	nd Slurry	
5. Material Used To Fill Well / Dril	lihole		From (ft.)	To (ft.)	No. Yards or Volur	s, Sacks Sealan ne (circle one)	t Mix Ra Mud W	itio or /eiaht
Chipped Bentonite			Surface	25	25 lbs	, <b>.</b> )		
6. Comments					-			

7. Supervision of Work	DNR Use Only						
Name of Person or Firm Doing Sealing Work			Date of Abandonment		Date Received	Noted By	
Dave Paulson			2/27/12				
Street or Route			Telephone N	lumber	Comments		
W6306 State Road 39			(608) 527	-2355			
City	State	ZIP Code	Э	Signature of Perso	on Doing Work		Date Signed
New Glarus	WI	53574					4/19/12

Douto to:

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Drinking Water Watershe	ed/Wastewater 🔲 Was	ste Management	✓ Remedi	ation/Redeve	lopment	Other:		
1. General Information			2. Facility	/ Owner In	formation			
WI Unique Well No. DNR We	II ID No. County		Facility Nam	е				
	Sauk		Badger Ar	my Ammunitio	n Plant			
Common Well Name	Gov't Lot # (if a	pplicable)	Facility ID		Lice	nse/Permit/Mon	itoring No.	
VIP-6			15705393	0				
1/4 / 1/4 Secti	on Township I	Range 🗹 E	Street Addre	ess of Well				
NW / SW 25	10 N	6 🗌 W	2 Badger Ro	ad				
Well Location ft. / M (Local G	Grid 🔲 )	Datum	City, Village	or Iown				
541455 (N) (S 31	15689	W NAD 83	Baraboo Present Wel			Original Well C	Jwper	
		Zone		Owner				
WTM-	ngitude- State Plane		Street Addre	ess or Route (	of Present O	wner		
Local Grid Origin ft. / M		Datum	2 Badger R	oad				
N,	E	/ W	Citv			State	ZIP Code	
		Zone	Baraboo			WI	53913	
	ngitude- State Plan		4. Pump. I	_iner. Scree	en. Casing	& Sealing Mat	erial	
Reason For Abandonment	VVI Unique Vvell No. o	Replacement We				<u>г у</u> Г	$\overline{\mathbf{y}}$	
Temp. Borenoie-vapor Collection				a piping remo	ived?	F		
3. Well / Drillhole / Borehole I	nformation		Liner(s) re	emoved?		F		
Monitoring Well	Driginal Construction Dat	e	Screen re	moved?		F		
	2/27/12		Casing le	ft in place?		<u>_</u>		
Borebole / Drillhole	If a Well Construction Re please attach.	port is available,	Was casi	ng cut off belo	ow surface?		JYes ∐No 기 □	
			Did sealir	ng material ris	se to surface	?	JYes ∐No ⊐ ⊡	
			Did mater	rial settle afte	r 24 hours?	F	JYes ĽNo	
		ug	If yes	, was hole ret	topped?	L hev bydrated	JYes ∐No	ĽN/А
Other (specify): Geoprobe			with water	from a know	n safe source	e?		ビ <sub>N/A</sub>
Formation Type:			Required Me	thod of Placir	ng Sealing Ma	aterial		
<ul> <li>Unconsolidated Formation</li> </ul>	Bedrock			ctor Pipe-Gra	vity 📙 Cor	nductor Pipe-Pun	nped	
Total Well Depth From Groundsurfa	ace (ft.) Casing Diame	ter (in.)	Bento	nite Chips)	ビ Oth	er (Explain): Pou	red / chipped bent	onite
45			Sealing Mate	erials		_		
Lower Drillhole Diameter (in.)	Casing Depth	(ft.)	Neat C	ement Grout		Clay-Sa	nd Slurry (11 l	b./gal. wt.)
2.125				Cement (Conc	crete) Grout	Bentoni	te-Sand Slurry	
Was well annular space grouted?	□ <sub>Yes</sub> ビ <sub>N</sub>	o 🗌 Unknown		ete		🖌 Bentoni	te Chips	
If ves to what depth (feet)?	Depth to Water (fee	t)	For Monitorii	ng Wells and I	Monitoring W	/ell Boreholes Or 1	nly:	
		()		nite Chips		Bentonite - Cer	nent Grout	
			Granul	ar Bentonite		Bentonite - Sar		tio or
5. Material Used To Fill Well / Dr	rillhole		From (ft.)	To (ft.)	or Volum	ne (circle one)	Mud W	eight
Chipped Bentonite			Surface	45	40 lbs			
6. Comments								

7. Supervision of Work	DNR Use Only						
Name of Person or Firm Doing Sealing Work			Date of Abandonment		Date Received	Noted By	
Dave Paulson			2/27/12				
Street or Route			Telephone N	lumber	Comments		
W6306 State Road 39			(608) 527	-2355			
City	State	ZIP Code	Э	Signature of Perso	on Doing Work		Date Signed
New Glarus	WI	53574					4/19/12
Douto to:

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Drinking Water Watershe	ed/Wastewater	aste Management	✓ Remed	ation/Redeve		Other:		
1. General Information			2. Facility	/ Owner In	formation			
WI Unique Well No. DNR We	II ID No. County		Facility Nam	ie				
	Sauk		Badger A	my Ammunitio	n Plant			
Common Well Name	Gov't Lot # (if a	applicable)	Facility ID		Lice	ense/Permit/Moni	itoring No.	
VIP-7			15705393	60				
1/4 / 1/4 Secti	ion Township	Range 🔽 E	Street Addre	ess of Well				
NW / NW 36	10 N	6 🗖 V	2 Badger Ro	ad				-
Well Location ft. / M (Local C	Grid 🔲 )	Datum	City, Village	or Town				
541541	14767 <b>(</b> E	W NAD 83	Baraboo					
		Zone	Present vve	ll Owner		Original Well O	wner	
WTM-	ngitude- State Plar	ne-🖌 🚺 🗌	U.S. Army	no or Douto	of Dropont C	)		
Local Grid Origin ft. / M		Datum	2 Badger R	and	or Present C	Jwhen		
Ν,	E	/ W	City			State	ZIP Code	
		Zone	Baraboo			WI	53913	
WTM-UTM-Latitude/Lo	ngitude- State Plar			iner Scree	en Casing	& Sealing Mat	erial	
Reason For Abandonment	WI Unique Well No. o	of Replacement W			in, casing			
Temp. Borehole-Vapor Collection	Pump an	Pump and piping removed?						
3. Well / Drillhole / Borehole I	nformation		Liner(s) r	emoved?		Ľ		ШN/А
	Driginal Construction Da	ate	Screen re	emoved?				M/A
	2/28/12		Casing le	ft in place?		L		<u>ЦN/A</u>
	If a Well Construction Re	eport is available,	Was casi	ng cut off belo	ow surface?	Ļ	Yes No	년 <sub>N/A</sub>
			Did sealir	ng material ris	se to surface	? Ľ	ĹYes ∐No	∐n/A
			Did mate	rial settle afte	er 24 hours?	Ļ	JYes ĽNo	∐n/A
Drilled Driven (S	andpoint)	Dug	If yes	, was hole ret	topped?	L	J <sub>Yes</sub> ∐ <sub>No</sub>	ビ <sub>N/A</sub>
✓ Other (specify): Geoprobe			<ul> <li>If bentoning</li> <li>with wate</li> </ul>	te chips were r from a know	used, were t n safe sourc	e?	] <sub>Yes</sub> □ <sub>No</sub>	⊡ <sub>N/A</sub>
Formation Type:			Required Me	ethod of Placir	ng Sealing M	aterial		
<ul> <li>Unconsolidated Formation</li> </ul>	Bedrock		Condu	ctor Pipe-Gra	vity 🔲 Co	nductor Pipe-Pun	nped	
Total Well Depth From Groundsurf	ace (ft.) Casing Diame	eter (in.)	Screer     (Bento)	ned & Poured	ビ Oth	ner (Explain): Pour	red / chipped bento	onite
45		. ,	Sealing Mate	erials				
Lower Drillhole Diameter (in.)	Casing Depth	n (ft.)	Neat C	ement Grout		🗌 Clay-Sa	nd Slurry (11 lt	o./gal. wt.)
2.125			Sand-	Cement (Cond	crete) Grout	Bentonit	te-Sand Slurry	
Was well appular space drouted?				ete		🖌 Bentonit	te Chips	
The second secon			For Monitori	ng Wells and	Monitoring V	Vell Boreholes Or	nly:	
If yes, to what depth (leet)?		et)	Bentor	nite Chips	Ļ	Bentonite - Cer	ment Grout	
			Granu	ar Bentonite	L.,	Bentonite - Sar	nd Slurry	
5. Material Used To Fill Well / Dr	rillhole		From (ft.)	To (ft.)	No. Yards or Volu	s, Sacks Sealant me (circle one)	Mix Rat Mud W	tio or eight
Chipped Bentonite			Surface	45	40 lbs			
6. Comments								

7. Supervision of Work	Supervision of Work						
Name of Person or Firm Doing Sealing Work			Date of Abandonment		Date Received	Noted By	
Dave Paulson			2/28/12				
Street or Route			Telephone Number		Comments		
W6306 State Road 39			(608) 527-2355				
City New Glarus	State WI	ZIP Code 53574	Э	Signature of Perso	on Doing Work		Date Signed 4/19/12

Douto to:

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Drinking Water Watershe	d/Wastewater	ste Management	✓ Remedi	ation/Redeve		Other:		
1. General Information			2. Facility	/ Owner In	formation			
WI Unique Well No. DNR Wel	II ID No. County		Facility Nam	ne				
	Sauk		Badger Ar	my Ammunitio	n Plant			
Common Well Name	Gov't Lot # (if a	pplicable)	Facility ID	_	Lice	nse/Permit/Moni	toring No.	
VIP-8		_	15705393	<u> </u>				
<sup>1</sup> / <sub>4</sub> / <sup>1</sup> / <sub>4</sub> Section	on Lownship	Range 🗹 E	Street Addre	ess of Well				
SW / SW 25	10 N	6 🗌 W	City Villago	ad or Town				
Well Location ft. / M (Local G	Grid 🔲 )	Datum	Barahoo	OF TOWN				
541463 (N) (S 31	5099 <b>E</b>	W NAD 83	Present We	ll Owner		Original Well O	)wner	
			U.S. Army					
	ngitudeState Plan		Street Addre	ess or Route	of Present O	wner		
		Datum	2 Badger R	oad				
N,	E	/W	City			State	ZIP Code	
WTM- UTM- Latitude/Lor	ngitude- State Plan		Baraboo			WI	53913	
Reason For Abandonment	WI Unique Well No. o	f Replacement W	4. Pump, l	Liner, Scree	en, Casing	& Sealing Mate	erial	
Temp. Borehole-Vapor Collection			Pump and	d piping remo	oved?		] <sub>Yes</sub> □ <sub>No</sub>	⊡ <sub>N/A</sub>
3. Well / Drillhole / Borehole Ir	Liner(s) r	emoved?			] <sub>Yes</sub> □ <sub>No</sub>	Ŀ N/A		
	Driginal Construction Da	te	Screen re	emoved?			] <sub>Yes</sub> □ <sub>No</sub>	
Monitoring Well	2/28/12		Casing le	ft in place?			] <sub>Yes</sub> D <sub>No</sub>	۲ <sub>N/A</sub>
Water Well	If a Well Construction Re	port is available,	Was casi	ng cut off belo	ow surface?		JYes DNo	Ŀ N/A
Borehole / Drillhole	please attach.		Did sealir	na material ris	se to surface	? •	J <sub>Yes</sub> □ <sub>No</sub>	
Construction Type:	_		Did mate	rial settle afte	r 24 hours?		JYes INO	
Drilled Driven (Sa	andpoint)	ug	If yes	, was hole ret	topped?		] <sub>Yes</sub> □ <sub>No</sub>	۲. N/A
✓ Other (specify): <u>Geoprobe</u>			If bentonit	te chips were	used, were the	hey hydrated		
Formation Type:			Required Me	thod of Placir	ng Sealing Ma	aterial		
	Bedrock		Condu	ctor Pipe-Gra	vity 🗌 Cor	nductor Pipe-Pur	nped	
Total Well Depth From Groundsurfa	ace (ft.) Casing Diame	ter (in.)		ned & Poured	Oth	er (Explain): _Pour	red / chipped bento	onite
45			Sealing Mate	erials		· · · / <u> </u>		
Lower Drillhole Diameter (in.)	Casing Depth	(ft.)		Cement Grout		Clay-Sa	nd Slurry (11 lł	b./gal. wt.)
2.125			Sand-	Cement (Cond	crete) Grout	Bentonit	te-Sand Slurry	" "
Was well appular space drouted?				ete		🖌 Bentonit	te Chips	
Was well allitudal space grouted?			For Monitori	ng Wells and	Monitoring W	ell Boreholes Or	nly:	
If yes, to what depth (feet)?	Depth to water (ree	et)	Bentor	nite Chips		Bentonite - Cer	nent Grout	
			Granu	ar Bentonite		Bentonite - Sar	nd Slurry	
5. Material Used To Fill Well / Dr	illhole		From (ft.)	To (ft.)	No. Yards or Volun	s, Sacks Sealant ne (circle one)	Mix Rat Mud W	tio or eight
Chipped Bentonite			Surface	45	40 lbs			
6. Comments								

7. Supervision of Work	Supervision of Work						
Name of Person or Firm Doing Sealing Work			Date of Abandonment		Date Received	Noted By	
Dave Paulson			2/28/12				
Street or Route			Telephone Number		Comments		
W6306 State Road 39			(608) 527	-2355			
City	State	ZIP Code	Э	Signature of Perso	on Doing Work		Date Signed
New Glarus	WI	53574					4/19/12

Douto to:

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Drinking Water Watershe	ed/Wastewater	aste Management	✓ Remedi	ation/Redeve	lopment	Other:		
1. General Information			2. Facility	/ Owner In	formation			
WI Unique Well No. DNR We	ell ID No. County		Facility Nam	ie				
	Sauk		Badger Ar	my Ammunitio	n Plant			
Common Well Name	Gov't Lot # (if a	applicable)	Facility ID		Lic	ense/Permit/Mor	nitoring No.	
VIP-9		1	15705393	0				
1/4 / 1/4 Secti	ion Township	Range 🗹 E	Street Addre	ess of Well				
SW / SW 25	10 N	6 🗌 W	City Villago	ad or Town				
Well Location ft. / M (Local C	Grid 🔲 )	Datum	Barahaa	OFTOWN				
541465 (N) (S) 31	15363 <b>(</b> E	W NAD 83	Present Wel	ll Owner		Original Well (	Owner	
			U.S. Army					
	ongitudeState Plan		Street Addre	ess or Route	of Present (	 Owner		
Local Glid Oligin ft. / M		Datum	2 Badger R	oad				
N,	E	/ W	City			State	ZIP Code	
WTM- UTM- Latitude/Lo	ngitude-		Baraboo			WI	53913	
Reason For Abandonment	WI Unique Well No. d	of Replacement W	4. Pump, l	Liner, Scree	en, Casing	& Sealing Ma	terial	
Temp. Borehole-Vapor Collection			Pump and	d piping remo	oved?	Γ	∃ <sub>Yes</sub> □ <sub>No</sub>	, <b>∠</b> <sub>N/A</sub>
3. Well / Drillhole / Borehole I	nformation		Liner(s) r	emoved?		Γ	J <sub>Yes</sub> □ <sub>No</sub>	, <b>L</b> <sub>N/A</sub>
	Driginal Construction Da	ite	Screen re	emoved?		Γ	∃ <sub>Yes</sub> □ <sub>No</sub>	, 🗹 <sub>N/A</sub>
Monitoring Well	2/28/12		Casing le	ft in place?			∃ <sub>Yes</sub> □ <sub>No</sub>	, ⊡ <sub>N/A</sub>
Water Well	If a Well Construction Re	eport is available,	Was casi	na cut off belo	ow surface?	2 [		. ⊡ <sub>N/A</sub>
✓ Borehole / Drillhole	please attach.		Did sealir	na material ris	se to surface	e?	IJ <sub>Yes</sub> □ <sub>No</sub>	$\Box_{N/A}$
Construction Type:			Did mate	rial settle afte	er 24 hours?		J <sub>Yes</sub> $\square_{No}$	$\square_{N/A}$
Drilled Driven (S	Sandpoint)	Dug	If yes	, was hole ret	topped?	Γ	J <sub>Yes</sub> □ <sub>No</sub>	, ⊡ <sub>N/A</sub>
✓ Other (specify): Geoprobe			If bentonit	te chips were r from a know	used, were	they hydrated		
Formation Type:			Required Me	thod of Placir	ng Sealing N	Aaterial		
	Bedrock		Condu	ctor Pipe-Gra	ivity 🗌 Co	onductor Pipe-Pu	mped	
Total Well Depth From Groundsurf	ace (ft.) Casing Diame	eter (in.)		ned & Poured	🗹 <sub>Ot</sub>	her (Explain): Pou	ured / chipped bent	onite
45		,	Sealing Mate	erials				
Lower Drillhole Diameter (in.)	Casing Depth	(ft.)		Cement Grout		Clay-Sa	and Slurry (11 I	b./gal. wt.)
2.125			Sand-	Cement (Cond	crete) Grout	Benton	ite-Sand Slurry	,
Was well appular space drouted?				ete		🖌 Benton	ite Chips	
If you to what donth (foot)?	Depth to Water (for		For Monitori	ng Wells and	Monitoring \	Well Boreholes O	nly:	
in yes, to what depth (leet)?		50)	Bentor	nite Chips	Ļ	Bentonite - Ce	ment Grout	
			Granul	ar Bentonite		Bentonite - Sa	nd Slurry	tio or
5. Material Used To Fill Well / Dr	rillhole		From (ft.)	To (ft.)	or Volu	ime (circle one)	Mud W	lio or leight
Chipped Bentonite			Surface	45	45 lbs			
6. Comments								

7. Supervision of Work	Supervision of Work						
Name of Person or Firm Doing Sealing Work			Date of Abandonment		Date Received	Noted By	
Dave Paulson			2/28/12				
Street or Route			Telephone Number		Comments		
W6306 State Road 39			(608) 527-2355				
City New Glarus	State WI	ZIP Code 53574	Э	Signature of Perso	on Doing Work		Date Signed 4/19/12

Douto to:

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Drinking Water Watershe	ed/Wastewater	ste Management	🖌 Remedi	ation/Redeve	lopment	Other:		
1. General Information			2. Facility	/ Owner In	formation			
WI Unique Well No. DNR We	II ID No. County		Facility Nam	e				
	Sauk		Badger Ar	my Ammunitio	n Plant			
Common Well Name	Gov't Lot # (if a	applicable)	Facility ID		Lice	ense/Permit/Mor	itoring No.	
VIP-10			15705393	0				
1/4 / 1/4 Secti	on Township	Range 🖌 E	Street Addre	ess of Well				
SW / NW 26	10 N	6 🗌 W	2 Badger Ro	ad -				
Well Location ft. / M (Local G	Grid 🔲 )	Datum	City, Village	or Iown				
541565 (N) S 31	16021 <b>(</b> E	W NAD 83	Baraboo	Quiner			Junor	
		Zone		Owner		Original Well C	Jwher	
WTM-	ngitude- State Plan	e– <b>∠ (</b> s)C ⊾	Street Addre	ss or Route	of Present (	Jwper		
Local Grid Origin ft. / M		Datum	2 Badger R	oad		JWIIEI		
N,	E	/ W	City			State	ZIP Code	
		Zone	Baraboo			WI	53913	
WTM- UTM- Latitude/Lo	ngitude- State Plan		4. Pump. I	_iner. Scree	en. Casing	& Sealing Ma	terial	
Reason For Abandonment	WI Unique Well No. c	of Replacement W			, eac	<u>с с с с с с с с с с с с с с с с с с с </u>		<b>.</b>
Temp. Borenoie-vapor Collection	Pump and	d piping remo	oved?	L F				
3. Well / Drillhole / Borehole I	nformation		Liner(s) r	emoved?		L F		, ĽN/A
	Driginal Construction Da	ite	Screen re	emoved?		L	JYes ∐No ⊐ □	, ĽN/A
	2/28/12		Casing le	ft in place?		L		) ĽN/A
	If a Well Construction Re	eport is available,	Was casi	ng cut off belo	ow surface?	, L r		) ĽN/A
			Did sealir	ng material ris	se to surface	∋? Ľ	∐Yes ∐No	, ∐N/A
			Did mater	rial settle afte	r 24 hours?	Ļ	_Yes ĽNo	, ∐N/A
		Jug	If yes	, was hole ret	topped?	they bydrated -		,ビ <sub>N/A</sub>
Other (specify): Geoprobe			with water	from a know	n safe sourc	ce?		,
Formation Type:			Required Me	thod of Placir	ng Sealing M	laterial		
Unconsolidated Formation	Bedrock		Condu	ctor Pipe-Gra	vity 📙 Co	onductor Pipe-Pu	mped	
Total Well Depth From Groundsurf	ace (ft.) Casing Diame	eter (in.)	Creer     (Bento	ned & Poured nite Chips)	Otl Otl	her (Explain): <u>Pou</u>	ured / chipped bent	onite
45			Sealing Mate	erials				
Lower Drillhole Diameter (in.)	Casing Depth	(ft.)	Neat C	ement Grout		Clay-Sa	and Slurry (11 I	b./gal. wt.)
2.125			Sand-(	Cement (Cond	crete) Grout	Benton	ite-Sand Slurry	, <b>н</b> н
Was well annular space grouted?				ete		<ul> <li>Benton</li> </ul>	ite Chips	
If yos, to what dopth (foot)?	Depth to Water (for	ot)	For Monitorii	ng Wells and I	Monitoring V	Vell Boreholes O ¬	nly:	
in yes, to what depth (leet):		50)	Bentor	ite Chips	Ļ	Bentonite - Ce	ment Grout	
			Granul	ar Bentonite		Bentonite - Sa	nd Slurry	410.07
5. Material Used To Fill Well / Dr	rillhole		From (ft.)	To (ft.)	or Volu	me (circle one)	Mud W	lio or /eight
Chipped Bentonite			Surface	45	40 lbs			
6. Comments								

7. Supervision of Work	Supervision of Work						
Name of Person or Firm Doing Sealing Work			Date of Abandonment		Date Received	Noted By	
Dave Paulson			2/28/12				
Street or Route			Telephone Number		Comments		
W6306 State Road 39			(608) 527-2355				
City New Glarus	State WI	ZIP Code 53574	Э	Signature of Perso	on Doing Work		Date Signed 4/19/12

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Route to:	vater Waste Management	✓ Remedi	ation/Redevel				
1 General Information		2 Facility	/ Owner Inf	formation			
WI Unique Well No. DNR Well ID No. C	County	Facility Nam	e				
	Sauk	Badger Arr	my Ammunitior	n Plant			
Common Well Name	Gov't Lot # (if applicable)	Facility ID		Lice	ense/Permit/Mor	nitoring No.	
VIP-1A		15705393	D				
1/4 / 1/4 1/4 Section T	Fownship Range	Street Addre	ss of Well				
NE / SW 14	10 N 6 $\square$ W	2 Badger Roa	ad				
Well Location ft. / M (Local Grid )	Datum	City, Village	or Town				
540043	(E) W NAD 83	Baraboo	Ourreat			2	
			Owner		Original well C	Jwner	
WTM-	State Plane-	Street Addre	ss or Route (	of Present (	)wner		
Local Grid Origin ft. / M	Datum	2 Badger Ro	bad		When		
N,	E / W	City			State	ZIP Code	
		Baraboo			WI	53913	
WIM- UIM- Latitude/Longitude-		4. Pump, L	iner, Scree	en, Casing	& Sealing Ma	terial	
Reason For Abandonment WI Unit	que vveil No. or Replacement vveil	Bump one		vod?			
2 Well / Drillhele / Berehele Informati	<u></u>	Lipor(o) ro		weu?	ī		
5. Well / Drinnole / Borenole Information		Soroon ro	moved?		Ē		
Monitoring Well 7/3/12	Distruction Date	Cosing lof			Ī		
Water Well	Construction Poport is available				<u>_</u> _		
Borehole / Drillhole please atta	ach.	vvas casir		ow surface?			
Construction Type:		Did sealin	g material ris	se to surface	γ L Γ		
Drilled Driven (Sandpoint)	Dug	Did mater	was hole ret	r 24 nours?	Г		
Cother (specify): Geoprobe		If bentonit	e chips were	used, were t	hey hydrated		
		with water	from a known	n safe sourc	e? L		) ビN/A
Formation Type:	-				nduotor Dino Du	mod	
Unconsolidated Formation	Bedrock		ed & Poured			rred / chipped ben	tonite
I otal Well Depth From Groundsurface (ft.)	Casing Diameter (in.)	Bentor	nite Chips)	L Otr	ier (Explain):	area, empped sen	
45	Cooing Donth (ft.)	Sealing Mate	rials			a a d Oliverni (44	lla (malt.)
2 125	Casing Depth (it.)		ement Grout	roto) Crout		and Siurry (11	ib./gai. wt.)
2.123			to	(rele) Grout		ite-Sand Slurry	/
Was well annular space grouted?	Yes No Unknown	For Monitorin	ne Na Wells and I	Monitorina V	Vell Boreholes O	nle Chips Inly:	
If yes, to what depth (feet)? Depth	n to Water (feet)	Benton	ite Chips		Bentonite - Ce	ment Grout	
		Granula	ar Bentonite	Ē	Bentonite - Sa	nd Slurry	
5. Material Used To Fill Well / Drillhole		From (ft.)	To (ft.)	No. Yards or Volu	s, Sacks Sealan me (circle one)	t Mix Ra Mud V	atio or Veight
Chipped Bentonite		Surface	45	40 lbs			
6. Comments							

7. Supervision of Work	DNR Use Only						
Name of Person or Firm Doing Sealing Work			Date of Abandonment		Date Received	Noted By	
Dave Paulson			7/3/12				
Street or Route		Telephone Number		Comments			
W6306 State Road 39			(608) 527	-2355			
City	State	ZIP Code	Э	Signature of Perso	n Doing Work		Date Signed
New Glarus	WI	53574					7/12/12

Douto to:

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Drinking Water Watershe	ed/Wastewater	ste Management	🖌 Remedi	ation/Redevel		Other:		
1. General Information			2. Facility	/ Owner In	formation			
WI Unique Well No. DNR We	ell ID No. County		Facility Nam	е				
	Sauk		Badger Ar	my Ammunitio	n Plant			
Common Well Name	Gov't Lot # (if a	pplicable)	Facility ID	_	Licer	nse/Permit/Moni	toring No.	
VIP-4A			15705393	0				
1/4 / 1/4 Sect	ion Township H	Range 🗹 E	Street Addre	ess of Well				
SW / NE 26	10 N	6 🗌 W	2 Badger Roa	ad or Town				
Well Location ft. / M (Local C	Grid 🔲 )	Datum	Barahoo					
540766 N S 3	15809 <b>E</b>	W NAD 83	Present Wel	l Owner		Original Well O	wner	
			U.S. Army					
	ongitudeState Plane		Street Addre	ess or Route of	of Present Ov	wner		
Eccar Ghu Ongin ft. / M	_	Datum	2 Badger R	oad				
N,	E	/ W	City			State	ZIP Code	
WTM- UTM- Latitude/Lo	ongitude- State Plane		Baraboo			WI	53913	
Reason For Abandonment	WI Unique Well No. of	Replacement We	4. Pump, I	_iner, Scree	en, Casing &	& Sealing Mate	erial	
Temp. Borehole-Vapor Collection	· · · · · · · · · · · · · · · · · · ·	·	Pump and	d piping remo	ved?		] <sub>Yes</sub> □ <sub>No</sub>	☑ <sub>N/A</sub>
3. Well / Drillhole / Borehole I	nformation		Liner(s) re	emoved?			] <sub>Yes</sub> □ <sub>No</sub>	⊡ <sub>N/A</sub>
	Driginal Construction Dat	e	Screen re	moved?			] <sub>Yes</sub> □ <sub>No</sub>	☑ <sub>N/A</sub>
Monitoring Well	6/27/12		Casing le	ft in place?			] <sub>Yes</sub> D <sub>No</sub>	⊡ <sub>N/A</sub>
Water Well	If a Well Construction Re	port is available,	Was casi	ng cut off belo	ow surface?		] <sub>Yes</sub> □ <sub>No</sub>	⊡ <sub>N/A</sub>
✓ Borehole / Drillhole	please attach.		Did sealir	ng material ris	e to surface?	?	] <sub>Yes</sub> □ <sub>No</sub>	$\Box_{N/A}$
Construction Type:	_		Did mater	rial settle after	r 24 hours?		] <sub>Yes</sub> ∠ <sub>No</sub>	□ <sub>N/A</sub>
Drilled Driven (S	Sandpoint)	ug	If yes, was hole retopped?					
✓ Other (specify): Geoprobe			If bentonite chips were used, were they hydrated $\Box_{\text{Yes}}$ $\Box_{\text{No}}$ $\Box'_{\text{N/A}}$					
Formation Type:			Required Me	thod of Placin	ng Sealing Ma	aterial		
Unconsolidated Formation	Bedrock		Condu	ctor Pipe-Grav	vity 🔲 Con	ductor Pipe-Pum	nped	
Total Well Depth From Groundsurf	face (ft.) Casing Diamet	ter (in.)	- Screen	ned & Poured	C Othe	er (Explain): Pour	ed / chipped bento	onite
37		( )	Sealing Mate	erials				
Lower Drillhole Diameter (in.)	Casing Depth	(ft.)	Neat C	ement Grout		Clay-Sa	nd Slurry (11 lb	o./gal. wt.)
2.125			Sand-C	Cement (Conc	rete) Grout	Bentonit	e-Sand Slurry	
Was well annular space grouted?	Yes VN			ete		🖌 Bentonit	e Chips	
If yes, to what depth (feet)?	Depth to Water (fee	e <u> </u>	For Monitorii	ng Wells and I	Monitoring W	ell Boreholes On	ly:	
in yes, to what depth (leet):		()	Benton	ite Chips	님	Bentonite - Cen	nent Grout	
			Granul	ar Bentonite		Bentonite - San	d Slurry	tio or
5. Material Used To Fill Well / D	rillhole		From (ft.)	To (ft.)	or Volum	ne (circle one)	Mud W	eight
Chipped Bentonite			Surface	37	30 lbs			
6. Comments								

7. Supervision of Work	DNR Use Only						
Name of Person or Firm Doing Sealing Work			Date of Abar	ndonment	Date Received	Noted By	
Dan Bendorf			6/27/12				
Street or Route			Telephone Number		Comments		
W1225 South Shore Drive			(262) 470-4768				
City	State	ZIP Code	9	Signature of Perso	on Doing Work		Date Signed
Palmyra	WI	53156					7/12/12

Douto to:

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Drinking Water Watershe	ed/Wastewater	ste Management	🖌 Remedi	ation/Redeve	lopment	Other:		
1. General Information			2. Facility	/ Owner In	formation			
WI Unique Well No. DNR We	ell ID No. County		Facility Nam	е				
	Sauk		Badger Ar	my Ammunitio	n Plant			
Common Well Name	Gov't Lot # (if a	applicable)	Facility ID	_	Lice	nse/Permit/Moni	itoring No.	
VIP-5A		L_	15705393	0				
<sup>1</sup> / <sub>4</sub> / <sup>1</sup> / <sub>4</sub> Sect	tion llownship	Range $\checkmark$ E	Street Addre	ess of Well				
SE / NE 26	10 N	6 🗌 W	2 Badger Ros	ad or Town				
Well Location ft. / M (Local (	Grid 🔲 )	Datum	Barahoo					
541097 N S 3	15811 <b>(</b> E	W NAD 83	Present Wel	l Owner		Original Well O	)wner	
		Zone	U.S. Army					
	State Plan		Street Addre	ess or Route	of Present O	wner		
		Datum	2 Badger R	oad				
N,	E	/ W	City			State	ZIP Code	
WTM- UTM- Latitude/Lo	ongitude-		Baraboo			WI	53913	
Reason For Abandonment	WI Unique Well No. c	of Replacement We	4. Pump, l	_iner, Scree	en, Casing	& Sealing Mat	erial	
Temp. Borehole-Vapor Collection			Pump and	d piping remo	oved?		] <sub>Yes</sub> □ <sub>No</sub>	₽ N/A
3. Well / Drillhole / Borehole I	Information		Liner(s) re	emoved?			] <sub>Yes</sub> □ <sub>No</sub>	⊡ <sub>N/A</sub>
	Original Construction Da	te	Screen re	moved?			] <sub>Yes</sub> □ <sub>No</sub>	☑ <sub>N/A</sub>
Monitoring Well	6/27/12		Casing le	ft in place?			] <sub>Yes</sub> D <sub>No</sub>	Ľ <sub>N/A</sub>
Water Well	If a Well Construction Re	eport is available,	Was casi	ng cut off belo	ow surface?		J <sub>Yes</sub> D <sub>No</sub>	⊡ <sub>N/A</sub>
✓ Borehole / Drillhole	please attach.		Did sealir	ng material ris	se to surface	? 🔽	'] <sub>Yes</sub> □ <sub>No</sub>	
Construction Type:	_		Did mater	ial settle afte	r 24 hours?		] <sub>Yes</sub> ∠ <sub>No</sub>	
Drilled Driven (S	Sandpoint)	Dug	If yes, was hole retopped?					
✓ Other (specify): Geoprobe			If bentonite chips were used, were they hydrated $\Box_{Ves}$ $\Box_{No}$ $\nabla_{N/A}$					
Formation Type:			Required Me	thod of Placir	ng Sealing M	aterial		
Unconsolidated Formation	Bedrock		Condu	ctor Pipe-Gra	vity 🔲 Cor	nductor Pipe-Pum	nped	
Total Well Depth From Groundsurf	face (ft.) Casing Diame	eter (in.)	- Screer	ned & Poured	🖸 Oth	er (Explain): Pour	red / chipped bento	onite
40		( )	Sealing Mate	erials				
Lower Drillhole Diameter (in.)	Casing Depth	(ft.)	Neat C	ement Grout		🗌 Clay-Sa	nd Slurry (11 lt	b./gal. wt.)
2.125			Sand-C	Cement (Conc	crete) Grout	Bentonit	te-Sand Slurry	" "
Was well annular space grouted?				ete		<ul> <li>Bentonit</li> </ul>	te Chips	
If ves, to what denth (feet)?	Depth to Water (fee		For Monitorii	ng Wells and I	Monitoring W	/ell Boreholes Or 1	ıly:	
				nite Chips		Bentonite - Cer	nent Grout	
			Granul	ar Bentonite	No Vards	Bentonite - Sar		tio or
5. Material Used To Fill Well / D	rillhole		From (ft.)	To (ft.)	or Volur	ne (circle one)	Mud W	eight
Chipped Bentonite			Surface	40	35 lbs			
6. Comments								

7. Supervision of Work	DNR Use Only						
Name of Person or Firm Doing Sealing Work			Date of Abar	ndonment	Date Received	Noted By	
Dan Bendorf			6/27/12				
Street or Route			Telephone Number		Comments		
W1225 South Shore Drive			(262) 470-4768				
City	State	ZIP Code	9	Signature of Perso	on Doing Work		Date Signed
Palmyra	WI	53156					7/12/12

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Route to:	ewater Waste Management	Remedia	ation/Redevel		Other:		
1. General Information		2. Facility	/ Owner In	formation			
WI Unique Well No. DNR Well ID No.	o. County	Facility Nam	e				
	Sauk	Badger Ar	my Ammunitio	n Plant			
Common Well Name	Gov't Lot # (if applicable)	Facility ID		Licens	e/Permit/Monite	oring No.	
VIP-7A		15705393	0				
1/4 / 1/4 Section	Township Range	Street Addre	ss of Well				
NW / NW 36	$  10 N   6 \square W$	2 Badger Roa	ad				
Well Location ft. / M (Local Grid	) Datum	City, Village	or Town				
541541 (N) S 314767	(E) W NAD 83	Baraboo	0				
		Present Well	Owner	C	Driginal Well Ov	vner	
WTM-	- State Plane- 🖌 (S) C N	U.S. Anny Stroot Addro	co or Pouto /	of Procent Own	or		
Local Grid Origin ft. / M	Datum	2 Badger R	bad	or Fresent Owr	lei		
Ν,	E/W	City			State	ZIP Code	
		Baraboo			WI	53913	
WTM- UTM- Latitude/Longitude	- State Plane S C N	4. Pump. I	iner, Scree	en. Casing &	Sealing Mate	rial	
Reason For Abandonment WI	Jnique Well No. of Replacement Well						<b>E</b>
Temp. Borenoie-Vapor Collection		Pump and	piping remo	oved?	H		
3. Well / Drillhole / Borehole Information	ation	Liner(s) re	emoved?				
Monitoring Well	Construction Date	Screen re	moved?		H		
Water Well		Casing let	t in place?		<u>_</u>		
Borebole / Drillbole   Drease	Il Construction Report is available, attach	Was casir	ng cut off belo	ow surface?			
Construction Type:		Did sealin	g material ris	se to surface?			
		Did mater	ial settle afte	r 24 hours?			
		If yes,	was hole ret	topped?			ĽN/А
Other (specify): Geoprobe		with water	from a know	n safe source?		Yes No	ビ <sub>N/A</sub>
Formation Type:		Required Me	thod of Placir	ng Sealing Mate	erial		
<ul> <li>Unconsolidated Formation</li> </ul>	Bedrock		ctor Pipe-Gra	vity 📙 Condu	uctor Pipe-Pump	ped	
Total Well Depth From Groundsurface (ft.	) Casing Diameter (in.)	Bentor	nite Chips)	C Other	(Explain): Poure	d / chipped bento	nite
37		Sealing Mate	rials				
Lower Drillhole Diameter (in.)	Casing Depth (ft.)	Neat C	ement Grout		Clay-San	id Slurry (11 lb	./gal. wt.)
2.125		Sand-C	Cement (Conc	crete) Grout	Bentonite	Sand Slurry "	
Was well annular space grouted?	Yes VNo Unknown		te		Bentonite	Chips	
If ves. to what depth (feet)?	pth to Water (feet)	For Monitorir	ng Wells and I	Monitoring Well	I Boreholes Onl	y:	
	F		ite Chips	H	Bentonite - Cem	ent Grout	
			ar bentonite		Senionite - Sand	Mix Rat	io or
5. Material Used To Fill Well / Drillhole		From (ft.)	To (ft.)	or Volume	(circle one)	Mud We	eight
Chipped Bentonite		Surface	37	30 lbs			
6. Comments							

7. Supervision of Work	DNR Use Only						
Name of Person or Firm Doing Sealing Work			Date of Abar	ndonment	Date Received	Noted By	
Dan Bendorf			6/27/12				
Street or Route		Telephone Number		Comments			
W1225 South Shore Drive			(262) 470-4768				
City	State	ZIP Code	Э	Signature of Perso	n Doing Work		Date Signed
Palmyra	WI	53156					7/12/12

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Route to:							
Drinking Water Watershe	d/Wastewater 🗌 Waste Management	🖌 Remedi	ation/Redeve	elopment 🗌 Othe	er:		
1. General Information		2. Facility	2. Facility / Owner Information				
WI Unique Well No. DNR We	II ID No. County	Facility Nam	Facility Name				
	Sauk	Badger Ar	my Ammunitio	n Plant			
Common Well Name	Gov't Lot # (if applicable)	Facility ID		License/Pe	rmit/Monitor	ring No.	
VIP-9A		15705393	0				
1/4 / 1/4 Secti	on Township Range	Street Addre	ess of Well				
SW / SW 25	10 N 6 🗖 V	2 Badger Ro	ad				
Well Location ft. / M (Local G	Grid ) Datum	City, Village	or Iown				
541465 (N) (S 31	5363 (E) W NAD 83	Baraboo	Owner	Origin			
	Zone		Owner	Ongin		liei	
WTM-	ngitude- State Plane- 🖌 (S) C I	Street Addre	ss or Route	of Present Owner			
Local Grid Origin ft. / M	Datum	2 Badger R	oad				
N,	E / W	City			State	ZIP Code	
	Zone	Baraboo			WI	53913	
	ngitude– State Plane– S C I	<b>4.</b> Pump. I	iner. Scree	en. Casing & Seal	ing Mater	ial	
Reason For Abandonment	WI Unique Well No. of Replacement W	ell					
Temp. Borenoie-vapor Collection			a piping remo	oved?			
3. Well / Drillhole / Borehole li	nformation		emoved?				
Monitoring Well	Jriginal Construction Date	Screen re	moved?				
Water Well	6/27/12	Casing le	It in place?				
Borebole / Drillhole	If a Well Construction Report is available, please attach.	Was casi	ng cut off belo	ow surface?			
		Did sealir	☐ Did sealing material rise to surface? ☐ Yes ☐ No ☐ N/A				
		Did mater	ial settle afte	er 24 hours?			
		If yes	, was hole rei e chins were	topped?	irated 🗖	/es ∐No ĽN/A	
Other (specify): Geoprobe		- with water	from a know	n safe source?		<u>/es 凵No └N/A</u>	
Formation Type:		Required Me	thod of Placir	ng Sealing Material			
<ul> <li>Unconsolidated Formation</li> </ul>	Bedrock		ctor Pipe-Gra		Pipe-Pumpe	ed	
Total Well Depth From Groundsurfa	ace (ft.) Casing Diameter (in.)	Bento	nite Chips)	Other (Expl	ain): Poured	/ chipped bentonite	
39		Sealing Mate	erials				
Lower Drillhole Diameter (in.)	Casing Depth (ft.)	Neat C	ement Grout		Clay-Sand	l Slurry (11 lb./gal. wt.)	
2.125			Cement (Cond	crete) Grout	Bentonite-	Sand Slurry " "	
Was well annular space grouted?	Yes 🗹 No 🗌 Unknow		ete	~	Bentonite	Chips	
If yes, to what depth (feet)?	Depth to Water (feet)	For Monitorii	ng Wells and	Monitoring Well Bore	sholes Only:	:	
			ite Chips		nite - Ceme	int Grout	
		Granui	ar Bentonite	No Varde Sacks	nite - Sand	Siurry Mix Patio or	
5. Material Used To Fill Well / Dr	fillhole	From (ft.)	To (ft.)	or Volume (circ	le one)	Mud Weight	
Chipped Bentonite		Surface	39	35 lbs			
6. Comments							

7. Supervision of Work					DNR Use Only		
Name of Person or Firm Doing Sealing Work			Date of Abandonment		Date Received	Received Noted By	
Dan Bendorf		6/27/12					
Street or Route			Telephone Number		Comments		
W1225 South Shore Drive			(262) 470-4768				
City	State	ZIP Code	Э	Signature of Perso	n Doing Work		Date Signed
Palmyra	WI	53156					7/12/12

Douto to:

Well /	Drillhole /	Borehole	Abandonm	ent
Well /	Drillhole /	Borehole	Abandonm	en

Form 3300-005 (R 12/04)

Page 1 of 2

Drinking Water Watershee	d/Wastewater 🗌 Was	ste Management	🖌 Remedi	ation/Redeve	lopment	Other:		
1. General Information			2. Facility / Owner Information					
WI Unique Well No. DNR Well	I ID No. County		Facility Nam	е				
	Sauk		Badger Ar	my Ammunitio	n Plant			
Common Well Name	Gov't Lot # (if a	pplicable)	Facility ID		Lice	ense/Permit/Mon	itoring No.	
VIP-10A			15705393	0				
1/4 / 1/4 Section	on Township I	Range 🗹 E	Street Addre	ess of Well				
SW / NW 26	10 N	6 🗌 W	2 Badger Roa	ad				
Well Location ft. / M (Local G	irid 🔲 )	Datum	City, Village	or Iown				
541565 (N) (S) 316	6021	W NAD 83	Baraboo Present Wel				Jwner	
		Zone		Owner			JWIIEI	
WTM-	ngitude- State Plane		Street Addre	ess or Route (	of Present O	l Wher		
Local Grid Origin ft. / M		Datum	2 Badger R	oad				
N,	E	/ W	Citv			State	ZIP Code	
		Zone	Baraboo			WI	53913	
	ngitude- State Plane		4. Pump. L	iner. Scree	en. Casing	& Sealing Mat	terial	
Reason For Abandonment	WI Unique Well No. of	Replacement We				Г		
		·		a piping remo	oved?	F		
3. Well / Drillhole / Borehole In	nformation		Liner(s) re	emoved?		F		
Monitoring Well	riginal Construction Dat	e	Screen re	moved?		F		
	//3/12			It in place?		<u>F</u>		$\frac{1}{\sqrt{N/A}}$
Borehole / Drillhole	f a Well Construction Re please attach.	port is available,	Was casi	ng cut off belo	ow surface?		JYes ∐No 刀 □	, Ľ N/A
Construction Type:			Did sealir	ig material ris	se to surface	? Ľ	'∐Yes ∐No ⊐ III	, ∐N/A
			Did mater	ial settle afte	r 24 hours?	F		) L N/A
		ug	If yes, was hole retopped?					
Other (specify): Geoprobe			with water	from a know	n safe source	e?		, ビ <sub>N/A</sub>
Formation Type:			Required Me	thod of Placir	ng Sealing M	aterial		
<ul> <li>Unconsolidated Formation</li> </ul>	Bedrock			ctor Pipe-Gra	vity 📙 Cor	nductor Pipe-Pur	mped	
Total Well Depth From Groundsurfa	ace (ft.) Casing Diame	ter (in.)	Bento	nite Chips)	ビ Oth	er (Explain): Pou	red / chipped bent	onite
45			Sealing Mate	erials		_		
Lower Drillhole Diameter (in.)	Casing Depth	(ft.)		ement Grout		Clay-Sa	and Slurry (11 I	b./gal. wt.)
2.125				Cement (Cond	crete) Grout	Bentoni	ite-Sand Slurry	,
Was well annular space grouted?				ete		🖌 Bentoni	ite Chips	
If yes, to what depth (feet)?	Depth to Water (fee	t)	For Monitorin	ng Wells and I	Monitoring W	/ell Boreholes Oi	nly:	
		-)		ite Chips		Bentonite - Cei	ment Grout	
			Granui	ar Bentonite	No Vards	Bentonite - Sai		tio or
5. Material Used To Fill Well / Dri	illhole		From (ft.)	To (ft.)	or Volur	ne (circle one)	Mud W	/eight
Chipped Bentonite			Surface	45	40 lbs			
6. Comments								

7. Supervision of Work					DNR Use Only		
Name of Person or Firm Doing Sealing Work			Date of Abandonment		Date Received	Noted By	
Dave Paulson			7/3/12				
Street or Route			Telephone Number		Comments		
W6306 State Road 39			(608) 527-2355				
City	State	ZIP Code	e	Signature of Perso	on Doing Work		Date Signed
New Glarus	WI	53574					7/12/12



19 April 2012

Contracting Officer's Representative Badger Army Ammunition Plant 2 Badger Road Baraboo, WI 53913-5000

Subject: Vapor Intrusion Pathway Analysis Report

Dear Sir:

The Wisconsin Department of Natural Resources (WDNR) issued a letter dated September 9, 2011 reminding responsible parties that a vapor intrusion investigation should be conducted at all sites where volatile organic compounds (VOC) are present in the soil and groundwater. The Propellant Burning Ground (PBG) Plume contains the following VOCs: carbon tetrachloride, chloroform and trichloroethylene (TCE). The Deterrent Burning Ground and Central Plumes contain primarily dinitrotoluene (DNT), which does not pose a vapor pathway risk.

The *Alternative Feasibility Study* – *Groundwater Remedial Strategy* (Alt FS) document states that a vapor intrusion pathway analysis would be conducted along the PBG Plume to determine the level of risk to down-gradient receptors. Based on the WDNR letter and the Alt FS, Badger Technical Services, LLC (BTS) conducted a vapor intrusion pathway analysis for the PBG Plume. BTS performed this work under Performance Work Statement (PWS) #0050.

The PBG Plume extends along the southwest portion of Badger Army Ammunition Plant (BAAAP) and off-site toward the Wisconsin River in the direction of groundwater flow. Figure 1 shows the PBG Plume area in relation to BAAAP. On February 27 and 28, 2012, BTS personnel conducted ten vapor intrusion pathway (VIP) Geoprobe<sup>®</sup> borings to a depth of 25 or 45 feet approximately half-way to the groundwater table (per WDNR vapor intrusion guidance, *Addressing Vapor Intrusion at Remediation & Redevelopment Sites in Wisconsin*, PUB-RR-800, December 2010) using post-run tubing vapor sampling technique. VIP borings were placed at well nest locations to compare groundwater contaminant concentrations to the soil vapor results. VIP boring locations are depicted on Figure 2.

After drilling the VIP boring to the desired depth, new high-density polyethylene 3/8-inch or 1/4-inch OD tubing was placed down to the bottom of the borehole. The tubing was connected to a flow controller/sampler attached to a 6-liter summa canister. Under negative pressure, the canister was allowed to draw vapor from the bottom of the borehole until filled, approximately 30 to 50 minutes. To ensure sample integrity, quality control leak detection tracer techniques were used during sample collection at borings VIP-1 through VIP-9. Isopropyl alcohol wetted rags were placed around fitting connections to determine whether any leaking or short-circuiting of air through the fittings had occurred. Boring VIP-10 was inadvertently omitted from the leak detection methodology. Soil borings were abandoned following sample collection.

Contracting Officer's Representative 19 April 2012 Page 2 of 2

Soil vapor samples were submitted to the Wisconsin State Laboratory of Hygiene (certification #113133790) for chloroform, carbon tetrachloride and TCE Method TO-15 laboratory analysis. Chloroform was detected within sample VIP-10 at a concentration of 22.3 parts per billion vapor (ppbV) exceeding the WDNR Vapor Risk Screening Level (RSL) for Deep Soil Gas (see *Addressing Vapor Intrusion at Remediation & Redevelopment Sites in Wisconsin*, PUB-RR-800, December 2010). However, because this sample was collected beyond the extent of the PBG Plume, and based on its proximity to the Windings subdivision, the chloroform is likely the result of well chlorination activities. Analytical results also detected chloroform, carbon tetrachloride and/or TCE at concentrations below WDNR Vapor RSLs at borings VIP-1, VIP-5, VIP-8, and VIP-9. A summary of the analytical results is presented on Table 1. Laboratory reports and chain of custody records are attached.

Please note that due to the presence of isopropanol, analytical results for samples VIP-1, VIP-4, VIP-5, VIP-7, and VIP-9 are considered questionable. The presence of the isopropanol is due to the shortcircuiting of air through the hardware/fitting connections. Sample results for VIP-6 and VIP-8 were also flagged due to isopropanol related issues. However, according to Wisconsin State Laboratory of Hygiene personnel, the flag is the result of a failed quality control check and had no bearing on the results for the targeted compounds.

Other than the carbon tetrachloride concentrations identified in monitoring well PBN-9101C and borings VIP-8 and VIP-9, there does not appear to be a definitive groundwater to soil vapor contaminant correlation. Due to the short-circuiting of soil vapor in the five above-referenced borings, it is difficult to provide a reasonable correlative assessment between the soil vapor and most recent groundwater monitoring results shown in Table 2.

Based on the investigation results, which were inconclusive for half the samples due to ineffectively sealed fittings, BTS recommends that the Army conduct further investigation of this pathway at five locations. BTS suggests that soil gas samples be collected again, utilizing a refined sample collection procedure, at VIP-1, VIP-4, VIP-5, VIP-7 and VIP-9. VIP-10 will be added to confirm the level of chloroform at that location. Helium will be used as a leak detection tracer due to the ability of the laboratory to quantify this element. Additional steps will be taken to ensure that fittings are adequately sealed.

Please contact Clair Ruenger or myself if you have any questions.

Respectfully,

F. C. Constatt

F. A. Anstett Sr. Program Manager

BHB:dkf Att. a/s

let\_sitton\_041912\_Vapor Intrusion Investigation Report

Figures





#### Legend

Badger Army Ammunition Plant Boundary

- ▲ Vapor Intrusion Pathway Boring
- Monitoring Well
- Groundwater Plume
  - Paved Road

Existing Building

L\_\_\_\_ Former Building Footprint

Figure 2

Boring Location Map Vapor Intrusion Pathway Analysis Report Badger Army Ammunition Plant









D:\GIS\_Projects\BTS Projects\ Vapor Instrusion\MXDs\Fig\_1\_Vapor\_Inv Tables

#### Table 1

#### Soil Vapor Sample Analytical Results

#### Vapor Intrusion Pathway Analysis Report

#### **Badger Army Ammunition Plant**

Boring/Soil Vapor Sample ID	Location Description	Date Sampled	Sample Depth (feet bgs)	Chloroform	Carbon Tetrachloride	Trichloroethylene
VIP-1	Badger Army Ammuniton Plant	2/27/12	45.0	<0.085* (IS)	0.296* (IS)	<0.085* (IS)
VIP-2	Badger Army Ammuniton Plant	2/27/12	25.0	<0.085 (IS)	<0.085 (IS)	<0.085 (IS)
VIP-3	County Road Z Right of Way	2/27/12	45.0	<0.085 (IS)	<0.085 (IS)	<0.085 (IS)
VIP-4	County Road Z Right of Way	2/27/12	45.0	<0.085* (IS)	<0.085* (IS)	<0.085* (IS)
VIP-5	County Road Z Right of Way	2/27/12	45.0	0.441*^	<0.085*^	<0.085*^
VIP-6	State Highway 78 Right-of-Way	2/27/12	45.0	< 0.085^	<0.085^	< 0.085^
VIP-7	State Highway 78 Right-of-Way	2/28/12	45.0	<0.085*^	<0.085*^	<0.085*^
VIP-8	State Highway 78 Right-of-Way	2/28/12	45.0	0.159^ (J)	29.2^ (U)	0.316^
VIP-9	State Highway 78 Right-of-Way	2/28/12	45.0	<0.085*^	0.814*^	<0.085*^
VIP-10	State Highway 78 Right-of-Way	2/28/12	45.0	22.3^ (U)	<0.085^	<0.085^
Wisconsin	Department of Natural Resources - Va Soil Gas	22	64	38		

Results expressed in parts per billion vapor (ppbV)

Bold text identifies a vapor risk screening level exceedance.

bgs - Below ground surface

\* Due to the high amount of isopropanol in this sample (there was a leak in the probe setup), the results are questionable.

^ Because of residual isopropanol (IPA) in the instrumentation, the 4-bromofluorobenzene (BFB) tune check did not pass, so the results are approximate.

IS - The internal standard QC limit is exceeded.

J - Analytical result is between limit of detection (LOD) and limit of quantitation (LOQ).

U - Results are approximate, above upper calibration range.

#### Table 2

#### **Groundwater Analytical Results**

#### Vapor Intrusion Pathway Analysis Report

#### **Badger Army Ammunition Plant**

Boring/Soil Vapor Sample ID	Monitoring Well ID (nearest soil vapor sample)	Groundwater Depth (feet bgs)	Groundwater Monitoring Well Screen Interval (feet bgs)	Chloroform	Carbon Tetrachloride	Trichloroethylene
VIP-1	PBN-8501A	98.0	112.9-121.9	0.42 J	3.91	1.02
VIP-2	PBN-9902A	43.5	45.0-60.0	<0.10	<0.10	<0.10
VIP-3	SWN-9102C	76.7	142.5-152.5	< 0.10	<0.10	<0.10
VIP-4	SWN-9103B	79.1	103.4-113.4	1.58	14	1.13
VIP-5	SWN-9104C	79.3	154.0-164.0	0.47 J	1.28	<0.10
VIP-6	SWN-9105B	80.7	102.5-112.5	0.51	<0.10	<0.10
VIP-7	PBN-9102B	76.7	105.0-115.0	0.12 J	<0.10	<0.10
VIP-8	PBN-9101C	85.0	142.5-152.5	6.32	44.6	14.5
VIP-9	PBN-9101C	85.0	142.5-152.5	6.32	44.6	14.5
VIP-10	SWN-0502B	103.5	145.8-155.8	1.11	0.15 J	< 0.10

Results expressed in micrograms per liter (ug/l)

bgs - Below ground surface

J - Analytical result is between limit of detection (LOD) and limit of quantitation (LOQ).

Note - Groundwater monitoring activities were not conducted during the Vapor Intrusion Pathway Analysis project. The groundwater results presented in this table are associated with September and December 2011 quarterly groundwater monitoring at the Badger Army Ammunition Plant.

#### Laboratory Reports and Chain of Custody Records



# Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division		Orga	anic Chemistry	
WDNR LAB ID: 113133790 NELAP LAB ID: E3	87658	EPA LAB	WI00007	WI DATCP ID: 105-415
WSLH Sample	: O	W003671		
BADGER TECHNICAL SERVICES			Bill To	
1 BADGER ROAD			Billing ID:	7305879
BARABOO, WI 53913			Customer I TRACKING 2601 AGRI MADISON	D: 320225 3 4920 CULTURAL DRIVE WI 53718
Field #:			ID#:	
Collection Start: 02/27/2012 09:11:00			Point/Well:	Outrall ID:
Collection End:			Account #:	LH034
Collected By: JML-BADGER TECHNICAL			Project No:	
County:			Date Recei	ved: 03/02/2012 09:20:00
Sample Source: AIR			Date Repor	rted: 03/20/2012
Sample Depth:			Sample Re	ason:
Sample Information:				
Sample Location: VIP-1 @ BADGER ARMY AMM	MUNIT	ION PLANT	г	

Sample Description: 6 L CANISTER / PASSIVE CANISTER SAMPLER (CS1200E)

Analysis Date	Lab Comment				
03/07/2012	SEE OW003671.MM1	SEE OW003671.MM1			
Analysis Method	Result	Units	LOD	LOQ	Report Limit
CHLOROFORM	*IS ND	PPB V	0.085	0.280	
CARBON TETRACHLORIDE	*IS 0.296	PPB V	0.085	0.280	
TRICHLOROETHYLENE	*IS ND	PPB V	0.085	0.280	
OW003671.MM1:					
WISCONSIN STATE LABORATORY OF FOLLOWING FLAGS.	HYGIENE SAMPLE OW003	671 CONTAI	NS THE		
THE INTERNAL STANDARD QC LIMIT IS EXCEEDED - *IS.					
DUE TO THE HIGH AMOUNT OF ISOPROPANOL IN THIS SAMPLE, THERE WAS A LEAK IN THE PROBE SETUP, THEREFORE THE RESULTS ARE QUESTIONABLE.					



### Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division Organic Chemistry

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007 V

WI DATCP ID: 105-415

WSLI	H Sample:	OW003671				
Analysis Date 03/07/2012	Lab Comm	ent				
Analysis Method		Result	Units	LOD	LOQ	Report Limit
TOXIC ORGANIC COMPOUNDS IN AMBIENT AIR T015 - PREP	(	COMPLETE				1

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations: LOD = Level of detection LOQ = Level of quantification ND = None detected. Results are less than the LOD

Responsible Party: \_\_\_\_\_\_\_ Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.



# Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division	Organic Chemistry
WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EF	PA LAB WI00007 WI DATCP ID: 105-415
WSLH Sample: OW0	03672
BADGER TECHNICAL SERVICES	Bill To
1 BADGER ROAD	Billing ID: 7305879
	Customer ID: 320225
BARABOO, WI 53913	TRACKING 4920
	2601 AGRICULTURAL DRIVE
	MADISON WI 53718
	ID#:
Field #:	Waterbody/Outfall ID:
Collection Start: 02/27/2012 10:25:00	Point/Well:
Collection End:	Account #: LH034
Collected By: JML-BADGER TECHNICAL	Project No:
County:	Date Received: 03/02/2012 09:20:00
Sample Source: AIR	Date Reported: 03/20/2012
Sample Depth:	Sample Reason:
Sample Information:	
Sample Location: VIP-2 @ BADGER ARMY AMMUNITION	I PLANT
Sample Description: 6 L CANISTER / PASSIVE CANISTER S	SAMPLER (CS1200E)

Analysis Date 03/07/2012	Lab Comment THE INTERNAL STANDARD QC LIMIT IS EXCEEDED - *IS.				
Analysis Method	Result	Units	LOD	LOQ	Report Limit
CHLOROFORM	*IS ND	PPB V	0.085	0.280	
CARBON TETRACHLORIDE	*IS ND	PPB V	0.085	0.280	
TRICHLOROETHYLENE	*IS ND	PPB V	0.085	0.280	
Analysis Date 03/07/2012	Lab Comment				
Analysis Method	Result	Units	LOD	LOQ	Report Limit
TOXIC ORGANIC COMPOUNDS IN AMBIENT AIR T015 - PREP	COMPLETE				1



### Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division

Organic Chemistry

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007

WI DATCP ID: 105-415

WSLH Sample: OW003672

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations:

LOD = Level of detection

LOQ = Level of quantification ND = None detected. Results are less than the LOD

Responsible Party: <u>the Aup</u> Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.



# Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Healt	h Division	Orga	anic Chemistry	4
WDNR LAB ID: 113133790	NELAP LAB ID: E37658	EPA LAB	WI00007	WI DATCP ID: 105-415
	WSLH Sample: O	W003673		
BADGER TECHNICAL	SERVICES		Bill To	
1 BADGER ROAD			Billing ID	0: 7305879
BARABOO, WI 53913			Custome TRACKI 2601 AG MADISC	er ID: 320225 NG 4920 GRICULTURAL DRIVE DN WI 53718
Field #:			ID#: Waterbo	ody/Outfall ID:
Collection Start: 02/27/2012 12:	02:00		Point/We	ell:
Collection End:			Account	#: LH034
Collected By: JML-BADGER TE	ECHNICAL		Project I	No:
County:			Date Re	ceived: 03/02/2012 09:20:00
Sample Source: AIR			Date Re	ported: 03/20/2012
Sample Depth:			Sample	Reason:
Sample Information:				
Sample Location: VIP-3-CTY	Z ROW			
Sample Description: 6 L CANIS	TER / PASSIVE CANISTE	R SAMPLE	R (CS1200E)	

Analysis Date 03/07/2012	Lab Comment THE INTERNAL STANDARD QC LIMIT IS EXCEEDED - *IS.				
Analysis Method	Result	Units	LOD	LOQ	Report Limit
CHLOROFORM	*IS ND	PPB V	0.085	0.280	
CARBON TETRACHLORIDE	*IS ND	PPB V	0.085	0.280	
TRICHLOROETHYLENE	*IS ND	PPB V	0.085	0.280	
Analysis Date 03/07/2012	Lab Comment				
Analysis Method	Result	Units	LOD	LOQ	Report Limit
TOXIC ORGANIC COMPOUNDS IN AMBIENT AIR T015 - PREP	COMPLETE				1



### Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

**Organic Chemistry** 

Environmental Health Division

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007

WI DATCP ID: 105-415

WSLH Sample: OW003673

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations:

LOD = Level of detection

LOQ = Level of quantification ND = None detected. Results are less than the LOD

Responsible Party: \_\_\_\_\_\_ Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.

The results in this report apply only to the sample specifically listed above. This report is not to be reproduced except in full.

Report #: 9500946



# Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Healt	h Division	Orga	anic Chemistry	
WDNR LAB ID: 113133790	NELAP LAB ID: E37658	EPA LAB	WI00007 WI	DATCP ID: 105-415
	WSLH Sample: O	W003674		
BADGER TECHNICAL	SERVICES		Bill To	
1 BADGER ROAD			Billing ID:	7305879
BARABOO, WI 53913			Customer ID: TRACKING 49 2601 AGRICUI MADISON WI	320225 20 LTURAL DRIVE 53718
Field #: Collection Start: 02/27/2012 13: Collection End: Collected By: JML-BADGER TE	59:00 ECHNICAL		ID#: Waterbody/Our Point/Well: Account #: LH Project No:	tfall ID: 1034
County:			Date Received	: 03/02/2012 09:20:00
Sample Source: AIR			Date Reported	: 03/20/2012
Sample Depth:			Sample Reaso	n:
Sample Information:				
Sample Location: VIP-4-CTY	Z ROW			

#### Sample Description: 6 L CANISTER / PASSIVE CANISTER SAMPLER (CS1200E)

#### Analyses and Results:

Analysis Date 03/07/2012	Lab Comment SEE OW003674.MM1							
Analysis Method	Result	Units	LOD	LOQ	Report Limit			
CHLOROFORM	*IS ND	PPB V	0.085	0.280				
CARBON TETRACHLORIDE	*IS ND	PPB V	0.085	0.280				
TRICHLOROETHYLENE	*IS ND	PPB V	0.085	0.280				
OW003674.MM1:								
WISCONSIN STATE LABORATORY OF HYGIENE SAMPLE OW003674 CONTAINS THE FOLLOWING FLAGS.								
THE INTERNAL STANDARD QC LIMIT IS EXCEEDED - *IS.								
DUE TO THE HIGH AMOUNT OF ISC WAS A LEAK IN THE PROBE SETUR QUESTIONABLE.	DPROPANOL IN THIS SAMP P, THEREFORE THE RESUL	LE, THERE TS ARE						

IF YOU HAVE ANY QUESTIONS, CONTACT STEVE GEIS AT (608) 224-6269.



### Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division Organic Chemistry

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007 V

WI DATCP ID: 105-415

WSLI	H Sample:	OW003674				
Analysis Date 03/07/2012	Lab Comm	ent				
Analysis Method		Result	Units	LOD	LOQ	Report Limit
TOXIC ORGANIC COMPOUNDS IN AMBIENT AIR T015 - PREP	(	COMPLETE				1

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations: LOD = Level of detection LOQ = Level of quantification ND = None detected. Results are less than the LOD

Responsible Party: \_\_\_\_\_\_\_ Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.



# Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Healt	n Division	Orga	anic Chemistry	
WDNR LAB ID: 113133790	NELAP LAB ID: E37658	EPA LAB	WI00007	WI DATCP ID: 105-415
	WSLH Sample: O	W003675		
BADGER TECHNICAL	SERVICES		Bill To	
1 BADGER ROAD			Billing ID:	7305879
BARABOO, WI 53913			Customer TRACKIN 2601 AGR MADISON	ID: 320225 G 4920 CICULTURAL DRIVE I WI 53718
Field #:			ID#: Waterbody	//Qutfall ID:
Collection Start: 02/27/2012 15:	10:00		Point/Well	:
Collection End:			Account #	: LH034
Collected By: JML-BADGER TE	CHNICAL		Project No	):
County:			Date Rece	eived: 03/02/2012 09:20:00
Sample Source: AIR			Date Repo	orted: 03/20/2012
Sample Depth:			Sample Re	eason:
Sample Information:				
Sample Location: VIP-5-CTY	Z ROW			
Sample Description: 6 L CANIS	FER / PASSIVE CANISTE	R SAMPLE	R (CS1200E)	

Analysis Date	Lab Comment						
03/14/2012	SEE OW003675.MM1						
Analysis Method	Result	Units	LOD	LOQ	Report Limit		
CHLOROFORM	0.441	PPB V	0.085	0.280			
CARBON TETRACHLORIDE	ND	PPB V	0.085	0.280			
TRICHLOROETHYLENE	ND	PPB V	0.085	0.280			
OW003675.MM1:							
WISCONSIN STATE LABORATORY OF FOLLOWING FLAGS.	HYGIENE SAMPLE OW003	675 CONTAI	NS THE				
DUE TO THE HIGH AMOUNT OF ISOPROPANOL IN THIS SAMPLE, THERE WAS A LEAK IN THE PROBE SETUP, THEREFORE THE RESULTS ARE QUESTIONABLE.							
BECAUSE OF THE IPA LEAK, THE BFB TUNE CHECK DID NOT PASS SO THE RESULTS ARE APPROXIMATE.							
IF YOU HAVE ANY QUESTIONS, CO	NTACT STEVE GEIS AT (	608) 224-6	269.				



### Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division Organic Chemistry

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007 V

WI DATCP ID: 105-415

WSLI	H Sample:	OW003675				
Analysis Date 03/14/2012	Lab Comme	ent				
Analysis Method		Result	Units	LOD	LOQ	Report Limit
TOXIC ORGANIC COMPOUNDS IN AMBIENT AIR T015 - PREP	C	OMPLETE				1

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations: LOD = Level of detection LOQ = Level of quantification ND = None detected. Results are less than the LOD

Responsible Party: \_\_\_\_\_\_\_ Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.



## Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Healt	h Division	Orga	anic Chemistry	
WDNR LAB ID: 113133790	NELAP LAB ID: E37658	EPA LAB	WI00007	WI DATCP ID: 105-415
	WSLH Sample: O	W003676		
BADGER TECHNICAL	SERVICES		Bill To	
1 BADGER ROAD			Billing ID:	7305879
BARABOO, WI 53913			Customer II TRACKING 2601 AGRI MADISON	D: 320225 6 4920 CULTURAL DRIVE WI 53718
Field #:			ID#: Waterbody/	Outfall ID:
Collection Start: 02/27/2012 16:	25:00		Point/Well:	
Collection End: Collected By: JML-BADGER TE County:	ECHNICAL		Account #: Project No: Date Receiv	LH034 ved: 03/02/2012 09:20:00
Sample Source: AIR			Date Repor	ted: 03/20/2012
Sample Depth:			Sample Rea	ason:
Sample Information:				
Sample Location: VIP-6-HW	( 78 ROW			
Sample Description: 6 L CANIS	TER / PASSIVE CANISTE	R SAMPLE	R (CS1200E)	

Analysis Date	Lab Comment					
03/14/2012	SEE OW003676.MM1					
Analysis Method	Result	Units	LOD	LOQ	Report Limit	
CHLOROFORM	ND	PPB V	0.085	0.280		
CARBON TETRACHLORIDE	ND	PPB V	0.085	0.280		
TRICHLOROETHYLENE	ND	PPB V	0.085	0.280		
OW003676.MM1:						
WISCONSIN STATE LABORATORY OF FOLLOWING FLAGS.	HYGIENE SAMPLE OW00	3676 CONTAI	NS THE			
BECAUSE OF THE IPA LEAK, THE BFB TUNE CHECK DID NOT PASS SO THE RESULTS ARE APPROXIMATE.						
IF YOU HAVE ANY QUESTIONS, CO	NTACT STEVE GEIS AT	(608) 224-6	269.			



### Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division Organic Chemistry

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007 W

WI DATCP ID: 105-415

WSLI	H Sample:	OW003676				
Analysis Date 03/14/2012	Lab Comm	ent				
Analysis Method		Result	Units	LOD	LOQ	Report Limit
TOXIC ORGANIC COMPOUNDS IN AMBIENT AIR T015 - PREP	(	COMPLETE				1

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations: LOD = Level of detection LOQ = Level of quantification ND = None detected. Results are less than the LOD

Responsible Party: \_\_\_\_\_\_\_ Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.



# Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Healt	anic Chemistry			
WDNR LAB ID: 113133790	NELAP LAB ID: E37658	EPA LAB	WI00007	WI DATCP ID: 105-415
	WSLH Sample: O	W003677		
BADGER TECHNICAL	SERVICES		Bill To	
1 BADGER ROAD			Billing ID:	7305879
BARABOO, WI 53913			Customer ID TRACKING 2601 AGRIC MADISON	0: 320225 4920 CULTURAL DRIVE WI 53718
Field #:			ID#: Waterbody/0	Outfall ID:
Collection Start: 02/28/2012 08:	20:00		Point/Well:	
Collection End:			Account #:	LH034
Collected By: JML-BADGER TE	ECHNICAL		Project No:	
County:			Date Receiv	red: 03/02/2012 09:20:00
Sample Source: AIR			Date Report	ed: 03/20/2012
Sample Depth:			Sample Rea	ison:
Sample Information:				
Sample Location: VIP-7-HW	( 78 ROW			
Sample Description: 6 L CANIS	TER / PASSIVE CANISTE	R SAMPLE	R (CS1200E)	

Analysis Date	Lab Comment						
03/14/2012	SEE OW003677.MM1						
Analysis Method	Result	Units	LOD	LOQ	Report Limit		
CHLOROFORM	ND	PPB V	0.085	0.280			
CARBON TETRACHLORIDE	ND	PPB V	0.085	0.280			
TRICHLOROETHYLENE	ND	PPB V	0.085	0.280			
OW003677.MM1:							
WISCONSIN STATE LABORATORY OF HYGIENE SAMPLE OW003677 CONTAINS THE FOLLOWING FLAGS.							
DUE TO THE HIGH AMOUNT OF ISOPROPANOL IN THIS SAMPLE, THERE WAS A LEAK IN THE PROBE SETUP, THEREFORE THE RESULTS ARE QUESTIONABLE.							
BECAUSE OF THE IPA LEAK, THE BFB TUNE CHECK DID NOT PASS SO THE RESULTS ARE APPROXIMATE.							
IF YOU HAVE ANY QUESTIONS, CO	IF YOU HAVE ANY QUESTIONS, CONTACT STEVE GEIS AT (608) 224-6269.						



### Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division Organic Chemistry

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007 V

WI DATCP ID: 105-415

WSLI	H Sample:	OW003677				
Analysis Date 03/14/2012	Lab Comm	ent				
Analysis Method		Result	Units	LOD	LOQ	Report Limit
TOXIC ORGANIC COMPOUNDS IN AMBIENT AIR T015 - PREP	(	COMPLETE				1

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations: LOD = Level of detection LOQ = Level of quantification ND = None detected. Results are less than the LOD

Responsible Party: \_\_\_\_\_\_\_ Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.



# Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division Organic				
WDNR LAB ID: 113133790	NELAP LAB ID: E37658	EPA LAB	WI00007	WI DATCP ID: 105-415
	WSLH Sample: O	W003678		
BADGER TECHNICAL	SERVICES		Bill To	
1 BADGER ROAD			Billing ID:	7305879
			Customer	ID: 320225
BARABOO, WI 53913			TRACKIN	G 4920
			2601 AGF	RICULTURAL DRIVE
			MADISON	N WI 53718
			ID#:	
Field #:			Waterbod	y/Outfall ID:
Collection Start: 02/28/2012 10:	00:00		Point/Wel	l:
Collection End:			Account #	: LH034
Collected By: JML-BADGER TE	ECHNICAL		Project No	D:
County:			Date Rece	eived: 03/02/2012 09:20:00
Sample Source: AIR			Date Repo	orted: 03/20/2012
Sample Depth:			Sample R	eason:
Sample Information:				
Sample Location: VIP-8 HWY	′ 78 RPW			
Sample Description: 6 L CANIS	TER / PASSIVE CANISTE	R SAMPLE	R (CS1200E)	

Analysis Date	Lab Comment SEE OW003678 MM1				
Analysis Method	Result	Units	LOD	LOQ	Report Limit
CHLOROFORM	0.159	PPB V	0.085	0.280	-
Note: The reported value above	is equal to or greater than the	LOD and les	s than the l	LOQ.	
CARBON TETRACHLORIDE	*U 29.2	PPB V	0.085	0.280	
TRICHLOROETHYLENE	0.316	PPB V	0.085	0.280	



AMBIENT AIR T015 - PREP

Wisconsin State Laboratory of Hygiene 2601 Agriculture Drive, PO Box 7996 Madison, WI 53707-7996 (800)442-4618 • FAX (608)224-6213 http://www.slh.wisc.edu

### Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division

**Organic Chemistry** 

NELAP LAB ID: E37658 EPA LAB WI00007 WDNR LAB ID: 113133790 WI DATCP ID: 105-415

WSLH Sample: OW003678							
OW003678.MM1:							
WISCONSIN STATE LABORATORY OF HY FOLLOWING FLAGS.	GIENE SAMPLE OW003	678 CONTAI	INS THE				
RESULTS ARE APPROXIMATE, ABOVE U	JPPER CALIBRATION F	ANGE - *U.					
BECAUSE OF THE IPA LEAK, THE BFE SO THE RESULTS ARE APPROXIMATE.	3 TUNE CHECK DID NO	T PASS					
IF YOU HAVE ANY QUESTIONS, CONTA	ACT STEVE GEIS AT (	608) 224-6	269.				
Analysis Date La 03/14/2012	b Comment						
Analysis Method	Result	Units	LOD	LOQ	Report Limit		
TOXIC ORGANIC COMPOUNDS IN	COMPLETE				1		

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations: LOD = Level of detection LOQ = Level of quantification ND = None detected. Results are less than the LOD

Responsible Party: \_\_\_\_\_\_ Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.



# Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health	Division	anic Chemistry		
WDNR LAB ID: 113133790	NELAP LAB ID: E37658	EPA LAB	WI00007	WI DATCP ID: 105-415
	WSLH Sample: 0	W003679		
BADGER TECHNICAL	SERVICES		Bill To	
1 BADGER ROAD			Billing ID	: 7305879
			Custome	r ID: 320225
BARABOO, WI 53913			TRACKIN	NG 4920
			2601 AG	RICULTURAL DRIVE
			MADISO	N WI 53718
			ID#:	
Field #:			Waterboo	dy/Outfall ID:
Collection Start: 02/28/2012 11:3	2:00		Point/We	II:
Collection End:			Account	#: LH034
Collected By: JML-BADGER TE	CHNICAL		Project N	lo:
County:			Date Rec	ceived: 03/02/2012 09:20:00
Sample Source: AIR			Date Rep	oorted: 03/20/2012
Sample Depth:			Sample F	Reason:
Sample Information:				
Sample Location: VIP-9-HWY	78 ROW			
Sample Description: 6 L CANIST	ER / PASSIVE CANIST	ER SAMPLE	R (CS1200E)	

Analysis Date 03/14/2012	Lab Comment SEE OW003679.MM1				
Analysis Method	Result	Units	LOD	LOQ	Report Limit
CHLOROFORM	ND	PPB V	0.085	0.280	
CARBON TETRACHLORIDE	0.814	PPB V	0.085	0.280	
TRICHLOROETHYLENE	ND	PPB V	0.085	0.280	


## Laboratory Report

MD M. P. ID. - - -D DIL D' DEV .

		D	.F. Kurtycz, M.D., I	Medical Director	• Charles D. B	rokopp, Dr.P.H., Direc
Environmental Healt	h Division	Org	anic Chemistry	,		
/DNR LAB ID: 113133790	NELAP LAB ID: E376	58 EPA LAB	WI00007	WI DAT(	CP ID: 105-	-415
	WSLH Sample:	OW003679				
OW003679.MM1:						
WISCONSIN STATE LABORA FOLLOWING FLAGS.	ATORY OF HYGIENE	SAMPLE OW00	3679 CONTA:	INS THE		
DUE TO THE HIGH AMOUNT WAS A LEAK IN THE PROF QUESTIONABLE.	[ OF ISOPROPANOL 3E SETUP, THEREFO	IN THIS SAM RE THE RESU	PLE, THERE LTS ARE			
BECAUSE OF THE IPA LEA SO THE RESULTS ARE APP	AK, THE BFB TUNE PROXIMATE.	CHECK DID N	OT PASS			
IF YOU HAVE ANY QUESTI	IONS, CONTACT STE	VE GEIS AT	(608) 224-6	5269.		
Analysis Date 03/14/2012	Lab Comm	ent				
Analysis Method		Result	Units	LOD	LOQ	Report Limit
TOXIC ORGANIC COMPO AMBIENT AIR T015 - PREF	UNDS IN (	COMPLETE				1

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations: LOD = Level of detection LOQ = Level of quantification ND = None detected. Results are less than the LOD

Responsible Party: \_\_\_\_\_\_ Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.



## Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Healt	h Division	Orga	anic Chemistry	
WDNR LAB ID: 113133790	NELAP LAB ID: E37658	EPA LAB	WI00007 WI	DATCP ID: 105-415
	WSLH Sample: O	W003680		
BADGER TECHNICAL	SERVICES		Bill To	
1 BADGER ROAD			Billing ID: 7	305879
BARABOO, WI 53913			Customer ID: TRACKING 492 2601 AGRICUL MADISON WI	320225 20 TURAL DRIVE 53718
Field #: Collection Start: 02/28/2012 13: Collection End:	27:00		ID#: Waterbody/Outf Point/Well: Account #: LH	fall ID: 1034
Collected By: JML-BADGER TE	CHNICAL		Project No:	03/02/2012 09:20:00
Sample Source: AIR Sample Depth: Sample Information:			Date Reported: Sample Reasor	03/20/2012 n:
Sample Location: VIP-10-HW	'Y 78 ROW			

#### Sample Description: 6 L CANISTER / PASSIVE CANISTER SAMPLER (CS1200E)

#### Analyses and Results:

Analysis Date	Lab Comment							
03/14/2012	SEE OW003680.MM1							
Analysis Method	Result	Units	LOD	LOQ	Report Limit			
CHLOROFORM	*U 22.3	PPB V	0.085	0.280				
CARBON TETRACHLORIDE	ND	PPB V	0.085	0.280				
TRICHLOROETHYLENE	ND	PPB V	0.085	0.280				
OW003680.MM1:								
WISCONSIN STATE LABORATORY OF HYGIENE SAMPLE OW003680 CONTAINS THE FOLLOWING FLAGS.								
RESULTS ARE APPROXIMATE, ABOVE UPPER CALIBRATION RANGE - *U.								
BECAUSE OF THE IPA LEAK, THE SO THE RESULTS ARE APPROXIMAT	BFB TUNE CHECK DID NO E.	T PASS						
IF YOU HAVE ANY QUESTIONS, CO	NTACT STEVE GEIS AT (	608) 224-6	269.					



## Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division Organic Chemistry

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007 V

WI DATCP ID: 105-415

WSLI	- Sample:	OW003680				
Analysis Date 03/14/2012	Lab Comm	ent				
Analysis Method		Result	Units	LOD	LOQ	Report Limit
TOXIC ORGANIC COMPOUNDS IN AMBIENT AIR T015 - PREP	(	COMPLETE				1

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations: LOD = Level of detection LOQ = Level of quantification ND = None detected. Results are less than the LOD

Responsible Party: \_\_\_\_\_\_\_ Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.

Generation     Sample Collector(s) Name       JEFFLARK(N       Property Owner       Property       Prope       Property <tr< th=""><th>PO # Return Report A:</th><th></th><th></th><th>·</th><th>Chain of</th><th></th><th>ľ.</th></tr<>	PO # Return Report A:			·	Chain of		ľ.
Sample Collector(s) Name JEFFLARK(N Property Owner Property Owner Property Owner Property Owner Split Samples: Offened? Types No. of Field ID No. Data Time Container	Refum Report A:	1121000212.					
Property Owner Deporturent tof 42 e Arr Split Samples: Offend? Tres NK Accepted? Tres NK Accepted? Tres NK. of Field ID No. Date Time Contained		\s: (select one) Email or Postal Address   Hand Conv _ je そそんセッ・  ロル Ki ላ @ S Δ e C	- PAD - 14 C. C	MO	Phone N	umber (inclu	la al
Split Samples: Offered? Tyes Ne Ne Split Samples: Offered? Yes Ne Ne Stell ID No. Data Time Contained	J.	Property Address	the Plan	+	Phone N	umber (inclu	de al
Accepted? Tyes No. of Field ID No. Date Time Contained		Baraboo, WI 53913	- 5000				
Field ID No. Data Time Container	Accepted By (Signal	thure):			l de l		
-1160	0	Station Location Sammle Description	Lab (D	Cracked /	Improperty	Good	ľ
	Badger Arn	my Aumuri Her Plant	INUINDEL		Sealed	Condition	อี
1 = ++07211/18/C T	For: Corbo	Atetrachloride, chloroform +					
112-20 2 12 1025- 1	Endrer Army /	Annunition Plant For: corben Chlorocount TCE analyses (To-15)					
-2021	CHY & ROL	W For: corbou talmachloride,					
VI-7-5 21/21/2 2-41/	chlonoferm +	+ TCE onalyses (TO-15)					
-1324-	CHJ Z ROU	2 Por: corbou tetrachloride,					
I Peri Siltele H-4/1	chlorophim	+ TCE onelyses (TO-15)					•
111 D_ 6 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	CHy Z Rou	O For! carbon telhachlonide,				-	
	Chloveform	1 + TCE enclysos (PO-15)					
VIP- ( ) - ( ) - ( )	Huy 78 Roc	W Por: corbon tetrachlonide,					
VII 6 0/0/11C [706	chlorotorum	1+ TCE omilyses (no-15)			·		
Method of Shipment:	for Sample Collection		Was the sample shi	ipping contain	ier sealed on i	eceipt?	Xes Ves
	nhydrous Ammonia Spi	xil Desticide Spill * - Specify Pest	icide:				
	nimal Waste	🔲 Hazardous Waste Release *					
	oen Burning	🛄 Petroleum Product Release * -	Specify Product:				
	airy Product Spill	Industrial Spill/Runoff * - Speci	fy Industry Type:				
	ensuration/Storm Wate act the leborations with :	Ber Runoff L Other Specify:					
hereby cartify that I received and monochy hour	Inter Company and the	Provide initiation and for consumation. Also, inc	lude sample of sus	spected spill	ed product.		
Relinition Brind Ru (Simatura)	ured upese samples as	s noted below:		Disp	osition of Ur	nused Portic	0 5
all all		Received By (Signature)	Date / Tim	2		. 950	
Relinquished By (Signature)	/IJS/Date / Time	Received By (Strinstruct)		2		E	
			Date / Tim			in until furth	ler no
Kelinquished By (Signature)	Date / Time	Received for Laboratory By (Signature)	Date / Tim		₽ E	-	

**.** 

	•						1090	с Г	1
State of Wisco Department of	nsin Natural Re	sources		PO# BTS	0001344		Chain of Form 4100-1	f Custod 145 (R 3/09)	y Record
Sample Collector	(e) Name RK(Z			Return Report A:	s: (select are) Email ar Postal Address, Hard Copy 」」こそぞうでく、「ロットドッの Sp Ei	cpro-inc.com	Phone (& C	Number (inclu)	de area code) -SS7 (
Property Owner De porting	ento.	t tr	e Arm	5	Property Address Bacfur Bruny Ammun	itten Plant	Phone	Number (inch	ide area code)
Split Samples:	Offered?	Ó	2 [] []		Banaboo, WI 53913	0005-1			
	Accepted:	ן בא ג	۶ ۲	Accepted By (Signar	ture):			Use Only	
Field ID No.	Date	Time	No. of Containers		Station Location Sample Description	Lab ID Crack Number Brot	ed / Improperty	V Good	Other
VIP-7	21/96/6	5160		Hwy 70 Ra	ud For: corbanteknehlovido, + TCE onglyses (70-15)				
VIP-B	21/80/0	1000- 1032	-	NWY 78 6	COLD For: corbon te Kachlonide				
VIP-9	zilaele	1132-	-	Huy 78 R	ow For: corbon tetrichbide				
VIPIO	zilaele	-5251	-	Hwy 70 RC	200 For. Cothan Yeyna chloride				
			,						
Method of Shi	oment	]	Reason f	or Sample Collection		Was the sample shipping .	container sealed o	n receipt?	Vae No.
	ti Centino		₹ □[	hydrous Ammonia Sp	ili 🗌 Pesticide Spill * - Specify Pes	ticide:			
			ة ة 100	imal Waste en Buming	Li Hazardous Waste Release *	- Specify Product			
	ecify:		8 8 	iiry Product Spill Instruction/Storm Watu	Industrial Spiil/Runoff *- Spa Runoff Dather - Specify;	dify industry Type:			
			• Conta	act the laboratory with	product information and for consultation. Also, in	clude sample of suspecto	d spilled product.		
I Nereby Certify I	nat I racein	ad and bu	operty hand	led these samples as	noted below:		Disposition of	Unused Parl	ion Sample:
TYP: 5			11/6		Received By (Signature)	Date / Time		esods	-
Relinquished By (	(Signature)			Date / Time	Received By (Signature)	Date / Time	<u>ٹ</u> 1	atum atain until fur	ther motice
Relinquished By (	(Signature)			Date / Time	Received for Laboratory By (Signature)	Date / Time		ther	
							use the bac	accuration of the for	oom tor hotes, m.

C Q

63



16 July 2012

Contracting Officer's Representative Badger Army Ammunition Plant 2 Badger Road Baraboo, WI 53913-5000

Subject: Supplemental Vapor Intrusion Pathway Analysis Report

Dear Sir:

On April 19, 2012, Badger Technical Services, LLC (BTS) submitted the *Vapor Intrusion Pathway Analysis Report* to the Department of the Army. The investigation included conducting ten Geoprobe<sup>®</sup> borings to a depth half way to the water table depth and collecting a soil gas sample at that depth. The purpose of the investigation was to determine the risk to human receptors should vapors from the Propellant Burning Ground (PBG) Plume partition from the groundwater to the soil. The PBG plume is the one of three plumes migrating from the Badger Army Ammunition Plant and is the only one plume associated with contaminants (volatile organic compounds) with the potential for vapor pathway concerns.

A complete evaluation of the vapor pathway was not possible based on the initial information because several of the soil gas samples reported elevated levels of the leak detection tracer gas. Therefore, BTS proposed the locations where the soil gas analytical data were deemed invalid be re-sampled using helium as the leak-detection tracer gas using an alternate leak detection methodology. BTS performed this work under Performance Work Statement (PWS) #0067.

The previous vapor intrusion pathway analysis resulted in five soil gas sample locations (VIP-1, VIP-4, VIP-5, VIP-7, and VIP-9) considered to have questionable analytical results due to the presence of isopropyl alcohol, the leak detection tracer gas, in the samples. In addition, vapor intrusion probe boring VIP-10 was inadvertently omitted from the leak detection methodology and the sample also reported an elevated level of chloroform. These six locations were selected to be re-sampled to either obtain reliable analytical data and/or to confirm the previous analytical result. Figure 1 shows the PBG Plume in relation to the Badger Army Ammunition Plant (BAAAP) and the locations of the initial ten probe borings with corresponding well locations.

On June 27, 2012, Probe Technologies of Palmyra, Wisconsin advanced four Geoprobe<sup>®</sup> borings to a depth of approximately 40 feet below grade (bg). These probe boring locations were VIP-4A, VIP-5A, VIP-7A, and VIP-9A. The Geoprobe<sup>®</sup> encountered refusal at approximately 15 to 20 feet bg preventing the advancement and sample collection of VIP-1A and VIP-10A. Several attempts were made in the

Contracting Officer's Representative 16 July 2012 Page 2 of 3

general area to penetrate a denser layer, but were not successful. On July 3, 2012, Soil Essentials of New Glarus, Wisconsin advanced the probe borings VIP-1A and VIP-10A to 45 feet bg with a more powerful Geoprobe<sup>®</sup>.

After drilling the probe boring to the desired depth, new high-density polyethylene 3/8-inch or 1/4-inch OD tubing was placed down to the bottom of the borehole (post-run tubing vapor sampling technique) and threaded onto the tip of the hollow steel drill rods. The down-hole tubing connects to a laboratorysupplied flow controller/sampler attached to a laboratory-supplied 6-liter summa canister. To ensure sample integrity, the following quality control leak detection tracer technique was used prior to sample collection. The flow controller/sampler with vinyl tubing connected to each end was placed inside an airtight container. Helium gas was injected into the airtight container. A helium gas detector (RadioDetection MGD-2002) was inserted into the tubing connected to the flow controller/sampler and the other end of the tubing was left open to the ambient air. BTS personnel monitored the helium detector to see whether any leaking or short-circuiting of air through the fittings was occurring. None of the tracer tests indicated leaks in the fittings. The flow controller/sampler was then connected to the summa canister, and the tubing was connected to the down-hole tubing with hose clamps. The valve on the summa canister was opened and under negative pressure, the canister was allowed to draw vapor from the bottom of the borehole until filled, approximately 30 minutes. The valve was then closed on the summa canister and the flow controller removed. Soil borings were abandoned with bentonite chips following sample collection.

Soil vapor samples were submitted to the Wisconsin State Laboratory of Hygiene (certification #113133790) for chloroform, carbon tetrachloride, and trichloroethylene (TCE) Method TO-15 laboratory analysis. Carbon tetrachloride was detected within sample VIP-1A at a concentration of 68.3 parts per billion vapor (ppbV) exceeding the WDNR Vapor Risk Screening Level (RSL) for Deep Soil Gas of 64 ppbV (see *Addressing Vapor Intrusion at Remediation & Redevelopment Sites in Wisconsin*, PUB-RR-800, December 2010). Analytical results also reported concentrations below WDNR Vapor RSLs at VIP-9A and VIP-10A. Carbon tetrachloride was reported at 0.563 ppbV at VIP-9A, and at VIP-10A, carbon tetrachloride and TCE were reported at concentrations of 25.5 ppbV and 1.08 ppbV, respectively. Chloroform was not detected at VIP-10A. It should be noted that the property closest to this data point has an in-ground swimming pool which, along with potential of private well chlorination impacts could be source(s) of the chloroform previously detected. Table 1 provides a summary of the analytical results. Laboratory reports and chain of custody records are attached.

There does not appear to be a strong correlation between groundwater concentrations and soil gas for the selected volatile organic compounds. Table 2 provides the groundwater analytical data for the three contaminants of concern and groundwater depth information for the shallow monitoring wells nearest to the probe boring locations. The well location with the highest groundwater contamination is PBN-9101C with carbon tetrachloride at 44.6 micrograms per liter ( $\mu$ g/l), which is in proximity to VIP-8 and VIP-9A. Carbon tetrachloride in soil gas was reported at 29.2 ppbV at VIP-8 and only 0.563 ppbV at VIP-9A. The soil gas sample with the highest reported concentration of carbon tetrachloride was VIP-1A with 68.3 ppbV; however, the carbon tetrachloride groundwater concentration at that location (PBN-8501A) is only 3.91  $\mu$ g/l. VIP-10A is located outside the PBG Plume; however, carbon tetrachloride was reported at 25.5 ppbV at that location.

There seems to be more of a correlation between density of lithology and soil gas concentrations. Probe borings that encountered difficulty in penetrating at shallow depth (15 to 20 feet bg) were the locations where the soil gas concentrations were highest. As stated previously, VIP-1A and VIP-10A required a

Contracting Officer's Representative 16 July 2012 Page 3 of 3

more powerful Geoprobe<sup>®</sup> to penetrate to total depth. The more dense lithologic layer could be acting as a confining layer, concentrating the low-level vapors below the dense layer and preventing the soil gas from diffusing to the surface. It should be noted, valid concentrations reported off-site are still below any WDNR Vapor RSLs.

The supplemental vapor intrusion pathway analysis filled the data gaps and provided reliable data to adequately evaluate the vapor pathway from the PBG Plume. Based on the findings of this analysis and the previous investigation, the data indicate that the PBG Plume does not present a significant risk to human health via vapor intrusion off-site. Analytical results of soil gas samples collected off-site do not exceed the WDNR Vapor RSLs for Deep Soil Gas.

Please contact Clair Ruenger or myself if you have any questions.

Respectfully,

F. C. Constat

F. A. Anstett Sr. Program Manager

BHB:dkf Att. a/s

let\_sitton\_071612\_Supplemental VIPA Report.doc

Figures





#### Legend

Badger Army Ammunition Figure 2 Plant Boundary  $\star$ Supplemental Vapor Intrusion Probe (VIP-1A) **Boring Location Map** Vapor Intrusion Probe (VIP-1) ▲ Supplemental Vapor Intrusion Pathway Analysis Report Monitoring Well • **Badger Army Ammunition Plant** Groundwater Plume 1 inch = 1,042 feet Paved Road Existing Building Feet D:\GIS\_Projects\BTS Projects\ Vapor Instrusion\Supplemental VIPA\Fig\_2\_Vapor\_Investigation 1,400 0 2,800 L\_\_\_\_ Former Building Footprint

Tables

#### Table 1

#### Soil Vapor Sample Analytical Results

#### Supplemental Vapor Intrusion Pathway Analysis Report

#### **Badger Army Ammunition Plant**

Boring/Soil Vapor Sample ID	Location Description	Date Sampled	Sample Depth (feet bgs)	Chloroform	Carbon Tetrachloride	Trichloroethylene
VIP-1	Badger Army Ammuniton Plant	2/27/12	45.0	<0.085* (IS)	0.296* (IS)	<0.085*(IS)
VIP-1A	Badger Army Ammuniton Plant	7/3/12	45.0	< 0.085	68.3	<0.694 (I)
VIP-2	Badger Army Ammuniton Plant	2/27/12	25.0	<0.085 (IS)	<0.085 (IS)	<0.085 (IS)
VIP-3	County Road Z Right-of-Way	2/27/12	45.0	<0.085 (IS)	<0.085 (IS)	<0.085 (IS)
VIP-4	County Road Z Right-of-Way	2/27/12	45.0	<0.085* (IS)	<0.085* (IS)	<0.085* (IS)
VIP-4A	County Road Z Right-of-Way	6/27/12	37.0	< 0.085	< 0.085	<0.085
VIP-5	County Road Z Right-of-Way	2/27/12	45.0	0.441*^	<0.085*^	<0.085*^
VIP-5A	County Road Z Right-of-Way	6/27/12	40.0	< 0.085	< 0.085	< 0.085
VIP-6	State Highway 78 Right-of-Way	2/27/12	45.0	<0.085^	<0.085^	<0.085^
VIP-7	State Highway 78 Right-of-Way	2/28/12	45.0	<0.085*^	<0.085*^	<0.085*^
VIP-7A	State Highway 78 Right-of-Way	6/27/12	37.0	< 0.085	< 0.085	< 0.085
VIP-8	State Highway 78 Right-of-Way	2/28/12	45.0	0.159^ (J)	29.2^ (U)	0.316^
VIP-9	State Highway 78 Right-of-Way	2/28/12	45.0	<0.085*^	0.814*^	<0.085*^
VIP-9A	State Highway 78 Right-of-Way	6/27/12	39.0	< 0.085	0.563	<0.085
VIP-10	State Highway 78 Right-of-Way	2/28/12	45.0	22.3^ (U)	<0.085^	<0.085^
VIP-10A	State Highway 78 Right-of-Way	7/3/12	45.0	< 0.085	25.5	1.08
Wisconsin	Department of Natural Resources - Va Soil Gas	apor Risk Screening	g Levels for Deep	22	64	38

Results expressed in parts per billion vapor (ppbV)

Bold text identifies a vapor risk screening level exceedance.

bgs - Below ground surface

\* Due to the high amount of isopropanol in this sample (there was a leak in the probe setup), the results are questionable.

^ Because of residual isopropanol (IPA) in the instrumentation, the 4-bromofluorobenzene (BFB) tune check did not pass, so the results are approximate.

IS - The internal standard QC limit is exceeded.

J - Analytical result is between limit of detection (LOD) and limit of quantitation (LOQ).

U - Results are approximate, above upper calibration range.

Italics indicate previously reported data.

I - elevated detection limit due to matrix interference

### Table 2

### **Groundwater Analytical Results**

### Supplemental Vapor Intrusion Pathway Analysis Report

### **Badger Army Ammunition Plant**

Boring/Soil Vapor Sample ID	Monitoring Well ID (nearest soil vapor sample)	Groundwater Depth (feet bgs)	Groundwater Monitoring Well Screen Interval (feet bgs)	Chloroform	Carbon Tetrachloride	Trichloroethylene
VIP-1, VIP-1A	PBN-8501A	98.0	112.9-121.9	0.42 J	3.91	1.02
VIP-2	PBN-9902A	43.5	45.0-60.0	<0.10	< 0.10	<0.10
VIP-3	SWN-9102C	76.7	142.5-152.5	<0.10	< 0.10	<0.10
VIP-4, VIP-4A	SWN-9103B	79.1	103.4-113.4	1.58	14	1.13
VIP-5, VIP-5A	SWN-9104C	79.3	154.0-164.0	0.47 J	1.28	<0.10
VIP-6	SWN-9105B	80.7	102.5-112.5	0.51	< 0.10	<0.10
VIP-7, VIP-7A	PBN-9102B	76.7	105.0-115.0	0.12 J	< 0.10	<0.10
VIP-8	PBN-9101C	85.0	142.5-152.5	6.32	44.6	14.5
VIP-9, VIP-9A	PBN-9101C	85.0	142.5-152.5	6.32	44.6	14.5
VIP-10, VIP-10A	SWN-0502B	103.5	145.8-155.8	1.11	0.15 J	<0.10

Results expressed in micrograms per liter (ug/l)

bgs - Below ground surface

J - Analytical result is between limit of detection (LOD) and limit of quantitation (LOQ).

Note - Groundwater monitoring activities were not conducted during the Vapor Intrusion Pathway Analysis project. The groundwater results presented in this table are associated with September and December 2011 quarterly groundwater monitoring at the Badger Army Ammunition Plant.

## Laboratory Reports and Chain of Custody Records



# Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Heal	th Division	Orga	anic Chemistry	
WDNR LAB ID: 113133790	NELAP LAB ID: E37658	EPA LAB	WI00007	WI DATCP ID: 105-415
	WSI H Sample: O	X000053		
	WOLH Gample. O	A000033		
BADGER TECHNICAL	. SERVICES		Bill To	
1 BADGER RD.			Billing ID:	7305879
			Customer	ID: 320225
BARABOO WI 33913			TRACKIN	G 4920
			2601 AGF	RICULTURAL DRIVE
			MADISON	I WI 53718
			ID#:	
Field #:			Waterbod	y/Outfall ID:
Collection Start: 07/03/2012 08	:26:00		Point/Well	:
Collection End: 07/03/2012 08	56:00		Account #	: LH034
Collected By: BHB-BADGER T	ECH. SVC		Project No	):
County:			Date Rece	eived: 07/05/2012
Sample Source: AIR			Date Repo	orted: 07/10/2012
Sample Depth: 45'			Sample R	eason:
Sample Information: SAMPLER	R 5473			
Sample Location: VIP-1A BA	DGER ARMY AMMUNITI	ON PLANT		
Sample Description: 6L CANIS	TER/PASSIVE SAMPLER			
Analyses and Results:				

Analysis Date	Lab Comment				
07/09/2012 10:48:45	INTERFERENCE INDICATED	BY *I.			
Analysis Method	Result	Units	LOD	LOQ	Report Limit
CHLOROFORM	ND	PPB V	0.085	0.280	
CARBON TETRACHLORIDE	68.3	PPB V	0.085	0.280	
TRICHLOROETHYLENE	*I< 0.694	PPB V	0.085	0.280	





D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division

Organic Chemistry

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007

WI DATCP ID: 105-415

WSLH Sample: OX000053

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations:

LOD = Level of detection LOQ = Level of quantification ND = None detected. Results are less than the LOD

Responsible Party: \_\_\_\_\_\_ Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.



# Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007 WI DATCP ID: 105-415  WSLH Sample: OW004797 BADGER TECHNICAL SERVICES 1 BADGER RD BARABOO, WI 53913 BARABOO, WI 53913 BARABOO, WI 53913 BARABOO, WI 53913 BARABOO, WI 53718 BIH To B
WSLH Sample:       OW004797         BADGER TECHNICAL SERVICES       Bill To         1 BADGER RD       Billing ID: 7305879         BARABOO, WI 53913       Customer ID: 320225         TRACKING 4920       2601 AGRICULTURAL DRIVE         MADISON WI 53718       ID#:         Field #:       Waterbody/Outfall ID:
WSLH Sample: OW004797 BADGER TECHNICAL SERVICES Bill To DIBADGER RD BARABOO, WI 53913 BARABOO, WI 5391
BADGER TECHNICAL SERVICES Bill To 1 BADGER RD Billing ID: 7305879 Customer ID: 320225 BARABOO, WI 53913 TRACKING 4920 2601 AGRICULTURAL DRIVE MADISON WI 53718 ID#: Field #: 00/07/0010 00:0000
1 BADGER RD       Billing ID: 7305879         Customer ID: 320225         BARABOO, WI 53913         TRACKING 4920         2601 AGRICULTURAL DRIVE         MADISON WI 53718         ID#:         Field #:       Waterbody/Outfall ID:
BARABOO, WI 53913 BARABOO, WI 53913 Customer ID: 320225 TRACKING 4920 2601 AGRICULTURAL DRIVE MADISON WI 53718 ID#: Field #: Waterbody/Outfall ID: Callection Ot/07/0040 00:00:00
BARABOO, WI 53913       TRACKING 4920         2601 AGRICULTURAL DRIVE         MADISON WI 53718         ID#:         Field #:       Waterbody/Outfall ID:         Oallastian Otath
Field #: 00/07/0010 00:00:00
Field #: Waterbody/Outfall ID:
Field #: Uterto 00/07/0010 00:00:00
Field #: Waterbody/Outfall ID:
Collection Start: 06/27/2012 08:20:00 Point/Well:
Collection End: 06/27/2012 08:45:00 Account #: LH034
Collected By: BHB Project No:
County: Date Received: 06/29/2012 11:12:00
Sample Source: AIR Date Reported: 07/09/2012
Sample Depth: Sample Reason:
Sample Information:
Sample Location: VIP-4A BADGER ARMY AMMUNITION PLANT
Sample Description: 6 L CANISTER / PASSIVE SAMPLER
Analyses and Results:

Analysis Date 07/06/2012	Lab Comment				
Analysis Method	Result	Units	LOD	LOQ	Report Limit
CHLOROFORM	ND	PPB V	0.085	0.280	
CARBON TETRACHLORIDE	ND	PPB V	0.085	0.280	
TRICHLOROETHYLENE	ND	PPB V	0.085	0.280	



## Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

**Organic Chemistry** 

Environmental Health Division

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007

WI DATCP ID: 105-415

WSLH Sample: OW004797

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations:

LOD = Level of detection

LOQ = Level of quantification ND = None detected. Results are less than the LOD

Responsible Party: <u>the Aup</u> Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.



# Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Heal	th Division	Orga	anic Chemistry	,
WDNR LAB ID: 113133790	NELAP LAB ID: E37658	EPA LAB	WI00007	WI DATCP ID: 105-415
	WSLH Sample: O	W004800		
BADGER TECHNICAL	SERVICES		Bill To	
1 BADGER RD			Billing ID	2: 7305879
			Custome	er ID: 320225
BARADOO, WI 33913			TRACKI	NG 4920
			2601 AG	RICULTURAL DRIVE
			MADISC	N WI 53718
			ID#:	
Field #:	-50-00		Waterbo	dy/Outfall ID:
	.50:00		Point/We	
Collection End: 06/27/2012 10:	17:00		Account	#: LH034
Collected By: BHB			Project N	NO: $06/20/2012 11:12:00$
County:			Date Rei	cerved: 00/29/2012 11.12.00
Sample Source: AIR			Sample	Reason:
Sample Deptn:			Sample	
Sample Information.				
Sample Location: 61 CANIS				
Analyses and Results:	TEN / FAGOIVE GAMIFLE			

Analysis Date 07/06/2012	Lab Comment				
Analysis Method	Result	Units	LOD	LOQ	Report Limit
CHLOROFORM	ND	PPB V	0.085	0.280	
CARBON TETRACHLORIDE	ND	PPB V	0.085	0.280	
TRICHLOROETHYLENE	ND	PPB V	0.085	0.280	





D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division

Organic Chemistry

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007

WI DATCP ID: 105-415

WSLH Sample: OW004800

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations:

LOD = Level of detection LOQ = Level of quantification

ND = None detected. Results are less than the LOD

Responsible Party: <u>Hun</u> Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.



# Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division	Orga	anic Chemistry	
WDNR LAB ID: 113133790 NELAP LAB ID: E376	58 EPA LAB	WI00007	WI DATCP ID: 105-415
WSLH Sample:	OW004799		
BADGER TECHNICAL SERVICES		Bill To	
1 BADGER RD		Billing ID:	7305879
		Customer	ID: 320225
BARABOO, WI 33313		TRACKING	G 4920
		2601 AGR	
		MADISON	WI 53718
		ID#:	
		Waterbody	//Outfall ID:
Collection Start: 06/27/2012 14:40:00		Point/Well:	
Collection End: 06/27/2012 15:10:00		Account #:	LH034
Collected By: BHB		Project No	:
County:		Date Rece	vived: 06/29/2012 11:12:00
Sample Source: AIR		Date Repo	orted: 07/09/2012
Sample Depth:		Sample Re	eason:
Sample Information:			
Sample Location: VIP-7A BADGER ARMY AMMUNI	TION PLANT		
Sample Description: 6 L CANISTER / PASSIVE SAMP	LER		
Analyses and Results:			

Analysis Date 07/06/2012	Lab Comment				
Analysis Method	Result	Units	LOD	LOQ	Report Limit
CHLOROFORM	ND	PPB V	0.085	0.280	
CARBON TETRACHLORIDE	ND	PPB V	0.085	0.280	
TRICHLOROETHYLENE	ND	PPB V	0.085	0.280	





D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division

Organic Chemistry

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007

WI DATCP ID: 105-415

WSLH Sample: OW004799

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations:

LOD = Level of detection LOQ = Level of quantification

ND = None detected. Results are less than the LOD

Responsible Party: <u>Hun</u> Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.



# Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Healt	h Division	Orga	anic Chemistry	
WDNR LAB ID: 113133790	NELAP LAB ID: E37658	EPA LAB	WI00007	WI DATCP ID: 105-415
	WSLH Sample: O	W004798		
BADGER TECHNICAL	SERVICES		Bill To	
1 BADGER RD			Billing ID:	7305879
			Customer I	D: 320225
DARADUU, WI 53913			TRACKING	G 4920
			2601 AGR	
			MADISON	WI 53718
			ID#:	
Field #:			Waterbody	/Outfall ID:
Collection Start: 06/27/2012 13:	30:00		Point/Well:	
Collection End: 06/27/2012 13:	57:00		Account #:	LH034
Collected By: BHB			Project No	:
County:			Date Rece	ived: 06/29/2012 11:12:00
Sample Source: AIR			Date Repo	rted: 07/09/2012
Sample Depth:			Sample Re	eason:
Sample Information:				
Sample Location: VIP-9A BA	DGER ARMY AMMUNITI	ON PLANT		
Sample Description: 6 L CANIS	TER / PASSIVE SAMPLE	R		
Analyses and Results:				

Analysis Date 07/06/2012	Lab Comment				
Analysis Method	Result	Units	LOD	LOQ	Report Limit
CHLOROFORM	ND	PPB V	0.085	0.280	
CARBON TETRACHLORIDE	0.563	PPB V	0.085	0.280	
TRICHLOROETHYLENE	ND	PPB V	0.085	0.280	





D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division

Organic Chemistry

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007

WI DATCP ID: 105-415

WSLH Sample: OW004798

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations:

LOD = Level of detection LOQ = Level of quantification

ND = None detected. Results are less than the LOD

Responsible Party: <u>Hun</u> Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.



# Laboratory Report

D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Hea	alth Division	Orga	anic Chemistr	ý
WDNR LAB ID: 113133790	NELAP LAB ID: E37658	EPA LAB	WI00007	WI DATCP ID: 105-415
		V000054		
	WSLH Sample: 0	X000054		
BADGER TECHNICA	L SERVICES		Bill To	
1 BADGER RD.			Billing I	D: 7305879
			Custom	er ID: 320225
BARABOO WI 53913			TRACK	NG 4920
			2601 AC	GRICULTURAL DRIVE
			MADISC	DN WI 53718
			ID#:	
Field #:			Waterbo	ody/Outfall ID:
Collection Start: 07/03/2012 09	9:58:00		Point/W	ell:
Collection End: 07/03/2012 10	):28:00		Account	#: LH034
Collected By: BHB-BADGER	TECH. SVC		Project I	No:
County:			Date Re	ceived: 07/05/2012
Sample Source: AIR			Date Re	ported: 07/10/2012
Sample Depth: 45'			Sample	Reason:
Sample Information: SAMPLE	R 2432			
Sample Location: VIP-10A	BADGER ARMY AMMUNIT	ION PLANT	Γ	
Sample Description: 6L CANIS	STER/PASSIVE SAMPLER			
Analyses and Results:				
Analysis Date	Lab Comment			

Analysis Date 07/09/2012 12:35:21	Lab Comment				
Analysis Method	Result	Units	LOD	LOQ	Report Limit
CHLOROFORM	ND	PPB V	0.085	0.280	
CARBON TETRACHLORIDE	25.5	PPB V	0.085	0.280	
TRICHLOROETHYLENE	1.08	PPB V	0.085	0.280	





D.F. Kurtycz, M.D., Medical Director • Charles D. Brokopp, Dr.P.H., Director

Environmental Health Division

Organic Chemistry

WDNR LAB ID: 113133790 NELAP LAB ID: E37658 EPA LAB WI00007

WI DATCP ID: 105-415

WSLH Sample: OX000054

Test results for NELAP accredited tests are certified to meet the requirements of the NELAC standards. For a list of accredited analytes see http://www.slh.wisc.edu/nelap/

List of Abbreviations:

LOD = Level of detection LOQ = Level of quantification

ND = None detected. Results are less than the LOD

Responsible Party: <u>Hun</u> Steve Geis, Chemist Supervisor

If there are questions about this report, please contact Steve Geis at 608-224-6269.

. of Wisconsin Irtment of Natural Resources	P0# BTS 0001832	Chain of C Form 4100-145	ustody Record (R 3/09)
llector(s) Name evda BDUC	Return Report As: (select one) Email or Postal Address	Spechvo-inc.com (008	ber (include area code)
wher J partment of	the Army Badger Army F	monunition Plant	tber (include area code)
hples: Offered? Yes Accepted? Yes	No Accepted By (Signature):	06129112 Lab Use	Only
No. Date Time C	No. of Station Location ontainers Sample Description	Tracked / Improperty Broken Sealed C	Good Other ondition Comments
1A 6-27128:20	1 Vapor Sample tor Chlorotorn TEE and Carbon tetrachier	+ 00004797	
A 6-27-12 9:50		0\v004800	
04:5 21-26-9 Y1		OW004799	
3 A 6-27-12 1:30		0W004798	
of Shipment:	Reason for Sample Collection:	Was the sample shipping container sealed on r	ceipt?YesNo
taff	Anhydrous Ammonia Spill Pesticide Spill * – Specify	'esticide:	
I.S. Postal Service IPS	Animal Waste L Hazardous Waste Releas	* • * – Specify Product:	
edEx	Dairy Product Spill	oecify Industry Type:	
viner-specify.	Construction/Storm Water Kunoff Contact the laboratory with product information and for consultation. Als	, include sample of suspected spilled product.	
certify that I received and pro	oerly handled these samples as noted below:	Disposition of Ur	used Portion Sample:
shed By (Signature)	Date / Time Received By (Signature) しこてる-1つんは化	Date / Time	se
hed By (Signature)	Date / Time / Received By (Signature)	Date / Time	n until further notice
hed By (Signature)	Date / Time Received for Laboratory By (Signature)	Date / Time If you need ac	ditional room for notes

Chain of Custody Record Form 4100-145 (R 3/09)	Phone Number (include area code) . Cow UOS-434-5733 Phone Number (include area code)	Lab Use Only	cked / Improperly Good Other oken Sealed Condition Comments		g container sealed on receipt? 🖄 Yes 🔲 No		ted spilled product.		Retain until further notice	If you need additional room for notes, use the back of this form.
	pecpro-im		Lab ID Number Nover53 Nover53		Vas the sample shippi	Specify Product:	ude sample of suspe	Date / Time	Date / Time	Date / Time
PO# BTS 0001832	Return Report As: (select one) Email or Postal Address Email  Hard Copy  Drewda.  Doy ce  Sad ce  Nrun  Mull	Accepted By (Signature):	Sample Description Sample Description VaDON Sample FON Chlonoforn, c TLE and carbon tetrachlonde analysis		for Sample Collection:	nimal Waste	tact the laboratory with product information and for consultation. Also, incl	dled these samples as noted below. Date / Time Received By (Signature) フィューム/えん	Date / Time Received By (Signature)	Date / Time Received for Laboratory By (Signature)
State of Wisconsin Department of Natural Resources	Sample Collector(s) Name Breenda Boyce Property Owner Dont of Arian II	Split Samples: Offered? TYes No Accepted? Yes No	Field ID No.         Date         Time         No. of No. of %: 24           V1P-1A         7-3-1A         8:51         1           V1P-10A         7-3-1A         9:53         1		Method of Shipment: Reason 1	UPS UPS Postal Service A A A A O C C C C	* Con	I hereby certify that I received and properly han Relinguished By (Signature)	Relinquished By (Signature)	Relinquished By (Signature)

#### Indoor Air Vapor Action Levels for Various VOCs Quick Look-Up Table<sup>1</sup> Based on November 2011 Regional Screening Level Summary Table

Chemical	Non-Re (1-in-100,000 risk	Non-ResidentialResidentialMolecular Weight(1-in-100,000 risk for carcinogens)(1-in-100,000 risk for carcinogens)(MW)			Molecular Weight (MW)	Basis of RSL <sup>2</sup>
	ppbV <sup>*</sup>	µg/m³	ppbV <sup>*</sup>	µg/m³	g/mole	
Benzene	4.9	16.0	0.95	3.1	78.11	С
Carbon Tetrachloride	3.1	20	0.64	4.1	153.82	С
Chloroform	1.1	5.3	0.22	1.1	119.38	С
Chloromethane	190	390	45	94	50.49	n
Dichlorodifluoromethane	88	440	20	100	120.91	n
1,1 – Dichloroethane (1,1-DCA)	19	77	3.6	15	98.96	С
1,2-Dichloroethane (1,2-DCA)	1.1	4.7	0.23	0.94	98.96	С
1,1 -Dichloroethylene (1,1-DCE)	220	880	52	210	96.94	n
1,2-Dichloroethene (cis and mixed)	NA	NA	NA	NA	96.94	n
1,2-Dichloroethene (trans)	65	260	16	63	96.94	n
Ethylbenzene	11	49	2.2	9.7	106.17	С
Methyl-tert-Butyl Ether (MTBE)	130	470	26	94	88.15	С
Methylene Chloride	74	260	15	52	84.93	С
Naphthalene	0.68	3.6	0.14	0.72	128.18	С
Tetrachloroethylene	3.0	21	0.60	4.1	165.83	С
Toluene	5700	22,000	1400	5200	92.14	n
1,1,1 - Trichloroethane	4000	22,000	940	5200	133.41	n
Trichloroethylene	1.6	8.8	0.38	2.1	131.39	n
Trichlorofluromethane	540	3100	130	730	137.37	n
Trimethylbenzene (1,2,4)	6.2	31	1.5	7.3	120.2	n
Trimethylbenzene (1,3,5)	NA	NA	NA	NA	120.2	n
Vinyl Chloride	11	28	0.62	1.6	62.5	С
Xylene (mix)	100	440	23	100	106.17	n
Xylene (n,m,o separately)	100	440	23	100	106.17	n

<sup>1</sup>Regional Screening Tables: <u>http://www.epa.gov/reg3hwmd/risk/human/rb-concentration\_table/index.htm</u>

<sup>2</sup> Basis for Regional Screening Level – n = non-carcinogen; c = carcinogen. Non-carcinogen RSL table values are based on a HI = 1; therefore, <u>no</u> multiple should be applied to the table values. Carcinogen RSL (cRSL) table values are listed as 1-in-1,000,000; in Wisconsin indoor air, 1-in-100,000 excess lifetime cancer risk is acceptable. This table of Vapor Action Levels was developed by multiplying the cRSL values by 10. Screening levels in this table are rounded to 2 significant digits.

\* Conversions from  $\mu g/m^3$  to ppbV in this table based on T = 20°C or 68 °F; P = 1 atm or 101.325 kPA (see next page)

### Convert µg/m<sup>3</sup> to ppbV

On-line calculator: Indoor Air Unit Conversion <u>http://www.epa.gov/athens/learn2model/part-two/onsite/ia\_unit\_conversion.html</u> At 20°C and 1 atm:

$$ppbV = \frac{\mu g}{MW} \times 8.3144 \left[ \frac{L \cdot kPa}{mol \cdot {}^{\circ}K} \right] \times \left[ T_{\circ c} + 273.15 \right] K \times \frac{1}{101.325 \, kPa} \quad \text{OR} \quad \text{ppbV} = (\mu g/\text{m}^3 \text{ x } 24.05)/\text{molecular weight}$$

#### Using indoor vapor action levels (VAL) to determine vapor risk screening levels (VRSL)

Vapor risk screening levels are used to estimate indoor air concentrations from sub-slab vapor, soil gas or groundwater concentrations. Standard attenuation factors are applied to each media. This table lists the attenuation factor ( $AF = C_{IA}/C_{source}$ ) and the dilution factor (inverse of the AF). The VAL is <u>divided by the AF</u> or <u>multiplied by the dilution factor</u> to calculate the vapor risk screening level.

Media Screened	<b>Residential / Small (</b>	<b>Commercial Buildings</b>	Large Commercial / Industrial Building		
	Attenuation Factor	Dilution Factor	Attenuation Factor	Dilution Factor	
Sub-slab vapor	0.1	10	0.01	100	
Deep soil gas	0.01	100	0.001	1000	
Groundwater	0.001	1000	0.0001	10,000	

#### <u>Determining the Vapor Risk Screening Level for Groundwater</u> (at what concentration would groundwater potentially cause an indoor air exceedance)

$$C_{gw} = \frac{C_{IA}}{\left(H \times AF_{gw} \times 1000 \, \frac{L}{m^3}\right)}$$

Where:  $C_{gw}$  = groundwater concentration (µg/L)

 $C_{IA}$  = indoor air concentration (from Quick look-up table,  $\mu g/m^3$ )

H = Henry's Law constant (dimensionless) from Chemical Specific Parameter Table: <u>http://www.epa.gov/reg3hwmd/risk/human/rb-</u>

concentration\_table/Generic\_Tables/index.htm

 $AF_{gw}$  = attenuation factor between groundwater and indoor air

<u>Note</u>: The default attenuation factor for groundwater to indoor air is 0.001. However, if the contaminated groundwater is located at the building foundation, the attenuation factor should be increased to 0.1 (i.e., treated as a sub-slab concentration). If contaminated groundwater is located close to the foundation (but not in contact with the foundation), the default attenuation factor of 0.001 may not be predictive of indoor air concentration. In that case, sub-slab sampling should be conducted.

Update: December 6, 2011

### Appendix G

Screening Level Groundwater Risk Evaluation (Draft) - Exponent

## Exponent®

#### EXTERNAL MEMORANDUM

To:	Joel Janssen, P.G., SpecPro Professional Services, LLC
FROM:	Michael Kierski, Ph.D., and Michael Garry, Ph.D.
DATE:	December 17, 2018
PROJECT:	Badger Army Ammunition Plant, Risk Assessment Consulting Support
SUBJECT:	Screening Level Groundwater Risk Evaluation (Draft)

### Introduction

SpecPro Professional Services, LLC (SpecPro) requested Exponent evaluate the potential risk associated with exposure to groundwater at the Badger Army Ammunition Plant (BAAP) on behalf of the Department of the Army (Army). Risks associated with groundwater contamination can arise from domestic use of groundwater (e.g., drinking, bathing, and washing) from private residential wells. In addition, exposure to the contamination in groundwater can occur through chemical vapors emanating from groundwater due to transport of chemicals from the water table through soil resulting in release of the vapors into homes, which is referred to as vapor intrusion. We were asked to calculate potential risks associated with domestic use of groundwater from existing residential wells, offsite monitoring wells that are within areas not restricted with respect to the use of the groundwater onsite is used as source of domestic water in the future.<sup>1</sup> The vapor intrusion exposure pathway will be separately evaluated by SpecPro.

A groundwater risk evaluation was conducted to estimate the cumulative risk associated with both current and hypothetical future exposure to groundwater by residents. Current risks were estimated by evaluating groundwater data collected from both monitoring wells and residential wells located offsite and associated with the three groundwater plumes that have migrated offsite into residential areas. The potential for future groundwater risks was evaluated using groundwater data collected from monitoring wells located onsite and associated with the four onsite groundwater plumes. Groundwater data from 2015, 2016, 2017, and 2018 were used for the initial screening level risk evaluation to best represent current and potential future groundwater quality. We used this data set because these data best represent current groundwater quality at BAAP. Source removal and groundwater remediation activities occurred

<sup>&</sup>lt;sup>1</sup> A monitoring well is placed in an area for the specific purpose of monitoring groundwater quality as part of a site investigation, but a monitoring well is not constructed in a way that would allow it to be used to provide drinking water. A residential well, on the other hand, is constructed specifically to provide drinking water.

at BAAP until 2015, so historical groundwater data collected before these activities occurred would not accurately reflect current groundwater quality conditions.

The analyses were focused on areas influenced by the following four groundwater plumes:

- Propellant Burning Ground Plume
- Deterrent Burning Ground Plume
- Central Plume
- Nitrocellulose Production Area Plume.

The assessment used standard U.S. Environmental Protection Agency (EPA) risk assessment methods to evaluate current groundwater risks from consumption. The approach used to estimate groundwater risks for this screening level evaluation was conservative in nature (i.e., more likely to overestimate risk). Maximum concentrations of analytes in offsite or onsite wells associated with each plume were used to estimate risks. Therefore, the risks presented in this screening level groundwater risk evaluation should be viewed as upper bound estimates of the potential groundwater risks within a specific area and do not reflect the risk associated with drinking water from any specific residential or groundwater monitoring well. The remaining sections document the screening risk methods used to estimate the groundwater risks by area and the results of the screening level groundwater risk evaluation by plume.

### **Screening Level Groundwater Risk Assessment Methods**

A screening level groundwater risk evaluation was conducted for each of the four plume areas using EPA human health risk assessment (HHRA) methods (U.S. EPA 1989, 1991). The screening risk evaluation was conducted in two steps. First, site concentrations were compared to health-based screening levels to identify chemicals of potential concern (COPCs). Second, risk estimates were calculated for COPCs that exceeded screening levels.

### Methods for Screening of Chemicals of Potential Concern

For this screening evaluation, we relied on the EPA's November 2018 tapwater regional screening levels (RSLs)<sup>2</sup> and Wisconsin Department of Natural Resources' (WDNR's) NR140 groundwater standards (WDNR 2017) to screen the groundwater data. A summary of the groundwater screening levels from these two sources are provided in Table 1. For purposes of the groundwater screening evaluation, we compared the maximum concentration of the chemicals detected in each plume area to the lowest groundwater screening value available for

<sup>&</sup>lt;sup>2</sup> The tapwater RSLs table used for the screening process was based on a target cancer risk of one-in-a-million (1E-06) and a noncancer target hazard quotient (THQ) of 0.1.

BAAP Groundwater Risk Evaluation – Draft December 17, 2018 Page 3

each chemical.<sup>3</sup> Chemicals with a maximum detected concentration that exceeded the lowest available groundwater screening value for the chemical are considered at the screening level stage to be COPCs.

#### **Groundwater Risk Calculation Methods**

Based on EPA risk assessment guidance, COPCs can be further evaluated in the HHRA to provide site-specific risk estimates. The site-specific HHRA is part of the remedial investigation process that serves to document potential risks associated with exposure to chemicals in environmental media at a specific site. This screening level groundwater evaluation is an initial step in the HHRA being completed by SpecPro. The groundwater risk estimates were calculated by plume area using a simple EPA scaling method described in Section 2.6.1 of the online Regional Screening Levels (RSLs) – User's Guide (November 2018), a copy of which is provided as Appendix A of this memorandum.<sup>4</sup> For each COPC, the following calculations were used to estimate potential cancer and noncancer risks, as applicable.

 $Cancer \ Risk = (C_{gw} \times TR) / \ RSL_{tapwater \ (based \ on \ cancer \ effect)}$ Noncancer  $HQ = (C_{gw} \times THQ) / \ RSL_{tapwater \ (based \ on \ noncancer \ effect)}$ 

where:

Cancer Risk = a unitless probability of an individual developing cancer over a lifetime.

- HQ = hazard quotient; a unitless ratio of exposure to chemicals in groundwater to a reference dose at which no health effects are expected to occur.
- $C_{gw}$  = Groundwater concentration in units of  $\mu g/L$  or mg/L.
- TR = Target cancer risk that the RSL is based on (1E-06).
- THQ = Target hazard quotient that the RSL is based on.
- $RSL_{tapwater} = Tapwater RSL$  (U.S. EPA November 2018) in the same units as  $C_{gw}$  for the applicable effect (i.e., cancer or noncancer).

We calculated cancer risks for all COPCs considered potential carcinogens. Cancer risk is expressed as an upper bound probability that an individual will develop cancer as a result of exposure to a chemical in the groundwater over their lifetime. For example, a 1E-06 cancer risk represents a one-in-a-million upper bound risk of an individual contracting cancer during their lifetime from the specific chemical exposure. This cancer risk is in addition to the background level risk of contracting cancer of any kind during one's lifetime unrelated to groundwater

<sup>&</sup>lt;sup>3</sup> For the screening assessment, all dinitrotoluene isomers (e.g., 2,4-dinitrotoluene, 3,4-dinitrotoluene, etc.) were summed together to calculate a total dinitrotoluene for each sample. The total dinitrotoluene value was then compared to the lowest screening value available for the dinitrotoluene isomers. This conservative approach was used because some of the dinitrotoluene isomers did not have screening values.

<sup>&</sup>lt;sup>4</sup> <u>https://www.epa.gov/risk/regional-screening-levels-rsls-users-guide</u>

chemical exposure, which is approximately 40.8% in males (a little less than one in two) and approximately 37.5% in females (a little more than one in three).<sup>5</sup>

We also calculated noncancer risks for each COPC. The chemical-specific noncancer risks are represented by hazard quotient (HQ) values derived by comparing the groundwater chemical concentrations to chemical-specific tapwater RSLs.<sup>6</sup> If the resultant HQ value is less than or equal to 1, then adverse health effects associated with exposure to that chemical in the groundwater are unlikely to occur even among sensitive individuals. An HQ greater than 1 does not necessarily indicate that a health effect will occur but does indicate that there is the potential for a health effect with increasing exposures and that additional analysis is necessary.

EPA's tapwater RSLs are risk-based concentrations developed using specific generic exposure assumptions that represent reasonable maximum exposure (RME) to groundwater. The tapwater RSLs were developed considering potential exposure to chemicals in groundwater associated with domestic use of the groundwater as a drinking water source, as well as other normal domestic water uses, such as bathing, doing laundry, and washing dishes. Exposure to chemicals in groundwater are incorporated into the tapwater RSL for ingestion, dermal contact with the water, and inhalation of the portion of the chemicals in groundwater that are volatilized from the water as it is used (e.g., for bathing). Tapwater RSLs based on noncancer effects are also developed separately for adults and children, and then the lower of the two RSLs is selected for evaluating risks to people. RSLs based on cancer incorporate exposure during both childhood and adulthood. For this reason, the tapwater RSLs are considered a conservative risk-based benchmark on which to calculate risk associated with groundwater chemical exposure.

The potential risk associated with groundwater in each plume area was calculated for each COPC using maximum groundwater concentrations and tapwater RSLs in the equations presented above.<sup>7</sup> The total groundwater risks for an area were estimated by adding the individual cancer risks or noncancer HQs for all COPCs together for a given area. The process of adding these COPC-specific risks together is described in detail in Section 5.15.2 of the online Regional Screening Levels (RSLs) – User's Guide (May 2018). The sum of all cancer risks for the COPC within an area is referred to as the cumulative cancer risk. The sum of all noncancer risks (i.e., HQs) for the COPC within an area is referred to as the hazard index (HI). The groundwater risk evaluation approach used to develop the cumulative cancer risk and

<sup>&</sup>lt;sup>5</sup> American Cancer Society PowerPoint presentation titled "cancer-statistics-presentation-2017.pptx" located at https://www.cancer.org/research/cancer-facts-statistics/all-cancer-facts-figures/cancer-facts-figures-2017.html.

<sup>&</sup>lt;sup>6</sup> The noncancer HQ for each chemical was calculated using the EPA noncancer-based tapwater RSLs based on a THQ of 0.1. Therefore, the noncancer-based risks were estimated using a THQ of 0.1 in the equation presented herein.

<sup>&</sup>lt;sup>7</sup> For total dinitrotoluene, the risk was calculated two ways (refer to Tables 2b through 8b). First, the risk was estimated using the total dinitrotoluene concentration, and then the risk was calculated by summing the risks for the individual dinitrotoluene isomer(s) that made up the total dinitrotoluene concentration. The second method was used as a check because one of the isomers is more toxic (i.e., 2,6-dinitrotoluene) than the other isomers. Therefore, if the total dinitrotoluene concentration is dominated by 2,6-dinitrotoluene, the risk using the total value can slightly underestimate the risk. The maximum total dinitrotoluene risk calculated by the two methods was used as the total risk estimate for an area.
noncancer HI is conservative in nature because it assumes people would be exposed to the maximum concentrations of all COPCs, although not all maximum COPC concentrations within an area occur in the same well. For this reason, the well with the maximum COPC concentration within the area for which the risk was estimated is identified in the risk tables presented in the next section.

## Interpretation of Risk Estimates

Although the determination of an acceptable site-specific target cancer risk level is ultimately a decision for risk managers, we provide perspective on the calculated values by comparing them to the range of cancer risk levels cited in EPA's National Contingency Plan (NCP) (U.S. EPA 1990), which EPA describes as the "blueprint for the Superfund law." The NCP (40 CFR 300.430 [e] [2]) states, "For known or suspected carcinogens, acceptable exposure levels are generally concentration levels that represent an excess upper bound lifetime cancer risk to an individual of between  $10^{-4}$  [1E-04] and  $10^{-6}$  [1E-06] using information on the relationship between dose and response." A later EPA memo states that "where the cumulative site risk to an individual based on reasonable maximum exposure for both current and future land use is less than  $10^{-4}$  [1E-04] and the non-carcinogenic hazard quotient is less than  $1,^8$  action generally is not warranted unless there are adverse environmental impacts." The memo further states:

A risk manager may also decide that a baseline [cancer] risk level less than 10<sup>-4</sup> is unacceptable due to site-specific reasons, and that remedial action is warranted ... Other chemical-specific ARARs [Applicable or Relevant and Appropriate Requirements] may also be used to determine whether a site warrants remediation. (U.S. EPA 1991)

To provide perspective on the estimated risks associated with groundwater use, the values are compared to the risk management criteria described above (i.e., cumulative cancer risk range or an HI of 1). Based on the NCP and EPA guidance, cumulative carcinogenic risks below 1E-06 are generally considered to represent a negligible risk, cumulative risks between 1E-06 and 1E-04 are within a range considered acceptable under most conditions, and cumulative cancer risks above 1E-04 indicate unacceptable levels of risk where remedial action or further evaluation needs to be considered. In other words, cumulative cancer risks that fall within the range of 1E-06 and 1E-04 are generally considered acceptable and require no action, but exceptions to this general rule can be made on a site-specific basis as discussed above and presented below. For noncancer effects there is no similar "risk range"; rather, any HI greater than 1 indicates further evaluation is required to determine whether the exposure presents a health concern. In other words, an HI equal to or less than 1 is not considered to pose a potential health concern.

However, as described above, risk managers may elect to consider risks within the risk range of 1E-06 and 1E-04 for remedial action. The Army risk managers have decided that in offsite areas where the Army does not have control over the use of the groundwater as a drinking water

<sup>&</sup>lt;sup>8</sup> While the NCP refers to an HQ less than 1 as the risk management criterion, in practice, the cumulative noncancer HI is used as the risk management criterion to determine whether potential noncancer health concerns require further evaluation.

source they will consider cumulative cancer risks above 1E-06 for potential action or additional evaluation. For onsite areas where the Army has control over the use of the groundwater as a drinking water source, they will use a cumulative risk management goal of less than or equal to 1E-04. Using a different remedial cumulative cancer risk goal onsite (1E-04) versus offsite (1E-06) affects the selection of contaminants of concern (COCs) in these two areas (offsite versus onsite), as discussed below.

A summary of the COPCs identified as COCs based on the results of the screening level groundwater risk assessment is presented by plume area in the next sections of this memorandum. COCs are analytes found to significantly contribute to the cumulative risk in a particular area (onsite or offsite) where risk was estimated to be above the risk management criteria selected for that area. For onsite areas, where the Army controls the use of groundwater, the risk management criteria used for defining a chemical as a COC is a cumulative cancer risk >1E-04 or an HI >1. For offsite areas, where the Army does not control the use of the groundwater, the risk management criteria used for defining a chemical as a COC is a cumulative cancer risk >1E-06 or an HI >1. As mentioned above, a cumulative cancer risk greater than 1E-04 is typically considered unacceptable, and an HI greater than 1 indicates a level of exposure that needs to be evaluated further to determine whether a health concern exists. However, in the case of the offsite areas, the Army has decided to take a much more health protective approach and use a lower cumulative cancer risk criterion of greater than 1E-06. For this reason, two different sets of risk management criteria are used in the following sections to put the cumulative risk estimates into perspective in terms of the need for potential action or additional evaluation.

Because of the different risk management criteria applied to onsite and offsite areas, risks by plume area are assessed and described separately for onsite and offsite areas in the following sections. There are currently no residential drinking water wells onsite, so only onsite monitoring well data were used to estimate risks. For offsite portions of each plume, both monitoring well and residential well data were used to estimate risks. Like the onsite area, the monitoring wells in offsite areas are not used as a drinking water source. However, the residential wells in offsite areas are used as a drinking water source. Therefore, risks calculated for a particular offsite plume area may have been based on monitoring well data or residential well data depending on the type of well in which the maximum chemical concentration was detected. For offsite portions of a plume, we specify whether the risk estimate is based on monitoring well or residential well data or a combination of both. The plume-specific risk tables specify the well where the maximum concentration of each chemical was identified.

# Screening Risk Evaluation for the Propellant Burning Ground Plume

Risks associated with hypothetical future and current use of groundwater were calculated based on data collected from the Propellant Burning Ground Plume in 2015, 2016, 2017, and 2018. Onsite monitoring well data were used to estimate hypothetical future onsite risks. The Army

currently prohibits the use of onsite groundwater as a drinking water source and anticipates continuing this practice in the future. Therefore, it is unlikely that onsite groundwater will be used as a drinking water source in the future. Data from both offsite monitoring wells and residential drinking water wells were used to evaluate current offsite risks.

## **Future Onsite Risk Evaluation**

Table 2a summarizes the results of the onsite COPC screening for the Propellant Burning Ground Plume. Hypothetical future risks were calculated for the COPCs identified in the onsite monitoring well data. Cancer and noncancer risk estimates are summarized in Table 2b for individual COPCs and for cumulative risks. A COPC may be identified as a COC because of its noncancer and/or cancer risks.

For onsite areas *where groundwater use is under the control of the Army* the risk management criteria used to determine which chemicals are COCs were a cumulative cancer risk >1E-04 or an HI >1. The cumulative risk estimates are considered hypothetical in nature, as groundwater onsite would not be expected to be used in these areas in the future because of the restrictions placed on groundwater use by the Army.

## **Cancer Risk Summary**

Cancer risks calculated using the simple scaling method for a hypothetical future onsite residential scenario, along with the maximum observed concentration of each COPC identified in the onsite monitoring wells from the Propellant Burning Ground Plume area, yielded a cumulative cancer risk estimate of 6E-03, which exceeds the upper limit of EPA's target cancer risk range. The only COC that contributed to that exceedance was dinitrotoluenes.

## **Noncancer Risk Summary**

Based on the same hypothetical future onsite residential scenario, the noncancer HI (which is the sum of all HQs for individual COPCs) for the onsite Propellant Burning Ground Plume was 53, indicating the need for additional analysis. The COCs contributing to the HI >1 were dinitrotoluenes, ethyl ether, and trichloroethene.

## **Current Offsite Risk Evaluation**

Table 3a summarizes the results of the offsite COPC screening for the Propellant Burning Ground Plume. Current risks were calculated for the COPCs identified in the offsite monitoring and residential well data. Cancer and noncancer risk estimates are summarized in Table 3b for individual COPCs and for cumulative risks. For offsite areas *where groundwater use is not under the control of the Army* the risk management criteria used to determine which chemicals are COCs were a cumulative cancer risk >1E-06 or an HI >1.

## **Cancer Risk Summary**

Cancer risks calculated using the simple scaling method for a current offsite residential scenario, along with the maximum observed concentration of each COPC identified in the offsite portion of the Propellant Burning Ground Plume area, yielded a cumulative cancer risk estimate of 1E-04. This is within EPA's target cancer risk range where remedial action or additional evaluation is typically considered unnecessary. However, the risk estimate exceeds the lower cumulative cancer risk goal of 1E-06 selected by the Army. All maximum concentrations of COPCs were from offsite monitoring wells, and none were from a residential well. The COCs that contributed to that exceedance were carbon tetrachloride, chloroform, dinitroluenes, and trichloroethene.

## **Noncancer Risk Summary**

Based on the same current offsite residential scenario, the noncancer HI for the offsite portion of the Propellant Burning Ground Plume was 5, indicating the need for additional analysis. The only COC contributing to the HI >1 was trichloroethene (monitoring well).

## Summary of Risks for the Propellant Burning Ground Plume

Based on the maximum risk scenario, both the onsite and offsite areas of the Propellant Burning Ground Plume represent zones where cumulative risk estimates exceed the risk management criteria selected by the Army. The cumulative cancer and noncancer risks (i.e., HI) are summarized separately in the table below. A COPC may be a COC because it exceeds the noncancer HI and/or cumulative cancer risk management criterion. As discussed previously, for offsite areas *where groundwater use is not under the control of the Army* the risk management criteria used to determine which chemicals are COCs are a cumulative cancer risk >1E-06 or an HI >1, whereas the risk management criteria for onsite areas are a cumulative cancer risk >1E-04 or an HI >1.

Location	Cumulative Cancer Risk	Noncancer Hazard Index	Contaminants of Concern*
Onsite (Hypothetical Future Risk)	6E-03	53	Ethyl Ether, Dinitrotoluenes, Trichloroethene
Offsite (Current Risk)	1E-04	5	Carbon Tetrachloride, Chloroform, Dinitrotoluenes, Trichloroethene

#### Summary of Risk Estimates for Propellant Burning Ground Plume

\* A chemical is considered a contaminant of concern (COC) if either the cancer risk or the noncancer hazard quotient (HQ) for that chemical *exceeds* the risk management criteria.

# Screening Risk Evaluation for the Deterrent Burning Ground Plume

Risks associated with hypothetical future and current use of groundwater were calculated based on data collected from the Deterrent Burning Ground Plume in 2015, 2016, 2017, and 2018. Onsite monitoring well data were used to estimate hypothetical future onsite risks. The Army currently prohibits the use of onsite groundwater as a drinking water source and anticipates continuing this practice in the future. Therefore, it is unlikely that onsite groundwater will be used as a drinking water source in the future. Data from both offsite monitoring wells and residential drinking water wells were used to evaluate current offsite risks.

## **Future Onsite Risk Evaluation**

Table 4a summarizes the results of the onsite COPC screening for the Deterrent Burning Ground Plume. Hypothetical future risks were calculated for the COPCs identified in the onsite monitoring well data. Cancer and noncancer risk estimates are summarized in Table 4b for individual COPCs and for cumulative risks. A COPC may be identified as a COC because of its noncancer and/or cancer risks.

For onsite areas *where groundwater use is under the control of the Army* the risk management criteria used to determine which chemicals are COCs were a cumulative cancer risk >1E-04 or an HI >1. The cumulative risk estimates are considered hypothetical in nature, as groundwater onsite would not be expected to be used in these areas in the future because of the restrictions placed on groundwater use by the Army.

## **Cancer Risk Summary**

Cancer risks calculated using the simple scaling method for a hypothetical future onsite residential scenario, along with the maximum observed concentration of each COPC identified in the onsite monitoring wells from the Deterrent Burning Ground Plume area, yielded a cumulative cancer risk estimate of 9E-05, which is within EPA's target cancer risk range of 1E-06 to 1E-04 and meets the risk management criterion for the site. Therefore, there are no COCs based on cancer risk.

## **Noncancer Risk Summary**

Based on the same hypothetical future onsite residential scenario, the noncancer HI for the onsite Deterrent Burning Ground Plume was 3, indicating the need for additional analysis. The only COC contributing to the HI >1 was 1,1,2-trichloroethane.

## **Current Offsite Risk Evaluation**

Table 5a summarizes the results of the offsite COPC screening for the Deterrent Burning Ground Plume. Current risks were calculated for the COPCs identified in the offsite monitoring and residential well data. Cancer and noncancer risk estimates are summarized in Table 5b for individual COPCs and for cumulative risks. For offsite areas *where groundwater use is not under the control of the Army* the risk management criteria used to determine which chemicals are COCs were a cumulative cancer risk >1E-06 or an HI >1.

## **Cancer Risk Summary**

Cancer risks calculated using the simple scaling method for a current offsite residential scenario, along with the maximum observed concentration of each COPC identified in the offsite residential or monitoring wells from the Deterrent Burning Ground Plume area, yielded a cumulative cancer risk estimate of 2E-05. This is within EPA's target cancer risk range where remedial action or additional evaluation is typically considered unnecessary. However, the risk estimate exceeds the lower cumulative cancer risk goal of 1E-06 selected by the Army. The COCs that contributed to that exceedance were chloroform (residential well), dinitrotoluenes (monitoring well), and trichloroethene (residential well).

## **Noncancer Risk Summary**

Based on the same current offsite residential scenario, the noncancer HI for the offsite portion of the Deterrent Burning Ground Plume was 2, indicating the need for additional analysis. The only COC contributing to the HI >1 was trichloroethene (residential well).

The HI for the offsite Deterrent Burning Ground Plume area is the result of one residential well with a maximum concentration of trichloroethene (4.7  $\mu$ g/L) associated with an HQ<sup>9</sup> of 2; HQs for all other chemicals were less than 1. There were no other residential wells within the offsite Deterrent Burning Ground Plume area with chemical concentrations that would be associated with an HQ greater than 1. These results indicate that noncancer risks associated with use of groundwater from residential wells in this plume area would be within the risk management range except for the single residential well located in the Deterrent Burning Ground Plume area. However, this cumulative risk estimate was based on the maximum concentration of trichloroethene indicates that over the last twelve years (2007 through 2018) the concentration of trichloroethene in the Hendershot residential well ranged from a minimum value of 0.4  $\mu$ g/L to a maximum value of 4.7  $\mu$ g/L, with arithmetic and geometric mean concentrations of 1.2 and 0.86  $\mu$ g/L, respectively (Table 5c).

<sup>&</sup>lt;sup>9</sup> The term hazard quotient or HQ is used to represent the risk associated with a given chemical in contrast to the term hazard index or HI, which refers to the sum of the multiple chemical-specific HQs. When discussing risk of a given chemical, the term HQ is used. If a given chemical in an area has an HQ greater than 1, then by default, the HI will be greater than 1 too.

## Summary of Risks for Deterrent Burning Ground Plume

Based on the maximum risk scenario, both the onsite and offsite areas of the Deterrent Burning Ground Plume represent zones where cumulative risk estimates exceed the risk management criteria selected by the Army. The cumulative cancer and noncancer risks (i.e., HI) are summarized separately in the table below. A COPC may be a COC because it exceeds the noncancer HI and/or cumulative cancer risk management criterion. As discussed previously, for offsite areas *where groundwater use is not under the control of the Army* the risk management criteria used to determine which chemicals are COCs are a cumulative cancer risk >1E-06 or an HI >1, whereas the risk management criteria for onsite areas are cumulative cancer risk >1E-04 or an HI >1.

Location	Cumulative Cancer Risk	Noncancer Hazard Index	Contaminants of Concern*	
Onsite (Hypothetical Future Risk)	9E-05	3	1,1,2-Trichloroethane	
Offsite (Current Risk)	2E-05	2	Chloroform, Dinitrotoluenes, Trichloroethene	

#### Summary of Risk Estimates for the Deterrent Burning Ground Plume

\* A chemical is considered a contaminant of concern (COC) if either the cancer risk or the noncancer hazard quotient (HQ) for that chemical *exceeds* the risk management criteria.

## Screening Risk Evaluation for the Central Plume

Risks associated with hypothetical future and current use of groundwater were calculated based on data collected from the Central Plume in 2015, 2016, 2017, and 2018. Onsite monitoring well data were used to estimate hypothetical future onsite risks. The Army currently prohibits the use of onsite groundwater as a drinking water source and anticipates continuing this practice in the future. Therefore, it is unlikely that onsite groundwater will be used as a drinking water source in the future. Data from both offsite monitoring wells and residential drinking water wells were used to evaluate current offsite risks.

## **Future Onsite Risk Evaluation**

Table 6a summarizes the results of the onsite COPC screening for the Central Plume. Hypothetical future risks were calculated for the COPCs identified in the onsite monitoring well data. Cancer and noncancer risk estimates are summarized in Table 6b for individual COPCs and for cumulative risks. A COPC may be identified as a COC because of its noncancer and/or cancer risks.

For onsite areas *where groundwater use is under the control of the Army* the risk management criteria used to determine which chemicals are COCs were a cumulative cancer risk >1E-04 or an HI >1. The cumulative risk estimates are considered hypothetical in nature, as groundwater onsite would not be expected to be used in these areas in the future because of the restrictions placed on groundwater use by the Army.

## **Cancer Risk Summary**

Cancer risks calculated using the simple scaling method for a hypothetical future onsite residential scenario, along with the maximum observed concentration of each COPC identified in the onsite monitoring wells from the Central Plume area, yielded a cumulative cancer risk estimate of 3E-06, which is within EPA's target cancer risk range of 1E-06 to 1E-04 and meets the risk management criterion for the site. Therefore, there are no COCs based on cancer risk.

## **Noncancer Risk Summary**

Based on the same hypothetical future onsite residential scenario, the noncancer HI for the onsite portion of the Central Plume was 0.02, which meets the risk management criterion (HI $\leq$ 1). Therefore, there are no COCs for the onsite Central Plume area and additional analysis is unnecessary.

## **Current Offsite Risk Evaluation**

Table 7a summarizes the results of the offsite COPC screening for the Central Plume. Current risks were calculated for the COPCs identified in the offsite monitoring and residential well data. Cancer and noncancer risk estimates are summarized in Table 7b for individual COPCs and for cumulative risks. For offsite areas *where groundwater use is not under the control of the Army* the risk management criteria used to determine which chemicals are COCs were a cumulative cancer risk >1E-06 or an HI >1.

## **Cancer Risk Summary**

Cancer risks calculated using the simple scaling method for a current offsite residential scenario, along with the maximum observed concentration of each COPC identified in the offsite portion of the Central Plume area, yielded a cumulative cancer risk estimate of 4E-05. This is within EPA's target cancer risk range where remedial action or additional evaluation is typically considered unnecessary. However, the risk estimate exceeds the lower cumulative cancer risk criteria of 1E-06 selected by the Army. The COCs that contributed to that exceedance were 1,2-dichloroethane (monitoring well), benzene (monitoring well), chloroform (residential well), and dinitrotoluenes (monitoring well).

## **Noncancer Risk Summary**

Based on the same current offsite residential scenario, the noncancer HI for the offsite portion of the Central Plume was 0.4, which meets the risk management criterion (HI $\leq$ 1). Therefore, there are no COCs for the offsite portion of the Central Plume area and additional analysis is unnecessary.

## Summary of Risks for Central Plume

Based on the maximum risk scenario, the offsite area of the Central Plume represents a zone where cumulative risk estimates exceed the risk management criteria selected by the Army. The onsite area of the Central Plume meets the risk management criteria. The cumulative cancer and noncancer risks (i.e., HI) are summarized separately in the table below. A COPC may be a COC because it exceeds the noncancer HI and/or cumulative cancer risk management criterion. As discussed previously, for offsite areas *where groundwater use is not under the control of the Army* the risk management criteria used to determine which chemicals are COCs are a cumulative cancer risk >1E-06 or an HI >1, whereas the risk management criteria for onsite areas are cumulative cancer risk >1E-04 or an HI >1.

Location	Cumulative Cancer Risk	Noncancer Hazard Index	Contaminants of Concern*
Onsite (Hypothetical Future Risk)	3E-06	0.02	None
Offsite (Current Risk)	4E-05	0.4	1,2-Dichloroethane, Benzene, Chloroform, Dinitrotoluenes

#### Summary of Risk Estimates for the Central Plume

<sup>k</sup> A chemical is considered a contaminant of concern (COC) if either the cancer risk or the noncancer hazard quotient (HQ) for that chemical *exceeds* the risk management criteria.

# Screening Risk Evaluation for the Nitrocellulose Production Area Plume

Risks associated with hypothetical future use of groundwater were calculated based on data collected from the Nitrocellulose Production Area Plume in 2015, 2016, 2017, and 2018. Onsite monitoring well data were used to estimate hypothetical future onsite risks. The Army currently prohibits the use of onsite groundwater as a drinking water source and anticipates continuing this practice in the future. Therefore, it is unlikely that onsite groundwater will be used as a drinking water source in the future. The Nitrocellulose Production Area Plume is contained

onsite, so there are no offsite exposures associated with this plume and a current offsite risk evaluation was not conducted.

## **Future Onsite Risk Evaluation**

Table 8a summarizes the results of the onsite COPC screening for the Nitrocellulose Production Area Plume. Hypothetical future risks were calculated for the COPCs identified in the onsite monitoring well data. Cancer and noncancer risk estimates are summarized in Table 8b for individual COPCs and for cumulative risks. A COPC may be identified as a COC because of its noncancer and/or cancer risks.

For onsite areas *where groundwater use is under the control of the Army* the risk management criteria used to determine which chemicals are COCs were a cumulative cancer risk >1E-04 or an HI >1. The cumulative risk estimates are considered hypothetical in nature, as groundwater onsite would not be expected to be used in these areas in the future because of the restrictions placed on groundwater use by the Army.

## **Cancer Risk Summary**

Cancer risks calculated using the simple scaling method for a hypothetical future onsite residential scenario, along with the maximum observed concentration of each COPC identified in the onsite monitoring wells from the Nitrocellulose Production Area Plume area, yielded a cumulative cancer risk estimate of 4E-06, which is within EPA's target cancer risk range of 1E-06 to 1E-04 and meets the risk management criterion for the site. Therefore, there are no COCs based on cancer risk.

## **Noncancer Risk Summary**

Based on the same hypothetical future onsite residential scenario, the noncancer HI for the onsite Nitrocellulose Production Area Plume was 0.04, which meets the risk management criterion (HI $\leq$ 1). Therefore, there are no COCs for the onsite Nitrocellulose Production Area Plume and additional analysis is unnecessary.

## Summary of Risks for the Nitrocellulose Production Area Plume

Based on the maximum risk scenario, the onsite area of the Nitrocellulose Production Area Plume meets the risk management criteria selected by the Army and there are no COCs. This plume is contained onsite and so there is no offsite exposure. The cumulative cancer and noncancer risks (i.e., HI) associated with hypothetical future onsite exposures are summarized separately in the table below.

Location	Cumulative Cancer Risk	Noncancer Hazard Index	Contaminants of Concern*
Onsite (Hypothetical Future Risk)	4E-06	0.04	None
Offsite (Current Risk)	NA	NA	NA

\* A chemical is considered a contaminant of concern (COC) if either the cancer risk or the noncancer hazard quotient (HQ) for that chemical *exceeds* the risk management criteria.

## Summary of Hypothetical Future and Current Groundwater Risks

A groundwater risk evaluation was conducted to estimate cumulative risks associated with hypothetical future residential exposure to onsite groundwater and current exposure to offsite groundwater. The hypothetical groundwater risks were evaluated using monitoring well data collected from four plume areas located onsite, where the Army maintains control over the use of the groundwater, and therefore, residential wells are not expected to be constructed in the future. Current risks were estimated by evaluating groundwater data collected for residential wells and offsite monitoring wells sampled downgradient from three plume areas that have migrated offsite. Groundwater data from 2015, 2016, 2017, and 2018 were used for the initial screening level risk evaluation to best represent current and future groundwater quality for each evaluation unless otherwise noted.

## Hypothetical Future Onsite Groundwater Risk Evaluation

Based on the results of the future onsite groundwater risk evaluation of the monitoring well data in each plume area, the potential for future risks was evaluated under the hypothetical scenario that residential wells are constructed in each plume area where no residential wells currently exist. For onsite areas where groundwater use is under the control of the Army, the risk management criteria used to determine which chemicals are COCs were a cumulative cancer risk >1E-04 or an HI >1. The cumulative risk estimates are considered hypothetical in nature, as groundwater onsite would not be expected to be used in these areas in the future because of the restrictions placed on groundwater use by the Army.

Based on the maximum risk scenario for each of four onsite plume areas using the monitoring well data, the Propellant Burning Ground Plume and Deterrent Burning Ground Plume areas represent zones that would be associated with cumulative risks that exceed the risk management criteria if onsite groundwater were used as a source of residential drinking water in the future. The onsite portions of the Propellant Burning Ground Plume and Deterrent Burning Ground Plume exceed the noncancer criterion (HI>1), whereas only the Propellant Burning Ground Plume exceeds the cancer criterion (>1E-04).

The following COCs were identified in onsite areas:

- Propellant Burning Ground Plume ethyl ether, dinitrotoluenes, and trichloroethene
- Deterrent Burning Ground Plume 1,1,2-trichloroethane.

The cumulative risk estimates meet the risk management criteria in the onsite portions of the Central Plume and Nitrocellulose Production Area Plume, and no COCs were identified.

## **Current Offsite Groundwater Risk Evaluation**

Based on the maximum risk scenario for the offsite portions of the Propellant Burning Ground Plume, Deterrent Burning Ground Plume, and the Central Plume using both monitoring well and private residential well data, all three plume areas are associated with cumulative risks that exceed the risk management criteria. As noted previously, the Army has elected to use a stricter cancer risk management criterion in offsite areas (>1E-06) to provide an extra level of public health protection. All three plumes exceed this lower cancer risk management criterion, whereas only the Propellant Burning Ground Plume and Deterrent Burning Ground Plume exceed the noncancer criterion (HI>1). No risk evaluation of the Nitrocellulose Production Area Plume was necessary offsite, because this plume is contained onsite and so has not affected offsite residential areas.

The following COCs were identified in offsite areas:

- Propellant Burning Ground Plume carbon tetrachloride, chloroform, dinitrotoluenes, and trichloroethene
- Deterrent Burning Ground Plume chloroform, dinitrotoluenes, and trichloroethene
- Central Plume 1,2-dichloroethane, benzene, chloroform, and dinitrotoluenes.

## References

U.S. EPA. 1989. Risk assessment guidance for Superfund—Volumes I and II: human health evaluation manual (Part A). Interim final. EPA/540/1-89/002. U.S. Environmental Protection Agency, Office of Solid Waste and Remedial Response, Washington, DC.

U.S. EPA. 1990. National oil and hazardous substances pollution contingency plan. U.S. Environmental Protection Agency, Office of Solid Waste and Emergency Response. Washington, DC. (40 CFR Part 300, revised March 1990.) U.S. EPA. 1991. Internal memorandum from D.R. Clay to Directors, Waste Management Division, Regions I, IV, V, VII, VIII; Director, Emergency and Remedial Response Division, Region II; Directors, Hazardous Waste Management Division, Regions III, VI, IX; and Director, Hazardous Waste Division, Region X, regarding role of the baseline risk assessment in Superfund remedy selection decision, dated April 22, 1991. U.S. Environmental Protection Agency, Office of Solid Waste and Emergency Response, Washington, DC.

U.S. EPA. 2018. Regional Screening Levels (RSLs) - Generic Tables (November 2018). Available at: <u>https://www.epa.gov/risk/regional-screening-levels-rsls-generic-tables</u>. Updated November 2018. U.S. Environmental Protection Agency.

WDNR. 2017. Chapter NR 140. Groundwater quality. Wisconsin Administrative Code Register, No. 715. Available at: <u>https://docs.legis.wisconsin.gov/code/admin\_code/nr/100/140</u>. Wisconsin Department of Natural Resources. February.

## Tables

## Table 1. Summary of Groundwater Screening Levels Used for the Screening Level Groundwater Risk Evaluation Screening Level Groundwater Risk Evaluation (Draft)

Badger Army Ammunition Plant

CAS         PARAPYE         Value         NAC         INC-011         CS         PAR         Office           71-55-6         1.1.1-Trichloroethane         0.041         0.28         0.041         5         0.5         µg/L           73-34         1.1-Dichloroethane         0.28         2.8         380         850         85         µg/L           95-35-4         1.2-Dichloroethane         0.7         NA         28         480         96         µg/L           95-50-1         1.2-Dichloroethane         0.17         0.17         1.3         5         0.5         µg/L           107-66-2         1.2-Dichloroptopane         0.5         0.85         0.82         5         0.5         µg/L           108-67-8         1.3.5-Trimethylbenzene         6         NA         6         480         96         µg/L           121-14-2         2.4-Dinitrotoluene         NA         Ig/L         Ig/L	CAS	Analista	Minimum	EPA Cancer- based Tapwater	EPA Noncancer- based Tapwater RSL (Based on	NR 140	NR 140	Linita
J.1-5-b.         J.1.1-inchloroethane         40         NA         800         200         40         Hg/L           75-36-3         J.1.2-Trichloroethane         2.8         2.8         380         850         85         µg/L           75-34-3         J.1-Dichloroethane         0.7         NA         2.8         380         850         85         µg/L           95-36-1         J.2-Dichloroethane         0.7         NA         30         600         60         µg/L           95-36-1         J.2-Dichloroopane         0.5         0.85         0.82         5         0.5         µg/L           107-06-2         J.2-Dichloroopane         0.5         0.85         0.82         5         0.5         µg/L           102-01-7         2.3-Dinitrotoluene         NA         NA         NA         NA         NA         NA         NA         µg/L           11-14-2         2.4-Dinitrotoluene         NA         NA         NA         NA         NA         NA         µg/L           66-20-2         2.6-Dinitrotoluene         NA         NA         NA         NA         NA         µg/L           619-33-9         3.5-Dinitrotoluene         NA         NA		Analyte	Value	KSL	1HQ=0.1)	ES	PAL	Units
(J-00-5)         1.1.2-inthrootentane         0.041         0.2.8         (0.041)         5.         0.5         1.40           57-34-3         1.10-inthrootentane         0.7         NA         2.8         380         850         85         1.9           95-63-6         1.2,4-Trimethylbenzene         5.6         NA         3.6         480         96         1.42           107-06-2         1.2-0ichtoroethane         0.17         0.17         1.3         5         0.5         1.42           108-67-8         1.3.5-Trimethylbenzene         6         NA         6         480         96         1.42/.           108-67-8         1.3.5-Trimethylbenzene         6         NA	71-55-6	1,1,1-Trichloroethane	40	NA 0.20	800	200	40	μg/L
/>3-3         1,1-bickhoroethene         2.8         2.8         380         850         85         µg/L           95-53-4         1,2-bickhoroethene         3.0         NA         3.0         600         60         µg/L           95-63-6         1,2-bickhoroethane         3.0         NA         3.0         600         60         µg/L           95-50-1         1,2-bickhoroethane         0.17         0.17         1.3         5         0.5         µg/L           107-65-2         1,2-bickhoroethane         0.17         0.17         1.3         5         0.5         µg/L           121-14-2         2,3-binitrotoluene         NA	79-00-5	1,1,2-Trichloroethane	0.041	0.28	0.041	5	0.5	µg/L
Y=33-4         1,1-10000000000000000000000000000000000	75-34-3	1,1-Dichloroethane	2.8	2.8	380	850	85	µg/L
95-63-6         1,2,4-11metrybehzene         5.6         NA         5.6         480         96         1g/L           107-06-2         1,2-Dichlorobenzene         30         NA         30         600         60         lg/L           107-06-2         1,2-Dichloropropane         0.5         0.85         0.82         5         0.5         lg/L           108-67-8         1,3-5-Timetryblenzene         6         NA         6         480         NA         S0 <t< td=""><td>75-35-4</td><td>1,1-Dichloroethene</td><td>0.7</td><td>NA</td><td>28</td><td>/</td><td>0.7</td><td>μg/L</td></t<>	75-35-4	1,1-Dichloroethene	0.7	NA	28	/	0.7	μg/L
95-50-1         1,2-Dichoroberhane         30         NA         30         600         60         µµ/L           78-75         1,2-Dichoroperhane         0.17         0.17         1.3         5         0.5         µµ/L           108-67-8         1,3,5-Trimethylbenzene         6         NA	95-63-6	1,2,4-Trimethylbenzene	5.6	NA	5.6	480	96	μg/L
10/-bc-2         1,2-bichloroethane         0.17         0.17         1.3         5         0.5         µg/L           108-67-8         1,3,5-Trimethylbenzene         6         NA         6         480         96         µg/L           108-67-8         1,3,5-Trimethylbenzene         6         NA         MA         µg/L         606-20-2         2,6-Dinitrotoluene         NA         µg/L         618-85-9         3,5-Dinitrotoluene         NA         NA         NA         NA         NA         NA         NA         µg/L         7143-2         Bernzene         0.46         0.46         3.3         5         0.5         µg/L         75-15-0         Carbon disulfide         81         NA         81         1000         200	95-50-1	1,2-Dichlorobenzene	30	NA	30	600	60	μg/L
78-87-5         1,2-Ditchloropropane         0.5         0.85         0.82         5         0.5         µg/L           602-01-7         2,3-Dinitrotoluene         NA         µg/L         6:6:13:3         5         0.5         µg/L         6:7:2:7:4         Bromodichloromethane         0.06         0.13         38         0.6         0.06         µg/L         7:5:2:7:4         Bromodichloromethane         0.46         0.46         4.9         5         0.5 <t< td=""><td>107-06-2</td><td>1,2-Dichloroethane</td><td>0.17</td><td>0.17</td><td>1.3</td><td>5</td><td>0.5</td><td>μg/L</td></t<>	107-06-2	1,2-Dichloroethane	0.17	0.17	1.3	5	0.5	μg/L
108-67-8         1,3,5-Trimethylbenzene         6         NA         6         480         96         µg/L           602-01-7         2,3-Dinitrotoluene         NA         IA	78-87-5	1,2-Dichloropropane	0.5	0.85	0.82	5	0.5	μg/L
602-01-7         2,3-Dinitrotoluene         NA         L <thl< th="">         L         L</thl<>	108-67-8	1,3,5-Trimethylbenzene	6	NA	6	480	96	μg/L
1211-14-2         2,4-Dinitrotoluene         0.005         0.24         3.8         0.05         0.005         µ/L           619-15-8         2,5-Dinitrotoluene         NA         NA         NA         NA         NA         NA         NA         NA         µ/L           78-93-3         2-Butanone         560         NA         S60         4000         800         µ/L           610-39-9         3,4-Dinitrotoluene         NA         µ/L         Catoro         µ/L         Z/S-27-4         Bromodichloromthane         0.06         0.13         38         0.6         0.05         µ/L           57-15-0         Carbon tetrachloride         0.46         0.46         4.9         5         0.5         µ/L           57-15-2         Carbon disufide         80         NA         2100         400         80         µ/L           56-59-2         Cis-1,2-Dichloromethane         3.6	602-01-7	2,3-Dinitrotoluene	NA	NA	NA	NA	NA	μg/L
619-15-8         2,5-Dinitrotoluene         NA         NA         NA         NA         NA         NA         µZ/L           606-20-2         2,6-Dinitrotoluene         0.005         0.049         0.57         0.05         0.005         µg/L           610-39-9         3,4-Dinitrotoluene         NA         µg/L           618-85-9         3,5-Dinitrotoluene         NA         IAA         NA         IAA         NA         NA         IAA         µg/L           67-64-1         Acetone         1400         NA         1400         9000         1800         µg/L           71-32-2         Benzene         0.46         0.46         0.46         4.9         5         0.5         µg/L           56-23-5         Carbon tetrachloride         0.46         0.46         4.9         5         0.5         µg/L           75-03-3         Chloroethane         33         NA         190         30         3         µg/L	121-14-2	2,4-Dinitrotoluene	0.005	0.24	3.8	0.05	0.005	μg/L
606-20-2         2,6-Dinitrotoluene         0.005         0.049         0.57         0.05         0.005         µg/L           78-93-3         2-Butanone         560         NA         560         4000         800         µg/L           610-39-9         3,4-Dinitrotoluene         NA         SA         0.5         µg/L         7.5.7         Catoro disulfide         81         NA         81         1000         200         µg/L         75.5.0         Catoro disulfide         0.46         0.46         4.9         5         0.5         µg/L         7.6         6.6         µg/L         7.487.3         Chloromethane         0.22         0.22         9.7         6         0.6         µg/L	619-15-8	2,5-Dinitrotoluene	NA	NA	NA	NA	NA	μg/L
78-93-3         2-Butanone         560         NA         560         4000         800         µµ/L           610-39-9         3,4-Dinitrotoluene         NA         MA         NA         NA         NA         NA         NA         NA         NA         MA         NA         NA         NA         NA         MA         NA         NA         MA         NA         MA         NA         MA         NA         MA         NA         MA         NA         MA         MA         NA         MA	606-20-2	2,6-Dinitrotoluene	0.005	0.049	0.57	0.05	0.005	μg/L
610-39-9         3,4-Dinitrotoluene         NA         NA         NA         NA         NA         NA         µg/L           618-85-9         3,5-Dinitrotoluene         NA         NA         NA         NA         NA         µg/L           71-43-2         Benzene         0.46         0.46         3.3         5         0.5         µg/L           71-43-2         Benzene         0.46         0.46         3.3         5         0.5         µg/L           75-27-4         Bromodichloromethane         0.06         0.13         38         0.6         0.06         µg/L           56-23-5         Carbon tetrachloride         0.46         0.46         4.9         5         0.5         µg/L           75-00-3         Chloroform         0.22         0.22         9.7         6         0.6         µg/L           74-87-3         Chloromethane         3.6         NA         19         30         3         µg/L           124-48-1         Dibromochloromethane         0.87         0.87         38         60         6         µg/L           75-43-4         Dichlorofluoromethane         NA         NA         NA         NA         µg/L	78-93-3	2-Butanone	560	NA	560	4000	800	μg/L
618-85-9         3,5-Dinitrotoluene         NA         NA         NA         NA         NA         MA         µg/L           67-64-1         Acetone         1400         NA         1400         9000         1800         µg/L           71-43-2         Benzene         0.46         0.46         3.3         5         0.5         µg/L           75-27-4         Bromodichloromethane         0.06         0.13         38         0.6         0.06         µg/L           56-23-5         Carbon tetrachloride         0.46         0.46         4.9         5         0.5         µg/L           75-00-3         Chloromethane         80         NA         2100         400         80         µg/L           76-65-3         Chloromethane         3.         NA         19         30         3         µg/L           126-59-2         cis-1,2-Dichloroethane         3.6         NA         3.6         70         7         µg/L           126-48-1         Dichlorofluoromethane         0.87         0.87         38         60         6         µg/L           75-43-4         Dichlorofluoromethane         NA         NA         NA         NA         NA         µg/L <td>610-39-9</td> <td>3,4-Dinitrotoluene</td> <td>NA</td> <td>NA</td> <td>NA</td> <td>NA</td> <td>NA</td> <td>μg/L</td>	610-39-9	3,4-Dinitrotoluene	NA	NA	NA	NA	NA	μg/L
67-64-1         Acetone         1400         NA         1400         9000         1800         µg/L           71-43-2         Benzene         0.46         0.46         3.3         5         0.5         µg/L           75-27-4         Bromodichloromethane         0.06         0.13         38         0.6         0.06         µg/L           55-15-0         Carbon disulfide         81         NA         81         1000         200         µg/L           56-23-5         Carbon tetrachloride         0.46         0.46         4.9         5         0.5         µg/L           67-66-3         Chloroethane         80         NA         2100         400         80         µg/L           74-87-3         Chloromethane         3         NA         19         30         3         µg/L           126-59-2         cis-1,2-Dichloroethene         3.6         NA         36         70         7         µg/L           124-48-1         Dichorofibuoromethane         0.87         0.87         38         60         6         µg/L           124-48-1         Dichlorofibuoromethane         NA         NA         NA         NA         µg/L           124-38 </td <td>618-85-9</td> <td>3,5-Dinitrotoluene</td> <td>NA</td> <td>NA</td> <td>NA</td> <td>NA</td> <td>NA</td> <td>μg/L</td>	618-85-9	3,5-Dinitrotoluene	NA	NA	NA	NA	NA	μg/L
21-43-2         Benzene         0.46         0.46         3.3         5         0.5         µg/L           75-27-4         Bromodichloromethane         0.06         0.13         38         0.6         0.06         µg/L           75-15-0         Carbon disulfide         81         NA         81         1000         200         µg/L           56-23-5         Carbon tetrachloride         0.46         0.46         4.9         5         0.5         µg/L           67-66-3         Chloroethane         80         NA         2100         400         80         µg/L           74-87-3         Chloromethane         3.         NA         19         30         3         µg/L           126-59-2         cis-1,2-Dichloroethene         3.6         NA         3.6         70         7         µg/L           124-48-1         Dichlorofluoromethane         0.87         0.87         38         60         6         µg/L           60-29-7         Ethyl ether         100         NA         390         1000         100         µg/L           100-41-4         Ethylbenzene         1.5         1.5         81         700         140         µg/L	67-64-1	Acetone	1400	NA	1400	9000	1800	μg/L
75-27-4         Bromodichloromethane         0.06         0.13         38         0.6         0.06         µg/L           75-15-0         Carbon disulfide         81         NA         81         1000         200         µg/L           56-23-5         Carbon tetrachloride         0.46         0.46         4.9         5         0.5         µg/L           56-03-3         Chloroform         0.22         0.22         9.7         6         0.6         µg/L           74-87-3         Chloromethane         3         NA         19         30         3         µg/L           126-59-2         cis-1,2-Dichloroethane         3.6         NA         3.6         70         7         µg/L           124-48-1         Dibromochloromethane         0.87         0.87         38         60         6         µg/L           124-48-1         Dichlorofiluoromethane         0.87         0.87         38         60         6         µg/L           124-48-1         Dichlorofiluoromethane         0.87         0.87         38         60         6         µg/L           126-3-3         Ethyl ether         100         NA         NA         NA         NA         µg/L     <	71-43-2	Benzene	0.46	0.46	3.3	5	0.5	μg/L
75-15-0         Carbon disulfide         81         NA         81         1000         200         µg/L           56-23-5         Carbon tetrachloride         0.46         0.46         4.9         5         0.5         µg/L           75-00-3         Chloroethane         80         NA         2100         400         80         µg/L           67-66-3         Chloroform         0.22         0.22         9.7         6         0.6         µg/L           74-87-3         Chloromethane         3         NA         19         30         3         µg/L           124-48-1         Dibromochloroethane         3.6         NA         3.6         70         7         µg/L           124-48-1         Dichlorodifluoromethane         0.87         0.87         38         60         6         µg/L           75-71-8         Dichlorofluoromethane         NA         NA         NA         NA         NA         NA         NA         µg/L           60-29-7         Ethyl ether         100         NA         390         1000         100         µg/L           98-82-8         Isoproylbenzene         45         NA         45         NA         NA <td< td=""><td>75-27-4</td><td>Bromodichloromethane</td><td>0.06</td><td>0.13</td><td>38</td><td>0.6</td><td>0.06</td><td>μg/L</td></td<>	75-27-4	Bromodichloromethane	0.06	0.13	38	0.6	0.06	μg/L
56-23-5         Carbon tetrachloride         0.46         0.46         4.9         5         0.5         µg/L           75-00-3         Chloroethane         80         NA         2100         400         80         µg/L           67-66-3         Chloroform         0.22         0.22         9.7         6         0.6         µg/L           74-87-3         Chloromethane         3         NA         19         30         3         µg/L           156-59-2         cis-1,2-Dichloroethene         3.6         NA         36         70         7         µg/L           124-48-1         Dibromochloromethane         0.87         0.87         38         60         6         µg/L           75-71-8         Dichlorofluoromethane         20         NA         20         1000         200         µg/L           60-29-7         Ethyl ether         100         NA         390         1000         100         µg/L           10-41-4         Ethylbenzene         1.5         1.5         81         700         140         µg/L           192-20-3         Naphthalene         0.17         0.61         100         µg/L         14797-55-8         Nitrate         2 </td <td>75-15-0</td> <td>Carbon disulfide</td> <td>81</td> <td>NA</td> <td>81</td> <td>1000</td> <td>200</td> <td>μg/L</td>	75-15-0	Carbon disulfide	81	NA	81	1000	200	μg/L
75-00-3         Chloroethane         80         NA         2100         400         80         µg/L           67-66-3         Chloroform         0.22         0.22         9.7         6         0.6         µg/L           74-87-3         Chloromethane         3         NA         19         30         3         µg/L           156-59-2         cis-1,2-Dichloroethene         3.6         NA         3.5         70         7         µg/L           124-48-1         Dibromochloromethane         0.87         0.87         38         60         6         µg/L           75-71-8         Dichlorofluoromethane         20         NA         20         1000         200         µg/L           10-41-4         Ethyl ether         100         NA         390         1000         100         µg/L           10-41-4         Ethyl benzene         1.5         1.5         81         700         140         µg/L           19601-23-1         m & p-Xylene         19         NA         19         2000         400         µg/L           197905-58         Nitrate         2         NA         3.2         10         2         mg/L           103-65-1 <td>56-23-5</td> <td>Carbon tetrachloride</td> <td>0.46</td> <td>0.46</td> <td>4.9</td> <td>5</td> <td>0.5</td> <td>μg/L</td>	56-23-5	Carbon tetrachloride	0.46	0.46	4.9	5	0.5	μg/L
67-66-3         Chloroform         0.22         0.22         9.7         6         0.6         μg/L           74-87-3         Chloromethane         3         NA         19         30         3         μg/L           156-59-2         cis-1,2-Dichloroethene         3.6         NA         3.6         70         7         μg/L           124-48-1         Dibromochloromethane         0.87         0.87         38         60         6         μg/L           75-71-8         Dichlorodifluoromethane         0.87         0.87         38         60         6         μg/L           60-29-7         Ethyl ether         100         NA         390         1000         100         μg/L           10-41-4         Ethylbenzene         1.5         1.5         81         700         140         μg/L           98-82-8         Isopropylbenzene         45         NA         45         NA         NA         μg/L           91-20-3         Naphthalene         0.17         0.17         0.61         100         μg/L           137-65-1         n-Propylbenzene         66         NA         66         NA         NA         μg/L           10-42-5	75-00-3	Chloroethane	80	NA	2100	400	80	μg/L
74-87-3         Chloromethane         3         NA         19         30         3         µg/L           156-59-2         cis-1,2-Dichloroethene         3.6         NA         3.6         70         7         µg/L           124-48-1         Dibromochloromethane         0.87         0.87         38         60         6         µg/L           75-71-8         Dichlorodifluoromethane         20         NA         20         1000         200         µg/L           60-29-7         Ethyl ether         100         NA         390         1000         100         µg/L           100-41-4         Ethylbenzene         1.5         1.5         81         700         140         µg/L           98-82-8         Isopropylbenzene         45         NA         45         NA         NA         µg/L           91-20-3         Naphthalene         0.17         0.17         0.61         100         10         µg/L           179601-23-1         m & p-Xylene         19         NA         3.2         10         2         mg/L           179755-8         Nitrate         2         NA         3.2         10         2         mg/L           100-42-5	67-66-3	Chloroform	0.22	0.22	9.7	6	0.6	μg/L
156-59-2         cis-1,2-Dichloroethene         3.6         NA         3.6         70         7         μg/L           124-48-1         Dibromochloromethane         0.87         0.87         38         60         6         μg/L           75-71-8         Dichlorodifluoromethane         20         NA         20         1000         200         μg/L           75-43-4         Dichlorofluoromethane         NA         NA         NA         NA         NA         NA         NA         NA         MA         NA         MA         NA         MA         NA         MA         MA         NA         NA         MA         MA         NA         MA         NA         MA         MA         MA         MA         NA         MA         µg/L         Mg/L	74-87-3	Chloromethane	3	NA	19	30	3	μg/L
124-48-1         Dibromochloromethane         0.87         0.87         38         60         6         μg/L           75-71-8         Dichlorodifluoromethane         20         NA         20         1000         200         μg/L           75-43-4         Dichlorofluoromethane         NA         MA         NA         NA         NA         MA         NA         NA         MA         NA         NA         MA         MA         NA         MA	156-59-2	cis-1,2-Dichloroethene	3.6	NA	3.6	70	7	μg/L
75-71-8         Dichlorodifluoromethane         20         NA         20         1000         200         µg/L           75-43-4         Dichlorofluoromethane         NA         µg/L           60-29-7         Ethyl ether         100         NA         390         1000         100         µg/L           100-41-4         Ethylbenzene         1.5         1.5         81         700         140         µg/L           98-82-8         Isopropylbenzene         45         NA         45         NA         NA         µg/L           179601-23-1         m & p-Xylene         19         NA         19         2000         400         µg/L           14797-55-8         Nitrate         2         NA         3.2         10         2         mg/L           103-65-1         n-Propylbenzene         66         NA         66         NA         NA         µg/L           100-42-5         Styrene         10         NA         120         100         10         µg/L           <	124-48-1	Dibromochloromethane	0.87	0.87	38	60	6	μg/L
75-43-4         Dichlorofluoromethane         NA         NA         NA         NA         NA         NA         NA         µg/L           60-29-7         Ethyl ether         100         NA         390         1000         100         µg/L           100-41-4         Ethylbenzene         1.5         1.5         81         700         140         µg/L           98-82-8         Isopropylbenzene         45         NA         45         NA         NA         µg/L           179601-23-1         m & p-Xylene         19         NA         19         2000         400         µg/L           179601-23-1         m & p-Xylene         19         NA         19         2000         400         µg/L           14797-55-8         Nitrate         2         NA         3.2         10         2         mg/L           103-65-1         n-Propylbenzene         66         NA         66         NA         NA         µg/L           100-42-5         Styrene         10         NA         120         100         10         µg/L           127-18-4         Tetrachloroethene         69         NA         A69         NA         NA         µg/L	75-71-8	Dichlorodifluoromethane	20	NA	20	1000	200	μg/L
60-29-7         Ethyl ether         100         NA         390         1000         100         µg/L           100-41-4         Ethylbenzene         1.5         1.5         81         700         140         µg/L           98-82-8         Isopropylbenzene         45         NA         45         NA         NA         µg/L           179601-23-1         m & p-Xylene         19         NA         19         2000         400         µg/L           91-20-3         Naphthalene         0.17         0.17         0.61         100         10         µg/L           14797-55-8         Nitrate         2         NA         3.2         10         2         mg/L           103-65-1         n-Propylbenzene         66         NA         66         NA         NA         µg/L           105-47-6         o-Xylene         19         NA         19         2000         400         µg/L           100-42-5         Styrene         10         NA         120         100         10         µg/L           14808-79-8         Sulfate         125         NA         NA         250         125         mg/L           10-42-5         Styrene	75-43-4	Dichlorofluoromethane	NA	NA	NA	NA	NA	μg/L
100-41-4         Ethylbenzene         1.5         1.5         81         700         140         µg/L           98-82-8         Isopropylbenzene         45         NA         45         NA         NA         µg/L           179601-23-1         m & p-Xylene         19         NA         19         2000         400         µg/L           91-20-3         Naphthalene         0.17         0.17         0.61         100         10         µg/L           14797-55-8         Nitrate         2         NA         3.2         10         2         mg/L           103-65-1         n-Propylbenzene         66         NA         66         NA         NA         µg/L           95-47-6         o-Xylene         19         NA         19         2000         400         µg/L           100-42-5         Styrene         10         NA         120         100         10         µg/L           14808-79-8         Sulfate         125         NA         NA         250         125         mg/L           10-42-5         Styrene         0.5         11         4.1         5         0.5         µg/L           14808-79-8         Sulfate	60-29-7	Ethyl ether	100	NA	390	1000	100	μg/L
98-82-8         Isopropylbenzene         45         NA         45         NA         NA         µg/L           179601-23-1         m & p-Xylene         19         NA         19         2000         400         µg/L           91-20-3         Naphthalene         0.17         0.17         0.61         100         10         µg/L           14797-55-8         Nitrate         2         NA         3.2         10         2         mg/L           103-65-1         n-Propylbenzene         66         NA         66         NA         NA         µg/L           95-47-6         o-Xylene         19         NA         19         2000         400         µg/L           100-42-5         Styrene         10         NA         120         100         10         µg/L           14808-79-8         Sulfate         125         NA         NA         250         125         mg/L           127-18-4         Tetrachloroethene         0.5         11         4.1         5         0.5         µg/L           108-88-3         Toluene         110         NA         340         50         10         µg/L           12521-14-6         Total Dinitro	100-41-4	Ethylbenzene	1.5	1.5	81	700	140	μg/L
179601-23-1m & p-Xylene19NA192000400μg/L91-20-3Naphthalene0.170.170.6110010μg/L14797-55-8Nitrate2NA3.2102mg/L103-65-1n-Propylbenzene66NA66NANAμg/L95-47-6o-Xylene19NA192000400μg/L100-42-5Styrene10NA12010010μg/L14808-79-8Sulfate125NANA250125mg/L98-06-6tert-Butylbenzene69NA69NANAμg/L127-18-4Tetrachloroethene0.5114.150.5μg/L108-88-3Toluene110NA110800160μg/L156-60-5trans-1,2-Dichloroethene20NA3610020μg/L75-69-4Trichlorofluoromethane520NA5015μg/L	98-82-8	Isopropylbenzene	45	NA	45	NA	NA	μg/L
91-20-3         Naphthalene         0.17         0.17         0.61         100         µg/L           14797-55-8         Nitrate         2         NA         3.2         10         2         mg/L           103-65-1         n-Propylbenzene         66         NA         66         NA         NA         µg/L           95-47-6         o-Xylene         19         NA         19         2000         400         µg/L           100-42-5         Styrene         10         NA         120         100         10         µg/L           14808-79-8         Sulfate         125         NA         NA         250         125         mg/L           14808-79-8         Sulfate         125         NA         NA         250         125         mg/L           14808-79-8         Sulfate         125         NA         NA         250         125         mg/L           127-18-4         Tetrachloroethene         0.5         11         4.1         5         0.5         µg/L           109-99-9         Tetrahydrofuran         10         NA         340         50         10         µg/L           108-88-3         Toluene         110	179601-23-1	m & p-Xvlene	19	NA	19	2000	400	ug/L
14797-55-8         Nitrate         2         NA         3.2         10         2         mg/L           103-65-1         n-Propylbenzene         66         NA         66         NA         NA         µg/L           95-47-6         o-Xylene         19         NA         19         2000         400         µg/L           100-42-5         Styrene         10         NA         120         100         10         µg/L           14308-79-8         Sulfate         125         NA         NA         250         125         mg/L           14308-79-8         Sulfate         125         NA         NA         250         125         mg/L           127-18-4         Tetrachloroethene         0.5         11         4.1         5         0.5         µg/L           109-99-9         Tetrahydrofuran         10         NA         340         50         10         µg/L           108-88-3         Toluene         110         NA         110         800         160         µg/L           156-60-5         trans-1,2-Dichloroethene         20         NA         36         100         20         µg/L           156-60-5         trans-1,2-	91-20-3	Naphthalene	0.17	0.17	0.61	100	10	ug/L
103-65-1         n-Propylbenzene         66         NA         66         NA         NA         μg/L           95-47-6         o-Xylene         19         NA         19         2000         400         μg/L           100-42-5         Styrene         10         NA         120         100         10         μg/L           14808-79-8         Sulfate         125         NA         NA         250         125         mg/L           98-06-6         tert-Butylbenzene         69         NA         69         NA         NA         μg/L           127-18-4         Tetrachloroethene         0.5         11         4.1         5         0.5         μg/L           109-99-9         Tetrahydrofuran         10         NA         340         50         10         μg/L           108-88-3         Toluene         110         NA         110         800         160         μg/L           156-60-5         trans-1,2-Dichloroethene         20         NA         36         100         20         μg/L           79-01-6         Trichloroethene         0.28         0.49         0.28         5         0.5         μg/L           75-69-4	14797-55-8	Nitrate	2	NA	3.2	10	2	mg/L
95-47-6         o-Xylene         19         NA         19         2000         400         μg/L           100-42-5         Styrene         10         NA         120         100         10         μg/L           14808-79-8         Sulfate         125         NA         NA         250         125         mg/L           98-06-6         tert-Butylbenzene         69         NA         69         NA         NA         μg/L           127-18-4         Tetrachloroethene         0.5         11         4.1         5         0.5         μg/L           109-99-9         Tetrachloroethene         10         NA         340         50         10         μg/L           108-88-3         Toluene         110         NA         110         800         160         μg/L           25321-14-6         Total Dinitrotoluenes         0.005         0.1         1.1         0.05         0.005         μg/L           156-60-5         trans-1,2-Dichloroethene         20         NA         36         100         20         μg/L           79-01-6         Trichloroethene         0.28         0.49         0.28         5         0.5         μg/L <td< td=""><td>103-65-1</td><td>n-Propylbenzene</td><td>66</td><td>NA</td><td>66</td><td>NA</td><td>NA</td><td>ug/L</td></td<>	103-65-1	n-Propylbenzene	66	NA	66	NA	NA	ug/L
International and the second structure         International and the second structure <thinternatinget structure<="" th=""> <thinternational and="" se<="" td="" the=""><td>95-47-6</td><td>o-Xvlene</td><td>19</td><td>NA</td><td>19</td><td>2000</td><td>400</td><td>ug/L</td></thinternational></thinternatinget>	95-47-6	o-Xvlene	19	NA	19	2000	400	ug/L
14808-79-8         Sulfate         125         NA         NA         250         125         mg/L           98-06-6         tert-Butylbenzene         69         NA         69         NA         NA         µg/L           127-18-4         Tetrachloroethene         0.5         11         4.1         5         0.5         µg/L           109-99-9         Tetrachloroethene         10         NA         340         50         10         µg/L           108-88-3         Toluene         110         NA         110         800         160         µg/L           25321-14-6         Total Dinitrotoluenes         0.005         0.1         1.1         0.05         0.005         µg/L           156-60-5         trans-1,2-Dichloroethene         20         NA         36         100         20         µg/L           79-01-6         Trichloroethene         0.28         0.49         0.28         5         0.5         µg/L           75-69-4         Trichlorofluoromethane         520         NA         520         3490         698         µg/L	100-42-5	Styrene	10	NA	120	100	10	ug/L
98-06-6         tert-Butylbenzene         69         NA         69         NA         NA         µg/L           127-18-4         Tetrachloroethene         0.5         11         4.1         5         0.5         µg/L           109-99-9         Tetrachloroethene         10         NA         340         50         10         µg/L           108-88-3         Toluene         110         NA         110         800         160         µg/L           25321-14-6         Total Dinitrotoluenes         0.005         0.1         1.1         0.05         0.005         µg/L           156-60-5         trans-1,2-Dichloroethene         20         NA         36         100         20         µg/L           79-01-6         Trichloroethene         0.28         0.49         0.28         5         0.5         µg/L           75-69-4         Trichlorofluoromethane         520         NA         520         3490         698         µg/L	14808-79-8	Sulfate	125	NA	NA	250	125	mg/L
127-18-4         Tetrachloroethene         0.5         11         4.1         5         0.5         µg/L           109-99-9         Tetrahydrofuran         10         NA         340         50         10         µg/L           108-88-3         Toluene         110         NA         110         800         160         µg/L           25321-14-6         Total Dinitrotoluenes         0.005         0.1         1.1         0.05         0.005         µg/L           156-60-5         trans-1,2-Dichloroethene         20         NA         36         100         20         µg/L           79-01-6         Trichloroethene         0.28         0.49         0.28         5         0.5         µg/L           75-69-4         Trichlorofluoromethane         520         NA         520         3490         698         µg/L	98-06-6	tert-Butylbenzene	69	NA	69	NA	NA	цд/І
International construction         International constructinction         International construction	127-18-4	Tetrachloroethene	0.5	11	4.1		0.5	цд/Г
100         100         104         100 <td>109-99-9</td> <td>Tetrabydrofuran</td> <td>10</td> <td>ΝΔ</td> <td>340</td> <td>50</td> <td>10</td> <td>11g/l</td>	109-99-9	Tetrabydrofuran	10	ΝΔ	340	50	10	11g/l
100 color         110         100         110         100         1	108-88-3	Toluene	110	NA	110	800	160	110/1
11         0.03         0.003         0.03         0.003         µg/L           156-60-5         trans-1,2-Dichloroethene         20         NA         36         100         20         µg/L           79-01-6         Trichloroethene         0.28         0.49         0.28         5         0.5         µg/L           75-69-4         Trichloroethane         520         NA         520         3490         698         µg/L	25321-14-6	Total Dinitrotoluenes	0.005	0.1	1 1	0.05	0.005	- νö/⊏ ⊔σ/Ι
Tool of an ans-s/2-bindroethene         20         NA         50         100         20         µg/L           79-01-6         Trichloroethene         0.28         0.49         0.28         5         0.5         µg/L           75-69-4         Trichlorofluoromethane         520         NA         520         3490         698         µg/L	156-60-5	trans-1 2-Dichloroetheno	20	NA	36	100	20	μσ/ι μσ/ι
75-69-4 Trichlorofluoromethane 520 NA 520 3490 698 ug/L	79-01-6	Trichloroethene	0.28	0.49	0.28	5	0.5	μσ/ι μσ/ι
	75-69-4	Trichlorofluoromethane	520	NA	520	3490	698	μg/L

Footnote 1. The U.S. Environmental Protection Agency (EPA) noncancer-based tapwater regional screening levels (RSLs) presented in this table are based on a target hazard quotient (THQ) of 0.1. A THQ of 0.1 is used at the screening step in the risk assessment as a conservative means to select chemicals of potential concern (COPCs).

Table 2a. Summary of Screening Assessment - Propellant Burning Ground Plume - Onsite Monitoring Wells Screening Level Groundwater Risk Evaluation (Draft) Badger Army Ammunition Plant

		Screening						Result
CAS	Analyte	Level	Units	Well Type	Well ID	Well Name	Date Sampled	(maximum)
71-55-6	1,1,1-Trichloroethane	40	μg/L	Monitoring	674	PBN-9303C	9/24/18	1.9
75-34-3	1,1-Dichloroethane	2.8	μg/L	Monitoring	793	PBN-1404D	9/28/15	0.53
75-35-4	1,1-Dichloroethene	0.7	μg/L	Monitoring	674	PBN-9303C	9/24/18	0.37
95-63-6	1,2,4-Trimethylbenzene	5.6	μg/L	Monitoring	655	PBN-8912B	9/27/16	0.11
107-06-2	1,2-Dichloroethane	0.17	μg/L	Monitoring	686	PBN-9304C	3/3/16	0.064
108-67-8	1,3,5-Trimethylbenzene	6	μg/L	Monitoring	655	PBN-8912B	9/27/16	0.037
71-43-2	Benzene	0.46	μg/L	Monitoring	655	PBN-8912B	4/12/18	0.3
75-27-4	Bromodichloromethane	0.06	μg/L	Monitoring	669	PBN-9301C	4/23/18	0.16
75-15-0	Carbon disulfide	81	μg/L	Monitoring	794	PBN-1405F	9/26/16	0.39
56-23-5	Carbon tetrachloride	0.46	μg/L	Monitoring	632	PBN-8502A	4/18/18	16
75-00-3	Chloroethane	80	μg/L	Monitoring	775	PBN-1303B	9/12/17	1.4
67-66-3	Chloroform	0.22	μg/L	Monitoring	632	PBN-8502A	9/28/15	4.3
74-87-3	Chloromethane	3	μg/L	Monitoring	633	PBN-8503A	9/25/18	0.14
75-71-8	Dichlorodifluoromethane	20	μg/L	Monitoring	726	SPN-9104D	9/24/15	0.037
60-29-7	Ethyl ether	100	μg/L	Monitoring	687	PBN-9304D	4/14/15	6900
100-41-4	Ethylbenzene	1.5	μg/L	Monitoring	655	PBN-8912B	9/27/16	0.069
179601-23-1	m & p-Xylene	19	μg/L	Monitoring	655	PBN-8912B	9/27/16	0.17
91-20-3	Naphthalene	0.17	μg/L	Monitoring	655	PBN-8912B	9/27/16	0.053
14797-55-8	Nitrate	2	mg/L	Monitoring	368	PBM-0002	9/20/17	4.6
95-47-6	o-Xylene	19	μg/L	Monitoring	655	PBN-8912B	9/27/16	0.075
98-06-6	tert-Butylbenzene	69	μg/L	Monitoring	655	PBN-8912B	9/27/16	0.047
127-18-4	Tetrachloroethene	0.5	μg/L	Monitoring	655	PBN-8912B	4/12/18	0.12
108-88-3	Toluene	110	μg/L	Monitoring	655	PBN-8912B	9/27/16	3.5
25321-14-6	Total Dinitrotoluenes	0.005	μg/L	Monitoring	613	PBN-8202A	5/14/18	420.294
602-01-7	2,3-Dinitrotoluene	NA	μg/L	Monitoring	613	PBN-8202A	5/14/18	78
121-14-2	2,4-Dinitrotoluene	0.005	μg/L	Monitoring	613	PBN-8202A	5/14/18	33
619-15-8	2,5-Dinitrotoluene	NA	μg/L	Monitoring	613	PBN-8202A	5/14/18	0.094
606-20-2	2,6-Dinitrotoluene	0.005	μg/L	Monitoring	613	PBN-8202A	5/14/18	270
610-39-9	3,4-Dinitrotoluene	NA	μg/L	Monitoring	613	PBN-8202A	5/14/18	35
618-85-9	3,5-Dinitrotoluene	NA	μg/L	Monitoring	613	PBN-8202A	5/14/18	4.2
79-01-6	Trichloroethene	0.28	μg/L	Monitoring	686	PBN-9304C	4/5/16	7.3

Notes:

1. Those analytes detected at least once in a well in 2015, 2016, 2017 or 2018 within this specific plume area are presented in this table.

2. Those analytes that have a maximum concentration greater than the screening level are highlighted in yellow and represent chemicals of potential concern (COPCs) for which further evaluation of risk will be conducted.

3. For the screening assessment, all dinitrotoluene isomers (e.g., 2,4-dinitrotoluene, 3,4-dinitrotoluene, etc.) were summed together to calculate a total dinitrotoluenes for each sample. The total dinitrotoluenes value was then compared to the lowest screening value available for the dinitrotoluene isomers. This conservative approach was used because many of the dinitrotoluene isomers did not have screening values. The individual isomers that make up the total dinitrotoluenes concentration for the water sample are provided below the total value in gray highlighting for informational purposes.

NA - A screening value is not available for the analyte.

#### Table 2b. Summary of Hypothetical Future Risks - Propellant Burning Ground Plume - Onsite Monitoring Wells Screening Level Groundwater Risk Evaluation (Draft) Badger Army Ammunition Plant

										EPA Noncancer-		
									EPA Cancer-	based Tapwater		Noncancer
		Screening						Result	based	RSL (Based on		Hazard Quotient
CAS	Analyte	Level	Units	Well Type	Well ID	Well Name	Date Sampled	(maximum)	Tapwater RSL	THQ=0.1)	Cancer Risk <sup>1</sup>	(HQ) <sup>1</sup>
75-27-4	Bromodichloromethane	0.06	μg/L	Monitoring	669	PBN-9301C	4/23/18	0.16	0.13	38	1E-06	0.0004
56-23-5	Carbon tetrachloride	0.46	μg/L	Monitoring	632	PBN-8502A	4/18/18	16	0.46	4.9	3E-05	0.3
67-66-3	Chloroform	0.22	μg/L	Monitoring	632	PBN-8502A	9/28/15	4.3	0.22	9.7	2E-05	0.04
60-29-7	Ethyl ether	100	μg/L	Monitoring	687	PBN-9304D	4/14/15	6900	NA	390	NA	2
14797-55-8	Nitrate	2	mg/L	Monitoring	368	PBM-0002	9/20/17	4.6	NA	3.2	NA	0.1
25321-14-6	Total Dinitrotoluenes	0.005	μg/L	Monitoring	613	PBN-8202A	5/14/18	420.294	0.1	1.1	4E-03	38
602-01-7	2,3-Dinitrotoluene	NA	μg/L	Monitoring	613	PBN-8202A	5/14/18	78	NA	NA	NA	NA
121-14-2	2,4-Dinitrotoluene	0.005	μg/L	Monitoring	613	PBN-8202A	5/14/18	33	0.24	3.8	1E-04	0.9
619-15-8	2,5-Dinitrotoluene	NA	μg/L	Monitoring	613	PBN-8202A	5/14/18	0.094	NA	NA	NA	NA
606-20-2	2,6-Dinitrotoluene	0.005	μg/L	Monitoring	613	PBN-8202A	5/14/18	270	0.049	0.57	6E-03	47
610-39-9	3,4-Dinitrotoluene	NA	μg/L	Monitoring	613	PBN-8202A	5/14/18	35	NA	NA	NA	NA
618-85-9	3,5-Dinitrotoluene	NA	μg/L	Monitoring	613	PBN-8202A	5/14/18	4.2	NA	NA	NA	NA
79-01-6	Trichloroethene	0.28	μg/L	Monitoring	686	PBN-9304C	4/5/16	7.3	0.49	0.28	1E-05	3
											6E-03	53
											Cumulative	Hazard Index
											Cancer Risk	(HI)

Notes:

1. For each chemical of potential concern (COPC) identified for the plume area, a cancer risk and noncancer hazard quotient (HQ) were calculated if appropriate U.S. Environmental Protection Agency (EPA) tapwater regional screening levels (RSLs) were available for an analyte.

2. The noncancer HQ for each chemical was calculated using the EPA noncancer-based tapwater RSLs based on a target hazard quotient (THQ) of 0.1.

3. The cumulative cancer risk is calculated by summing the individual cancer risks for each COPC. The total noncancer risk is calculated by summing the analyte-specific HQs to develop a hazard index (HI).

4. The total dinitrotoluenes concentration represents the sum of all isomers of dinitrotoluene detected in the water sample. The individual isomers that make up the total dinitrotoluenes concentration for the water samples are provided below the total value in gray highlighting. The risks associated with dinitrotoluene are based on the total value and the individual isomers. The highest of the two risk estimates (i.e., based on total or the sum of individual isomers) are used in calculating the total risk for the plume area.

NA - A screening value and/or tapwater RSL was not available for the analyte. Where a tapwater RSL was not available, risk was not estimated.

Footnote:

1. All risk values are rounded to one significant figure by convention. In some cases the cumulative cancer risk or hazard index may be different from the sum of the individual cancer risks or HQs as presented because they are summed from the unrounded values.

Table 3a. Summary of Screening Assessment - Propellant Burning Ground Plume - Offsite Wells Screening Level Groundwater Risk Evaluation (Draft) Badger Army Ammunition Plant

		Screening						Result
CAS	Analyte	Level	Units	Well Type	Well ID	Well Name	Date Sampled	(maximum)
71-55-6	1,1,1-Trichloroethane	40	μg/L	Monitoring	561	PBN-9101C	10/3/16	0.45
75-34-3	1,1-Dichloroethane	2.8	μg/L	Monitoring	695	PBN-9903D	9/26/16	0.3
75-35-4	1,1-Dichloroethene	0.7	μg/L	Monitoring	561	PBN-9101C	10/3/16	0.084
107-06-2	1,2-Dichloroethane	0.17	μg/L	Monitoring	574	SWN-9103E	10/4/16	0.032
71-43-2	Benzene	0.46	μg/L	Monitoring	574	SWN-9103E	9/14/17	0.22
75-27-4	Bromodichloromethane	0.06	μg/L	Monitoring	561	PBN-9101C	4/9/15	0.079
75-15-0	Carbon disulfide	81	μg/L	Monitoring	561	PBN-9101C	10/3/16	0.072
56-23-5	Carbon tetrachloride	0.46	μg/L	Monitoring	561	PBN-9101C	4/9/15	29
75-00-3	Chloroethane	80	μg/L	Residential	875	Krumenauer	10/2/15	0.075
67-66-3	Chloroform	0.22	μg/L	Monitoring	561	PBN-9101C	9/17/15	3
60-29-7	Ethyl ether	100	μg/L	Monitoring	695	PBN-9903D	4/14/15	3100
108-88-3	Toluene	110	μg/L	Monitoring	574	SWN-9103E	9/14/17	0.79
25321-14-6	Total Dinitrotoluenes	0.005	μg/L	Monitoring	561	PBN-9101C	4/12/18	0.082
606-20-2	2,6-Dinitrotoluene	0.005	μg/L	Monitoring	561	PBN-9101C	4/12/18	0.082
79-01-6	Trichloroethene	0.28	μg/L	Monitoring	561	PBN-9101C	4/9/15	9.6

Notes:

1. Those analytes detected at least once in a well in 2015, 2016, 2017 or 2018 within this specific plume area are presented in this table.

2. Those analytes that have a maximum concentration greater than the screening level are highlighted in yellow and represent chemicals of potential concern (COPCs) for which further evaluation of risk will be conducted.

3. For the screening assessment, all dinitrotoluene isomers (e.g., 2,4-dinitrotoluene, 3,4-dinitrotoluene, etc.) were summed together to calculate a total dinitrotoluenes for each sample. The total dinitrotoluenes value was then compared to the lowest screening value available for the dinitrotoluene isomers. This conservative approach was used because many of the dinitrotoluene isomers did not have screening values. The individual isomers that make up the total dinitrotoluenes concentration for the water sample are provided below the total value in gray highlighting for informational purposes.

NA - A screening value is not available for the analyte.

## Table 3b. Summary of Current Risks - Propellant Burning Ground Plume - Offsite Wells Screening Level Groundwater Risk Evaluation (Draft) Badger Army Ammunition Plant

										EPA Noncancer-		
									EPA Cancer-	based Tapwater		Noncancer
		Screening					Date	Result	based	RSL (Based on		Hazard
CAS	Analyte	Level	Units	Well Type	Well ID	Well Name	Sampled	(maximum)	Tapwater RSL	THQ=0.1)	Cancer Risk <sup>1</sup>	Quotient $(HQ)^1$
75-27-4	Bromodichloromethane	0.06	μg/L	Monitoring	561	PBN-9101C	4/9/15	0.079	0.13	38	6E-07	0.0002
56-23-5	Carbon tetrachloride	0.46	μg/L	Monitoring	561	PBN-9101C	4/9/15	29	0.46	4.9	6E-05	0.6
67-66-3	Chloroform	0.22	μg/L	Monitoring	561	PBN-9101C	9/17/15	3	0.22	9.7	1E-05	0.03
60-29-7	Ethyl ether	100	μg/L	Monitoring	695	PBN-9903D	4/14/15	3100	NA	390	NA	0.8
25321-14-6	Total Dinitrotoluenes	0.005	μg/L	Monitoring	561	PBN-9101C	4/12/18	0.082	0.1	1.1	8E-07	0.01
606-20-2	2,6-Dinitrotoluene	0.005	μg/L	Monitoring	561	PBN-9101C	4/12/18	0.082	0.049	0.57	2E-06	0.01
79-01-6	Trichloroethene	0.28	μg/L	Monitoring	561	PBN-9101C	4/9/15	9.6	0.49	0.28	2E-05	3
											1E-04	5
											Cumulative	Hazard Index
											Cancer Risk	(HI)

#### Notes:

1. For each chemical of potential concern (COPC) identified for the plume area, a cancer risk and noncancer hazard quotient (HQ) were calculated if appropriate U.S. Environmental Protection Agency (EPA) tapwater regional screening levels (RSLs) were available for an analyte.

2. The noncancer HQ for each chemical was calculated using the EPA noncancer-based tapwater RSLs based on a target hazard quotient (THQ) of 0.1.

3. The cumulative cancer risk is calculated by summing the individual cancer risks for each COPC. The total noncancer risk is calculated by summing the analyte-specific HQs to develop a hazard index (HI).

4. The total dinitrotoluenes concentration represents the sum of all isomers of dinitrotoluene detected in the water sample. The individual isomers that make up the total dinitrotoluenes concentration for the water samples are provided below the total value in gray highlighting. The risks associated with dinitrotoluene are based on the total value and the individual isomers. The highest of the two risk estimates (i.e., based on total or the sum of individual isomers) are used in calculating the total risk for the plume area.

NA - A screening value and/or tapwater RSL was not available for the analyte. Where a tapwater RSL was not available, risk was not estimated.

Footnote:

1. All risk values are rounded to one significant figure by convention. In some cases the cumulative cancer risk or hazard index may be different from the sum of the individual cancer risks or HQs as presented because they are summed from the unrounded values.

Table 4a. Summary of Screening Assessment - Deterrent Burning Ground Plume - Onsite Monitoring Wells Screening Level Groundwater Risk Evaluation (Draft) Badger Army Ammunition Plant

Screening Level Units Well Type Well ID Well Name CAS Analyte **Date Sampled** Result (maximum) 71-55-6 1,1,1-Trichloroethane 40 μg/L Monitoring 472 DBN-1001B 4/21/16 1.7 79-00-5 1,1,2-Trichloroethane 0.041 μg/L Monitoring 211 ELN-8203B 4/26/18 0.98 75-34-3 1,1-Dichloroethane 2.8 μg/L Monitoring 210 ELN-8203A 4/18/16 0.044 75-35-4 0.054 1,1-Dichloroethene 0.7 μg/L Monitoring 216 ELM-8901 4/21/16 95-50-1 1,2-Dichlorobenzene 30 μg/L Monitoring 236 S1134R 4/20/15 0.15 107-06-2 1,2-Dichloroethane 0.17 μg/L Monitoring 234 ELM-9501 4/28/15 0.17 78-87-5 1,2-Dichloropropane 0.5 μg/L Monitoring 211 ELN-8203B 4/24/17 0.41 67-66-3 Chloroform 0.22 μg/L Monitoring 301 DBM-8201 4/21/16 0.075 74-87-3 220 Chloromethane 3 μg/L Monitoring ELM-8907 4/18/16 0.034 156-59-2 cis-1,2-Dichloroethene μg/L 210 ELN-8203A 4/20/15 0.057 3.6 Monitoring 75-71-8 Dichlorodifluoromethane 20 μg/L Monitoring 211 ELN-8203B 4/26/18 0.38 Dichlorofluoromethane NA ELN-8203A 4/18/16 0.029 75-43-4 μg/L Monitoring 210 Ethyl ether 100 Monitoring 60-29-7 μg/L 210 ELN-8203A 4/24/17 0.77 10 100-42-5 Styrene μg/L 316 DBN-9501C 4/27/15 0.03 Monitoring mg/L 14808-79-8 Sulfate 125 Monitoring 210 ELN-8203A 4/26/18 1100 127-18-4 Tetrachloroethene 0.5 μg/L Monitoring 225 ELN-8904A 4/20/15 0.12 109-99-9 Tetrahydrofuran 10 μg/L Monitoring 211 ELN-8203B 4/18/16 20 Total Dinitrotoluenes 0.005 8.58 25321-14-6 μg/L Monitoring 302 DBM-8202 4/24/17 5 602-01-7 2,3-Dinitrotoluene NA μg/L Monitoring 302 DBM-8202 4/24/17 606-20-2 2.6-Dinitrotoluene 0.005 μg/L Monitoring 302 DBM-8202 4/24/17 0.22 μg/L 610-39-9 3,4-Dinitrotoluene NA Monitoring 302 DBM-8202 4/24/17 0.26 618-85-9 3,5-Dinitrotoluene NA μg/L Monitoring 302 DBM-8202 4/24/17 3.1 75-69-4 Trichlorofluoromethane 520 DBM-8202 4/21/15 0.043 μg/L Monitoring 302

Notes:

1. Those analytes detected at least once in a well in 2015, 2016, 2017 or 2018 within this specific plume area are presented in this table.

2. Those analytes that have a maximum concentration greater than the screening level are highlighted in yellow and represent chemicals of potential concern (COPCs) for which further evaluation of risk will be conducted.

3. For the screening assessment, all dinitrotoluene isomers (e.g., 2,4-dinitrotoluene, 3,4-dinitrotoluene, etc.) were summed together to calculate a total dinitrotoluenes for each sample. The total dinitrotoluenes value was then compared to the lowest screening value available for the dinitrotoluene isomers. This conservative approach was used because many of the dinitrotoluene isomers did not have screening values. The individual isomers that make up the total dinitrotoluenes concentration for the water sample are provided below the total value in gray highlighting for informational purposes.

NA - A screening value is not available for the analyte.

## Table 4b. Summary of Hypothetical Future Risks - Deterrent Burning Ground Plume - Onsite Monitoring Wells Screening Level Groundwater Risk Evaluation (Draft) Badger Army Ammunition Plant

CAS	Analyte	Screening Level	Units	Well Type	Well ID	Well Name	Date Sampled	Result (maximum)	EPA Cancer- based Tapwater RSL	EPA Noncancer- based Tapwater RSL (Based on THQ=0.1)	Cancer Risk <sup>1</sup>	Noncancer Hazard Quotient (HQ) <sup>1</sup>
79-00-5	1,1,2-Trichloroethane	0.041	μg/L	Monitoring	211	ELN-8203B	4/26/18	0.98	0.28	0.041	4E-06	2
107-06-2	1,2-Dichloroethane	0.17	μg/L	Monitoring	234	ELM-9501	4/28/15	0.17	0.17	1.3	1E-06	0.01
14808-79-8	Sulfate	125	mg/L	Monitoring	210	ELN-8203A	4/26/18	1100	NA	NA	NA	NA
109-99-9	Tetrahydrofuran	10	μg/L	Monitoring	211	ELN-8203B	4/18/16	20	NA	340	NA	0.01
25321-14-6	Total Dinitrotoluenes	0.005	μg/L	Monitoring	302	DBM-8202	4/24/17	8.58	0.1	1.1	9E-05	0.8
602-01-7	2,3-Dinitrotoluene	NA	μg/L	Monitoring	302	DBM-8202	4/24/17	5	NA	NA	NA	NA
606-20-2	2,6-Dinitrotoluene	0.005	μg/L	Monitoring	302	DBM-8202	4/24/17	0.22	0.049	0.57	4E-06	0.04
610-39-9	3,4-Dinitrotoluene	NA	μg/L	Monitoring	302	DBM-8202	4/24/17	0.26	NA	NA	NA	NA
618-85-9	3,5-Dinitrotoluene	NA	μg/L	Monitoring	302	DBM-8202	4/24/17	3.1	NA	NA	NA	NA
											9E-05	3
											Cumulative	Hazard Index
											Cancer Risk	(HI)

#### Notes:

1. For each chemical of potential concern (COPC) identified for the plume area, a cancer risk and noncancer hazard quotient (HQ) were calculated if appropriate U.S. Environmental Protection Agency (EPA) tapwater regional screening levels (RSLs) were available for an analyte.

2. The noncancer HQ for each chemical was calculated using the EPA noncancer-based tapwater RSLs based on a target hazard quotient (THQ) of 0.1.

3. The cumulative cancer risk is calculated by summing the individual cancer risks for each COPC. The total noncancer risk is calculated by summing the analyte-specific HQs to develop a hazard index (HI).

4. The total dinitrotoluenes concentration represents the sum of all isomers of dinitrotoluene detected in the water sample. The individual isomers that make up the total dinitrotoluenes concentration for the water sample are provided below the total value in gray highlighting. The risks associated with dinitrotoluene are based on the total value and the individual isomers. The highest of the two risk estimates (i.e., based on total or the sum of individual isomers) are used in calculating the total risk for the plume area.

NA - A screening value and/or tapwater RSL was not available for the analyte. Where a tapwater RSL was not available, risk was not estimated.

Footnote:

1. All risk values are rounded to one significant figure by convention. In some cases the cumulative cancer risk or hazard index may be different from the sum of the individual cancer risks or HQs as presented because they are summed from the unrounded values.

Table 5a. Summary of Screening Assessment - Deterrent Burning Ground Plume - Offsite Wells Screening Level Groundwater Risk Evaluation (Draft) Badger Army Ammunition Plant

		Screening					Date	
CAS	Analyte	Level	Units	Well Type	Well ID	Well Name	Sampled	Result (maximum)
71-55-6	1,1,1-Trichloroethane	40	μg/L	Residential	163	Purcell-D	4/23/18	0.11
79-00-5	1,1,2-Trichloroethane	0.041	μg/L	Residential	803	Spear	10/5/15	0.25
78-93-3	2-Butanone	560	μg/L	Residential	428	Schumann	10/1/15	1.7
71-43-2	Benzene	0.46	μg/L	Residential	411	Anderson-R	10/1/15	0.012
67-66-3	Chloroform	0.22	μg/L	Residential	426	Cornelius	10/1/15	0.37
74-87-3	Chloromethane	3	μg/L	Residential	426	Cornelius	8/21/18	0.11
75-71-8	Dichlorodifluoromethane	20	μg/L	Residential	412	Curto	8/21/18	0.17
91-20-3	Naphthalene	0.17	μg/L	Residential	428	Schumann	10/1/15	0.072
100-42-5	Styrene	10	μg/L	Residential	428	Schumann	10/1/15	0.054
108-88-3	Toluene	110	μg/L	Residential	428	Schumann	10/1/15	1.8
25321-14-6	Total Dinitrotoluenes	0.005	μg/L	Monitoring	468	ELN-1003B	11/15/18	0.32
602-01-7	2,3-Dinitrotoluene	NA	μg/L	Monitoring	468	ELN-1003B	11/15/18	0.078
606-20-2	2,6-Dinitrotoluene	0.005	μg/L	Monitoring	468	ELN-1003B	11/15/18	0.072
610-39-9	3,4-Dinitrotoluene	NA	μg/L	Monitoring	468	ELN-1003B	11/15/18	0.17
156-60-5	trans-1,2-Dichloroethene	20	μg/L	Residential	428	Schumann	10/1/15	0.37
79-01-6	Trichloroethene	0.28	μg/L	Residential	418	Hendershot	8/29/16	4.7
75-69-4	Trichlorofluoromethane	520	μg/L	Residential	803	Spear	10/5/15	0.043

Notes:

1. Those analytes detected at least once in a well in 2015, 2016, 2017 or 2018 within this specific plume area are presented in this table.

2. Those analytes that have a maximum concentration greater than the screening level are highlighted in yellow and represent chemicals of potential concern (COPCs) for which further evaluation of risk will be conducted.

3. For the screening assessment, all dinitrotoluene isomers (e.g., 2,4-dinitrotoluene, 3,4-dinitrotoluene, etc.) were summed together to calculate a total dinitrotoluenes for each sample. The total dinitrotoluenes value was then compared to the lowest screening value available for the dinitrotoluene isomers. This conservative approach was used because many of the dinitrotoluene isomers did not have screening values. The individual isomers that make up the total dinitrotoluenes concentration for the water sample are provided below the total value in gray highlighting for informational purposes.

NA - A screening value is not available for the analyte.

#### Table 5b. Summary of Current Risks - Deterrent Burning Ground Plume - Offsite Wells Screening Level Groundwater Risk Evaluation (Draft) Badger Army Ammunition Plant

CAS	Analyte	Screening Level	Units	Well Type	Well ID	Well Name	Date Sampled	Result (maximum)	EPA Cancer- based Tapwater RSL	EPA Noncancer- based Tapwater RSL (Based on THQ=0.1)	Cancer Risk <sup>1</sup>	Noncancer Hazard Quotient (HQ) <sup>1</sup>
79-00-5	1,1,2-Trichloroethane	0.041	μg/L	Residential	803	Spear	10/5/15	0.25	0.28	0.041	9E-07	0.6
67-66-3	Chloroform	0.22	μg/L	Residential	426	Roll	10/1/15	0.37	0.22	9.7	2E-06	0.004
25321-14-6	Total Dinitrotoluenes	0.005	μg/L	Monitoring	468	ELN-1003B	11/15/18	0.32	0.1	1.1	3E-06	0.03
602-01-7	2,3-Dinitrotoluene	NA	μg/L	Monitoring	468	ELN-1003B	11/15/18	0.078	NA	NA	NA	NA
606-20-2	2,6-Dinitrotoluene	0.005	μg/L	Monitoring	468	ELN-1003B	11/15/18	0.072	0.049	0.57	1E-06	0.013
610-39-9	3,4-Dinitrotoluene	NA	μg/L	Monitoring	468	ELN-1003B	11/15/18	0.17	NA	NA	NA	NA
79-01-6	Trichloroethene	0.28	μg/L	Residential	418	Hendershot	8/29/16	4.7	0.49	0.28	1E-05	2
											2E-05	2
											Cumulative	Hazard Index
											Cancer Risk	(HI)

#### Notes:

1. For each chemical of potential concern (COPC) identified for the plume area, a cancer risk and noncancer hazard quotient (HQ) were calculated if appropriate U.S. Environmental Protection Agency (EPA) tapwater regional screening levels (RSLs) were available for an analyte.

2. The noncancer HQ for each chemical was calculated using the EPA noncancer-based tapwater RSLs based on a target hazard quotient (THQ) of 0.1.

3. The cumulative cancer risk is calculated by summing the individual cancer risks for each COPC. The total noncancer risk is calculated by summing the analyte-specific HQs to develop a hazard index (HI).

4. The total dinitrotoluenes concentration represents the sum of all isomers of dinitrotoluene detected in the water sample. The individual isomers that make up the total dinitrotoluenes concentration for the water sample are provided below the total value in gray highlighting. The risks associated with dinitrotoluene are based on the total value and the individual isomers. The highest of the two risk estimates (i.e., based on total or the sum of individual isomers) are used in calculating the total risk for the plume area.

NA - A screening value and/or tapwater RSL was not available for the analyte. Where a tapwater RSL was not available, risk was not estimated.

Footnote:

1. All risk values are rounded to one significant figure by convention. In some cases the cumulative cancer risk or hazard index may be different from the sum of the individual cancer risks or HQs as presented because they are summed from the unrounded values.

# Table 5c. Time Trends of Trichloroethene in Hendershot Residential Well: 2007 to 2018Screening Level Groundwater Risk Evaluation (Draft)Badger Army Ammunition Plant

CAS	Analyte	Well ID	Well Name	Date Sampled	Result	Units
79-01-6	Trichloroethene	418	Hendershot	8/6/07	0.40	μg/L
79-01-6	Trichloroethene	418	Hendershot	8/24/10	0.60	μg/L
79-01-6	Trichloroethene	418	Hendershot	8/24/11	0.49	μg/L
79-01-6	Trichloroethene	418	Hendershot	8/22/12	0.61	μg/L
79-01-6	Trichloroethene	418	Hendershot	8/20/13	1.28	μg/L
79-01-6	Trichloroethene	418	Hendershot	8/5/14	0.76	μg/L
79-01-6	Trichloroethene	418	Hendershot	10/2/15	0.42	μg/L
79-01-6	Trichloroethene	418	Hendershot	8/29/16	4.70	μg/L
79-01-6	Trichloroethene	418	Hendershot	8/28/17	0.82	μg/L
79-01-6	Trichloroethene	418	Hendershot	8/21/18	2	μg/L
				Arithmetic Mean	1.2	μg/L
				Geometric Mean	0.86	μg/L

Table 6a. Summary of Screening Assessment - Central Plume - Onsite Monitoring Wells Screening Level Groundwater Risk Evaluation (Draft) Badger Army Ammunition Plant

		Screening						
CAS	Analyte	Level	Units	Well Type	Well ID	Well Name	Date Sampled	Result (maximum)
56-23-5	Carbon tetrachloride	0.46	μg/L	Monitoring	542	RIN-1502C	12/7/15	0.028
67-66-3	Chloroform	0.22	μg/L	Monitoring	540	RIN-1501D	12/7/15	0.27
25321-14-6	Total Dinitrotoluenes	0.005	μg/L	Monitoring	331	NLN-1001A	6/26/18	0.209
602-01-7	2,3-Dinitrotoluene	NA	μg/L	Monitoring	331	NLN-1001A	6/26/18	0.061
606-20-2	2,6-Dinitrotoluene	0.005	μg/L	Monitoring	331	NLN-1001A	6/26/18	0.058
610-39-9	3,4-Dinitrotoluene	NA	μg/L	Monitoring	331	NLN-1001A	6/26/18	0.09

Notes:

Those analytes detected at least once in a well in 2015, 2016, 2017 or 2018 ithin this specific plume area are presented in this table.
 Those analytes that have a maximum concentration greater than the screening level are highlighted in yellow and represent chemicals of potential concern (COPCs) for which further evaluation of risk will be conducted.

3. For the screening assessment, all dinitrotoluene isomers (e.g., 2,4-dinitrotoluene, 3,4-dinitrotoluene, etc.) were summed together to calculate a total dinitrotoluenes for each sample. The total dinitrotoluenes value was then compared to the lowest screening value available for the dinitrotoluene isomers. This conservative approach was used because many of the dinitrotoluene isomers did not have screening values. The individual isomers that make up the total dinitrotoluenes concentration for the water sample are provided below the total value in gray highlighting for informational purposes.

NA - A screening value is not available for the analyte.

Table 6b. Summary of Hypothetical Future Risks - Central Plume - Onsite Monitoring Wells Screening Level Groundwater Risk Evaluation (Draft) Badger Army Ammunition Plant

CAS	Analyte	Screening Level	Units	Well Type	Well ID	Well Name	Date Sampled	Result (maximum)	EPA Cancer- based Tapwater RSL	EPA Noncancer- based Tapwater RSL (Based on THQ=0.1)	Cancer Risk <sup>1</sup>	Noncancer Hazard Quotient (HQ) <sup>1</sup>
67-66-3	Chloroform	0.22	μg/L	Monitoring	540	RIN-1501D	12/7/15	0.27	0.22	9.7	1E-06	0.003
25321-14-6	Total Dinitrotoluenes	0.005	μg/L	Monitoring	331	NLN-1001A	6/26/18	0.209	0.1	1.1	2E-06	0.02
602-01-7	2,3-Dinitrotoluene	NA	μg/L	Monitoring	331	NLN-1001A	6/26/18	0.061	NA	NA	NA	NA
606-20-2	2,6-Dinitrotoluene	0.005	μg/L	Monitoring	331	NLN-1001A	6/26/18	0.058	0.049	0.57	1E-06	0.01
610-39-9	3,4-Dinitrotoluene	NA	μg/L	Monitoring	331	NLN-1001A	6/26/18	0.09	NA	NA	NA	NA
											3E-06	0.02
											Cumulative	Hazard Index
											Cancer Risk	(HI)

#### Notes:

1. For each chemical of potential concern (COPC) identified for the plume area, a cancer risk and noncancer hazard quotient (HQ) were calculated if appropriate U.S. Environmental Protection Agency (EPA) tapwater regional screening levels (RSLs) were available for an analyte.

2. The noncancer HQ for each chemical was calculated using the EPA noncancer-based tapwater RSLs based on a target hazard quotient (THQ) of 0.1.

3. The cumulative cancer risk is calculated by summing the individual cancer risks for each COPC. The total noncancer risk is calculated by summing the analyte-specific HQs to develop a hazard index (HI).

4. The total dinitrotoluenes concentration represents the sum of all isomers of dinitrotoluene detected in the water sample. The individual isomers that make up the total dinitrotoluenes concentration for the water sample are provided below the total value in gray highlighting. The risks associated with dinitrotoluene are based on the total value and the individual isomers. The highest of the two risk estimates (i.e., based on total or the sum of individual isomers) are used in calculating the total risk for the plume area.

NA - A screening value and/or tapwater RSL was not available for the analyte. Where a tapwater RSL was not available, risk was not estimated.

Footnote:

1. All risk values are rounded to one significant figure by convention. In some cases the cumulative cancer risk or hazard index may be different from the sum of the individual cancer risks or HQs as presented because they are summed from the unrounded values.

Table 7a. Summary of Screening Assessment - Central Plume - Offsite Wells Screening Level Groundwater Risk Evaluation (Draft) Badger Army Ammunition Plant

		Screening						
CAS	Analyte	Level	Units	Well Type	Well ID	Well Name	Date Sampled	Result (maximum)
95-63-6	1,2,4-Trimethylbenzene	5.6	μg/L	Monitoring	586	SEN-0503B	6/12/17	0.68
107-06-2	1,2-Dichloroethane	0.17	μg/L	Monitoring	586	SEN-0503B	6/12/17	0.3
108-67-8	1,3,5-Trimethylbenzene	6	μg/L	Monitoring	586	SEN-0503B	6/12/17	0.34
71-43-2	Benzene	0.46	μg/L	Monitoring	586	SEN-0503B	6/12/17	10
75-27-4	Bromodichloromethane	0.06	μg/L	Monitoring	581	SEN-0501B	4/22/15	0.031
75-15-0	Carbon disulfide	81	μg/L	Monitoring	580	SEN-0501A	11/7/16	0.11
56-23-5	Carbon tetrachloride	0.46	μg/L	Monitoring	581	SEN-0501B	6/12/17	0.22
67-66-3	Chloroform	0.22	μg/L	Residential	164	WE-SQ017	10/5/15	2
100-41-4	Ethylbenzene	1.5	μg/L	Monitoring	586	SEN-0503B	6/12/17	1.9
98-82-8	Isopropylbenzene	45	μg/L	Monitoring	586	SEN-0503B	11/7/16	0.03
179601-23-1	m & p-Xylene	19	μg/L	Monitoring	586	SEN-0503B	6/12/17	6.7
103-65-1	n-Propylbenzene	66	μg/L	Monitoring	586	SEN-0503B	6/12/17	0.21
95-47-6	o-Xylene	19	μg/L	Monitoring	586	SEN-0503B	6/12/17	2.6
100-42-5	Styrene	10	μg/L	Monitoring	586	SEN-0503B	6/12/17	0.21
108-88-3	Toluene	110	μg/L	Monitoring	586	SEN-0503B	6/12/17	35
25321-14-6	Total Dinitrotoluenes	0.005	μg/L	Monitoring	586	SEN-0503B	11/15/18	0.08
606-20-2	2,6-Dinitrotoluene	0.005	μg/L	Monitoring	586	SEN-0503B	11/15/18	0.08

Notes:

1. Those analytes detected at least once in a well in 2015, 2016, 2017 or 2018 within this specific plume area are presented in this table.

2. Those analytes that have a maximum concentration greater than the screening level are highlighted in yellow and represent chemicals of potential concern (COPCs) for which further evaluation of risk will be conducted.

3. For the screening assessment, all dinitrotoluene isomers (e.g., 2,4-dinitrotoluene, 3,4-dinitrotoluene, etc.) were summed together to calculate a total dinitrotoluenes for each sample. The total dinitrotoluenes value was then compared to the lowest screening value available for the dinitrotoluene isomers. This conservative approach was used because many of the dinitrotoluene isomers did not have screening values. The individual isomers that make up the total dinitrotoluenes concentration for the water sample are provided below the total value in gray highlighting for informational purposes.

NA - A screening value is not available for the analyte.

#### Table 7b. Summary of Current Risks - Central Plume - Offsite Wells Screening Level Groundwater Risk Evaluation (Draft) Badger Army Ammunition Plant

									EPA Cancer- based	EPA Noncancer- based Tapwater		Noncancer
		Screening					Date	Result	Tapwater	RSL (Based on		Hazard Quotient
CAS	Analyte	Level	Units	Well Type	Well ID	Well Name	Sampled	(maximum)	RSL	THQ=0.1)	Cancer Risk <sup>1</sup>	(HQ) <sup>1</sup>
107-06-2	1,2-Dichloroethane	0.17	μg/L	Monitoring	586	SEN-0503B	6/12/17	0.3	0.17	1.3	2E-06	0.02
71-43-2	Benzene	0.46	μg/L	Monitoring	586	SEN-0503B	6/12/17	10	0.46	3.3	2E-05	0.3
67-66-3	Chloroform	0.22	μg/L	Residential	164	WE-SQ017	10/5/15	2	0.22	9.7	9E-06	0.02
100-41-4	Ethylbenzene	1.5	μg/L	Monitoring	586	SEN-0503B	6/12/17	1.9	1.5	81	1E-06	0.002
25321-14-6	Total Dinitrotoluenes	0.005	μg/L	Monitoring	586	SEN-0503B	11/15/18	0.08	0.1	1.1	8E-07	0.007
606-20-2	2,6-Dinitrotoluene	0.005	μg/L	Monitoring	586	SEN-0503B	11/15/18	0.08	0.049	0.57	2E-06	0.01
											4E-05	0.4
											Cumulative	
											Cancer Risk	Hazard Index (HI)

Notes:

1. For each chemical of potential concern (COPC) identified for the plume area, a cancer risk and noncancer hazard quotient (HQ) were calculated if appropriate U.S. Environmental Protection Agency (EPA) tapwater regional screening levels (RSLs) were available for an analyte.

2. The noncancer HQ for each chemical was calculated using the EPA noncancer-based tapwater RSLs based on a target hazard quotient (THQ) of 0.1.

3. The cumulative cancer risk is calculated by summing the individual cancer risks for each COPC. The total noncancer risk is calculated by summing the analyte-specific HQs to develop a hazard index (HI).

4. The total dinitrotoluenes concentration represents the sum of all isomers of dinitrotoluene detected in the water sample. The individual isomers that make up the total dinitrotoluenes concentration for the water sample are provided below the total value in gray highlighting. The risks associated with dinitrotoluene are based on the total value and the individual isomers. The highest of the two risk estimates (i.e., based on total or the sum of individual isomers) are used in calculating the total risk for the plume area.

NA - A screening value and/or tapwater RSL was not available for the analyte. Where a tapwater RSL was not available, risk was not estimated.

Footnote:

1. All risk values are rounded to one significant figure by convention. In some cases the cumulative cancer risk or hazard index may be different from the sum of the individual cancer risks or HQs as presented because they are summed from the unrounded values.

Table 8a.Summary of Screening Assessment - Nitrocellulose Production Area Plume - Onsite Monitoring WellsScreening Level Groundwater Risk Evaluation (Draft)Badger Army Ammunition Plant

		Screening						
CAS	Analyte	Level	Units	Well Type	Well ID	Well Name	Date Sampled	Result (maximum)
25321-14-6	Total Dinitrotoluenes	0.005	μg/L	Monitoring	478	RIM-1002	9/26/18	0.22
606-20-2	2,6-Dinitrotoluene	0.005	μg/L	Monitoring	478	RIM-1002	9/26/18	0.22

Notes:

1. Those analytes detected at least once in a well in 2015, 2016, 2017 or 2018 within this specific plume area are presented in this table.

2. Those analytes that have a maximum concentration greater than the screening level are highlighted in yellow and represent chemicals of potential concern (COPCs) for which further evaluation of risk will be conducted.

3. For the screening assessment, all dinitrotoluene isomers (e.g., 2,4-dinitrotoluene, 3,4-dinitrotoluene, etc.) were summed together to calculate a total dinitrotoluenes for each sample. The total dinitrotoluenes value was then compared to the lowest screening value available for the dinitrotoluene isomers. This conservative approach was used because many of the dinitrotoluene isomers did not have screening values. The individual isomers that make up the total dinitrotoluenes concentration for the water sample are provided below the total value in gray highlighting for informational purposes.

Table 8b.Summary of Hypothetical Future Risks - Nitrocellulose Production Area Plume - Onsite Monitoring WellsScreening Level Groundwater Risk Evaluation (Draft)Badger Army Ammunition Plant

CAS	Analyte	Screening Level	Units	Well Type	Well ID	Well Name	Date Sampled	Result (maximum)	EPA Cancer- based Tapwater RSL	EPA Noncancer- based Tapwater RSL (Based on THQ=0.1)	Cancer Risk <sup>1</sup>	Noncancer Hazard Quotient (HQ) <sup>1</sup>
25321-14-6	Total Dinitrotoluenes	0.005	μg/L	Monitoring	478	RIM-1002	9/26/2018	0.22	0.1	1.1	2E-06	0.02
606-20-2	2,6-Dinitrotoluene	0.005	μg/L	Monitoring	478	RIM-1002	9/26/2018	0.22	0.049	0.57	4E-06	0.04
											4E-06	0.04
											Cumulative	
											Cancer Risk	Hazard Index (HI)

#### Notes:

1. For each chemical of potential concern (COPC) identified for the plume area, a cancer risk and noncancer hazard quotient (HQ) were calculated if appropriate U.S. Environmental Protection Agency (EPA) tapwater regional screening levels (RSLs) were available for an analyte.

2. The noncancer HQ for each chemical was calculated using the EPA noncancer-based tapwater RSLs based on a target hazard quotient (THQ) of 0.1.

3. The cumulative cancer risk is calculated by summing the individual cancer risks for each COPC. The total noncancer risk is calculated by summing the analyte-specific HQs to develop a hazard index (HI).

4. The total dinitrotoluenes concentration represents the sum of all isomers of dinitrotoluene detected in the water sample. The individual isomers that make up the total dinitrotoluenes concentration for the water sample are provided below the total value in gray highlighting. The risks associated with dinitrotoluene are based on the total value and the individual isomers. The highest of the two risk estimates (i.e., based on total or the sum of individual isomers) are used in calculating the total risk for the plume area.

NA - A screening value and/or tapwater RSL was not available for the analyte. Where a tapwater RSL was not available, risk was not estimated.

Footnote:

1. All risk values are rounded to one significant figure by convention. In some cases the cumulative cancer risk or hazard index may be different from the sum of the individual cancer risks or HQs as presented because they are summed from the unrounded values.

Appendix A

Regional Screening Levels (RSLs) – User's Guide (November 2018) An official website of the United States government.

We've made some changes to EPA.gov. If the information you are looking for is not here, you may be able to find it on the EPA Web Archive or the January 19, 2017 Web Snapshot.

Close



## **Regional Screening Levels (RSLs) - User's Guide**

## November 2018

**Regional Screening Levels (RSLs)** 

- <u>Home Page</u>
- User's Guide
- What's New
- Frequent Questions
- Equations
- <u>RSL Calculator</u>
- Generic Tables
- Contact Us

You may need a PDF reader to view some of the files on this page. See EPA's <u>About PDF page</u> to learn more.

For assistance/questions please use the Regional Screening Levels (RSLs) contact us page.

## Disclaimer

This guidance sets forth a recommended, but not mandatory, approach based upon currently available information with respect to risk assessment for response actions at CERCLA sites. This document does not establish binding rules. Alternative approaches for risk assessment may be found to be more appropriate at specific sites (e.g., where site circumstances do not match the underlying assumptions, conditions and models of the guidance). The decision whether to use an alternative approach and a description of any such approach should be documented for such sites.

Accordingly, when comments are received at individual CERCLA sites questioning the use of the approaches recommended in this guidance, the comments should be considered and an explanation provided for the selected approach.

It should also be noted that the screening levels (SLs) in these tables are based upon human health risk and do not address potential ecological risk. Some sites in sensitive ecological settings may also need to be evaluated for potential ecological risk. EPA's guidance "Ecological Risk Assessment Guidance for Superfund: Process for Designing and Conducting Ecological Risk Assessment" contains an eight step process for using benchmarks for ecological effects in the remedy selection process.

On this page:

- **1. Introduction**
- 2. Understanding the Screening Tables
- 2.1 General Considerations
- 2.2 Exposure Assumptions
- 2.3 Toxicity Values
- 2.4 Chemical-specific Parameters
- 2.5 Maximum Contaminant Levels (MCLs)
- 2.6 Understanding Risk Output on the RSL Website

## 3. Using the SL Tables

- 3.1 Developing a Conceptual Site Model
- 3.2 Background
- 3.3 Potential Problems
- 4. Land Use Descriptions, Equations and Technical Documentation
- 4.1 Resident
- 4.2 Composite Worker
- 4.3 Outdoor Worker
- 4.4 Indoor Worker
- 4.5 Construction Worker
- 4.6 Recreator
- 4.7 Ingestion of Fish
- 4.8 Soil to Groundwater
- 4.9 Supporting Equations and Parameter Discussion

## **<u>5. Special Considerations</u>**

- 5.1 Cadmium
- 5.2 Lead
- 5.3 Manganese
- 5.4 Vanadium Compounds
- 5.5 Uranium
- 5.6 Chromium (VI)
- 5.7 Aminodinitrotoluenes
- 5.8 PCBs
- 5.9 Xylenes
- 5.10 Arsenic
- 5.11 Total Petroleum Hydrocarbons (TPHs)
- 5.12 Soil Saturation Limit (C<sub>sat</sub>)
- 5.13 SL Theoretical Ceiling Limit
- 5.14 Target Risk
- 5.15 Screening Sites with Multiple Contaminants

5.16 Deriving Soil Gas SLs
5.17 Mutagens
5.18 Trichloroethylene (TCE)
5.19 Mercuric Chloride (and other Mercury salts)
5.20 Cyanide (CN-)
5.21 Thallic Oxide and Thallium Selenite
5.22 Polycyclic Aromatic Hydrocarbon
5.23 Refractory Ceramic Fibers
5.24 Lanthanum Salts
6. Using the Calculator
7. References

## **1. Introduction**

The purpose of this website is to provide default screening tables and a calculator to assist Remedial Project Managers (RPMs), On Scene Coordinators (OSC's), risk assessors and others involved in decision-making concerning CERCLA hazardous waste sites and to determine whether levels of contamination found at the site may warrant further investigation or site cleanup, or whether no further investigation or action may be required.

Users within and outside the CERCLA program should use the tables or calculator results at their own discretion and they should take care to understand the assumptions incorporated in these results and to apply the SLs appropriately.

The SLs presented in the Generic Tables are chemical-specific concentrations for individual contaminants in air, drinking water and soil that may warrant further investigation or site cleanup. The SLs generated from the calculator may be site-specific concentrations for individual chemicals in soil, air, water and fish. It should be emphasized that SLs are not cleanup standards. We also do not recommend that the RSLs be used as cleanup levels for Superfund Sites until the recommendations in EPA's Supplemental Guidance to Risk Assessment Guidance for Superfund, Volume I, Part A ("Community Involvement in Superfund Risk Assessments (PDF)" (24 pp, 156 K) have been addressed. SLs should not be used as cleanup levels for a CERCLA site until the other remedy selections identified in the relevant portions of the National Contingency Plan (NCP), 40 CFR Part 300, have been evaluated and considered. PRGs (Preliminary Remediation Goals) is a term used to describe a project team's early and evolving identification of possible remedial goals. PRGs may be initially identified early in the Remedial Investigation/ Feasibility Study (RI/FS) process (e.g., at RI scoping) to select appropriate detection limits for RI sampling. Typically, it is necessary for PRGs to be more generic early in the process and to become more refined and site-specific as data collection and assessment progress. The SLs identified on this website are likely to serve as PRGs early in the process--e.g., at RI scoping and at screening of chemicals of potential concern (COPCs) for the baseline risk assessment. However, once the baseline risk assessment has been performed, PRGs can be derived from the calculator using site-specific risks, and the SLs in the Generic Tables are less likely to apply. PRGs developed in the FS will usually be based on site-specific risks and Applicable or Relevant and Appropriate Requirements (ARARs) and not on generic SLs.

## 2. Understanding the Screening Tables

## **2.1 General Considerations**

Risk-based SLs are derived from equations combining exposure assumptions with chemical-specific toxicity

values.

## **2.2 Exposure Assumptions**

Generic SLs are based on default exposure parameters and factors that represent Reasonable Maximum Exposure (RME) conditions for long-term/chronic exposures and are based on the methods outlined in EPA's <u>Risk Assessment Guidance for Superfund</u>, <u>Part B Manual (1991) (PDF)</u> (68 pp, 721 K) and Soil Screening Guidance documents (<u>1996 (PDF)</u> (89 pp, 863 K) and <u>2002 (PDF)</u> (187 pp, 2.2MB).

Site-specific information may warrant modifying the default parameters in the equations and calculating sitespecific SLs, which may differ from the values in these tables. In completing such calculations, the user should answer some fundamental questions about the site. For example, information is needed on the contaminants detected at the site, the land use, impacted media and the likely pathways for human exposure.

Whether these generic SLs or site-specific screening levels are used, it is important to clearly demonstrate the equations and exposure parameters used in deriving SLs at a site. A discussion of the assumptions used in the SL calculations should be included in the documentation for a CERCLA site.

## 2.3 Toxicity Values

In 2003, EPA's Superfund program revised its hierarchy of human health toxicity values, providing three tiers of toxicity values in a <u>memo (PDF)</u> (4 pp, 225 K). Three tier 3 sources were identified in that guidance, but it was acknowledged that additional tier 3 sources may exist. The 2003 guidance did not attempt to rank or put the identified tier 3 sources into a hierarchy of their own. However, when developing the screening tables and calculator presented on this website, EPA needed to establish a hierarchy among the tier 3 sources. The toxicity values used as "defaults" in these tables and calculator are consistent with the 2003 guidance. Chronic and subchronic toxicity values from the following sources, in the order in which they are presented below, are used as the defaults in these tables and calculator.

- 1. EPA's Integrated Risk Information System (IRIS).
- 2. The Provisional Peer Reviewed Toxicity Values (<u>PPRTVs</u>) derived by EPA's Superfund Health Risk Technical Support Center (STSC) for the EPA Superfund program. PPRTVs are archived (removed) when an IRIS profile is released, even if the IRIS profile indicates a toxicity value could not be derived. PPRTVs will retain subchronic values if IRIS releases a profile without subchronic values.
- 3. EPA's Office of Pesticide Programs (OPP) Human Health Benchmarks for Pesticides (<u>HHBPs</u>). IRIS has archived <u>51</u> chemical assessments for pesticides and for these pesticides has instead recommended the use of the toxicity values presented in the HHBP table. These include RfDs (also referred to as chronic PADs) and OSFs (referred to as cancer quantification values). OPP lists 363 pesticides in the HHBP table. Only the 51 archived by IRIS will be used in the RSL calculations.
- 4. The Agency for Toxic Substances and Disease Registry (<u>ATSDR</u>) minimal risk levels (MRLs). An <u>MRL</u> is an estimate of the daily human exposure to a hazardous substance that is likely to be without appreciable risk of adverse non-cancer health effects over a specified duration of exposure. These substance specific estimates, which are intended to serve as screening levels, are used by ATSDR health assessors and other responders to identify contaminants and potential health effects that may be of concern at hazardous waste sites.

- 5. The California Environmental Protection Agency Office of Environmental Health Hazard Assessment (<u>OEHHA</u>) provides toxicity values for the State of California. The OEHHA Toxicity Criteria Database website should be monitored for any updates to the toxicity values.
- 6. In the Fall 2009, this new source of toxicity values used was added: screening toxicity values in an appendix to certain PPRTV assessments. While we have less confidence in a screening toxicity value than in a PPRTV, we put these ahead of HEAST toxicity values because these appendix screening toxicity values are more recent and use current EPA methodologies in the derivation, and because the PPRTV appendix screening toxicity values also receive external peer review. To alert users when these values are used, the key presents an "X" (for Appendix) rather than a "P" (for PPRTV). The following is taken from a PPRTV appendix and states the intended usage of appendix screening levels.

However, information is available for this chemical, which although insufficient to support derivation of a provisional toxicity value, under current guidelines, may be of limited use to risk assessors. In such cases, the Superfund Health Risk Technical Support Center summarizes available information in an appendix and develops a "screening value." Appendices receive the same level of internal and external scientific peer review as the PPRTV documents to ensure their appropriateness within the limitations detailed in the document. Users of screening toxicity values in an appendix to a PPRTV assessment should understand that there is considerably more uncertainty associated with the derivation of an appendix screening toxicity value than for a value presented in the body of the assessment. Questions or concerns about the appropriate use of screening values should be directed to the Superfund Health Risk Technical Support Center.

7. The EPA Superfund program's <u>Health Effects Assessment Summary Table</u>. Values in HEAST are archived (removed) when an IRIS profile or a PPRTV paper is released, even if the PPRTV paper indicates a toxicity value could not be derived.

Users of these screening tables and calculator wishing to consider using other toxicity values, including toxicity values from additional sources, may find the discussions and seven preferences on selecting toxicity values in the attached Environmental Council of States paper useful for this purpose (ECOS website, ECOS paper(DOC)).

When using toxicity values, users are encouraged to carefully review the basis for the value and to document the basis of toxicity values used on a CERCLA site.

Please contact a Superfund risk assessor in your Region for help with chemicals that lack toxicity values in the sources outlined above.

## **2.3.1 Reference Doses**

The current, or recently completed, EPA toxicity assessments used in these screening tables (IRIS and PPRTVs) define a reference dose, or RfD, as an estimate (with uncertainty spanning perhaps an order of magnitude) of a daily oral exposure to the human population (including sensitive subgroups) that is likely to be without an appreciable risk of deleterious effects during a lifetime. It can be derived from a NOAEL, LOAEL, or benchmark dose, or using categorical regression, with uncertainty factors generally applied to reflect limitations of the data used. RfDs are generally the toxicity value used most often in evaluating noncancer health effects at Superfund sites. Various types of RfDs are available depending on the critical effect (developmental or other) and the length of exposure being evaluated (chronic or subchronic). Some of the SLs in these tables also use Agency for Toxic Substances and Disease Registry (ATSDR) chronic oral minimal risk levels (MRLs) as an oral chronic RfD. Screening toxicity values in an appendix to certain
PPRTV assessments were added to the hierarchy in the fall of 2009. The HEAST RfDs used in these SLs were based upon then current EPA toxicity methodologies, but did not use the more recent benchmark dose or categorical regression methodologies. Chronic oral reference doses and ATSDR chronic oral MRLs are expressed in units of (mg/kg-day).

## 2.3.1.1 Chronic Reference Doses

Chronic oral RfDs are specifically developed to be protective for long-term exposure to a compound. As a guideline for Superfund program risk assessments, chronic oral RfDs generally should be used to evaluate the potential noncarcinogenic effects associated with exposure periods greater than 7 years (approximately 10 percent of a human lifetime). However, this is not a bright line. Note, that ATSDR defines chronic exposure as greater than 1 year for use of their values. The calculator requires the user to select between chronic and subchronic toxicity values.

## 2.3.1.2 Subchronic Reference Doses

Subchronic oral RfDs are specifically developed to be protective for short-term exposure to a compound. As a guideline for Superfund program risk assessments, subchronic oral RfDs should generally be used to evaluate the potential noncarcinogenic effects of exposure periods between two weeks and seven years. However, this is not a bright line. Note, that ATSDR defines subchronic exposure as less than 1 year for use of their values. The calculator requires the user to select between chronic and subchronic toxicity values.

# 2.3.2 Reference Concentrations

The current, or recently completed, EPA toxicity assessments used in these screening tables (IRIS and PPRTV assessments) define a reference concentration (RfC) as an estimate (with uncertainty spanning perhaps an order of magnitude) of a continuous inhalation exposure to the human population (including sensitive subgroups) that is likely to be without an appreciable risk of deleterious effects during a lifetime. It can be derived from a NOAEL, LOAEL, or benchmark concentration, or using categorical regression with uncertainty factors generally applied to reflect limitations of the data used. Various types of RfCs are available depending on the critical effect (developmental or other) and the length of exposure being evaluated (chronic or subchronic). These screening tables also use ATSDR chronic inhalation MRLs as a chronic RfC, intermediate inhalation MRLs as a subchronic RfC and California Environmental Protection Agency (chronic) Reference Exposure Levels (RELs) as chronic RfCs. Screening toxicity values in an appendix to certain PPRTV assessments were added to the hierarchy in the fall of 2009. These screening tables may also use Some RfCs from EPA's HEAST tables.

# **2.3.2.1** Chronic Reference Concentrations

The chronic inhalation reference concentration is generally used for continuous or near continuous inhalation exposures that occur for 7 years or more. However, this is not a bright line, and ATSDR chronic MRLs are based on exposures longer than 1 year. EPA chronic inhalation reference concentrations are expressed in units of (mg/m<sup>3</sup>). Cal EPA RELs are presented in  $\mu$ g/m<sup>3</sup> and have been converted to mg/m<sup>3</sup> for use in these screening tables. Some ATSDR inhalation MRLs are derived in parts per million (ppm) and some in mg/m<sup>3</sup>. For use in this table all were converted into mg/m<sup>3</sup>. The calculator requires the user to select between chronic and subchronic toxicity values.

#### 2.3.2.2 Subchronic reference Concentrations

The subchronic inhalation reference concentration is generally used for exposures that are between 2 weeks and 7 years. However, this is not a bright line, and ATSDR subchronic MRLs are based on exposures less than 1 year. EPA subchronic inhalation reference concentrations are expressed in units of  $(mg/m^3)$ . Cal EPA RELs are presented in  $\mu g/m^3$  and have been converted to  $mg/m^3$  for use in these screening tables. Some ATSDR intermediate inhalation MRLs are derived in parts per million (ppm) and some in  $mg/m^3$ . For use in this table all were converted into  $mg/m^3$ . The calculator requires the user to select between chronic and subchronic toxicity values.

## 2.3.3 Slope Factors

A slope factor and the accompanying weight-of-evidence determination are the toxicity data most commonly used to evaluate potential human carcinogenic risks. Generally, the slope factor is a plausible upper-bound estimate of the probability of a response per unit intake of a chemical over a lifetime. The slope factor is used in risk assessments to estimate an upper-bound lifetime probability of an individual developing cancer as a result of exposure to a particular level of a potential carcinogen. Slope factors should always be accompanied by the weight-of-evidence classification to indicate the strength of the evidence that the agent is a human carcinogen.

Oral slope factors are toxicity values for evaluating the probability of an individual developing cancer from oral exposure to contaminant levels over a lifetime. Oral slope factors are expressed in units of (mg/kg-day) <sup>-1</sup>. When available, oral slope factors from EPA's IRIS or PPRTV assessments are used. The ATSDR does not derive cancer toxicity values (e.g. slope factors or inhalation unit risks). Some oral slope factors used in these screening tables were derived by the California Environmental Protection Agency, whose methodologies are quite similar to those used by EPA's IRIS and PPRTV assessments. Screening toxicity values in an appendix to certain PPRTV assessments were added to the hierarchy in the fall of 2009. When oral slope factors are not available in IRIS then PPRTVs, Cal EPA assessments, PPRTV appendices or values from HEAST are used.

# 2.3.4 Inhalation Unit Risk

The IUR is defined as the upper-bound excess lifetime cancer risk estimated to result from continuous exposure to an agent at a concentration of  $1 \ \mu g/m^3$  in air. Inhalation unit risk toxicity values are expressed in units of  $(\mu g/m^3)^{-1}$ .

When available, inhalation unit risk values from EPA's IRIS or PPRTV assessments are used. The ATSDR does not derive cancer toxicity values (e.g. slope factors or inhalation unit risks). Some inhalation unit risk values used in these screening tables were derived by the California Environmental Protection Agency, whose methodologies are quite similar to those used by EPA's IRIS and PPRTV assessments. Screening toxicity values in an appendix to certain PPRTV assessments were added to the hierarchy in the fall of 2009. When inhalation unit risk values are not available in IRIS then PPRTVs, Cal EPA assessments, PPRTV appendices or values from HEAST are used.

# 2.3.5 Toxicity Equivalence Factors

Some chemicals are members of the same family and exhibit similar toxicological properties; however, they differ in the degree of toxicity. Therefore, a toxicity equivalence factor (TEF) must first be applied to adjust the measured concentrations to a toxicity equivalent concentration.

The following table contains the various dioxin-like toxicity equivalency factors for Dioxins, Furans and dioxin-like PCBs (Van den Berg et al. 2006 (PDF) (19 pp, 290 K)), which are the World Health Organization 2005 values. These TEFs are also presented in the May 2013 fact sheet, "Use of Dioxin TEFs in Calculating Dioxin TEQs at CERCLA and RCRA Sites (PDF)" (8 pp, 105 K) which references the 2010 EPA report, "Recommended Toxicity Equivalence Factors (TEFs) for Human Health Risk Assessments of 2,3,7,8Tetrachlorodibenzo-p-dioxin and Dioxin-Like Compounds (PDF)" (38 pp, 532 K).

#### **Dioxin Toxicity Equivalence Factors**

CASRN	Dioxins and Furans	TEF	
Chlorinated dibenzo-p-dioxins			
1746-01-6	2,3,7,8-TCDD	1	
40321-76-4	1,2,3,7,8-PeCDD	1	
39227-28-6	1,2,3,4,7,8-HxCDD	0.1	
57653-85-7	1,2,3,6,7,8-HxCDD	0.1	
57653-85-7	1,2,3,7,8,9-HxCDD	0.1	
35822-46-9	1,2,3,4,6,7,8-HpCDD	0.01	
3268-87-9	OCDD	0.0003	
Chlorinated dibenzofurans			
51207-31-9	2,3,7,8-TCDF	0.1	
57117-41-6	1,2,3,7,8-PeCDF	0.03	
57117-31-4	2,3,4,7,8-PeCDF	0.3	
70648-26-9	1,2,3,4,7,8-HxCDF	0.1	
57117-44-9	1,2,3,6,7,8-HxCDF	0.1	

CASRN	Dioxins and Furans		TEF	
72918-21-9	1,2,3,7,8,9-HxCDF		0.1	
60851-34-5	2,3,4,6,7,8-HxCDF		0.1	
35822-46-9	1,2,3,4,6,7,8-HpCDF		0.01	
55673-89-7	1,2,3,4,7,8,9-HpCDF		0.01	
39001-02-0	OCDF		0.0003	
PCBs				
	IUPAC No.	Structure		
	Non- <i>ortho</i>			
32598-13-3	77	3,3',4,4'-TetraCB	0.0001	
70362-50-4	81	3,4,4',5-TetraCB	0.0003	
57465-28-8	126	3,3',4,4',5-PeCB	0.1	
32774-16-6	169	3,3',4,4',5,5'-HxCB	0.03	
Mono- <i>ortho</i>				
32598-14-4	105	2,3,3',4,4'-PeCB	0.00003	
74472-37-0	114	2,3,4,4',5-PeCB	0.00003	
31508-00-6	118	2,3',4,4',5-PeCB	0.00003	
65510-44-3	123	2',3,4,4',5-PeCB	0.00003	
38380-08-4	156	2,3,3',4,4',5-HxCB	0.00003	

CASRN	Dioxins and Furans		TEF	
69782-90-7	157	2,3,3',4,4',5'-HxCB	0.00003	
52663-72-6	167	2,3',4,4',5,5'-HxCB	0.00003	
39635-31-9	189	2,3,3',4,4',5,5'-HpCB	0.00003	
Di-ortho*				
35065-30-6	170	2,2',3,3',4,4',5-HpCB	0.0001	
35065-29-3	180	2,2',3,4,4',5,5'-HpCB	0.00001	

\* Di-ortho values come from Ahlborg, U.G., et al. (1994), which are the WHO 1994 values from Toxic equivalency factors for dioxin-like PCBs: Report on WHO-ECEH and IPCS consultation, December 1993 <u>Chemosphere, Volume 28, Issue 6, March 1994, Pages 1049-1067 (PDF)</u> (19 pp, 880 K).

# 2.3.6 Relative Potency Factors (RPFs)

Some chemicals are members of the same family and exhibit similar toxicological properties; however, they differ in the degree of toxicity. Therefore, a relative potency factor (RPF) must first be applied to adjust the oral slope factor or inhalation unit risk based on the realtive potency to the primary compound.

#### Carcinogenic polycyclic aromatic hydrocarbons

*Provisional Guidance for Quantitative Risk Assessment of Polycyclic Aromatic Hydrocarbons* (EPA/600/R-93/089, July 1993), recommends that a RPF be used to convert concentrations of carcinogenic polycyclic aromatic hydrocarbons (cPAHs) to an equivalent concentration of benzo(a)pyrene when assessing the cancer risks posed by these substances from oral exposures. These RPFs are based on the potency of each compound relative to that of benzo(a)pyrene. For the toxicity value database, these RPFs have been applied to the toxicity values. Although this is not in complete agreement with the direction in the aforementioned documents, this approach was used so that toxicity values could be generated for each cPAH. Additionally, it should be noted that computationally it makes little difference whether the RPFs are applied to the concentrations of cPAHs found in environmental samples or to the toxicity values as long as the RPFs are not applied to both. However, if the adjusted toxicity values are used, the user will need to sum the risks from all cPAHs is what is derived when the RPFs are applied to the environmental concentrations of cPAHs and not to the toxicity values. These RPFs are not needed, and should not be used, with the Cal EPA toxicity values, **nor should they be used when calculating non-cancer risk**.

The <u>IRIS Profile</u> gives the following instructions for RPF application:

"It (BaP) also serves as an index chemical for deriving relative potency factors to estimate the carcinogenicity of other PAH congeners, such as in EPA's Relative Potency Factor approach for the assessment of the carcinogenicity of PAHs (U.S. EPA, 1993) (PDF) (28 pp, 1.4 MB)."

and

"The inhalation unit risk for benzo[a]pyrene is derived with the intention that it will be paired with EPA's relative potency factors for the assessment of the carcinogenicity of PAH mixtures. In addition, regarding the assessment of early life exposures, because cancer risk values calculated for benzo[a]pyrene were derived from adult animal exposures, and because benzo[a]pyrene carcinogenicity occurs via a mutagenic mode of action, exposures that occur during development should include the application of ADAFs (see Section 2.5)."

The following table presents the RPFs for cPAHs recommended in <u>Provisional Guidance for Quantitative</u> <u>Risk Assessment of Polycyclic Aromatic Hydrocarbons (PDF)</u> (28 pp, 1.4 MB).

<b>Relative</b> Po	tency Factors	s for Carcin	logenic Polyc	velic Aromatic	Hydrocarbons
iterative i o	icity racion		logenne i orge	y che i si onnatie	iiyui ocai bolis

CASRN	Compound	RPF
50-32-8	Benzo(a)pyrene	1.0
56-55-3	Benz(a)anthracene	0.1
205-99-2	Benzo(b)fluoranthene	0.1
207-08-9	Benzo(k)fluoranthene	0.01
218-01-9	Chrysene	0.001
53-70-3	Dibenz(a,h)anthracene	1.0
193-39-5	Indeno(1,2,3-c,d)pyrene	0.1

# **2.4 Chemical-specific Parameters**

Several chemical specific parameters are needed for development of the SLs.

#### 2.4.1 Sources

Many sources are used to populate the database of chemical-specific parameters. They are briefly described below.

- 1. The Physical Properties Database (<u>PhysProp</u> EXIT ) was developed by Syracuse Research Corporation (SRC). The PhysProp database contains chemical structures, names and physical properties for over 41,000 chemicals. Physical properties collected from a wide variety of sources include experimental, extrapolated and estimated values.
- 2. The Estimation Programs Interface (<u>EPI Suite<sup>TM</sup></u>) was developed by the US Environmental Protection Agency's Office of Pollution Prevention and Toxics and SRC. These programs estimate various chemical-specific properties. The calculations for these SL tables use the experimental values for a property over the estimated values.
- 3. EPA Supplemental Guidance for Developing Soil Screening Levels for Superfund Sites (<u>SSL</u>) and <u>Appendix A-C</u> (39 pp, 681 K), "Chemical Properties and Regulatory/Human Health Benchmarks for SSL Calculations". Table C-1: Chemical-Specific Properties used in SSL Calculations and Table C-4: Metal Kd Values (L/kg) as a Function of pH.
- 4. <u>WATER9 Version 2.0</u> is the Windows-based wastewater treatment model containing a database listing many organic compounds and procedures for obtaining reports of constituent fates, including air emissions and treatment effectiveness. This program supersedes WATER8, Chem9, and Chemdat8 WATER9.
- 5. CHEMFATE Database. CHEMFATE is part of the Environmental Fate Data Bases (<u>EFDB</u>) software developed by SRC under sponsorship of the U.S. Environmental Protection Agency. CHEMFATE contains physical property values, rate constants, and monitoring data for approximately 1700 chemicals.
- 6. Yaws' Handbook of Thermodynamic and Physical Properties of Chemical Compounds. Knovel, 2003.
- 7. Baes, C.F. 1984. Oak Ridge National Laboratory. A Review and Analysis of Parameters for Assessing Transport of Environmentally Released Radionuclides through Agriculture. Values are also found in Superfund Chemical Data Matrix (SCDM).
- 8. NIOSH Pocket Guide to Chemical Hazards (NPG), NIOSH Publication No. 97-140, February 2004.
- 9. CRC Handbook of Chemistry and Physics EXIT . (Various Editions)
- 10. Perry's Chemical Engineers' Handbook (Various Editions).McGraw-Hill. Online version available <u>here</u> EXIT . Green, Don W.; Perry, Robert H. (2008).
- 11. Lange's Handbook of Chemistry (Various Editions). Online version available <u>here</u> EXIT . Speight, James G. (2005). McGraw-Hill.
- U.S. EPA 2004. Risk Assessment Guidance for Superfund Volume I: Human Health Evaluation Manual (Part E, Supplemental Guidance for Dermal Risk Assessment) Final. OSWER 9285.7-02EP. July 2004. <u>Document (PDF)</u> (186 pp, 4.2 MB) and <u>website</u>.
- 13. The ARS Pesticide Properties Database: U.S. Department of Agriculture, Agricultural Research Service. 2009. <u>Document (PDF)</u> (393 pp, 775 K) and <u>website</u>.
- 14. The <u>PubChem</u> website published by the National Center for Biotechnology Information, U.S. National Library of Medicine, 8600 Rockville Pike, Bethesda, MD20894, USA.

- The <u>Hazardous Substance Data Bank (HSDB)</u> website published by the U.S. National Library of Medicine 8600 Rockville Pike, Bethesda, MD 20894 National Institutes of Health, Health & Human Services.
- 16. The <u>Agency for Toxic Substances & Disease Registry (ATSDR)</u> Toxicological Profiles. Agency for Toxic Substances and Disease Registry, 4770 Buford Hwy NE, Atlanta, GA 30341.

#### 2.4.2 Hierarchy by Parameter

Generally, the hierarchies below will work for organic and inorganic compounds.

 Organic Carbon Partition Coefficient (K<sub>oc</sub>) (L/kg). Not applicable for inorganics. EPI estimated values; SSL, Yaw estimated values; EPI experimental values; Yaw Experimental values. The exception to this hierarchy are the nine ionizable organics identified in table 42 of Part 5 of the <u>Soil Screening Guidance</u> <u>Technical Background Document (PDF)</u> (28 pp, 523 K). Appendix L goes into detail on the derivation of these values. The table is reproduced below:

Compound	K <sub>oc</sub> pH=6.8F
Benzoic acid	0.6
2-chlorophenol	388
2,4-dichlorophenol	147
2,4-dinitrophenol	0.01
pentachlorophenol (PCP)	592
2,3,4,5-tetrachlorophenol	4742
2,3,4,6-tetrachlorophenol	280
2,4,5-trichlorophenol	1597
2,4,6-trichlorophenol	381

- 2. Dermal Permeability Coefficient (K<sub>p</sub>) (cm/hour). EPI estimated values; RAGS Part E.
- 3. Effective Predictive Domain (EPD). Calculated based on RAGS Part E criteria for MW and log Kow.
- 4. Fraction Absorbed (FA). RAGS Part E Exhibit B-3; Calculated. Calculated FA values less than zero

are set to zero.

- 5. Molecular Weight (MW) (g/mole). PHYSPROP; EPI; CRC89; Perry's; Lange's; Yaws.
- Water Solubility (S) (mg/L at 25 °C, unless otherwise stated in the source). PHYSPROP experimental values; EPI experimental values; CRC; YAWS experimental values; PERRY; LANGE; PHYSPROP estimated values; Yaws estimated values; EPI estimated values (WATERNT v.1.01, WSKOWWIN v1.42 respectively).
- 7. Unitless Henry's Law Constant (H' at 25 °C, unless otherwise stated in the source.). PHYSPROP experimental values; EPI experimental values; YAWS experimental values; PHYSPROP extrapolated values; PHYSPROP estimated values; EPI group-estimated values; EPI bond-estimated values; PHYSPROP.
- 8. Henry's Law Constant (atm-m<sup>3</sup>/mole at 25 °C, unless otherwise stated in the source). PHYSPROP experimental values; EPI experimental values; YAWS experimental values; PHYSPROP extrapolated values; PHYSPROP estimated values; EPI group-estimated values; EPI bond-estimated values; PHYSPROP.
- 9. Diffusivity in Air (D<sub>ia</sub>) (cm<sup>2</sup>/s). WATER9 equations.
- 10. Diffusivity in Water ( $D_{iw}$ ) (cm<sup>2</sup>/s). WATER9 equations.
- 11. Fish Bioconcentration Factor (BCF) (L/kg). EPI experimental values; EPI estimated values.
- 12. Soil-Water Partition Coefficient (K<sub>d</sub>) (cm<sup>3</sup>/g). SSL; BAES.
- 13. Density (g/cm<sup>3</sup>). CRC; Perry's; Lange's; IRIS.
- 14. Melting Point (MP °C). PHYSPROP; EPI experimental values; CRC; Perry's; Lange's; Yaws freezing point; EPI estimated values.
- 15. log Octanol-Water Partition Coefficient (logKow). PHYSPROP, EPI experimental values; Yaws experimental values; EPI estimated values; Yaws estimated values.
- 16. Vapor Pressure (VP). PHYSPROP experimental values, EPI experimental values; PHYSPROP extrapolated values; PHYSPROP estimated values; EPI estimated values.
- 17. Critical Temperature (T<sub>c</sub> °K). CRC; Yaws Experimental; Yaws Estimated.
- 18. Enthalpy of vaporization at the normal boiling point (cal/mol). CRC, Yaws Extrapolated, Yaws Estimated.

#### 2.5 Maximum Contaminant Levels (MCLs)

The Safe Drinking Water Act (<u>SDWA</u>) was originally passed by Congress in 1974 to protect public health by regulating the nation's public drinking water supply. SDWA authorizes the United States Environmental Protection Agency (US EPA) to set national health based standards for drinking water to protect against both naturally-occurring and man-made contaminants that may be found in drinking water.

US EPA sets national standards for drinking water based on sound science to protect against health risks, considering available technology and costs. These National Primary Drinking Water Regulations set enforceable maximum contaminant levels (MCLs) for contaminants in drinking water or required ways to treat water to remove contaminants. The <u>MCLs are published here (PDF)</u> (6 pp, 924 K).

US EPA sets primary drinking water standards through a three-step process: First, US EPA identifies contaminants that may adversely affect public health and occur in drinking water with a frequency and at levels that pose a threat to public health. Second, US EPA determines a maximum contaminant level goal (MCLG) for contaminants it decides to regulate. This goal is the level of a contaminant in drinking water below which there is no known or expected risk to health. Third, US EPA specifies a MCL, the maximum permissible level of a contaminant in drinking water which is delivered to any user of a public water system. These levels are enforceable standards, and are set as close to the MCLGs as feasible.

MCLs are provided in the RSL tables and the calculator output for users' information. A few things should be understood about the differences between RSLs and MCLs.

- RSLs are generated by and for the Superfund program, and MCLs are generated by the Office of Water although they are both used by other federal and state programs.
- The RSL calculations may result in a lower concentration than the MCL for a contaminant. The most common reason for this is that the RSLs represent risk-based concentrations considering only the relationship between exposure and risk.
- RSLs are calculated considering ingestion, inhalation, and dermal exposure to tap water.
- MCLs are set as close to risk-based goals, or Maximum Contaminant Level Goal (MCLG), as feasible. MCLGs are non-enforceable public health goals. MCLGs consider only public health and not the limits of detection and treatment technology effectiveness. Conversely, MCLs take the best available analytical and treatment technologies and cost into consideration.
- MCLGs for noncancer effects are based on a Reference Dose and consider ingestion of drinking water with a relative source contribution. The relative source contribution is the percentage of total drinking water exposure for the general population, after considering other exposure routes (for example, food, inhalation).
- For chemical contaminants that are carcinogens, EPA sets the MCLG at zero if: 1) there is evidence that a chemical may cause cancer and 2) there is no dose below which the chemical is considered safe. If a chemical is carcinogenic, and a safe dose can be determined, EPA sets the MCLG at a level above zero that is safe.
- If you have questions about the use of MCLs and/or RSLs at a Superfund site, consult your regional risk assessor.

# 2.6 Understanding Risk Output on the RSL Website

The RSL calculator provides an option to select risk output. In the calculator, select yes if risk output is desired. Selecting risk output requires the calculator to be run in "Site Specific" mode. In site specific mode, the user will be required to enter site concentrations for each media and chemical selected. The "Soil to Groundwater" medium does not have risk output and the risk option will become disabled when selected. The risk and hazard index values presented on this site are chemical-specific values for individual contaminants in air, water, soil, and fish that may warrant further investigation or site cleanup.

This portion of the risk assessment process is generally referred to as "Risk Characterization". This step incorporates the outcome of the exposure and toxicity assessments to calculate the risk resulting from potential exposure to chemicals via the pathways and routes of exposure determined appropriate for the source area.

# 2.6.1 How Risk is Calculated

The process used to calculate risk (carcinogenic risk and hazard quotient) in this calculator does not follow the traditional method of first calculating a Chronic Daily Intake (CDI). Rather, risk is derived using a simple method that relies on the linear nature of the relationship between concentration and risk. Using the equation below, an RSL, the target risk or target hazard quotient used to calculate the RSL, and a concentration entered by the user are all that is required to calculate risk.

Carcinogenic: TR / RSL = Risk / C

Noncarcinogenic: THQ / RSL = HQ / C

The linear equation above is then rearranged to solve for risk:

Carcinogenic:  $Risk = (C \times TR) / RSL$ 

Noncarcinogenic:  $HQ = (C \times THQ) / RSL$ 

where:

Risk = a unitless probability of an individual developing cancer over a lifetime, determined with the equation above;

HQ = a unitless ratio of exposure concentration to reference concentration where a value greater than unity indicates an individual will likely experience adverse health effects;

C = Concentration entered by the user in site-specific mode [mg/kg;  $\mu$ g/m<sup>3</sup>;  $\mu$ g/L]

TR = Target Risk provided by the user in site-specific mode

THQ = Target Hazard Quotient provided by the user in site-specific mode

RSL = Regional Screening Level, determined by the values entered by the user in site-specific mode [mg/kg;  $\mu g/m^3$ ;  $\mu g/L$ ]

# 2.6.2 One-Hit Rule for Carcinogenic Risk

The linear risk equation, listed above, is valid only at low risk levels (below estimated risks of 0.01). For sites where chemical intakes might be high (estimated risks above 0.01, an alternate calculation should be used. The one-hit equation, which is consistent with the linear low-dose model, should be used instead (RAGS, part A, ch. 8). The results presented use this rule. In the following instances, the one-hit rule is used independently in the risk output tables:

- Risk from a single exposure route for a single chemical.
- Summation of single chemical risk (without one-hit rule applied to single chemical results) for multiple exposure routes (right of each row).
- Summation of risk (without one-hit rule applied to single chemical results) from a single exposure route for multiple chemicals (bottom of each column).
- Summation of total risk (without one-hit rule applied to single chemical results or summations listed above) from multiple chemicals across multiple exposure routes (bottom right hand cell).

# 3. Using the SL Tables

The "<u>Generic Tables</u>" page provides generic concentrations in the absence of site-specific exposure assessments. These concentrations can be used for:

- Prioritizing multiple sites or operable units or areas of concern within a facility or exposure units
- Setting risk-based detection limits for contaminants of potential concern (COPCs)
- Focusing future site investigation and risk assessment efforts (e.g., selecting COPCs for the baseline risk assessment)
- Identifying contamination which may warrant cleanup
- Identifying sites, or portions of sites, which warrant no further action or investigation
- Initial cleanup goals when site-specific data are lacking

Generic SLs are provided for multiple exposure pathways and for chemicals with both carcinogenic and noncarcinogenic effects. A Summary Table is provided that contains SLs corresponding to either a 10<sup>-6</sup> risk level for carcinogens or a Hazard Quotient (HQ) of 1 for non-carcinogens. The summary table identifies whether the SL is based on cancer or noncancer effects by including a "c" or "n" after the SL. The Supporting Tables provide SLs corresponding to a 10<sup>-6</sup> risk level for carcinogens and an HQ of 1 for noncarcinogens. Site specific SLs corresponding to an HQ of less than 1 may be appropriate for those sites where multiple chemicals are present that have RfDs or RfCs based on the same toxic endpoint. Site specific SLs based upon a cancer risk greater than 10<sup>-6</sup> can be calculated and may be appropriate based upon site specific considerations. However, caution is recommended to ensure that cumulative cancer risk for all actual and potential carcinogenic contaminants found at the site does not have a residual (after site cleanup, or when it has been determined that no site cleanup is required) cancer risk exceeding 10<sup>-4</sup>. Also, changing the target risk or HI may change the balance between the cancer and noncancer endpoints. At some concentrations, the cancer-risk concerns predominate; at other concentrations, noncancer-HI concerns predominate. The user must take care to consider both when adjusting target risks and hazards.

Tables are provided in either MS Excel or in PDF format. The following lists the tables provided and a description of what is contained in each:

- Summary Table provides a list of contaminants, toxicity values, MCLs and the lesser (more protective) of the cancer and noncancer SLs for resident soil, industrial soil, resident air, industrial air and tapwater.
- Residential Soil Supporting Table provides a list of contaminants, toxicity values and the cancer and noncancer SLs for resident soil.
- Industrial Soil Supporting Table provides a list of contaminants, toxicity values and the cancer and noncancer SLs for industrial soil.
- Residential Air Supporting Table provides a list of contaminants, toxicity values and the cancer and noncancer SLs for resident air.
- Industrial Air Supporting Table provides a list of contaminants, toxicity values and the cancer and noncancer SLs for industrial air.
- Residential Tapwater Supporting Table provides a list of contaminants, toxicity values, MCLs and the cancer and noncancer SLs for tapwater.

# **3.1 Developing a Conceptual Site Model**

When using generic SLs at a site, the exposure pathways of concern and site conditions should match those used in developing the SLs presented here. (Note, however, that future uses may not match current uses. Future uses are potential site uses that may occur in the future. At Superfund sites, future uses should be considered as well as current uses. RAGS Part A, Chapter 6, provides guidance on selecting future-use receptors.) Thus, it is necessary to develop a conceptual site model (CSM) to identify likely contaminant source areas, exposure pathways, and potential receptors. This information can be used to determine the applicability of SLs at the site and the need for additional information. The final CSM diagram represents linkages among contaminant sources, release mechanisms, exposure pathways, and routes and receptors based on historical information. It summarizes the understanding of the contamination problem. A separate CSM for ecological receptors can be useful. Part 2 and Attachment A of the Soil Screening Guidance for Superfund: Users Guide (EPA 1996) contains the steps for developing a CSM.

As a final check, the CSM should address the following questions:

- Are there potential ecological concerns?
- Is there potential for land use other than those used in the SL calculations (i.e., residential and commercial/industrial)?
- Are there other likely human exposure pathways that were not considered in development of the SLs?
- Are there unusual site conditions (e.g. large areas of contamination, high fugitive dust levels, potential for indoor air contamination)?

The SLs and later PRGs may need to be adjusted to reflect the answers to these questions.

Below is a potential CSM of the quantified pathways addressed in the SL Tables.



Conceptual Site Model of Qantified Exposure Pathways for Regional Screening Levels. Click image to view

full size image.

# **3.2 Background**

EPA may be concerned with two types of background at sites: naturally occurring and anthropogenic. Natural background is usually limited to metals whereas anthropogenic (i.e. human-made) "background" includes both organic and inorganic contaminants.

Please note that the SL tables, which are purely risk-based, may yield SLs lower than naturally occurring background concentrations of some chemicals in some areas. However, background considerations may be incorporated into the assessment and investigation of sites, as acknowledged in existing EPA guidance. Background levels should be addressed as they are for other contaminants at CERCLA sites. For further information see EPA's guidance <u>Role of Background in the CERCLA Cleanup Program (PDF)</u> (13 pp, 144 K), April 2002, (OSWER 9285.6-07P) and <u>Guidance for Comparing Background and Chemical Concentration in Soil for CERCLA Sites (PDF)</u> (89 pp, 1.2 MB), September 2002, (OSWER 9285.7-41).

Generally EPA does not clean up below natural background. In some cases, the predictive risk-based models generate SL concentrations that lie within or even below typical background concentrations for the same element or compound. Arsenic, aluminum, iron and manganese are common elements in soils that have background levels that may exceed risk-based SLs. This does not mean that these metals cannot be site-related, or that these metals should automatically be attributed to background. Attribution of chemicals to background is a site-specific decision; consult your regional risk assessor.

Where anthropogenic "background" levels exceed SLs and EPA has determined that a response action is necessary and feasible, EPA's goal will be to develop a comprehensive response to the widespread contamination. This will often require coordination with different authorities that have jurisdiction over the sources of contamination in the area.

# **3.3 Potential Problems**

As with any risk based screening table or tool, the potential exists for misapplication. In most cases, this results from not understanding the intended use of the SLs or PRGs. In order to prevent misuse of the SLs, the following should be avoided:

- Applying SLs to a site without adequately developing a conceptual site model that identifies relevant exposure pathways and exposure scenarios.
- Not considering the effects from the presence of multiple contaminants, where appropriate.
- Use of the SLs as cleanup levels without adequate consideration of the other NCP remedy selection criteria on CERCLA sites.
- Use of SL as cleanup levels without verifying numbers with a toxicologist or regional risk assessor.
- Use of outdated SLs when tables have been superseded by more recent values.
- Not considering the effects of additivity when screening multiple chemicals.

- Applying inappropriate target risks or changing a cancer target risk without considering its effect on noncancer, or vice versa.
- Not performing additional screening for pathways not included in these SLs (e.g., vapor intrusion, fish consumption).
- Adjusting SLs upward by factors of 10 or 100 without consulting a toxicologist or regional risk assessor.

# 4. Land Use Descriptions, Equations and Technical Documentation

The SLs consider human exposure to individual contaminants in air, drinking water and soil. The equations and technical discussion are aimed at developing risk-based SLs or PRGs. The following text presents the land use equations and their exposure routes. Table 1 presents the definitions of the variables and their default values. Any alternative values or assumptions used in developing SLs on a site should be presented with supporting rationale in the decision document on CERCLA sites.

#### 4.1 Resident

## 4.1.1 Resident Soil

This receptor spends most, if not all, of the day at home. The activities for this receptor involve typical home making chores (cooking, cleaning and laundering) as well as outdoor activities. The resident is assumed to be exposed to contaminants via the following pathways: incidental ingestion of soil, dermal contact with soil, inhalation of volatiles and fugitive dust. Adults and children exhibit different ingestion rates for soil. For example, the child resident is assumed to ingest 200 mg per day while the adult ingests 100 mg per day. To account for changes in intake as the receptor ages, age adjusted intake equations were developed.

Note that the soil ingestion rates are intended to also represent ingestion of indoor dust. According to U.S. EPA 2011, "The source of the soil in these recommendations could be outdoor soil, indoor containerized soil used to support growth of indoor plants, or a combination of both outdoor soil and containerized indoor soil. The inhalation and subsequent swallowing of soil particles is accounted for in these recommended values, therefore, this pathway does not need to be considered separately." Further, according to U.S. EPA 1997, "Although the recommendations presented below are derived from studies which were mostly conducted in the summer, exposure during the winter months when the ground is frozen or snow covered should not be considered as zero. Exposure during these months, although lower than in the summer months, would not be zero because some portion of the house dust comes from outdoor soil."

# This land use is for developing residential default screening levels that are presented in the RSL Generic Tables.

#### 4.1.1.1 Noncarcinogenic-child

The residential soil land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil

$$SL_{res-soil-nc-ing-c}(mg/kg) = \frac{THQ \times AT_{res-c}\left(\frac{365 \text{ days}}{\text{year}} \times ED_{res-c}(6 \text{ years})\right) \times BW_{res-c}(15 \text{ kg})}{EF_{res-c}\left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res-c}(6 \text{ years}) \times \frac{RBA}{RfD_0\left(\frac{mg}{\text{kg-day}}\right)} \times IRS_{res-c}\left(\frac{200 \text{ mg}}{\text{day}}\right) \times \frac{10^{-6} \text{kg}}{1 \text{ mg}}}$$

• dermal contact with soil

$$SL_{res-soil-nc-der-c}(mg/kg) = \frac{THQ \times AT_{res-c}\left(\frac{365 \text{ days}}{\text{year}} \times ED_{res-c}\left(6 \text{ years}\right)\right) \times BW_{res-c}(15 \text{ kg})}{EF_{res-c}\left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res-c}\left(6 \text{ years}\right) \times \frac{1}{\left(RfD_{o}\left(\frac{mg}{\text{kg-day}}\right) \times GIABS\right)} \times SA_{res-c}\left(\frac{2373 \text{ cm}^{2}}{\text{day}}\right) \times AF_{res-c}\left(\frac{0.2 \text{ mg}}{\text{cm}^{2}}\right) \times ABS_{d} \times \frac{10^{-6} \text{kg}}{1 \text{ mg}}}{1 \text{ mg}}$$

• inhalation of volatiles and particulates emitted from soil

$$SL_{res-soil-nc-inh-c} (mg/kg) = \frac{THQ \times AT_{res-c} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{res-c} \left(6 \text{ years}\right)\right)}{EF_{res-c} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res-c} \left(6 \text{ years}\right) \times ET_{res-c} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \frac{1}{RfC \left(\frac{mg}{m3}\right)} \times \left(\frac{1}{VF_{ulim} \left(\frac{m3}{kg}\right)} + \frac{1}{PEF\left(\frac{m3}{kg}\right)}\right)}$$

Total

$$SL_{res \cdot soil - nc \cdot tot - c} \left( mg/kg \right) = \frac{1}{\frac{1}{SL_{res \cdot soil - nc \cdot ing - c}} + \frac{1}{SL_{res \cdot soil - nc \cdot der - c}} + \frac{1}{$$

#### 4.1.1.2 Noncarcinogenic-adult

The residential soil land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil

$$SL_{res \cdot soil - nc \cdot ing \cdot a} (mg/kg) = \frac{THQ \times AT_{res \cdot a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{res} (26 \text{ years})\right) \times BW_{res \cdot a} (80 \text{ kg})}{EF_{res \cdot a} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res} (26 \text{ years}) \times \frac{RBA}{RfD_0 \left(\frac{mg}{\text{kg} \cdot \text{day}}\right)} \times IRS_{res \cdot a} \left(\frac{100 \text{ mg}}{\text{day}}\right) \times \frac{10^{-6} \text{kg}}{1 \text{ mg}}}{1 \text{ mg}}}$$

• dermal contact with soil

$$SL_{res-soil-nc-der-a}(mg/kg) = \frac{THQ \times AT_{res-a}\left(\frac{365 \text{ days}}{\text{year}} \times ED_{res}(26 \text{ years})\right) \times BW_{res-a}(80 \text{ kg})}{EF_{res-a}\left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res}(26 \text{ years}) \times \frac{1}{\left(\frac{RfD_{o}\left(\frac{mg}{\text{kg-day}}\right) \times GIABS}{\sqrt{\frac{RfD_{o}\left(\frac{mg}{\text{kg-day}}\right)} \times GIABS}\right)} \times SA_{res-a}\left(\frac{6032 \text{ cm}^{2}}{\text{day}}\right) \times AF_{res-a}\left(\frac{0.07 \text{ mg}}{\text{cm}^{2}}\right) \times ABS_{d} \times \frac{10^{-6} \text{ kg}}{1 \text{ mg}}}{1 \text{ mg}}$$

• inhalation of volatiles and particulates emitted from soil

$$SL_{res-soil-nc-inh-a} (mg/kg) = \frac{THQ \times AT_{res-a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{res} (26 \text{ years})\right)}{EF_{res-a} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res} (26 \text{ years}) \times ET_{res-a} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \frac{1}{RfC \left(\frac{mg}{m^3}\right)} \times \left(\frac{1}{VF_{ulim} \left(\frac{m^3}{kg}\right)} + \frac{1}{PEF \left(\frac{m^3}{kg}\right)}\right)}$$

• Total



#### 4.1.1.3 Carcinogenic

The residential soil land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil

$$SL_{res-soil-ca-ing} (mg/kg) = \frac{TR \times AT_{res} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right)}{CSF_{0} \left(\frac{mg}{\text{kg}-\text{day}}\right)^{-1} \times RBA \times IFS_{res-adj} \left(\frac{36,750 \text{ mg}}{\text{kg}}\right) \times \left(\frac{10^{-6}\text{kg}}{\text{mg}}\right)}{\text{where:}}$$

$$IFS_{res-adj} \left(\frac{36,750 \text{ mg}}{\text{kg}}\right) = \left(\frac{EF_{res-c} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res-c} \left(6 \text{ years}\right) \times IRS_{res-c} \left(\frac{200 \text{ mg}}{\text{day}}\right)}{BW_{res-c} (15 \text{ kg})} + \frac{EF_{res-a} \left(\frac{350 \text{ days}}{\text{year}}\right) \times \left(ED_{res} (26 \text{ years}) - ED_{res-c} \left(6 \text{ years}\right)\right) \times IRS_{res-a} \left(\frac{100 \text{ mg}}{\text{day}}\right)}{BW_{res-a} \left(80 \text{ kg}\right)}\right)$$

• dermal contact with soil

$$SL_{res-soil-ca-der}(mg/kg) = \frac{TR \times AT_{res}\left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right)}{\left(\frac{CSF_{0}\left(\frac{mg}{\text{kg-day}}\right)^{-1}}{G|ABS}\right) \times DFS_{res-adj}\left(\frac{103,390 \text{ mg}}{\text{kg}}\right) \times ABS_{d} \times \left(\frac{10^{-6}\text{kg}}{\text{mg}}\right)}{\text{where:}}$$

$$DFS_{res-adj}\left(\frac{103,390 \text{ mg}}{\text{kg}}\right) = \left(\frac{EF_{res-c}\left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res-c}\left(6 \text{ years}\right) \times SA_{res-c}\left(\frac{2373 \text{ cm}^{2}}{\text{day}}\right) \times AF_{res-c}\left(\frac{0.2 \text{ mg}}{\text{cm}^{2}}\right)}{BW_{res-c}(15 \text{ kg})} + \frac{EF_{res-a}\left(\frac{350 \text{ days}}{\text{year}}\right) \times \left(ED_{res}\left(26 \text{ years}\right) - ED_{res-c}\left(6 \text{ years}\right)\right) \times SA_{res-a}\left(\frac{6032 \text{ cm}^{2}}{\text{day}}\right) \times AF_{res-a}\left(\frac{0.07 \text{ mg}}{\text{cm}^{2}}\right)}{BW_{res-a}(80 \text{ kg})}\right)$$

• inhalation of volatiles and particulates emitted from soil

$$SL_{res \cdot soil-ca-inh} (mg/kg) = \frac{TR \times AT_{res} \left( \frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years}) \right)}{IUR \left( \frac{\mu g}{mg} \right)^{-1} \times \left( \frac{1000 \ \mu g}{mg} \right) \times EF_{res} \left( \frac{350 \ \text{days}}{\text{year}} \right) \times \left( \frac{1}{VF_{ulim} \left( \frac{m^3}{kg} \right)^{+} PEF \left( \frac{m^3}{kg} \right)} \right) \times ED_{res} (26 \text{ years}) \times ET_{res} \left( \frac{24 \ hours}{day} \right) \times \left( \frac{1 \ day}{24 \ hours} \right) \times \left( \frac{1}{24 \$$

• Total



#### 4.1.1.4 Mutagenic

The residential soil land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil

$$SL_{res \cdot soil-mu-ing}(mg/kg) = \frac{TR \times AT_{res} \left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right)}{CSF_{0} \left(\frac{mg}{\text{kg-day}}\right)^{-1} \times RBA \times IFSM_{res \cdot adj} \left(\frac{166,833 \text{ mg}}{\text{kg}}\right) \times \left(\frac{10^{-6}\text{kg}}{\text{mg}}\right)}$$
where:
$$IFSM_{res \cdot adj} \left(\frac{166,833 \text{ mg}}{\text{kg}}\right) = \frac{\left(\frac{EF_{0-2} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{0-2}(2 \text{ years}) \times IRS_{0-2} \left(\frac{200 \text{ mg}}{\text{day}}\right) \times 10}{BW_{0-2}(15 \text{ kg})} + \frac{EF_{2-6} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{2-6}(4 \text{ years}) \times IRS_{2-6} \left(\frac{200 \text{ mg}}{\text{day}}\right) \times 3}{BW_{2-6}(15 \text{ kg})} + \frac{EF_{6-16} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{6-16}(10 \text{ years}) \times IRS_{6-16} \left(\frac{100 \text{ mg}}{\text{day}}\right) \times 3}{BW_{6-16}(80 \text{ kg})} + \frac{EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26}(10 \text{ years}) \times IRS_{16-26} \left(\frac{100 \text{ mg}}{\text{day}}\right) \times 1}{BW_{16-26}(80 \text{ kg})}$$

• dermal contact with soil

$$SL_{res-soil-mu-der} (mg/kg) = \frac{TR \times AT_{res} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right)}{\left(\frac{CSF_{0} \left(\frac{mg}{\text{kg-day}}\right)^{-1}}{\text{GIABS}}\right) \times DFSM_{res-adj} \left(\frac{428,260 \text{ mg}}{\text{kg}}\right) \times ABS_{d} \times \left(\frac{10^{-6} \text{kg}}{\text{mg}}\right)}{\text{where:}}$$

$$where:$$

$$DFSM_{res-adj} \left(\frac{428,260 \text{ mg}}{\text{kg}}\right) = \left(\frac{EF_{0-2} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{0-2} (2 \text{ years}) \times AF_{0-2} \left(\frac{0.2 \text{ mg}}{\text{cm}^{2}}\right) \times SA_{0-2} \left(\frac{2373 \text{ cm}^{2}}{\text{day}}\right) \times 10}{BW_{0-2} (15 \text{ kg})} + \frac{EF_{2-6} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{2-6} (4 \text{ years}) \times AF_{2-6} \left(\frac{0.2 \text{ mg}}{\text{cm}^{2}}\right) \times SA_{2-6} \left(\frac{2373 \text{ cm}^{2}}{\text{day}}\right) \times 3}{BW_{2-6} (15 \text{ kg})} + \frac{EF_{6-16} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{6-16} (10 \text{ years}) \times AF_{6-16} \left(\frac{0.07 \text{ mg}}{\text{cm}^{2}}\right) \times SA_{6-16} \left(\frac{6032 \text{ cm}^{2}}{\text{day}}\right) \times 3}{BW_{6-16} (80 \text{ kg})} + \frac{EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26} (10 \text{ years}) \times AF_{16-26} \left(\frac{0.07 \text{ mg}}{\text{cm}^{2}}\right) \times SA_{16-26} \left(\frac{6032 \text{ cm}^{2}}{\text{day}}\right) \times 1}{BW_{16-26} (80 \text{ kg})} \right)$$

• inhalation of volatiles and particulates emitted from soil

$$SL_{res-soil-mu-inh} (mg/kg) = \frac{TR \times AT_{res} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right)}{IUR \left(\frac{\mu g}{m3}\right)^{-1} \times \left(\frac{1}{VF_{ulim} \left(\frac{m^3}{kg}\right)} + \frac{1}{PEF \left(\frac{m^3}{kg}\right)}\right) \times \left(\frac{1000 \ \mu g}{mg}\right) \times \left(\frac{1000$$

• Total  

$$SL_{res-soil-mu-tot} (mg/kg) = \frac{1}{\frac{1}{\frac{1}{SL_{res-soil-mu-ing}} + \frac{1}{SL_{res-soil-mu-der}} + \frac{1}{SL_{res-soil-mu-inh}}}}$$

# 4.1.1.5 Vinyl Chloride - Carcinogenic

The residential soil land use equations, presented here, contain the following exposure routes:

incidental ingestion of soil



dermal contact with soil

$$SL_{res-soil-ca-vc-der}(mg/kg) = \frac{1R}{\left(\frac{CSF_{o}\left(\frac{mg}{kg\cdot day}\right)^{-1}}{GIABS} \times DFS_{res-adj}\left(\frac{103,390 mg}{kg}\right) \times ABS_{d} \times \frac{10^{-6}kg}{1 mg}}{AT_{res}\left(\frac{365 days}{year} \times LT (70 years)\right)}\right)^{+} + \frac{\frac{CSF_{o}\left(\frac{mg}{kg\cdot day}\right)^{-1}}{GIABS} \times SA_{res-c}\left(\frac{2373 cm^{2}}{day}\right) \times AF_{res-c}\left(\frac{0.2 mg}{cm^{2}}\right) \times ABS \times \frac{10^{-6}kg}{1 mg}}{BW_{res-c}(15 kg)}\right)}$$
where:  

$$DFS_{res-adj}\left(\frac{103,390 mg}{kg}\right) = \left(\frac{EF_{res-c}\left(\frac{350 days}{year}\right) \times ED_{res-c}(6 years) \times SA_{res-c}\left(\frac{2373 cm^{2}}{day}\right) \times AF_{res-c}\left(\frac{0.2 mg}{cm^{2}}\right)}{BW_{res-c}(15 kg)} + \frac{EF_{res-a}\left(\frac{350 days}{year}\right) \times ED_{res-c}(6 years) \times SA_{res-c}\left(\frac{2373 cm^{2}}{day}\right) \times AF_{res-a}\left(\frac{6032 cm^{2}}{day}\right)}{BW_{res-a}(80 kg)} + \frac{EF_{res-a}\left(\frac{350 days}{year}\right) \times \left(ED_{res}(26 years) - ED_{res-c}(6 years)\right) \times SA_{res-a}\left(\frac{6032 cm^{2}}{day}\right) \times AF_{res-a}\left(\frac{0.07 mg}{cm^{2}}\right)}{BW_{res-a}(80 kg)}$$

• inhalation of volatiles and particulates emitted from soil

$$SL_{res-soil-ca-vc-inh}(mg/kg) = \frac{TR}{\left[\frac{IUR\left(\frac{\mu g}{m3}\right)^{-1} \times EF_{res}\left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res}\left(26 \text{ years}\right) \times ET_{res}\left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \left(\frac{1000 \mu g}{mg}\right)\right]} + AT_{res}\left(\frac{365 \text{ days}}{\text{year}} \times LT\left(70 \text{ years}\right)\right) \times VF_{ulim}\left(\frac{m^3}{kg}\right)}{\left(\frac{IUR\left(\frac{\mu g}{m3}\right)^{-1}}{VF_{ulim}\left(\frac{m^3}{kg}\right)} \times \left(\frac{1000 \mu g}{mg}\right)\right)}\right]}$$
  
Total  
$$SL_{res-soil-ca-vc-tot}\left(mg/kg\right) = \frac{1}{\frac{1}{SL_{res-soil-ca-vc-ing}} + \frac{1}{SL_{res-soil-ca-vc-der}} + \frac{1}{SL_{res-soil-ca-vc-inh}}}$$

#### 4.1.1.6 Trichloroethylene - Carcinogenic and Mutagenic

SL

The residential soil land use equations, presented here, contain the following exposure routes:

res-soi⊦ca-vc-ing

• incidental ingestion of soil

$$IFSM_{res-adj}\left(\frac{166,833 \text{ mg}}{\text{kg}}\right) = \frac{TR \times AT_{res}\left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right)}{CSF_{0}\left(\frac{\text{mg}}{\text{kg}-\text{day}}\right)^{-1} \times RBA \times \left(\frac{10^{-6}\text{kg}}{\text{mg}}\right) \times \left[\left(CAF_{0}\left(0.804\right) \times IFS_{res-adj}\left(\frac{37,650 \text{ mg}}{\text{kg}}\right)\right) + \left(MAF_{0}\left(0.202\right) \times IFSM_{res-adj}\left(\frac{166,833 \text{ mg}}{\text{kg}}\right)\right)\right]}$$
where:  

$$IFS_{res-adj}\left(\frac{36,750 \text{ mg}}{\text{kg}}\right) = \left(\frac{ED_{res-c}\left(6 \text{ years}\right) \times EF_{res-c}\left(\frac{350 \text{ days}}{\text{year}}\right) \times IRS_{res-c}\left(\frac{200 \text{ mg}}{\text{day}}\right)}{BW_{res-a}\left(80 \text{ kg}\right)} \times IRS_{res-a}\left(\frac{100 \text{ mg}}{\text{day}}\right)}\right)$$
where:  

$$IFSM_{res-adj}\left(\frac{166,833 \text{ mg}}{\text{kg}}\right) = \left(\frac{ED_{0-2}\left(2 \text{ years}\right) \times EF_{0-2}\left(\frac{350 \text{ days}}{\text{year}}\right) \times IRS_{0-2}\left(\frac{200 \text{ mg}}{\text{day}}\right) \times IRS_{res-a}\left(\frac{100 \text{ mg}}{\text{day}}\right)}{BW_{res-a}\left(80 \text{ kg}\right)} + \frac{ED_{2-6}\left(4 \text{ years}\right) \times EF_{2-6}\left(\frac{350 \text{ days}}{\text{year}}\right) \times IRS_{2-6}\left(\frac{200 \text{ mg}}{\text{day}}\right) \times 1}{BW_{2-6}\left(15 \text{ kg}\right)} + \frac{ED_{6-16}\left(10 \text{ years}\right) \times EF_{6-16}\left(\frac{350 \text{ days}}{\text{year}}\right) \times IRS_{6-16}\left(\frac{100 \text{ mg}}{\text{day}}\right) \times 1}{BW_{2-6}\left(15 \text{ kg}\right)} + \frac{ED_{6-16}\left(10 \text{ years}\right) \times EF_{6-16}\left(\frac{350 \text{ days}}{\text{year}}\right) \times IRS_{16-26}\left(\frac{100 \text{ mg}}{\text{day}}\right) \times 1}{BW_{16-26}\left(80 \text{ kg}\right)} + \frac{ED_{16-26}\left(10 \text{ years}\right) \times EF_{16-26}\left(\frac{350 \text{ days}}{\text{year}}\right) \times IRS_{16-26}\left(\frac{100 \text{ mg}}{\text{day}}\right) \times 1}{BW_{16-26}\left(80 \text{ kg}\right)} + \frac{ED_{16-26}\left(10 \text{ years}\right) \times EF_{16-26}\left(\frac{350 \text{ days}}{\text{year}}\right) \times IRS_{16-26}\left(\frac{100 \text{ mg}}{\text{day}}\right) \times 1}{BW_{16-26}\left(80 \text{ kg}\right)} + \frac{ED_{16-26}\left(10 \text{ years}\right) \times EF_{16-26}\left(\frac{350 \text{ days}}{\text{year}}\right) \times IRS_{16-26}\left(\frac{100 \text{ mg}}{\text{day}}\right) \times 1}{BW_{16-26}\left(80 \text{ kg}\right)} \times 1$$

• dermal contact with soil



• inhalation of volatiles and particulates emitted from soil

$$SL_{res-soil-tce-inh}(mg/kg) = \frac{TR \times AT_{res}\left[\frac{360 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right]}{IUR\left(\frac{\mu g}{m_{3}}\right)^{-1} \times \left[\frac{1}{VF_{ulim}\left(\frac{m^{3}}{kg}\right)} + \frac{1}{PEF\left(\frac{m^{3}}{kg}\right)}\right] \times \left[\frac{1000 \ \mu g}{mg}\right] \times \left(\frac{1 \ day}{24 \ hours}\right) \times \left[\left(\frac{1000 \ \mu g}{mg}\right) \times \left(\frac{1 \ day}{24 \ hours}\right) \times ET_{0.2}\left(\frac{24 \ hours}{day}\right) \times MAF_{i}(0.244) \times 10\right] + \left[\left(\frac{CAF_{i}(0.756) \times EF_{res}\left(\frac{350 \ days}{year}\right) \times ET_{c.6}\left(\frac{350 \ days}{year}\right) \times ET_{2.6}\left(\frac{24 \ hours}{day}\right) \times MAF_{i}(0.244) \times 3\right] + \left[ED_{res}(26 \ years) \times ET_{res}\left(\frac{24 \ hours}{day}\right)\right] + \left[\left(\frac{ED_{0.2}(2 \ years) \times EF_{2.6}\left(\frac{350 \ days}{year}\right) \times ET_{2.6}\left(\frac{24 \ hours}{day}\right) \times MAF_{i}(0.244) \times 3\right] + \left(\frac{ED_{c.16}(10 \ years) \times EF_{6.16}\left(\frac{350 \ days}{year}\right) \times ET_{6.16}\left(\frac{24 \ hours}{day}\right) \times MAF_{i}(0.244) \times 3\right] + \left(\frac{ED_{6.16}(10 \ years) \times EF_{6.16}\left(\frac{350 \ days}{year}\right) \times ET_{6.26}\left(\frac{24 \ hours}{day}\right) \times MAF_{i}(0.244) \times 3\right) + \left(\frac{ED_{6.26}(10 \ years) \times EF_{6.26}\left(\frac{350 \ days}{year}\right) \times ET_{6.26}\left(\frac{24 \ hours}{day}\right) \times MAF_{i}(0.244) \times 3\right) + \left(\frac{ED_{6.26}(10 \ years) \times EF_{6.26}\left(\frac{350 \ days}{year}\right) \times ET_{6.26}\left(\frac{24 \ hours}{day}\right) \times MAF_{i}(0.244) \times 3\right) + \left(\frac{ED_{6.26}(10 \ years) \times EF_{6.26}\left(\frac{350 \ days}{year}\right) \times ET_{6.26}\left(\frac{24 \ hours}{day}\right) \times MAF_{i}(0.244) \times 3\right) + \left(\frac{ED_{6.26}(10 \ years) \times EF_{6.26}\left(\frac{350 \ days}{year}\right) \times ET_{6.26}\left(\frac{24 \ hours}{day}\right) \times MAF_{i}(0.244) \times 3\right) \right) \right)$$
Total.  
SL\_{res-soil-tce-tot}\left(mg/kg\right) = \frac{1}{\frac{1}{SL\_{res-soil-tce-ing}}} + \frac{1}{SL\_{res-soil-tce-der}}} + \frac{1}{SL\_{res-soil-tce-inh}}

A number of studies have shown that inadvertent ingestion of soil is common among children 6 years old and younger (Calabrese et al. 1989, Davis et al. 1990, Van Wijnen et al. 1990). Therefore, the dose method uses an age-adjusted soil ingestion factor that takes into account the difference in daily soil ingestion rates, body weights, and exposure duration for children from 1 to 6 years old and others from 7 to 26 years old. The

equation is presented below. This health-protective approach is chosen to take into account the higher daily rates of soil ingestion in children as well as the longer duration of exposure that is anticipated for a long-term resident. For more on this method, see <u>RAGS Part B</u>.

#### 4.1.1.7 Supporting Equations

• Child

$$\mathsf{BW}_{\mathsf{res-c}}(15 \text{ kg}) = \frac{\mathsf{BW}_{0-2}(15 \text{ kg}) \times \mathsf{ED}_{0-2}(2 \text{ years}) + \mathsf{BW}_{2-6}(15 \text{ kg}) \times \mathsf{ED}_{2-6}(4 \text{ years})}{\mathsf{ED}_{0-2}(2 \text{ years}) + \mathsf{ED}_{2-6}(4 \text{ years})}$$

$$\mathsf{EF}_{\mathsf{res-c}}\left(\frac{350 \text{ days}}{\mathsf{year}}\right) = \frac{\mathsf{EF}_{0-2}\left(\frac{350 \text{ days}}{\mathsf{year}}\right) \times \mathsf{ED}_{0-2}\left(2 \text{ years}\right) + \mathsf{EF}_{2-6}\left(\frac{350 \text{ days}}{\mathsf{year}}\right) \times \mathsf{ED}_{2-6}\left(4 \text{ years}\right)}{\mathsf{ED}_{0-2}\left(2 \text{ years}\right) + \mathsf{ED}_{2-6}\left(4 \text{ years}\right)}$$

$$\mathsf{ET}_{\mathsf{res-c}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) = \frac{\mathsf{ET}_{0-2}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{0-2}\left(2 \text{ years}\right) + \mathsf{ET}_{2-6}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{2-6}\left(4 \text{ years}\right)}{\mathsf{ED}_{0-2}\left(2 \text{ years}\right) + \mathsf{ED}_{2-6}\left(4 \text{ years}\right)}$$

$$AF_{res-c}\left(\frac{0.2 \text{ mg}}{cm^2}\right) = \frac{AF_{0-2}\left(\frac{0.2 \text{ mg}}{cm^2}\right) \times ED_{0-2}\left(2 \text{ years}\right) + AF_{2-6}\left(\frac{0.2 \text{ mg}}{cm^2}\right) \times ED_{2-6}\left(4 \text{ years}\right)}{ED_{0-2}\left(2 \text{ years}\right) + ED_{2-6}\left(4 \text{ years}\right)}$$

$$SA_{res-c}\left(\frac{2373 \text{ cm}^2}{\text{day}}\right) = \frac{SA_{0-2}\left(\frac{2373 \text{ cm}^2}{\text{day}}\right) \times ED_{0-2}\left(2 \text{ years}\right) + SA_{2-6}\left(\frac{2373 \text{ cm}^2}{\text{day}}\right) \times ED_{2-6}\left(4 \text{ years}\right)}{ED_{0-2}\left(2 \text{ years}\right) + ED_{2-6}\left(4 \text{ years}\right)}$$

$$\mathsf{IRS}_{\mathsf{res}\text{-}\mathsf{c}}\left(\frac{200\ \mathsf{mg}}{\mathsf{day}}\right) = \frac{\mathsf{IRS}_{0\text{-}2}\left(\frac{200\ \mathsf{mg}}{\mathsf{day}}\right) \times \mathsf{ED}_{0\text{-}2}\left(2\ \mathsf{years}\right) + \mathsf{IRS}_{2\text{-}6}\left(\frac{200\ \mathsf{mg}}{\mathsf{day}}\right) \times \mathsf{ED}_{2\text{-}6}\left(4\ \mathsf{years}\right)}{\mathsf{ED}_{0\text{-}2}\left(2\ \mathsf{years}\right) + \mathsf{ED}_{2\text{-}6}\left(4\ \mathsf{years}\right)}$$

• Adult

$$ED_{res-a} (20 \text{ years}) = ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})$$

$$EW_{res-a} (80 \text{ kg}) = \frac{EW_{6-16} (80 \text{ kg}) \times ED_{6-16} (10 \text{ years}) + EW_{16-26} (80 \text{ kg}) \times ED_{16-26} (10 \text{ years})}{ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})}$$

$$EF_{res-a} \left(\frac{350 \text{ days}}{\text{year}}\right) = \frac{EF_{6-16} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{6-16} (10 \text{ years}) + EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26} (10 \text{ years})}}{ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})}$$

$$ET_{res-a} \left(\frac{24 \text{ hours}}{\text{day}}\right) = \frac{ET_{6-16} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times ED_{6-16} (2 \text{ years}) + ET_{16-26} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times ED_{16-26} (4 \text{ years})}{ED_{6-16} (2 \text{ years}) + ED_{16-26} (4 \text{ years})}$$

$$AF_{res-a} \left(\frac{0.07 \text{ mg}}{\text{cm}^2}\right) = \frac{AF_{6-16} \left(\frac{0.07 \text{ mg}}{\text{cm}^2}\right) \times ED_{6-16} (10 \text{ years}) + AF_{16-26} \left(\frac{0.07 \text{ mg}}{\text{cm}^2}\right) \times ED_{16-26} (10 \text{ years})}{ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})}$$

$$SA_{res-a} \left(\frac{6032 \text{ cm}^2}{\text{day}}\right) = \frac{SA_{6-16} \left(\frac{6032 \text{ cm}^2}{\text{day}}\right) \times ED_{6-16} (10 \text{ years}) + SA_{16-26} \left(\frac{6032 \text{ cm}^2}{\text{day}}\right) \times ED_{16-26} (10 \text{ years})}{ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})}$$

$$IRS_{res-a}\left(\frac{100 \text{ mg}}{day}\right) = \frac{IRS_{6-16}\left(\frac{100 \text{ mg}}{day}\right) \times ED_{6-16}\left(10 \text{ years}\right) + IRS_{16-26}\left(\frac{100 \text{ mg}}{day}\right) \times ED_{16-26}\left(10 \text{ years}\right)}{ED_{6-16}\left(10 \text{ years}\right) + ED_{16-26}\left(10 \text{ years}\right)}$$

• Age-adjusted

 $ED_{res}(26 \text{ years}) = ED_{0-2}(2 \text{ years}) + ED_{2-6}(4 \text{ years}) + ED_{6-16}(10 \text{ years}) + ED_{16-26}(10 \text{ years})$ 

$$EF_{0-2}\left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{0-2}\left(2 \text{ years}\right) + EF_{2-6}\left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{2-6}\left(4 \text{ years}\right) + \\ EF_{res}\left(\frac{350 \text{ days}}{\text{year}}\right) = \frac{EF_{6-16}\left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{6-16}\left(10 \text{ years}\right) + EF_{16-26}\left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26}\left(10 \text{ years}\right)}{ED_{0-2}\left(2 \text{ years}\right) + ED_{2-6}\left(4 \text{ years}\right) + ED_{6-16}\left(10 \text{ years}\right) + ED_{16-26}\left(10 \text{ years}\right)}{ED_{0-2}\left(2 \text{ years}\right) + ED_{2-6}\left(2 \text{ years}\right) + ED_{2-6}\left(\frac{24 \text{ hours}}{\text{day}}\right) \times ED_{2-6}\left(4 \text{ years}\right) + \\ ET_{res}\left(\frac{24 \text{ hours}}{\text{day}}\right) = \frac{ET_{6-16}\left(\frac{24 \text{ hours}}{\text{day}}\right) \times ED_{6-16}\left(10 \text{ years}\right) + ET_{16-26}\left(\frac{24 \text{ hours}}{\text{day}}\right) \times ED_{16-26}\left(10 \text{ years}\right)}{ED_{0-2}\left(2 \text{ years}\right) + ED_{2-6}\left(4 \text{ years}\right) + ET_{16-26}\left(\frac{24 \text{ hours}}{\text{day}}\right) \times ED_{16-26}\left(10 \text{ years}\right)}{ED_{0-2}\left(2 \text{ years}\right) + ED_{2-6}\left(4 \text{ years}\right) + ET_{16-26}\left(\frac{24 \text{ hours}}{\text{day}}\right) \times ED_{16-26}\left(10 \text{ years}\right)}{ED_{0-2}\left(2 \text{ years}\right) + ED_{2-6}\left(4 \text{ years}\right) + ED_{6-16}\left(10 \text{ years}\right) + ED_{16-26}\left(10 \text{ years}\right)}{ED_{0-2}\left(2 \text{ years}\right) + ED_{2-6}\left(4 \text{ years}\right) + ED_{6-16}\left(10 \text{ years}\right) + ED_{16-26}\left(10 \text{ years}\right)}{ED_{0-2}\left(2 \text{ years}\right) + ED_{2-6}\left(4 \text{ years}\right) + ED_{6-16}\left(10 \text{ years}\right) + ED_{16-26}\left(10 \text{ years}\right)}{ED_{0-2}\left(2 \text{ years}\right) + ED_{2-6}\left(4 \text{ years}\right) + ED_{6-16}\left(10 \text{ years}\right) + ED_{16-26}\left(10 \text{ years}\right)}{ED_{0-2}\left(2 \text{ years}\right) + ED_{2-6}\left(4 \text{ years}\right) + ED_{6-16}\left(10 \text{ years}\right) + ED_{16-26}\left(10 \text{ years}\right)}{ED_{0-2}\left(2 \text{ years}\right) + ED_{2-6}\left(4 \text{ years}\right) + ED_{6-16}\left(10 \text{ years}\right) + ED_{16-26}\left(10 \text{ years}\right)}{ED_{16-26}\left(10 \text{ years}\right)}$$

# 4.1.2 Resident Tapwater

This receptor is exposed to chemicals in water that are delivered into a residence from sources such as groundwater or surface water. Ingestion of drinking water is an appropriate pathway for all chemicals. The inhalation exposure route is only calculated for volatile compounds. Activities such as showering, laundering, and dish washing contribute to contaminants in the air for inhalation. Dermal contact with tapwater is also considered for analytes determined to be within the effective predictive domain as described in Section 4.9.8.

# This land use is for developing residential default screening levels that are presented in the RSL Generic Tables.

#### 4.1.2.1 Noncarcinogenic-child

The tapwater land use equation, presented here, contains the following exposure routes:

• ingestion of water

$$SL_{res-wat-nc-ing-c}(\mu g/L) = \frac{THQ \times AT_{res-c}\left(\frac{365 \text{ days}}{\text{year}} \times ED_{res-c}\left(6 \text{ years}\right)\right) \times BW_{res-c}\left(15 \text{ kg}\right) \times \left(\frac{1000 \text{ }\mu g}{\text{ mg}}\right)}{EF_{res-c}\left(\frac{350 \text{ days}}{\text{ year}}\right) \times ED_{res-c}\left(6 \text{ years}\right) \times \frac{1}{RfD_{0}\left(\frac{mg}{\text{ kg-d}}\right)} \times IRW_{res-c}\left(\frac{0.78 \text{ L}}{\text{ day}}\right)}$$

• dermal

FOR INORGANICS:  $SL_{res-wat-nc-der-c}(\mu g/L) = \frac{DA_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) \times \left(\frac{1000 \text{ cm}^{3}}{L}\right)}{K_{p}\left(\frac{cm}{hour}\right) \times ET_{event-res-c}\left(\frac{0.54 \text{ hours}}{event}\right)}$ FOR ORGANICS:  $IF ET_{event-res-c}\left(\frac{0.54 \text{ hours}}{event}\right) \leq t^{*} \text{ (hours), then } SL_{res-wat-nc-der}(\mu g/L) = \frac{DA_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) \times \left(\frac{1000 \text{ cm}^{3}}{L}\right)}{2 \times FA \times K_{p}\left(\frac{cm}{hour}\right) \sqrt{\frac{6 \times revent}{event}} \times ET_{event-res-c}\left(\frac{0.54 \text{ hours}}{L}\right)}$   $IF ET_{event-res-c}\left(\frac{0.54 \text{ hours}}{event}\right) > t^{*} \text{ (hours), then } SL_{res-wat-nc-der}(\mu g/L) = \frac{DA_{event}\left(\frac{hours}{event}\right) \times ET_{event-res-c}\left(\frac{0.54 \text{ hours}}{L}\right)}{\pi}$   $IF ET_{event-res-c}\left(\frac{0.54 \text{ hours}}{event}\right) > t^{*} \text{ (hours), then } SL_{res-wat-nc-der}(\mu g/L) = \frac{DA_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) \times \left(\frac{1000 \text{ cm}^{3}}{L}\right)}{FA \times K_{p}\left(\frac{cm}{hour}\right) \times \left(\frac{ET_{event-res-c}\left(\frac{0.54 \text{ hours}}{Event}\right) \times \left(\frac{1+38 + 38^{2}}{(1+8)^{2}}\right)\right)}{\frac{1}{1+8}}$ where:  $DA_{event}\left(\frac{ug}{mg}\right) = \frac{THQ \times AT_{res-c}\left(\frac{365 \text{ days}}{year} \times ED_{res-c}(6 \text{ years})\right) \times \left(\frac{1000 \mu g}{mg}\right) \times BW_{res-c}(15 \text{ kg})}{\frac{1}{mg}}$ 

$$\frac{1}{\left(\frac{1}{\operatorname{RfD}_{0}\left(\frac{\operatorname{mg}}{\operatorname{kg-day}}\right)\times\operatorname{GIABS}}\right)}\times\operatorname{EV}_{\operatorname{res-c}}\left(\frac{1 \operatorname{events}}{\operatorname{day}}\right)\times\operatorname{ED}_{\operatorname{res-c}}\left(6 \operatorname{years}\right)\times\operatorname{EF}_{\operatorname{res-c}}\left(\frac{350 \operatorname{days}}{\operatorname{year}}\right)\times\operatorname{SA}_{\operatorname{res-c}}\left(6365 \operatorname{cm}^{2}\right)$$

• inhalation of volatiles

$$SL_{res-wat-nc-inh-c} (\mu g/L) = \frac{THQ \times AT_{res-c} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{res-c} \left(6 \text{ years}\right)\right) \times \left(\frac{1000 \mu g}{\text{mg}}\right)}{EF_{res-c} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res-c} \left(6 \text{ years}\right) \times ET_{res-c} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \frac{1}{RfC \left(\frac{\text{mg}}{\text{m}3}\right)} \times K \left(\frac{0.5 \text{ L}}{\text{m}3}\right)}$$

• Total



#### 4.1.2.2 Noncarcinogenic-adult

The tapwater land use equation, presented here, contains the following exposure routes:

• ingestion of water

$$SL_{res-wat-nc-ing-a} (\mu g \Lambda L) = \frac{THQ \times AT_{res-a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{res} (26 \text{ years})\right) \times BW_{res-a} (80 \text{ kg}) \times \left(\frac{1000 \mu g}{\text{mg}}\right)}{EF_{res-a} \left(350 \frac{\text{days}}{\text{year}}\right) \times ED_{res} (26 \text{ years}) \times \frac{1}{RfD_0 \left(\frac{\text{mg}}{\text{kg-d}}\right)} \times IRW_{res-a} \left(\frac{2.5 \text{ L}}{\text{day}}\right)}$$

dermal

FOR INDERGANICS:  $SL_{res.wat.nc.der.a}(\mu gL) = \frac{DA_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) \times \left(\frac{1000 \text{ cm}^{3}}{L}\right)}{K_{p}\left(\frac{cm}{hour}\right) \times ET_{event.res.a}\left(\frac{0.71 \text{ hours}}{event}\right)}$ FOR ORGANICS: IF ET<sub>event.res.a}\left(\frac{0.71 \text{ hours}}{event}\right) \leq t^{\*} (hours), then SL\_{res.wat.nc.der}(\mu g/L) = \frac{DA\_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) \times \left(\frac{1000 \text{ cm}^{3}}{L}\right)}{2 \times FA \times K\_{p}\left(\frac{cm}{hour}\right) \sqrt{\frac{6 \times r\_{event}\left(\frac{hours}{event}\right) \times ET\_{event.res.a}\left(\frac{0.71 \text{ hours}}{L}\right)}{\pi}}
or, IF ET<sub>event.res.a}\left(\frac{0.71 \text{ hours}}{event}\right) > t^{\*} (hours), then SL\_{res.wat.nc.der}(\mu g/L) = \frac{DA\_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) \times ET\_{event.res.a}\left(\frac{0.71 \text{ hours}}{L}\right)}{\pi}
where:  $DA_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) > t^{*} (hours), then SL_{res.wat.nc.der}(\mu g/L) = \frac{DA_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) \times \left(\frac{1000 \text{ cm}^{3}}{L}\right)}{FA \times K_{p}\left(\frac{cm}{hour}\right) \times \left[\frac{ET_{event.res.a}\left(\frac{0.71 \text{ hours}}{L}\right) \times \left(\frac{1 + 3B + 3B^{2}}{(1 + B)^{2}}\right)\right]}$ where:  $DA_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) = \frac{THQ \times AT_{res.a}\left(\frac{365 \text{ days}}{year} \times ED_{res}(26 \text{ years})\right) \times \left(\frac{1000 \mu g}{mg}\right) \times EM_{res.a}(19652 \text{ cm}^{2})}{\left(\frac{1}{RD_{0}\left(\frac{mg}{Kg-day}\right) \times GIABS}\right)} \times EV_{res.a}\left(\frac{1 \text{ events}}{day}\right) \times ED_{res}(26 \text{ years}) \times EF_{res.a}\left(\frac{350 \text{ days}}{year}\right) \times SA_{res.a}(19652 \text{ cm}^{2})$ </sub></sub>

• inhalation of volatiles

$$SL_{res-wat-nc-inh-a} (\mu g L) = \frac{THQ \times AT_{res-a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{res} (26 \text{ years})\right) \times \left(\frac{1000 \mu g}{\text{mg}}\right)}{EF_{res-a} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res} (26 \text{ years}) \times ET_{res-a} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \frac{1}{RfC \left(\frac{\text{mg}}{\text{m}3}\right)} \times K \left(\frac{0.5 \text{ L}}{\text{m}^3}\right)}$$

Total

$$SL_{res-wat-nc-tot-a}(\mu g/L) = \frac{1}{\frac{1}{SL_{res-wat-nc-ing-a}} + \frac{1}{SL_{res-wat-nc-der-a}} + \frac{1}{SL_{res-wat-nc-inh-a}}}$$

# 4.1.2.3 Carcinogenic

The tapwater land use equation, presented here, contains the following exposure routes:

• ingestion of water

$$SL_{res-wat-ca-ing}(\mu g/L) = \frac{TR \times AT_{res} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right) \times \left(\frac{1000 \mu g}{\text{mg}}\right)}{C SF_0 \left(\frac{\text{mg}}{\text{kg-day}}\right)^{-1} \times \left(IFW_{res-adj} \left(\frac{327.95 \text{ L}}{\text{kg}}\right)\right)}$$
where:
$$IFW_{res-adj} \left(\frac{327.95 \text{ L}}{\text{kg}}\right) = \left(\frac{EF_{res-c} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res-c} (6 \text{ years}) \times IRW_{res-c} \left(\frac{0.78 \text{ L}}{\text{day}}\right)}{BW_{res-c} (15 \text{ kg})} + \frac{1000 \mu g}{\text{kg}}\right) + \frac{1000 \mu g}{\text{kg}} + \frac{1000 \mu g}{\text{kg}} + \frac{1000 \mu g}{\text{kg}} + \frac{1000 \mu g}{\text{kg}}\right)}{BW_{res-c} (15 \text{ kg})}$$

• dermal



• inhalation of volatiles

$$SL_{res-wat-ca-inh} (\mu g/L) = \frac{TR \times AT_{res} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right)}{EF_{res} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res} (26 \text{ years}) \times ET_{res} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times IUR \left(\frac{\mu g}{m^3}\right)^{-1} \times K \left(\frac{0.5 \text{ L}}{m^3}\right)}$$

• Total

$$SL_{res-wat-ca-tot} (\mu g/L) = \frac{1}{\frac{1}{SL_{res-wat-ca-ing}} + \frac{1}{SL_{res-wat-ca-der}} + \frac{1}{SL_{res-wat-ca-inh}}}$$

#### 4.1.2.4 Mutagenic

The tapwater land use equation, presented here, contains the following exposure routes:

• ingestion of water

$$SL_{res-wat-mu-ing} (\mu g/L) = \frac{TR \times AT_{res} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right) \times \left(\frac{1000 \mu g}{\text{mg}}\right)}{CSF_0 \left(\frac{mg}{\text{kg-day}}\right)^{-1} \times IFWM_{res-adj} \left(\frac{1019.9 \text{ L}}{\text{kg}}\right)}$$
where:
$$IFWM_{res-adj} \left(\frac{1019.9 \text{ L}}{\text{kg}}\right) = \left(\frac{\frac{EF_{0-2} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{0-2} (\text{years}) \times IRW_{0-2} \left(\frac{0.78 \text{ L}}{\text{day}}\right) \times 10}{EW_{0-2} (15 \text{ kg})} + \frac{EF_{2-6} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{2-6} (\text{years}) \times IRW_{2-6} \left(\frac{0.78 \text{ L}}{\text{day}}\right) \times 3}{EW_{2-6} (15 \text{ kg})} + \frac{EF_{6-16} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{6-16} (\text{years}) \times IRW_{6-16} \left(\frac{2.5 \text{ L}}{\text{day}}\right) \times 3}{EW_{6-16} (80 \text{ kg})} + \frac{EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26} (\text{years}) \times IRW_{16-26} \left(\frac{2.5 \text{ L}}{\text{day}}\right) \times 1}{EW_{16-26} (80 \text{ kg})} + \frac{EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26} (\text{years}) \times IRW_{16-26} \left(\frac{2.5 \text{ L}}{\text{day}}\right) \times 1}{EW_{16-26} (80 \text{ kg})} + \frac{EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26} (\text{years}) \times IRW_{16-26} \left(\frac{2.5 \text{ L}}{\text{day}}\right) \times 1}{EW_{16-26} (80 \text{ kg})} + \frac{EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26} (\text{years}) \times IRW_{16-26} \left(\frac{2.5 \text{ L}}{\text{day}}\right) \times 1}{EW_{16-26} (80 \text{ kg})} + \frac{EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26} (80 \text{ kg})}{EW_{16-26} (80 \text{ kg})} + \frac{EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26} (80 \text{ kg})}{EW_{16-26} (80 \text{ kg})} + \frac{EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26} (80 \text{ kg})}{EW_{16-26} (80 \text{ kg})} + \frac{EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26} (80 \text{ kg})}{EW_{16-26} (80 \text{ kg})} + \frac{EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26} (80 \text{ kg})}{EW_{16-26} (80 \text{ kg})} + \frac{EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26} (80 \text{ kg})}{EW_{16-26} (80 \text{ kg})} + \frac{EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26} \left(\frac{350 \text{ days}}{\text{year}$$

• dermal



• inhalation of volatiles

$$SL_{res-wat-mu-inh} (\mu g/L) = \frac{TR \times AT_{res} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right)}{IUR \left(\frac{\mu g}{/m3}\right)^{-1} \times \left(\frac{0.5 \text{ L}}{m^3}\right) \times} \\ \left( \left( EF_{0-2} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{0-2} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times ED_{0-2} (2 \text{ years}) \times 10\right) + \\ \left( EF_{2-6} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{2-6} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times ED_{2-6} (4 \text{ years}) \times 3\right) + \\ \left( EF_{6-16} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{6-16} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times ED_{6-16} (10 \text{ years}) \times 3\right) + \\ \left( EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{16-26} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times ED_{16-26} (10 \text{ years}) \times 1\right) \right)$$

$$SL_{res-wat-mu-tot}(\mu g/L) = \frac{1}{\frac{1}{SL_{res-wat-mu-ing}} + \frac{1}{SL_{res-wat-mu-der}} + \frac{1}{SL_{res-wat-mu-inh}}}$$

#### 4.1.2.5 Vinyl Chloride - Carcinogenic

The tapwater land use equation, presented here, contains the following exposure routes:

• ingestion of water

$$SL_{res-wat-ca-vc-ing}(\mu g/L) = \frac{TR}{\left[ \frac{CSF_{o}\left(\frac{mg}{kg \cdot day}\right)^{-1} \times IFW_{res-adj}\left(\frac{327.95 \text{ L}}{kg}\right) \times \left(\frac{mg}{1000 \ \mu g}\right)}{AT_{res}\left(\frac{365 \ days}{year} \times LT \ (70 \ years)\right)} \right] + \left( \frac{CSF_{o}\left(\frac{mg}{kg \cdot day}\right)^{-1} \times IRW_{res-c}\left(\frac{0.78 \text{ L}}{day}\right) \times \left(\frac{mg}{1000 \ \mu g}\right)}{BW_{res-c} (15 \ kg)} \right)}{W_{res-c} \left(15 \ kg\right)} \right]$$
  
where:  
$$IFW_{res-adj}\left(\frac{327.95 \text{ L}}{kg}\right) = \left( \frac{EF_{res-c}\left(\frac{350 \ days}{year}\right) \times ED_{res-c} \left(6 \ years\right) \times IRW_{res-c}\left(\frac{0.78 \ L}{day}\right)}{BW_{res-c} (15 \ kg)} + \frac{EF_{res-a}\left(\frac{350 \ days}{year}\right) \times (ED_{res-c} (26 \ years) - ED_{res-c} (6 \ years)) \times IRW_{res-a}\left(\frac{2.5 \ L}{day}\right)}{BW_{res-a} \left(80 \ kg\right)} \right)$$

• dermal



· inhalation of volatiles

$$SL_{res-wat-ca-vc-inh} (\mu g/L) = \frac{TR}{\left[\frac{IUR\left(\frac{\mu g}{m3}\right)^{-1} \times EF_{res}\left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res}(26 \text{ years}) \times ET_{res}\left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times K\left(\frac{0.5 \text{ L}}{m^3}\right)\right]} + AT_{res}\left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right)$$

$$\left(IUR\left(\frac{\mu g}{m3}\right)^{-1} \times K\left(\frac{0.5 \text{ L}}{m^3}\right)\right)$$
• Total

$$SL_{res-wat-ca-vc-tot}(\mu g/L) = \frac{1}{\frac{1}{SL_{res-wat-ca-vc-ing}} + \frac{1}{SL_{res-wat-ca-vc-der}} + \frac{1}{SL_{res-wat-ca-vc-inh}}}$$

#### 4.1.2.6 Trichloroethylene - Carcinogenic and Mutagenic

The tapwater land use equation, presented here, contains the following exposure routes:

• ingestion of water

$$SL_{res-wat-tce-ing} (\mu g/L) = \frac{TR \times AT_{res} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right) \times \left(\frac{1000 \mu g}{\text{mg}}\right)}{CSF_{0} \left(\frac{\text{mg}}{\text{kg-day}}\right)^{-1} \times \left(\left(CAF_{0} (0.804) \times IFW_{res-adj} \left(\frac{327.95 \text{ L}}{\text{kg}}\right)\right)\right) + \left(MAF_{0} (0.202) \times IFWM_{res-adj} \left(\frac{1019.9 \text{ L}}{\text{kg}}\right)\right)}$$
where:
$$IFW_{res-adj} \left(\frac{327.95 \text{ L}}{\text{kg}}\right) = \left(\frac{ED_{res-c} (6 \text{ years}) \times EF_{res-c} \left(\frac{350 \text{ days}}{\text{year}}\right) \times IRW_{res-c} \left(\frac{0.78 \text{ L}}{\text{day}}\right)}{BW_{res-c} (15 \text{ kg})} + \frac{BW_{res-c} (15 \text{ kg})}{BW_{res-a} (80 \text{ kg})} \right)$$

where:  

$$IFWM_{res-adj}\left(\frac{1019.9 \text{ L}}{\text{kg}}\right) = \left(\frac{\frac{ED_{0-2}(2 \text{ years}) \times EF_{0-2}\left(\frac{350 \text{ days}}{\text{year}}\right) \times IRW_{0-2}\left(\frac{0.78 \text{ L}}{\text{day}}\right) \times 10}{BW_{0-2}(15 \text{ kg})} + \frac{ED_{2-6}\left(4 \text{ years}\right) \times EF_{2-6}\left(\frac{350 \text{ days}}{\text{year}}\right) \times IRW_{2-6}\left(\frac{0.78 \text{ L}}{\text{day}}\right) \times 3}{BW_{2-6}(15 \text{ kg})} + \frac{ED_{6-16}\left(10 \text{ years}\right) \times EF_{6-16}\left(\frac{350 \text{ days}}{\text{year}}\right) \times IRW_{6-16}\left(\frac{2.5 \text{ L}}{\text{day}}\right) \times 3}{BW_{6-16}(80 \text{ kg})} + \frac{ED_{16-26}\left(10 \text{ years}\right) \times EF_{16-26}\left(\frac{350 \text{ days}}{\text{year}}\right) \times IRW_{6-26}\left(\frac{2.5 \text{ L}}{\text{day}}\right) \times 1}{BW_{16-26}(80 \text{ kg})}\right)}\right)$$

• dermal



• inhalation of volatiles

$$SL_{res-wat-tce-inh} (\mu g/L) = \frac{TR \times AT_{res} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right)}{IUR \left(\frac{\mu g}{m^3}\right)^{-1} \times K \left(\frac{0.5 \text{ L}}{m^3}\right) \times} \\ \left( \left(EF_{res} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res} (26 \text{ years}) \times ET_{res} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times CAF_i (0.756)\right) + \\ \left( \left(\left(ED_{0-2} (2 \text{ years}) \times EF_{0-2} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{0-2} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times MAF_i (0.244) \times 10\right) + \\ \left(ED_{2-6} (4 \text{ years}) \times EF_{2-6} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{2-6} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times MAF_i (0.244) \times 3\right) + \\ \left(ED_{6-16} (10 \text{ years}) \times EF_{6-16} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{6-16} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times MAF_i (0.244) \times 3\right) + \\ \left(ED_{16-26} (10 \text{ years}) \times EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{16-26} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times MAF_i (0.244) \times 3\right) + \\ \left(ED_{16-26} (10 \text{ years}) \times EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{16-26} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times MAF_i (0.244) \times 1\right) \right) \right)$$

• Total  

$$SL_{res-wat-tce-tot} (\mu g/L) = \frac{1}{\frac{1}{SL_{res-wat-tce-ing}} + \frac{1}{SL_{res-wat-tce-der}} + \frac{1}{SL_{res-wat-tce-inh}}}$$

# 4.1.2.7 Supporting Equations

• Child

$$ED_{res-c} (6 \text{ years}) = ED_{0-2} (2 \text{ years}) + ED_{2-6} (4 \text{ years})$$

$$BW_{res-c} (15 \text{ kg}) = \frac{BW_{0-2} (15 \text{ kg}) \times ED_{0-2} (2 \text{ years}) + EW_{2-6} (15 \text{ kg}) \times ED_{2-6} (4 \text{ years})}{ED_{0-2} (2 \text{ years}) + ED_{2-6} (4 \text{ years})}$$

$$EV_{res-c} \left(\frac{1 \text{ event}}{\text{day}}\right) = \frac{EV_{0-2} \left(\frac{1 \text{ event}}{\text{day}}\right) \times ED_{0-2} (2 \text{ years}) + EV_{2-6} \left(\frac{1 \text{ event}}{\text{day}}\right) \times ED_{2-6} (4 \text{ years})}{ED_{0-2} (2 \text{ years}) + ED_{2-6} (4 \text{ years})}$$

$$EF_{res-c} \left(\frac{350 \text{ days}}{\text{year}}\right) = \frac{EF_{0-2} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{0-2} (2 \text{ years}) + ED_{2-6} (4 \text{ years})}{ED_{0-2} (2 \text{ years}) + ED_{2-6} (4 \text{ years})}$$

$$EF_{res-c} \left(\frac{350 \text{ days}}{\text{year}}\right) = \frac{EF_{0-2} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{0-2} (2 \text{ years}) + ET_{2-6} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{2-6} (4 \text{ years})}{ED_{0-2} (2 \text{ years}) + ED_{2-6} (4 \text{ years})}$$

$$ET_{event \text{ res-c}} \left(\frac{0.54 \text{ hours}}{\text{event}}\right) = \frac{ET_{event} (0.2) \left(\frac{0.54 \text{ hours}}{\text{event}}\right) \times ED_{0-2} (2 \text{ years}) + ET_{event} (2-4) \left(\frac{0.54 \text{ hours}}{\text{event}}\right) \times ED_{2-6} (4 \text{ years})}{ED_{0-2} (2 \text{ years}) + ED_{2-6} (4 \text{ years})}$$

$$ET_{res-c} \left(\frac{24 \text{ hours}}{\text{day}}\right) = \frac{ET_{0-2} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times ED_{0-2} (2 \text{ years}) + ET_{2-6} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times ED_{2-6} (4 \text{ years})}{ED_{0-2} (2 \text{ years}) + ED_{2-6} (4 \text{ years})}$$

$$ET_{res-c} \left(6365 \text{ cm}^2\right) = \frac{ET_{0-2} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times ED_{0-2} (2 \text{ years}) + ED_{2-6} (4 \text{ years})}{ED_{0-2} (2 \text{ years}) + ED_{2-6} (4 \text{ years})}$$

$$ET_{event} \left(\frac{0.78 \text{ L}}{\text{day}}\right) = \frac{1RW_{0-2} \left(\frac{0.78 \text{ L}}{\text{day}}\right) \times ED_{0-2} (2 \text{ years}) + EN_{2-6} \left(\frac{0.78 \text{ L}}{\text{day}}\right) \times ED_{2-6} (4 \text{ years})}$$

$$IRW_{res-c} \left(\frac{0.78 \text{ L}}{\text{day}}\right) = \frac{1RW_{0-2} \left(\frac{0.78 \text{ L}}{\text{day}}\right) \times ED_{0-2} (2 \text{ years}) + ED_{2-6} (4 \text{ years})}{ED_{0-2} (2 \text{ years}) + ED_{2-6} (4 \text{ years})}$$
• Adult

$$ED_{res-a} (20 \text{ years}) = ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})$$

$$EW_{res-a} (80 \text{ kg}) = \frac{BW_{6-16} (60 \text{ kg}) \times ED_{6-16} (10 \text{ years}) + BW_{16-26} (80 \text{ kg}) \times ED_{16-26} (10 \text{ years})}{ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})}$$

$$EV_{res-a} \left(\frac{1 \text{ event}}{\text{day}}\right) = \frac{EV_{6-16} \left(\frac{1 \text{ event}}{\text{day}}\right) \times ED_{6-16} (10 \text{ years}) + EV_{16-26} \left(\frac{1 \text{ event}}{\text{day}}\right) \times ED_{16-26} (10 \text{ years})}{ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})}$$

$$EF_{res-a} \left(\frac{350 \text{ days}}{\text{year}}\right) = \frac{EF_{6-16} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{6-16} (10 \text{ years}) + ED_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{16-26} (10 \text{ years})}{ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})} \times ED_{16-26} (10 \text{ years})$$

$$EF_{res-a} \left(\frac{0.71 \text{ hours}}{\text{event}}\right) = \frac{EF_{6-16} \left(\frac{271 \text{ hours}}{\text{event}}\right) \times ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})}{ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})} \times ED_{16-26} (10 \text{ years})$$

$$ET_{event-res-a} \left(\frac{0.71 \text{ hours}}{\text{event}}\right) = \frac{ET_{6-16} \left(\frac{24 \text{ hours}}{\text{event}}\right) \times ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})}{ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})} \times ED_{16-26} (10 \text{ years})$$

$$ET_{res-a} \left(\frac{24 \text{ hours}}{\text{day}}\right) = \frac{ET_{6-16} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})}{ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})} \times ED_{16-26} (10 \text{ years})$$

$$ET_{res-a} \left(\frac{24 \text{ hours}}{\text{day}}\right) = \frac{ET_{6-16} \left(\frac{19652 \text{ cm}^2}{\text{day}}\right) \times ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})}{ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})} \times ED_{16-26} (10 \text{ years})$$

$$ET_{res-a} \left(19652 \text{ cm}^2\right) = \frac{SA_{6-16} \left(\frac{19652 \text{ cm}^2}{\text{day}}\right) \times ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})}{ED_{6-16} (10 \text{ years}) + ED_{16-26} (10 \text{ years})} \times ED_{16-26} (10 \text{ years})}$$

$$RW_{res-a} \left(\frac{2.5 \text{ L}}{\text{day}}\right) = \frac{RW_{6-16} \left(\frac{2.5 \text{ L}}{\text{day}}\right) \times ED_{6-16} (10 \text{ years}) + RW_{16-26} \left(\frac{2.5 \text{ L}}{\text{day}}\right) \times ED_{16-26} (10 \text{$$

• Age-adjusted  $\mathsf{ED}_{\mathsf{res}}(\mathsf{26 years}) = \mathsf{ED}_{\mathsf{0-2}}(\mathsf{2 years}) + \mathsf{ED}_{\mathsf{2-6}}(\mathsf{4 years}) + \mathsf{ED}_{\mathsf{6-16}}(\mathsf{10 years}) + \mathsf{ED}_{\mathsf{16-26}}(\mathsf{10 years})$ 

$$\mathsf{EF}_{\mathsf{res}}\left(\frac{350 \text{ days}}{\text{year}}\right) \times \mathsf{ED}_{\mathsf{0-2}}\left(2 \text{ years}\right) + \mathsf{EF}_{\mathsf{2-6}}\left(\frac{350 \text{ days}}{\text{year}}\right) \times \mathsf{ED}_{\mathsf{2-6}}\left(4 \text{ years}\right) + \\ \mathsf{EF}_{\mathsf{res}}\left(\frac{350 \text{ days}}{\text{year}}\right) = \frac{\mathsf{EF}_{\mathsf{6-16}}\left(\frac{350 \text{ days}}{\text{year}}\right) \times \mathsf{ED}_{\mathsf{6-16}}\left(10 \text{ years}\right) + \mathsf{EF}_{\mathsf{16-26}}\left(\frac{350 \text{ days}}{\text{year}}\right) \times \mathsf{ED}_{\mathsf{16-26}}\left(10 \text{ years}\right)}{\mathsf{ED}_{\mathsf{0-2}}\left(2 \text{ years}\right) + \mathsf{ED}_{\mathsf{2-6}}\left(4 \text{ years}\right) + \mathsf{ED}_{\mathsf{6-16}}\left(10 \text{ years}\right) + \mathsf{ED}_{\mathsf{16-26}}\left(10 \text{ years}\right)}{\mathsf{day}} \times \mathsf{ED}_{\mathsf{2-6}}\left(4 \text{ years}\right) + \\ \mathsf{ET}_{\mathsf{res}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) = \frac{\mathsf{ET}_{\mathsf{6-16}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{\mathsf{6-16}}\left(10 \text{ years}\right) + \mathsf{ET}_{\mathsf{16-26}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{\mathsf{2-6}}\left(4 \text{ years}\right) + \\ \mathsf{ET}_{\mathsf{res}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) = \frac{\mathsf{ET}_{\mathsf{6-16}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{\mathsf{6-16}}\left(10 \text{ years}\right) + \mathsf{ET}_{\mathsf{16-26}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{\mathsf{16-26}}\left(10 \text{ years}\right)}{\mathsf{ED}_{\mathsf{0-2}}\left(2 \text{ years}\right) + \mathsf{ED}_{\mathsf{2-6}}\left(4 \text{ years}\right) + \mathsf{ET}_{\mathsf{16-26}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{\mathsf{16-26}}\left(10 \text{ years}\right)}{\mathsf{ED}_{\mathsf{16-26}}\left(10 \text{ years}\right)} + \\ \\ \mathsf{ET}_{\mathsf{res}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) = \frac{\mathsf{ET}_{\mathsf{6-16}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{\mathsf{6-16}}\left(10 \text{ years}\right) + \mathsf{ET}_{\mathsf{16-26}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{\mathsf{16-26}}\left(10 \text{ years}\right)}{\mathsf{ED}_{\mathsf{16-26}}\left(10 \text{ years}\right)} + \\ \\ \mathsf{ET}_{\mathsf{16-26}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) = \frac{\mathsf{ET}_{\mathsf{6-16}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{\mathsf{6-16}}\left(10 \text{ years}\right) + \\ \\ \mathsf{ET}_{\mathsf{6-16}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) = \frac{\mathsf{ET}_{\mathsf{6-16}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{\mathsf{6-16}}\left(10 \text{ years}\right) + \\ \\ \mathsf{ET}_{\mathsf{6-16}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) = \frac{\mathsf{ET}_{\mathsf{6-16}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{\mathsf{6-16}}\left(10 \text{ years}\right) + \\ \\ \\ \mathsf{ET}_{\mathsf{6-16}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) = \frac{\mathsf{ET}_{\mathsf{6-16}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) \times \mathsf{ET}_{\mathsf{6-16}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) + \\ \\ \\ \\ \mathsf{ET}_{\mathsf{6-16}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) = \frac{\mathsf{ET}_{\mathsf{6-16}}\left(\frac{24 \text{ hours}}{\mathsf{day}}\right) + \\ \\ \\ \\ \mathsf{ET}_{\mathsf{6-16}}\left(\frac{24 \text{ ho$$

## 4.1.3 Resident Air

This receptor spends most, if not all, of the day at home. The activities for this receptor involve typical home making chores (cooking, cleaning and laundering) as well as outdoor activities. The resident is assumed to be exposed to contaminants via the following pathway: inhalation of ambient air. This land use has no assumptions of how contaminants get into the air and the RSLs derived should be compared to air samples.

## This land use is for developing residential default screening levels that are presented in the RSL Generic Tables.

#### 4.1.3.1 Noncarcinogenic

The air land use equation, presented here, contains the following exposure routes:

• inhalation

$$SL_{res-air-nc}\left(\mu g/m^{3}\right) = \frac{THQ \times AT_{res-a}\left(\frac{365 \text{ days}}{\text{year}} \times ED_{res}(26 \text{ years})\right) \times \left(\frac{1000 \mu g}{mg}\right)}{EF_{res}\left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res}(26 \text{ years}) \times ET_{res}\left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \frac{1}{RfC\left(\frac{mg}{m3}\right)}$$

#### 4.1.3.2 Carcinogenic

The air land use equation, presented here, contains the following exposure routes:

• inhalation

$$SL_{res-air-ca}(\mu g/m^{3}) = \frac{TR \times AT_{res}\left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right)}{EF_{res}\left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res}(26 \text{ years}) \times ET_{res}\left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times IUR\left(\frac{\mu g}{m^{3}}\right)^{-1}}$$

#### 4.1.3.3 Mutagenic

The air land use equation, presented here, contains the following exposure routes:

• inhalation

$$SL_{res-air-mu} \left(\mu g/m^{3}\right) = \frac{TR \times AT_{res} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right)}{IUR \left(\frac{\mu g}{m^{3}}\right)^{-1} \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times}{\left(ED_{0-2} \left(2 \text{ years}\right) \times EF_{0-2} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{0-2} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times 10\right) +}{\left(ED_{2-6} \left(4 \text{ years}\right) \times EF_{2-6} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{2-6} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times 3\right) +}{\left(ED_{6-16} \left(10 \text{ years}\right) \times EF_{6-16} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{6-16} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times 3\right) +}{\left(ED_{16-26} \left(10 \text{ years}\right) \times EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{16-26} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times 1\right)\right)}$$

#### 4.1.3.4 Vinyl Chloride - Carcinogenic

The air land use equation, presented here, contains the following exposure routes:

• inhalation  $SL_{res-air-ca-vc}\left(\mu g/m^{3}\right) = \frac{TR}{IUR\left(\frac{\mu g}{m^{3}}\right)^{-1} + \left(\frac{IUR\left(\frac{\mu g}{m^{3}}\right)^{-1} \times EF_{res}\left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res}\left(26 \text{ years}\right) \times ET_{res}\left(\frac{24 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right)}{AT_{res}\left(\frac{365 \text{ days}}{\text{year}} \times LT\left(70 \text{ years}\right)\right)}\right)}$ 

#### 4.1.3.5 Trichloroethylene - Carcinogenic and Mutagenic

The air land use equation, presented here, contains the following exposure routes:

• inhalation

$$SL_{res-air-tce} \left(\mu g/m^{3}\right) = \frac{TR \times AT_{res} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right)}{IUR \left(\frac{\mu g}{m^{3}}\right)^{-1} \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times}{\left(ED_{res} (26 \text{ years}) \times EF_{res} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{res} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times CAF_{i} (0.756)\right) + \left(ED_{0-2} (2 \text{ years}) \times EF_{0-2} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{0-2} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times MAF_{i} (0.244) \times 10\right) + \left(ED_{2-6} (4 \text{ years}) \times EF_{2-6} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{2-6} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times MAF_{i} (0.244) \times 3\right) + \left(ED_{6-16} (10 \text{ years}) \times EF_{6-16} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{6-16} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times MAF_{i} (0.244) \times 3\right) + \left(ED_{16-26} (10 \text{ years}) \times EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{16-26} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times MAF_{i} (0.244) \times 3\right) + \left(ED_{16-26} (10 \text{ years}) \times EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{16-26} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times MAF_{i} (0.244) \times 3\right) + \left(ED_{16-26} (10 \text{ years}) \times EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{16-26} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times MAF_{i} (0.244) \times 3\right) + \left(ED_{16-26} (10 \text{ years}) \times EF_{16-26} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ET_{16-26} \left(\frac{24 \text{ hours}}{\text{day}}\right) \times MAF_{i} (0.244) \times 1\right)\right)\right)$$

(005 J

5

## **4.2** Composite Worker

## 4.2.1 Composite Worker Soil

This is a long-term receptor exposed during the work day who is a full-time employee working on-site and spends most of the workday conducting maintenance activities outdoors. The activities for this receptor (e.g., moderate digging, landscaping) typically involve on-site exposure to surface soils. The composite worker is expected to have an elevated soil ingestion rate (100 mg per day) and is assumed to be exposed to contaminants via the following pathways: incidental ingestion of soil, dermal contact with soil, inhalation of volatiles and fugitive dust. The composite worker combines the most protective exposure assumptions of the outdoor and indoor workers. The only difference between the outdoor worker and the composite worker is that the composite worker uses the more protective exposure frequency of 250 days/year from the indoor worker scenario.

# This land use is for developing industrial default screening levels that are presented in the RSL Generic Tables.

#### 4.2.1.1 Noncarcinogenic

The composite worker soil land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil

$$SL_{w-soil-nc-ing}(mg/kg) = \frac{THQ \times AT_{w-a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_w \left(25 \text{ years}\right)\right) \times BW_w \left(80 \text{ kg}\right)}{EF_w \left(250 \frac{\text{days}}{\text{year}}\right) \times ED_w \left(25 \text{ years}\right) \times \frac{RBA}{RfD_0 \left(\frac{mg}{\text{kg-day}}\right)} \times IR_w \left(100 \frac{mg}{\text{day}}\right) \times \left(\frac{10^{-6} \text{ kg}}{1 \text{ mg}}\right)}$$

• dermal exposure

$$SL_{w-soil-nc-der}(mg/kg) = \frac{THQ \times AT_{w-a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_w (25 \text{ years})\right) \times BW_w (80 \text{ kg})}{EF_w \left(250 \frac{\text{days}}{\text{year}}\right) \times ED_w (25 \text{ years}) \times \left(\frac{1}{RfD_0 \left(\frac{mg}{\text{kg-day}}\right) \times GIABS}\right) \times SA_w \left(\frac{3527 \text{ cm}^2}{\text{day}}\right) \times AF_w \left(\frac{0.12 \text{ mg}}{\text{cm}^2}\right) \times ABS_d \times \left(\frac{10^{-6} \text{ kg}}{1 \text{ mg}}\right)$$

• inhalation of volatiles and particulates emitted from soil

$$SL_{w-soil-nc-inh} (mg/kg) = \frac{THQ \times AT_{w-a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_w (25 \text{ years})\right)}{EF_w \left(250 \frac{\text{days}}{\text{year}}\right) \times ED_w (25 \text{ years}) \times ET_w \left(\frac{8 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \frac{1}{RfC \left(\frac{mg}{m3}\right)} \times \left(\frac{1}{VF_{ulim} \left(\frac{m^3}{kg}\right)} + \frac{1}{PEF \left(\frac{m^3}{kg}\right)}\right)}$$

• Total

$$SL_{w-soil-nc-tot}(mg/kg) = \frac{1}{\frac{1}{SL_{w-soil-nc-ing}} + \frac{1}{SL_{w-soil-nc-der}} + \frac{1}{SL_{w-soil-nc-inh}}}$$

### 4.2.1.2 Carcinogenic

The composite worker soil land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil

$$SL_{w-soil-ca-ing} (mg/kg) = \frac{TR \times AT_{w} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right) \times BW_{w} (80 \text{ kg})}{EF_{w} \left(250 \frac{\text{days}}{\text{year}}\right) \times ED_{w} (25 \text{ years}) \times CSF_{o} \left(\frac{mg}{\text{kg-day}}\right)^{-1} \times RBA \times IR_{w} \left(100 \frac{mg}{\text{day}}\right) \times \left(\frac{10^{-6} \text{ kg}}{1 \text{ mg}}\right)}$$

• dermal exposure

$$SL_{w-soil-ca-der}(mg/kg) = \frac{TR \times AT_{w} \left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right) \times BW_{w}(80 \text{ kg})}{EF_{w} \left(250 \frac{\text{days}}{\text{year}}\right) \times ED_{w}(25 \text{ years}) \times \left(\frac{CSF_{0} \left(\frac{mg}{\text{kg-day}}\right)^{-1}}{\text{GIABS}}\right) \times SA_{w} \left(\frac{3527 \text{ cm}^{2}}{\text{day}}\right) \times AF_{w} \left(\frac{0.12 \text{ mg}}{\text{cm}^{2}}\right) \times ABS_{d} \times \left(\frac{10^{-6} \text{ kg}}{1 \text{ mg}}\right)$$

• inhalation of volatiles and particulates emitted from soil

$$SL_{w-soil-ca-inh}(mg/kg) = \frac{TR \times AT_{w} \left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right)}{EF_{w} \left(250 \frac{\text{days}}{\text{year}}\right) \times ED_{w}(25 \text{ years}) \times ET_{w} \left(\frac{8 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times IUR \left(\frac{\mu g}{m3}\right)^{-1} \times \left(\frac{1000 \mu g}{mg}\right) \times \left(\frac{1}{VF_{ulim} \left(\frac{m3}{kg}\right)} + \frac{1}{PEF\left(\frac{m3}{kg}\right)}\right)$$

$$SL_{w-soil-ca-tot} (mg/kg) = \frac{1}{\frac{1}{SL_{w-soil-ca-ing}} + \frac{1}{SL_{w-soil-ca-der}} + \frac{1}{SL_{w-soil-ca-inh}}}$$
  
Total

## 4.2.2 Composite Worker Air

This is a long-term receptor exposed during the work day who is a full-time employee working on-site and spends most of the workday conducting maintenance activities indoors. The composite worker is assumed to be exposed to contaminants via the following pathway: inhalation of ambient air. The composite worker combines the most protective exposure assumptions of the outdoor and indoor workers. The only difference between the outdoor worker and the composite worker is that the composite worker uses the more protective exposure frequency of 250 days/year from the indoor worker scenario. This land use has no assumptions of how contaminants get into the air and the RSLs derived should be compared to air samples.

## This land use is for developing industrial default screening levels that are presented in the RSL Generic Tables.

#### 4.2.2.1 Noncarcinogenic

The air land use equation, presented here, contains the following exposure routes:

• Inhalation

$$SL_{w-air-nc}\left(\mu g/m^{3}\right) = \frac{THQ \times AT_{w}\left(\frac{365 \text{ days}}{\text{year}} \times ED_{w}\left(25 \text{ years}\right)\right) \times \left(\frac{1000 \ \mu g}{\text{mg}}\right)}{EF_{w}\left(\frac{250 \ \text{days}}{\text{year}}\right) \times ED_{w}\left(25 \text{ years}\right) \times ET_{w}\left(\frac{8 \ \text{hours}}{\text{day}}\right) \times \left(\frac{1 \ \text{day}}{24 \ \text{hours}}\right) \times \frac{1}{RfC\binom{mg}{m^{3}}}$$

#### 4.2.2.2 Carcinogenic

The air land use equation, presented here, contains the following exposure routes:

• Inhalation

$$SL_{w-air-ca}\left(\mu g/m^{3}\right) = \frac{TR \times AT_{w}\left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right)}{EF_{w}\left(\frac{250 \text{ days}}{\text{year}}\right) \times ED_{w}(25 \text{ years}) \times ET_{w}\left(\frac{8 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times IUR\left(\frac{\mu g}{m^{3}}\right)^{-1}}$$

#### 4.3 Outdoor Worker

#### 4.3.1 Outdoor Worker Soil

This is a long-term receptor exposed during the work day who is a full-time employee working on-site and spends most of the workday conducting maintenance activities outdoors. The activities for this receptor (e.g., moderate digging, landscaping) typically involve on-site exposure to surface soils. The outdoor worker is expected to have an elevated soil ingestion rate (100 mg per day) and is assumed to be exposed to contaminants via the following pathways: incidental ingestion of soil, dermal contact with soil, inhalation of volatiles and fugitive dust. The outdoor worker receives more exposure than the indoor worker under commercial/industrial conditions.

## The outdoor worker soil land use is not provided in the RSL Generic Tables but RSLs can be created by using the Calculator.

## 4.3.1.1 Noncarcinogenic

The outdoor worker soil land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil

$$SL_{ow-soil-nc-ing}(mg/kg) = \frac{THQ \times AT_{ow-a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{ow} (25 \text{ years})\right) \times BW_{ow} (80 \text{ kg})}{EF_{ow} \left(225 \frac{\text{days}}{\text{year}}\right) \times ED_{ow} (25 \text{ years}) \times \frac{RBA}{RfD_{o} \left(\frac{mg}{\text{kg-day}}\right)} \times IR_{ow} \left(100 \frac{mg}{\text{day}}\right) \times \left(\frac{10^{-6} \text{ kg}}{1 \text{ mg}}\right)}$$

• dermal exposure

$$SL_{ow-soil-nc-der}(mg/kg) = \frac{THQ \times AT_{oW-a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{oW} \left(25 \text{ years}\right)\right) \times BW_{oW}(80 \text{ kg})}{EF_{oW} \left(225 \frac{\text{days}}{\text{year}}\right) \times ED_{oW} \left(25 \text{ years}\right) \times \left(\frac{1}{RfD_{o} \left(\frac{mg}{\text{kg-day}}\right) \times GIABS}\right) \times SA_{oW} \left(\frac{3527 \text{ cm}^{2}}{\text{day}}\right) \times AF_{oW} \left(\frac{0.12 \text{ mg}}{\text{cm}^{2}}\right) \times ABS_{d} \times \left(\frac{10^{-6} \text{ kg}}{1 \text{ mg}}\right) \times GIABS}$$

• inhalation of volatiles and particulates emitted from soil

$$SL_{ow-soil-nc-inh} (mg/kg) = \frac{THQ \times AT_{ow-a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{ow} (25 \text{ years})\right)}{EF_{ow} \left(225 \frac{\text{days}}{\text{year}}\right) \times ED_{ow} (25 \text{ years}) \times ET_{ow} \left(\frac{8 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \frac{1}{RfC \left(\frac{mg}{m3}\right)} \times \left(\frac{1}{VF_{ulim} \left(\frac{m^3}{kg}\right)} + \frac{1}{PEF \left(\frac{m^3}{kg}\right)}\right)}$$

• Total

$$SL_{ow-soil-nc-tot}(mg/kg) = \frac{1}{\frac{1}{SL_{ow-soil-nc-ing}} + \frac{1}{SL_{ow-soil-nc-der}} + \frac{1}{SL_{ow-soil-nc-inh}}}$$

### 4.3.1.2 Carcinogenic

The outdoor worker soil land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil,

$$SL_{ow-soil-ca-ing}(mg/kg) = \frac{TR \times AT_{ow}\left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right) \times BW_{ow}(80 \text{ kg})}{EF_{ow}\left(225 \frac{\text{days}}{\text{year}}\right) \times ED_{ow}(25 \text{ years}) \times CSF_{o}\left(\frac{mg}{\text{kg-day}}\right)^{-1} \times RBA \times IR_{ow}\left(100 \frac{mg}{\text{day}}\right) \times \left(\frac{10^{-6} \text{ kg}}{1 \text{ mg}}\right)}$$

• dermal exposure,

$$SL_{ow-soil-ca-der}(mg/kg) = \frac{TR \times AT_{ow}\left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right) \times BW_{ow}(80 \text{ kg})}{EF_{ow}\left(225 \frac{\text{days}}{\text{year}}\right) \times ED_{ow}(25 \text{ years}) \times \left(\frac{CSF_{o}\left(\frac{mg}{\text{kg-day}}\right)^{-1}}{GIABS}\right) \times SA_{ow}\left(\frac{3527 \text{ cm}^{2}}{\text{day}}\right) \times AF_{ow}\left(\frac{0.12 \text{ mg}}{\text{cm}^{2}}\right) \times ABS_{d} \times \left(\frac{10^{-6} \text{ kg}}{1 \text{ mg}}\right)$$

• inhalation of volatiles and particulates emitted from soil,



#### 4.3.2 Outdoor Worker Air

This is a long-term receptor exposed during the work day who is a full-time employee working on-site and spends most of the workday conducting maintenance activities outdoors. The outdoor worker is assumed to be exposed to contaminants via the following pathway: inhalation of ambient air. This land use has no assumptions of how contaminants get into the air and the RSLs derived should be compared to air samples.

The outdoor worker air land use is not provided in the RSL Generic Tables but RSLs can be created by using the Calculator.

#### 4.3.2.1 Noncarcinogenic

The air land use equation, presented here, contains the following exposure routes:

• Inhalation

$$SL_{ow-air-nc}\left(\mu g/m^{3}\right) = \frac{THQ \times AT_{ow}\left(\frac{365 \text{ days}}{\text{year}} \times ED_{ow}\left(25 \text{ years}\right)\right) \times \left(\frac{1000 \ \mu g}{\text{mg}}\right)}{EF_{ow}\left(\frac{225 \ \text{days}}{\text{year}}\right) \times ED_{ow}\left(25 \text{ years}\right) \times ET_{ow}\left(\frac{8 \ \text{hours}}{\text{day}}\right) \times \left(\frac{1 \ \text{day}}{24 \ \text{hours}}\right) \times \frac{1}{RfC}\left(\frac{mg}{m^{3}}\right)}$$

#### 4.3.2.2 Carcinogenic

The air land use equation, presented here, contains the following exposure routes:

Inhalation

$$SL_{ow-air-ca}\left(\mu g/m^{3}\right) = \frac{TR \times AT_{ow}\left(\frac{365 \text{ days}}{\text{year}} \times LT\left(70 \text{ years}\right)\right)}{EF_{ow}\left(\frac{225 \text{ days}}{\text{year}}\right) \times ED_{ow}\left(25 \text{ years}\right) \times ET_{ow}\left(\frac{8 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times IUR\left(\frac{\mu g}{m^{3}}\right)^{-1}}$$

## 4.4 Indoor Worker

## 4.4.1 Indoor Worker Soil

This receptor spends most, if not all, of the workday indoors. Thus, an indoor worker has no direct dermal contact with outdoor soils. This worker may, however, be exposed to contaminants through ingestion of contaminated soils that have been incorporated into indoor dust and inhalation of volatiles and particulates from outside soils. RSLs calculated for this receptor are expected to be protective of both workers engaged in low intensity activities such as office work and those engaged in more strenuous activity (e.g., factory or warehouse workers).

The indoor worker soil land use is not provided in the Generic Tables but RSLs can be created by using the Calculator.

#### 4.4.1.1 Noncarcinogenic

The indoor worker soil land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil

$$SL_{iw-soil-nc-ing}(mg/kg) = \frac{THQ \times AT_{iw-a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{iw} \left(25 \text{ years}\right)\right) \times BW_{iw}(80 \text{ kg})}{EF_{iw} \left(250 \frac{\text{days}}{\text{year}}\right) \times ED_{iw}(25 \text{ years}) \times \frac{RBA}{RfD_0 \left(\frac{mg}{\text{kg-day}}\right)} \times IR_{iw} \left(50 \frac{mg}{\text{day}}\right) \times \left(\frac{10^{-6} \text{ kg}}{1 \text{ mg}}\right)}$$

• inhalation of volatiles and particulates emitted from soil

$$SL_{iw-soil-nc-inh} (mg/kg) = \frac{THQ \times AT_{iw-a} \left(\frac{305 \text{ days}}{\text{year}} \times ED_{iw} (25 \text{ years})\right)}{EF_{iw} \left(250 \frac{\text{days}}{\text{year}}\right) \times ED_{iw} (25 \text{ years}) \times ET_{iw}} \left(\frac{8 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \frac{1}{RfC \left(\frac{mg}{m3}\right)} \times \left(\frac{1}{VF_{ulim} \left(\frac{m^3}{kg}\right)} + \frac{1}{PEF \left(\frac{m^3}{kg}\right)}\right)}$$

• Total  $SL_{iw-soil-nc-tot} (mg/kg) = \frac{1}{\frac{1}{SL_{iw-soil-nc-ing}} + \frac{1}{SL_{iw-soil-nc-inh}}}$ 

#### 4.4.1.2 Carcinogenic

The indoor worker soil land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil

$$SL_{iw-soil-ca-ing}(mg/kg) = \frac{TR \times AT_{iw} \left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right) \times BW_{iw}(80 \text{ kg})}{EF_{iw} \left(250 \frac{\text{days}}{\text{year}}\right) \times ED_{iw}(25 \text{ years}) \times CSF_{o} \left(\frac{mg}{\text{kg-day}}\right)^{-1} \times RBA \times IR_{iw} \left(50 \frac{mg}{\text{day}}\right) \times \left(\frac{10^{-6} \text{ kg}}{1 \text{ mg}}\right)}$$

• inhalation of volatiles and particulates emitted from soil



#### 4.4.2 Indoor Worker Air

This is a long-term receptor exposed during the work day who is a full-time employee working on-site and spends most of the workday conducting maintenance activities indoors. The indoor worker is assumed to be exposed to contaminants via the following pathway: inhalation of ambient air. This land use has no assumptions of how contaminants get into the air and the RSLs derived should be compared to air samples.

The indoor worker air land use is not provided in the Generic Tables but RSLs can be created by using the Calculator.

#### 4.4.2.1 Noncarcinogenic

The air land use equation, presented here, contains the following exposure routes:

• Inhalation

$$SL_{iw-air-nc}\left(\mu g/m^{3}\right) = \frac{THQ \times AT_{iw}\left(\frac{365 \text{ days}}{\text{year}} \times ED_{iw}\left(25 \text{ years}\right)\right) \times \left(\frac{1000 \ \mu g}{\text{mg}}\right)}{EF_{iw}\left(\frac{250 \ \text{days}}{\text{year}}\right) \times ED_{iw}\left(25 \ \text{years}\right) \times ET_{iw}\left(\frac{8 \ \text{hours}}{\text{day}}\right) \times \left(\frac{1 \ \text{day}}{24 \ \text{hours}}\right) \times \frac{1}{RfC}\left(\frac{mg}{m^{3}}\right)}$$

#### 4.4.2.2 Carcinogenic

The air land use equation, presented here, contains the following exposure routes:

• Inhalation

$$SL_{iw-air-ca}\left(\mu g/m^{3}\right) = \frac{TR \times AT_{iw}\left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right)}{EF_{iw}\left(\frac{250 \text{ days}}{\text{year}}\right) \times ED_{iw}\left(25 \text{ years}\right) \times ET_{iw}\left(\frac{8 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times IUR\left(\frac{\mu g}{m^{3}}\right)^{-1}}$$

#### 4.5 Construction Worker

An assessment for the construction worker scenario is described in more detail in the Supplemental Soil Screening Guidance (SSSG, EPA, 2002). Despite the exposure duration of one year, carcinogenic risk is averaged over an assumed lifetime of 70 years, consistent with the assumption that the risk of developing cancer continues even after exposure has stopped. EPA guidance states that the averaging time for noncancer is to be set at the same length as exposure duration, even if the exposure duration is less than one year. For noncancer, the averaging time can be changed to be less than a year by changing the number of weeks worked (EW). Further, the examples given in the SSSG show that the time of traffic (Tt) is equivalent to EF and time of construction ( $T_c$ ) is the averaging time (length of project).

The particulate emission factor (PEF) and volatilization factor (VF) equations used are unique to this scenario. See Section 4.9 for further information on subchronic VFs and PEFs. The PEFs calculated in these scenarios may predict much higher air concentrations than the standard wind-driven PEFs; however, the inhalation screening level will likely be dominated by the VF in the case of a volatile contaminant. VFs are commonly 5 orders of magnitude more protective than PEFs. Additionally, the ingestion route typically is the driving factor in most RSL calculations. Two types of mechanical soil disturbance are addressed: standard vehicle traffic (unpaved) and other construction activities (wind, grading, dozing, tilling and excavating). In general, the intake and contact rates are all greater than the outdoor worker. Exhibit 5-1 in the supplemental soil screening guidance presents the exposure parameters

# The construction worker soil land use is not provided in the Generic Tables but RSLs can be created by using the Calculator.

## 4.5.1 Construction Worker Soil Exposure to Standard Vehicle Traffic

This is a short-term receptor exposed during the work day working around vehicles suspending dust in the air. The activities for this receptor (e.g., trenching, excavating) typically involve on-site exposure to surface soils. The construction worker is expected to have an elevated soil ingestion rate (330 mg per day) and is assumed to be exposed to contaminants via the following pathways: incidental ingestion of soil, dermal contact with contaminants in soil, inhalation of volatiles and fugitive dust. The only difference between this construction worker and the one described in section 4.5.2 is that this construction worker uses a different PEF. The construction worker soil land use is not provided in the Generic Tables but RSLs can be created by using the Calculator. The construction land use is described in the supplemental soil screening guidance. This land use is limited to an exposure duration of 1 year and is thus, subchronic. Other unique aspects of this scenario are that the PEF is based on mechanical disturbance of the soil. Two types of mechanical soil disturbance are addressed: standard vehicle traffic and other than standard vehicle traffic (e.g. wind, grading, dozing, tilling and excavating). In general, the intakes and contact rates are all greater than the outdoor worker. Exhibit 5-1 in the supplemental soil screening guidance presents the exposure parameters.

#### 4.5.1.1 Noncarcinogenic

The construction worker soil land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil

$$SL_{cw-soil-nc-ing} (mg/kg) = \frac{THQ \times AT_{cw-a} \left( EW_{cw} \frac{50 \text{ weeks}}{\text{year}} \times \frac{50 \text{ weeks}}{\text{year}} \times \frac{7 \text{ days}}{\text{week}} \times ED_{cw} (1 \text{ year}) \right) \times BW_{cw} (80 \text{ kg})}{EF_{cw} \left( EW_{cw} \frac{50 \text{ weeks}}{\text{year}} \times DW_{cw} \frac{5 \text{ days}}{\text{week}} \right) \times ED_{cw} (1 \text{ year}) \times \frac{RBA}{RfD_0 \left( \frac{mg}{\text{kg-day}} \right)} \times IR_{cw} \left( 330 \frac{mg}{\text{day}} \right) \times \left( \frac{10^{-6} \text{ kg}}{1 \text{ mg}} \right)}$$

• dermal exposure



#### 4.5.1.2 Carcinogenic

The construction worker soil land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil,

$$SL_{cw-soil-ca-ing}(mg/kg) = \frac{TR \times AT_{cw} \left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right) \times BW_{cw}(80 \text{ kg})}{EF_{cw} \left(EW_{cw} \frac{50 \text{ weeks}}{\text{year}} \times DW_{cw} \frac{5 \text{ days}}{\text{week}}\right) \times ED_{cw}(1 \text{ year}) \times CSF_{o} \left(\frac{mg}{\text{kg-day}}\right)^{-1} \times RBA \times IR_{cw} \left(330 \frac{mg}{\text{day}}\right) \times \left(\frac{10^{-6} \text{ kg}}{1 \text{ mg}}\right)$$

· dermal exposure,

$$SL_{cw-soil-ca-der}(mg/kg) = \frac{TR \times AT_{cw} \left(\frac{305 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right) \times BW_{cw}(80 \text{ kg})}{EF_{cw} \left(EW_{cw} \frac{50 \text{ weeks}}{\text{year}} \times DW_{cw} \frac{5 \text{ days}}{\text{week}}\right) \times ED_{cw}(1 \text{ year}) \times \left(\frac{CSF_0 \left(\frac{mg}{\text{kg-day}}\right)^{-1}}{GIABS}\right) \times SA_{cw} \left(\frac{3527 \text{ cm}^2}{\text{day}}\right) \times AF_{cw} \left(\frac{0.3 \text{ mg}}{\text{cm}^2}\right) \times ABS_d \times \left(\frac{10^{-6} \text{ kg}}{1 \text{ mg}}\right)$$

• inhalation of volatiles and particulates emitted from soil,

$$SL_{cw-soil-ca-inh}(mg/kg) = \frac{TR \times AT_{CW}\left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right)}{EF_{cw}\left(EW_{cw}\frac{50 \text{ weeks}}{\text{year}} \times DW_{cw}\frac{5 \text{ days}}{\text{week}}\right) \times ED_{cw}(1 \text{ year}) \times ET_{cw}\left(\frac{8 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times IUR\left(\frac{\mu g}{m3}\right)^{-1} \times \left(\frac{1000 \mu g}{mg}\right) \times \left(\frac{1}{VF_{ulim-sc}\left(\frac{m^3}{kg}\right)} + \frac{1}{PEF_{sc}\left(\frac{m^3}{kg}\right)}\right)$$

• Total.



## 4.5.2 Construction Worker Soil Exposure to Other Construction Activities

This is a short-term receptor exposed during the work day working around heavy vehicles suspending dust in the air. The activities for this receptor (e.g., dozing, grading, tilling, dumping, and excavating) typically involve on-site exposure to surface soils. The construction worker is expected to have an elevated soil ingestion rate (330 mg per day) and is assumed to be exposed to contaminants via the following pathways: incidental ingestion of soil, dermal contact with contaminants in soil, inhalation of volatiles and fugitive dust. The only difference between this construction worker and the one described in section 4.5.1 is that this construction worker uses a different PEF. The construction worker soil land use is not provided in the Generic Tables but RSLs can be created by using the Calculator. The construction land use is described in the supplemental soil screening guidance. This land use is limited to an exposure duration of 1 year and is thus, subchronic. Other unique aspects of this scenario are that the PEF is based on mechanical disturbance of the soil. Two types of mechanical soil disturbance are addressed: standard vehicle traffic (e.g. wind, grading, dozing, tilling and excavating). In general, the intakes and contact rates are all greater than the outdoor worker. Exhibit 5-1 in the supplemental soil screening guidance presents the exposure parameters.

#### 4.5.2.1 Noncarcinogenic

The construction worker soil land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil

$$SL_{cw-soil-nc-ing} (mg/kg) = \frac{THQ \times AT_{cw-a} \left(EW_{cw} \frac{50 \text{ weeks}}{\text{year}} \times \frac{7 \text{ days}}{\text{week}} \times ED_{cw} (1 \text{ year})\right) \times BW_{cw} (80 \text{ kg})}{EF_{cw} \left(EW_{cw} \frac{50 \text{ weeks}}{\text{year}} \times DW_{cw} \frac{5 \text{ days}}{\text{week}}\right) \times ED_{cw} (1 \text{ year}) \times \frac{RBA}{RfD_0 \left(\frac{mg}{kg \cdot day}\right)} \times IR_{cw} \left(330 \frac{mg}{day}\right) \times \left(\frac{10^{-6} \text{ kg}}{1 \text{ mg}}\right)}$$
  
dermal exposure
$$\frac{THQ \times AT_{cw-a} \left(EW_{cw} \frac{50 \text{ weeks}}{\text{year}} \times \frac{7 \text{ days}}{\text{week}} \times ED_{cw} (1 \text{ year})\right) \times BW_{cw} (80 \text{ kg})}{\frac{10^{-6} \text{ kg}}{1 \text{ mg}}}$$

$$\mathsf{EF}_{\mathsf{CW}}\left(\mathsf{EW}_{\mathsf{CW}}\frac{50\,\mathsf{weeks}}{\mathsf{year}}\times\mathsf{DW}_{\mathsf{CW}}\frac{5\,\mathsf{days}}{\mathsf{week}}\right)\times\mathsf{ED}_{\mathsf{CW}}\left(1\,\mathsf{year}\right)\times\left(\frac{1}{\mathsf{RfD}_0\left(\frac{\mathsf{mg}}{\mathsf{kg}\cdot\mathsf{day}}\right)}\times\mathsf{GIABS}\right)\times\mathsf{SA}_{\mathsf{CW}}\left(\frac{3527\,\mathsf{cm}^2}{\mathsf{day}}\right)\times\mathsf{AF}_{\mathsf{CW}}\left(\frac{0.3\,\mathsf{mg}}{\mathsf{cm}^2}\right)\times\mathsf{ABS}_{\mathsf{d}}\times\left(\frac{10^{-6}\,\mathsf{kg}}{1\,\mathsf{mg}}\right)$$

• inhalation of volatiles and particulates emitted from soil

$$SL_{cw-soil-nc-inh}(mg/kg) = \frac{THQ \times AT_{CW-a}\left(EW_{CW} \frac{50 \text{ weeks}}{\text{ year}} \times \frac{7 \text{ days}}{\text{ week}} \times ED_{CW}(1 \text{ year})\right)}{EF_{CW}\left(EW_{CW} \frac{50 \text{ weeks}}{\text{ year}} \times DW_{CW} \frac{5 \text{ days}}{\text{ week}}\right) \times ED_{CW}(1 \text{ year}) \times ET_{CW} \left(\frac{8 \text{ hours}}{\text{ day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \frac{1}{RfC\left(\frac{mg}{m}\right)} \times \left(\frac{1}{VF_{ulim-sc}\left(\frac{m^3}{kg}\right)} + \frac{1}{PEF_{sc}\left(\frac{m^3}{kg}\right)}\right)}$$

$$Total$$

$$SL_{cw-soil-nc-tot} (mg/kg) = \frac{1}{\frac{1}{SL_{cw-soil-nc-ing}} + \frac{1}{SL_{cw-soil-nc-der}} + \frac{1}{SL_{cw-soil-nc-inh}}}$$

#### 4.5.2.2 Carcinogenic

•

The construction worker soil land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil



## 4.6 Recreator

## 4.6.1 Recreator Soil or Sediment

This receptor spends time outside involved in recreational activities. The recreator is assumed to be exposed to contaminants via the following pathways: incidental ingestion of soil, dermal contact with contaminants in soil, and inhalation of volatiles and fugitive dust. There are no default RSLs for this scenario; only site-specific.

# The recreator soil land use is not provided in the Generic Tables but RSLs can be created by using the Calculator.

#### 4.6.1.1 Noncarcinogenic - Child

The recreator soil or sediment land use equation, presented here, contains the following exposure routes:

· incidental ingestion of soil or sediment

$$SL_{rec-soil-nc-ing-c} (mg/kg) = \frac{THQ \times AT_{rec-c} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{rec-c} (\text{years})\right) \times BW_{rec-c} (15 \text{ kg})}{EF_{rec-c} \left(\frac{\text{days}}{\text{year}}\right) \times ED_{rec-c} (\text{years}) \times \frac{RBA}{RfD_0 \left(\frac{mg}{\text{kg-day}}\right)} \times IRS_{rec-c} \left(\frac{200 \text{ mg}}{\text{day}}\right) \times \frac{10^{-6} \text{kg}}{1 \text{ mg}}}$$

• dermal contact with soil or sediment

$$SL_{rec-soil-nc-der-c}(mg/kg) = \frac{THQ \times AT_{rec-c}\left(\frac{365 \text{ days}}{\text{year}} \times ED_{rec-c}(\text{years})\right) \times BW_{rec-c}(15 \text{ kg})}{EF_{rec-c}\left(\frac{days}{\text{year}}\right) \times ED_{rec-c}(\text{years}) \times \frac{1}{\left(RfD_{o}\left(\frac{mg}{\text{kg-day}}\right) \times GIABS\right)} \times SA_{rec-c}\left(\frac{2373 \text{ cm}^{2}}{\text{day}}\right) \times AF_{rec-c}\left(\frac{0.2 \text{ mg}}{\text{cm}^{2}}\right) \times ABS_{d} \times \frac{10^{-6} \text{ kg}}{1 \text{ mg}}}{1 \text{ mg}}$$

inhalation of volatiles and particulates emitted from soil or sediment
 (365 days)

$$SL_{rec-soil-nc-inh-c}(mg/kg) = \frac{THQ \times AT_{rec-c}\left(\frac{365 \text{ days}}{\text{year}} \times ED_{rec-c}\left(6 \text{ years}\right)\right)}{EF_{rec-c}\left(\frac{days}{year}\right) \times ED_{rec-c}\left(6 \text{ years}\right) \times ET_{rec-c}\left(\frac{hours}{day}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \frac{1}{RfC\left(\frac{mg}{m3}\right)} \times \left(\frac{1}{VF_{ulim}\left(\frac{m^3}{kg}\right)} + \frac{1}{PEF\left(\frac{m^3}{kg}\right)}\right)}$$

• Total  $SL_{rec-soil-nc-tot-c} (mg/kg) = \frac{1}{\frac{1}{SL_{rec-soil-nc-ing}} + \frac{1}{SL_{rec-soil-nc-der}} + \frac{1}{SL_{rec-soil-nc-inh}}}$ 

#### 4.6.1.2 Noncarcinogenic - Adult

The recreator soil or sediment land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil or sediment

$$SL_{rec-soil-nc-ing-a} (mg/kg) = \frac{THQ \times AT_{rec-a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{rec-a} (\text{years})\right) \times BW_{rec-a} (80 \text{ kg})}{EF_{rec-a} \left(\frac{\text{days}}{\text{year}}\right) \times ED_{rec-a} (\text{years}) \times \frac{RBA}{RfD_0 \left(\frac{mg}{\text{kg-day}}\right)} \times IRS_{rec-a} \left(\frac{100 \text{ mg}}{\text{day}}\right) \times \frac{10^{-6} \text{kg}}{1 \text{ mg}}}$$

• dermal contact with soil or sediment

$$SL_{rec-soil-nc-der-a} (mg/kg) = \frac{THQ \times AT_{rec-a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{rec-a} (\text{years})\right) \times BW_{rec-a} (80 \text{ kg})}{EF_{rec-a} \left(\frac{days}{\text{year}}\right) \times ED_{rec-a} (\text{years}) \times \frac{1}{\left(RtD_{0} \left(\frac{mg}{\text{kg-day}}\right) \times GIABS\right)} \times SA_{rec-a} \left(\frac{6032 \text{ cm}^{2}}{\text{day}}\right) \times AF_{rec-a} \left(\frac{0.07 \text{ mg}}{\text{cm}^{2}}\right) \times ABS_{d} \times \frac{10^{-6} \text{kg}}{1 \text{ mg}}}{1 \text{ mg}}$$

• inhalation of volatiles and particulates emitted from soil or sediment

$$SL_{rec-soil-nc-inh-a} (mg/kg) = \frac{THQ \times AT_{rec-a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{rec-a} (20 \text{ years})\right)}{EF_{rec-a} \left(\frac{\text{days}}{\text{year}}\right) \times ED_{rec-a} (20 \text{ years}) \times ET_{rec-a} \left(\frac{\text{hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \frac{1}{RfC \left(\frac{mg}{m3}\right)} \times \left(\frac{1}{VF_{ulim} \left(\frac{m^3}{kg}\right)} + \frac{1}{PEF \left(\frac{m^3}{kg}\right)}\right)}$$

Total

$$SL_{rec-soil-nc-tot-a} (mg/kg) = \frac{1}{\frac{1}{\frac{1}{SL_{rec-soil-nc-ing-a}} + \frac{1}{\frac{1}{SL_{rec-soil-nc-der-a}} + \frac{1}{\frac{1}{SL_{rec-soil-nc-inh-a}}}}}$$

## 4.6.1.3 Carcinogenic

The recreator soil or sediment land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil or sediment

$$SL_{rec-soil-ca-ing}(mg/kg) = \frac{TR \times AT_{rec} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right)}{CSF_0 \left(\frac{mg}{\text{kg-day}}\right)^{-1} \times RBA \times IFS_{rec-adj} \left(\frac{mg}{\text{kg}}\right) \times \left(\frac{10^{-6} \text{kg}}{mg}\right)}$$

where:

$$\mathsf{IFS}_{\mathsf{rec-adj}}\left(\frac{\mathsf{mg}}{\mathsf{kg}}\right) = \left(\frac{\mathsf{ED}_{\mathsf{rec-c}}\left(\mathsf{years}\right) \times \mathsf{EF}_{\mathsf{rec-c}}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{IRS}_{\mathsf{rec-c}}\left(\frac{200\ \mathsf{mg}}{\mathsf{day}}\right)}{\mathsf{BW}_{\mathsf{rec-c}}\left(15\ \mathsf{kg}\right)} + \frac{\mathsf{ED}_{\mathsf{rec-a}}\left(\mathsf{years}\right) \times \mathsf{EF}_{\mathsf{rec-a}}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{IRS}_{\mathsf{rec-a}}\left(\frac{100\ \mathsf{mg}}{\mathsf{day}}\right)}{\mathsf{BW}_{\mathsf{rec-a}}\left(80\ \mathsf{kg}\right)}\right)$$

• dermal contact with soil or sediment

$$SL_{rec-soil-ca-der}(mg/kg) = \frac{TR \times AT_{rec} \left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right)}{\left(\frac{CSF_{o} \left(\frac{mg}{\text{kg-day}}\right)^{-1}}{\text{GIABS}}\right)} \times DFS_{rec-adj} \left(\frac{mg}{\text{Kg}}\right) \times ABS_{d} \times \left(\frac{10^{-6} \text{kg}}{\text{mg}}\right)}{\text{mg}}$$

where:

$$DFS_{rec-adj}\left(\frac{mg}{kg}\right) = \left( \frac{ED_{rec-c}\left(years\right) \times EF_{rec-c}\left(\frac{days}{year}\right) \times SA_{rec-c}\left(\frac{2373 \text{ cm}^2}{day}\right) \times AF_{rec-c}\left(\frac{0.2mg}{cm^2}\right)}{BW_{rec-c}\left(15 \text{ kg}\right)} + \frac{ED_{rec-a}\left(years\right) \times EF_{rec-a}\left(\frac{days}{year}\right) \times SA_{rec-a}\left(\frac{6032 \text{ cm}^2}{day}\right) \times AF_{rec-a}\left(\frac{0.07 \text{ mg}}{cm^2}\right)}{BW_{rec-a}\left(80 \text{ kg}\right)} \right)$$

• inhalation of volatiles and particulates emitted from soil or sediment

$$SL_{rec-soil-ca-inh} (mg/kg) = \frac{TR \times AT_{rec} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right)}{IUR \left(\frac{\mu g}{mg}\right)^{-1} \times \left(\frac{1000 \ \mu g}{mg}\right) \times EF_{rec} \left(\frac{\text{days}}{\text{year}}\right) \times \left(\frac{1}{VF_{ulim} \left(\frac{m^3}{kg}\right)} + \frac{1}{PEF \left(\frac{m^3}{kg}\right)}\right) \times ED_{rec} (26 \text{ years}) \times ET_{rec} \left(\frac{\text{hours}}{\text{day}}\right) \times \left(\frac{1 \ \text{day}}{24 \ \text{hours}}\right)$$

• Total

$$SL_{rec-soil-ca-tot}(mg/kg) = \frac{1}{\frac{1}{SL_{rec-soil-ca-ing}} + \frac{1}{SL_{rec-soil-ca-der}} + \frac{1}{SL_{rec-soil-ca-inh}}}$$

## 4.6.1.4 Mutagenic

The recreator soil or sediment land use equation, presented here, contains the following exposure routes:

• incidental ingestion of soil or sediment,

$$SL_{rec-soil-mu-ing} (mg/kg) = \frac{TR \times AT_{rec} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right)}{CSF_{0} \left(\frac{mg}{\text{kg-day}}\right)^{-1} \times RBA \times IFSM_{rec-adj} \left(\frac{mg}{\text{kg}}\right) \times \left(\frac{10^{-6} \text{kg}}{\text{mg}}\right)}{\text{where:}}$$

$$\text{where:} = \frac{\left(\frac{ED_{0-2} (\text{years}) \times EF_{0-2} \left(\frac{\text{days}}{\text{year}}\right) \times IRS_{0-2} \left(\frac{200 \text{ mg}}{\text{day}}\right) \times 10}{BW_{0-2} (15 \text{ kg})} + \frac{ED_{2-6} (\text{years}) \times EF_{2-6} \left(\frac{\text{days}}{\text{year}}\right) \times IRS_{2-6} \left(\frac{200 \text{ mg}}{\text{day}}\right) \times 3}{BW_{2-6} (15 \text{ kg})} + \frac{ED_{6-16} (\text{years}) \times EF_{6-16} \left(\frac{\text{days}}{\text{year}}\right) \times IRS_{6-16} \left(\frac{100 \text{ mg}}{\text{day}}\right) \times 3}{BW_{6-16} (80 \text{ kg})} + \frac{ED_{16-26} (\text{years}) \times EF_{16-26} \left(\frac{\text{days}}{\text{year}}\right) \times IRS_{16-26} \left(\frac{100 \text{ mg}}{\text{day}}\right) \times 1}{BW_{16-26} (80 \text{ kg})}$$

• dermal contact with soil or sediment

• inhalation of volatiles and particulates emitted from soil or sediment

$$SL_{rec-soil-mu-inh}(mg/kg) = \frac{TR \times AT_{rec} \left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right)}{IUR \left(\frac{\mu g}{m^3}\right)^{-1} \times \left(\frac{1}{VF_{ulim} \left(\frac{m^3}{kg}\right)^{+} + \frac{1}{PEF \left(\frac{m^3}{kg}\right)}\right) \times \left(\frac{1000 \ \mu g}{mg}\right) \times \left(\frac{100$$

• Total



### 4.6.1.5 Vinyl Chloride - Carcinogenic

The recreator soil or sediment land use equations, presented here, contain the following exposure routes:

• incidental ingestion of soil or sediment

$$SL_{rec-soil-ca-vc-ing}(mg/kg) = \frac{TR}{\left(\frac{CSF_{o}\left(\frac{mg}{kg-day}\right)^{-1} \times RBA \times IFS_{rec-adj}\left(\frac{mg}{kg}\right) \times \frac{10^{-6}kg}{1 mg}}{AT_{rec}\left(\frac{365 \text{ days}}{\text{ year}} \times LT (70 \text{ years})\right)}\right)^{+}}$$
$$\frac{\left(\frac{CSF_{o}\left(\frac{mg}{kg-day}\right)^{-1} \times RBA \times IRS_{rec-c}\left(\frac{200 mg}{day}\right) \times \frac{10^{-6}kg}{1 mg}}{BW_{rec-c}(15 kg)}\right)}{BW_{rec-c}(15 kg)}$$

· dermal contact with soil or sediment

$$SL_{rec-soil-ca-vc-der}(mg/kg) = \frac{TR}{\left(\frac{CSF_{0}\left(\frac{mg}{kg-day}\right)^{-1} \times DFS_{rec-adj}\left(\frac{mg}{kg}\right) \times ABS_{d} \times \frac{10^{-6}kg}{1 mg}}{AT_{rec}\left(\frac{365 \text{ days}}{\text{ year}} \times LT(70 \text{ years})\right)}\right) + \left(\frac{CSF_{0}\left(\frac{mg}{kg-day}\right)^{-1} \times SA_{rec-c}\left(\frac{2373 \text{ cm}^{2}}{day}\right) \times AF_{rec-c}\left(\frac{0.2 \text{ mg}}{cm^{2}}\right) \times ABS \times \frac{10^{-6}kg}{1 \text{ mg}}}{BWV_{rec-c}(15 \text{ kg})}\right)}\right)}$$

• inhalation of volatiles and particulates emitted from soil or sediment



#### 4.6.1.6 Trichloroethylene - Carcinogenic and Mutagenic

The recreator soil or sediment land use equations, presented here, contain the following exposure routes:

• incidental ingestion of soil or sediment

$$SL_{rec-soil-tce-ing}(mg/kg) = \frac{TR \times AT_{rec} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right)}{CSF_{o} \left(\frac{mg}{\text{kg-day}}\right)^{-1} \times RBA \times \left(\frac{10^{-6} \text{ kg}}{mg}\right) \times \left[\left(\frac{CAF_{o} (0.804) \times IFS_{rec-adj} \left(\frac{mg}{\text{kg}}\right)\right) + \left(\frac{MAF_{o} (0.202) \times IFSM_{rec-adj} \left(\frac{mg}{\text{kg}}\right)\right)\right]}{CSF_{o} \left(\frac{mg}{\text{kg-day}}\right)^{-1} \times RBA \times \left(\frac{10^{-6} \text{ kg}}{mg}\right) \times \left[\left(\frac{MAF_{o} (0.202) \times IFSM_{rec-adj} \left(\frac{mg}{\text{kg}}\right)\right) + \left(\frac{MAF_{o} (0.202) \times IFSM_{rec-adj} \left(\frac{mg}{\text{kg}}\right)\right)\right]}{CSF_{o} \left(\frac{mg}{\text{kg-day}}\right)^{-1} \times RBA \times \left(\frac{10^{-6} \text{ kg}}{mg}\right) \times \left(\frac{MAF_{o} (0.202) \times IFSM_{rec-adj} \left(\frac{mg}{\text{kg}}\right)\right)}{CSF_{o} \left(\frac{mg}{\text{kg-day}}\right)^{-1} \times RBA \times \left(\frac{10^{-6} \text{ kg}}{mg}\right) \times \left(\frac{MAF_{o} (0.202) \times IFSM_{rec-adj} \left(\frac{mg}{\text{kg}}\right)}{CSF_{o} \left(\frac{mg}{\text{kg}}\right)} \times \left(\frac{10^{-6} \text{ kg}}{mg}\right) \times \left(\frac{10^{-6} \text{ kg}}{MAF_{o} (0.202) \times IFSM_{rec-adj} \left(\frac{mg}{\text{kg}}\right)}\right) \times \left(\frac{10^{-6} \text{ kg}}{MAF_{o} (0.202) \times IFSM_{rec-adj} \left(\frac{mg}{\text{kg}}\right)}\right)$$

where:

$$\mathsf{IFS}_{\mathsf{rec-adj}}\left(\frac{\mathsf{mg}}{\mathsf{kg}}\right) = \left( \begin{array}{c} \mathsf{ED}_{\mathsf{rec-c}}\left(\mathsf{years}\right) \times \mathsf{EF}_{\mathsf{rec-c}}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{IRS}_{\mathsf{rec-c}}\left(\frac{200\ \mathsf{mg}}{\mathsf{day}}\right)}{\mathsf{BW}_{\mathsf{rec-c}}\left(\mathsf{15\ kg}\right)} + \\ \frac{\mathsf{ED}_{\mathsf{rec}}\left(\mathsf{years}\right) - \mathsf{ED}_{\mathsf{rec-c}}\left(\mathsf{years}\right)\right) \times \mathsf{EF}_{\mathsf{rec-a}}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{IRS}_{\mathsf{rec-a}}\left(\frac{\mathsf{100\ mg}}{\mathsf{day}}\right)}{\mathsf{BW}_{\mathsf{rec-a}}\left(\mathsf{80\ kg}\right)} \right)$$

where:

$$IFSM_{rec-adj} \cdot \left(\frac{mg}{kg}\right) = \begin{pmatrix} ED_{0-2} (years) \times EF_{0-2} \left(\frac{days}{year}\right) \times IRS_{0-2} \left(\frac{200 mg}{day}\right) \times 10 \\ BW_{0-2} (15 kg) \\ ED_{2-6} (years) \times EF_{2-6} \left(\frac{days}{year}\right) \times IRS_{2-6} \left(\frac{200 mg}{day}\right) \times 3 \\ BW_{2-6} (15 kg) \\ ED_{6-16} (years) \times EF_{6-16} \left(\frac{days}{year}\right) \times IRS_{6-16} \left(\frac{100 mg}{day}\right) \times 3 \\ BW_{6-16} (80 kg) \\ ED_{16-26} (years) \times EF_{16-26} \left(\frac{days}{year}\right) \times IRS_{16-26} \left(\frac{100 mg}{day}\right) \times 1 \\ BW_{16-26} (80 kg) \end{pmatrix}$$

• dermal contact with soil or sediment

$$SL_{\text{rec-soll-tce-der}}(mg/kg) = \frac{TR \times AT_{\text{rec}}\left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right)}{\left[\frac{CSF_{0}\left(\frac{mg}{kg \cdot day}\right)^{-1}}{GlABS}\right] \times \left(\frac{10^{6} \text{kg}}{\text{mg}}\right) \times \left(\left[CAF_{0}\left(0.804\right) \times DFS_{\text{rec-adj}}\left(\frac{mg}{kg}\right) \times ABS_{d}\right] + \left(MAF_{0}\left(0.202\right) \times DFSM_{\text{rec-adj}}\left(\frac{mg}{kg}\right) \times ABS_{d}\right)\right)}$$
where:
$$DFS_{\text{rec-adj}}\left(\frac{mg}{kg}\right) = \left[\frac{ED_{\text{rec-c}}\left(\text{ years}\right) \times EF_{\text{rec-c}}\left(\frac{days}{year}\right) \times SA_{\text{rec-c}}\left(\frac{2373 \text{ cm}^{2}}{day}\right) \times AF_{\text{rec-c}}\left(\frac{0.2 \text{ mg}}{cm^{2}}\right)}{BW_{\text{rec-c}}(15 \text{ kg})} + \left(\frac{ED_{\text{rec}}\left(\text{ years}\right) \cdot ED_{\text{rec-c}}\left(\text{ years}\right) \times EF_{\text{rec-adj}}\left(\frac{days}{year}\right) \times SA_{\text{rec-a}}\left(\frac{5032 \text{ cm}^{2}}{day}\right) \times AF_{\text{rec-a}}\left(\frac{0.07 \text{ mg}}{cm^{2}}\right)}{BW_{\text{rec-a}}(30 \text{ kg})} + \left(\frac{ED_{0-2}\left(\text{ years}\right) \times EF_{0-2}\left(\frac{days}{year}\right) \times AF_{0-2}\left(\frac{2373 \text{ cm}^{2}}{day}\right) \times AF_{\text{rec-a}}\left(\frac{0.07 \text{ mg}}{cm^{2}}\right)}{BW_{\text{rec-a}}(30 \text{ kg})} + \left(\frac{ED_{0-2}\left(\text{ years}\right) \times EF_{0-2}\left(\frac{days}{year}\right) \times AF_{0-2}\left(\frac{2373 \text{ cm}^{2}}{day}\right) \times AF_{\text{rec-a}}\left(\frac{0.07 \text{ mg}}{cm^{2}}\right) \times AF_{\text{rec-a}}\left(\frac{0.07 \text{ mg}}{day}\right)}{BW_{\text{rec-a}}(30 \text{ kg})} + \left(\frac{ED_{0-2}\left(\text{ years}\right) \times EF_{0-2}\left(\frac{days}{year}\right) \times AF_{0-2}\left(\frac{0.2 \text{ mg}}{cm^{2}}\right) \times SA_{0-2}\left(\frac{2373 \text{ cm}^{2}}{day}\right) \times AF_{\text{rec-a}}\left(\frac{0.07 \text{ mg}}{cm^{2}}\right) \times AF_{\text{rec-a}}\left(\frac{10.07 \text{ mg}}{cm^{2}}\right) \times AF_{\text{rec-a}}\left(\frac{10.07 \text{ mg}}{cm^{2}}\right) \times AF_{\text{rec-a}}\left(\frac{10.07 \text{ mg}}{day}\right) \times AF_{$$

• inhalation of volatiles and particulates emitted from soil or sediment

$$SL_{rec} \cdot soil-tce-inh} (mg/kg) = \frac{TR \times AT_{rec} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right)}{IUR \left(\frac{\mu g}{m3}\right)^{-1} \times \left(\frac{1}{VF_{ulim} \left(\frac{m^3}{kg}\right)} + \frac{1}{PEF \left(\frac{m^3}{kg}\right)}\right) \times \left(\frac{1000 \ \mu g}{mg}\right) \times \left(\frac{1 \ day}{24 \ hours}\right) \times \left(\frac{1000 \ \mu g}{mg}\right) \times \left(\frac{1 \ day}{24 \ hours}\right) \times \left(\frac{1000 \ \mu g}{24 \ hours}\right) \times ET_{0-2} \left(\frac{hours}{day}\right) \times MAF_i (0.244) \times 10\right) + \left(\frac{CAF_i (0.756) \times EF_{rec} \left(\frac{days}{year}\right) \times }{ED_{rec} (26 \ years) \times ET_{rec} \left(\frac{hours}{day}\right)}\right) + \left(\frac{ED_{0-2} (2 \ years) \times EF_{0-2} \left(\frac{days}{year}\right) \times ET_{2-6} \left(\frac{hours}{day}\right) \times MAF_i (0.244) \times 3\right) + \left(ED_{6-16} (10 \ years) \times EF_{6-16} \left(\frac{days}{year}\right) \times ET_{6-16} \left(\frac{hours}{day}\right) \times MAF_i (0.244) \times 3\right) + \left(ED_{16-26} (10 \ years) \times EF_{16-26} \left(\frac{days}{year}\right) \times ET_{16-26} \left(\frac{hours}{day}\right) \times MAF_i (0.244) \times 1\right)\right)$$

• Total  

$$SL_{rec-soil-tce-tot} (mg/kg) = \frac{1}{\frac{1}{SL_{rec-soil-tce-ing}} + \frac{1}{SL_{rec-soil-tce-der}} + \frac{1}{SL_{rec-soil-tce-inh}}}$$

A number of studies have shown that inadvertent ingestion of soil is common among children 6 years old and younger (Calabrese et al. 1989, Davis et al. 1990, Van Wijnen et al. 1990). Therefore, the dose method uses an age-adjusted soil ingestion factor that takes into account the difference in daily soil ingestion rates, body weights, and exposure duration for children from 1 to 6 years old and others from 7 to 26 years old. The equation is presented below. This health-protective approach is chosen to take into account the higher daily rates of soil ingestion in children as well as the longer duration of exposure that is anticipated for a long-term resident. For more on this method, see <u>RAGS Part B (PDF)</u> (68 pp, 721 K).

#### **4.6.1.7 Supporting Equations**

• Child

$$ED_{rec-c}$$
 (years) =  $ED_{0-2}$  (years) +  $ED_{2-6}$  (years)

$$\mathsf{BW}_{\mathsf{rec-c}}\left(\mathsf{kg}\right) = \frac{\mathsf{BW}_{0\text{-}2}\left(\mathsf{kg}\right) \times \mathsf{ED}_{0\text{-}2}\left(\mathsf{years}\right) + \mathsf{BW}_{2\text{-}6}\left(\mathsf{kg}\right) \times \mathsf{ED}_{2\text{-}6}\left(\mathsf{years}\right)}{\mathsf{ED}_{0\text{-}2}\left(\mathsf{years}\right) + \mathsf{ED}_{2\text{-}6}\left(\mathsf{years}\right)}$$

$$\mathsf{AF}_{\mathsf{rec-c}}\left(\frac{\mathsf{events}}{\mathsf{day}}\right) = \frac{\mathsf{AF}_{\mathsf{O}-2}\left(\frac{\mathsf{events}}{\mathsf{day}}\right) \times \mathsf{ED}_{\mathsf{O}-2}\left(\mathsf{years}\right) + \mathsf{AF}_{\mathsf{2-6}}\left(\frac{\mathsf{events}}{\mathsf{day}}\right) \times \mathsf{ED}_{\mathsf{2-6}}\left(\mathsf{years}\right)}{\mathsf{ED}_{\mathsf{O}-2}\left(\mathsf{years}\right) + \mathsf{ED}_{\mathsf{2-6}}\left(\mathsf{years}\right)}$$

$$\mathsf{EF}_{\mathsf{rec-c}}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) = \frac{\mathsf{EF}_{\mathsf{O}-2}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{ED}_{\mathsf{O}-2}\left(\mathsf{years}\right) + \mathsf{EF}_{\mathsf{2-6}}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{ED}_{\mathsf{2-6}}\left(\mathsf{years}\right)}{\mathsf{ED}_{\mathsf{O}-2}\left(\mathsf{years}\right) + \mathsf{ED}_{\mathsf{2-6}}\left(\mathsf{years}\right)}$$

$$\mathsf{ET}_{\mathsf{rec-c}}\left(\frac{\mathsf{hours}}{\mathsf{day}}\right) = \frac{\mathsf{ET}_{0-2}\left(\frac{\mathsf{hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{0-2}\left(\mathsf{years}\right) + \mathsf{ET}_{2-6}\left(\frac{\mathsf{hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{2-6}\left(\mathsf{years}\right)}{\mathsf{ED}_{0-2}\left(\mathsf{years}\right) + \mathsf{ED}_{2-6}\left(\mathsf{years}\right)}$$

$$SA_{rec-c}\left(\frac{cm^2}{day}\right) = \frac{SA_{0-2}\left(\frac{cm^2}{day}\right) \times ED_{0-2}\left(years\right) + SA_{2-6}\left(\frac{cm^2}{day}\right) \times ED_{2-6}\left(years\right)}{ED_{0-2}\left(years\right) + ED_{2-6}\left(years\right)}$$

$$\mathsf{IRS}_{\mathsf{rec-c}}\left(\frac{\mathsf{mg}}{\mathsf{day}}\right) = \frac{\mathsf{IRS}_{\mathsf{O}\text{-}2}\left(\frac{\mathsf{mg}}{\mathsf{day}}\right) \times \mathsf{ED}_{\mathsf{O}\text{-}2}\left(\mathsf{years}\right) + \mathsf{IRS}_{\mathsf{2}\text{-}\mathsf{6}}\left(\frac{\mathsf{mg}}{\mathsf{day}}\right) \times \mathsf{ED}_{\mathsf{2}\text{-}\mathsf{6}}\left(\mathsf{years}\right)}{\mathsf{ED}_{\mathsf{O}\text{-}2}\left(\mathsf{years}\right) + \mathsf{ED}_{\mathsf{2}\text{-}\mathsf{6}}\left(\mathsf{years}\right)}$$

• Adult

$$ED_{rec-a}(years) = ED_{6-16}(years) + ED_{16-26}(years)$$

$$\mathsf{BW}_{\mathsf{rec-a}}\left(\mathsf{kg}\right) = \frac{\mathsf{BW}_{6-16}\left(\mathsf{kg}\right) \times \mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{BW}_{16-26}\left(\mathsf{kg}\right) \times \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}{\mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}$$

$$\mathsf{AF}_{\mathsf{rec-a}}\left(\frac{\mathsf{events}}{\mathsf{day}}\right) = \frac{\mathsf{AF}_{6-16}\left(\frac{\mathsf{events}}{\mathsf{day}}\right) \times \mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{AF}_{16-26}\left(\frac{\mathsf{events}}{\mathsf{day}}\right) \times \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}{\mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}$$

$$\mathsf{EF}_{\mathsf{rec-a}}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) = \frac{\mathsf{EF}_{6-16}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{EF}_{16-26}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}{\mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}$$

$$\mathsf{ET}_{\mathsf{rec-a}}\left(\frac{\mathsf{hours}}{\mathsf{day}}\right) = \frac{\mathsf{ET}_{6-16}\left(\frac{\mathsf{hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ET}_{16-26}\left(\frac{\mathsf{hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}{\mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}$$

$$SA_{rec-a}\left(\frac{cm^2}{day}\right) = \frac{SA_{6-16}\left(\frac{cm^2}{day}\right) \times ED_{6-16}\left(years\right) + SA_{16-26}\left(\frac{cm^2}{day}\right) \times ED_{16-26}\left(years\right)}{ED_{6-16}\left(years\right) + ED_{16-26}\left(years\right)}$$

$$\operatorname{IRS}_{\operatorname{rec-a}}\left(\frac{\operatorname{mg}}{\operatorname{day}}\right) = \frac{\operatorname{IRS}_{6-16}\left(\frac{\operatorname{mg}}{\operatorname{day}}\right) \times \operatorname{ED}_{6-16}\left(\operatorname{years}\right) + \operatorname{IRS}_{16-26}\left(\frac{\operatorname{mg}}{\operatorname{day}}\right) \times \operatorname{ED}_{16-26}\left(\operatorname{years}\right)}{\operatorname{ED}_{6-16}\left(\operatorname{years}\right) + \operatorname{ED}_{16-26}\left(\operatorname{years}\right)}$$

• Age-adjusted

 $\mathsf{ED}_{\mathsf{rec}}(\mathsf{years}) = \mathsf{ED}_{0\text{-}2}(\mathsf{years}) + \mathsf{ED}_{2\text{-}6}(\mathsf{years}) + \mathsf{ED}_{6\text{-}16}(\mathsf{years}) + \mathsf{ED}_{16\text{-}26}(\mathsf{years})$ 

$$\mathsf{EF}_{\mathsf{rec}}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) = \frac{\mathsf{EF}_{0-2}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{ED}_{0-2}\left(\mathsf{years}\right) + \mathsf{EF}_{2-6}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{ED}_{2-6}\left(\mathsf{years}\right) + \mathsf{EF}_{6-16}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{EF}_{16-26}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}{\mathsf{ED}_{0-2}\left(\mathsf{years}\right) + \mathsf{ED}_{2-6}\left(\mathsf{years}\right) + \mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}}{\mathsf{ED}_{0-2}\left(\mathsf{years}\right) + \mathsf{ED}_{2-6}\left(\mathsf{years}\right) + \mathsf{ED}_{6-16}\left(\frac{\mathsf{hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ET}_{16-26}\left(\frac{\mathsf{hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}}{\mathsf{ED}_{0-2}\left(\mathsf{years}\right) + \mathsf{ED}_{2-6}\left(\mathsf{years}\right) + \mathsf{ET}_{6-16}\left(\frac{\mathsf{hours}}{\mathsf{day}}\right) \times \mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ET}_{16-26}\left(\mathsf{years}\right)}}{\mathsf{ED}_{0-2}\left(\mathsf{years}\right) + \mathsf{ED}_{2-6}\left(\mathsf{years}\right) + \mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}}$$

## 4.6.2 Recreator Surface Water

This receptor is exposed to chemicals that are present in surface water. Ingestion of water and dermal contact with water are appropriate pathways. Dermal contact with surface water is also considered for analytes determined to be within the effective predictive domain as described in Section 4.9.8. Inhalation is not considered due to mixing with outdoor air. There are no default RSLs for this scenario; only site-specific.

## The recreator surface water land use is not provided in the Generic Tables but RSLs can be created by using the Calculator.

## 4.6.2.1 Noncarcinogenic - Child

The recreator surface water land use equation, presented here, contains the following exposure routes:

• incidental ingestion of water

$$SL_{rec-wat-nc-ing-c} (\mu g/L) = \frac{THQ \times AT_{rec-c} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{rec-c} (6 \text{ years})\right) \times BW_{rec-c} (15 \text{ kg}) \times \left(\frac{1000 \ \mu g}{\text{mg}}\right)}{EF_{rec-c} \left(\frac{\text{days}}{\text{year}}\right) \times ED_{rec-c} (6 \text{ years}) \times \frac{1}{RfD_o \left(\frac{mg}{\text{kg-d}}\right)} \times IRW_{rec-c} \left(\frac{0.12 \ L}{\text{hour}}\right) \times EV_{rec-c} \left(\frac{\text{events}}{\text{day}}\right) \times ET_{event-rec-c} \left(\frac{\text{hours}}{\text{event}}\right)}$$

• dermal

FOR INORGANICS:

$$SL_{rec-wat-nc-der-c}(\mu g L) = \frac{DA_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) \times \left(\frac{1000 \text{ cm}^{3}}{L}\right)}{K_{p}\left(\frac{cm}{hour}\right) \times ET_{event-rec-c}\left(\frac{hours}{event}\right)}$$
FOR ORGANICS:  
IF ET\_{event-rec-c}\left(\frac{hours}{event}\right) \le t^{\*} (hour), then SL\_{rec-wat-nc-der}(\mu g L) = \frac{DA\_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) \times \left(\frac{1000 \text{ cm}^{3}}{L}\right)}{2 \times FA \times K\_{p}\left(\frac{cm}{hour}\right) \sqrt{\frac{6 \times r\_{event}\left(\frac{hours}{event}\right) \times ET\_{event-rec-c}\left(\frac{hours}{event}\right)}{\pi}}
or,  
IF ET\_{event-rec-c}\left(\frac{hours}{event}\right) > t^{\*} (hour), then SL\_{rec-wat-nc-der}(\mu g L) = \frac{DA\_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) \times \left(\frac{1000 \text{ cm}^{3}}{L}\right)}{FA \times K\_{p}\left(\frac{cm}{hour}\right) \times \left[\frac{1}{1 + B} + \frac{1}{2} \times r\_{event}\left(\frac{hours}{event}\right) \times \left(\frac{1 + 3B + 3B^{2}}{(1 + B)^{2}}\right)\right]}
where:  

$$DA_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) = \frac{THQ \times AT_{rec-c}\left(\frac{365 \text{ days}}{year} \times ED_{rec-c}(6 \text{ years})\right) \times \left(\frac{1000 \text{ }\mu g}{mg}\right) \times SM_{rec-c}(15 \text{ kg})}{\left(\frac{1}{RrD_{0}\left(\frac{mg}{kg \cdot day}\right) \times GIABS}\right) \times EV_{rec-c}\left(\frac{events}{day}\right) \times ED_{rec-c}(6 \text{ years}) \times EF_{rec-c}\left(\frac{days}{year}\right) \times SA_{rec-c}(6365 \text{ cm}^{2})}$$
Total

$$SL_{rec-wat-nc-tot-c} (\mu g/L) = \frac{1}{\frac{1}{SL_{rec-wat-nc-ing-c}} + \frac{1}{SL_{rec-wat-nc-der-c}}}$$

#### 4.6.2.2 Noncarcinogenic - Adult

The recreator surface water land use equation, presented here, contains the following exposure routes:

• incidental ingestion of water

$$SL_{rec-wat-nc-ing-a}(\mu gAL) = \frac{THQ \times AT_{rec-a}\left(\frac{365 \text{ days}}{\text{year}} \times ED_{rec-a}\left(20 \text{ years}\right)\right) \times BW_{rec-a}\left(80 \text{ kg}\right) \times \left(\frac{1000 \text{ }\mu g}{\text{ mg}}\right)}{EF_{rec-a}\left(\frac{days}{\text{ year}}\right) \times ED_{rec-a}\left(20 \text{ years}\right) \times \frac{1}{RfD_{o}\left(\frac{mg}{\text{ kg-d}}\right)} \times IRW_{rec-a}\left(\frac{0.071 \text{ L}}{\text{hour}}\right) \times EV_{rec-a}\left(\frac{\text{events}}{\text{ day}}\right) \times ET_{event-rec-a}\left(\frac{\text{hours}}{\text{ event}}\right)}$$

• dermal

•

#### Regional Screening Levels (RSLs) - User's Guide | Risk Assessment | US EPA

FOR INDEGRANCE:  

$$SL_{rec-wat-nc-der-a}(\mu g L) = \frac{DA_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) * \left(\frac{1000 \text{ cm}^{3}}{L}\right)}{V_{\mu}\left(\frac{cm}{hour}\right) * ET_{event-rec-c}\left(\frac{hours}{event}\right)}$$
FOR ORGANICS:  
IF ET\_{event-rec-a}\left(\frac{hours}{event}\right) \leq t^{\*} (hour) then SL\_{rec-wat-nc-der}(\mu g L) = \frac{DA\_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) \* \left(\frac{1000 \text{ cm}^{3}}{L}\right)}{2 \* FA \* K\_{p}\left(\frac{cm}{hour}\right) \sqrt{\frac{6 \times revent}{event}\left(\frac{hours}{event}\right) \times ET\_{event-rec-a}\left(\frac{hours}{event}\right)}}{\pi}
or,  
IF ET\_{event-rec-a}\left(\frac{hours}{event}\right) > t^{\*} (hour), then SL\_{rec-wat-nc-der}(\mu g L) = \frac{DA\_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) \times \left(\frac{1000 \text{ cm}^{3}}{L}\right)}{FA \times K\_{p}\left(\frac{cm}{hour}\right) \times \left[\frac{ET\_{event-rec-a}\left(\frac{hours}{event}\right) \times \left(\frac{1 + 3B + 3B^{2}}{(1 + B)^{2}}\right)\right]}{Where:}
where:  

$$DA_{event}\left(\frac{ug}{cm^{2} \cdot event}\right) = \frac{TH0 \times AT_{rec-a}\left(\frac{365 \text{ days}}{y_{ear}} \times ED_{rec-a}(20 \text{ years})\right) \times \left(\frac{1000 \text{ ug}}{mg}\right) \times SW_{rec-a}(80 \text{ kg})}{\left(\frac{1}{RD_{0}\left(\frac{mg}{kg}\text{ day}\right)} \times GIABS}\right) \times EV_{rec-a}\left(\frac{events}{day}\right) \times ED_{rec-a}(20 \text{ years}) \times EF_{rec-a}\left(\frac{days}{y_{ear}}\right) \times SA_{rec-a}\left(19652 \text{ cm}^{2}\right)}$$
• Total

$$SL_{rec-wat-nc-tot-a} (\mu g \Lambda L) = \frac{1}{\frac{1}{SL_{rec-wat-nc-ing-a}} + \frac{1}{SL_{rec-wat-nc-der-a}}}$$

#### 4.6.2.3 Carcinogenic

The recreator surface water land use equation, presented here, contains the following exposure routes:

• incidental ingestion of water  

$$SL_{rec-wat-ca-ing}(\mu g/L) = \frac{TR \times AT_{rec} \left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right) \times \left(\frac{1000 \mu g}{\text{mg}}\right)}{CSF_{0} \left(\frac{\text{mg}}{\text{kg}-\text{day}}\right)^{-1} \times IFW_{rec-adj} \left(\frac{L}{\text{kg}}\right)}$$
where:  

$$IFW_{rec-adj} \left(\frac{L}{\text{kg}}\right) = \left(\frac{EV_{rec-c} \left(\frac{\text{events}}{\text{day}}\right) \times ED_{rec-c}(6 \text{ years}) \times EF_{rec-c} \left(\frac{\text{days}}{\text{year}}\right) \times ET_{event-rec-c} \left(\frac{\text{hours}}{\text{event}}\right) \times IRW_{rec-c} \left(\frac{0.12 \text{ L}}{\text{hour}}\right)}{\frac{EV_{rec-a} \left(\frac{\text{events}}{\text{day}}\right) \times ED_{rec-a}(20 \text{ years}) \times EF_{rec-a} \left(\frac{\text{days}}{\text{year}}\right) \times ET_{event-rec-a} \left(\frac{\text{hours}}{\text{event}}\right) \times IRW_{rec-a} \left(\frac{0.071 \text{ L}}{\text{hour}}\right)}{BW_{rec-a}(80 \text{ kg})}$$

• dermal

#### Regional Screening Levels (RSLs) - User's Guide | Risk Assessment | US EPA





#### 4.6.2.4 Mutagenic

The recreator surface water land use equation, presented here, contains the following exposure routes:

· incidental ingestion of water

$$SL_{rec-wat-mu-ing}(\mu g/L) = \frac{TR \times AT_{rec} \left(\frac{365 \text{ days}}{\text{year}} \times LT (70 \text{ years})\right) \times \left(\frac{1000 \ \mu g}{\text{mg}}\right)}{CSF_{0} \left(\frac{\text{mg}}{\text{kg-day}}\right)^{-1} \times IFWM_{rec-adj} \left(\frac{L}{\text{kg}}\right)}$$
where:
$$IFWM_{rec-adj} \left(\frac{L}{\text{kg}}\right) = \left(\frac{\frac{ED_{0-2}(2 \text{ years}) \times EF_{0-2} \left(\frac{\text{days}}{\text{year}}\right) \times IRW_{0-2} \left(\frac{0.12 \ L}{\text{hour}}\right) \times EV_{0-2} \left(\frac{\text{events}}{\text{day}}\right) \times ET_{0-2} \left(\frac{\text{hours}}{\text{event}}\right) \times 10}{BW_{0-2} (15 \ \text{kg})} + \frac{ED_{2-6} \left(4 \ \text{years}\right) \times EF_{2-6} \left(\frac{\text{days}}{\text{year}}\right) \times IRW_{2-6} \left(\frac{0.12 \ L}{\text{hour}}\right) \times EV_{2-6} \left(\frac{\text{events}}{\text{day}}\right) \times ET_{2-6} \left(\frac{\text{hours}}{\text{event}}\right) \times 4}{BW_{2-6} (15 \ \text{kg})} + \frac{ED_{6-16} \left(10 \ \text{years}\right) \times EF_{6-16} \left(\frac{\text{days}}{\text{year}}\right) \times IRW_{6-16} \left(\frac{0.071 \ L}{\text{hour}}\right) \times EV_{6-16} \left(\frac{\text{events}}{\text{day}}\right) \times ET_{6-16} \left(\frac{\text{hours}}{\text{event}}\right) \times 4}{BW_{6-16} (80 \ \text{kg})} + \frac{ED_{16-26} \left(10 \ \text{years}\right) \times EF_{16-26} \left(\frac{\text{days}}{\text{year}}\right) \times IRW_{16-26} \left(\frac{0.071 \ L}{\text{hour}}\right) \times EV_{16-26} \left(\frac{\text{events}}{\text{day}}\right) \times ET_{16-26} \left(\frac{\text{hours}}{\text{event}}\right) \times 1}{BW_{16-26} (80 \ \text{kg})} + \frac{ED_{16-26} \left(\frac{\text{hours}}{\text{event}}\right) \times 1}{BW_{16-26} (80 \ \text{kg})} \times ET_{16-26} \left(\frac{\text{hours}}{\text{event}}\right) \times 1}{BW_{16-26} (80 \ \text{kg})}$$

• dermal

#### Regional Screening Levels (RSLs) - User's Guide | Risk Assessment | US EPA



#### 4.6.2.5 Vinyl Chloride - Carcinogenic

The recreator surface water land use equation, presented here, contains the following exposure routes:

• incidental ingestion of water

$$SL_{\text{tree.watccavcing}}\left[igd_{-}\right] = \frac{1}{\left[\frac{CSF_{0}\left(\frac{mg}{k_{0}}d_{3}\right)^{-1}rIFW_{\text{tree}}cdg\left(\frac{k_{0}}{k_{0}}\right)^{2}\left(\frac{mg}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}\right)^{2}\left(\frac{mg}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}kI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}KI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}KI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}KI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}KI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}KI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}KI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}KI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}KI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}KI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k_{0}}d_{3}\right)^{2}KI_{1}(rog_{\text{tree}}dg\left(\frac{k_{0}}{k$$

SL<sub>rec-wat-ca-vc-ing</sub>+SL<sub>rec-wat-ca-vc-der</sub>

٠

## 4.6.2.6 Trichloroethylene - Carcinogenic and Mutagenic

The recreator surface water land use equation, presented here, contains the following exposure routes:

• incidental ingestion of water

$$SL_{rec.wat-tce-ing}(\mu g/L) = \frac{TR \times AT_{rec} \left(\frac{365 \text{ days}}{\text{ year}} \times LT (70 \text{ years})\right) \times \left(\frac{1000 \ \mu g}{\text{ mg}}\right)}{CSF_{0} \left(\frac{mg}{\text{ kg-day}}\right)^{-1} \times RBA \times \left(\left(CAF_{0} (0.804) \times IFW_{rec-adj}\left(\frac{L}{\text{ kg}}\right)\right)\right) + \left(MAF_{0} (0.202) \times IFWM_{rec-adj}\left(\frac{L}{\text{ kg}}\right)\right)}$$
where:
$$IFW_{rec-adj}\left(\frac{L}{\text{ kg}}\right) = \left(\frac{EV_{rec-c}\left(\frac{events}{\text{ day}}\right) \times ED_{rec-c} (6 \text{ years}) \times EF_{rec-c}\left(\frac{\text{ days}}{\text{ year}}\right) \times ET_{event-rec-c}\left(\frac{\text{ hours}}{\text{ event}}\right) \times IRW_{rec-c}\left(\frac{0.12 \ L}{\text{ hour}}\right)}{W_{rec-c} (15 \text{ kg})} + \frac{EV_{rec-a}\left(\frac{events}{\text{ day}}\right) \times ED_{rec-a} (20 \text{ years}) \times EF_{rec-a}\left(\frac{\text{ days}}{\text{ year}}\right) \times ET_{event-rec-a}\left(\frac{\text{ hours}}{\text{ event}}\right) \times IRW_{rec-a}\left(\frac{0.071 \ L}{\text{ hour}}\right)}{W_{rec-a} (80 \ \text{ kg})} + \frac{EO_{0.2} \left(2 \text{ years}\right) \times EF_{0.2}\left(\frac{\text{ days}}{\text{ year}}\right) \times IRW_{0.2}\left(\frac{0.12 \ L}{\text{ hour}}\right) \times ET_{0.2}\left(\frac{\text{ hours}}{\text{ event}}\right) \times IRW_{rec-a}\left(\frac{0.071 \ L}{\text{ hour}}\right)}{W_{rec-a} (80 \ \text{ kg})} + \frac{ED_{0.2} \left(2 \text{ years}\right) \times EF_{0.2}\left(\frac{\text{ days}}{\text{ year}}\right) \times IRW_{0.2}\left(\frac{0.12 \ L}{\text{ hour}}\right) \times ET_{0.2}\left(\frac{\text{ hours}}{\text{ event}}\right) \times 10}{W_{0.2} (15 \ \text{ kg})} + \frac{ED_{2.6} \left(4 \text{ years}\right) \times EF_{2.6}\left(\frac{\text{ days}}{\text{ year}}\right) \times IRW_{2.6}\left(\frac{0.071 \ L}{\text{ hour}}\right) \times ET_{2.6}\left(\frac{\text{ hours}}{\text{ day}}\right) \times \frac{3}{\pi} + \frac{ED_{2.6} \left(10 \text{ years}\right) \times EF_{6.16}\left(\frac{\text{ days}}{\text{ year}}\right) \times IRW_{2.6}\left(\frac{0.071 \ L}{\text{ hour}}\right) \times ET_{2.6}\left(\frac{\text{ hours}}{\text{ day}}\right) \times \frac{3}{\pi} + \frac{ED_{2.6} \left(10 \text{ years}\right) \times EF_{6.16}\left(\frac{\text{ days}}{\text{ year}}\right) \times IRW_{6.16}\left(\frac{0.071 \ L}{\text{ hour}}\right) \times EV_{6.16}\left(\frac{\text{ events}}{\text{ day}}\right) \times ET_{6.16}\left(\frac{\text{ hours}}{\text{ day}}\right) \times 1} + \frac{ED_{6.16} \left(10 \text{ years}\right) \times EF_{16.26}\left(\frac{\text{ days}}{\text{ year}}\right) \times IRW_{6.16}\left(\frac{0.071 \ L}{\text{ hour}}\right) \times EV_{16.26}\left(\frac{\text{ events}}{\text{ day}}\right) \times ET_{16.26}\left(\frac{\text{ hours}}{\text{ day}}\right) \times 1} + \frac{EV_{6.16} \left(\frac{100 \ Hours}{100 \ Hours}\right) \times 1}{EV_{16.26} \left(\frac{10 \ Hours}{100 \ Hours}\right) \times 1} + \frac{EV_{6.16} \left(\frac{10 \ Hours}{100 \ Hours}\right) \times 1}{EV_{16.26} \left(\frac{10 \ Hours}{100 \ Hours}\right) \times 1} + \frac{EV_{6.26} \left(\frac{10 \ Hours}{100 \ Hours}\right)}{EV_{1$$

• dermal

#### Regional Screening Levels (RSLs) - User's Guide | Risk Assessment | US EPA



#### Regional Screening Levels (RSLs) - User's Guide | Risk Assessment | US EPA



#### 4.6.2.7 Supporting Equations

• Child

$$ED_{rec-c}(years) = ED_{0-2}(years) + ED_{2-6}(years)$$

$$\mathsf{BW}_{\mathsf{rec-c}}\left(\mathsf{kg}\right) = \frac{\mathsf{BW}_{0\text{-}2}\left(\mathsf{kg}\right) \times \mathsf{ED}_{0\text{-}2}\left(\mathsf{years}\right) + \mathsf{BW}_{2\text{-}6}\left(\mathsf{kg}\right) \times \mathsf{ED}_{2\text{-}6}\left(\mathsf{years}\right)}{\mathsf{ED}_{0\text{-}2}\left(\mathsf{years}\right) + \mathsf{ED}_{2\text{-}6}\left(\mathsf{years}\right)}$$

$$\mathsf{EV}_{\mathsf{rec-c}}\left(\frac{\mathsf{events}}{\mathsf{day}}\right) = \frac{\mathsf{EV}_{0-2}\left(\frac{\mathsf{events}}{\mathsf{day}}\right) \times \mathsf{ED}_{0-2}\left(\mathsf{years}\right) + \mathsf{EV}_{2-6}\left(\frac{\mathsf{events}}{\mathsf{day}}\right) \times \mathsf{ED}_{2-6}\left(\mathsf{years}\right)}{\mathsf{ED}_{0-2}\left(\mathsf{years}\right) + \mathsf{ED}_{2-6}\left(\mathsf{years}\right)}$$

$$\mathsf{EF}_{\mathsf{rec-c}}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) = \frac{\mathsf{EF}_{0-2}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{ED}_{0-2}\left(\mathsf{years}\right) + \mathsf{EF}_{2-6}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{ED}_{2-6}\left(\mathsf{years}\right)}{\mathsf{ED}_{0-2}\left(\mathsf{years}\right) + \mathsf{ED}_{2-6}\left(\mathsf{years}\right)}$$

$$\mathsf{ET}_{\mathsf{rec-c}}\left(\frac{\mathsf{hours}}{\mathsf{event}}\right) = \frac{\mathsf{ET}_{0-2}\left(\frac{\mathsf{hours}}{\mathsf{event}}\right) \times \mathsf{ED}_{0-2}\left(\mathsf{years}\right) + \mathsf{ET}_{2-6}\left(\frac{\mathsf{hours}}{\mathsf{event}}\right) \times \mathsf{ED}_{2-6}\left(\mathsf{years}\right)}{\mathsf{ED}_{0-2}\left(\mathsf{years}\right) + \mathsf{ED}_{2-6}\left(\mathsf{years}\right)}$$

$$\mathsf{SA}_{\mathsf{rec-c}}\left(\mathsf{cm}^{2}\right) = \frac{\mathsf{SA}_{\mathsf{O}\text{-}2}\left(\mathsf{cm}^{2}\right) \times \mathsf{ED}_{\mathsf{O}\text{-}2}\left(\mathsf{years}\right) + \mathsf{SA}_{\mathsf{2}\text{-}6}\left(\mathsf{cm}^{2}\right) \times \mathsf{ED}_{\mathsf{2}\text{-}6}\left(\mathsf{years}\right)}{\mathsf{ED}_{\mathsf{O}\text{-}2}\left(\mathsf{years}\right) + \mathsf{ED}_{\mathsf{2}\text{-}6}\left(\mathsf{years}\right)}$$

$$\mathsf{IRW}_{\mathsf{rec-c}}\left(\frac{\mathsf{L}}{\mathsf{hour}}\right) = \frac{\mathsf{IRW}_{\mathsf{0-2}}\left(\frac{\mathsf{L}}{\mathsf{hour}}\right) \times \mathsf{ED}_{\mathsf{0-2}}\left(\mathsf{years}\right) + \mathsf{IRW}_{\mathsf{2-6}}\left(\frac{\mathsf{L}}{\mathsf{hour}}\right) \times \mathsf{ED}_{\mathsf{2-6}}\left(\mathsf{years}\right)}{\mathsf{ED}_{\mathsf{0-2}}\left(\mathsf{years}\right) + \mathsf{ED}_{\mathsf{2-6}}\left(\mathsf{years}\right)}$$

• Adult

$$ED_{rec-a}(years) = ED_{6-16}(years) + ED_{16-26}(years)$$

$$\mathsf{BW}_{\mathsf{rec-a}}\left(\mathsf{kg}\right) = \frac{\mathsf{BW}_{6-16}\left(\mathsf{kg}\right) \times \mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{BW}_{16-26}\left(\mathsf{kg}\right) \times \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}{\mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}$$

$$\mathsf{EV}_{\mathsf{rec-a}}\left(\frac{\mathsf{events}}{\mathsf{day}}\right) = \frac{\mathsf{EV}_{6-16}\left(\frac{\mathsf{events}}{\mathsf{day}}\right) \times \mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{EV}_{16-26}\left(\frac{\mathsf{events}}{\mathsf{day}}\right) \times \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}{\mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}$$

$$\mathsf{EF}_{\mathsf{rec-a}}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) = \frac{\mathsf{EF}_{6-16}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{EF}_{16-26}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \times \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}{\mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}$$

$$\mathsf{ET}_{\mathsf{rec-a}}\left(\frac{\mathsf{hours}}{\mathsf{event}}\right) = \frac{\mathsf{ET}_{6-16}\left(\frac{\mathsf{hours}}{\mathsf{event}}\right) \times \mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ET}_{16-26}\left(\frac{\mathsf{hours}}{\mathsf{event}}\right) \times \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}{\mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}$$

$$SA_{rec-a}(cm^{2}) = \frac{SA_{6-16}(cm^{2}) \times ED_{6-16}(years) + SA_{16-26}(cm^{2}) \times ED_{16-30}(years)}{ED_{6-16}(years) + ED_{16-26}(years)}$$

$$\mathsf{IRW}_{\mathsf{rec-a}}\left(\frac{\mathsf{L}}{\mathsf{hour}}\right) = \frac{\mathsf{IRW}_{6-16}\left(\frac{\mathsf{L}}{\mathsf{hour}}\right) \times \mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{IRW}_{16-26}\left(\frac{\mathsf{L}}{\mathsf{hour}}\right) \times \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}{\mathsf{ED}_{6-16}\left(\mathsf{years}\right) + \mathsf{ED}_{16-26}\left(\mathsf{years}\right)}$$

## 4.7 Ingestion of Fish

The fish RSL represents the concentration, in the fish, that can be consumed. Note: the consumption rate for fish is not age adjusted for this land use. Also, the SL calculated for fish is not for surface water or soil but is for fish tissue.

## The ingestion of fish land use is not provided in the Generic Tables but RSLs can be created by using the Calculator.

#### 4.7.1 Noncarcinogenic

The ingestion of fish equation, presented here, contains the following exposure route:

• consumption of fish.

$$SL_{res-fsh-nc-ing}(mg/kg) = \frac{THQ \times AT_{res-a} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{res} \left(26 \text{ years}\right)\right) \times BW_{res-a} \left(80 \text{ kg}\right)}{EF_{res-a} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res} \left(26 \text{ years}\right) \times \frac{1}{RfD_0 \left(\frac{mg}{\text{kg-day}}\right)} \times IRF_{res-a} \left(\frac{mg}{\text{day}}\right) \times \frac{10^{-6} \text{kg}}{1 \text{ mg}}}$$

https://www.epa.gov/risk/regional-screening-levels-rsls-users-guide

#### 4.7.2 Carcinogenic

The ingestion of fish equation, presented here, contains the following exposure route:

• consumption of fish

$$SL_{res-fsh-ca-ing}(mg/kg) = \frac{TR \times AT_{res} \left(\frac{365 \text{ days}}{\text{year}} \times LT(70 \text{ years})\right) \times BW_{res-a}(80 \text{ kg})}{EF_{res} \left(\frac{350 \text{ days}}{\text{year}}\right) \times ED_{res}(26 \text{ years}) \times CSF_{o} \left(\frac{mg}{\text{kg-day}}\right)^{-1} \times IRF_{res-a} \left(\frac{mg}{\text{day}}\right) \times \frac{10^{-6} \text{kg}}{1 \text{ mg}}}$$

## 4.8 Soil to Groundwater

The soil to groundwater scenario was developed to identify concentrations in soil that have the potential to contaminate groundwater above risk based RSLs or MCLs. Migration of contaminants from soil to groundwater can be envisioned as a two-stage process: (1) release of contaminant from soil to soil leachate and (2) transport of the contaminant through the underlying soil and aquifer to a receptor well. The soil to groundwater scenario considers both of these fate and transport mechanisms. First, the acceptable groundwater concentration is multiplied by a dilution factor to obtain a target leachate concentration. For example, if the dilution factor is 10 and the MCL is 0.05 mg/L, the target soil leachate concentration would be 0.5 mg/L. The partition equation (presented in the Soil Screening Guidance documents) is then used to calculate the total soil concentration corresponding to this soil leachate concentration.

These equations are used to calculate screening levels in soil (SSLs) that are protective of groundwater. SSLs are either back-calculated from protective risk-based ground water concentrations or based on MCLs. The SSLs were designed for use during the early stages of a site evaluation when information about subsurface conditions may be limited. Because of this constraint, the equations used are based on conservative, simplifying assumptions about the release and transport of contaminants in the subsurface. Migration of contaminants from soil to groundwater can be envisioned as a two-stage process: (1) release of contaminant in soil leachate and (2) transport of the contaminant through the underlying soil and aquifer to a receptor well. The SSL methodology considers both of these fate and transport mechanisms.

The SSLs protective of groundwater, provided in the generic tables and the calculator, are all risk-based concentrations based on three phases (vapor, soil and water). No substitution for  $C_{sat}$  is performed. If the risk-based concentration exceeds  $C_{sat}$ , the resulting SSL concentration may be overly protective. This is because the dissolved, absorbed and vapor concentrations cease to rise linearly as soil concentration increases above the  $C_{sat}$  level (pure product or nonaqueous phase liquid (NAPL) is present). The SSL model used in the RSL calculator is not a four phase model. If a NAPL is present at your site more sophisticated models may be necessary.

SSLs are provided for metals in the Generic Tables based on Kds from the <u>Soil Screening Guidance Exhibit</u>  $\underline{C-4}$ . According to Appendix C,

"Exhibit C-4 provides pH-specific soil-water partition coefficients (Kd) for metals. Site-specific soil pH measurements can be used to select appropriate Kd values for these metals. Where site-specific soil pH values are not available, values corresponding to a pH of 6.8 should be used."

If a metal is not listed in Exhibit C-4, Kds were taken from <u>Baes, C. F. 1984 (PDF)</u> (167 pp, 3.0 MB). Kds for organic coumponds are calculated from  $K_{oc}$  and the fraction of organic carbon in the soil ( $f_{oc}$ ). Kds for metals are listed below.
## This land use is for developing residential default soil screening levels for the protection of groundwater that are presented in the RSL Generic Tables.

Chemical	CAS	Kd	Reference
Aluminum	7429- 90-5	1.50E+03	Baes, C.F. 1984
Antimony (metallic)	7440- 36-0	4.50E+01	SSG 9355.4-23 July 1996
Arsenic, Inorganic	7440- 38-2	2.90E+01	SSG 9355.4-23 July 1996
Barium	7440- 39-3	4.10E+01	SSG 9355.4-23 July 1996
Beryllium and compounds	7440- 41-7	7.90E+02	SSG 9355.4-23 July 1996
Boron And Borates Only	7440- 42-8	3.00E+00	Baes, C.F. 1984
Bromate	15541- 45-4	7.50E+00	Baes, C.F. 1984
Cadmium (Diet)	7440- 43-9	7.50E+01	SSG 9355.4-23 July 1996
Cadmium (Water)	7440- 43-9	7.50E+01	SSG 9355.4-23 July 1996
Chlorine	7782- 50-5	2.50E-01	Baes, C.F. 1984
Chromium (III) (Insoluble Salts)	16065- 83-1	1.80E+06	SSG 9355.4-23 July 1996
Chromium Salts	0-00-3	8.50E+02	Baes, C.F. 1984

Chemical	CAS	Kd	Reference
Chromium VI (chromic acid mists)	18540- 29-9	1.90E+01	SSG 9355.4-23 July 1996
Chromium VI (particulates)	18540- 29-9	1.90E+01	SSG 9355.4-23 July 1996
Chromium, Total (1:6 ratio Cr VI : Cr III)	7440- 47-3	1.80E+06	SSG 9355.4-23 July 1996
Cobalt	7440- 48-4	4.50E+01	Baes, C.F. 1984
Copper	7440- 50-8	3.50E+01	Baes, C.F. 1984
Cyanide (CN-)	57-12-5	9.90E+00	SSG 9355.4-23 July 1996
Fluoride	16984- 48-8	1.50E+02	Surrogate Value from Fluorine (Soluble Fluoride)
Fluorine (Soluble Fluoride)	7782- 41-4	1.50E+02	Baes, C.F. 1984
Hydrogen Cyanide (HCN)	74-90-8	9.90E+00	Surrogate value from Cyanide
Iron	7439- 89-6	2.50E+01	Baes, C.F. 1984
Lead and Compounds	7439- 92-1	9.00E+02	Baes, C.F. 1984
Lithium	7439- 93-2	3.00E+02	Baes, C.F. 1984
Magnesium	7439- 95-4	4.50E+00	Baes, C.F. 1984

Chemical	CAS	Kd	Reference
Manganese (Diet)	7439- 96-5	6.50E+01	Baes, C.F. 1984
Manganese (Water)	7439- 96-5	6.50E+01	Baes, C.F. 1984
Mercury (elemental)	7439- 97-6	5.20E+01	SSG 9355.4-23 July 1996
Mercury, Inorganic Salts	0-01-7	5.20E+01	SSG 9355.4-23 July 1996
Molybdenum	7439- 98-7	2.00E+01	Baes, C.F. 1984
Nickel Soluble Salts	7440- 02-0	6.50E+01	SSG 9355.4-23 July 1996
Phosphorus, White	7723- 14-0	3.50E+00	Baes, C.F. 1984
Selenium	7782- 49-2	5.00E+00	SSG 9355.4-23 July 1996
Silver	7440- 22-4	8.30E+00	SSG 9355.4-23 July 1996
Sodium	7440- 23-5	1.00E+02	Baes, C.F. 1984
Sodium Fluoride	7681- 49-4	1.50E+02	Surrogate Value from Fluorine (Soluble Fluoride)
Strontium, Stable	7440- 24-6	3.50E+01	Baes, C.F. 1984
Thallium (Soluble Salts)	7440- 28-0	7.10E+01	SSG 9355.4-23 July 1996

Chemical	CAS	Kd	Reference
Thorium	0-23-2	1.50E+05	Baes, C.F. 1984
Tin	7440- 31-5	2.50E+02	Baes, C.F. 1984
Titanium	7440- 32-6	1.00E+03	Baes, C.F. 1984
Uranium (Soluble Salts)	0-23-8	4.50E+02	Baes, C.F. 1984
Vanadium and Compounds	0-06-6	1.00E+03	SSG 9355.4-23 July 1996
Vanadium, Metallic	7440- 62-2	1.00E+03	SSG 9355.4-23 July 1996
Zinc (Metallic)	7440- 66-6	6.20E+01	SSG 9355.4-23 July 1996
Zirconium	7440- 67-7	3.00E+03	Baes, C.F. 1984

Because Kds vary greatly by soil type, it is highly recommended that site-specific Kds be determined and used to develop SSLs.

The more protective of the carcinogenic and noncarcinogenic SLs is selected to calculate the SSL.

#### 4.8.1 Noncarcinogenic Tapwater Equations for SSLs

The tapwater equations, presented in Section 4.1.2.1, are used to calculate the noncarcinogenic SSLs for volatiles and nonvolatiles. If the contaminant is a volatile, ingestion, dermal and inhalation exposure routes are considered. If the contaminant is not a volatile, only ingestion and dermal are considered.

#### 4.8.2 Carcinogenic Tapwater Equations for SSLs

The tapwater equations, presented in Section 4.1.2.3, are used to calculate the carcinogenic SSLs for volatiles and nonvolatiles. Sections 4.1.2.4 and 4.1.2.5 present the mutagenic and vinyl chloride equations, respectively. If the contaminant is a volatile, ingestion, dermal and inhalation exposure routes are considered. If the contaminant is not a volatile, only ingestion and dermal are considered.

#### 4.8.3 Method 1 for SSL Determination

Method 1 employs a partitioning equation for migration to groundwater and defaults are provided. This method is used to generate the download default tables. If H' is not available, SSL can still be calculated. H' changes when groundwater temperature changes. Please see section 4.9.9 of this user guide for H' determination at temperature other than 25°C.

• method 1.

$$SSL(mg/kg) = C_{water} \left(\frac{mg}{L}\right) \times \left[ K_{d} \left(\frac{L}{kg}\right) + \left(\frac{\left(\theta_{w} \left(\frac{L_{water}}{L_{soil}}\right) + \theta_{a} \left(\frac{L_{air}}{L_{soil}}\right) \times H'\right)}{\rho_{b} \left(\frac{1.5 \text{ kg}}{L}\right)} \right) \right]$$

$$C_{water} \left(\frac{mg}{L}\right) = MCL \left(\frac{ug}{L}\right) \times \left(\frac{1 mg}{1000 ug}\right) \times DAF$$
  
or:  
$$C_{water} \left(\frac{mg}{L}\right) = PRG \left(\frac{ug}{L}\right) \times \left(\frac{1 mg}{1000 ug}\right) \times DAF$$

and:  

$$\theta_{a} \left( \frac{L_{air}}{L_{soil}} \right) = n \left( \frac{L_{water}}{L_{soil}} \right) - \theta_{w} \left( \frac{0.3 \ L_{water}}{L_{soil}} \right);$$

$$n \left( \frac{L_{pore}}{L_{soil}} \right) = 1 - \left( \frac{\rho_{b} \left( \frac{1.5 \ kg}{L} \right)}{\rho_{s} \left( \frac{2.65 \ kg}{L} \right)} \right)$$

$$K_{d} \left( \frac{L}{kg} \right) = K_{oc} \left( \frac{L}{kg} \right) \times f_{oc} \left( \frac{0.002 \ g\text{-carbon}}{g\text{-soil}} \right), \text{ for organic compounds;}$$

$$K_{d} \text{ values for inorganic compounds are listed in the user guide.}$$

The fraction of organic carbon ( $f_{oc}$ ) selected for this equation is 0.002. This is the default for subsurface soil identified in U.S. EPA 1996b, Sections 2.5.2 and 2.5.7. According to this source, soil organic carbon decreases rapidly with depth. Note that the default  $f_{oc}$  in section "4.9.4 Infinite Source Chronic Volatilization Factor (VF<sub>ulim</sub>)" is 0.006, which is the default for surface soil from the same study.

#### 4.8.4 Method 2 for SSL Determination

Method 2 employs a mass-limit equation for migration to groundwater and site-specific information is required. This method can be used in the calculator portion of this website.

• method 2.

$$SSL(mg/kg) = \frac{C_{w}\left(\frac{mg}{L}\right) \times I\left(\frac{0.18 \text{ m}}{\text{year}}\right) \times ED (70 \text{ years})}{\rho_{b}\left(\frac{1.5 \text{ kg}}{L}\right) \times d_{s}(m)}$$
where:  

$$C_{w}\left(\frac{mg}{L}\right) = MCL\left(\frac{ug}{L}\right) \times \left(\frac{1 \text{ mg}}{1000 \text{ ug}}\right) \times DAF$$
or:  

$$C_{w}\left(\frac{mg}{L}\right) = PRG\left(\frac{ug}{L}\right) \times \left(\frac{1 \text{ mg}}{1000 \text{ ug}}\right) \times DAF$$

#### 4.8.5 Determination of the Dilution Factor

The SSL values in the download tables are based on a dilution factor of 1. If one wishes to use the calculator to calculate screening levels using the SSL guidance for a source up to 0.5 acres, then a dilution factor of 20 can be used. If all of the parameters needed to calculate a site-specific dilution factor are known, they may be entered.

• Dilution Attenuation Factor.

Dilution Attenuation Factor (DAF) = 1 + 
$$\frac{K\left(\frac{m}{year}\right) \times i\left(\frac{m}{m}\right) \times d(m)}{I\left(\frac{0.18 \text{ m}}{year}\right) \times L(m)}$$

where:

$$d(m) = \left(0.0112 \times L^{2}(m)\right)^{0.5} + d_{a} \times \left[1 - exp\left(\frac{-L(m) \times I\left(\frac{m}{year}\right)}{K\left(\frac{m}{year}\right) \times I\left(\frac{m}{m}\right) \times d_{a}(m)}\right)\right]$$

#### 4.9 Supporting Equations and Parameter Discussion

There are two parts of the above land use equations that require further explanation. They are the inhalation variables: the particulate emission factor (PEF) and the volatilization factor (VF).

#### 4.9.1 Wind-driven Particulate Emission Factor (PEF)

Inhalation of contaminants adsorbed to respirable particles (PM10) was assessed using a default PEF equal to  $1.36 \times 10^9 \text{ m}^3/\text{kg}$ . This equation relates the contaminant concentration in soil with the concentration of respirable particles in the air due to fugitive dust emissions from contaminated soils. The generic PEF was derived using default values that correspond to a receptor point concentration of approximately  $0.76 \,\mu\text{g/m}^3$ . The relationship is derived by Cowherd (1985) for a rapid assessment procedure applicable to a typical hazardous waste site, where the surface contamination provides a relatively continuous and constant potential for emission over an extended period of time (e.g., years). This represents an annual average emission rate based on wind erosion that should be compared with chronic health criteria; it is not appropriate for evaluating the potential for more acute exposures. Definitions of the input variables are in Table 1.

1 x<sup>4</sup>

With the exception of specific heavy metals, the PEF does not appear to significantly affect most soil screening levels. The equation forms the basis for deriving a generic PEF for the inhalation pathway. For more details regarding specific parameters used in the PEF model, refer to Appendix D of the <u>Supplemental Soil Screening Guidance</u>. The use of alternate values on a specific site should be justified and presented in an Administrative Record if considered in CERCLA remedy selection.

$$\mathsf{PEF}\left(\frac{\mathsf{m}_{\mathsf{air}}^{3}}{\mathsf{kg}_{\mathsf{soil}}}\right) = \frac{\mathsf{Q}}{\mathsf{C}_{\mathsf{wind}}} \left(\frac{\left(\frac{\mathsf{g}}{\mathsf{m}^{2}},\mathsf{s}\right)}{\left(\frac{\mathsf{kg}}{\mathsf{m}^{3}}\right)}\right) \times \frac{3,600\left(\frac{\mathsf{s}}{\mathsf{hour}}\right)}{0.036\times(1-\mathsf{V})\times\left(\frac{\mathsf{U}_{\mathsf{m}}\left(\frac{\mathsf{m}}{\mathsf{s}}\right)}{\mathsf{U}_{\mathsf{t}}\left(\frac{\mathsf{m}}{\mathsf{s}}\right)}\right)^{3}\times\mathsf{F}(\mathsf{x})}$$

where:

$$\frac{Q}{C_{wind}} = A \times exp\left[\frac{\left(\ln A_{s} (acre) - B\right)^{2}}{C}\right]$$
  
and:  
if x < 2, F (x) = 1.91207 - 0.0278085 x + 0.48113 x<sup>2</sup> - 1.09871 x<sup>3</sup> + 0.33534  
if x ≥ 2, F (x) = 0.18 (8x<sup>3</sup> + 12x) e^{(-x<sup>2</sup>)}  
where:  
x = 0.886 ×  $\left(\frac{U_{t}}{U_{m}}\right)$ 

Note: the generic PEF evaluates wind-borne emissions and does not consider dust emissions from traffic or other forms of mechanical disturbance that could lead to greater emissions than assumed here.

#### 4.9.2 Vehicle traffic-driven Particulate Emission Factor (PEFsc)

The equation to calculate the subchronic particulate emission factor ( $PEF_{sc}$ ) is significantly different from the residential and non-residential PEF equations. The  $PEF_{sc}$  focuses exclusively on emissions from truck traffic on unpaved roads, which typically contribute the majority of dust emissions during construction. This equation requires estimates of parameters such as the number of days with at least 0.01 inches of rainfall, the mean vehicle weight, and the sum of fleet vehicle distance traveled during construction.

The number of days with at least 0.01 inches of rainfall can be estimated using Exhibit 5-2 in the <u>supplemental soil screening guidance (PDF)</u> (187 pp, 2.2 MB). Mean vehicle weight (W) can be estimated by assuming the numbers and weights of different types of vehicles. For example, assuming that the daily unpaved road traffic consists of 20 two-ton cars and 10 twenty-ton trucks, the mean vehicle weight would be:

W = [(20 cars x 2 tons/car) + (10 trucks x 20 tons/truck)]/30 vehicles = 8 tons

The sum of the fleet vehicle kilometers traveled during construction ( $\sum$  VKT) can be estimated based on the size of the area of surface soil contamination, assuming the configuration of the unpaved road, and the amount of vehicle traffic on the road. For example, if the area of surface soil contamination is 0.5 acres (or 2,024 m<sup>2</sup>), and one assumes that this area is configured as a square with the unpaved road segment dividing the square evenly, the road length would be equal to the square root of 2,024 m<sup>2</sup>, 45 m (or 0.045 km). Assuming that each vehicle travels the length of the road once per day, 5 days per week for a total of 6 months, the total fleet vehicle kilometers traveled would be:

 $\sum$  VKT = 30 vehicles x 0.045 km/day x (52 weeks/year  $\div$  2) x 5 days/wk = 175.5 km

Г

$$\begin{aligned} \mathsf{PEF}_{sc}\left(\frac{\mathsf{m}_{air}^{3}}{\mathsf{kg}_{soil}}\right) &= \frac{\mathsf{Q}}{\mathsf{C}_{sr}}\left(\frac{\left(\frac{\mathsf{g}}{\mathsf{m}^{2}}, \mathsf{s}\right)}{\left(\frac{\mathsf{kg}}{\mathsf{m}^{3}}\right)}\right) \times \frac{1}{\mathsf{F}_{D}} \times \left(\frac{\mathsf{I}}{\mathsf{F}_{D}}\right)^{0.8} \times \frac{\mathsf{T}_{t}\left(\mathsf{s}\right) \times \mathsf{A}_{R}\left(\mathsf{m}^{2}\right)}{\frac{2.6 \times \left(\frac{\mathsf{s}}{12}\right)^{0.8} \times \left(\frac{\mathsf{W}(\mathsf{tons})}{3}\right)^{0.4}}{\left(\frac{\mathsf{M}_{dry}}{\mathsf{0.2}}\right)^{0.3}} \times \frac{\mathsf{365}\left(\frac{\mathsf{days}}{\mathsf{year}}\right) \cdot \mathsf{p}\left(\frac{\mathsf{days}}{\mathsf{year}}\right)}{\mathsf{365}\left(\frac{\mathsf{days}}{\mathsf{year}}\right)} \times 281.9 \times \mathsf{EVKT} \right) \\ &= \frac{\mathsf{Q}}{\mathsf{C}_{sr}}\left(\frac{\left(\frac{\mathsf{m}}{\mathsf{m}^{2}}, \mathsf{s}\right)}{\left(\frac{\mathsf{kg}}{\mathsf{m}^{3}}\right)}\right) = \mathsf{A} \times \mathsf{exp}\left[\frac{\left(\mathsf{InA}_{\mathsf{s}}\left(\mathsf{acre}\right) \cdot \mathsf{B}\right)^{2}}{\mathsf{C}}\right] \\ &= \mathsf{A}_{R}\left(\mathsf{m}^{2}\right) = \mathsf{L}_{R}\left(\mathsf{ff}\right) \times \mathsf{W}_{R}\left(\mathsf{20\ feet}\right) \times \mathsf{0.092903}\left(\frac{\mathsf{m}^{2}}{\mathsf{feet}^{2}}\right) \\ &= \mathsf{W}\left(\mathsf{tons}\right) = \frac{\left(\mathsf{number}\ of\ cas \times \frac{\mathsf{tons}}{\mathsf{car}} + \mathsf{number}\ of\ trucks \times \frac{\mathsf{tons}}{\mathsf{truck}}\right)}{\mathsf{total}\ \mathsf{vehcles}} \\ &= \mathsf{EVKT}\left(\mathsf{km}\right) = \mathsf{total}\ \mathsf{vehcles} \times \mathsf{distance}\left(\frac{\mathsf{km}}{\mathsf{day}}\right) \times \mathsf{EW}_{\mathsf{cw}}\left(\frac{\mathsf{Weeks}}{\mathsf{year}}\right) \times \mathsf{DW}_{\mathsf{cw}}\left(\frac{\mathsf{days}}{\mathsf{week}}\right) \\ &\mathsf{T}_{t}\left(\mathsf{7200000}\ \mathsf{s}\right) = \mathsf{ED}_{\mathsf{cw}}\left(\mathsf{1}\ \mathsf{years}\right) \times \mathsf{EF}_{\mathsf{cw}}\left(\frac{\mathsf{250}\ \mathsf{days}}{\mathsf{year}}\right) \times \mathsf{ET}_{\mathsf{cw}}\left(\frac{\mathsf{8}\ \mathsf{hours}}{\mathsf{day}}\right) \times \left(\frac{\mathsf{24}\ \mathsf{hours}}{\mathsf{hour}}\right) \\ &\mathsf{F}_{D}\left(\mathsf{0.185e4}\right) = \mathsf{0.1852} + \left(\mathsf{5.3537}\{t}_{c}\right) + \left(\mathsf{-9.6318}\{t}_{c}^{2}\right) \\ &\mathsf{t}_{c}\left(\mathsf{8400\ hours}\right) = \mathsf{ED}_{\mathsf{cw}}\left(\mathsf{1}\ \mathsf{years}\right) \times \mathsf{EW}_{\mathsf{cw}}\left(\frac{\mathsf{50}\ \mathsf{weeks}}{\mathsf{year}}\right) \times \left(\frac{\mathsf{24}\ \mathsf{hours}}{\mathsf{day}}\right) \\ &\mathsf{T}_{c}\left(\mathsf{2400\ hours}\right) = \mathsf{ED}_{\mathsf{cw}}\left(\mathsf{1}\ \mathsf{years}\right) \times \mathsf{EW}_{\mathsf{cw}}\left(\frac{\mathsf{50}\ \mathsf{weeks}}{\mathsf{year}}\right) \times \left(\frac{\mathsf{24}\ \mathsf{hours}}{\mathsf{day}}\right) \\ &\mathsf{T}_{c}\left(\mathsf{2400\ hours}\right) = \mathsf{ED}_{\mathsf{cw}}\left(\mathsf{1}\ \mathsf{years}\right) \times \mathsf{EW}_{\mathsf{cw}}\left(\frac{\mathsf{50}\ \mathsf{weeks}}{\mathsf{year}\right) \times \left(\frac{\mathsf{24}\ \mathsf{hours}}{\mathsf{day}}\right) \\ &\mathsf{t}\left(\mathsf{24}\ \mathsf{hours}\right) \\ &\mathsf{t}\left(\mathsf{24}\ \mathsf{hours}\right) \\ &\mathsf{t}\left(\mathsf{24}\ \mathsf{hours}\right) \\ &\mathsf{t}\left(\mathsf{24}\\mathsf{hours}\right) \\ &\mathsf{t}\left(\mathsf{24}\\mathsf{hours}\right) \\ &\mathsf{t}\left(\mathsf{24}\\mathsf{hours}\right) \\ &\mathsf{t}\left(\mathsf{24}\\mathsf{hours}\right) \\ &\mathsf{t}\left(\mathsf{44}\mathsf{hours}\right) \\ &\mathsf{t}\left(\mathsf{44}\mathsf{hours}\right) \\ &\mathsf{t}\left(\mathsf{44}\mathsf{hours}\right) \\ &\mathsf{t}\left(\mathsf{44}\mathsf{hours}\right) \\ &\mathsf{t}\left(\mathsf{44}\mathsf{hours}\right) \\ &\mathsf{t}\left(\mathsf{44}\mathsf{hours}\right)$$

#### 4.9.3 Other than vehicle traffic-driven Particulate Emission Factor (PEF'sc)

Other than emissions from unpaved road traffic, the construction worker may also be exposed to particulate matter emissions from wind erosion, excavation soil dumping, dozing, grading, and tilling or similar operations PEF'<sub>sc</sub>. These operations may occur separately or concurrently and the duration of each operation may be different. For these reasons, the total unit mass emitted from each operation is calculated separately and the sum is normalized over the entire area of contamination and over the entire time during which construction activities take place. Equation E-26 in the <u>supplemental soil screening guidance (PDF)</u> (187 pp, 2.2 MB) was used.

17 - 55

$$\mathsf{PEF}_{sc}^{'}\left(\frac{m_{air}^{3}}{kg_{soil}}\right) = \frac{Q}{C_{sa}}\left[\frac{\left(\frac{g}{m^{2}},s\right)}{\left(\frac{kg}{m^{3}}\right)}\right] \times \frac{1}{F_{D}} \times \frac{1}{\langle J_{T}^{'} \rangle \left(\frac{g}{m^{2}}-s\right)}$$

where:

$$\begin{split} & \frac{Q}{C_{sa}} \left[ \left( \frac{g}{m^2 \cdot s} \right) \\ \frac{\left( \frac{g}{m^2} \right)}{\left( \frac{kg}{m^3} \right)} \right] = A \times exp \left[ \frac{\left[ \ln A_c \left( acre \right) \cdot B \right]^2}{C} \right] \\ < & \int_{T} > \left( \frac{g}{m^2 - s} \right) = \frac{M_{wind}^{PC} \left( g \right) + M_{excav} \left( g \right) + M_{doz} \left( g \right) + M_{grade} \left( g \right) + M_{till} \left( g \right)}{A_{surf} \left( m^2 \right) \times T_t \left( s \right)} \\ & M_{wind}^{PC} \left( g \right) = 0.036 \times (1 \cdot V) \times \left( \frac{U_m \left( \frac{m}{s} \right)}{U_t \left( \frac{m}{s} \right)} \right)^3 \times F \left( x \right) \times A_{surf} \left( m^2 \right) \times ED \left( years \right) \times 8760 \left( \frac{hours}{year} \right) \\ & M_{excav} \left( g \right) = 0.35 \times 0.0016 \times \left( \frac{U_m \left( \frac{m}{s} \right)}{\left( \frac{M_{m-excav} \left( \% \right)}{2} \right)^{1.4}} \times \rho_{soil} \left( \frac{Mg}{m^3} \right) \times A_{excav} \left( m^2 \right) \times d_{excav} \left( m \right) \times N_{A,dump} \times 1000 \\ & M_{doz} \left( g \right) = 0.75 \times \left( \frac{0.45 \times s_{doz} \left( \% \right)^{1.5}}{\left( M_{m-doz} \left( \% \right) \right)^{1.4}} \times \frac{\Sigma \vee KT_{doz} \left( km \right)}{S_{doz} \left( \frac{km}{h_T} \right)} \times 1000 \left( \frac{g}{kg} \right) \\ & M_{grade} \left( g \right) = 0.60 \times 0.0056 \times S_{grade} \left( \frac{km}{hour} \right)^{2.0} \times \Sigma \vee KT_{grade} \left( km \right) \times 1000 \left( \frac{g}{kg} \right) \\ & ard \end{split}$$

$$\begin{split} \mathsf{M}_{\mathsf{till}}\left(\mathsf{g}\right) &= 1.1 \times \mathsf{s}_{\mathsf{till}}\left(\%\right)^{0.6} \times \mathsf{A}_{\mathsf{c-till}}\left(\mathsf{acres}\right) \times 4047 \left(\frac{\mathsf{m}^2}{\mathsf{acre}}\right) \times 10^{-4} \left(\frac{\mathsf{ha}}{\mathsf{m}^2}\right) \times 1000 \left(\frac{\mathsf{g}}{\mathsf{kg}}\right) \times \mathsf{N}_{\mathsf{A}\mathsf{-till}} \end{split}$$
 where:

$$\Sigma \vee KT_{\text{grade}}(km) = A_{\text{c-grade}}(acres) \times 4047 \left(\frac{m^2}{acre}\right) \times \frac{1}{B_{\text{g}}(m)} \times \frac{1}{1000 \left(\frac{m}{km}\right)} \times N_{\text{A-grade}}$$

where:

$$\Sigma \vee \mathsf{KT}_{doz}(\mathsf{km}) = \mathsf{A}_{c\text{-}doz}(\mathsf{acres}) \times 4047 \left(\frac{\mathsf{m}^2}{\mathsf{acre}}\right) \times \frac{1}{\mathsf{B}_g(\mathsf{m})} \times \frac{1}{1000 \left(\frac{\mathsf{m}}{\mathsf{km}}\right)} \times \mathsf{N}_{A\text{-}doz}$$
$$\mathsf{T}_t(7200000 \text{ s}) = \mathsf{ED}_{cw}(1 \text{ years}) \times \mathsf{EF}_{cw}\left(\frac{250 \text{ days}}{\mathsf{year}}\right) \times \mathsf{ET}_{cw}\left(\frac{8 \text{ hours}}{\mathsf{day}}\right) \times \left(\frac{3600 \text{ s}}{\mathsf{hour}}\right)$$
$$\mathsf{F}_D(0.18584) = 0.1852 + (5.3537 / \mathsf{t}_c) + (-9.6318 / \mathsf{t}_c^2)$$

#### https://www.epa.gov/risk/regional-screening-levels-rsls-users-guide

$$t_{c} (8400 \text{ hours}) = ED_{cw} (1 \text{ years}) \times EW_{cw} \left(\frac{50 \text{ weeks}}{\text{year}}\right) \times \left(\frac{7 \text{ days}}{\text{week}}\right) \times \left(\frac{24 \text{ hours}}{\text{day}}\right)$$
  
and:  
if x < 2, F (x) = 1.91207 - 0.0278085 x + 0.48113 x<sup>2</sup> - 1.09871 x<sup>3</sup> + 0.335341 x<sup>4</sup>  
if x ≥ 2, F (x) = 0.18 (8x<sup>3</sup>+12x) e^{(-x<sup>2</sup>)}  
where:  
x = 0.886 ×  $\left(\frac{U_{t}}{U_{m}}\right)$ 

#### 4.9.4 Infinite Source Chronic Volatilization Factor (VFulim)

The soil-to-air VF is used to define the relationship between the concentration of the contaminant in soil and the flux of the volatilized contaminant to air. VF is calculated from the equation below using chemical-specific properties and either site-measured or default values for soil moisture, dry bulk density, and fraction of organic carbon in soil. The <u>Soil Screening Guidance: User's Guide (PDF)</u> (89 pp, 863 K) describes how to develop site measured values for these parameters.

VF is only calculated for volatile compounds. Volatiles, for the purpose of this guidance, are chemicals with a Henry's Law constant greater than or equal to  $1 \times 10^{-5}$  atm-m<sup>3</sup>/mole or a vapor pressure greater than or equal to 1 mm Hg. The volatile status of a chemical is important for some exposure routes. According to <u>RAGS Part E</u>, dermal absorption to soil is not assessed for volatiles. For the purposes of this guidance, dermal exposure to soil is only quantified if <u>RAGS Part E</u> provides a dermal absorption value in Exhibit 3-4 or the website, regardless of volatility status. The rationale for this is that in the considered soil exposure scenarios, volatile organic compounds would tend to be volatilized from the soil on skin and should be accounted for via inhalation routes in the combined exposure pathway analysis. Further, a chemical must be volatile in order to be included in the calculation of tapwater inhalation. H' changes when groundwater temperature changes. Please see section 4.9.9 of this user guide for H' determination at temperature other than 25°C.

unlimited source model for chronic exposure

$$VF_{ulim}\left(\frac{m_{air}^{3}}{kg_{soil}}\right) = \frac{\frac{Q}{C_{vol}}\left(\frac{\left(\frac{g}{m^{2} \cdot s}\right)}{\left(\frac{kg}{m^{3}}\right)}\right) \times \left(3.14 \times D_{A}\left(\frac{cm^{2}}{s}\right) \times T(s)\right)^{1/2} \times 10^{-4}\left(\frac{m^{2}}{cm^{2}}\right)}{2 \times \rho_{b}\left(\frac{g}{cm^{3}}\right) \times D_{A}\left(\frac{cm^{2}}{s}\right)}$$

where:  

$$\frac{Q}{C_{vol}} \left( \frac{\left(\frac{g}{m^2 - s}\right)}{\left(\frac{kg}{m^3}\right)} \right) = A \times exp \left[ \frac{\left(\ln A_s (acre) - B\right)^2}{C} \right]$$

$$D_{A}\left(\frac{cm^{2}}{s}\right) = \frac{\left(\theta_{a}\left(\frac{L_{air}}{L_{soil}}\right)^{10/3} \times D_{ia}\left(\frac{cm^{2}}{s}\right) \times H' + \theta_{w}\left(\frac{0.15 \ L_{water}}{L_{soil}}\right)^{10/3} \times D_{iw}\left(\frac{cm^{2}}{s}\right)\right) / n^{2}\left(\frac{L_{pore}}{L_{soil}}\right)}{\rho_{b}\left(\frac{1.5g}{cm^{3}}\right) \times K_{d}\left(\frac{cm^{3}}{g}\right) + \theta_{w}\left(\frac{0.15 \ L_{water}}{L_{soil}}\right) + \theta_{a}\left(\frac{L_{air}}{L_{soil}}\right) \times H'$$

where:

where:

$$\theta_{a}\left(\frac{L_{air}}{L_{soil}}\right) = n\left(\frac{L_{pore}}{L_{soil}}\right) - \theta_{w}\left(\frac{0.15 \ L_{water}}{L_{soil}}\right) \text{and } n\left(\frac{L_{pore}}{L_{soil}}\right) = 1 - \begin{pmatrix}\rho_{b}\left(\frac{1.5g}{cm^{3}}\right) \\ \rho_{s}\left(\frac{2.65 \ g}{cm^{3}}\right) \end{pmatrix}$$

where:

$$\begin{split} \mathsf{K}_{\mathsf{d}}\!\left(\frac{\mathsf{L}}{\mathsf{k}\mathsf{g}}\right) &= \mathsf{K}_{\mathsf{oc}}\left(\frac{\mathsf{L}}{\mathsf{k}\mathsf{g}}\right) \times \mathsf{f}_{\mathsf{oc}}\left(\frac{0.006 \text{ g-carbon}}{\mathsf{g-soil}}\right) \text{, for organic compounds;} \\ \mathsf{K}_{\mathsf{d}} \text{ values for inorganic compounds are listed in the user guide.} \end{split}$$

The fraction of organic carbon ( $f_{oc}$ ) selected for this equation is 0.006. This is the default for surface soil identified in U.S. EPA 1996b, Sections 2.4.2 and 2.5.7, and represents the mean value for the top 0.3m of Class B soils. According to this source, soil organic carbon decreases rapidly with depth. Note that the default  $f_{oc}$  in section "4.8.3 Method 1 for SSL Determination" is 0.002, which is the default for subsurface soil from the same study.

#### Diffusivity in Water (cm<sup>2</sup>/s)

Diffusivity in water can be calculated from the chemical's molecular weight and density, using the following

correlation equation based on WATER9 (U.S. EPA, 2001 (PDF) (38 pp, 185 K)):

$$\mathsf{D}_{\mathsf{iW}}\left(\frac{\mathsf{cm}^2}{\mathsf{s}}\right) = 0.0001518 \times \left(\frac{\mathsf{T}\,{}^{\mathsf{o}}\mathsf{C}\,\mathsf{+}273.16}{298.16}\right) \times \left(\frac{\mathsf{MW}\left(\frac{\mathsf{g}}{\mathsf{mol}}\right)}{\rho\left(\frac{\mathsf{g}}{\mathsf{cm}^3}\right)}\right)^{-0.6}$$

where:

T typically =  $25^{\circ}$  C

If density is not available,

$$D_{iw}\left(\frac{cm^2}{s}\right) = 0.000222 \times (MW)^{-\left(\frac{2}{3}\right)}$$

If density is not available, diffusivity in water can be calculated using the correlation equation based on U.S. EPA (1987). The value for diffusivity in water must be greater than zero. No maximum limit is enforced.

#### Diffusivity in Air (cm<sup>2</sup>/s).

Diffusivity in air can be calculated from the chemical's molecular weight and density, using the following correlation equation based on WATER9 (U.S. EPA, 2001 (PDF) (38 pp, 185 K)). If density is not available, an alternate equation is provided.:

$$D_{ia}\left(\frac{cm^{2}}{s}\right) = \frac{0.00229 \times \left(T^{0}C+273.16\right)^{1.5} \times \sqrt{0.034 + \left(\frac{1}{MW\left(\frac{g}{mol}\right)}\right)} \times MW_{cor}}{\left(\left(\frac{MW\left(\frac{g}{mol}\right)}{2.5 \times p\left(\frac{g}{cm^{3}}\right)}\right)^{0.333} + 1.8\right)^{2}}$$

where:

T typically =  $25^{\circ}$  C

$$MW_{cor} = (1-0.000015 \times MW^2) \text{ If } MW_{cor} \text{ is less than } 0.4 \text{, then } MW_{cor} \text{ is set to } 0.4 \text{.}$$

If density is not available use,

$$D_{ia}\left(\frac{cm^2}{s}\right) = 1.9 \times \left(MW\left(\frac{g}{mol}\right)^{-}\left(\frac{2}{3}\right)\right)$$

For dioxins, furans, and dioxin-like PCBs always use,

$$D_{ia}\left(\frac{cm^2}{s}\right) = \left(\frac{154\left(\frac{g}{mol}\right)}{MW\left(\frac{g}{mol}\right)}\right)^{0.5} \times 0.068\left(\frac{cm^2}{s}\right)$$

For dioxins, furans, and dioxin-like PCBs, diffusivity in air should always be calculated from the molecular weight using the Graham's Law correlation equation based on <u>December 2003 NAS Review Draft Part I:</u> <u>Volume 3 (pg 4-38 )(PDF)</u> (148 pp, 1.9 MB). In this equation, the unknown diffusivity is solved by correlation to the known diphenyl diffusivity of 0.068 cm<sup>2</sup>/s and MW of 154 g/mol.

#### 4.9.5 Mass-limit Chronic Volatilization Factor (VF<sub>mlim</sub>)

This Equation presents a model for calculating mass-limit SSLs for the outdoor inhalation of volatiles. This model can be used only if the depth and area of contamination are known or can be estimated with confidence. This equation is presented in the <u>Soil Screening Guidance: User's Guide (PDF)</u> (89 pp, 863 K<u>F</u>) and the <u>Supplemental Soil Screening Guidance (PDF)</u> (187 pp, 2.2 MB).

Use of infinite source models to estimate volatilization can violate mass balance considerations, especially for small sources. To address this concern, the Soil Screening Guidance includes a model for calculating a mass-limit SSL that provides a lower limit to the SSL when the area and depth (i.e., volume) of the source are known or can be estimated reliably.

A mass-limit SSL represents the level of contaminant in the subsurface that is still protective when the entire volume of contamination volatilizes over the 26-year exposure duration and the level of contaminant at the receptor does not exceed the health-based limit.

To use mass-limit SSLs, determine the area and depth of the source, calculate both standard and mass-limit SSLs, compare them for each chemical of concern and select the higher of the two values.

Note that the equation requires a site-specific determination of the average depth of contamination in the source. Step 3, in the SSG, provides guidance for conducting subsurface sampling to determine source depth. Where the actual average depth of contamination is uncertain, a conservative estimate should be used (e.g., the maximum possible depth in the unsaturated zone). At many sites, the average water table depth may be used unless there is reason to believe that contamination extends below the water table. In this case SSLs do not apply and further investigation of the source in question is needed.

mass limit model for chronic exposure

$$\forall \mathsf{F}_{\mathsf{m}\mathsf{lim}}\left(\frac{\mathsf{m}_{\mathsf{air}}^3}{\mathsf{k}\mathsf{g}_{\mathsf{soil}}}\right) = \frac{\mathsf{Q}}{\mathsf{C}_{\mathsf{vol}}}\left(\frac{\left(\frac{\mathsf{g}}{\mathsf{m}^2 \cdot \mathsf{s}}\right)}{\left(\frac{\mathsf{k}\mathsf{g}}{\mathsf{m}^3}\right)}\right) \times \frac{\left[\mathsf{T}\left(\mathsf{year}\right) \times \left(3.15 \times 10^7 \left(\frac{\mathsf{s}}{\mathsf{year}}\right)\right)\right]}{\mathsf{p}_{\mathsf{b}}\left(\frac{\mathsf{m}\mathsf{g}}{\mathsf{m}^3}\right) \times \mathsf{d}_{\mathsf{s}}\left(\mathsf{m}\right) \times 10^6 \left(\frac{\mathsf{g}}{\mathsf{m}\mathsf{g}}\right)}$$

where:

$$\frac{Q}{C_{vol}} \left( \frac{\left(\frac{g}{m^2 - s}\right)}{\left(\frac{kg}{m^3}\right)} \right) = A \times exp\left[ \frac{\left(\ln A_s (acre) - B\right)^2}{C} \right]$$

# **4.9.6 Unlimited Source Subchronic Volatilization Factor for Construction Worker** (VF<sub>ulim-sc</sub>)

Equation 5-14 of the <u>supplemental soil screening guidance (PDF)</u> (187 pp, 2.2 MB) is appropriate for calculating the soil-to-air volatilization factor (VF<sub>ulim-sc</sub>) that relates the concentration of a contaminant in soil to the concentration in air resulting from volatilization. The equation for the subchronic dispersion factor for volatiles,  $Q/C_{sa}$ , is presented in Equation 5-15 of the <u>supplemental soil screening guidance (PDF)</u> (187 pp, 2.2 MB).  $Q/C_{sa}$  was derived using EPA's SCREEN3 dispersion model for a hypothetical site under a wide range of meteorological conditions. Unlike the Q/C values for the other scenarios, the  $Q/C_{sa}$  for the construction scenario's simple site-specific approach can be modified only to reflect different site sizes between 0.5 and 500 acres; it cannot be modified for climatic zone. Site managers conducting a detailed site-specific analysis for the construction scenario can develop a site-specific Q/C value by running the SCREEN3 model. Further details on the derivation of Q/C<sub>sa</sub> can be found in <u>Appendix E (PDF)</u> (42 pp, 779 K) of the <u>supplemental soil screening guidance (PDF)</u> (187 pp, 2.2 MB). If H' is not available, D<sub>A</sub> can still be calculated. H' changes when groundwater temperature changes. Please see section 4.9.9 of this user guide for H' determination at temperature other than 25°C.

unlimited source model for subchronic exposure

$$VF_{ulim-sc}\left(\frac{m_{air}^{3}}{kg_{soil}}\right) = \frac{Q}{C_{sa}}\left(\frac{\left(\frac{g}{m^{2}-s}\right)}{\left(\frac{kg}{m^{3}}\right)}\right) \times \frac{1}{F_{D}} \times \left[\frac{\left(3.14 \times D_{A}\left(\frac{cm^{2}}{s}\right) \times T_{c}(s)\right)^{1/2}}{2 \times \rho_{b}\left(\frac{1.5g}{cm^{3}}\right) \times D_{A}\left(\frac{cm^{2}}{s}\right)}\right] \times 10^{-4}\left(\frac{m^{2}}{cm^{2}}\right)$$

г

where:

$$\frac{Q}{C_{sa}} \left( \frac{\left(\frac{g}{m^2 \cdot s}\right)}{\left(\frac{kg}{m^3}\right)} \right) = A \times exp \left[ \frac{\left(\ln A_s \left(a \operatorname{cre}\right) \cdot B\right)^2}{C} \right]$$

$$T_c \left(30240000 \text{ s}\right) = ED_{cw} \left(1 \text{ year}\right) \times EW_{cw} \left(\frac{50 \text{ weeks}}{\text{ year}}\right) \times \left(\frac{7 \text{ days}}{\text{ week}}\right) \times \left(\frac{24 \text{ hours}}{\text{ day}}\right) \times \left(\frac{3600 \text{ s}}{\text{ hour}}\right)$$

$$F_D \left(0.18584\right) = 0.1852 + \left(5.3537 / t_c\right) + \left(-9.6318 / t_c^2\right)$$

$$t_c \left(8400 \text{ hours}\right) = ED_{cw} \left(1 \text{ year}\right) \times EW_{cw} \left(\frac{50 \text{ weeks}}{\text{ year}}\right) \times \left(\frac{7 \text{ days}}{\text{ week}}\right) \times \left(\frac{24 \text{ hours}}{\text{ day}}\right)$$

$$C_C \left(\frac{cm^2}{L_{soil}}\right) = \left(\theta_a \left(\frac{L_{air}}{L_{soil}}\right)^{10/3} \times D_{ia} \left(\frac{cm^2}{s}\right) \times H' + \theta_w \left(\frac{0.15 \text{ L}_{water}}{L_{soil}}\right)^{10/3} \times D_{iw} \left(\frac{cm^2}{s}\right)\right) / n^2 \left(\frac{L_{pore}}{L_{soil}}\right)$$

$$\mathsf{D}_{\mathsf{A}}\left(\frac{\mathsf{cm}^{2}}{\mathsf{s}}\right) = \frac{1}{\mathsf{p}_{\mathsf{b}}\left(\frac{1.5\mathsf{g}}{\mathsf{cm}^{3}}\right) \times \mathsf{K}_{\mathsf{d}}\left(\frac{\mathsf{cm}^{3}}{\mathsf{g}}\right) + \theta_{\mathsf{w}}\left(\frac{0.15 \,\mathsf{L}_{\mathsf{water}}}{\mathsf{L}_{\mathsf{soil}}}\right) + \theta_{\mathsf{a}}\left(\frac{\mathsf{L}_{\mathsf{air}}}{\mathsf{L}_{\mathsf{soil}}}\right) \times \mathsf{H}'}$$
where:

where:

$$\theta_{a}\left(\frac{L_{air}}{L_{soil}}\right) = n\left(\frac{L_{pore}}{L_{soil}}\right) - \theta_{w}\left(\frac{0.15 L_{water}}{L_{soil}}\right) \text{ and } n\left(\frac{L_{pore}}{L_{soil}}\right) = 1 - \left(\begin{array}{c} \rho_{b}\left(\frac{1.5 g}{cm^{3}}\right) \\ \rho_{s}\left(\frac{2.65 g}{cm^{3}}\right) \end{array}\right)$$

and:

$$K_{d}\left(\frac{L}{kg}\right) = K_{oc}\left(\frac{L}{kg}\right) \times f_{oc}\left(\frac{0.006 \text{ g-carbon}}{\text{g-soil}}\right)$$
, for organic compounds;

 ${\rm K}_{\rm d}$  values for inorganic compounds are listed in the user guide.

#### 4.9.7 Mass-limit Subchronic Volatilization Factor for Construction Worker (VF<sub>mlim-sc</sub>)

Because the equations developed to calculate SSLs for the inhalation of volatiles outdoors assume an infinite source, they can violate mass-balance considerations, especially for small sources. To address this concern, a mass-limit SSL equation for this pathway may be used (Equation 5-17 of the supplemental soils screening guidance (PDF) (187 pp, 2.2 MB)). This equation can be used only when the volume (i.e., area and depth) of the contaminated soil source is known or can be estimated with confidence. As discussed above, the simple

site-specific approach for calculating construction scenario SSLs uses the same emission model for volatiles as that used in the residential and non-residential scenarios. However, the conservative nature of this model (i.e., it assumes all contamination is at the surface) makes it sufficiently protective of construction worker exposures to volatiles.

mass limit model for subchronic exposure

$$VF_{m \text{ lim-sc}}\left(\frac{m_{air}^{3}}{kg_{soil}}\right) = \frac{Q}{C_{sa}}\left(\frac{\left(\frac{g}{m^{2}-s}\right)}{\left(\frac{kg}{m^{3}}\right)}\right) \times \frac{1}{F_{D}} \times \frac{1}{\rho_{b}\left(\frac{1.5 \text{ Mg}}{m^{3}}\right) \times d_{s}(m) \times 10^{6}\left(\frac{g}{mg}\right)}$$

where:

$$\frac{Q}{C_{sa}} \left( \frac{\left(\frac{g}{m^2 \cdot s}\right)}{\left(\frac{kg}{m^3}\right)} \right) = A \times exp\left[ \frac{\left(\ln A_s (a \operatorname{cre}) \cdot B\right)^2}{C} \right]$$

$$T_c (30240000 \text{ s}) = ED_{cW} (1 \text{ year}) \times EW_{cW} \left( \frac{50 \text{ weeks}}{\text{ year}} \right) \times \left( \frac{7 \text{ days}}{\text{ week}} \right) \times \left( \frac{24 \text{ hours}}{\text{ day}} \right) \times \left( \frac{3600 \text{ s}}{\text{ hour}} \right)$$

$$F_D (0.18584) = 0.1852 + (5.3537 / t_c) + (-9.6318 / t_c^2)$$

$$t_c (8400 \text{ hours}) = ED_{cW} (1 \text{ year}) \times EW_{cW} \left( \frac{50 \text{ weeks}}{\text{ year}} \right) \times \left( \frac{7 \text{ days}}{\text{ week}} \right) \times \left( \frac{24 \text{ hours}}{\text{ day}} \right)$$

#### 4.9.8 Dermal Contact with Water Supporting Equations

• EPD = Effective Predictive Domain. The EPD is an area on a X/Y plot that symbolizes 95% statistical confidence levels of a regression equation to accurately estimate a dermal permeability constant ( $K_p$ ). Only if a chemical is within the EPD, will a  $K_p$  be estimated and the dermal exposure to water exposure route quantified. The EPD is determined by investigating the predictive power of a regression equation using MW and log Kow values for a compound. If the intersection of the values falls within the designated plotted area, the chemical is determined to be in the EPD and a  $K_p$  is estimated. The boundaries of MW and log  $K_{ow}$  for the regression equation are presented below. The EPD is depicted in <u>RAGS Part E</u> in Appendix A; Exhibit A-1.

-0.06831  $\leq 5.103 \times 10^{-4}$  MW + 0.5616 log  $K_{ow} \leq 0.5577$  and -0.3010  $\leq$  -5.103  $\times 10^{-4}$  MW + 0.05616 log  $K_{ow} \leq 0.1758$ 

FA = fraction absorbed water. The FA is described in <u>RAGS Part E</u> in Appendix A. The FA term should be applied to account for the loss of chemical due to the desquamation of the outer skin layer and a corresponding reduction in the absorbed dermal dose. To determine FA vales for the RSLs, the following regression analysis was performed. This analysis builds on the RAGS Part E data. logds=(-2.805063-0.0056118\*mw); dsclc=10\*\*logds; dsc = dsclc\*&lsc; B = kp\*(mw\*\*0.5)/2.6; tau = &lsc\*\*2/(6\*dsc);

logB=log10(B);

 $\label{eq:states} \begin{array}{l} \mbox{logtau} = \log 10(\mbox{tau}) \ ; \\ \mbox{if } B <= 0.1 \ \mbox{then } FA \mbox{calc} = 0.9589849087 \ ..0163393790* \mbox{logB} \ ..1451565908* \mbox{logtau} \\ \ ..0534664095* \mbox{logB}* \mbox{logtau} \ ; \\ \mbox{else if } B > 0.1 \ \mbox{and } B <= 1 \ \mbox{then } FA \mbox{calc} = 1.051232292 \ + \ 0.091016187* \mbox{logB} \ - 0.286735467* \mbox{logtau} \\ \ ..180504367* \mbox{logB}* \mbox{logtau} \ ; \\ \mbox{else if } B > 1 \ \mbox{then } FA \mbox{calc} = 0.992336792 \ + \ 0.479643809* \mbox{logB} \ - 0.114381522* \mbox{logtau} \\ \ ..1263647642* \mbox{logB}* \mbox{logtau} \ ; \\ \ FA = \mbox{ifn}(FA \mbox{calc} >= 1,1, \mbox{round}(FA \mbox{calc}, 0.1)); \\ \ \mbox{if } FA < 0 \ \mbox{then } FA = 0 \ ; \end{array}$ 

• B = Dimensionless ratio of the permeability coefficient of a compound through the stratum corneum relative to its permeability coefficient across the viable epidermis (ve)

$$B = \frac{K_p\left(\frac{cm}{hour}\right)}{K_{p,ve}\left(\frac{cm}{hour}\right)} \approx K_p\left(\frac{cm}{hour}\right) \times \frac{\sqrt{MW\left(\frac{g}{mole}\right)}}{2.6} \text{ (as an approximation)}$$

where:

$$K_{p,ve}\left(\frac{cm}{hour}\right) = \frac{K_{ew} \times D_{e}\left(\frac{cm^{2}}{hour}\right)}{L_{e}(cm)}$$

where:

$$\begin{split} & \mathsf{K}_{ew} = 1 \text{ (assuming epidermis behaves essentially as water);} \\ & \mathsf{L}_e = 10^{-2} \text{ (cm);} \\ & \mathsf{D}_e = \frac{7.1 \times 10^{-6} \left(\frac{\mathrm{cm}^2}{\mathrm{sec}}\right)}{\sqrt{\mathsf{MW}\left(\frac{\mathsf{g}}{\mathsf{mole}}\right)}} \text{ (assumes } \mathsf{D}_e = 10^{-6} \left(\frac{\mathrm{cm}^2}{\mathrm{sec}}\right) \text{ when } \mathsf{MW} = 50) \end{split}$$

•  $t^* = \text{Time to reach steady-state (hours)} = 2.4 \tau_{\text{event}}$ If B  $\leq 0.6$ , then  $t^*(\text{hours}) = 2.4 \times \tau_{\text{event}} \left(\frac{\text{hours}}{\text{event}}\right)$ or,

If B > 0.6, then 
$$t^*(hours) = 6 \times r_{event}\left(\frac{hours}{event}\right) \times \left(b \cdot \sqrt{b^2 \cdot c^2}\right)$$

where:

$$b = \frac{2 \times (1+B)^2}{\pi} - c$$
 and  $c = \frac{1+3 \times B + 3 \times B^2}{3 \times (1+B)}$ 

•  $\tau_{\text{event}} = \text{Lag time per event (hours/event)}$ 

$$r_{event}\left(\frac{hours}{event}\right) = \frac{l_{sc}^{2}(cm)}{6 \times D_{sc}\left(\frac{cm^{2}}{hour}\right)}$$

where:

$$\log \frac{\mathsf{D}_{\mathsf{sc}}\left(\frac{\mathsf{cm}^2}{\mathsf{hour}}\right)}{\mathsf{I}_{\mathsf{sc}}\left(\mathsf{cm}\right)} = -2.80 - 0.0056 \times \mathsf{MW}\left(\frac{\mathsf{g}}{\mathsf{mole}}\right) \text{ or } \frac{\mathsf{D}_{\mathsf{sc}}\left(\frac{\mathsf{cm}^2}{\mathsf{hr}}\right)}{\mathsf{I}_{\mathsf{sc}}\left(\mathsf{cm}\right)} = 10^{\left(-2.80 - 0.0056 \times \mathsf{MW}\left(\frac{\mathsf{g}}{\mathsf{mole}}\right)\right)}$$

thus:

$$I_{sc}(cm) = \frac{10^{\left(-2.80 - 0.0056 \times MW\left(\frac{g}{mole}\right)\right)}}{D_{sc}\left(\frac{cm^2}{hour}\right)} \text{ and } D_{sc}\left(\frac{cm^2}{hour}\right) = I_{sc}(cm) \times 10^{\left(-2.80 - 0.0056 \times MW\left(\frac{g}{mole}\right)\right)}$$

#### 4.9.9 H' Determination at Temperature Other Than 25°C

In site-specific mode for soil and soil to groundwater land uses, users are given the option to the change groundwater temperature from the default of 25°C to a site-specific value. Since the unitless Henry's Law Constant (H') is derived based on the partial pressure of a gas in equilibrium with a liquid and the equilibrium changes when temperature changes, H' is changed to reflect the equilibrium at the given temperature. The equation below illustrates how H' is derived when groundwater temperature is changed. An EPA Fact Sheet describing the process can be found at

https://www.epa.gov/vaporintrusion/fact-sheet-correcting-henrys-law-constant-temperature.

$$H' = \left(\frac{\exp\left[-\frac{\Delta H_{v,w}\left(\frac{cal}{mol}\right)}{R_{c}\left(1.9872 \text{ cal/mol-K}\right)} \times \left(\frac{1}{T_{w}\left(K\right)} - \frac{1}{T_{R}\left(298.15 \text{ K}\right)}\right)\right] \times HLC\left(atm \cdot m^{3} / mol\right)}{R\left(8.205 \text{ E-05 atm} \cdot m^{3} / mol \cdot K\right) \times T_{w}\left(K\right)}\right)$$

where:

and:

$$\Delta H_{v,w}\left(\frac{cal}{mol}\right) = \Delta H_{v,b}\left(\frac{cal}{mol}\right) \times \left[\frac{1 - T_{w}(k)/T_{c}(k)}{1 - T_{b}(k)/T_{c}(k)}\right]^{\eta}$$
where:

$$\begin{split} & \text{IF } \left(\frac{T_b}{T_c}\right) < 0.57 \text{ , then: } \eta = 0.3; \\ & \text{IF } \left(\frac{T_b}{T_c}\right) > 0.71 \text{ , then: } \eta = 0.41; \\ & \text{IF } 0.57 < \left(\frac{T_b}{T_c}\right) \le 0.71 \text{ , then: } \eta = \left(0.74 \times \left(\frac{T_b}{T_c}\right) - 0.116\right) \end{split}$$

## **5.** Special Considerations

Most of the SLs are readily derived by referring to the above equations. However, there are some cases for which the standard equations do not apply and/or external adjustments to the SLs are recommended. These special case chemicals are discussed below.

#### 5.1 Cadmium

IRIS presents an oral "water" RfD for cadmium for use in assessment of risks to water of 0.0005 mg/kg-day. IRIS also presents an oral "food" RfD for cadmium for use in assessment of risks to soil and biota of 0.001 mg/kg-day. The SLs for Cadmium are based on the appropriate oral RfD based on the media. The "water" RfD is slightly more conservative (by a factor of 2) than the RfD for "food" and it could be argued that the more conservative RfD should be used to develop screening levels. RAGS Part E, in Exhibit 4-1, presents a GIABS for soil of 2.5% and for water of 5%.

#### **5.2 Lead**

EPA has no consensus RfD or SFO for inorganic lead, so it is not possible to calculate SLs as we have done for other chemicals. EPA considers lead to be a special case because of the difficulty in identifying the classic "threshold" needed to develop an RfD.

EPA therefore evaluates lead exposure by using blood-lead modeling, such as the Integrated Exposure-Uptake Biokinetic Model (IEUBK). The EPA Office of Solid Waste has also released a detailed directive on risk assessment and cleanup of residential soil lead. The directive recommends that soil lead levels less than 400 mg/kg are generally safe for residential use. Above that level, the document suggests collecting data and modeling blood-lead levels with the IEUBK model. For the purposes of screening, therefore, 400 mg/kg is recommended for residential soils. For water, we suggest 15  $\mu$ g/L (the EPA Action Level in water), and for air, the National Ambient Air Quality Standard of 0.15  $\mu$ g/m<sup>3</sup>. An updated screening level for soil lead at commercial/industrial (i.e., non-residential) sites of 800 part per million (ppm) is based on a recent analysis of the combined phases of the National Health and Nutrition Examination Survey (NHANES III) that choose a cleanup goal protective for all subpopulations. More information can be found <u>here</u>.

However, caution should be used when both water and soil are being assessed. The IEUBK model shows that if the average soil concentration is 400 mg/kg, an average tap water concentration above 5  $\mu$ g/L would yield more than 5% of the population above a 10  $\mu$ g/dL blood-lead level. If the average tap water concentration is 15  $\mu$ g/L, an average soil concentration greater than 250 mg/kg would yield more than 5% of the population above a 10  $\mu$ g/dL blood-lead level.

EPA uses a second Adult Lead Model to estimate SLs for an industrial setting. This SL is intended to protect a fetus that may be carried by a pregnant female worker. It is assumed that a cleanup goal that is protective of a fetus will also afford protection for male or female adult workers. The model equations were developed to calculate cleanup goals such that the fetus of a pregnant female worker would not likely have an unsafe concentration of lead in blood.

For lead in soil, the default values for absolute bioavailability (ABA) in the IEUBK Model for Lead in Children are 0.3 for soil and dust and 0.5 for food and water. This corresponds to an RBA for soil of 0.6 (ABA\_soil / ABA\_water = 0.6). It's important to note that the ABA values in the IEUBK model are central estimates and the oral RBA at any given site may be higher or lower than the default oral RBA for lead. For this reason, and because it provides a more comprehensive characterization of exposure at a site, the TRW recommends using EPA SW846 Method 1340 to estimate site-specific RBA. Guidance related to these topics can be found in the <u>Soil Bioavailability at Superfund Site Guidance</u>. Documents <u>OSWER 9285.7-77</u> both contain the value for lead.

For more information on EPA's lead models and other lead-related topics, please go to <u>Addressing Lead at</u> <u>Superfund Sites</u>.

#### 5.3 Manganese

The IRIS RfD (0.14 mg/kg-day) includes manganese from all sources, including diet. The author of the IRIS assessment for manganese recommended that the dietary contribution from the normal U.S. diet (an upper limit of 5 mg/day) be subtracted when evaluating non-food (e.g., drinking water or soil) exposures to manganese, leading to a RfD of 0.071 mg/kg-day for non-food items. The explanatory text in IRIS further recommends using a modifying factor of 3 when calculating risks associated with non-food sources due to a number of uncertainties that are discussed in the IRIS file for manganese, leading to a RfD of 0.024 mg/kg-day. This modified RfD has been used in the derivation of some manganese screening levels for soil and water. For more information regarding the Manganese RfD, users are advised to contact the author of the IRIS assessment on Manganese.

#### **5.4 Vanadium Compounds**

The oral RfD toxicity value for Vanadium, used in this website, is derived from the IRIS oral RfD for Vanadium Pentoxide by factoring out the molecular weight (MW) of the oxide ion. Vanadium Pentoxide  $(V_20_5)$  has a molecular weight of 181.88. The two atoms of Vanadium contribute 56% of the MW. Vanadium Pentoxide's oral RfD of 9E-03 mg/kg-day multiplied by 56% gives a Vanadium oral RfD of 5.04E-03 mg/kg-day.

#### 5.5 Uranium

The "Uranium Soluble Salts" RSL uses the ATSDR intermediate MRL of 2E-04 mg/kg-day instead of the IRIS oral RfD of 3E-03 mg/kg-day. This is a deviation from the typical RSL toxicity hierarchy. This deviation was justified by the 2003 hierarchy <u>memo (PDF)</u> (4 pp, 25 K) that acknowledges and "recognizes that EPA should use the best science available on which to base risk assessments." In December 2016, the EPA Office of Superfund Remediation and Technology Innovation (OSRTI) announced its determination that the ATSDR intermediate MRL generally reflects a better scientific basis for assessing the chronic health risks of soluble uranium than the RfD currently available in IRIS." The rationale for this determination is summarized in an accompanying <u>memorandum (PDF)</u> (11 pp, 2.5 MB), which recommends use of the ATSDR intermediate MRL for assessing chronic and subchronic human exposures at Superfund sites nationwide.

#### 5.6 Chromium (VI)

It is recommended that valence-specific data for chromium be collected whenever possible when chromium is likely to be an important contaminant at a site, and when hexavalent chromium (Cr (VI)) may exist. For Cr (VI), IRIS shows an air inhalation unit risk (IUR) of 1.2E-2 per ( $\mu$ g/m<sup>3</sup>). While the exact ratio of Cr(VI) to Cr(III) in the data used to derive the IRIS IUR value is not known, it is likely that both Cr(VI) and Cr(III) were present. The RSLs, calculated using the IRIS IUR, assume that the Cr(VI) to Cr(III) ratio is 1:6. Because of various sources of uncertainty, this assumption may overestimate or underestimate the risk calculated. Users are invited to review the document "Toxicological Review of Hexavalent Chromium" in support of the summary information on Cr(VI) on IRIS to determine whether they believe this ratio applies to their site and to consider consulting with an EPA regional risk assessor. The uncertainty section of the risk assessment may want to address the potential for overestimating or underestimating the risk and provide quantitative analysis by deriving different IUR values based on different Cr(VI) to Cr(III) ratios from more recent studies.

In the RSL Table, the Cr(VI) specific value (assuming 100% Cr(VI)) is derived by multiplying the IRIS Cr(VI) value by 7. This is considered to be a health-protective assumption, and is also consistent with the State of California's interpretation of the Mancuso study that forms the basis for their estimated cancer potency of Cr(VI).

If you are working on a chromium site, you may want to contact the appropriate regulatory officials in your region to determine what their position is on this issue.

The Maximum Contaminant Level (MCL) of 100  $\mu$ g/L for "Chromium (total)", from the EPA's <u>MCL</u> listing is applied to the "Chromium, Total" analyte on this website.

The State of California Environmental Protection Agency (CalEPA) determined that Cr(VI) by ingestion is likely to be carcinogenic in humans. CalEPA derived an oral cancer slope factor, based on a dose-related increase of tumors of the small intestine in male mice conducted by the <u>National Toxicology Program (PDF)</u> (162 pp, 1.9 MB). CalEPA determined that Cr(VI) was carcinogenic by mutagenic by mode of action.

EPA's <u>Office of Pesticide Programs (OPP) (PDF)</u> (23 pp, 414 K) made a determination that Cr(VI) has a mutagenic mode of action for carcinogenesis in all cells regardless of type, following administration via drinking water. OPP recommended that Age-Dependent Adjustment Factors (ADAFs) be applied when assessing cancer risks from early-life exposure (< 16 years of age). This determination was reviewed by OPP's Cancer Assessment Review Committee and published in a peer review journal (PDF) (23 pp, 414 K)).

Therefore, the RSL workgroup adopted the Tier III CalEPA value and the OPP recommendation with respect to mutagenicity. More recently, in 2011, external peer reviewers provided input on the EPA's Office of Research and Development Integrated Risk Information System draft <u>Toxicological Review of Hexavalent</u> <u>Chromium</u>. The majority of reviewers questioned the evidence used to support a mutagenic mode of action for carcinogenesis for Cr(VI). Furthermore, in 2011 California Environmental Protection Agency finalized its drinking water Public Health Goal for Cr(VI). <u>CalEPA's Technical Support Document</u> concluded in numerous studies that Cr(VI) is both genotoxic and mutagenic.

Therefore, the RSL workgroup acknowledges that there is uncertainty associated with the assessment of hexavalent chromium. However, no updated consensus IRIS assessment (Tier I) has yet appeared, and chromium is still under review by the IRIS program. With respect to RSLs, the more health-protective approach of applying ADAFs for early life exposure via ingestion, dermal and inhalation was used to calculate screening levels for all exposure pathways. Application of ADAFs for all exposure pathways results in more health-protective screening levels.

As always, consult EPA toxicologists in the Superfund program of the regional office when developing site specific screening levels.

#### 5.7 Aminodinitrotoluenes

The IRIS oral RfD of 2E-03 mg/kg-day for 2,4-Dinitrotoluene is used as a surrogate for 2-Amino-4,6-Dinitrotoluene and 4-Amino-2,6-Dinitrotoluene.

#### **5.8 PCBs**

Aroclor 1016 is considered "lowest risk" and assigned appropriate toxicity values. All other Aroclors are assigned the high risk toxicity values.

#### 5.9 Xylenes

The IRIS oral RfD of 2E-01 mg/kg-day for xylene, mixture is used as a surrogate for the 3 xylene congeners. The earlier RfD values for some xylene isomers were withdrawn from our electronic version of HEAST. Also, the IRIS inhalation RfC of 1E-01 mg/m<sup>3</sup> for xylene, mixture is used as a surrogate for the 3 xylene congeners.

#### 5.10 Arsenic

Arsenic screening levels for ingestion of soil are now calculated with the default <u>relative bioavailability</u> <u>factor</u> (RBA) of 0.6. The RBA can be adjusted using the calculator in site-specific/user-provided mode the same way toxicity values can be changed. The RBA for soil ingestion is shown in the calculator output. The 2012 document, <u>Compilation and Review of Data on Relative Bioavailability of Arsenic in Soil (PDF) (58 pp, 474 K) provides supporting information.</u>

In 2017, the EPA has released a standard operating procedure for an in vitro bioaccessibility assay for arsenic in soil. The in vitro method for predicting oral RBA of arsenic in soil (EPA SW846 Method 1340) has been validated, and it is now recommended that the in vitro method be used to estimate site-specific RBA, when site-specific RBA is needed. This method can provide a more comprehensive characterization of RBA variability at the site. The default value represents the 95th percentile of many arsenic soil samples, and it is expected that the site-specific RBA will be less than 0.6 at most sites, which means that the default should be protective for screening. Site-specific RBAs derived with the in vitro method should be verified with your Regional Risk Assessor. Guidance related to these topics can be found in the <u>Soil Bioavailability at</u> <u>Superfund Sites: Guidance</u>.

Absolute bioavailability can be thought of as the <u>absorption fraction (PDF)</u> (20 pp, 133 K). Relative bioavailability accounts for differences in the bioavailability of a contaminant between the medium of exposure (e.g., soil) and the media associated with the toxicity value (e.g., the arsenic RfD and SFO are derived from drinking water studies). The 60% oral RBA for arsenic in soil is empirically-based. It represents an upper-bound estimate from numerous studies where the oral RBA of soil-borne arsenic in samples collected from across the U.S. was experimentally determined against the water-soluble form. This RBA does not apply to dermal exposures to arsenic in soil for which the absorbed dose is calculated using a dermal absorption fraction (ABS<sub>d</sub>) of 0.03 (Exhibit 3-4 of USEPA, 2004).

#### 5.11 Total Petroleum Hydrocarbons (TPHs)

The six TPH fractions were assigned representative compounds for determination of toxicity values and chemical-specific parameters to calculate RSLs. The <u>PPRTV (PDF)</u> (60 pp, 678 K) paper was the principal source for the derivation of these values.

The carbon ranges and representative compounds are listed in the table below. An average of the chemicalspecific parameters for 2-methylnaphthalene and naphthalene was calculated for the medium aromatic fraction.

TPH Fractions	Number of Carbons	Equivalent Carbon Number Index	Representative Compound (RfD/RfC)
Low aliphatic	C5-C8	EC5-EC8	n-hexane
Medium aliphatic	C9-C18	EC>8-EC16	hydrocarbon streams*
High aliphatic	C19-C32	EC>16-EC35	white mineral oil
Low aromatic	C6-C8	EC6-EC<9	benzene
Medium aromatic	C9-C16	EC9-EC<22	2-methylnaphthalene/naphthalene

TPH	Number of	Equivalent Carbon	Representative Compound
Fractions	Carbons	Number Index	(RfD/RfC)
High aromatic	C17-C32	EC>22-EC35	fluoranthene

\*Medium aliphatic representative compound was not listed in the PPRTV paper so n-nonane was selected by the RSL work-group to represent the chemical-specific parameters.

#### 5.12 Soil Saturation Limit (Csat)

The soil saturation concentration,  $C_{sat}$ , corresponds to the contaminant concentration in soil at which the absorptive limits of the soil particles, the solubility limits of the soil pore water, and saturation of soil pore air have been reached. Above this concentration, the soil contaminant may be present in free phase (i.e., nonaqueous phase liquids (NAPLs) for contaminants that are liquid at ambient soil temperatures and pure solid phases for compounds that are solid at ambient soil temperatures).  $C_{sat}$  is not calculated for chemicals that are solid at ambient soil temperatures. The following decision criteria was established from SSL guidance, Table C-3: if melting point is less than 20 °C, chemical is a liquid; if melting point is above 20 °C, chemical is solid.

Equation 4-10 is used to calculate  $C_{sat}$  for each volatile contaminant. As an update to RAGS HHEM, Part B (USEPA 1991a), this equation takes into account the amount of contaminant that is in the vapor phase in soil in addition to the amount dissolved in the soil's pore water and sorbed to soil particles. If H' is not available,  $C_{sat}$  can still be calculated.

Chemical-specific  $C_{sat}$  concentrations must be compared with each VF-based inhalation SL because a basic principle of the SL volatilization model is not applicable when free-phase contaminants are present. How these cases are handled depends on whether the contaminant is liquid or solid at ambient temperatures. Liquid contaminants that have a VF-based inhalation SL that exceeds the  $C_{sat}$  concentration are set equal to  $C_{sat}$ . For organic compounds that are solids (e.g., PAHs), soil screening decisions are based on the appropriate SLs for other pathways of concern at the site (e.g., ingestion). Note, that the SLs presented for soil inhalation in the RSL tool combine the VF and the PEF components. If the  $C_{sat}$  substitution is performed, the whole SL is replaced and not just the VF component.

The RSL tables and the default calculator settings do not substitute  $C_{sat}$  for risk-based calculations. If the risk-based concentration exceeds  $C_{sat}$ , the resulting SSL concentration may be overly protective. This is because the dissolved, absorbed and vapor concentrations cease to rise linearly as soil concentration increases above the  $C_{sat}$  level (pure product or nonaqueous phase liquid (NAPL) is present). The SSL model used in the RSL calculator is not a four phase model. If a NAPL is present at your site more sophisticated models may be necessary. The calculator, if operated in site-specific mode, will give the option to apply the  $C_{sat}$  substitution rule. H' changes when groundwater temperature changes. Please see section 4.9.9 of this user guide for H' determination at temperature other than 25°C.

$$C_{sat}\left(\frac{mg}{kg}\right) = \frac{S\left(\frac{mg}{L}\right)}{\rho_{b}\left(\frac{kg}{L}\right)} \times \left(K_{d}\left(\frac{L}{kg}\right) \times \rho_{b}\left(\frac{kg}{L}\right) + \theta_{w}\left(\frac{L_{water}}{L_{soil}}\right) + H' \times \theta_{a}\left(\frac{L_{air}}{L_{soil}}\right)\right)$$

where:

$$K_{d}\left(\frac{L}{kg}\right) = K_{oc}\left(\frac{L}{kg}\right) \times f_{oc}\left(\frac{0.006 \text{ g-carbon}}{\text{g-soil}}\right)$$
, for organic compounds;

 ${\rm K}_{\rm d}$  values for inorganic compounds are listed in the user guide,

$$\theta_{a}\left(\frac{L_{air}}{L_{soil}}\right) = n\left(\frac{L_{pore}}{L_{soil}}\right) - \theta_{w}\left(\frac{L_{water}}{L_{soil}}\right)$$

and:

$$n = 1 - \left(\frac{\rho_{b}\left(\frac{kg}{L}\right)}{\rho_{s}\left(\frac{kg}{L}\right)}\right)$$

#### 5.13 SL Theoretical Ceiling Limit

The ceiling limit of  $10^{+5}$  mg/kg is equivalent to a chemical representing 10% by weight of the soil sample. At this contaminant concentration (and higher), the assumptions for soil contact may be violated (for example, soil adherence and wind-borne dispersion assumptions) due to the presence of the foreign substance itself.

The RSL tables and the default calculator settings do not substitute the theoretical ceiling limit for risk-based calculations but they do indicate if the resulting RSL has exceeded the theoretical ceiling limit in the key. The calculator, if operated in site-specific mode, will give the option to apply the theoretical ceiling limit.

#### 5.14 Target Risk

With the exceptions described previously, SLs are chemical concentrations that correspond to fixed levels of risk (i.e., either a one-in-one million [10<sup>-6</sup>] for cancer risk or a noncarcinogenic hazard quotient of 1) in soil, air, and water. In noncarcinogenic equations, THQ represents the target hazard quotient and is used for individual substances or exposure routes like: ingestion, dermal, and inhalation. The target hazard index (THI), is the target across multiple substances or exposure routes. In most cases, where a substance causes both cancer and noncancer (systemic) effects, the 10<sup>-6</sup> cancer risk will result in a more stringent criteria and consequently this value is presented in the printed copy of the Table. SL concentrations that equate to a 10<sup>-6</sup> cancer risk are indicated by 'ca' in the calculator and 'c' in the generic tables. SL concentrations that equate to a hazard quotient (HQ) of 1 for noncarcinogenic concerns are indicated by 'nc' in the calculator and 'n' in the generic tables.

If the SLs are to be used for site screening, it is recommended that both cancer and noncancer-based SLs be used. Both carcinogenic and noncarcinogenic values may be obtained in the Supporting Tables.

Some users of this SL Table may plan to multiply the cancer SL concentrations by 10 or 100 to set 'action levels' for triggering remediation or to set less stringent cleanup levels for a specific site after considering non-risk-based factors such as ambient levels, detection limits, or technological feasibility. This risk management practice recognizes that there may be a range of values that may be 'acceptable' for carcinogenic risk (EPA's risk management range is one-in-a-million [10<sup>-6</sup>] to one-in-ten thousand [10<sup>-4</sup>]). However, this practice could lead one to overlook serious noncancer health threats and it is strongly recommended that the user consult with a toxicologist or regional risk assessor before doing this. Carcinogens are indicated by an asterisk ('\*') in the SL Table where the noncancer SLs would be exceeded if the cancer value that is displayed is multiplied by 100. ('\*\*') indicate that the noncancer values would be exceeded if the cancer SL were multiplied by 10. There is no range of 'acceptable' noncarcinogenic 'risk' for CERCLA sites. Therefore, the noncancer SLs should not be multiplied by 10 or 100 when setting final cleanup criteria. In the rare case where noncancer SLs are more stringent than cancer SLs set at one-in-one-million risk, a similar approach has been applied (e.g. 'max').

SL concentrations in the printed Table are risk-based, but for soil there are two important exceptions: (1) for several volatile chemicals, SLs may exceed the soil saturation level ('sat') and (2) SLs may exceed a non-risk based 'ceiling limit' concentration of  $10^{+5}$  mg/kg ('max') for relatively less toxic inorganic and semivolatile contaminants. For more information on the 'sat' value in the SL Table, please see the discussion in Section 5.11. For more information on the 'max' value in the SL Table, please see the discussion in Section 5.13.

With respect to applying a 'ceiling limit' for chemicals other than volatiles, it is recognized that this is not a universally accepted approach. Some within the agency argue that all values should be risk-based to allow for scaling (for example, if the risk-based SL is set at a hazard quotient = 1.0, and the user would like to set the hazard quotient to 0.1 to take into account multiple chemicals, then this is as simple as multiplying the risk-based SL by 1/10th). If scaling is necessary, SL users can do this simply by referring to the Supporting Tables at this website where risk-based soil concentrations are presented for all chemicals.

In spite of the fact that applying a ceiling limit is not a universally accepted approach, this table applies a 'max' soil concentration to the SL Table for the following reasons:

- Risk-based SLs for some chemicals in soil exceed unity (>1,000,000 mg/kg), which is not possible.
- The ceiling limit of 10<sup>+5</sup> mg/kg is equivalent to a chemical representing 10% by weight of the soil sample. At this contaminant concentration (and higher), the assumptions for soil contact may be violated (for example, soil adherence and wind-borne dispersion assumptions) due to the presence of the foreign substance itself.
- SLs currently do not address short-term exposures (e.g., pica children and construction workers). Although extremely high soil SLs are likely to represent relatively non-toxic chemicals, such high values may not be justified if in fact more toxicological data were available for evaluating short-term and/or acute exposures.

#### 5.15 Screening Sites with Multiple Contaminants

The screening levels in the tables are calculated under the assumption that only one contaminant is present. Users needing to screen sites with multiple contaminants should consult with their regional risk assessors. The following sections describe how target risks can be changed to screen against multiple contaminants and how the ratio of concentration to RSL can be used to estimate total risk.

#### 5.15.1 Adjusting Target Risk and Target Hazard Quotient

When multiple contaminants are present at a site the target hazard quotient (THQ) may be modified. The following options are among the commonly used methods to modify the THQ:

The <u>calculator</u> on this website can be used to generate SLs based on any THQ or target cancer risk (TR) deemed appropriate by the user. The THQ input to the calculator can be modified from the default of 1. How much it should be modified is a user decision, but it could be based upon the number of contaminants being screened together. For example, if one is screening two contaminants together, then the THQ could be modified to 0.5. If ten contaminants are being screened together, then the THQ could be modified to 0.1. The above example weights each chemical equally; it is also possible to weight the chemicals unequally, as long as the total risk meets the desired goal. The decision of how to weight the chemicals is likely to be site-specific, and it is recommended that this decision be made in consultation with the regional risk assessor.

Note that when the TR or THQ is altered, the relationship between cancer-based and noncancer-based SLs may change. At certain risk levels, the cancer-based number may be more conservative; at different risk levels, the noncancer-based number may be more conservative. The data user needs to consider both cancer and noncancer endpoints.

Similar to the above approach of using the calculator to recalculate SLs based on non-default target levels, the values in the screening tables themselves can be addressed directly. Consistent with the above logic, although the EPA Superfund Program has not developed guidance on this, it is not uncommon that Superfund sites are screened at a THQ of 0.1. (The cancer-based SLs are already at a target risk of 1E-6 and are usually not adjusted further in this scenario.) SLs based on a THQ of 0.1 can be derived by dividing a default SL by 10. Again, note that altering the target HQ can change the relationship between cancer-based and noncancerbased screening levels; the data user needs to consider both endpoints. Additional approaches or alternatives may exist. When screening actual or potential Superfund sites, users are encouraged to consult with risk assessors in that EPA Regional Office when evaluating or screening contamination at a site with multiple contaminants to see if they may know of another approach or if they have a preference.

#### 5.15.2 Using RSLs to Sum Risk from Multiple Contaminants

RSLs can be used to estimate the total risk from multiple contaminants at a site as part of a screening procedure used by some regions. This methodology, which does not substitute for a baseline risk assessment, is often called the "sum of the ratios" approach. A step-wise approach follows:

- 1. Perform an extensive records search and compile existing data.
- 2. Identify site contaminants in the SL Table. Record the SL concentrations for various media and note whether SL is based on cancer risk (indicated by 'c') or noncancer hazard (indicated by 'n'). Segregate cancer SLs from non-cancer SLs and exclude (but don't eliminate) non-risk based SLs 's' or 'm'.
- 3. For cancer risk estimates, take the site-specific concentration (maximum or 95th percent of the upper confidence limit on the mean (UCL)) and divide by the SL concentrations that are designated for cancer evaluation 'c'. Multiply this ratio by 10-6 to estimate chemical-specific risk for a reasonable maximum exposure (RME). For multiple pollutants, simply add the risk for each chemical. See

equation below.

Total Cancer Risk = 
$$\left[ \left( \frac{C_x}{SL_x} \right) + \left( \frac{C_y}{SL_y} \right) + \left( \frac{C_z}{SL_z} \right) \right] \times TR$$

where:

TR = target cancer risk C = site contaminant concentration

4. For non-cancer hazard estimates, divide the concentration term by its respective non-cancer SL designated as 'n' and sum the ratios for multiple contaminants. The cumulative ratio represents a non-carcinogenic hazard index (HI). A hazard index of 1 or less is generally considered 'safe'. A ratio greater than 1 suggests further evaluation. Note that carcinogens may also have an associated non-cancer SL that is not listed in the SL Table. To obtain these values, the user should view the Supporting Tables. See equation below

Total Hazard Index = 
$$\left[ \left( \frac{C_x}{SL_x} \right) + \left( \frac{C_y}{SL_y} \right) + \left( \frac{C_z}{SL_z} \right) \right] \times THQ$$

where:

THQ = target hazard quotient C = site contaminant concentration

#### 5.16 Deriving Soil Gas SLs

The air SLs could apply to indoor air from, e.g., a vapor intrusion scenario. To model indoor air concentrations from other media (e.g., soil gas, groundwater), consult with regional experts in vapor intrusion.

For more information on EPA's current understanding of this emerging exposure pathway, please refer to EPA's recent draft guidance Evaluating the Vapor Intrusion to Indoor Air Pathway from Groundwater and Soils (Subsurface Vapor Intrusion Guidance) (USEPA 2002).

#### 5.17 Mutagens

Some of the cancer causing analytes in this tool operate by a mutagenic mode of action for carcinogenesis. There is reason to surmise that some chemicals with a mutagenic mode of action, which would be expected to cause irreversible changes to DNA, would exhibit a greater effect in early-life versus later-life exposure. Cancer risk to children in the context of the U.S. Environmental Protection Agency's cancer guidelines (U.S. EPA, 2005 (PDF) (166 pp, 468 K)) includes both early-life exposures that may result in the occurrence of cancer during childhood and early-life exposures that may contribute to cancers later in life. In keeping with this guidance, separate cancer risk equations are presented for mutagens. The mutagen vinyl chloride has a unique set of equations. Consult <u>Supplemental Guidance for Assessing Susceptibility from Early-Life Exposure to Carcinogens, EPA/630/R-03/003F, March 2005 (PDF)</u> (126 pp, 1.78 MB) for further information.

The below table lists the chemicals considered to be carcinogenic by mutagenic mode of action for the purposes of the RSLs. Also provided in the table is a link to the source as to why the chemical is considered to be a mutagen.

Chemical	CASRN	Reference
	79-06-1	IRIS
Benz[a]anthracene	56-55-3	Benzo[a]pyrene*
Benzidine	92-87-5	Supplemental Guidance
Benzo[a]pyrene	50-32-8	Supplemental Guidance
Benzo[b]fluoranthene	205-99-2	Benzo[a]pyrene*
Benzo[k]fluoranthene	207-08-9	Benzo[a]pyrene*
Chromium(VI)	18540-29-9	CalEPA and OPP
Chrysene	218-01-9	Benzo[a]pyrene*
Coke Oven Emissions	8007-45-2	70 Federal Register 19992
Dibenz[a,h]anthracene	53-70-3	Supplemental Guidance
Dibromo-3-chloropropane, 1,2-	96-12-8	PPRTV
Dimethylbenz(a)anthracene, 7,12-	57-97-6	Supplemental Guidance
Ethylene Oxide	75-21-8	IRIS
Indeno[1,2,3-cd]pyrene	193-39-5	Benzo[a]pyrene*
Methylcholanthrene, 3-	56-49-5	Supplemental Guidance
Methylene Chloride	75-09-2	IRIS
Methylene-bis(2-chloroaniline), 4,4'-	101-14-4	PPRTV
Nitrosodiethylamine, N-	55-18-5	Supplemental Guidance

Nitrosodimethylamine, N-	62-75-9	Supplemental Guidance
Nitroso-N-ethylurea, N-	759-73-9	Supplemental Guidance
Nitroso-N-methylurea, N-	684-93-5	Supplemental Guidance
Safrole	94-59-7	Supplemental Guidance
Trichloroethylene	79-01-6	IRIS
Trichloropropane, 1,2,3-	96-18-4	IRIS
Urethane	51-79-6	Supplemental Guidance
Vinyl Chloride	75-01-4	Supplemental Guidance

\* Please see section 2.3.6 of this user guide regarding Relative Potency Factors (RPFs).

#### 5.18 Trichloroethylene (TCE)

It is recommended that a regional risk assessor be consulted when evaluating TCE in any medium especially when less than chronic exposure scenarios are considered. The <u>Superfund program</u> issued a <u>Compilation of</u> <u>Information Relating of Early/Interim Actions at Superfund Sites and the TCE IRIS Assessment (PDF)</u> (3 pp, 929 K) memo in August 2014. Several regions have issued their own guidance as well.

In order to make the calculator display the correct results for TCE, the standard cancer and mutagen equations needed to be combined. Since TCE requires the use of different toxicity values for cancer and mutagen equations, it was decided to make a toxicity value adjustment factor for cancer (CAF) and mutagens (MAF). The adjustments were done for oral (o) and inhalation (i). These adjustment factors are used in the TCE equation images presented in section 4. The equations used are presented below. The adjustment factors are based on the adult-based toxicity values and these are the cancer toxicity values presented in the Generic Tables.

$$CAF_{0}(0.804) = \frac{CSF_{0}\left(\frac{3.7 \times 10^{-2} \text{mg}}{\text{kg} \cdot \text{day}}\right)^{-1} \text{NHL+Liver oral slope factor}}{CSF_{0}\left(\frac{4.6 \times 10^{-2} \text{mg}}{\text{kg} \cdot \text{day}}\right)^{-1} \text{Adult} - \text{based oral slope factor}} MAF_{0}(0.202) = \frac{CSF_{0}\left(\frac{9.3 \times 10^{-3} \text{mg}}{\text{kg} \cdot \text{day}}\right)^{-1} \text{Adult} - \text{based oral slope factor}}}{CSF_{0}\left(\frac{4.6 \times 10^{-2} \text{mg}}{\text{kg} \cdot \text{day}}\right)^{-1} \text{Adult} - \text{based oral slope factor}} MAF_{0}(0.202) = \frac{CSF_{0}\left(\frac{4.6 \times 10^{-2} \text{mg}}{\text{kg} \cdot \text{day}}\right)^{-1} \text{Adult} - \text{based oral slope factor}}}{CSF_{0}\left(\frac{4.6 \times 10^{-2} \text{mg}}{\text{kg} \cdot \text{day}}\right)^{-1} \text{Adult} - \text{based oral slope factor}}} MAF_{1}(0.244) = \frac{IUR\left(\frac{1 \times 10^{-6} \text{mg}}{\text{m}^{3}}\right)^{-1} \text{Kidney unit risk estimate}}{IUR\left(\frac{4.1 \times 10^{-6} \text{mg}}{\text{m}^{3}}\right)^{-1} \text{Adult} - \text{based unit risk estimate}}}{IUR\left(\frac{4.1 \times 10^{-6} \text{mg}}{\text{m}^{3}}\right)^{-1} \text{Adult} - \text{based unit risk estimate}}}{IUR\left(\frac{4.1 \times 10^{-6} \text{mg}}{\text{m}^{3}}\right)^{-1} \text{Adult} - \text{based unit risk estimate}}}$$

#### 5.19 Mercuric Chloride (and other Mercury salts)

The IRIS RfC for "Mercury (elemental)" is used as a surrogate for "Mercuric Chloride (and other Mercury salts)". Note, that the VF for "Mercury (elemental)" is not used as a surrogate for "Mercuric Chloride (and other Mercury salts)". The use of the surrogate RfC would appear to be a violation of the RSL toxicity hierarchy because Cal EPA offers a RfC for Mercuric Chloride. However, the actual form of mercury evaluated for the Cal EPA RfC was elemental mercury. Since IRIS already had a RfC for "Mercury (elemental)", it was decided to use the tier 1 source over a tier 3 source.

#### 5.20 Cyanide (CN-)

The IRIS RfC for "Hydrogen Cyanide" is used as a surrogate for "Cyanide (CN-)".

#### 5.21 Thallic Oxide and Thallium Selenite

The oral RfD for thallic oxide, used in this website, is derived from the PPRTV oral RfD for thallium sulfate by molecular weight (MW) adjustments and stoichiometric calculations. Thallic oxide (Tl<sub>2</sub>O<sub>3</sub>) has a MW of 456.765 and thallium sulfate (Tl<sub>2</sub>SO<sub>4</sub>) has a MW of 504.82. To derive the oral RfD of 2E-05 mg/kg-day for thallic oxide, the thallium sulfate RfD of 2E-05 mg/kg-day is multiplied by the MW of thallic oxide (456.765) divided by the MW of thallium sulfate (504.82). The oral RfD for thallium selenite, used in this website, is derived from the PPRTV oral RfD for thallium by molecular weight (MW) adjustments and stoichiometric calculations. Thallic oxide (TlSe) has a MW of 283.34 and thallium (Tl) has a MW of 204.38. To derive the oral RfD of 1E-05 mg/kg-day for thallium selenite, the thallium RfD of 1E-05 mg/kg-day is multiplied by the MW of thallium RfD of 1E-05 mg/kg-day for thallium selenite, the thallium RfD of 1E-05 mg/kg-day is multiplied by the MW of thallium selenite (283.34) divided by the MW of thallium (204.38).

#### **5.22** Polycyclic Aromatic Hydrocarbons (PAHs)

For PAHs in soil, we have not made any recommendations on a default value, which is to say that the default assumption remains that these are 100% bioavailable. There is also no available in vitro method to estimate the oral RBA of PAHs. A small number of sites have elected to run swine or rat models to assess oral RBA, and the TRW has reviewed them before the RBA was accepted for use at the site. Guidance related to these topics can be found in the <u>Soil Bioavailability at Superfund Sites Guidance</u>.

#### 5.23 Refractory Ceramic Fibers

The <u>ATSDR</u> chronic RfC for refractory ceramic fibers is presented in units of fibers/cm<sup>3</sup>. The RfC presented in the tables and calculator is 0.03 fibers/cm<sup>3</sup>, which differs from all other chemicals where the RfC unit is mg/m<sup>3</sup>. When the chronic RfC is used in the standard RSL air inhalation equations, the resulting units are not in  $\mu$ g/m<sup>3</sup> like all the other chemicals. RSLs are only calculated for air as the medium. The air values in the RSL table are calculated using the equations below to give RSLs in fibers/m<sup>3</sup>.

Resident Air Noncancer Equation

$$SL_{res-air-rcf} \left( \frac{f}{m^3} \right) = \frac{THQ \times AT_{res-a} \left( \frac{365 \text{ days}}{\text{year}} \times ED_{res} \left( 26 \text{ years} \right) \right) \times \left( \frac{1.0 \times 10^6 \text{ cm}^3}{\text{m}^3} \right)}{EF_{res} \left( \frac{350 \text{ days}}{\text{year}} \right) \times ED_{res} \left( 26 \text{ years} \right) \times ET_{res} \left( \frac{24 \text{ hours}}{\text{day}} \right) \times \left( \frac{1 \text{ day}}{24 \text{ hours}} \right) \times \frac{1}{RfC \left( \frac{f}{cm^3} \right)}}$$

Composite Worker Air Noncancer Equation

$$SL_{w-air-rcf} \begin{pmatrix} f \\ m^3 \end{pmatrix} = \frac{THQ \times AT_{w} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{w} \left(25 \text{ years}\right)\right) \times \left(\frac{1.0 \times 10^{6} \text{ cm}^{3}}{\text{m}^{3}}\right)}{EF_{w} \left(\frac{250 \text{ days}}{\text{year}}\right) \times ED_{w} \left(25 \text{ years}\right) \times ET_{w} \left(\frac{8 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \frac{1}{RfC \left(\frac{f}{\text{cm}^{3}}\right)}$$

Indoor Worker Air Noncancer Equation

$$SL_{iw-air-rcf} \begin{pmatrix} f \\ m^3 \end{pmatrix} = \frac{THQ \times AT_{iw} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{iw} \left(25 \text{ years}\right)\right) \times \left(\frac{1.0 \times 10^6 \text{ cm}^3}{\text{m}^3}\right)}{EF_{iw} \left(\frac{250 \text{ days}}{\text{year}}\right) \times ED_{iw} \left(25 \text{ years}\right) \times ET_{iw} \left(\frac{8 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \frac{1}{RfC \left(\frac{f}{\text{cm}^3}\right)}}$$

Outdoor Worker Air Noncancer Equation

$$SL_{ow-air-rcf} \begin{pmatrix} f \\ m^{3} \end{pmatrix} = \frac{THQ \times AT_{ow} \left(\frac{365 \text{ days}}{\text{year}} \times ED_{ow} \left(25 \text{ years}\right)\right) \times \left(\frac{1.0 \times 10^{6} \text{ cm}^{3}}{\text{m}^{3}}\right)}{EF_{ow} \left(\frac{225 \text{ days}}{\text{year}}\right) \times ED_{ow} \left(25 \text{ years}\right) \times ET_{ow} \left(\frac{8 \text{ hours}}{\text{day}}\right) \times \left(\frac{1 \text{ day}}{24 \text{ hours}}\right) \times \frac{1}{RfC \left(\frac{f}{cm^{3}}\right)}}$$

Land Use	THI = 0.1	THI = 1.0
Resident	3128	31286
Composite Worker	13140	131400
Indoor Worker	13140	131400
Outdoor Worker	14600	146000

Refractory Ceramic Fibers Air RSLs (fibers/m<sup>3</sup>)

#### 5.24 Lanthanum Salts

The oral chronic RfDs for lanthanum salts, used in this website, are derived from the <u>PPRTV</u> oral chronic RfD for lanthanum by molecular weight (MW) adjustments and stoichiometric calculations. Lanthanum chloride, anhydrous (LaCl<sub>3</sub>) has a MW of 245.27, and lanthanum (La) has a MW of 138.91. To derive the chronic oral RfD of 2.83E-05 mg/kg-day for lanthanum chloride, anhydrous, the lanthanum RfD of 5E-05 mg/kg-day is multiplied by the MW of lanthanum (138.91) divided by the MW of lanthanum chloride, anhydrous (245.27). To derive the chronic oral RfD of 1.87E-05 mg/kg-day for lanthanum chloride heptahydrate, the lanthanum RfD of 5E-05 mg/kg-day is multiplied by the MW of 5E-05 mg/kg-day is multiplied by the MW of lanthanum chloride heptahydrate, the lanthanum RfD of 5E-05 mg/kg-day is multiplied by the MW of 1.60E-05 mg/kg-day for lanthanum nitrate hexahydrate, the lanthanum nitrate hexahydrate (433.01). To derive the chronic oral RfD of 2.08E-05 mg/kg-day for lanthanum acetate hydrate, the lanthanum RfD of 5E-05 mg/kg-day for lanthanum (38.91) divided by the MW of lanthanum (138.91) divided by the MW of lanthanum (138.91). To derive the chronic oral RfD of 2.08E-05 mg/kg-day for lanthanum nitrate hexahydrate, the lanthanum RfD of 5E-05 mg/kg-day is multiplied by the (33.01). To derive the chronic oral RfD of 2.08E-05 mg/kg-day for lanthanum acetate hydrate, the lanthanum RfD of 5E-05 mg/kg-day is multiplied by the MW of lanthanum (138.91) divided by the MW of lanthanum (38.91) divided by the MW of lanthanum acetate hydrate (433.01). To derive the chronic oral RfD of 2.08E-05 mg/kg-day for lanthanum acetate hydrate, the lanthanum RfD of 5E-05 mg/kg-day is multiplied by the MW of lanthanum (38.91) divided by the MW of lanthanum acetate hydrate (334.05).

### 6. Using the Calculator

The <u>Calculator</u> can be used to generate site-specific SLs or PRGs. The calculator requires the user to make some simple selections. To use the calculator Select a land use. Next, select whether you want Default or Site-specific SLs. Selecting default screening levels will reproduce the results in the generic <u>Generic Tables</u>. Selecting Site-Specific will allow you to change exposure parameters. Now pick your analytes. To pick several in a row, depress the left mouse button and drag, then release. Or hold the Ctrl key down and select multiple analytes that are not in a row. Select the output option. Hit the retrieve button. If you selected Site-Specific, the next page allows you to change exposure parameters. Hit the retrieve button. SLs are being calculated. The first table presents the input parameters that were selected. The next table contains the screening levels. This table can be too big to print. The easiest way to manage this table is to move it to a spreadsheet or a database. To copy this table, hold the left mouse key down and drag across the entire table. when done, press Ctrl c to copy. Switch to a spreadsheet and press Ctrl v to paste.

Symbol	Definition (units)	Default	Reference	
SLs				
	Resident SLs			
SL <sub>res-sol-nc-</sub> ing	Resident Soil Noncarcinogenic Ingestion (mg/kg)	Contaminant- specific	Determined in this calculator	
SL <sub>res-sol-nc-</sub> der	Resident Soil Noncarcinogenic Dermal (mg/kg)	Contaminant- specific	Determined in this calculator	

• Table 1. Standard Default Factors

SL <sub>res-sol-nc-</sub> inh	Resident Soil Noncarcinogenic Inhalation (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>res-sol-nc-</sub> tot	Resident Soil Noncarcinogenic Total (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>res-sol-ca-</sub> ing	Resident Soil Carcinogenic Ingestion (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>res-sol-ca-</sub> der	Resident Soil Carcinogenic Dermal (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>res-sol-ca-</sub> inh	Resident Soil Carcinogenic Inhalation (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>res-sol-ca-</sub> tot	Resident Soil Carcinogenic Total (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>res-sol-mu-</sub> ing	Resident Soil Mutagenic Ingestion (mg/kg)	Mutagen-specific	Determined in this calculator
SL <sub>res-sol-mu-</sub> der	Resident Soil Mutagenic Dermal (mg/kg)	Mutagen-specific	Determined in this calculator
SL <sub>res-sol-mu-</sub> inh	Resident Soil Mutagenic Inhalation (mg/kg)	Mutagen-specific	Determined in this calculator
SL <sub>res-sol-mu-</sub> tot	Resident Soil Mutagenic Total (mg/kg)	Mutagen-specific	Determined in this calculator
SL <sub>res-sol-ca-</sub> vc-ing	Resident Soil Carcinogenic Vinyl Chloride Ingestion (mg/kg)	Vinyl Chloride -specific	Determined in this calculator

SL <sub>res-sol-ca-</sub> vc-der	Resident Soil Carcinogenic Vinyl Chloride Dermal (mg/kg)	Vinyl Chloride- specific	Determined in this calculator
SL <sub>res-sol-ca-</sub> vc-inh	Resident Soil Carcinogenic Vinyl Chloride Inhalation (mg/kg)	Vinyl Chloride- specific	Determined in this calculator
SL <sub>res-sol-ca-</sub> vc-tot	Resident Soil Carcinogenic Vinyl Chloride Total (mg/kg)	Vinyl Chloride- specific	Determined in this calculator
SL <sub>res-sol-tce-</sub> ing	Resident Soil Trichloroethylene Ingestion (mg/kg)	Trichloroethylene- specific	Determined in this calculator
SL <sub>res-sol-tce-</sub> der	Resident Soil Trichloroethylene Dermal (mg/kg)	Trichloroethylene- specific	Determined in this calculator
SL <sub>res-sol-tce-</sub> inh	Resident Soil Trichloroethylene Inhalation (mg/kg)	Trichloroethylene- specific	Determined in this calculator
SL <sub>res-sol-tce-</sub> tot	Resident Soil Trichloroethylene Total (mg/kg)	Trichloroethylene- specific	Determined in this calculator
SL <sub>water-nc-</sub> ing	Resident Tapwater Noncarcinogenic Ingestion (µg/L)	Contaminant- specific	Determined in this calculator
SL <sub>water-nc-</sub> der	Resident Tapwater Noncarcinogenic Dermal (µg/L)	Contaminant- specific	Determined in this calculator
SL <sub>water-nc-</sub> inh	Resident Tapwater Noncarcinogenic Inhalation (µg/L)	Contaminant- specific	Determined in this calculator

SL <sub>water-nc-</sub> tot	Resident Tapwater Noncarcinogenic Total (µg/L)	Contaminant- specific	Determined in this calculator
SL <sub>water-ca-</sub> ing	Resident Tapwater Carcinogenic Ingestion (µg/L)	Contaminant- specific	Determined in this calculator
SL <sub>water-ca-</sub> der	Resident Tapwater Carcinogenic Dermal (µg/L)	Contaminant- specific	Determined in this calculator
SL <sub>water-ca-</sub> inh	Resident Tapwater Carcinogenic Inhalation (µg/L)	Contaminant- specific	Determined in this calculator
SL <sub>water-ca-</sub> tot	Resident Tapwater Carcinogenic Total (µg/L)	Contaminant- specific	Determined in this calculator
SL <sub>water-mu-</sub> ing	Resident Tapwater Mutagenic Ingestion (µg/L)	Mutagen-specific	Determined in this calculator
SL <sub>water-mu-</sub> der	Resident Tapwater Mutagenic Dermal (µg/L)	Mutagen-specific	Determined in this calculator
SL <sub>water-mu-</sub> inh	Resident Tapwater Mutagenic Inhalation (µg/L)	Mutagen-specific	Determined in this calculator
SL <sub>water-mu-</sub> tot	Resident Tapwater Mutagenic Total (µg/L)	Mutagen-specific	Determined in this calculator
SL <sub>res-water-</sub> ca-vc-ing	Resident Tapwater Carcinogenic Vinyl Chloride Ingestion (µg/L)	Vinyl Chloride- specific	Determined in this calculator
SL <sub>res-water-</sub> ca-vc-der	Resident Tapwater Carcinogenic Vinyl Chloride Dermal (µg/L)	Vinyl Chloride- specific	Determined in this calculator
SL <sub>res-water-</sub> ca-vc-inh	Resident Tapwater Carcinogenic Vinyl Chloride Inhalation (µg/L)	Vinyl Chloride- specific	Determined in this calculator
--	--	--------------------------------	-------------------------------
SL <sub>res-water-</sub> ca-vc-tot	Resident Tapwater Carcinogenic Vinyl Chloride Total (µg/L)	Vinyl Chloride- specific	Determined in this calculator
SL <sub>water-tce-</sub> ing	Resident Tapwater Trichloroethylene Ingestion (µg/L)	Trichloroethylene- specific	Determined in this calculator
SL <sub>water-tce-</sub> der	Resident Tapwater Trichloroethylene Dermal (µg/L)	Trichloroethylene- specific	Determined in this calculator
SL <sub>water-tce-</sub> inh	Resident Tapwater Trichloroethylene Inhalation (µg/L)	Trichloroethylene- specific	Determined in this calculator
SL <sub>water-tce-</sub> tot	Resident Tapwater Trichloroethylene Total (µg/L)	Trichloroethylene- specific	Determined in this calculator
SL <sub>res-air-nc</sub>	Resident Air Noncarcinogenic (µg/m <sup>3</sup> )	Contaminant- specific	Determined in this calculator
SL <sub>res-air-ca</sub>	Resident Air Carcinogenic (µg/m <sup>3</sup> )	Contaminant- specific	Determined in this calculator
SL <sub>res-air-mu</sub>	Resident Air Mutagenic (µg/m <sup>3</sup> )	Mutagen-specific	Determined in this calculator
SL <sub>res-air-ca-</sub> vinyl chloride	Resident Air Carcinogenic Vinyl Chloride (µg/m <sup>3</sup> )	Vinyl Chloride- specific	Determined in this calculator
SL <sub>res-air-tce</sub>	Resident Air Trichloroethylene (µg/m <sup>3</sup> )	Trichloroethylene- specific	Determined in this calculator

	Worker SLs			
SL <sub>w-sol-nc-</sub> ing	Composite Worker Soil Noncarcinogenic Ingestion (mg/kg)	Contaminant- specific	Determined in this calculator	
SL <sub>w-sol-nc-</sub> der	Composite Worker Soil Noncarcinogenic Dermal (mg/kg)	Contaminant- specific	Determined in this calculator	
SL <sub>w-sol-nc-</sub> inh	Composite Worker Soil Noncarcinogenic Inhalation (mg/kg)	Contaminant- specific	Determined in this calculator	
SL <sub>w-sol-nc-</sub> tot	Composite Worker Soil Noncarcinogenic Total (mg/kg)	Contaminant- specific	Determined in this calculator	
SL <sub>w-sol-ca-</sub> ing	Composite Worker Soil Carcinogenic Ingestion (mg/kg)	Contaminant- specific	Determined in this calculator	
SL <sub>w-sol-ca-</sub> der	Composite Worker Soil Carcinogenic Dermal (mg/kg)	Contaminant- specific	Determined in this calculator	
$SL_{w-sol-ca-}$ inh	Composite Worker Soil Carcinogenic Inhalation (mg/kg)	Contaminant- specific	Determined in this calculator	
SL <sub>w-sol-ca-</sub> tot	Composite Worker Soil Carcinogenic Total (mg/kg)	Contaminant- specific	Determined in this calculator	
SL <sub>w-air-nc</sub>	Composite Worker Air Noncarcinogenic (µg/m <sup>3</sup> )	Contaminant- specific	Determined in this calculator	
SL <sub>w-air-ca</sub>	Composite Worker Air Carcinogenic (µg/m <sup>3</sup> )	Contaminant- specific	Determined in this calculator	

SLow-sol-nc-ing	Outdoor Worker Soil Noncarcinogenic Ingestion (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>ow-sol-nc-</sub> der	Outdoor Worker Soil Noncarcinogenic Dermal (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>ow-sol-nc-</sub> inh	Outdoor Worker Soil Noncarcinogenic Inhalation (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>ow-sol-nc-</sub> tot	Outdoor Worker Soil Noncarcinogenic Total (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>ow-sol-ca-</sub> ing	Outdoor Worker Soil Carcinogenic Ingestion (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>ow-sol-ca-</sub> der	Outdoor Worker Soil Carcinogenic Dermal (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>ow-sol-ca-</sub> inh	Outdoor Worker Soil Carcinogenic Inhalation (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>ow-sol-ca-</sub> tot	Outdoor Worker Soil Carcinogenic Total (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>ow-air-nc</sub>	Outdoor Worker Air Noncarcinogenic (µg/m <sup>3</sup> )	Contaminant- specific	Determined in this calculator
SL <sub>ow-air-ca</sub>	Outdoor Worker Air Carcinogenic (µg/m <sup>3</sup> )	Contaminant- specific	Determined in this calculator
SL <sub>iw-sol-nc-</sub> ing	Indoor Worker Soil Noncarcinogenic Ingestion (mg/kg)	Contaminant- specific	Determined in this calculator

SL <sub>iw-sol-nc-</sub>	Indoor Worker Soil Noncarcinogenic Inhalation (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>iw-sol-nc-</sub> tot	Indoor Worker Soil Noncarcinogenic Total (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>iw-sol-ca-</sub> ing	Indoor Worker Soil Carcinogenic Ingestion (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>iw-sol-ca-</sub> inh	Indoor Worker Soil Carcinogenic Inhalation (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>iw-sol-ca-</sub> tot	Indoor Worker Soil Carcinogenic Total (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>iw-air-nc</sub>	Indoor Worker Air Noncarcinogenic (µg/m <sup>3</sup> )	Contaminant- specific	Determined in this calculator
SL <sub>iw-air-ca</sub>	Indoor Worker Air Carcinogenic (µg/m <sup>3</sup> )	Contaminant- specific	Determined in this calculator
SL <sub>cw-sol-nc-</sub> ing	Construction Worker Soil Noncarcinogenic Ingestion (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>cw-sol-nc-</sub> der	Construction Worker Soil Noncarcinogenic Dermal (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>cw-sol-nc-</sub> inh	Construction Worker Soil Noncarcinogenic Inhalation (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>cw-sol-nc-</sub> tot	Construction Worker Soil Noncarcinogenic Total (mg/kg)	Contaminant- specific	Determined in this calculator

SL <sub>cw-sol-ca-</sub> ing	Construction Worker Soil Carcinogenic Ingestion (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>cw-sol-ca-</sub> der	Construction Worker Soil Carcinogenic Dermal (mg/kg)	Contaminant- specific	Determined in this calculator
$SL_{cw-sol-ca-}$ inh	Construction Worker Soil Carcinogenic Inhalation (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>cw-sol-ca-</sub> tot	Construction Worker Soil Carcinogenic Total (mg/kg)	Contaminant- specific	Determined in this calculator
	Re	ecreator SLs	
SL <sub>rec-sol-nc-</sub> ing	Recreator Soil Noncarcinogenic Ingestion (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>rec-sol-nc-</sub> der	Recreator Soil Noncarcinogenic Dermal (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>rec-sol-nc-</sub> inh	Recreator Soil Noncarcinogenic Inhalation (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>rec-sol-nc-</sub> tot	Recreator Soil Noncarcinogenic Total (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>rec-sol-ca-</sub> ing	Recreator Soil Carcinogenic Ingestion (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>rec-sol-ca-</sub> der	Recreator Soil Carcinogenic Dermal (mg/kg)	Contaminant- specific	Determined in this calculator

SL <sub>rec-sol-ca-</sub> inh	Recreator Soil Carcinogenic Inhalation (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>rec-sol-ca-</sub> tot	Recreator Soil Carcinogenic Total (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>rec-sol-</sub> mu-ing	Recreator Soil Mutagenic Ingestion (mg/kg)	Mutagenic-specific	Determined in this calculator
SL <sub>rec-sol-</sub> mu-der	Recreator Soil Mutagenic Dermal (mg/kg)	Mutagenic-specific	Determined in this calculator
SL <sub>rec-sol-</sub> mu-inh	Recreator Soil Mutagenic Inhalation (mg/kg)	Mutagenic-specific	Determined in this calculator
SL <sub>rec-sol-</sub> mu-tot	Recreator Soil Mutagenic Total (mg/kg)	Mutagenic-specific	Determined in this calculator
SL <sub>rec-sol-ca-</sub> vc-ing	Recreator Soil Carcinogenic Vinyl Chloride Ingestion (mg/kg)	Vinyl Chloride- specific	Determined in this calculator
SL <sub>rec-sol-ca-</sub> vc-der	Recreator Soil Carcinogenic Vinyl Chloride Dermal (mg/kg)	Vinyl Chloride- specific	Determined in this calculator
SL <sub>rec-sol-ca-</sub> vc-inh	Recreator Soil Carcinogenic Vinyl Chloride Inhalation (mg/kg)	Vinyl Chloride- specific	Determined in this calculator
SL <sub>rec-sol-ca-</sub> vc-tot	Recreator Soil Carcinogenic Vinyl Chloride Total (mg/kg)	Vinyl Chloride- specific	Determined in this calculator
SL <sub>rec-sol-tce-</sub> ing	Recreator Soil Trichloroethylene Ingestion (mg/kg)	Trichloroethylene- specific	Determined in this calculator

			u
SL <sub>rec-sol-tce-</sub> der	Recreator Soil Trichloroethylene Dermal (mg/kg)	Trichloroethylene- specific	Determined in this calculator
SL <sub>rec-sol-tce-</sub> inh	Recreator Soil Trichloroethylene Inhalation (mg/kg)	Trichloroethylene- specific	Determined in this calculator
SL <sub>rec-sol-tce-</sub> tot	Recreator Soil Trichloroethylene Total (mg/kg)	Trichloroethylene- specific	Determined in this calculator
SL <sub>rec-water-</sub> nc-ing	Recreator Surface Water Non-Carcinogenic Ingestion (µg/L)	Contaminant- specific	Determined in this calculator
SL <sub>rec-water-</sub> nc-der	Recreator Surface Water Non-Carcinogenic Dermal (µg/L)	Contaminant- specific	Determined in this calculator
SL <sub>rec-water-</sub> nc-tot	Recreator Surface Water Non-Carcinogenic Total (µg/L)	Contaminant- specific	Determined in this calculator
SL <sub>rec-water-</sub> ca-ing	Recreator Surface Water Carcinogenic Ingestion (µg/L)	Contaminant- specific	Determined in this calculator
SL <sub>rec-water-</sub> ca-der	Recreator Surface Water Carcinogenic Dermal (µg/L)	Contaminant- specific	Determined in this calculator
SL <sub>rec-water-</sub> ca-tot	Recreator Surface Water Carcinogenic Total (µg/L)	Contaminant- specific	Determined in this calculator
SL <sub>rec-water-</sub> mu-ing	Recreator Surface Water Mutagenic Ingestion (µg/L)	Mutagen-specific	Determined in this calculator
SL <sub>rec-water-</sub> mu-der	Recreator Surface Water Mutagenic Dermal (µg/L)	Mutagen-specific	Determined in this calculator

SL <sub>rec-water-</sub> mu-tot	Recreator Surface Water Mutagenic Total (µg/L)	Mutagen-specific	Determined in this calculator
SL <sub>rec-water-</sub> vc-ing	Recreator Surface Water Carcinogenic Vinyl Chloride Ingestion (µg/L)	Vinyl Chloride- specific	Determined in this calculator
S <sub>rec-water-vc-</sub> der	Recreator Surface Water Carcinogenic Vinyl Chloride Dermal (µg/L)	Vinyl Chloride- specific	Determined in this calculator
SL <sub>rec-water-</sub> vc-tot	Recreator Surface Water Carcinogenic Vinyl Chloride Total (µg/L)	Vinyl Chloride- specific	Determined in this calculator
SL <sub>rec-water-</sub> tce-ing	Recreator Surface Water Trichlroethylene Ingestion (µg/L)	Trichlroethylene- specific	Determined in this calculator
SL <sub>rec-water-</sub> tce-der	Recreator Surface Water Trichlroethylene Dermal (µg/L)	Trichlroethylene- specific	Determined in this calculator
SLrec-water- tce-tot	Recreator Surface Water Trichlroethylene Total (µg/L)	Trichlroethylene- specific	Determined in this calculator
		Fish SLs	
SL <sub>res-fsh-nc-</sub> ing	Resident Fish Noncarcinogenic Ingestion (mg/kg)	Contaminant- specific	Determined in this calculator
SL <sub>res-fsh-ca-</sub> ing	Resident Fish Carcinogenic Ingestion (mg/kg)	Contaminant- specific	Determined in this calculator
	Το	oxicity Values	
RfD <sub>o</sub> or RFDOC	Chronic Oral Reference Dose (mg/kg-day)	Contaminant- specific	EPA Superfund hierarchy

RfC or RFCIC	Chronic Inhalation Reference Concentration (mg/m <sup>3</sup> )	Contaminant- specific	EPA Superfund hierarchy
CSF <sub>o</sub> or SFO	Oral Slope Factor (mg/kg- day) <sup>-1</sup>	Contaminant- specific	EPA Superfund hierarchy
IUR	Inhalation Unit Risk (µg/m <sup>3</sup> ) <sup>-1</sup>	Contaminant- specific	EPA Superfund hierarchy
	Miscell	aneous Variables	
TR	target risk	1 x 10 <sup>-6</sup>	Selected by user
THQ	target hazard quotient	0.1	Selected by user
THI	target hazard index	0.1	Selected by user
RBA	relative bioavailability factor	Arsenic = 0.6 All Others = 1	<u>U.S. EPA 2012</u>
К	Andelman Volatilization Factor (L/m <sup>3</sup> )	0.5	U.S. EPA 1991b (pg. 20)
K <sub>p</sub>	Dermal Permeability Constant (cm/hour)	Contaminant- specific Inorganic default = 0.001	U.S. EPA 2004 Exhibit 3-1 and Section 3.1.2.1
K <sub>p,ve</sub>	Steady-state Permeability Coefficient (cm/hour)	Contaminant- specific	U.S. EPA 2004
K <sub>ew</sub>	Equillibrium Partition Coefficient between epidermis and water (unitless)	1 - assuming epidermis behaves essentially as water	U.S. EPA 2004

De	Effective Diffusivity of absorbing chemical in the epidermis (cm <sup>2</sup> /sec)	(7.1 × 10 <sup>-6</sup> ) / (√MW)	U.S. EPA 2004
Le	Effective Thickness of the Epidermis (cm)	10-2	U.S. EPA 2004
AT <sub>res-c</sub>	Averaging time - resident child (days)	$365 \text{ x ED}_{\text{res-c}} = 2190$	U.S. EPA 1989 (pg. 6-23)
AT <sub>res-a</sub>	Averaging time - resident adult (days)	$365 \text{ x ED}_{\text{res}} = 9490$	U.S. EPA 1989 (pg. 6-23)
AT <sub>res</sub>	Averaging time - resident age adjusted (days)	365 x LT = 25550	U.S. EPA 1989 (pg. 6-23)
AT <sub>w-a</sub>	Averaging time - composite worker (days)	365 x ED <sub>w</sub> = 9125 (non-carcinogenic)	U.S. EPA 1989 (pg. 6-23)
AT <sub>w</sub>	Averaging time - composite worker (days)	365 x LT = 25550 (carcinogenic)	U.S. EPA 1989 (pg. 6-23)
AT <sub>iw-a</sub>	Averaging time - indoor worker (days)	365 x ED <sub>iw</sub> = 9125 (non-carcinogenic)	U.S. EPA 1989 (pg. 6-23)
AT <sub>iw</sub>	Averaging time - indoor worker soil (days)	365 x LT = 25550 (carcinogenic)	U.S. EPA 1989 (pg. 6-23)
AT <sub>ow-a</sub>	Averaging time - outdoor worker (days)	365 x ED <sub>ow</sub> = 9125 (non-carcinogenic)	U.S. EPA 1989 (pg. 6-23)
AT <sub>ow</sub>	Averaging time - outdoor worker (days)	365 x LT = 25550 (carcinogenic)	U.S. EPA 1989 (pg. 6-23)
AT <sub>cw-a</sub>	Averaging time - construction worker (days)	EWcw x 7 (d/wk) x EDcw = 350 (non- carcinogenic)	U.S. EPA 1989 (pg. 6-23)

AT <sub>cw</sub>	Averaging time - construction worker (days)	365 x LT = 25550 (carcinogenic)	U.S. EPA 1989 (pg. 6-23)
AT <sub>rec-c</sub>	Averaging time - recreator child (days)	365 x ED <sub>rec-c</sub>	U.S. EPA 1989 (pg. 6-23)
AT <sub>rec-a</sub>	Averaging time - recreator adult (days)	365 x ED <sub>rec-a</sub>	U.S. EPA 1989 (pg. 6-23)
AT <sub>rec</sub>	Averaging time - recreator (days)	365 x LT	U.S. EPA 1989 (pg. 6-23)
LT	Lifetime (years)	70	U.S. EPA 1989 (pg. 6-22)
ΔH <sub>v,b</sub>	Enthalpy of vaporization at the normal boiling point (cal/mol)	Contaminant- specific	See Chemical-specific hierarchy
ΔH <sub>v,gw</sub>	Enthalpy of vaporization at temperature of groundwater (cal/mol)	Contaminant- specific	Determined in this calculator
HLC	Henry's Law Constant at specified groundwater temperature (atm-m <sup>3</sup> /mol)	Contaminant- specific	See Chemical-specific hierarchy
T <sub>w</sub>	Groundwater Temperatures (Kelvin)	Site-specific	Site-specific
T <sub>C</sub>	Critical Temperatures (Kelvin)	Contaminant- specific	See Chemical-specific hierarchy

Ть	Normal Boiling Point (Kelvin)	Contaminant- specific	See Chemical-specific hierarchy
n	If $(T_b/T_C < 0.57)$ If $(T_b/T_C > 0.71)$ If $(0.57 < T_b/T_C \le 0.71)$	n = 0.3 n = 0.41 $n = (0.74 \text{ x T}_{b}/T_{C} - 0.116)$	U.S. EPA <u>Fact Sheet</u> Unitless exponent values used to determine $\Delta$ Hv,gw
	Ingestion and	Dermal Contact Rates	
IRW <sub>res-c</sub>	Resident Drinking Water Ingestion Rate - Child (L/day)	0.78	U.S. EPA 2011, Tables 3-15 and 3-33; weighted average of 90th percentile consumer-only ingestion of drinking water (birth to <6 years)
IRW <sub>res-a</sub>	Resident Drinking Water Ingestion Rate - Adult (L/day)	2.5	U.S. EPA 2011, Table 3-33; 90th percentile of consumer-only ingestion of drinking water (>= 21 years)
IFW <sub>res-adj</sub>	Resident Drinking Water Ingestion Rate - Age- adjusted (L/kg)	327.95	Calculated using the age adjusted intake factors equation
IFWM <sub>res-</sub> adj	Resident Mutagenic Drinking Water Ingestion Rate - Age-adjusted (L/kg)	1019.9	Calculated using the age adjusted intake factors equation
IRS <sub>res-c</sub>	Resident Soil Ingestion Rate - Child (mg/day)	200	U.S. EPA 1991a (pg. 15)
IRS <sub>res-a</sub>	Resident Soil Ingestion Rate - Adult (mg/day)	100	U.S. EPA 1991a (pg. 15)
IFS <sub>res-adj</sub>	Resident Soil Ingestion Rate - Age-adjusted (mg/kg)	36750	Calculated using the age adjusted intake factors equation

IFSM <sub>res-adj</sub>	Resident Mutagenic Soil Ingestion Rate - Age- adjusted (mg/kg)	166833.33	Calculated using the age adjusted intake factors equation
IR <sub>iw</sub>	Indoor Worker Soil Ingestion Rate (mg/day)	50	U.S. EPA 1991a (pg. 15)
IR <sub>ow</sub>	Outdoor Worker Soil Ingestion Rate (mg/day)	100	U.S. EPA 1991a (pg. 15)
IR <sub>cw</sub>	Construction Worker Soil Ingestion Rate (mg/day)	330	U.S. EPA 2002 Exhibit 5-1
IRw	Composite Worker Soil Ingestion Rate (mg/day)	100	U.S. EPA 1991a (pg. 15)
IRW <sub>rec-c</sub>	Recreator Surface Water Ingestion Rate - Child (L/hour)	0.12	U.S. EPA 2011, Table 3.5
IRW <sub>rec-a</sub>	Recreator Surface Water Ingestion Rate - Adult (L/hour)	0.071	U.S. EPA 2011, Table 3.5
IFW <sub>rec-adj</sub>	Recreator Surface Water Ingestion Rate - Age- adjusted (L/kg)	Site-specific	Calculated using the age adjusted intake factors equation
IRW <sub>0-2</sub>	Surface Water Ingestion Rate - Age Segment 0-2 (L/hour)	0.12	U.S. EPA 2011, Table 3.5
IRW <sub>2-6</sub>	Surface Water Ingestion Rate - Age Segment 2-6 (L/hour)	0.12	U.S. EPA 2011, Table 3.5
IRW <sub>6-16</sub>	Surface Water Ingestion Rate - Age Segment 6-16 (L/hour)	0.071	U.S. EPA 2011, Table 3.5

IRW16-26	Surface Water Ingestion Rate - Age Segment 16- 26 (L/hour)	0.071	U.S. EPA 2011, Table 3.5
IFWM <sub>rec-</sub> adj	Recreator Mutagenic Surface Water Ingestion Rate - Age-adjusted (L/kg)	Site-specific	Calculated using the age adjusted intake factors equation
IRS <sub>rec-c</sub>	Recreator Soil Ingestion Rate - Child (mg/day)	200	U.S. EPA 1991a (pg. 15)
IRS <sub>rec-a</sub>	Recreator Soil Ingestion Rate - Adult (mg/day)	100	U.S. EPA 1991a (pg. 15)
IFS <sub>rec-adj</sub>	Recreator Soil Ingestion Rate - Age-adjusted (mg/kg)	Site-specific	Calculated using the age adjusted intake factors equation
IRS <sub>0-2</sub>	Soil Ingestion Rate - Age- segment 0-2 (mg/day)	200	U.S. EPA 1991a (pg. 15)
IRS <sub>2-6</sub>	Soil Ingestion Rate - Age- segment 2-6 (mg/day)	200	U.S. EPA 1991a (pg. 15)
IRS <sub>6-16</sub>	Soil Ingestion Rate - Age- segment 6-16 (mg/day)	100	U.S. EPA 1991a (pg. 15)
IRS <sub>16-26</sub>	Soil Ingestion Rate - Age- segment 16-26 (mg/day)	100	U.S. EPA 1991a (pg. 15)
IFSM <sub>rec-adj</sub>	Recreator Mutagenic Soil Ingestion Rate - Age- adjusted (mg/kg)	Site-specific	Calculated using the age adjusted intake factors equation
DFS <sub>res-adj</sub>	Resident soil dermal contact factor- age- adjusted (mg/kg)	103390	Calculated using the age adjusted intake factors equation
DFSM <sub>res-</sub> adj	Resident Mutagenic soil dermal contact factor- age-adjusted (mg/kg)	428260	Calculated using the age adjusted intake factors equation

DFS <sub>rec-adj</sub>	Recreator soil dermal contact factor- age- adjusted (mg/kg)	Site-specific	Calculated using the age adjusted intake factors equation
DFSM <sub>rec-</sub> adj	Recreator Mutagenic soil dermal contact factor- age-adjusted (mg/kg)	Site-specific	Calculated using the age adjusted intake factors equation
DFW <sub>res-adj</sub>	Resident water dermal contact factor- age- adjusted (cm <sup>2</sup> - event/kg)	2610650	Calculated using the age adjusted intake factors equation
DFWM <sub>res-</sub> adj	Resident Mutagenic water dermal contact factor- age-adjusted (cm <sup>2</sup> - event/kg)	8191633	Calculated using the age adjusted intake factors equation
DFW <sub>rec-adj</sub>	Recreator water dermal contact factor- age- adjusted (cm <sup>2</sup> - event/kg)	Site-specific	Calculated using the age adjusted intake factors equation
DFWM <sub>rec-</sub> adj	Recreator Mutagenic water dermal contact factor- age-adjusted (cm <sup>2</sup> - event/kg)	Site-specific	Calculated using the age adjusted intake factors equation
IRF <sub>res-a</sub>	Fish Ingestion Rate (mg/day)	Site-specific	Recommend using site- specific values
SA <sub>res-c</sub>	Resident surface area soil - child (cm <sup>2</sup> /day)	2373	U.S. EPA 2011a, Tables 7-2 and 7-8; weighted average of mean values for head, hands, forearms, lower legs, and feet (male and female, birth to $< 6$ years)(forearm and lower leg-specific data used when available, ratios for nearest available age group used elsewhere)

SA <sub>res-a</sub>	Resident surface area soil - adult (cm <sup>2</sup> /day)	6032	U.S. EPA 2011, Tables 7-2 and 7-12; weighted average of mean values for head, hands, forearms, lower legs, and feet (male and female, 21+ years) (forearm and lower leg- specific data used for males and female lower leg; ratio of male forearm to arm applied to female arm data.
SA <sub>res-c</sub>	Resident surface area water - child (cm <sup>2</sup> )	6365	U.S. EPA 2014, weighted average of mean values for children <6 years.
SA <sub>res-a</sub>	Resident surface area water - adult (cm <sup>2</sup> )	19652	U.S. EPA 2014, weighted average of mean values for adults, male and female 21+.
SA <sub>ow</sub>	Outdoor Worker soil surface area - adult (cm <sup>2</sup> /day)	3527	US EPA 2011a, Table 7-2; weighted average of mean values for head, hands, and forearms (male and female, 21+years)
SA <sub>cw</sub>	Construction Worker soil surface area - adult (cm <sup>2</sup> /day)	3527	US EPA 2011a, Table 7-2; weighted average of mean values for head, hands, and forearms (male and female, 21+years)
SAw	Composite Worker soil surface area - adult (cm <sup>2</sup> /day)	3527	US EPA 2011a, Table 7-2; weighted average of mean values for head, hands, and forearms (male and female, 21+years)

SA <sub>rec-c</sub>	Recreator surface area soil - child (cm <sup>2</sup> /day)	2373	U.S. EPA 2011a, Tables 7-2 and 7-8; weighted average of mean values for head, hands, forearms, lower legs, and feet (male and female, birth to $< 6$ years)(forearm and lower leg-specific data used when available, ratios for nearest available age group used elsewhere)
SA <sub>rec-a</sub>	Recreator surface area soil - adult (cm <sup>2</sup> /day)	6032	U.S. EPA 2011, Tables 7-2 and 7-12; weighted average of mean values for head, hands, forearms, lower legs, and feet (male and female, 21+ years) (forearm and lower leg- specific data used for males and female lower leg; ratio of male forearm to arm applied to female arm data.
SA <sub>rec-c</sub>	Recreator surface area water - child (cm <sup>2</sup> )	6365	U.S. EPA 2014, weighted average of mean values for children <6 years.
SA <sub>rec-a</sub>	Recreator surface area water - adult (cm <sup>2</sup> )	19652	U.S. EPA 2014, weighted average of mean values for adults, male and female 21+.
SA <sub>0-2</sub>	Resident/Recreator surface area soil - age segment 0-2 (cm <sup>2</sup> /day)	2373	U.S. EPA 2011a, Tables 7-2 and 7-8; weighted average of mean values for head, hands, forearms, lower legs, and feet (male and female, birth to < 6 years)(forearm and lower leg-specific data used when available, ratios for nearest available age group used elsewhere)

SA <sub>2-6</sub>	Resident/Recreator surface area soil - age segment 2-6 (cm <sup>2</sup> /day)	2373	U.S. EPA 2011a, Tables 7-2 and 7-8; weighted average of mean values for head, hands, forearms, lower legs, and feet (male and female, birth to $< 6$ years)(forearm and lower leg-specific data used when available, ratios for nearest available age group used elsewhere)
SA <sub>6-16</sub>	Resident/Recreator surface area soil - age segment 6-16 (cm <sup>2</sup> /day)	6032	U.S. EPA 2011, Tables 7-2 and 7-12; weighted average of mean values for head, hands, forearms, lower legs, and feet (male and female, 21+ years) (forearm and lower leg- specific data used for males and female lower leg; ratio of male forearm to arm applied to female arm data.
SA <sub>16-26</sub>	Resident/Recreator surface area soil - age segment 16-26 (cm <sup>2</sup> /day)	6032	U.S. EPA 2011, Tables 7-2 and 7-12; weighted average of mean values for head, hands, forearms, lower legs, and feet (male and female, 21+ years) (forearm and lower leg- specific data used for males and female lower leg; ratio of male forearm to arm applied to female arm data.
SA <sub>0-2</sub>	Resident/Recreator surface area water - age segment 0-2 (cm <sup>2</sup> )	6365	U.S. EPA 2014, weighted average of mean values for children <6 years.
SA <sub>2-6</sub>	Resident/Recreator surface area water - age segment 2-6 (cm <sup>2</sup> )	6365	U.S. EPA 2014, weighted average of mean values for children <6 years.

SA <sub>6-16</sub>	Resident/Recreator surface area water - age segment 6-16 (cm <sup>2</sup> )	19652	U.S. EPA 2014, weighted average of mean values for adults, male and female 21+.
SA <sub>16-26</sub>	Resident/Recreator surface area water - age segment 16-26 (cm <sup>2</sup> )	19652	U.S. EPA 2014, weighted average of mean values for adults, male and female 21+.
AF <sub>res-c</sub>	Resident soil adherence factor - child (mg/cm <sup>2</sup> )	0.2	U.S. EPA 2002 (Exhibit 1-2)
AF <sub>res-a</sub>	Resident soil adherence factor - adult (mg/cm <sup>2</sup> )	0.07	U.S. EPA 2002 (Exhibit 1-2)
AF <sub>ow</sub>	Outdoor Worker soil adherence factor (mg/cm <sup>2</sup> )	0.12	U.S. EPA 2011, Table 7-20 and Section 7.2.2; arithmetic mean of weighted average of body part- specific (hands, forearms, and face) mean adherence factors for adult commercial/industrial activities
AFw	Composite Worker soil adherence factor (mg/cm <sup>2</sup> )	0.12	U.S. EPA 2011, Table 7-20 and Section 7.2.2; arithmetic mean of weighted average of body part- specific (hands, forearms, and face) mean adherence factors for adult commercial/industrial activities
AF <sub>cw</sub>	Construction Worker soil adherence factor (mg/cm <sup>2</sup> )	0.3	U.S. EPA 2002 (Exhibit 5-1)
AF <sub>rec-c</sub>	Recreator soil adherence factor - child (mg/cm <sup>2</sup> )	0.2	U.S. EPA 2002 (Exhibit 1-2)

			r
AF <sub>rec-a</sub>	Recreator soil adherence factor - adult (mg/cm <sup>2</sup> )	0.07	U.S. EPA 2002 (Exhibit 1-2)
AF <sub>0-2</sub>	Resident/Recreator soil adherence factor - age segment 0-2 (mg/cm <sup>2</sup> )	0.2	U.S. EPA 2002 (Exhibit 1-2)
AF <sub>2-6</sub>	Resident/Recreator soil adherence factor - age segment 2-6 (mg/cm <sup>2</sup> )	0.2	U.S. EPA 2002 (Exhibit 1-2)
AF <sub>6-16</sub>	Resident/Recreator soil adherence factor - age segment 6-16 (mg/cm <sup>2</sup> )	0.07	U.S. EPA 2002 (Exhibit 1-2)
AF <sub>16-26</sub>	Resident/Recreator soil adherence factor - age segment 16-26 (mg/cm <sup>2</sup> )	0.07	U.S. EPA 2002 (Exhibit 1-2)
BW <sub>res-c</sub>	Resident Body Weight - child (kg)	15	U.S. EPA 1991a (pg. 15)
BW <sub>res-a</sub>	Resident Body Weight - adult (kg)	80	U.S. EPA 2011, Table 8-3; weighted mean values for adults 21 - 78
BW <sub>rec-c</sub>	Recreator Body Weight - child (kg)	15	U.S. EPA 1991a (pg. 15)
BW <sub>rec-a</sub>	Recreator Body Weight - adult (kg)	80	U.S. EPA 2011, Table 8-3; weighted mean values for adults 21 - 78
BW <sub>0-2</sub>	Resident/Recreator Body Weight - age segment 0-2 (kg)	15	U.S. EPA 1991a (pg. 15)
BW <sub>2-6</sub>	Resident/Recreator Body Weight - age segment 2-6 (kg)	15	U.S. EPA 1991a (pg. 15)

			u-
BW <sub>6-16</sub>	Resident/Recreator Body Weight - age segment 6-16 (kg)	80	U.S. EPA 2011, Table 8-3; weighted mean values for adults 21 - 78
BW <sub>16-26</sub>	Resident/Recreator Body Weight - age segment 16- 26 (kg)	80	U.S. EPA 2011, Table 8-3; weighted mean values for adults 21 - 78
BW <sub>ow</sub>	Outdoor Worker Body Weight (kg)	80	U.S. EPA 1991a (pg. 15)
BW <sub>cw</sub>	Construction Worker Body Weight (kg)	80	U.S. EPA 2011, Table 8-3; weighted mean values for adults 21 - 78
$\mathrm{BW}_{\mathrm{iw}}$	Indoor Worker Body Weight (kg)	80	U.S. EPA 2011, Table 8-3; weighted mean values for adults 21 - 78
BWw	Composite Worker Body Weight (kg)	80	U.S. EPA 2011, Table 8-3; weighted mean values for adults 21 - 78
ABS <sub>d</sub>	Fraction of contaminant absorbed dermally from soil (unitless)	Contaminant- specific Inorganic default = none VOC default = none SVOC default = 0.1	U.S. EPA 2004 (Exhibit 3-4 and section 3.2.2.4)
GIABS	Fraction of contaminant absorbed in gastrointestinal tract (unitless) Note: if the GIABS is >50% then it is set to 100% for the calculation of dermal toxicity values.	Contaminant- specific Inorganic default = 1.0 VOC default = 1.0 SVOC default = 1.0	U.S. EPA 2004 (Exhibit 4-1 and section 4.2)
DA <sub>event</sub>	Absorbed dose per event $(\mu g/cm^2 - event)$	Contaminant- specific	U.S. EPA 2004 (Equation 3.2 and 3.3)

Exposure Frequency, Exposure Duration, and Exposure Time Variables			
EF <sub>res</sub>	Resident Exposure Frequency (days/year)	350	U.S. EPA 1991a (pg. 15)
EF <sub>res-a</sub>	Resident Exposure Frequency - adult (days/year)	350	U.S. EPA 1991a (pg. 15)
EF <sub>res-c</sub>	Resident Exposure Frequency - child (days/year)	350	U.S. EPA 1991a (pg. 15)
$\mathrm{EF}_{\mathrm{w}}$	Composite Worker Exposure Frequency (days/year)	250	U.S. EPA 1991a (pg. 15)
EF <sub>iw</sub>	Indoor Worker Exposure Frequency (days/year)	250	U.S. EPA 1991a (pg. 15)
EF <sub>ow</sub>	Outdoor Worker Exposure Frequency (days/year)	225	U.S. EPA 2002 (Exhibit 1-2)
EF <sub>cw</sub>	Construction Worker Exposure Frequency (days/year)	250	U.S. EPA 2002 Exhibit 5-1
EF <sub>rec</sub>	Recreator Exposure Frequency (days/year)	Site-specific	Site-specific
EF <sub>rec-c</sub>	Recreator Exposure Frequency - child (days/year)	Site-specific	Site-specific
EF <sub>rec-a</sub>	Recreator Exposure Frequency - adult (days/year)	Site-specific	Site-specific
EF <sub>0-2</sub>	Resident/Recreator Exposure Frequency - age segment 0-2 (days/year)	Resident - 350 Recreator - Site- specific	Resident - U.S. EPA 1991a (pg. 15) Recreator - Site-specific

EF <sub>2-6</sub>	Resident/Recreator Exposure Frequency - age segment 2-6 (days/year)	Resident - 350 Recreator - Site- specific	Resident - U.S. EPA 1991a (pg. 15) Recreator - Site-specific
EF <sub>6-16</sub>	Resident/Recreator Exposure Frequency - age segment 6-16 (days/year)	Resident - 350 Recreator - Site- specific	Resident - U.S. EPA 1991a (pg. 15) Recreator - Site-specific
EF <sub>16-26</sub>	Resident/Recreator Exposure Frequency - age segment 16-26 (days/year)	Resident - 350 Recreator - Site- specific	Resident - U.S. EPA 1991a (pg. 15) Recreator - Site-specific
ED <sub>res</sub>	Resident Exposure Duration (years)	26	EPA 2011, Table 16-108; 90th percentile for current residence time.
ED <sub>res-c</sub>	Resident Exposure Duration - child (years)	6	U.S. EPA 1991a (pg. 15)
ED <sub>res-a</sub>	Resident Exposure Duration - adult (years)	20	ED <sub>res</sub> (26 years) - ED <sub>res-c</sub> (6 years)
EDw	Composite Worker Exposure Duration - (years)	25	U.S. EPA 1991a (pg. 15)
ED <sub>iw</sub>	Indoor Worker Exposure Duration - (years)	25	U.S. EPA 1991a (pg. 15)
ED <sub>ow</sub>	Outdoor Worker Exposure Duration (years)	25	U.S. EPA 1991a (pg. 15)
ED <sub>cw</sub>	Construction Worker Exposure Duration (years)	1	U.S. EPA 2002 Exhibit 5-1
ED <sub>rec</sub>	Recreator Exposure Duration (years)	26	EPA 2011, Table 16-108; 90th percentile for current residence time.

ED <sub>rec-c</sub>	Recreator Exposure Duration - child (years)	6	U.S. EPA 1991a (pg. 15)
ED <sub>rec-a</sub>	Recreator Exposure Duration - adult (years)	20	ED <sub>rec</sub> (26 years) - ED <sub>rec-c</sub> (6 years)
ED <sub>0-2</sub>	Resident/Recreator Exposure Duration - age segment 0-2 (years)	2	U.S. EPA 2005 (pg. 37)
ED <sub>2-6</sub>	Resident/Recreator Exposure Duration - age segment 2-6 (years)	4	U.S. EPA 2005 (pg. 37)
ED <sub>6-16</sub>	Resident/Recreator Exposure Duration - age segment 6-16 (years)	10	U.S. EPA 2005 (pg. 37)
ED <sub>16-26</sub>	Resident/Recreator Exposure Duration - age segment 16-26 (years)	10	U.S. EPA 2005 (pg. 37)
ET <sub>res-a</sub>	Resident Exposure Time (hours/day)	24	The whole day
ET <sub>res-c</sub>	Resident Exposure Time (hours/day)	24	The whole day
ET <sub>res</sub>	Resident Exposure Time (hours/day)	24	The whole day
ETw	Composite Worker Exposure Time (hours/day)	8	The work day
ET <sub>iw</sub>	Indoor Worker Exposure Time (hours/day)	8	The work day
ET <sub>ow</sub>	Outdoor Worker Exposure Time (hours/day)	8	The work day

			u
ET <sub>cw</sub>	Construction Worker Exposure Time (hours/day)	8	The work day
ET <sub>rec</sub>	Recreator Exposure Time (hours/day)	Site-specific	Site-specific
ET <sub>rec-c</sub>	Recreator Exposure Time - child (hours/day)	Site-specific	Site-specific
ET <sub>rec-a</sub>	Recreator Exposure Time - adult (hours/day)	Site-specific	Site-specific
ET <sub>event-res-c</sub>	Resident Water Exposure Time - child (hours/event)	0.54	U.S. EPA 2011, Table 16- 28; weighted average of 90th percentile time spent bathing (birth to <6 years)
ET <sub>event-res-a</sub>	Resident Water Exposure Time - adult (hours/event)	0.71	U.S. EPA 2011, Tables 16- 30 and 16-31; weighted average of adult (21 to 78) 90th percentile of time spent bathing/ showering in a day, divided by mean number of baths/showers taken in a day.
ET <sub>event-res-</sub> adj	Resident Water Exposure Time - age-adjusted (hours/event)	0.6708	Calculated using the age adjusted intake factors equation
ET <sub>event</sub> -res- madj	Resident Exposure Time - age-adjusted (hours/event)	0.6708	Calculated using the age adjusted intake factors equation
ET <sub>event-rec-c</sub>	Recreator Surface Water Exposure Time - child (hours/event)	Site-specific	Site-specific
ET <sub>event-rec-a</sub>	Recreator Surface Water Exposure Time - adult (hours/event)	Site-specific	Site-specific

ET <sub>0-2</sub>	Resident/Recreator Exposure Time - age segment 0-2 (hours/day)	Resident - 24 Recreator - Site- specific	Resident - The whole day Recreator - Site-specific
ET <sub>2-6</sub>	Resident/Recreator Exposure Time - age segment 2-6 (hours/day)	Resident - 24 Recreator - Site- specific	Resident - The whole day Recreator - Site-specific
ET <sub>6-16</sub>	Resident/Recreator Exposure Time - age segment 6-16 (hours/day)	Resident - 24 Recreator - Site- specific	Resident - The whole day Recreator - Site-specific
ET <sub>16-26</sub>	Resident/Recreator Exposure Time - age segment 16-26 (hours/day)	Resident - 24 Recreator - Site- specific	Resident - The whole day Recreator - Site-specific
ET <sub>event</sub> -rec- adj	Recreator Exposure Time - age-adjusted (hours/event)	Site-specific	Calculated using the age adjusted intake factors equation
ET <sub>event-rec</sub> (0-2)	Recreator Exposure Time - age segment 0-2 (hours/event)	Site-specific	Site-specific
ET <sub>event-rec</sub> (2-6)	Recreator Exposure Time - age segment 2-6 (hours/event)	Site-specific	Site-specific
ET <sub>event</sub> -rec (6-16)	Recreator Exposure Time - age segment 6-16 (hours/event)	Site-specific	Site-specific
ET <sub>event-rec</sub> (16-26)	Recreator Exposure Time - age segment 16-26 (hours/event)	Site-specific	Site-specific
ET <sub>event-res</sub> (0-2)	Resident Exposure Time - age segment 0-2 (hours/event)	0.54	Calculated based on the ET given for ETevent-res-c

ET <sub>event-res</sub> (2-6)	Resident Exposure Time - age segment 2-6 (hours/event)	0.54	Calculated based on the ET given for ETevent-res-c
ET <sub>event-res</sub> (6-16)	Resident Exposure Time - age segment 6-16 (hours/event)	0.71	Calculated based on the ET given for ETevent-res-a
ET <sub>event-res</sub> (16-26)	Resident Exposure Time - age segment 16-26 (hours/event)	0.71	Calculated based on the ET given for ETevent-res-a
ET <sub>event-rec-</sub> madj	Recreator Exposure Time - age-adjusted (hours/event)	Site-specific	Calculated using the age adjusted intake factors equation
EV <sub>rec-c</sub>	Recreator Events - child (events/day)	Site-specific	Site-specific
EV <sub>rec-a</sub>	Recreator Events - adult (events/day)	Site-specific	Site-specific
EV <sub>res-c</sub>	Resident Events - child (events/day)	1	U.S. EPA 2004; Exhibit 3-2
EV <sub>res-a</sub>	Resident Events - adult (events/day)	1	U.S. EPA 2004; Exhibit 3-2
EV <sub>0-2</sub>	Resident/Recreator Events - age segment 0-2 (events/day)	Resident - 1 Recreator - Site- specific	U.S. EPA 2004; Exhibit 3-2
EV <sub>2-6</sub>	Resident/Recreator Events - age segment 2-6 (events/day)	Resident - 1 Recreator - Site- specific	U.S. EPA 2004; Exhibit 3-2
EV <sub>6-16</sub>	Resident/Recreator Events - age segment 6-16 (events/day)	Resident - 1 Recreator - Site- specific	U.S. EPA 2004; Exhibit 3-2

EV <sub>16-26</sub>	Resident/Recreator Events - age segment 16- 26 (events/day)	Resident - 1 Recreator - Site- specific	U.S. EPA 2004; Exhibit 3-2
	Soil to Groundw	ater SSL Factor Varial	bles
Cw	Target soil leachate concentration (mg/L)	nonzero MCL or RSL × DAF	U.S. EPA. 2002 Equation 4-14
DAF	Dilution attenuation factor (unitless)	1 (or site-specific)	U.S. EPA. 2002 Equation 4-11
ED	Exposure duration	70	U.S. EPA. 2002 Equation 4-14
Ι	Infiltration Rate (m/year)	0.18	U.S. EPA. 2002 Equation 4-11
L	source length parallel to ground water flow (m)	site-specific	U.S. EPA. 2002 Equation 4-11
i	hydraulic gradient (m/m)	site-specific	U.S. EPA. 2002 Equation 4-11
К	aquifer hydraulic conductivity (m/year)	site-specific	U.S. EPA. 2002 Equation 4-11
$\theta_{\rm w}$	water-filled soil porosity (L <sub>water</sub> /L <sub>soil</sub> )	0.3	U.S. EPA. 2002 Equation 4-10
$\theta_a$	air-filled soil porosity (L <sub>air</sub> /L <sub>soil</sub> )	$= n - \theta_w$	U.S. EPA. 2002 Equation 4-10
n	total soil porosity (L <sub>pore</sub> /L <sub>soil</sub> )	$= 1 - (\rho_b / \rho_s)$	U.S. EPA. 2002 Equation 4-10
$\rho_s$	soil particle density (Kg/L)	2.65	U.S. EPA. 2002 Equation 4-10
ρ <sub>b</sub>	dry soil bulk density (kg/L)	1.5	U.S. EPA. 2002 Equation 4-10

H'	Dimensionless Henry Law Constant (unitless)	Contaminant- specific	See Chemical-specific hierarchy
K <sub>d</sub>	soil-water partition coefficient (L/kg)	$= K_{oc} * f_{oc} \text{ for}$ organics	U.S. EPA. 2002 Equation 4-10
K <sub>oc</sub>	soil organic carbon/water partition coefficient (L/kg)	Contaminant- specific	See Chemical-specific hierarchy
f <sub>oc</sub>	fraction organic carbon in soil (g/g)	0.002	U.S. EPA. 2002 Equation 4-10
d <sub>a</sub>	aquifer thickness (m)	site-specific	U.S. EPA. 2002 Equation 4-10
ds	depth of source (m)	site-specific	U.S. EPA. 2002 Equation 4-10
d	mixing zone depth (m)	site-specific	U.S. EPA. 2002 Equation 4-12
	Wind Particulate	Emission Factor Varia	bles
PEF	Particulate Emission Factor - Minneapolis (m <sup>3</sup> /kg)	1.36 x 10 <sup>9</sup> (region-specific)	U.S. EPA 2002 Exhibit D-2
Q/C <sub>wind</sub>	Inverse of the Mean Concentration at the Center of a $0.5$ -Acre- Square Source (g/m <sup>2</sup> -s per kg/m <sup>3</sup> )	93.77 (region- specific)	U.S. EPA 2002 Exhibit D-2
V	Fraction of Vegetative Cover (unitless)	0.5	U.S. EPA. 2002 Equation 4-5
Um	Mean Annual Wind Speed (m/s)	4.69	U.S. EPA. 2002 Equation 4-5

Ut	Equivalent Threshold Value of Wind Speed at 7m (m/s)	11.32	U.S. EPA. 2002 Equation 4-5
F(x)	Function Dependent on $U_m / U_t$ (unitless)	0.194	U.S. EPA. 2002 Equation 4-5
А	Dispersion constant unitless	PEF and region- specific	U.S. EPA 2002 Exhibit D-2
A <sub>s</sub>	Areal extent of the site or contamination (acres)	0.5 (range 0.5 to 500 )	U.S. EPA 2002 Exhibit D-2
В	Dispersion constant unitless	PEF and region- specific	U.S. EPA 2002 Exhibit D-2
С	Dispersion constant unitless	PEF and region- specific	U.S. EPA 2002 Exhibit D-2
N	Iechanical Particulate Emissi	on Factor Variables fro	m Vehicle Traffic
PEF <sub>sc</sub>	Particulate Emission Factor - subchronic (m <sup>3</sup> /kg)	(site-specific)	U.S. EPA 2002 Equation 5-5
Q/C <sub>sr</sub>	Inverse of the ratio of the 1-h geometric mean concentration to the emission flux along a straight road segment bisecting a square site $(g/m^2$ -s per kg/m <sup>3</sup> )	23.02 (for 0.5 acre site)	U.S. EPA 2002 Equation 5-5
F <sub>D</sub>	Dispersion correction factor (unitless)	0.185	U.S. EPA 2002 Equation E-16
Т	Total time over which construction occurs (s)	site-specific	U.S. EPA 2002 Equation 5-5
A <sub>R</sub>	Surface area of contaminated road segment (m <sup>2</sup> )	$(A_R = L_R * W_R * 0.092903 m^2/ft^2)$	U.S. EPA 2002 Equation 5-5

	1			
L <sub>R</sub>	Length of road segment (ft)	Site-specific	U.S. EPA 2002 Equation 5-5	
W <sub>R</sub>	Width of road segment (ft)	20	U.S. EPA 2002 Equation E-18	
W	Mean vehicle weight (tons)	(number of cars x tons/car + number of trucks x tons/truck) / total vehicles)	U.S. EPA 2002 Equation 5-5	
р	Number of days with at least 0.01 inches of precipitation (days/year)	Site-specific	U.S. EPA 2002 Exhibit 5-2	
Σνκτ	Sum of fleet vehicle kilometers traveled during the exposure duration (km)	$\sum$ VKT = total vehicles x distance (km/day) x frequency (weeks/year) x (days/year)	U.S. EPA 2002 Equation 5-5	
A	Dispersion constant unitless	12.9351	U.S. EPA 2002 Equation 5-6	
A <sub>s</sub>	Areal extent of site surface soil contamination (acres)	0.5 (range 0.5 to 500 )	U.S. EPA 2002 Equation 5-6	
В	Dispersion constant unitless	5.7383	U.S. EPA 2002 Equation 5-6	
С	Dispersion constant unitless	71.7711	U.S. EPA 2002 Equation 5-6	
Mecha	Mechanical Particulate Emission Factor Variables from other than Vehicle Traffic			
PEF'sc	Particulate Emission Factor - subchronic (m <sup>3</sup> /kg)	(site-specific)	U.S. EPA 2002 Equation E-26	

Q/C <sub>sa</sub>	Inverse of the ratio of the 1-h. geometric mean air concentration and the emission flux at the center of the square emission source $(g/m^2$ -s per kg/m <sup>3</sup> )	Site-specific	U.S. EPA 2002 Equation E-15
FD	Dispersion correction factor (unitless)	Site-specific	U.S. EPA 2002 Equation E-16
А	Dispersion constant unitless	2.4538	U.S. EPA 2002 Equation E-15
В	Dispersion constant unitless	17.5660	U.S. EPA 2002 Equation E-15
С	Dispersion constant unitless	189.0426	U.S. EPA 2002 Equation E-15
As	Areal extent of site surface soil contamination (acres)	(range 0.5 to 500)	U.S. EPA 2002 Equation E-15
J' <sub>T</sub>	Total time-averaged $PM_{10}$ unit emission flux for construction activities other than traffic on unpaved roads (g/m <sup>2</sup> -s)	Site-specific	U.S. EPA 2002 Equation E-25
M <sup>PC</sup> <sub>wind</sub>	Unit mass emitted from wind erosion (g)	site-specific	U.S. EPA 2002 Equation E-20
V	Fraction of Vegetative Cover (unitless)	0	U.S. EPA 2002 Equation E-20
Um	Mean Annual Wind Speed (m/s)	4.69	U.S. EPA 2002 Equation E-20
Ut	Equivalent Threshold Value of Wind Speed at 7m (m/s)	11.32	U.S. EPA 2002 Equation E-20

			u
F(x)	Function Dependent on $U_m / U_t$ (unitless)	0.194	U.S. EPA 2002 Equation E-20
A <sub>surf</sub>	Areal extent of site surface soil contamination (m <sup>2</sup> )	(range 0.5 to 500)	U.S. EPA 2002 Equation E-20
ED	Exposure duration (years)	Site-specific	U.S. EPA 2002 Equation E-20
M <sub>excav</sub>	Unit mass emitted from excavation soil dumping (g)	site-specific	U.S. EPA 2002 Equation E-21
0.35	PM <sub>10</sub> particle size multiplier (unitless)	0.35	U.S. EPA 2002 Equation E-21
Um	Mean annual wind speed during construction (m/s)	4.69	U.S. EPA 2002 Equation E-21
M <sub>m-excav</sub>	Gravimetric soil moisture content (%)	12 (mean value for municipal landfill cover)	U.S. EPA 2002 Equation E-21
$ ho_{soil}$	In situ soil density (includes water) (mg/m <sup>3</sup> )	1.68	U.S. EPA 2002 Equation E-21
A <sub>excav</sub>	Areal extent of excavation $(m^2)$	(range 0.5 to 500)	U.S. EPA 2002 Equation E-21
d <sub>excav</sub>	Average depth of excavation (m)	Site-specific	U.S. EPA 2002 Equation E-21
N <sub>A-dump</sub>	Number of times soil is dumped (unitless)	2	U.S. EPA 2002 Equation E-21
M <sub>doz</sub>	Unit mass emitted from dozing operations (g)	site-specific	U.S. EPA 2002 Equation E-22
0.75	PM <sub>10</sub> scaling factor (unitless)	0.75	U.S. EPA 2002 Equation E-22

S <sub>doz</sub>	Soil silt content (%)	6.9	U.S. EPA 2002 Equation E-22
M <sub>m-doz</sub>	Gravimetric soil moisture content (%)	7.9 (mean value for overburden)	U.S. EPA 2002 Equation E-22
∑VKT <sub>doz</sub>	Sum of dozing kilometers traveled (km)	Site-specific	U.S. EPA 2002 Equation E-22
S <sub>doz</sub>	Average dozing speed (kph)	11.4 (mean value for graders)	U.S. EPA 2002 Equation E-22
N <sub>A-doz</sub>	Number of times site is dozed (unitless)	Site-specific	U.S. EPA 2002 Equation E-22
B <sub>d</sub>	Dozer blade length (m)	Site-specific	U.S. EPA 2002 Page E-28
M <sub>grade</sub>	Unit mass emitted from grading operations (g)	site-specific	U.S. EPA 2002 Equation E-23
0.60	PM10 scaling factor (unitless)	0.60	U.S. EPA 2002 Equation E-23
∑VKT <sub>grade</sub>	Sum of grading kilometers traveled (km)		U.S. EPA 2002 Equation E-23
S <sub>grade</sub>	Average grading speed (kph)	11.4 (mean value for graders)	U.S. EPA 2002 Equation E-23
N <sub>A-grade</sub>	Number of times site is graded (unitless)	Site-specific	U.S. EPA 2002 Equation E-23
Bg	Grader blade length (m)	Site-specific	U.S. EPA 2002 Page E-28
M <sub>till</sub>	Unit mass emitted from tilling operations (g)	site-specific	U.S. EPA 2002 Equation E-24
S <sub>till</sub>	Soil silt content (%)	18	U.S. EPA 2002 Equation E-24

A <sub>c-till</sub>	Areal extent of tilling (acres)	Site-specific	U.S. EPA 2002 Equation E-24
A <sub>c-grade</sub>	Areal extent of grading (acres)	Site-specific	Necessary to solve $\sum VKT_{grade}$ in U.S. EPA 2002 Equation E-23
A <sub>c-doz</sub>	Areal extent of dozinging (acres)	Site-specific	Necessary to solve $\sum VKT_{grade}$ in U.S. EPA 2002 Equation E-22
N <sub>A-till</sub>	Number of times soil is tilled (unitless)	2	U.S. EPA 2002 Equation E-24
	Chronic Volatilization Facto	or and Soil Saturation L	limit Variables
VF <sub>ulim</sub>	Volatilization Factor - Los Angeles (m <sup>3</sup> /kg)	Contaminant- specific	U.S. EPA. 2002 Equation 4-8
C <sub>sat</sub>	Soil saturation concentration (mg/kg)	Contaminant- specific	U.S. EPA. 2002 Equation 4-9
Q/C <sub>vol</sub>	Inverse of the Mean Concentration at the Center of a 0.5-Acre-Square Source $(g/m^2$ -s per kg/m <sup>3</sup> )	68.18	U.S. EPA. 2002 Equation 4-8
А	Dispersion constant unitless	11.9110 (region- specific)	U.S. EPA 2002 Exhibit D-3
As	Areal extent of the site contamination (acres)	0.5 (range 0.5 to 500 )	U.S. EPA 2002 Equation 4-8
В	Dispersion constant unitless	18.4385 (region- specific)	U.S. EPA 2002 Exhibit D-3
С	Dispersion constant unitless	209.7845 (region- specific)	U.S. EPA 2002 Exhibit D-3

D <sub>A</sub>	Apparent Diffusivity (cm <sup>2</sup> /s)	Contaminant- specific	U.S. EPA. 2002 Equation 4-8		
Т	Exposure interval (s)	8.2×108 (used for unlimited source model)	U.S. EPA. 2002 Equation 4-8		
Т	Exposure interval (years)	26 (used for mass- limit model)	U.S. EPA. 2002 Equation 4-13		
ρ <sub>b</sub>	Dry soil bulk density (g/cm <sup>3</sup> )	1.5	U.S. EPA. 2002 Equation 4-8		
$\theta_a$	Air-filled soil porosity (L <sub>air</sub> /L <sub>soil</sub> ) (n-θ <sub>w</sub> )	0.28	U.S. EPA. 2002 Equation 4-8		
n	Total soil porosity ( $L_{pore}/L_{soi}l$ ) (1-( $\rho_b/\rho_s$ )	0.43	U.S. EPA. 2002 Equation 4-8		
$\theta_{\rm w}$	Water-filled soil porosity (L <sub>water</sub> /L <sub>soil</sub> )	0.15	U.S. EPA. 2002 Equation 4-8		
ρ <sub>s</sub>	Soil particle density (g/cm <sup>3</sup> )	2.65	U.S. EPA. 2002 Equation 4-8		
S	Water Solubility Limit (mg/L)	Contaminant- specific	See Chemical-specific hierarchy		
R	Universal Gas Constant (L-atm/mole-K)	0.082057	U.S. EPAFact Sheet		
R <sub>c</sub>	Universal Gas Constant (cal/mole-K)	1.9872	U.S. EPA Fact Sheet		
D <sub>ia</sub>	Diffusivity in air (cm <sup>2</sup> /s)	Contaminant- specific	U.S. EPA. 2001		
Н'	Dimensionless Henry's Law Constant	Contaminant- specific	See Chemical-specific hierarchy		
D <sub>iw</sub>	Diffusivity in water (cm <sup>2</sup> /s)	Contaminant- specific	U.S. EPA. 2001		
-----------------------	--	-------------------------------------	------------------------------------	--	--
K <sub>d</sub>	Soil-water partition coefficient (L/Kg) (K <sub>oc</sub> ×f <sub>oc</sub> )	Contaminant- specific	U.S. EPA. 2002 Equation 4-8		
K <sub>oc</sub>	Soil organic carbon-water partition coefficient (L/Kg)	$= K_{oc} * f_{oc}$ for organics	See Chemical-specific hierarchy		
f <sub>oc</sub>	Organic carbon content of soil (g/g)	0.006	U.S. EPA. 2002 Equation 4-8		
ds	Average source depth (m)	Site-specific	U.S. EPA 2002 Equation 4-13		
Subch	Subchronic Volatilization Factor for Unlimited Source and Mass-limit Equations				
VF <sub>ulim-sc</sub>	Subchronic Volatilization Factor (m <sup>3</sup> /kg)	Contaminant- specific	U.S. EPA 2002 Equation 5-14		
Q/C <sub>sa</sub>	Inverse of the ratio of the 1-h geometric mean air concentration to the volatilization flux at the center of a square source $(g/m^2$ -s per kg/m <sup>3</sup> )	14.31 (for 0.5 acre site)	U.S. EPA 2002 Equation 5-14		
А	Dispersion constant unitless	2.4538	U.S. EPA 2002 Equation 5-15		
Ac	Areal extent of the site soil contamination (acres)	0.5 (range 0.5 to 500)	U.S. EPA 2002 Equation 5-15		
В	Dispersion constant unitless	17.5660	U.S. EPA 2002 Equation 5-15		
С	Dispersion constant unitless	189.0426	U.S. EPA 2002 Equation 5-15		

D <sub>A</sub>	Apparent Diffusivity (cm <sup>2</sup> /s)	Contaminant- specific	U.S. EPA 2002 Equation 5-14
Т	Total time over which construction occurs (s)	site-specific	U.S. EPA 2002 Equation 5-14
ρ <sub>b</sub>	Dry soil bulk density (g/cm <sup>3</sup> )	1.5	U.S. EPA 2002 Equation 5-14
F <sub>D</sub>	Dispersion correction factor (unitless)	0.185	U.S. EPA 2002 Equation 5-14
$\theta_a$	Air-filled soil porosity $(L_{air}/L_{soil}) (n-\theta_w)$	0.28	U.S. EPA 2002 Equation 5-14
n	Total soil porosity ( $L_{pore}/L_{soil}$ ) (1-( $\rho_b/\rho_s$ )	0.43	U.S. EPA 2002 Equation 5-14
$\theta_{\rm w}$	Water-filled soil porosity (L <sub>water</sub> /L <sub>soil</sub> )	0.15	U.S. EPA 2002 Equation 5-14
ρ <sub>s</sub>	Soil particle density (g/cm <sup>3</sup> )	2.65	U.S. EPA 2002 Equation 5-14
D <sub>ia</sub>	Diffusivity in air (cm <sup>2</sup> /s)	Contaminant- specific	U.S. EPA 2001
Η'	Dimensionless Henry's Law Constant	Contaminant- specific	See Chemical-specific hierarchy
D <sub>iw</sub>	Diffusivity in water (cm <sup>2</sup> /s)	Contaminant- specific	U.S. EPA 2001
K <sub>d</sub>	Soil-water partition coefficient (L/Kg) $(K_{oc} \times f_{oc})$	$= K_{oc} * f_{oc}$ for organics	See Chemical-specific hierarchy
K <sub>oc</sub>	Soil organic carbon-water partition coefficient (L/Kg)	Contaminant- specific	See Chemical-specific hierarchy

f <sub>oc</sub>	Organic carbon content of soil (g/g)	0.006 (0.6%)	U.S. EPA 2002 Equation 5-14
Т	Total time over which construction occurs (year)	site-specific (T=ED)	U.S. EPA 2002 Equation 5-17
ds	Average source depth (m)	Site-specific	U.S. EPA 2002 Equation 5-17

## 7. References

U.S. EPA (Environmental Protection Agency). 1987. Processes, Coefficients, and Models for Simulation Toxic Organics and Heavy Metals in Surface Waters. EPA/600/3-87/015. Office of Research and Development, Athens, GA.

U.S. EPA 1989. <u>Risk assessment guidance for Superfund. Volume I: Human health evaluation manual (Part A)(PDF)</u> (289 pp, 6.9 MB). Interim Final. Office of Emergency and Remedial Response. EPA/540/1-89/002.

U.S. EPA 1991a. Human health evaluation manual, supplemental guidance: "<u>Standard default exposure</u> factors (PDF) (28 pp, 248 K)". OSWER Directive 9285.6-03.

U.S. EPA 1991b. <u>Risk Assessment Guidance for Superfund</u>, Volume I: <u>Human Health Evaluation Manual</u> (Part B, Development of Risk-Based Preliminary Remediation Goals) (PDF) (68 pp, 721 K). Office of Emergency and Remedial Response. EPA/540/R-92/003. December 1991

U.S. EPA. 1996a. Soil Screening Guidance: User's Guide. Office of Emergency and Remedial Response. Washington, DC. OSWER No. 9355.4-23

U.S. EPA. 1996b. Soil Screening Guidance: Technical Background Document. Office of Emergency and Remedial Response. Washington, DC. OSWER No. 9355.4-17A

U.S. EPA. 1997a. <u>Exposure Factors Handbook (PDF)</u> (854 pp, 7.7 MB). Office of Research and Development, Washington, DC. EPA/600/P-95/002Fa.

U.S. EPA 2000. Exposure and Human Health Reassessment of 2,3,7,8-Tetrachlorodibenzo-p-Dioxin (TCDD) and Related Compounds. Part I: Estimating Exposure to Dioxin-Like Compounds. Volume 3--Properties, Environmental Levels, and Background Exposures. Draft Final Report. EPA/600/P- 00/001. Office of Research and Development, Washington, DC. September.

U.S. EPA, 2000b. <u>Methodology for Deriving Ambient Water Quality Criteria for the Protection of Human</u> <u>Health (PDF)</u> (185 pp, 790 K).

U.S. EPA, 2001 (PDF) (38 pp, 185 K). WATER9. Version 1.0.0. Office of Air Quality Planning and Standards, Research Triangle Park, NC. Web site.

U.S. EPA 2002. Supplemental Guidance for Developing Soil Screening Levels for Superfund Sites. OSWER 9355.4-24. December 2002.

U.S. EPA 2004. Risk Assessment Guidance for Superfund Volume I: Human Health Evaluation Manual (Part E, Supplemental Guidance for Dermal Risk Assessment) Final. OSWER 9285.7-02EP.July 2004. <u>Document (PDF)</u> (186 pp, 4.2 MB) and <u>website</u>.

U.S. EPA 2005. Supplemental Guidance for Assessing Susceptibility from Early-Life Exposure to Carcinogens. March 2005. <u>Document (PDF)</u> (166 pp, 468 K)

U.S. EPA 2009. Risk Assessment Guidance for Superfund Volume I: Human Health Evaluation Manual (Part F, Supplemental Guidance for Inhalation Risk Assessment) Final. OSWER 9285.7-82. January 2009. Document (PDF) (68 pp, 724 K), memo (PDF) (3 pp, 487 K) and website.

U.S. EPA 2011. Exposure Factors Handbook 2011 Edition (Final). National Center for Environmental Assessment, Office of Research and Development. Washington D.C. Currently available online at <u>http://cfpub.epa.gov/ncea/risk/recordisplay.cfm?deid=236252</u> or download <u>here (PDF)</u> (1436 pp, 21.3 MB).

U.S. EPA 2012. OSWER Directive 9200.1-113: Compilation and Review of Data on Relative Bioavailability of Arsenic in Soil. Washington, DC.

U.S. EPA 2014. OSWER Directive 9200.1-120 (PDF) (7 pp, 1.1 MB). Washington, DC.

For assistance/questions please use the <u>Regional Screening Levels (RSLs) contact us</u> page. For general risk assessment questions, separate from the RSLs, please use the link below.

LAST UPDATED ON NOVEMBER 19, 2018

## Appendix H

**Groundwater Conceptual Site Models – Exposure Routes** 

### Groundwater Conceptual Site Model – PBG Plume Badger Army Ammunition Plant



LEGEND:

• Pathway potentially complete under current land use conditions and warrants further evaluation.

• Pathway incomplete or considered insignificant under current land use conditions but potentially complete under hypothetical future onsite groundwater usage.

### Groundwater Conceptual Site Model – DBG Plume Badger Army Ammunition Plant



LEGEND:

• Pathway potentially complete under current land use conditions and warrants further evaluation.

• Pathway incomplete or considered insignificant under current land use conditions but potentially complete under hypothetical future onsite groundwater usage.

### Groundwater Conceptual Site Model – Central Plume Badger Army Ammunition Plant



LEGEND:

• Pathway potentially complete under current land use conditions and warrants further evaluation.

• Pathway incomplete or considered insignificant under current land use conditions but potentially complete under hypothetical future onsite groundwater usage.

### Groundwater Conceptual Site Model – NC Area Plume Badger Army Ammunition Plant



LEGEND:

• Pathway potentially complete under current land use conditions and warrants further evaluation.

• Pathway incomplete or considered insignificant under current land use conditions but potentially complete under hypothetical future onsite groundwater usage.

# Appendix I

**Remedial Alternative Cost Summaries** 

#### PBG Plume Alternative 2 Cost Summary Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Alternative	Item Description	Comments	Sub Totals	Total Costs
	Direct Capital Cost	No design cost to implement alternative	\$-	\$0
n and	Indirect Capital Cost	Engineering Design (10% of Direct Capital Cost)	\$-	\$0
nuatio upply		Legal/License/Permit (5% of Direct Capital Cost)	\$ -	
Monitored Natural Atter Alternate Water S		Start-up & Shake-down (10% of Direct Capital Cost)	\$-	
		Contingency (20% of Direct Capital Cost)	\$ -	
	Annual O&M	30 years of groundwater monitoring at current groundwater sampling program	\$ 4,511,746	
		Provision of bottled water - 20 gallons/month for 3 months, once every 3 years for 30 years	\$ 1,367	\$4,913,113
		Residential well replacement - 1 well every 3 years for 30 years	\$ 400,000	
				\$4,913,113

Notes:

Costs are based on current monitoring plans and engineering estimates.

O&M - Operations and Maintenance

#### PBG Plume Alternative 3 Cost Summary Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Alternative	Item Description	Comments	Sub Totals	Total Costs
reat		Four groundwater extraction wells	\$ 520,573	\$3,633,573
	Direct Capital Cost	Four mobile treatment units	\$ 2,460,000	
p & T		Treatment area preparation, piping, utilities, SCADA System, buoy system, restoration	\$ 653,000	
- Pum		Engineering Design (10% of Direct Capital Cost)	\$ 363,357	
lation	Indirect Capital Cost	Legal/License/Permit (5% of Direct Capital Cost)	\$ 181,679	\$1,635,108
Active Groundwater Remedi		Start-up & Shake-down (10% of Direct Capital Cost)	\$ 363,357	
		Contingency (20% of Direct Capital Cost)	\$ 726,715	
	Annual O&M	30 years of groundwater monitoring at current groundwater sampling program	\$ 4,511,746	
		Provision of bottled water - 20 gallons/month for 3 months, once every 3 years for 30 years	\$ 1,367	\$7,433,131
		Residential well replacement - 1 well every 3 years for 30 years	\$ 400,000	
		Mobile treatment units - 8 years of O&M (2 for 8 years, 1 for 6 years, 1 for 2 years)	\$ 2,520,018	
				\$12,701,812

Notes:

Costs are based on current monitoring plans and engineering estimates.

O&M - Operations and Maintenance

### PBG Plume Alternative 4 Cost Summary Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Alternative	Item Description	Comments	Sub Totals	Total Costs
- -		Biochemical product (59 tankers)	\$ 2,754,328	¢2.254.720
	Direct Capital Cost	Drilling, well installation & development, direct push, injection, abandonment & decontamination (9 wells, 150 DPTs)	\$ 500,402	\$3,234,729
diatio		Engineering Design (10% of Direct Capital Cost)	\$ 325,473	
Active Groundwater Reme Anaerobic Bioremedia	Indirect Capital Cost	Legal/License/Permit (5% of Direct Capital Cost)	\$ 162,736	\$1,464,628
		Start-up & Shake-down (10% of Direct Capital Cost)	\$ 325,473	
		Contingency (20% of Direct Capital Cost)	\$ 650,946	
	Annual O&M	30 years of groundwater monitoring at current groundwater sampling program	\$ 4,511,746	\$4,913,113
		Provision of bottled water - 20 gallons/month for 3 months, once every 3 years for 30 years	\$ 1,367	
		Residential well replacement - 1 well every 3 years for 30 years	\$ 400,000	
				\$9,632,470

Notes:

Costs are based on current monitoring plans and engineering estimates.

O&M - Operations and Maintenance

### PBG Plume Alternative 5 Cost Summary Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Alternative	Item Description	Comments	Sub Totals	Total Costs
Replacement - Plume Area	Direct Capital Cost	Replacement of 47 residential wells	\$ 2,350,000	\$2,350,000
	Indirect Capital Cost	Engineering Design (10% of Direct Capital Cost)	\$ 235,000	
		Legal/License/Permit (5% of Direct Capital Cost)	\$ 117,500	\$1,057,500
		Start-up & Shake-down (10% of Direct Capital Cost)	\$ 235,000	
		Contingency (20% of Direct Capital Cost)	\$ 470,000	
Well	Annual O&M	30 years of groundwater monitoring at current groundwater sampling program	\$ 4,511,746	\$4,511,746
				\$7,919,246

Notes:

Costs are based on current monitoring plans and engineering estimates. O&M - Operations and Maintenance

#### PBG Plume Alternative 6 Cost Summary Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Alternative	Item Description	Comments	Sub Totals	Total Costs
nt	Direct Capital Cost	Biochemical product (2 tankers)	\$ 83,150	\$201,433
		Drilling, well installation & development, injection & decontamination (9 locations)	\$ 118,283	
		Engineering Design (10% of Direct Capital Cost)	\$ 20,143	
reatmo	Indirect Capital Cost	Legal/License/Permit (5% of Direct Capital Cost)	\$ 10,072	\$90,645
Source Area Tr		Start-up & Shake-down (10% of Direct Capital Cost)	\$ 20,143	
		Contingency (20% of Direct Capital Cost)	\$ 40,287	
	Annual O&M	30 years of groundwater monitoring at current groundwater sampling program	\$ 4,511,746	
		Provision of bottled water - 20 gallons/month for 3 months, once every 3 years for 30 years	\$ 1,367	
		Residential well replacement - 1 well every 3 years for 30 years	\$ 400,000	
				\$5,205,190

Notes:

Costs are based on current monitoring plans and engineering estimates.

O&M - Operations and Maintenance

#### DBG Plume Alternative 2 Cost Summary Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Alternative	Item Description	Comments	Sub Totals	Total Costs
	Direct Capital Cost	No design cost to implement alternative	\$-	\$0
n and	Indirect Capital Cost	Engineering Design (10% of Direct Capital Cost)	\$ -	\$0
nuatio upply		Legal/License/Permit (5% of Direct Capital Cost)	\$-	
Monitored Natural Atter Alternate Water Si		Start-up & Shake-down (10% of Direct Capital Cost)	\$ -	
		Contingency (20% of Direct Capital Cost)	\$ -	
	Annual O&M	30 years of groundwater monitoring at current groundwater sampling program	\$ 3,839,123	
		Provision of bottled water - 20 gallons/month for 3 months, once every 3 years for 30 years	\$ 1,367	\$4,240,490
1		Residential well replacement - 1 well every 3 years for 30 years	\$ 400,000	
				\$4,240,490

Notes:

Costs are based on current monitoring plans and engineering estimates.

O&M - Operations and Maintenance

#### DBG Plume Alternative 3 Cost Summary Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Alternative	Item Description	Comments	Sub Totals	Total Costs
reat		Three groundwater extraction wells	\$ 390,430	\$2,776,030
	Direct Capital Cost	Three mobile treatment units	\$ 1,845,000	
p & T		Treatment area preparation, piping, utilities, SCADA System, buoy system, restoration	\$ 540,600	
- Pum		Engineering Design (10% of Direct Capital Cost)	\$ 277,603	
iation	Indirect Capital Cost	Legal/License/Permit (5% of Direct Capital Cost)	\$ 138,802	\$1,249,214
Active Groundwater Remedi		Start-up & Shake-down (10% of Direct Capital Cost)	\$ 277,603	
		Contingency (20% of Direct Capital Cost)	\$ 555,206	
	Annual O&M	24 years of groundwater monitoring at current groundwater sampling program	\$ 3,199,269	
		Provision of bottled water - 20 gallons/month for 3 months, once every 3 years for 24 years	\$ 1,184	\$8,522,395
		Residential well replacement - 1 well every 3 years for 24 years	\$ 333,333	
		Mobile treatment units - 22 years of O&M (2 for 22 years, 1 for 10 years)	\$ 4,988,608	
				\$12,547,639

Notes:

Costs are based on current monitoring plans and engineering estimates.

O&M - Operations and Maintenance

### DBG Plume Alternative 4 Cost Summary Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Alternative	Item Description	Comments	Sub Totals	Total Costs
		Biochemical product (149 tankers)	\$ 7,033,063	¢0 107 0C0
	Direct Capital Cost	Direct push, injection, hole abandonment & decontamination (406 locations)	\$ 1,074,805	\$8,107,808
diatio		Engineering Design (10% of Direct Capital Cost)	\$ 810,787	
Active Groundwater Reme Anaerobic Bioremedia	Indirect Capital Cost	Legal/License/Permit (5% of Direct Capital Cost)	\$ 405,393	\$3,648,540
		Start-up & Shake-down (10% of Direct Capital Cost)	\$ 810,787	
		Contingency (20% of Direct Capital Cost)	\$ 1,621,574	
	Annual O&M	4 years of groundwater monitoring at current groundwater sampling program	\$ 639,854	
		Provision of bottled water - 20 gallons/month for 3 months, once every 3 years for 4 years	\$ 228	
		Residential well replacement - 1 well every 3 years for 4 years	\$ 66,667	
				\$12,463,156

Notes:

Costs are based on current monitoring plans and engineering estimates.

O&M - Operations and Maintenance

### DBG Plume Alternative 5 Cost Summary Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Alternative	Item Description	Comments	Sub Totals	Total Costs
rrea	Direct Capital Cost	Replacement of 57 residential wells	\$ 2,280,000	\$2,280,000
ume /		Engineering Design (10% of Direct Capital Cost)	\$ 228,000	
l Replacement - Pl	Indirect Capital Cost	Legal/License/Permit (5% of Direct Capital Cost)	\$ 114,000	\$1,026,000
		Start-up & Shake-down (10% of Direct Capital Cost)	\$ 228,000	
		Contingency (20% of Direct Capital Cost)	\$ 456,000	
Well	Annual O&M	30 years of groundwater monitoring at current groundwater sampling program	\$ 3,839,123	\$3,839,123
				\$7,145,123

Notes:

Costs are based on current monitoring plans and engineering estimates. O&M - Operations and Maintenance

#### DBG Plume Alternative 6 Cost Summary Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Alternative	Item Description	Comments	Su	ub Totals	Total Costs
	Direct Capital Cost	Biochemical product (11 tankers)	\$	517,375	\$645,631
		Direct push, injection, hole abandonment & decontamination (56 locations)	\$	128,256	
ent	Indirect Capital Cost	Engineering Design (10% of Direct Capital Cost)	\$	64,563	\$290,534
reatm		Legal/License/Permit (5% of Direct Capital Cost)	\$	32,282	
Source Area Tı		Start-up & Shake-down (10% of Direct Capital Cost)	\$	64,563	
		Contingency (20% of Direct Capital Cost)	\$	129,126	
	Annual O&M	30 years of groundwater monitoring at current groundwater sampling program	\$	3,839,123	
		Provision of bottled water - 20 gallons/month for 3 months, once every 3 years for 30 years	\$	1,367	\$4,240,490
		Residential well replacement - 1 well every 3 years for 30 years	\$	400,000	
					\$5,176,654

Notes:

Costs are based on current monitoring plans and engineering estimates.

O&M - Operations and Maintenance

#### Central Plume Alternative 2 Cost Summary Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Alternative	Item Description	Comments	Sub Totals	Total Costs
Monitored Natural Attenuation and Alternate Water Supply	Direct Capital Cost	No design cost to implement alternative	\$ -	\$0
	Indirect Capital Cost	Engineering Design (10% of Direct Capital Cost)	\$-	\$0
		Legal/License/Permit (5% of Direct Capital Cost)	\$ -	
		Start-up & Shake-down (10% of Direct Capital Cost)	\$ -	
		Contingency (20% of Direct Capital Cost)	\$ -	
	Annual O&M	30 years of groundwater monitoring at current groundwater sampling program	\$ 1,997,172	
		Provision of bottled water - 20 gallons/month for 3 months, once every 3 years for 30 years	\$ 1,367	\$2,398,538
		Residential well replacement - 1 well every 3 years for 30 years	\$ 400,000	
				\$2,398,538

Notes:

Costs are based on current monitoring plans and engineering estimates.

O&M - Operations and Maintenance

### Central Plume Alternative 3 Cost Summary Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Alternative	Item Description	Comments	Sub Totals	Total Costs
p & Treat	Direct Capital Cost	Eight groundwater extraction wells	\$ 1,041,147	
		Eight mobile treatment units	\$ 4,920,000	\$6,939,247
		Treatment area preparation, piping, utilities, SCADA System, buoy system, restoration	\$ 978,100	
- Pum	Indirect Capital Cost	Engineering Design (10% of Direct Capital Cost)	\$ 693,925	
ation		Legal/License/Permit (5% of Direct Capital Cost)	\$ 346,962	\$3,122,661
Active Groundwater Remedi		Start-up & Shake-down (10% of Direct Capital Cost)	\$ 693,925	
		Contingency (20% of Direct Capital Cost)	\$ 1,387,849	
	Annual O&M	12 years of groundwater monitoring at current groundwater sampling program	\$ 865,441	\$7,953,709
		Provision of bottled water - 20 gallons/month for 3 months, once every 3 years for 12 years	\$ 592	
		Residential well replacement - 1 well every 3 years for 12 years	\$ 173,333	
		Mobile treatment units - 10 years of O&M	\$ 6,914,343	
				\$18,015,617

Notes:

Costs are based on current monitoring plans and engineering estimates.

O&M - Operations and Maintenance

#### Central Plume Alternative 4 Cost Summary Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Alternative	Item Description	Comments	Sub Totals	Total Costs
Active Groundwater Remediation - Anaerobic Bioremediation	Direct Capital Cost	Biochemical product (291 tankers)	\$ 13,627,063	\$16,082,742
		Direct push, injection, hole abandonment & decontamination (988 locations)	\$ 2,455,680	
	Indirect Capital Cost	Engineering Design (10% of Direct Capital Cost)	\$ 1,608,274	\$7,237,234
		Legal/License/Permit (5% of Direct Capital Cost)	\$ 804,137	
		Start-up & Shake-down (10% of Direct Capital Cost)	\$ 1,608,274	
		Contingency (20% of Direct Capital Cost)	\$ 3,216,548	
	Annual O&M	4 years of groundwater monitoring at current sampling program	\$ 332,862	
		Provision of bottled water - 20 gallons/month for 3 months, once every 3 years for 4 years	\$ 228	\$399,756
		Residential well replacement - 1 well every 3 years for 4 years	\$ 66,667	
				\$23,719,733

Notes:

Costs are based on current monitoring plans and engineering estimates.

O&M - Operations and Maintenance

### Central Plume Alternative 5 Cost Summary Remedial Investigation/Feasibility Study Badger Army Ammunition Plant

Alternative	Item Description	Comments	Sub Totals	Total Costs
Well Replacement - Plume Area	Direct Capital Cost	Replacement of 23 residential wells	\$ 920,000	\$920,000
	Indirect Capital Cost	Engineering Design (10% of Direct Capital Cost)	\$ 92,000	\$414,000
		Legal/License/Permit (5% of Direct Capital Cost)	\$ 46,000	
		Start-up & Shake-down (10% of Direct Capital Cost)	\$ 92,000	
		Contingency (20% of Direct Capital Cost)	\$ 184,000	
	Annual O&M	30 years of groundwater monitoring at current groundwater sampling program	\$ 1,997,172	\$1,997,172
				\$3,331,172

Notes:

Costs are based on current monitoring plans and engineering estimates. O&M - Operations and Maintenance

# Appendix J

**Remedial Alternative Treatment Area Drawings** 









Coordinate System: NAD 1983 StatePlane Wisconsin South FIPS 4803 Feet













Coordinate System: NAD 1983 StatePlane Wisconsin South FIPS 4803 Feet



















